

# Parish Annex Building Renovations (Phase I) (M.A. Project No. H1-22029-DA)

Beauregard Parish Police Jury

Project documents obtained from www.CentralBidding.com 06-Apr-2023 04:33:02 PM



# Meyer & Associates, Inc. Consulting Engineers

Vernon F. Meyer, P.E. President Richard T. Meyer, P.E. Vice President

#### Parish Annex Building Renovations (Phase I)

Beauregard Parish Police Jury Beauregard Parish, Louisiana

MA Project No. H1-22029-DA

#### Administrator Bryan McReynolds

#### **Police Jury**

Wayne Reeves — District #1
Jeffrey Meadows — District #2
Shanel Handy — District #3A
Eddie Ware — District #3B
Chuck Montgomery — District #3C
Mike Harper — District #3D
Jerry Shirley — District #3E
John Stebbins — District #4A
Ronnie Jackson — District #4B
Kelly Bailey — District #5

|    |           |      | BYRON D. RACCA License No. 33694 PROFESSIONAL ENGINEER IN  CANAL ENGIN |
|----|-----------|------|--|
| Ву | Revisions | Date | <i>3/31/23</i>   |

#### **PROJECT MANUAL & CONTRACT DOCUMENTS**

Beauregard Parish Police Jury
Parish Annex Building
Renovations (Phase I)
M.A. Project No. H1-22029-DA

#### **TABLE OF CONTENTS**

|   | Page Number |
|---|-------------|
| Cover   |             |
| Table of Contents   |             |
| BID PHASE DOCUMENTS   | 006-029     |
| Advertisement for Bids  |             |
| Instructions to Bidders for Construction Contract   |             |
| Louisiana Uniform Public Work Bid Form  |             |
| Bid Bond  |             |
| FORMS TO BE SUBMITTED WITHIN 10 DAYS OF THE BID OPENING   | 030-043     |
| Affidavits  |             |
| Attestations Affidavit: LA. R.S. 38:2227 Past Criminal Convictions of Bidders, LA. R.S. 38:2212.10 Verification of Employees, & LA. R.S. 23:1726(B) Certification Regarding Unpaid Workers Compensation Insurance |             |
| Affidavit for Attesting that Public Contract Was Not Secured Through the<br>Employment or Payment of a Solicitor  |             |
| Qualifications Statement  |             |
| POST BID CONTRACT RELATED DOCUMENTS   | 044-081     |
| Designation of Construction Contractor as Agent of a Governmental Entity and Exemption Certificate  |             |
| Notice of Award   |             |
| Agreement Between Owner and Contractor for Construction Contract (Stipulated Price)   |             |
| Performance Bond  |             |
| Payment Bond  |             |
| Notice to Proceed   |             |
| Work Change Directive   |             |

| Change Order       |  |
|--------------------|--|
| Field Order        |  |
|                    | tantial Completion                                 |
| Notice of Acceptab | •  |
| Pay Application    | Sility of Work                                     |
| Рау Арріісаціон    |  |
| GENERAL CONDIT     | <u>IONS</u>  |
| Standard General   | Conditions of the Construction Contract            |
|                    |  |
| SUPPLEMENTARY      | CONDITIONS 162-181                                 |
| Supplementary Co   | nditions of the Construction Contract              |
| TECHNICAL PROVI    | <u>SIONS</u>                                       |
| Section H-100 -    |  |
| Section 202 -      | Removing or Relocating Structures and Obstructions |
| Section 203 -      | Excavation and Embankment                          |
| Section 302 -      | Class II Base Course                               |
| Section 502 -      | Asphalt Concrete Mixtures                          |
| Section 706 -      | Concrete Walks, Drives and Incidental Paving       |
| Section 712 -      | Revetments   |
| Section 713 -      | Temporary Traffic Control                          |
| Section 727 -      | Mobilization                                       |
| Section 740 -      | Construction Layout                                |
|                    |  |
| DIVISION 1 -       | GENERAL REQUIREMENTS                               |
| Section 01 10 00   | Summary of Work                                    |
| Section 01 23 00   | Alternates   |
| Section 01 26 00   | Contract Modification Procedure                    |
| Section 01 29 00   | Payment Procedures                                 |
| Section 01 31 00   | Project Management and Coordination                |
| Section 01 32 00   | Construction Progress Documentation                |
| Section 01 32 33   | Photographic Documentation                         |

**Submittal Procedures** 

**Quality Requirements** 

Section 01 33 00

Section 01 40 00

#### Page Number

| Temporary Facilities and Controls             |
|---|
| Product Requirements                          |
| Execution                                     |
| Closeout Procedures                           |
| Operation and Maintenance Data                |
| Project Record Documents                      |
| Demonstration and Training                    |
|   |
| EXISTING CONDITIONS                           |
| Selective Demolition                          |
|   |
| CONCRETE                                      |
| Cast-In-Place Concrete                        |
|   |
| MASONRY                                       |
| 3 Brick Masonry Repair                        |
| Masonry Veneer                                |
|   |
| METALS  |
| Cold-Formed Metal Framing                     |
|   |
| WOOD, PLASTICS & COMPOSITES                   |
| Rough Carpentry                               |
| Interior Finish Carpentry                     |
| Plastic-Laminate-Faced Architectural Cabinets |
|   |
| THERMAL AND MOISTURE PROTECTION               |
| Thermal & Acoustical Insulation               |
| Fluid Applied Air Barrier                     |
| Sheet Metal Flashing and Trim                 |
| Joint Sealants                                |
|   |

#### Page Number

| DIVISION 8 -     | OPENINGS                                  |
|------------------|---|
| Section 08 11 13 | Hollow Metal Doors and Frames             |
| Section 08 14 16 | Flush Wood Doors                          |
| Section 08 41 13 | Aluminum-Framed Entrances and Storefronts |
| Section 08 71 00 | Door Hardware                             |
| Section 08 80 00 | Glazing                                   |
|                  |   |
| DIVISION 9 -     | FINISHES                                  |
| Section 09 22 16 | Non-Structural Metal Framing              |
| Section 09 29 00 | Gypsum Board                              |
| Section 09 30 00 | Tiling                                    |
| Section 09 51 13 | Acoustical Panel Ceiling                  |
| Section 09 65 13 | Resilient Base and Accessories            |
| Section 09 65 19 | Resilient Tile Flooring                   |
| Section 09 91 00 | Painting                                  |
|                  |   |
| DIVISION 10 -    | SPECIALTIES                               |
| Section 10 14 67 | Signage                                   |
| Section 10 28 00 | Toilet, Bath and Laundry Accessories      |
| Section 10 44 13 | Fire Protection Cabinets                  |
| Section 10 44 16 | Fire Extinguishers                        |

**BID PHASE DOCUMENTS** 

This page intentionally left blank

#### **ADVERTISEMENT FOR BIDS**

#### **Issued By**

Beauregard Parish Police Jury 202 W. 2<sup>nd</sup> Street DeRidder, LA 70634

#### **General Notice**

Separate sealed Bids for Parish Annex Building Renovations (Phase I), Beauregard Parish Police Jury; M.A. Project No. H1-22029-DA will be received by the Police Jury, at the Beauregard Parish Police Jury, 202 W. 2<sup>nd</sup> Street, DeRidder, LA 70634 until 10:00 AM Central Time Zone, on April 27, 2023, and then at said time and at said office publicly opened and read aloud. Work consists of removal and disposal of debris in laterals and drainage ways..

#### **Work Classification**

Work Classification: Municipal and Public Works Construction; Highway, Street and Bridge

Construction; Building Construction

#### **Obtaining the Bidding Documents**

Electronic copies of the Bidding Documents may be obtained from the Issuing Office of Meyer & Associates, Inc. (337) 625-8353, located at 600 N. Cities Service Hwy., Sulphur, LA 70663. A Bidding Document deposit is not required. In order to submit a bid, Bidders must obtain an original set of electronic Bidding Documents from Meyer & Associates, Inc. or the approved electronic bid website defined herein said advertisement.

Access to electronic bidding is available through the Bidding Documents Website <a href="https://www.centralbidding.com">https://www.centralbidding.com</a>.

#### **Run Dates**

Friday, March 31, 2023 Friday, April 14, 2023 Friday, April 21, 2023

/s/ Bryan McReynolds, Administrator

This page intentionally left blank

#### INSTRUCTIONS TO BIDDERS FOR CONSTRUCTION CONTRACT

#### **Prepared By**













#### Copyright<sup>©</sup> 2018

National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882

www.nspe.org

American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474

www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723

www.asce.org

The copyright for this EJCDC document is owned jointly by the three sponsoring organizations listed above. The National Society of Professional Engineers is the Copyright Administrator for the EJCDC documents; please direct all inquiries regarding EJCDC copyrights to NSPE.

NOTE: EJCDC publications may be purchased at <a href="www.ejcdc.org">www.ejcdc.org</a>, or from any of the sponsoring organizations above.

#### **INSTRUCTIONS TO BIDDERS FOR CONSTRUCTION CONTRACT**

#### **TABLE OF CONTENTS**

|   | Page |
|---|------|
| Article 1— Defined Terms  | 1    |
| Article 2— Bidding Documents  | 1    |
| Article 3— Qualifications of Bidders  | 2    |
| Article 4— Pre-Bid Conference   | 2    |
| Article 5— Site and Other Areas; Existing Site Conditions; Examination of S |      |
| Article 6— Bidder's Representations and Certifications                      | 4    |
| Article 7— Interpretations and Addenda                                      | 5    |
| Article 8— Bid Security   | 5    |
| Article 9— Contract Times   | 6    |
| Article 10— Substitute and "Or Equal" Items                                 | 6    |
| Article 11— Subcontractors, Suppliers, and Others                           | 6    |
| Article 12— Preparation of Bid  | 7    |
| Article 13— Basis of Bid  | 7    |
| Article 14— Submittal of Bid  | 8    |
| Article 15— Modification and Withdrawal of Bid                              | 8    |
| Article 16— Opening of Bids   | 8    |
| Article 17— Bids to Remain Subject to Acceptance                            | 9    |
| Article 18— Evaluation of Bids and Award of Contract                        | 9    |
| Article 19— Bonds and Insurance   | 9    |
| Article 20— Signing of Agreement  | 10   |
| Article 21— Sales and Use Taxes   | 10   |

| This page intentionally left blank |
|------------------------------------|
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |
|                                    |

#### **ARTICLE 1—DEFINED TERMS**

- 1.01 Terms used in these Instructions to Bidders have the meanings indicated in the General Conditions and Supplementary Conditions. Additional terms used in these Instructions to Bidders have the meanings indicated below:
  - A. *Issuing Office*—The office from which the Bidding Documents are to be issued, and which registers plan holders.
  - B. Bidding Documents Website The website identified in the Advertisement for Bids through which a Bidder may submit a Bid electronically. If a Bidding Document Website is not identified in the Advertisement for Bids, electronic submission of Bids will not be permitted.

#### **ARTICLE 2—BIDDING DOCUMENTS**

- 2.01 Bidder shall obtain a complete set of Bidding Requirements and proposed Contract Documents (together, the Bidding Documents). See the Agreement for a list of the Contract Documents. It is Bidder's responsibility to determine that it is using a complete set of documents in the preparation of a Bid. Bidder assumes sole responsibility for errors or misinterpretations resulting from the use of incomplete documents, by Bidder itself or by its prospective Subcontractors and Suppliers.
- 2.02 Bidding Documents are made available for the sole purpose of obtaining Bids for completion of the Project and permission to download or distribution of the Bidding Documents does not confer a license or grant permission or authorization for any other use. Authorization to download documents, or other distribution, includes the right for plan holders to print documents solely for their use, and the use of their prospective Subcontractors and Suppliers, provided the plan holder pays all costs associated with printing or reproduction. Printed documents may not be re-sold under any circumstances.
- 2.03 Bidder must obtain an original set of electronic or paper Bidding Documents from the Issuing Office or from Bidding Documents Website identified in the Advertisement for Bids. Bidders may rely that sets of Bidding Documents obtained from the Issuing Office or Bidding Documents Website are complete, unless an omission is blatant. Registered plan holders will receive Addenda issued by the Issuing Office or the Bidding Documents Website as applicable.
- 2.04 Plan rooms (including construction information subscription services, and electronic and virtual plan rooms) may distribute the Bidding Documents, or make them available for examination. Owner is not responsible for omissions in Bidding Documents or other documents obtained from plan rooms, or for a Bidder's failure to obtain Addenda from a plan room. In order to submit a Bid, Bidder must obtain an original set of electronic or paper Bidding Documents in accordance with Paragraph 2.03.

#### 2.05 Electronic Documents

- A. When the Bidding Requirements indicate that electronic (digital) copies of the Bidding Documents are available, such documents will be made available to the Bidders as Electronic Documents in the manner specified.
  - Bidding Documents will be provided in Adobe PDF (Portable Document Format) (.pdf).
    It is the intent of the Engineer and Owner that such Electronic Documents are to be
    exactly representative of the paper copies of the documents. However, because the
    Owner and Engineer cannot totally control the transmission and receipt of Electronic

Documents nor the Contractor's means of reproduction of such documents, the Owner and Engineer cannot and do not guarantee that transmissions and/or reproductions of Electronic Documents are identical in every manner to the paper copies.

B. Unless otherwise stated in the Bidding Documents, the Bidder may use and rely upon complete sets of Electronic Documents of the Bidding Documents, described in Paragraph 2.05.A above. However, Bidder assumes all risks associated with differences arising from transmission/receipt of Electronic Documents versions of Bidding Documents and reproductions prepared from those versions and, further, assumes all risks, costs, and responsibility associated with use of the Electronic Documents versions to derive information that is not explicitly contained in printed paper versions of the documents, and for Bidder's reliance upon such derived information.

#### **ARTICLE 3—QUALIFICATIONS OF BIDDERS**

- 3.01 In accordance with LA RS 37:2163 Bidders must hold an active Louisiana State Contractor's License unless an exemption to the referenced statute applies.
- 3.02 Within 10 days of the Bid opening, the apparent low Bidder shall submit to the Engineer:
  - A. Attestation Affidavit as required by LA RS 38:2227 (Past Criminal Convictions of Bidders), LA RS 38:2212.10 (Verification of Employees), LA RS 23:1726(B) (Certification regarding Unpaid Workers Compensation Insurance)
  - B. LA RS 38:2224 Affidavit for Attesting That Public Contract Was Not Secured Through Employment or Payment of Solicitor
  - C. EJCDC C-451, Qualifications Statement
  - D. Any other forms listed in the Table of Contents as "Forms to be Submitted Within 10 Days of the Bid Opening".
- 3.03 To demonstrate Bidder's qualifications to perform the Work, after submitting its Bid and within 10 days of Owner's request, Bidder must submit the following information:
  - A. Written evidence establishing its qualifications such as financial data, previous experience, and present commitments.
  - B. Subcontractor and Supplier qualification information.
  - C. Other required information regarding qualifications.
- 3.04 No requirement in this Article 3 to submit information will prejudice the right of Owner to seek additional pertinent information regarding Bidder's qualifications.

#### **ARTICLE 4—PRE-BID CONFERENCE**

- 4.01 If a pre-bid conference will be held, the date, time, and location will be stated in the Advertisement for Bids, and representatives of Owner and Engineer will be present to discuss the Project.
- 4.02 Mandatory Pre-Bid Conference
  - A. If the pre-bid conference is stated as mandatory in the Advertisement for Bids, Bids will not be accepted from Bidders who do not attend the conference. It is each Bidder's responsibility

015

to sign in at the pre-bid conference to verify its participation. Bidders must sign in using the name of the organization that will be submitting a Bid. A list of qualified Bidders that attended the pre-bid conference and are eligible to submit a Bid for this Project will be issued in an Addendum.

4.03 Information presented at a pre-Bid conference does not alter the Contract Documents. Owner will issue Addenda to make any changes to the Contract Documents that result from discussions at the pre-Bid conference. Information presented, and statements made at the pre-bid conference will not be binding or legally effective unless incorporated in an Addendum.

### ARTICLE 5—SITE AND OTHER AREAS; EXISTING SITE CONDITIONS; EXAMINATION OF SITE; OWNER'S SAFETY PROGRAM; OTHER WORK AT THE SITE

#### 5.01 Site and Other Areas

A. The Site is identified in the Bidding Documents. By definition, the Site includes rights-of-way, easements, and other lands furnished by Owner for the use of the Contractor. Any additional lands required for temporary construction facilities, construction equipment, or storage of materials and equipment, and any access needed for such additional lands, are to be obtained and paid for by Contractor.

#### 5.02 Existing Site Conditions

- A. Subsurface and Physical Conditions; Hazardous Environmental Conditions
  - 1. The Supplementary Conditions identify the following regarding existing conditions at or adjacent to the Site:
    - a. Those reports of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data.
    - b. Those drawings known to Owner of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to the Site (except Underground Facilities), that contain Technical Data.
    - c. Reports and drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site.
    - d. Technical Data contained in such reports and drawings.
  - 2. Owner will make copies of reports and drawings referenced above available to any Bidder on request. These reports and drawings are not part of the Contract Documents, but the Technical Data contained therein upon whose accuracy Bidder is entitled to rely, as provided in the General Conditions, has been identified and established in the Supplementary Conditions. Bidder is responsible for any interpretation or conclusion Bidder draws from any Technical Data or any other data, interpretations, opinions, or information contained in such reports or shown or indicated in such drawings.
  - 3. If the Supplementary Conditions do not identify Technical Data, the default definition of Technical Data set forth in Article 1 of the General Conditions will apply.
- B. *Underground Facilities:* Underground Facilities are shown or indicated on the Drawings, pursuant to Paragraph 5.05 of the General Conditions, and not in the drawings referred to in

Paragraph 5.02.A of these Instructions to Bidders. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data.

#### 5.03 Other Site-related Documents

A. No other Site-related documents are available.

#### 5.04 Site Visit and Testing by Bidders

- A. Bidder is required to visit the Site and conduct a thorough visual examination of the Site and adjacent areas. During the visit the Bidder must not disturb any ongoing operations at the Site.
- B. Bidders visiting the Site are required to arrange their own transportation to the Site.
- C. All access to the Site other than during a regularly scheduled Site visit must be coordinated through the Owner for visiting the Site. Bidder must conduct the required Site visit during normal working hours.
- D. On request, and to the extent Owner has control over the Site, and schedule permitting, the Owner will provide Bidder general access to the Site to conduct such additional examinations, investigations, explorations, tests, and studies as Bidder deems necessary for preparing and submitting a successful Bid. Owner will not have any obligation to grant such access if doing so is not practical because of existing operations, security or safety concerns, or restraints on Owner's authority regarding the Site. Bidder is responsible for establishing access needed to reach specific selected test sites.
- E. Bidder must comply with all applicable Laws and Regulations regarding excavation and location of utilities, obtain all permits, and comply with all terms and conditions established by Owner or by property owners or other entities controlling the Site with respect to schedule, access, existing operations, security, liability insurance, and applicable safety programs.
- F. Bidder must fill all holes and clean up and restore the Site to its former condition upon completion of such explorations, investigations, tests, and studies.

#### 5.05 *Owner's Safety Program*

A. Site visits and work at the Site may be governed by an Owner safety program.

#### 5.06 Other Work at the Site

A. Reference is made to Article 8 of the Supplementary Conditions for the identification of the general nature of other work of which Owner is aware (if any) that is to be performed at the Site by Owner or others (such as utilities and other prime contractors) and relates to the Work contemplated by these Bidding Documents. If Owner is party to a written contract for such other work, then on request, Owner will provide to each Bidder access to examine such contracts (other than portions thereof related to price and other confidential matters), if any.

#### ARTICLE 6—BIDDER'S REPRESENTATIONS AND CERTIFICATIONS

- 6.01 Express Representations and Certifications in Bid Form, Agreement
  - A. The Bid Form that each Bidder will submit contains express representations regarding the Bidder's examination of Project documentation, Site visit, and preparation of the Bid. Bidder

- should review these representations and certifications, and assure that Bidder can make the representations and certifications in good faith, before executing and submitting its Bid.
- B. If Bidder is awarded the Contract, Bidder (as Contractor) will make similar express representations and certifications when it executes the Agreement.

#### ARTICLE 7—INTERPRETATIONS AND ADDENDA

- 7.01 Owner on its own initiative may issue Addenda to clarify, correct, supplement, or change the Bidding Documents.
- 7.02 Bidder shall submit all questions about the meaning or intent of the Bidding Documents to Engineer in writing.
- 7.03 Interpretations or clarifications considered necessary by Engineer in response to such questions will be issued by Addenda delivered to all registered plan holders. Questions received less than seven working days prior to the date for opening of Bids may not be answered. Bidder is responsible for confirming that any questions submitted are received by the Engineer.
- 7.04 Only responses set forth in an Addendum will be binding. Oral and other interpretations or clarifications will be without legal effect. Responses to questions are not part of the Contract Documents unless set forth in an Addendum that expressly modifies or supplements the Contract Documents.

#### ARTICLE 8—BID SECURITY

- 8.01 A Bid must be accompanied by Bid security made payable to Owner in an amount of five percent (5%) of Bidder's maximum Bid price (determined by adding the base bid and all alternates). Security shall be in the form of a certified check, cashier's check, or Bid bond issued by a surety meeting the requirements of Paragraph 6.01 of the General Conditions. Such Bid bond shall be issued in the form included in the Bidding Documents.
- 8.02 The Bid security of the apparent Successful Bidder will be retained until Owner awards the contract to such Bidder, and such Bidder has executed the Contract, furnished the required Contract security, and met the other conditions of the Notice of Award, whereupon the Bid security will be released. If the Successful Bidder fails to execute and deliver the Contract and furnish the required Contract security within 15 days after the Notice of Award, Owner may consider Bidder to be in default, annul the Notice of Award, and the Bid security of that Bidder will be forfeited in whole. Such forfeiture will be Owner's exclusive remedy if Bidder defaults.
- 8.03 The Bid security of other Bidders that Owner believes to have a reasonable chance of receiving the award may be retained by Owner until the earlier of 7 days after the Effective Date of the Contract or 46 days after the Bid opening, whereupon Bid security furnished by such Bidders will be released.
- 8.04 Bid security of other Bidders that Owner believes do not have a reasonable chance of receiving the award will be released within 7 days after the Bid opening.

#### **ARTICLE 9—CONTRACT TIMES**

- 9.01 The number of days within which, or the dates by which, the Work is to be (a) substantially completed and (b) ready for final payment, and (c) Milestones (if any) are to be achieved, are set forth in the Agreement.
- 9.02 Provisions for liquidated damages, if any, for failure to timely attain a Milestone, Substantial Completion, or completion of the Work in readiness for final payment, are set forth in the Agreement.

#### ARTICLE 10—SUBSTITUTE AND "OR EQUAL" ITEMS

- 10.01 The Contract for the Work, as awarded, will be on the basis of materials and equipment specified or described in the Bidding Documents, and those "or-equal" or substitute materials and equipment approved by the Engineer.
- 10.02 Whenever the Bidding Documents specify that an "approved equal" material or equipment may be furnished or used, application for such acceptance shall be made via a Shop Drawing or Submittal after the Effective Date of the Contract in accordance with Paragraphs 7.05 and 7.06 of the General Conditions.
- 10.03 Whenever the Bidding Documents specify that a "prior-approved equal" material or equipment may be furnished or used, application for such acceptance shall be submitted in writing by the Bidder and received by the Engineer at least 7 working days prior to the Bid opening. Each such request must comply with the requirements of Paragraphs 7.05 and 7.06 of the General Conditions, and the review of the request will be governed by the principles in those paragraphs. The burden of proof of the merit of the proposed item is upon Bidder. Engineer's decision of approval or disapproval of a proposed item will be final. If Engineer approves any such proposed item, such approval will be set forth in an Addendum issued to all registered Bidders. Bidders cannot rely upon approvals made in any other manner.
- 10.04 All prices that Bidder sets forth in its Bid will be based on the presumption that the Contractor will furnish the materials and equipment specified or described in the Bidding Documents, as supplemented by Addenda. Any assumptions regarding the possibility of post-Bid approvals of "or-equal" or substitution requests are made at Bidder's sole risk.

#### ARTICLE 11—SUBCONTRACTORS, SUPPLIERS, AND OTHERS

- 11.01 Upon request from the Owner or Engineer, the apparent Successful Bidder, and any other Bidder so requested, must submit a list of the Subcontractors or Suppliers proposed for the Work within ten days of the request.
- 11.02 If requested, such list must be accompanied by an experience statement with pertinent information regarding similar projects and other evidence of qualification for each such Subcontractor or Supplier. If Owner or Engineer, after due investigation, has reasonable objection to any proposed Subcontractor or Supplier, Owner may, before the Notice of Award is given, request apparent Successful Bidder to submit an acceptable substitute, in which case apparent Successful Bidder will submit a substitute, without an increase in Bid price.
- 11.03 If apparent Successful Bidder declines to make any such substitution, Owner may award the Contract to the next lowest Bidder that proposes to use acceptable Subcontractors and Suppliers.

Declining to make requested substitutions will constitute grounds for forfeiture of the Bid security of any Bidder. Any Subcontractor or Supplier, so listed and against which Owner or Engineer makes no written objection prior to the giving of the Notice of Award will be deemed acceptable to Owner and Engineer subject to subsequent revocation of such acceptance as provided in Paragraph 7.07 of the General Conditions.

#### **ARTICLE 12—PREPARATION OF BID**

- 12.01 The Bid Form is included with the Bidding Documents.
  - A. All blanks on the Bid Form must be completed in ink and the Bid Form signed in ink. Alterations must be scratched through and initialed in ink by the Bidder or person who submits the Bid Form.
  - B. A Bid price must be indicated for each section, Bid item, alternate, adjustment unit price item, and unit price item listed therein. If Bidder elects to not furnish pricing for an alternate item, then Bidder may enter the words "No Bid" or "Not Applicable."
- 12.02 If Bidder has obtained the Bidding Documents as Electronic Documents, then Bidder shall prepare its Bid on a paper copy of the Bid Form printed from the Electronic Documents version of the Bidding Documents. The printed copy of the Bid Form must be clearly legible, printed on 8½ inch by 11-inch paper and as closely identical in appearance to the Electronic Document version of the Bid Form as may be practical.
- 12.03 A Bid must contain a corporate resolution or written evidence of authority of the person signing the Bid in accordance with LA RS 38:2212.B(5).
- 12.04 The Bid must contain an acknowledgment of receipt of all Addenda, the numbers of which must be filled in on the Bid Form.

#### **ARTICLE 13—BASIS OF BID**

#### 13.01 Lump Sum

A. Bidders must submit a Bid on a lump sum basis as set forth in the Bid Form.

#### 13.02 Base Bid with Alternates

A. Bidders must submit a Bid on a lump sum basis for the base Bid and include a separate price for each alternate described in the Bidding Documents and as provided for in the Bid Form. The price for each alternate will be the amount added to or deleted from the base Bid if Owner selects the alternate.

#### 13.03 Unit Price

- A. Bidders must submit a Bid on a unit price basis for each item of Work listed in the unit price section of the Bid Form.
- 3. The "Bid Price" (sometimes referred to as the extended price) for each unit price Bid item will be the product of the "Estimated Quantity", which Owner or its representative has set forth in the Bid Form, for the item and the corresponding "Bid Unit Price" offered by the Bidder. The total of all unit price Bid items will be the sum of these "Bid Prices"; such total will be used by Owner for Bid comparison purposes. The final quantities and Contract Price will be determined in accordance with Paragraph 13.03 of the General Conditions.

C. Discrepancies between the multiplication of units of Work and unit prices will be resolved in favor of the unit prices. Discrepancies between the indicated sum of any column of figures and the correct sum thereof will be resolved in favor of the correct sum.

#### **ARTICLE 14—SUBMITTAL OF BID**

- 14.01 A Bid must be received no later than the date and time prescribed and at the place indicated in the Advertisement or invitation to bid and must be enclosed in a plainly marked package with the Project title, and, if applicable, the designated portion of the Project for which the Bid is submitted and the name and address of Bidder. If a Bid is sent by mail or other delivery system, the sealed envelope containing the Bid must be enclosed in a separate package plainly marked on the outside with the notation "BID ENCLOSED." A mailed Bid must be addressed to the location designated in the Advertisement.
- 14.02 In accordance with LA RS 37:2163, all Bidders must show their Louisiana State Contractor's License number on the Bid envelope. Any Bid that does not require the Bidder to hold an active Louisiana State Contractor's License shall state the exemption on the Bid envelope. Any Bid failing to meet these provisions will not be opened.
- 14.03 Bids received after the date and time prescribed for the opening of bids, or not submitted at the correct location or in the designated manner, will not be accepted and will be returned to the Bidder unopened.

#### ARTICLE 15—MODIFICATION AND WITHDRAWAL OF BID

- 15.01 An unopened Bid may be withdrawn by an appropriate document duly executed in the same manner that a Bid must be executed and delivered to the place where Bids are to be submitted prior to the date and time for the opening of Bids. Upon receipt of such notice, the unopened Bid will be returned to the Bidder.
- 15.02 If a Bidder wishes to modify its Bid prior to Bid opening, Bidder must withdraw its initial Bid in the manner specified in Paragraph 15.01 and submit a new Bid prior to the date and time for the opening of Bids.
- 15.03 In accordance with LA RS 38:2214, Bids containing patently obvious, unintentional, and substantial mechanical, clerical, or mathematical errors, or errors of unintentional omission of a substantial quantity of work, labor, material, or services made directly in the compilation of the Bid, may be withdrawn by the Bidder if clear and convincing sworn, written evidence of such errors is furnished to the Owner within 48 hours after Bids are opened. If the Owner determines that the error was a patently obvious mechanical, clerical, or mathematical error, or unintentional omission of a substantial quantity of work, labor, material, or services, as opposed to an error in judgement, and that the Bid was submitted in good faith, the Bid security will be returned. Thereafter, if the Work is rebid, the Bidder will be disqualified from further bidding on the Work.

#### **ARTICLE 16—OPENING OF BIDS**

16.01 Bids will be opened at the time and place indicated in the Advertisement or invitation to bid and, unless obviously non-responsive, read aloud publicly.

#### ARTICLE 17—BIDS TO REMAIN SUBJECT TO ACCEPTANCE

17.01 All Bids will remain subject to acceptance for up to 45 days following the date of the Bid opening; however, the Owner and the lowest responsible and responsive Bidder, by mutually written consent, may agree to extend the deadline for award by one or more extensions of 30 calendar days. Owner may, at its sole discretion, release any Bid and return the Bid security prior to the end of this period.

#### ARTICLE 18—EVALUATION OF BIDS AND AWARD OF CONTRACT

- 18.01 Owner reserves the right to reject any or all Bids for just cause in accordance with LA RS 38:2214.
- 18.02 Owner will reject the Bid of any Bidder that Owner finds, after reasonable inquiry and evaluation, to not be responsible.
- 18.03 If Bidder purports to add terms or conditions to its Bid, takes exception to any provision of the Bidding Documents, or attempts to alter the contents of the Contract Documents for purposes of the Bid, whether in the Bid itself or in a separate communication to Owner or Engineer, then Owner will reject the Bid as nonresponsive.

#### 18.04 Evaluation of Bids

- A. In evaluating Bids, Owner will consider whether the Bids comply with the prescribed requirements, and such alternates, unit prices, and other data, as may be requested in the Bid Form or prior to the Notice of Award.
- B. For the determination of the apparent low Bidder when unit price bids are submitted, Bids will be compared on the basis of the total of the products of the estimated quantity of each item and unit price Bid for that item, together with any lump sum items.
- 18.05 In evaluating whether a Bidder is responsible, Owner will consider the qualifications of the Bidder and may consider the qualifications and experience of Subcontractors and Suppliers proposed for those portions of the Work for which the identity of Subcontractors and Suppliers must be submitted as provided in the Bidding Documents.
- 18.06 Owner may conduct such investigations as Owner deems necessary to establish the responsibility, qualifications, and financial ability of Bidders and any proposed Subcontractors or Suppliers.
- 18.07 If Owner awards the contract for the Work, such award will be to the responsible Bidder submitting the lowest responsive Bid.

#### ARTICLE 19—BONDS AND INSURANCE

- 19.01 Article 6 of the General Conditions, as may be modified by the Supplementary Conditions, sets forth Owner's requirements as to performance and payment bonds, other required bonds (if any), and insurance. When the Successful Bidder delivers the executed Agreement to Owner, it must be accompanied by required bonds and insurance documentation.
- 19.02 Article 8, Bid Security, of these Instructions, addresses any requirements for providing bid bonds as part of the bidding process.

#### **ARTICLE 20—SIGNING OF AGREEMENT**

- 20.01 Owner will issue the unexecuted Agreement and counterparts to the Successful Bidder. Within 15 days thereafter, Successful Bidder must execute and deliver the required number of copies of the Agreement and any bonds and insurance documentation required to be delivered by the Contract Documents to Owner.
- 20.02 In accordance with LA RS 38:2215, the Successful Bidder and Owner shall execute the Contract not later than 60 days after the date of Contract award.
- 20.03 Owner will deliver one fully executed counterpart of the Agreement to Successful Bidder, together with printed and electronic copies of the Contract Documents as stated in Paragraph 2.02 of the General Conditions at the preconstruction meeting.

#### **ARTICLE 21—SALES AND USE TAXES**

21.01 Owner is exempt from Louisiana state sales and use taxes on materials and equipment to be incorporated in the Work. Said taxes must not be included in the Bid. Refer to Paragraph SC-7.10 of the Supplementary Conditions for additional information.

#### LOUISIANA UNIFORM PUBLIC WORK BID FORM

**TO:** Beauregard Parish Police Jury 201 W. 2<sup>nd</sup> Street DeRidder, LA 70634

BID FOR: Parish Annex Building Renovations (Phase I)

M.A. Project No. H1-22029-DA

The undersigned bidder hereby declares and represents that she/he; a) has carefully examined and understands the Bidding Documents, b) has not received, relied on, or based his bid on any verbal instructions contrary to the Bidding Documents or any addenda, c) has personally inspected and is familiar with the project site, and hereby proposes to provide all labor, materials, tools, appliances and facilities as required to perform, in a workmanlike manner, all work and services for the construction and completion of the referenced project, all in strict accordance with the Bidding Documents prepared by: Meyer & Associates, Inc. and dated: March 2023.

Bidders must acknowledge all addenda. The Bidder acknowledges receipt of the following ADDENDA: (Enter the number the Designer has assigned to each of the addenda that the Bidder is acknowledging) \_\_\_\_\_\_\_\_.

TOTAL BASE BID: For all work required by the Bidding Documents (including any and all unit prices designated "Base Bid" \* but not

| alternates) the sum of:  | D. II.                        | (6                      | ,          |
|--|-------------------------------|-------------------------|------------|
|  | Dolla                         | ırs (\$                 | )          |
| <b>ALTERNATES:</b> For any and all work required by the Bidding Docum as alternates in the unit price description. | ents for Alternates including | any and all unit prices | designated |
| Alternate No. 1 – (Interior Build-Out) for the lump sum of:  |                               |                         |            |
|  | Dollars (\$                   |                         | )          |
| Alternate No. 2 – (Owner to provide description of alternate and sta   | ate whether add or deduct) fo | r the lump sum of:      |            |
| Not Applicable   | Dollars (\$                   | Not Applicable          | )          |
| Alternate No. 3 – (Owner to provide description of alternate and sta   | ate whether add or deduct) fo | r the lump sum of:      |            |
| Not Applicable   | Dollars (\$                   | Not Applicable          | )          |
| NAME OF BIDDER:  |                               |                         |            |
| ADDRESS OF BIDDER:   |                               |                         |            |
| LOUISIANA CONTRACTOR'S LICENSE NUMBER:   |                               |                         |            |
| NAME OF AUTHORIZED SIGNATORY OF BIDDER:  |                               |                         |            |
| TITLE OF AUTHORIZED SIGNATORY OF BIDDER:   |                               |                         |            |
| SIGNATURE OF AUTHORIZED SIGNATORY OF BIDDER **:  |                               |                         |            |

#### THE FOLLOWING ITEMS ARE TO BE INCLUDED WITH THE SUBMISSION OF THIS LOUISIANA UNIFORM PUBLIC WORK BID FORM:

- \* The <u>Unit Price Form</u> shall be used if the contract includes unit prices. Otherwise it is not required and need not be included with the form. The number of unit prices that may be included is not limited and additional sheets may be included if needed.
- \*\* A CORPORATE RESOLUTION OR WRITTEN EVIDENCE of the authority of the person signing the bid for the public work as prescribed by LA R.S. 38:2212(B)(5).

**BID SECURITY** in the form of a bid bond, certified check or cashier's check as prescribed by LA RS 38:2218(A) is attached to and made a part of this bid.

March 2020 Page 1 of 3

DATE:

This page intentionally left blank

UNIT PRICE EXTENSION (Quantity times Unit Price)

**UNIT PRICE** 

UNIT OF MEASURE: Lump Sum

QUANTITY:

713-01-00100 REF. NO.

026

# **LOUISIANA UNIFORM PUBLIC WORK BID FORM**

# **UNIT PRICE FORM**

Beauregard Parish Police Jury DeRidder, LA 70634 201 W. 2nd Street ë

Parish Annex Building Renovations (Phase I) **BID FOR:** 

M.A. Project No. H1-22029-DA

UNIT PRICES: This form shall be used for any and all work required by the Bidding Documents and described as unit prices. Amounts shall be stated in figures and only in figures.

| DESCRIPTION: | ■ Base Bid or □ Alt. # | Removal and Disposal of Pa | Removal and Disposal of Pavement (Asphaltic and Concrete) |  |
|--------------|------------------------|----------------------------|---|--|
| REF. NO.     | QUANTITY:              | UNIT OF MEASURE:           | UNIT PRICE  | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 202-02-32600 | 440                    | λS                         |   |  |
|              |                        |                            |   |  |

| DESCRIPTION: | ■ Base Bid or □ Alt. # | General Excavation |            |  |
|--------------|------------------------|--------------------|------------|--|
| REF. NO.     | QUANTITY:              | UNIT OF MEASURE:   | UNIT PRICE | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 203-01-00100 | 275                    | CY                 |            |  |
|              |                        |                    |            |  |

|  |            |                  |           | •            |
|--|------------|------------------|-----------|--------------|
|  |            | SY               | 062       | 302-02-06070 |
| UNIT PRICE EXTENSION (Quantity times Unit Price) | UNIT PRICE | UNIT OF MEASURE: | QUANTITY: | REF. NO.     |

Class II Base Course (6" Thick) (Crushed Stone or Recycled Portland Cement Concrete)

■ Base Bid or □ Alt. #

DESCRIPTION:

| DESCRIPTION: | ■ Base Bid or □ Alt. # | Asphalt Concrete, Drives, T | , Drives, Turnouts and Misc. |  |
|--------------|------------------------|-----------------------------|------------------------------|--|
| REF. NO.     | QUANTITY:              | UNIT OF MEASURE:            | UNIT PRICE                   | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 502-01-00200 | 155                    | Tons                        |                              |  |
|              |                        |                             |                              |  |

| DESCRIPTION: | ■ Base Bid or □ Alt. # | Concrete Walk (4" Thick) |            |  |
|--------------|------------------------|--------------------------|------------|--|
| REF. NO.     | QUANTITY:              | UNIT OF MEASURE:         | UNIT PRICE | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 706-01-00100 | 300                    | SY                       |            |  |
|              |                        |                          |            |  |

| DESCRIPTION: | ■ Base Bid or 🏻 Alt. # | <b>Concrete Swale Ditches</b>  |            |  |
|--------------|------------------------|--------------------------------|------------|--|
| REF. NO.     | QUANTITY:              | UNIT OF MEASURE:               | UNIT PRICE | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 712-06-00100 | 250                    | 17                             |            |  |
|              |                        |                                |            |  |
| DESCRIPTION: | ■ Base Bid or □ Alt. # | Temporary Signs and Barricades | ades       |  |

| DESCRIPTION:  | ■ Base Bid or □ Alt.#    | Mobilization                 |  |  |
|---------------|--------------------------|------------------------------|--|--|
| REF. NO.      | QUANTITY:                | UNIT OF MEASURE:             | UNIT PRICE   | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 727-01-00100  | 1                        | Lump Sum                     |  |  |
|               |                          |                              |  |  |
| DESCRIPTION:  | ■ Base Bid or □ Alt. #   | Construction Layout          |  |  |
| REF. NO.      | QUANTITY:                | UNIT OF MEASURE:             | UNIT PRICE   | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| 740-010-00100 | 1                        | Lump Sum                     |  |  |
|               |                          |                              |  |  |
| DESCRIPTION:  | ■ Base Bid or □ Alt. #   | Trench Drain (18" Width)     |  |  |
| REF. NO.      | QUANTITY:                | UNIT OF MEASURE:             | UNIT PRICE   | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| NS-700-00100  | 10                       | LF                           |  |  |
|               |                          |                              |  |  |
| DESCRIPTION:  | ■ Base Bid or □ Alt. #   | All site work and exterior w | xterior work to building envelope per plans and specifications | specifications                                   |
| REF. NO.      | QUANTITY:                | UNIT OF MEASURE:             | UNIT PRICE   | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| S-001         | 1                        | ST                           |  |  |
|               |                          |                              |  |  |
| DESCRIPTION:  | ☐ Base Bid or ■ Alt. # 1 | Interior Build-Out           |  |  |
| REF. NO.      | QUANTITY:                | UNIT OF MEASURE:             | UNIT PRICE   | UNIT PRICE EXTENSION (Quantity times Unit Price) |
| S-002         | 1                        | ST                           |  |  |

3 of 3

#### **BID BOND (PENAL SUM FORM)**

| Bidder   | Surety  |
|--|---|
| Name: [Full formal name of Bidder]                               | Name: [Full formal name of Surety]                                  |
| Address (principal place of business):                           | Address (principal place of business):                              |
| [Address of Bidder's principal place of business]                | [Address of Surety's principal place of business]                   |
|  |   |
|  |   |
| Owner  | Bid   |
| Name: [Full formal name of Owner]                                | Project (name and location):  |
| Address (principal place of business):                           | [Owner project/contract name, and location of                       |
| [Address of Owner's principal place of business]                 | the project]  |
|  |   |
|  |   |
|  | Bid Due Date: [Enter date bid is due]                               |
| Bond   |   |
| Penal Sum: [Amount]  |   |
| Date of Bond: [Date]   |   |
| Surety and Bidder, intending to be legally bound he              | ereby, subject to the terms set forth in this Bid Bond,             |
| do each cause this Bid Bond to be duly executed by               | an authorized officer, agent, or representative.                    |
| Bidder   | Surety  |
|  |   |
| (Full formal name of Bidder)                                     | (Full formal name of Surety) (corporate seal)                       |
| Ву:  | Ву:   |
| (Signature)  | (Signature) (Attach Power of Attorney)                              |
| Name: (Printed or typed)   | Name:(Printed or typed)   |
| Title:   | Title:  |
| Title.   | Title.  |
| Attest:  | Attest:   |
| (Signature)  | (Signature)   |
| Name:  | Name:   |
| (Printed or typed)   | (Printed or typed)  |
| Title:   | Title:  |
| Notes: (1) Note: Addresses are to be used for giving any require | ed notice. (2) Provide execution by any additional parties, such as |

- 1. Bidder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to pay to Owner upon default of Bidder the penal sum set forth on the face of this Bond. Payment of the penal sum is the extent of Bidder's and Surety's liability. Recovery of such penal sum under the terms of this Bond will be Owner's sole and exclusive remedy upon default of Bidder.
- 2. Default of Bidder occurs upon the failure of Bidder to deliver within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents.
- 3. This obligation will be null and void if:
  - 3.1. Owner accepts Bidder's Bid and Bidder delivers within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents, or
  - 3.2. All Bids are rejected by Owner, or
  - 3.3. Owner fails to issue a Notice of Award to Bidder within the time specified in the Bidding Documents (or any extension thereof agreed to in writing by Bidder and, if applicable, consented to by Surety when required by Paragraph 5 hereof).
- 4. Payment under this Bond will be due and payable upon default of Bidder and within 30 calendar days after receipt by Bidder and Surety of written notice of default from Owner, which notice will be given with reasonable promptness, identifying this Bond and the Project and including a statement of the amount due.
- 5. Surety waives notice of any and all defenses based on or arising out of any time extension to issue Notice of Award agreed to in writing by Owner and Bidder, provided that the total time for issuing Notice of Award including extensions does not in the aggregate exceed 120 days from the Bid due date without Surety's written consent.
- 6. No suit or action will be commenced under this Bond prior to 30 calendar days after the notice of default required in Paragraph 4 above is received by Bidder and Surety, and in no case later than one year after the Bid due date.
- 7. Any suit or action under this Bond will be commenced only in a court of competent jurisdiction located in the state in which the Project is located.
- 8. Notices required hereunder must be in writing and sent to Bidder and Surety at their respective addresses shown on the face of this Bond. Such notices may be sent by personal delivery, commercial courier, or by United States Postal Service registered or certified mail, return receipt requested, postage pre-paid, and will be deemed to be effective upon receipt by the party concerned.
- 9. Surety shall cause to be attached to this Bond a current and effective Power of Attorney evidencing the authority of the officer, agent, or representative who executed this Bond on behalf of Surety to execute, seal, and deliver such Bond and bind the Surety thereby.
- 10. This Bond is intended to conform to all applicable statutory requirements. Any applicable requirement of any applicable statute that has been omitted from this Bond will be deemed to be included herein as if set forth at length. If any provision of this Bond conflicts with any applicable statute, then the provision of said statute governs and the remainder of this Bond that is not in conflict therewith continues in full force and effect.
- 11. The term "Bid" as used herein includes a Bid, offer, or proposal as applicable.

# FORMS TO BE SUBMITTED WITHIN 10 DAYS OF THE BID OPENING

This page intentionally left blank

| Name of Project | Project No. |
|-----------------|-------------|
| STATE OF        |             |
| PARISH OF       |             |

#### **ATTESTATIONS AFFIDAVIT**

**Before me**, the undersigned notary public, duly commissioned and qualified in and for the parish and state aforesaid, personally came and appeared Affiant, who after being duly sworn, attested as follows:

#### LA. R.S. 38:2227 PAST CRIMINAL CONVICTIONS OF BIDDERS

- A. No sole proprietor or individual partner, incorporator, director, manager, officer, organizer, or member who has a minimum of a ten percent (10%) ownership in the bidding entity named below has been convicted of, or has entered a plea of guilty or nolo contendere to any of the following state crimes or equivalent federal crimes:
  - (a) Public bribery (R.S. 14:118)
- (c) Extortion (R.S. 14:66)
- (b) Corrupt influencing (R.S. 14:120)
- (d) Money laundering (R.S. 14:230)
- B. Within the past five years from the project bid date, no sole proprietor or individual partner, incorporator, director, manager, officer, organizer, or member who has a minimum of a ten percent (10%) ownership in the bidding entity named below has been convicted of, or has entered a plea of guilty or nolo contendere to any of the following state crimes or equivalent federal crimes, during the solicitation or execution of a contract or bid awarded pursuant to the provisions of Chapter 10 of Title 38 of the Louisiana Revised Statutes:
  - (a) Theft (R.S. 14:67)
  - (b) Identity Theft (R.S. 14:67.16)
  - (c) Theft of a business record (R.S.14:67.20)
  - (d) False accounting (R.S. 14:70)
  - (e) Issuing worthless checks (R.S. 14:71)

- (f) Bank fraud (R.S. 14:71.1)
- (g) Forgery (R.S. 14:72)
- (h) Contractors; misapplication of payments (R.S. 14:202)
- (i) Malfeasance in office (R.S. 14:134)

#### LA. R.S. 38:2212.10 Verification of Employees

- A. At the time of bidding, Appearer is registered and participates in a status verification system to verify that all new hires in the state of Louisiana are legal citizens of the United States or are legal aliens.
- B. If awarded the contract, Appearer shall continue, during the term of the contract, to utilize a status verification system to verify the legal status of all new employees in the state of Louisiana.
- C. If awarded the contract, Appearer shall require all subcontractors to submit to it a sworn affidavit verifying compliance with Paragraphs (A) and (B) of this Subsection.

July 2021

|           | Name of Project   |   | Project No.   |
|-----------|---|---|---|
| <u>LA</u> | . R.S. 23:1726(B) Certification Regard  | ding Unpaid Workers Compensa  | ation Insurance   |
| A.        | R.S. 23:1726 prohibits any entity again Louisiana Revised Statutes of 1950 (Alte to appeal that assessment is exhausted, to Chapter 10 of Title 38 of the Louisiana Revised Statutes of 1950. | ernative Collection Procedures & Ass<br>, from submitting a bid or proposal | sessments) is in effect, and whose right for or obtaining any contract pursuant |
| В.        | By signing this bid /proposal, Affiant cerentity.   | rtifies that no such assessment is in                                       | effect against the bidding / proposing  |
|           |   |   |   |
|           | NAME OF BIDDER  | NAME OF AURTHORIZED   | SIGNATORY OF BIDDER   |
|           | DATE  | TITLE OF AUTHORIZED SIG   | GANTORY OF BIDDER   |
|           |   | SIGNATURE OF AUTHORIZED SIGNATORY OF BIDDER/AFFIANT                         |   |
|           | Sworn to and subscribed before me   | e by Affiant on the day of _  | , 20  |
|           |   | Notary Public   |   |

#### AFFIDAVIT FOR

## ATTESTING THAT PUBLIC CONTRACT WAS NOT SECURED THROUGH EMPLOYMENT OR PAYMENT OF SOLICITOR

To be executed by all contractors, subcontractors, persons, corporations, firms, associations, or other organizations receiving value for services rendered in connection with this project:

| STATE OF  |   |  |
|---|---|--|
| PARISH OR COUNTY OF   |   |  |
|   |   | , Being first duly sworn,  |
| deposes and says that:  |   |  |
| (1) He is   | of  |  |
| (Title)<br>the/a Contractor, Subcontractor, Person, Cor<br>value for services rendered in connection wit  |   | (Company Name) Association, or other organization receiving  |
| or indirectly, to secure this public contract or regularly employed by the affiant whose sepublic contract were in the regular course of (3) That no part of the contract price person, corporation, firm, association, or opayment of their normal compensation to perconnection with the project are/were in the resulting to the connection with the project are/were in the resulting to the connection with the project are/were in the resulting to the contract of the | ervices in conn<br>their duties for<br>the to be receive<br>other organizat<br>ersons regularly | ection with this project or in securing this affiant; and  d by affiant was paid or will be paid to any ion for soliciting the contract, other than gemployed by the affiant whose services in |
|   | Signed:   |  |
|   | Title:  |  |
| SWORN TO AND SUBSCRIBED before me this day of, 20   |   |  |
| NOTARY PUBLIC   |   |  |
| My Commission Expires   |   |  |

This page intentionally left blank

# **QUALIFICATIONS STATEMENT**

# **ARTICLE 1—GENERAL INFORMATION**

1.02

1.03

1.01 Provide general information for the Bidder:

|   | -           |             |                 |  |            |              |               |         |
|---|-------------|-------------|-----------------|--|------------|--------------|---------------|---------|
| Legal Name of Bus   | ness:       |             |                 |  |            |              |               |         |
| Corporate Office (p   | rovide cont | tact inforn | nation fo       | or local office on separate sheet if applicable) |            |              |               |         |
| Name:   |             |             |                 | Phon   | e numbe    | er:          |               |         |
| Title:  |             |             |                 | Emai   | l address  | s:           |               |         |
| Business address o  | f corporate | office:     |                 | •  |            | •            |               |         |
|   |             |             |                 |  |            |              |               |         |
|   |             |             |                 |  |            |              |               |         |
| ID Numbers  |             |             |                 |  |            |              |               |         |
| Federal ID Number   | :           |             |                 |  |            |              |               |         |
| Unique Entity ID (L   | IEI):       |             |                 | Act  | tive per v | vww.s        | sam.gov: 🗆 Ye | es 🗆 No |
| Provide information on the Bidder's organizational structure:   |             |             |                 |  |            |              |               |         |
| Form of Business:   | □ Sole P    | ronrietors  | hin $\square$ P | artners  | hin □ Co   | rnora        | tion          |         |
| Form of Business: ☐ Sole Proprietorship ☐ Partnership ☐ Corporation ☐ Limited Liability Company ☐ Joint Venture comprised of the following companies: |             |             |                 | ·•   |            |              |               |         |
| 1.  |             |             |                 | ·· <u> </u>                                      |            |              |               |         |
| 2.  |             |             |                 |  |            |              |               |         |
| 3.  |             |             |                 |  |            |              |               |         |
| Note: Provide a separate Qualification Statement for each Joint Venturer.   |             |             |                 |  |            |              |               |         |
|   |             |             | -               |  |            | was formed:  |               |         |
| Is this Business authorized to operate in the Pro   |             |             | ject loca       | ation?   | ☐ Ye       | s □ No □ Pen | ding          |         |
| Is this Business authorized to operate in the Project location? ☐ Yes ☐ No ☐ Pending  |             |             |                 |  |            |              |               |         |
| Provide information regarding the Bidder's officers, partners, and limits of authority.   |             |             |                 |  |            |              |               |         |
| Name:   |             |             |                 | Title:   |            |              |               |         |
| Authorized to sign  | contracts:  | □ Yes □ N   | No              | Limit o  | of Author  | ity:         | \$            |         |
| Name:   |             |             |                 | Title:   |            |              |               |         |
| Authorized to sign  | contracts:  | □ Yes □ N   | No              | Limit  | of Author  | ity:         | \$            |         |
| Name:   |             |             |                 | Title:   |            |              |               |         |
| Authorized to sign  | contracts:  | □ Yes □ N   | No              | Limit  | of Author  | ity:         | \$            |         |
|   |             |             |                 |  |            |              |               |         |

### **ARTICLE 2—CONSTRUCTION EXPERIENCE**

2.01 Provide information regarding the Bidder's previous contracting experience.

| Years of experience with projects like the proposed project:  |  |                      |  |  |  |
|---|--|----------------------|--|--|--|
| As a general contractor:  |  | As a joint venturer: |  |  |  |
| Has Bidder, or a predecessor in interest, or an affiliate with 25% or more in interest:   |  |                      |  |  |  |
| Been disqualified as a bidder by any local, state, or federal agency within the last 5 years?  ☐ Yes ☐ No                               |  |                      |  |  |  |
| Been barred from contracting by any local, state, or federal agency within the last 5 years?   ☐ Yes ☐ No                               |  |                      |  |  |  |
| Been released from a bid in the past 5 years? $\square$ Yes $\square$ No  |  |                      |  |  |  |
| Defaulted on a project or failed to complete any contract awarded to it? $\Box$ Yes $\Box$ No   |  |                      |  |  |  |
| Refused to construct or refused to provide materials defined in the contract documents or in a change order? $\square$ Yes $\square$ No |  |                      |  |  |  |
| Been a party to any currently pending litigation or arbitration? ☐ Yes ☐ No   |  |                      |  |  |  |
| Been a party to any project related litigation or arbitration within the last 5 years?  ☐ Yes ☐ No                                      |  |                      |  |  |  |
| Provide full details in a separate attachment if the response to any of these questions is Yes.   |  |                      |  |  |  |

2.02 List a minimum of three and a maximum of six projects completed in the last 5 years in Schedule A. Provide indicated information to demonstrate the Bidder's experience with projects similar in type and cost of construction and satisfactory performance based on a <u>positive</u> reference from the Project Owner and Engineer/Architect.

### **ARTICLE 3—SUBCONTRACTORS**

3.01 List all proposed Subcontractors and provide category or scope of work and approximate subcontract value. Subcontracts amounting to less than \$5,000 or less than 1% of the total amount bid do not have to be listed.

| Legal Nam                    | ne of Subcontr  | actor:                    |  |                   |          |          |
|------------------------------|-----------------|---------------------------|--|-------------------|----------|----------|
| Business                     |                 |                           |  | Phone number:     |          |          |
| address:                     |                 |                           |  | Email address:    |          |          |
|                              |                 |                           |  |                   |          |          |
| LA Contractor's License No.: |                 | Date Business was formed: |  |                   |          |          |
| Unique Er                    | ntity ID (UEI): |                           |  | Active per www.sa | m.gov: 🗆 | Yes □ No |
| Scope of \                   | Nork:           |                           |  |                   |          |          |
|                              |                 |                           |  |                   |          |          |

| Legal Nan         | ne of Subcontr  | ractor:  |                   |                   |
|-------------------|-----------------|----------|-------------------|-------------------|
| Business          |                 |          | Phone number:     |                   |
| address:          |                 |          | Email address:    |                   |
|                   |                 |          |                   |                   |
| LA Contra         | ctor's License  | No.:     | Date Business was | formed:           |
| Unique Er         | ntity ID (UEI): |          | Active per www.sa | m.gov: ☐ Yes ☐ No |
| Scope of \        | Nork:           |          |                   |                   |
| Legal Nan         | ne of Subcontr  | ractor:  |                   |                   |
| Business          |                 |          | Phone number:     |                   |
| address:          |                 |          | Email address:    |                   |
|                   |                 |          |                   |                   |
| LA Contra         | ctor's License  | No.:     | Date Business was | formed:           |
| Unique Er         | ntity ID (UEI): |          | Active per www.sa | m.gov: ☐ Yes ☐ No |
| Scope of \        | Nork:           | 1        | 1                 |                   |
| I a seel Man      | f C  + -        |          |                   |                   |
|                   | ne of Subcontr  | ractor:  | 1                 |                   |
| Business address: |                 |          | Phone number:     |                   |
|                   |                 |          | Email address:    |                   |
|                   |                 |          |                   |                   |
| LA Contra         | ctor's License  | No.:     | Date Business was | formed:           |
| Unique Er         | ntity ID (UEI): |          | Active per www.sa | m.gov: ☐ Yes ☐ No |
| Scope of \        | Nork:           |          |                   |                   |
| Legal Nan         | ne of Subcontr  | ractor:  |                   |                   |
| Business          |                 | <u> </u> | Phone number:     |                   |
| address:          |                 |          | Email address:    |                   |
|                   |                 |          |                   |                   |
| LA Contra         | ctor's License  | No.:     | Date Business was | formed:           |
| Unique Er         | ntity ID (UEI): |          | Active per www.sa | m.gov: ☐ Yes ☐ No |
| Scope of \        | Work:           | 1        | •                 |                   |
| Legal Nan         | ne of Subcontr  | ractor:  |                   |                   |
| Business          |                 | I        | Phone number:     |                   |
| address:          |                 |          | Email address:    |                   |
|                   | _               |          |                   |                   |
| LA Contra         | ctor's License  | No.:     | Date Business was | formed:           |
| Unique Er         | ntity ID (UEI): |          | Active per www.sa | m.gov: ☐ Yes ☐ No |
| Scope of \        | Work:           | 1        |                   |                   |
|                   |                 |          |                   |                   |

| 3.02   | Note that in accordance with SC-7.07 of the Supplementary Conditions, the Contractor shall not award work valued at more than 50% of the Contract Price to a single Subcontractor without prior written approval of the Owner. If the Bidder proposes to award work valued at more than 50% of the Contract Price to a single Subcontractor, a Qualifications Statement shall be completed for that Subcontractor. |
|--------|--|
| ARTICL | E 4—MANUFACTURERS  |

4.01 Provide the names of the equipment manufacturers that Bidder proposes for furnish for the following key items:

| Equipment Description: | None | Specification Section: | NA |
|------------------------|------|------------------------|----|
| Proposed Manufacturer: |      |                        |    |
|                        |      |                        |    |
| Equipment Description: | None | Specification Section: | NA |
| Proposed Manufacturer: |      |                        |    |
|                        |      |                        |    |
| Equipment Description: | None | Specification Section: | NA |
| Proposed Manufacturer: |      |                        |    |
|                        |      |                        |    |
| Equipment Description: | None | Specification Section: | NA |
| Proposed Manufacturer: |      |                        |    |
|                        |      |                        |    |
| Equipment Description: | None | Specification Section: | NA |
| Proposed Manufacturer: |      |                        |    |
|                        | ,    |                        |    |

4.02 The names provided above are firm, with equal importance to the Contract Price, and can only be changed by Change Order. If Bidder writes in more than one manufacturer for a given item, the Owner or Engineer will select which manufacturer to use.

### **ARTICLE 5—FINANCIAL INFORMATION**

5.01 Provide information regarding the Business's financial stability. Provide the most recent audited financial statement, and if such audited financial statement is not current, also provide the most current financial statement.

| Financial Institution:                         |                 |            |  |
|--|-----------------|------------|--|
| Business address:                              |                 |            |  |
| Date of Business's mos                         |                 | ☐ Attached |  |
| Date of Business's mos                         | ☐ Attached      |            |  |
| Financial indicators fro                       |                 |            |  |
| Contractor's Current R                         |                 |            |  |
| Contractor's Quick Rat<br>Short Term Investmen | ts Receivable + |            |  |

| Has Business or predecessor business been involved in bankruptcy or reorganization? |  |
|---|--|
| ☐ Yes ☐ No  |  |

### **ARTICLE 6—ADDITIONAL INFORMATION**

- 6.01 After review of the Qualifications Statement, Owner may request additional information from the Bidder to demonstrate Bidder's qualifications to perform the Work. Within 10 days of Owner's request, Bidder must submit such requested information as:
  - A. Written evidence establishing its qualifications such as financial data and present commitments.
  - B. Subcontractor and Supplier qualification information.
  - C. Other required information regarding qualifications.
- 6.02 No requirement in this Qualifications Statement to submit information will prejudice the right of Owner to seek additional pertinent information regarding Bidder's qualifications.

| This Stater  | ment of Qualifications is offered by:   |
|--------------|---|
| Bidder:      |   |
|              | (typed or printed name of organization)   |
| Ву:          | (individual's signature)  |
| Name:        |   |
|              | (typed or printed)  |
| Title:       | (typed or printed)  |
| Date:        | (date signed)   |
| (If Business | s is a corporation, a partnership, or a joint venture, attach evidence of authority to sign.) |
| Attest:      | (individual's signature)  |
| Name:        |   |
| Ti+lo.       | (typed or printed)  |
| Title:       | (typed or printed)  |
| Address fo   | r giving notices:   |
|              |   |
| Designated   | <br>d Representative:   |
| Name:        |   |
|              | (typed or printed)  |
| Title:       | (typed or printed)  |
| Address:     |   |
|              |   |
|              |   |
| Phone:       |   |
| Email:       |   |

# Schedule A—Previous Experience with Similar Projects

| Bidder Name                    |   |   |                 |                |                   |  |
|--------------------------------|---|---|-----------------|----------------|-------------------|--|
| Project Owner                  |   | В   | Project Name    |                |                   |  |
| General Description of Project | oject   |   |                 |                |                   |  |
| Project Cost                   |   |   | Date of Project | ct             |                   |  |
| Key Project Personnel          | Project Manager   | Project Superintendent                                      | endent          | %              | of Work Performed | % of Work Performed with Your Own Forces |
| Name                           |   |   |                 |                |                   |  |
| Reference Contact Inform       | Reference Contact Information (listing names indicates approv   | approval to contacting the names individuals as a reference | e names indi    | viduals as a ı | eference)         |  |
|                                | Name  | Title/Position  | Organization    | ation          | Telephone         | Email                                    |
| Owner                          |   |   |                 |                |                   |  |
| Engineer/Architect             |   |   |                 |                |                   |  |
| Construction Manager           |   |   |                 |                |                   |  |
| Project Owner                  |   |   | Project Name    |                |                   |  |
| General Description of Project | oject   |   |                 |                |                   |  |
| Project Cost                   |   |   | Date of Project | ct             |                   |  |
| Key Project Personnel          | Project Manager   | Project Superintendent                                      | endent          | %              | of Work Performed | % of Work Performed with Your Own Forces |
| Name                           |   |   |                 |                |                   |  |
| Reference Contact Inform       | Reference Contact Information (listing names indicates approval to contacting the names individuals as a reference) | approval to contacting th                                   | e names indi    | viduals as a ı | eference)         |  |
|                                | Name  | Title/Position  | Organization    | ation          | Telephone         | Email                                    |
| Owner                          |   |   |                 |                |                   |  |
| Engineer/Architect             |   |   |                 |                |                   |  |
| Construction Manager           |   |   |                 |                |                   |  |
| Project Owner                  |   |   | Project Name    |                |                   |  |
| General Description of Project | oject   |   |                 |                |                   |  |
| Project Cost                   |   |   | Date of Project | ct             |                   |  |
| Key Project Personnel          | Project Manager   | Project Superintendent                                      | endent          | %              | of Work Performed | % of Work Performed with Your Own Forces |
| Name                           |   |   |                 |                |                   |  |
| Reference Contact Inform       | Reference Contact Information (listing names indicates approval to contacting the names individuals as a reference) | approval to contacting th                                   | e names indi    | viduals as a ı | eference)         |  |
|                                | Name  | Title/Position  | Organization    | ation          | Telephone         | Email                                    |
| Owner                          |   |   |                 |                |                   |  |
| Engineer/Architect             |   |   |                 |                |                   |  |
| Construction Manager           |   |   |                 |                |                   |  |

 $\rm EJCDC^{\otimes}$  C-451, Qualifications Statement—Schedule A—Previous Experience with Similar Projects. Page 1 of 2

# Schedule A—Previous Experience with Similar Projects

| Bidder Name                    |   |   |                 |                |                   |  |
|--------------------------------|---|---|-----------------|----------------|-------------------|--|
| Project Owner                  |   | <u>a</u>  | Project Name    |                |                   |  |
| General Description of Project | oject   |   |                 |                |                   |  |
| Project Cost                   |   |   | Date of Project | ct             |                   |  |
| Key Project Personnel          | Project Manager   | Project Superintendent                                      | ndent           | %              | of Work Performed | % of Work Performed with Your Own Forces |
| Name                           |   |   |                 |                |                   |  |
| Reference Contact Inform       | Reference Contact Information (listing names indicates approv   | approval to contacting the names individuals as a reference | e names indiv   | viduals as a ı | eference)         |  |
|                                | Name  | Title/Position  | Organization    | ation          | Telephone         | Email                                    |
| Owner                          |   |   |                 |                |                   |  |
| Engineer/Architect             |   |   |                 |                |                   |  |
| Construction Manager           |   |   |                 |                |                   |  |
| Project Owner                  |   |   | Project Name    |                |                   |  |
| General Description of Project | oject   |   |                 |                |                   |  |
| Project Cost                   |   |   | Date of Project | ct             |                   |  |
| Key Project Personnel          | Project Manager   | Project Superintendent                                      | ndent           | %              | of Work Performed | % of Work Performed with Your Own Forces |
| Name                           |   |   |                 |                |                   |  |
| Reference Contact Inform       | Reference Contact Information (listing names indicates approval to contacting the names individuals as a reference) | approval to contacting the                                  | e names indiv   | viduals as a ı | eference)         |  |
|                                | Name  | Title/Position  | Organization    | ation          | Telephone         | Email                                    |
| Owner                          |   |   |                 |                |                   |  |
| Engineer/Architect             |   |   |                 |                |                   |  |
| Construction Manager           |   |   |                 |                |                   |  |
| Project Owner                  |   | <u>a</u>  | Project Name    |                |                   |  |
| General Description of Project | oject   |   |                 |                |                   |  |
| Project Cost                   |   |   | Date of Project | ct             |                   |  |
| Key Project Personnel          | Project Manager   | Project Superintendent                                      | ndent           | %              | of Work Performed | % of Work Performed with Your Own Forces |
| Name                           |   |   |                 |                |                   |  |
| Reference Contact Inform       | Reference Contact Information (listing names indicates approval to contacting the names individuals as a reference) | approval to contacting the                                  | e names indiv   | viduals as a ı | eference)         |  |
|                                | Name  | Title/Position  | Organization    | ation          | Telephone         | Email                                    |
| Owner                          |   |   |                 |                |                   |  |
| Engineer/Architect             |   |   |                 |                |                   |  |
| Construction Manager           |   |   |                 |                |                   |  |

EJCDC® C-451, Qualifications Statement—Schedule A—Previous Experience with Similar Projects. Page 2 of 2

# POST BID CONTRACT RELATED DOCUMENTS

This page intentionally left blank



# Designation of Construction Contractor as Agent of a Governmental Entity Sales Tax Exemption Certificate

|                                    | , an agency of the officed                                   |
|------------------------------------|--|
|                                    | ouisiana or its political subdivisions, including            |
| the following contracto            | or as its agent for the purpose of making sales              |
|                                    |  |
|                                    |  |
|                                    |  |
| State                              | ZIP  |
| onent construction mat<br>project: | terials, taxable services and leases and rentals             |
|                                    | Contract Number  |
| od                                 |  |
| End Date (mm/dd/yyyy)              |  |
| d contractor shall be c            | onsidered as the legal equivalent of purchases               |
|                                    | State  Onent construction matoroject:  End Date (mm/dd/yyyy) |

directly by the governmental body. Any materials purchased by this agent shall immediately, upon the vendor's delivery to the agent, become the property of this government entity. This government entity, as principal, assumes direct liability to the vendor for the payment of any property, services, leases, or rentals made by this designated agent. This agreement does not void or supersede the obligations of any party created under any construction contract related to this project, including specifically any contractual obligation of the construction contractor to submit payment to the vendors of materials or services for the project.

This contractor-agent is not authorized to delegate this purchasing agency to others; separate designations of agency by this governmental entity are required for each contractor or sub-contractor who is to purchase on behalf of this governmental entity. The undersigned hereby certify that this designation is the entirety of the agency designation agreement between them. In order for a purchase for an eligible governmental entity through a designated agent to be eligible for sales tax exemption, the designation of agency must be made, accepted, and disclosed to the vendor before or at the time of the purchase transaction.

| Designation of A                   | gency |                   | Acceptance of Age                                     | ency    |                   |
|------------------------------------|-------|-------------------|---|---------|-------------------|
| Signature of Authorized Designator |       | Date (mm/dd/yyyy) | Signature of Contractor or Subcontractor Authorized A | cceptor | Date (mm/dd/yyyy) |
|                                    |       |                   |   |         |                   |
| Name of Authorized Designator      |       |                   | Name of Contractor's or Subcontractor's Acceptor      |         |                   |
|                                    |       |                   |   |         |                   |
| Name of Governmental Entity        |       |                   | Name of Contractor                                    |         |                   |
|                                    |       |                   |   |         |                   |
| Address                            |       | Address           |   |         |                   |
|                                    |       |                   |   |         |                   |
| City                               | State | ZIP               | City  | State   | ZIP               |
|                                    |       |                   |   |         |                   |
|                                    |       |                   |   |         |                   |

This designation of agency form, when properly executed by both the contractor and the governmental entity, shall serve as evidence of the sales tax exempt status that has been conferred onto the contractor. No other exemption certificate form is necessary to claim exemption from sales taxes. The agency agreement evidenced by this sales tax exemption certificate must be implemented at the time of contract execution with the governmental entity. The contract between the governmental entity and his agent must contain provisions to authenticate the conferment of agency.

R-1020I(11/04)



# Designation of Construction Contractor as Agent of a Governmental Entity and Exemption Certificate

### General Information

**Purpose of the R-1020 Form:** Agencies and instrumentalities of federal or Louisiana state or local government may designate a construction contractor as its authorized agent for the purpose of purchasing construction materials, leasing and renting tangible personal property, and purchasing taxable services. Form R-1020 serves as the documentation by which the government entity and contractor document the agency relationship to vendors of materials and services. It also serves as documentation that the contractor's purchases are sales tax exempt, and therefore serves as an exemption certificate, which the vendor must retain on file to support the deduction he will claim on his sales tax return. Effective 11-1-2004, the R-1032 exemption certificate will no longer be necessary.

**Use of the R-1020 Form:** The form must be signed by both parties, contractor and governmental entity. After signature, both the contractor/agent and the governmental entity must keep an original copy of the form on file, along with other documents that pertain to the construction project. (Effective 11-1-2004) Do not send a copy of the R-1020 form to the Louisiana Department of Revenue. Retain your copy of the original certificate on file. The contractor/agent must reproduce the original copy as needed to attach a copy to each purchase order for materials for the project. The reproduced copy will serve as the exemption certificate that will document the exempt sale of materials to the contractor/agent.

**Subcontractors.** A designated contractor may not re-designate his subcontractors as authorized agents for the governmental entity. Each subcontractor must obtain its own designation from the governmental entity.

**Title to Property:** Any materials purchased by the agent through the use of this certificate immediately become the property of the governmental entity upon delivery to the contractor/agent.

**Restrictions as to Vendors:** The governmental entity may choose to restrict the agent/contractor to making purchases from a pre-selected list of vendors and providers of services. This restriction, if applicable, must be incorporated into a contractual agreement between the governmental entity and the designated agent. If there are no vendor restrictions, the contractor/agent may use the R-1020 Exemption Certificate to make sales tax exempt purchases from any vendor.

### **NOTICE OF AWARD**

| Date of Award:     |   |   |                             |
|--------------------|---|---|-----------------------------|
| Owner:             |   | Owner's Project No.:                          |                             |
| Engineer:          |   | Engineer's Project No.:                       |                             |
| Project:           |   |   |                             |
| Bidder:            |   |   |                             |
| Bidder's Addre     | ss:   |   |                             |
|                    | that Owner has accepted your Bid dated nat you are the Successful Bidder.   | Click or tap to enter a date. <b>, for th</b> | e above                     |
| Price is subject t | is for the Total Base Bid. The Contract Price adjustment based on the provisions of ges and Unit Price Work, as applicable. |   | . Contract<br>ited to those |
| •                  | pies of the Agreement, Contract security, a execution, as applicable, either accompan                                       | 0 0   | •                           |
| Within 15 days o   | of your receipt of these items, you must d  | eliver the following to the Enginee           | r:                          |
| 1. Execute         | d Agreement(s)  |   |                             |
| 2. Execute         | d Contract security (such as required perf  | ormance and payment bonds)                    |                             |
| 3. Other ex        | xecuted contract or funding agency relate   | d forms, as applicable                        |                             |
|                    | ce documentation, as specified in the Instons, Articles 2 and 6.  | ructions to Bidders and in the Gene           | eral                        |
| •                  | ly with these conditions within the time spairs Notice of Award, and declare your Bid                                       |   | der you in                  |
|                    | rith LA RS 38:2215, Contractor and Owner fter the date of Contract award.   | shall execute the Agreement no la             | ater than 60                |
| Owner:             |   |   |                             |
| By (signature):    |   |   |                             |
| Name (printed)     | ):  |   |                             |
| Title:             |   |   |                             |
| Copy: Enginee      | r   |   |                             |
|                    |   |   |                             |

This page intentionally left blank

# AGREEMENT BETWEEN OWNER AND CONTRACTOR FOR CONSTRUCTION CONTRACT (STIPULATED PRICE)

# **Prepared By**









# **Endorsed By**





# Copyright<sup>©</sup> 2018

National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882

www.nspe.org

American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474

www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723

www.asce.org

The copyright for this EJCDC document is owned jointly by the three sponsoring organizations listed above. The National Society of Professional Engineers is the Copyright Administrator for the EJCDC documents; please direct all inquiries regarding EJCDC copyrights to NSPE.

NOTE: EJCDC publications may be purchased at <a href="www.ejcdc.org">www.ejcdc.org</a>, or from any of the sponsoring organizations above.

# AGREEMENT BETWEEN OWNER AND CONTRACTOR FOR CONSTRUCTION CONTRACT (STIPULATED PRICE)

This Agreement is by and between <u>Beauregard Parish Police Jury</u> ("Owner") and \_\_\_\_\_ ("Contractor").

Terms used in this Agreement have the meanings stated in the General Conditions and the Supplementary Conditions.

Owner and Contractor hereby agree as follows:

### **ARTICLE 1—WORK**

1.01 Contractor shall complete all Work as specified or indicated in the Contract Documents. The Work is generally described as follows: Site civil renovation improvements to the existing building infrastructure.

### **ARTICLE 2—THE PROJECT**

2.01 The Project, of which the Work under the Contract Documents is a part, is generally described as follows: Parish Annex Building Renovations (Phase I)

### **ARTICLE 3—ENGINEER**

- 3.01 The Owner has retained Meyer & Associates, Inc. ("Engineer") to act as Owner's representative, assume all duties and responsibilities of Engineer, and have the rights and authority assigned to Engineer in the Contract.
- 3.02 The part of the Project that pertains to the Work has been designed by Engineer.

### **ARTICLE 4—CONTRACT TIMES**

- 4.01 Time is of the Essence
  - A. All time limits for Milestones, if any, Substantial Completion, and completion and readiness for final payment as stated in the Contract Documents are of the essence of the Contract.
- 4.02 *Contract Times: Days* 
  - A. The Work will be substantially complete within <u>90</u> days after the date when the Contract Times commence to run as provided in Paragraph 4.01 of the General Conditions, and completed and ready for final payment in accordance with Paragraph 15.06 of the General Conditions within <u>135</u> days after the date when the Contract Times commence to run.
- 4.03 Liquidated Damages
  - A. Contractor and Owner recognize that time is of the essence as stated in Paragraph 4.01 above and that Owner will suffer financial and other losses if the Work is not completed and Milestones not achieved within the Contract Times, as duly modified. The parties also recognize the delays, expense, and difficulties involved in proving, in a legal or arbitration proceeding, the actual loss suffered by Owner if the Work is not completed on time. Accordingly, instead of requiring any such proof, Owner and Contractor agree that as liquidated damages for delay (but not as a penalty):

- 1. Substantial Completion: Contractor shall pay Owner \$500.00 for each day that expires after the time (as duly adjusted pursuant to the Contract) specified above for Substantial Completion, until the Work is substantially complete.
- Completion of Remaining Work: After Substantial Completion, if Contractor shall neglect, refuse, or fail to complete the remaining Work within the Contract Times (as duly adjusted pursuant to the Contract) for completion and readiness for final payment, Contractor shall pay Owner \$500.00 for each day that expires after such time until the Work is completed and ready for final payment.
- 3. Liquidated damages for failing to timely attain Milestones, Substantial Completion, and final completion are not additive, and will not be imposed concurrently.
- B. If Owner recovers liquidated damages for a delay in completion by Contractor, then such liquidated damages are Owner's sole and exclusive remedy for such delay, and Owner is precluded from recovering any other damages, whether actual, direct, excess, or consequential, for such delay, except for special damages (if any) specified in this Agreement.

### **ARTICLE 5—CONTRACT PRICE**

- 5.01 Owner shall pay Contractor for completion of the Work in accordance with the Contract Documents, the amounts that follow, subject to adjustment under the Contract:
  - A. For all Work, at the prices stated in Contractor's Bid, attached hereto as an exhibit, and as adjusted for the actual quantity of any unit price items.
    - 1. Contract price as of the Effective Date of the Contract is \$\_\_\_\_\_
    - 2. For all Unit Price Work, the actual payment amount will be equal to the sum of the extended prices (established for each separately identified item of Unit Price Work by multiplying the unit price times the actual quantity of that item). The extended prices for Unit Price Work set forth as of the Effective Date of the Contract are based on estimated quantities. As provided in Paragraph 13.03 of the General Conditions, estimated quantities are not guaranteed, and determinations of actual quantities and classifications are to be made by Engineer.

### **ARTICLE 6—PAYMENT PROCEDURES**

- 6.01 Submittal and Processing of Payments
  - A. Contractor shall submit Applications for Payment in accordance with Article 15 of the General Conditions. Applications for Payment will be processed by Engineer as provided in the General Conditions.
- 6.02 Progress Payments; Retainage
  - A. Owner shall make progress payments on the basis of Contractor's Applications for Payment on or about the date each month to be established at the preconstruction meeting based on Owner's meeting schedule during performance of the Work as provided in Paragraph 6.02.A.1 below, provided that such Applications for Payment have been submitted in a timely manner and otherwise meet the requirements of the Contract. All such payments will be measured by the Schedule of Values established as provided in the General Conditions (and in the case

of Unit Price Work based on the number of units completed) or, in the event there is no Schedule of Values, as provided elsewhere in the Contract.

- Prior to Final Completion, progress payments will be made in an amount equal to the
  percentage indicated below but, in each case, less the aggregate of payments previously
  made and less such amounts as Owner may withhold, including but not limited to
  liquidated damages, in accordance with the Contract. Retainage indicated below does not
  include withholdings for known incomplete work. Withholdings for known incomplete
  work will be made over and above the retainage values indicated below.
  - a. For Contract Prices less than \$500,000, 90% of the Work completed and/or cost of materials and equipment received and properly stored but not incorporated in the Work (with the balance of 10% being retainage).
  - b. For Contract Prices of \$500,000 or more, 95% of the Work completed and/or cost of materials and equipment received and properly stored but not incorporated in the Work (with the balance of 5% being retainage).

### 6.03 Final Payment

A. Upon final completion and acceptance of the Work, Owner shall pay the remainder of the Contract Price in accordance with Paragraph 15.06 of the General Conditions.

### 6.04 Interest

A. All amounts not paid within 45 days of when due will bear interest at the rate ½ percent accumulated daily, not to exceed 15 percent.

### **ARTICLE 7—CONTRACT DOCUMENTS**

### 7.01 Contents

- A. The Contract Documents consist of all of the following:
  - 1. This Agreement.
  - 2. Bonds:
    - a. Performance bond (together with power of attorney).
    - b. Payment bond (together with power of attorney).
  - 3. General Conditions.
  - 4. Supplementary Conditions.
  - 5. Specifications as listed in the table of contents of the project manual.
  - 6. Drawings (not attached but incorporated by reference) consisting of sheets with each sheet bearing the Project name and dated .
  - 8. Addenda (numbers to , inclusive).
  - 9. Exhibits to this Agreement (enumerated as follows):
    - a. Contractor's Bid.
    - b. Attestations Affidavit.

054

- c. Affidavit Attesting that Public Contract was not Secured through Employment or Payment of Solicitor.
- 10. The following which may be delivered or issued on or after the Effective Date of the Contract and are not attached hereto:
  - a. Notice to Proceed.
  - b. Work Change Directives.
  - c. Change Orders.
  - d. Field Orders.
  - e. Warranty Bond, if any.
- B. The Contract Documents listed in Paragraph 7.01.A are attached to this Agreement (except as expressly noted otherwise above).
- C. There are no Contract Documents other than those listed above in this Article 7.
- D. The Contract Documents may only be amended, modified, or supplemented as provided in the Contract.

### ARTICLE 8—REPRESENTATIONS, CERTIFICATIONS, AND STIPULATIONS

### 8.01 Contractor's Representations

- A. In order to induce Owner to enter into this Contract, Contractor makes the following representations:
  - 1. Contractor has examined and carefully studied the Contract Documents, including Addenda.
  - Contractor has visited the Site, conducted a thorough visual examination of the Site and adjacent areas, and become familiar with the general, local, and Site conditions that may affect cost, progress, and performance of the Work.
  - 3. Contractor is familiar with all Laws and Regulations that may affect cost, progress, and performance of the Work.
  - 4. Contractor has carefully studied the reports of explorations and tests of subsurface conditions at or adjacent to the Site and the drawings of physical conditions relating to existing surface or subsurface structures at the Site that have been identified in the Supplementary Conditions, with respect to the Technical Data in such reports and drawings.
  - Contractor has carefully studied the reports and drawings relating to Hazardous Environmental Conditions, if any, at or adjacent to the Site that have been identified in the Supplementary Conditions, with respect to Technical Data in such reports and drawings.
  - 6. Contractor has considered the information known to Contractor itself; information commonly known to contractors doing business in the locality of the Site; information and observations obtained from visits to the Site; the Contract Documents; and the Technical Data identified in the Supplementary Conditions or by definition, with respect to the effect of such information, observations, and Technical Data on (a) the cost, progress, and

- performance of the Work; (b) the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor; and (c) Contractor's safety precautions and programs.
- 7. Based on the information and observations referred to in the preceding paragraph, Contractor agrees that no further examinations, investigations, explorations, tests, studies, or data are necessary for the performance of the Work at the Contract Price, within the Contract Times, and in accordance with the other terms and conditions of the Contract.
- 8. Contractor is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Contract Documents.
- 9. Contractor has given Engineer written notice of all conflicts, errors, ambiguities, or discrepancies that Contractor has discovered in the Contract Documents, and of discrepancies between Site conditions and the Contract Documents, and the written resolution thereof by Engineer is acceptable to Contractor.
- 10. The Contract Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance and furnishing of the Work.
- 11. Contractor's entry into this Contract constitutes an incontrovertible representation by Contractor that without exception all prices in the Agreement are premised upon performing and furnishing the Work required by the Contract Documents.

### 8.02 Contractor's Certifications

- A. Contractor certifies that it has not engaged in corrupt, fraudulent, collusive, or coercive practices in competing for or in executing the Contract. For the purposes of this Paragraph 8.02:
  - "corrupt practice" means the offering, giving, receiving, or soliciting of anything of value likely to influence the action of a public official in the bidding process or in the Contract execution;
  - "fraudulent practice" means an intentional misrepresentation of facts made (a) to influence the bidding process or the execution of the Contract to the detriment of Owner, (b) to establish Bid or Contract prices at artificial non-competitive levels, or (c) to deprive Owner of the benefits of free and open competition;
  - 3. "collusive practice" means a scheme or arrangement between two or more Bidders, with or without the knowledge of Owner, a purpose of which is to establish Bid prices at artificial, non-competitive levels; and
  - 4. "coercive practice" means harming or threatening to harm, directly or indirectly, persons or their property to influence their participation in the bidding process or affect the execution of the Contract.

### 8.03 Standard General Conditions

A. Owner stipulates that if the General Conditions that are made a part of this Contract are EJCDC® C-700, Standard General Conditions for the Construction Contract (2018), published by the Engineers Joint Contract Documents Committee, and if Owner is the party that has furnished said General Conditions, then Owner has plainly shown all modifications to the

standard wording of such published document to the Contractor, through a process such as highlighting or "track changes" (redline/strikeout), or in the Supplementary Conditions.

IN WITNESS WHEREOF, Owner and Contractor have signed this Agreement. This Agreement will be effective on \_\_\_\_\_ (which is the Effective Date of the Contract). Contractor: Owner: (typed or printed name of organization) (typed or printed name of organization) By: By: (individual's signature) (individual's signature) Date: Date: (date signed) (date signed) Name: Name: (typed or printed) (typed or printed) Title: Title: (typed or printed) (typed or printed) (If Contractor is a corporation, a partnership, or a joint venture, attach evidence of authority to sign.) Attest: (individual's signature) (individual's signature) Title: Title: (typed or printed) (typed or printed) Address for giving notices: Address for giving notices: Designated Representative: Designated Representative: Name: Name: (typed or printed) (typed or printed) Title: Title: (typed or printed) (typed or printed) Address: Address: Phone: Phone: Email: Email: (If Owner is a corporation, attach evidence of authority License No.: to sign. If Owner is a public body, attach evidence of

State:

authority to sign and resolution or other documents

authorizing execution of this Agreement.)

(where applicable)

# **PERFORMANCE BOND**

| Contractor   | Surety   |
|--|--|
| Name:  | Name:  |
| Address (principal place of business):   | Address (principal place of business):             |
|  |  |
|  |  |
| Owner  | Contract   |
| Name:  | Description (name and location):                   |
| Mailing address (principal place of business):   |  |
|  |  |
|  | Contract Price:                                    |
|  | Effective Date of Contract:                        |
| Bond   | The cure bate of contract.                         |
| Bond Amount:   |  |
| Date of Bond:  |  |
| (Date of Bond cannot be earlier than Effective Date of Contract)   |  |
| Modifications to this Bond form:   |  |
| None □ See Paragraph 16  |  |
| Surety and Contractor, intending to be legally bound   |  |
| Performance Bond, do each cause this Performance agent, or representative.   | Bond to be duly executed by an authorized officer, |
| Contractor as Principal  | Surety   |
|  |  |
| (Full formal name of Contractor)   | (Full formal name of Surety) (corporate seal)      |
| Ву:  | Ву:  |
| (Signature)  | (Signature)(Attach Power of Attorney)              |
| Name:  | Name:  |
| (Printed or typed)   | (Printed or typed)                                 |
| Title:   | Title:   |
| Attest:  | Attest:  |
| (Signature)  | (Signature)  |
| Name:  | Name:  |
| (Printed or typed)   | (Printed or typed)                                 |
| Title:   | Title:   |
| Notes: (1) Provide supplemental execution by any additional pa<br>Contractor, Surety, Owner, or other party is considered plural w |  |

- 1. The Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to the Owner for the performance of the Construction Contract, which is incorporated herein by reference.
- 2. If the Contractor performs the Construction Contract, the Surety and the Contractor shall have no obligation under this Bond, except when applicable to participate in a conference as provided in Paragraph 3.
- 3. If there is no Owner Default under the Construction Contract, the Surety's obligation under this Bond will arise after:
  - 3.1. The Owner first provides notice to the Contractor and the Surety that the Owner is considering declaring a Contractor Default. Such notice may indicate whether the Owner is requesting a conference among the Owner, Contractor, and Surety to discuss the Contractor's performance. If the Owner does not request a conference, the Surety may, within five (5) business days after receipt of the Owner's notice, request such a conference. If the Surety timely requests a conference, the Owner shall attend. Unless the Owner agrees otherwise, any conference requested under this Paragraph 3.1 will be held within ten (10) business days of the Surety's receipt of the Owner's notice. If the Owner, the Contractor, and the Surety agree, the Contractor shall be allowed a reasonable time to perform the Construction Contract, but such an agreement does not waive the Owner's right, if any, subsequently to declare a Contractor Default;
  - 3.2. The Owner declares a Contractor Default, terminates the Construction Contract and notifies the Surety; and
  - 3.3. The Owner has agreed to pay the Balance of the Contract Price in accordance with the terms of the Construction Contract to the Surety or to a contractor selected to perform the Construction Contract.
- 4. Failure on the part of the Owner to comply with the notice requirement in Paragraph 3.1 does not constitute a failure to comply with a condition precedent to the Surety's obligations, or release the Surety from its obligations, except to the extent the Surety demonstrates actual prejudice.
- 5. When the Owner has satisfied the conditions of Paragraph 3, the Surety shall promptly and at the Surety's expense take one of the following actions:
  - 5.1. Arrange for the Contractor, with the consent of the Owner, to perform and complete the Construction Contract;
  - 5.2. Undertake to perform and complete the Construction Contract itself, through its agents or independent contractors;
  - 5.3. Obtain bids or negotiated proposals from qualified contractors acceptable to the Owner for a contract for performance and completion of the Construction Contract, arrange for a contract to be prepared for execution by the Owner and a contractor selected with the Owners concurrence, to be secured with performance and payment bonds executed by a qualified surety equivalent to the bonds issued on the Construction Contract, and pay to the Owner the amount of damages as described in Paragraph 7 in excess of the Balance of the Contract Price incurred by the Owner as a result of the Contractor Default; or
  - 5.4. Waive its right to perform and complete, arrange for completion, or obtain a new contractor, and with reasonable promptness under the circumstances:

- 5.4.1 After investigation, determine the amount for which it may be liable to the Owner and, as soon as practicable after the amount is determined, make payment to the Owner; or
- 5.4.2 Deny liability in whole or in part and notify the Owner, citing the reasons for denial.
- 6. If the Surety does not proceed as provided in Paragraph 5 with reasonable promptness, the Surety shall be deemed to be in default on this Bond seven days after receipt of an additional written notice from the Owner to the Surety demanding that the Surety perform its obligations under this Bond, and the Owner shall be entitled to enforce any remedy available to the Owner. If the Surety proceeds as provided in Paragraph 5.4, and the Owner refuses the payment, or the Surety has denied liability, in whole or in part, without further notice, the Owner shall be entitled to enforce any remedy available to the Owner.
- 7. If the Surety elects to act under Paragraph 5.1, 5.2, or 5.3, then the responsibilities of the Surety to the Owner will not be greater than those of the Contractor under the Construction Contract, and the responsibilities of the Owner to the Surety will not be greater than those of the Owner under the Construction Contract. Subject to the commitment by the Owner to pay the Balance of the Contract Price, the Surety is obligated, without duplication for:
  - 7.1. the responsibilities of the Contractor for correction of defective work and completion of the Construction Contract;
  - 7.2. additional legal, design professional, and delay costs resulting from the Contractor's Default, and resulting from the actions or failure to act of the Surety under Paragraph 5; and
  - 7.3. liquidated damages, or if no liquidated damages are specified in the Construction Contract, actual damages caused by delayed performance or non-performance of the Contractor.
- 8. If the Surety elects to act under Paragraph 5.1, 5.3, or 5.4, the Surety's liability is limited to the amount of this Bond.
- 9. The Surety shall not be liable to the Owner or others for obligations of the Contractor that are unrelated to the Construction Contract, and the Balance of the Contract Price will not be reduced or set off on account of any such unrelated obligations. No right of action will accrue on this Bond to any person or entity other than the Owner or its heirs, executors, administrators, successors, and assigns.
- 10. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders, and other obligations.
- 11. Any proceeding, legal or equitable, under this Bond must be instituted in any court of competent jurisdiction in the location in which the work or part of the work is located and must be instituted within two years after a declaration of Contractor Default or within two years after the Contractor ceased working or within two years after the Surety refuses or fails to perform its obligations under this Bond, whichever occurs first. If the provisions of this paragraph are void or prohibited by law, the minimum periods of limitations available to sureties as a defense in the jurisdiction of the suit will be applicable.
- 12. Notice to the Surety, the Owner, or the Contractor must be mailed or delivered to the address shown on the page on which their signature appears.
- 13. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement will be deemed deleted therefrom and provisions conforming to such

statutory or other legal requirement will be deemed incorporated herein. When so furnished, the intent is that this Bond will be construed as a statutory bond and not as a common law bond.

### 14. Definitions

- 14.1. Balance of the Contract Price—The total amount payable by the Owner to the Contractor under the Construction Contract after all proper adjustments have been made including allowance for the Contractor for any amounts received or to be received by the Owner in settlement of insurance or other claims for damages to which the Contractor is entitled, reduced by all valid and proper payments made to or on behalf of the Contractor under the Construction Contract.
- 14.2. *Construction Contract*—The agreement between the Owner and Contractor identified on the cover page, including all Contract Documents and changes made to the agreement and the Contract Documents.
- 14.3. *Contractor Default*—Failure of the Contractor, which has not been remedied or waived, to perform or otherwise to comply with a material term of the Construction Contract.
- 14.4. Owner Default—Failure of the Owner, which has not been remedied or waived, to pay the Contractor as required under the Construction Contract or to perform and complete or comply with the other material terms of the Construction Contract.
- 14.5. *Contract Documents*—All the documents that comprise the agreement between the Owner and Contractor.
- 15. If this Bond is issued for an agreement between a contractor and subcontractor, the term Contractor in this Bond will be deemed to be Subcontractor and the term Owner will be deemed to be Contractor.
- 16. Modifications to this Bond are as follows: None

# **PAYMENT BOND**

| Contractor   | Surety   |
|--|--|
| Name:  | Name:  |
| Address (principal place of business):   | Address (principal place of business):                   |
|  |  |
|  |  |
| Owner  | Contract   |
| Name:  | Description (name and location):                         |
| Mailing address (principal place of business):   | ,  |
|  |  |
|  | Contract Price:  |
|  | Effective Date of Contract:                              |
| Bond   |  |
| Bond Amount:   |  |
| Date of Bond:  |  |
| (Date of Bond cannot be earlier than Effective Date of Contract)   |  |
| Modifications to this Bond form:   |  |
| <ul><li>☑ None ☐ See Paragraph 18</li><li>Surety and Contractor, intending to be legally bound</li></ul> | d haraby subject to the terms set forth in this          |
|  | o be duly executed by an authorized officer, agent, or   |
| representative.  | o se daily exceded by an additionized officer, agent, or |
| Contractor as Principal  | Surety   |
|  |  |
| (Full formal name of Contractor)   | (Full formal name of Surety) (corporate seal)            |
| Ву:  | Ву:  |
| (Signature)  | (Signature)(Attach Power of Attorney)                    |
| Name:  | Name:  |
| (Printed or typed)   | (Printed or typed)                                       |
| Title:   | Title:   |
| Attest:  | Attest:  |
| (Signature)  | (Signature)  |
| Name:  | Name:  |
| (Printed or typed)   | (Printed or typed)                                       |
| Title:   | Title:   |
| Notes: (1) Provide supplemental execution by any additional po   |  |
| Contractor, Surety, Owner, or other party is considered plural v   | итете аррпсавте.   |

- 1. The Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to the Owner to pay for labor, materials, and equipment furnished for use in the performance of the Construction Contract, which is incorporated herein by reference, subject to the following terms.
- 2. If the Contractor promptly makes payment of all sums due to Claimants, and defends, indemnifies, and holds harmless the Owner from claims, demands, liens, or suits by any person or entity seeking payment for labor, materials, or equipment furnished for use in the performance of the Construction Contract, then the Surety and the Contractor shall have no obligation under this Bond.
- 3. If there is no Owner Default under the Construction Contract, the Surety's obligation to the Owner under this Bond will arise after the Owner has promptly notified the Contractor and the Surety (at the address described in Paragraph 13) of claims, demands, liens, or suits against the Owner or the Owner's property by any person or entity seeking payment for labor, materials, or equipment furnished for use in the performance of the Construction Contract, and tendered defense of such claims, demands, liens, or suits to the Contractor and the Surety.
- 4. When the Owner has satisfied the conditions in Paragraph 3, the Surety shall promptly and at the Surety's expense defend, indemnify, and hold harmless the Owner against a duly tendered claim, demand, lien, or suit.
- 5. The Surety's obligations to a Claimant under this Bond will arise after the following:
  - 5.1. Claimants who do not have a direct contract with the Contractor
    - 5.1.1. have furnished a written notice of non-payment to the Contractor, stating with substantial accuracy the amount claimed and the name of the party to whom the materials were, or equipment was, furnished or supplied or for whom the labor was done or performed, within ninety (90) days after having last performed labor or last furnished materials or equipment included in the Claim; and
    - 5.1.2. have sent a Claim to the Surety (at the address described in Paragraph 13).
  - 5.2. Claimants who are employed by or have a direct contract with the Contractor have sent a Claim to the Surety (at the address described in Paragraph 13).
- 6. If a notice of non-payment required by Paragraph 5.1.1 is given by the Owner to the Contractor, that is sufficient to satisfy a Claimant's obligation to furnish a written notice of non-payment under Paragraph 5.1.1.
- 7. When a Claimant has satisfied the conditions of Paragraph 5.1 or 5.2, whichever is applicable, the Surety shall promptly and at the Surety's expense take the following actions:
  - 7.1. Send an answer to the Claimant, with a copy to the Owner, within sixty (60) days after receipt of the Claim, stating the amounts that are undisputed and the basis for challenging any amounts that are disputed; and
  - 7.2. Pay or arrange for payment of any undisputed amounts.
  - 7.3. The Surety's failure to discharge its obligations under Paragraph 7.1 or 7.2 will not be deemed to constitute a waiver of defenses the Surety or Contractor may have or acquire as to a Claim, except as to undisputed amounts for which the Surety and Claimant have reached agreement. If, however, the Surety fails to discharge its obligations under Paragraph 7.1 or 7.2, the Surety shall indemnify the Claimant for the reasonable attorney's fees the Claimant incurs thereafter to recover any sums found to be due and owing to the Claimant.

- 8. The Surety's total obligation will not exceed the amount of this Bond, plus the amount of reasonable attorney's fees provided under Paragraph 7.3, and the amount of this Bond will be credited for any payments made in good faith by the Surety.
- 9. Amounts owed by the Owner to the Contractor under the Construction Contract will be used for the performance of the Construction Contract and to satisfy claims, if any, under any construction performance bond. By the Contractor furnishing and the Owner accepting this Bond, they agree that all funds earned by the Contractor in the performance of the Construction Contract are dedicated to satisfying obligations of the Contractor and Surety under this Bond, subject to the Owner's priority to use the funds for the completion of the work.
- 10. The Surety shall not be liable to the Owner, Claimants, or others for obligations of the Contractor that are unrelated to the Construction Contract. The Owner shall not be liable for the payment of any costs or expenses of any Claimant under this Bond, and shall have under this Bond no obligation to make payments to or give notice on behalf of Claimants, or otherwise have any obligations to Claimants under this Bond.
- 11. The Surety hereby waives notice of any change, including changes of time, to the Construction Contract or to related subcontracts, purchase orders, and other obligations.
- 12. No suit or action will be commenced by a Claimant under this Bond other than in a court of competent jurisdiction in the state in which the project that is the subject of the Construction Contract is located or after the expiration of one year from the date (1) on which the Claimant sent a Claim to the Surety pursuant to Paragraph 5.1.2 or 5.2, or (2) on which the last labor or service was performed by anyone or the last materials or equipment were furnished by anyone under the Construction Contract, whichever of (1) or (2) first occurs. If the provisions of this paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit will be applicable.
- 13. Notice and Claims to the Surety, the Owner, or the Contractor must be mailed or delivered to the address shown on the page on which their signature appears. Actual receipt of notice or Claims, however accomplished, will be sufficient compliance as of the date received.
- 14. When this Bond has been furnished to comply with a statutory or other legal requirement in the location where the construction was to be performed, any provision in this Bond conflicting with said statutory or legal requirement will be deemed deleted here from and provisions conforming to such statutory or other legal requirement will be deemed incorporated herein. When so furnished, the intent is that this Bond will be construed as a statutory bond and not as a common law bond.
- 15. Upon requests by any person or entity appearing to be a potential beneficiary of this Bond, the Contractor and Owner shall promptly furnish a copy of this Bond or shall permit a copy to be made.

### 16. Definitions

- 16.1. *Claim*—A written statement by the Claimant including at a minimum:
  - 16.1.1. The name of the Claimant;
  - 16.1.2. The name of the person for whom the labor was done, or materials or equipment furnished;
  - 16.1.3. A copy of the agreement or purchase order pursuant to which labor, materials, or equipment was furnished for use in the performance of the Construction Contract;
  - 16.1.4. A brief description of the labor, materials, or equipment furnished;

- 16.1.5. The date on which the Claimant last performed labor or last furnished materials or equipment for use in the performance of the Construction Contract;
- 16.1.6. The total amount earned by the Claimant for labor, materials, or equipment furnished as of the date of the Claim;
- 16.1.7. The total amount of previous payments received by the Claimant; and
- 16.1.8. The total amount due and unpaid to the Claimant for labor, materials, or equipment furnished as of the date of the Claim.
- 16.2. Claimant—An individual or entity having a direct contract with the Contractor or with a subcontractor of the Contractor to furnish labor, materials, or equipment for use in the performance of the Construction Contract. The term Claimant also includes any individual or entity that has rightfully asserted a claim under an applicable mechanic's lien or similar statute against the real property upon which the Project is located. The intent of this Bond is to include without limitation in the terms of "labor, materials, or equipment" that part of the water, gas, power, light, heat, oil, gasoline, telephone service, or rental equipment used in the Construction Contract, architectural and engineering services required for performance of the work of the Contractor and the Contractor's subcontractors, and all other items for which a mechanic's lien may be asserted in the jurisdiction where the labor, materials, or equipment were furnished.
- 16.3. Construction Contract—The agreement between the Owner and Contractor identified on the cover page, including all Contract Documents and all changes made to the agreement and the Contract Documents.
- 16.4. Owner Default—Failure of the Owner, which has not been remedied or waived, to pay the Contractor as required under the Construction Contract or to perform and complete or comply with the other material terms of the Construction Contract.
- 16.5. *Contract Documents*—All the documents that comprise the agreement between the Owner and Contractor.
- 17. If this Bond is issued for an agreement between a contractor and subcontractor, the term Contractor in this Bond will be deemed to be Subcontractor and the term Owner will be deemed to be Contractor.
- 18. Modifications to this Bond are as follows: None

# **NOTICE TO PROCEED**

| Owner:                       |  | Owner's Project No.:   |
|------------------------------|--|--|
| Engineer:                    |  | Engineer's Project No.:  |
| Contractor:                  |  |  |
| Project:                     |  |  |
| Effective Date of 0          | Contract:  |  |
| •                            | ifies Contractor that the Contract Tim<br>to enter a date., pursuant to Paragrap | nes under the above Contract will commence to oh 4.01 of the General Conditions.   |
| •                            | ractor shall start performing its oblig Site prior to such date.                 | ations under the Contract Documents. No Work   |
| In accordance with           | the Agreement:   |  |
| commencemer<br>the number of | nt of the Contract Times, resulting in   | tion is from the date stated above for the a date for Substantial Completion of; and yment is from the commencement date of for final payment of |
| Before starting any          | Work at the Site, Contractor must co   | mply with the following:   |
| Owner:                       |  | <u> </u>   |
| By (signature):              |  | <u> </u>   |
| Name (printed):              |  | <u> </u>   |
| Title:                       |  | <u> </u>   |
| Date Issued:                 |  | <u> </u>   |
| Copy: Engineer               |  |  |

This page intentionally left blank

# **WORK CHANGE DIRECTIVE NO.: [Number of Work Change Directive]**

| Owner:               |   | Owner's Project No.:  |
|----------------------|---|---|
| Engineer:            |   | Engineer's Project No.:   |
| Contractor:          |   | Contractor's Project No.:   |
| Project:             |   |   |
| Date Issued:         | Effe  | ective Date of Work Change Directive:                             |
| Contractor is direct | ted to proceed promptly witl                            | n the following change(s):  |
| Description:         |   |   |
| [Description of      | the change to the Work]                                 |   |
| Attachments:         |   |   |
| [List document       | ts related to the change to t                           | he Work]  |
| Purpose for the Wo   | ork Change Directive:                                   |   |
| [Describe the p      | ourpose for the change to th                            | e Work]   |
| •                    | ed promptly with the Work of<br>Time, is issued due to: | described herein, prior to agreeing to change in Contract         |
| Notes to User—Ch     | eck one or both of the follow                           | wing  |
| ☐ Non-agreement      | on pricing of proposed chan                             | ge. $\square$ Necessity to proceed for schedule or other reasons. |
| Estimated Change     | in Contract Price and Contrac                           | ct Times (non-binding, preliminary):                              |
| Contract Price:      | \$  | [increase] [decrease] [not yet estimated].                        |
| Contract Time:       | days  | [increase] [decrease] [not yet estimated].                        |
| Basis of estimated   | change in Contract Price:                               |   |
| ☐ Lump Sum ☐ Ui      | nit Price $\square$ Cost of the Work                    | ☐ Other   |
| Recomme              | ended by Engineer                                       | Authorized by Owner   |
| Necomme              | ended by Engineer                                       | Authorized by Owner   |
| Ву:                  |   |   |
| Title:               |   |   |
| Date:                |   |   |
|                      |   | <del></del>   |

EJCDC® C-940, Work Change Directive.

This page intentionally left blank

# **CHANGE ORDER NO.:**

| Owner<br>Engine<br>Contra<br>Project<br>Contra<br>Date Is | er: Meyer & Associates, Inc. ctor: : ct Date: Date | Owner's Project No.: Engineer's Project No.: Contractor's Project No.: of Notice to Proceed: tive Date of Change Order: |
|---|--|---|
|   | tract is modified as follows upon execution c      | f this Change Order:  |
| Descript  | ion:   |   |
| Attachm   | nents:   |   |
|   | Change in Contract Price                           | Change in Contract Times  |
| Origina   | Contract Price:                                    | Original Contract Times: Substantial Completion:  |
| \$  |  | Ready for final payment:  |
| [Increa   | se] [Decrease] from previously approved Change     | [Increase] [Decrease] from previously approved  |
|   | No. 1 to No. [Number of previous Change            | Change Orders No.1 to No. [Number of previous   |
| Order]:   |  | Change Order]: Substantial Completion:  |
| \$  |  | Ready for final payment:  |
|   | t Price prior to this Change Order:                | Contract Times prior to this Change Order:  |
|   |  | Substantial Completion:   |
| \$  |  | Ready for final payment:  |
| [Increa   | se] [Decrease] this Change Order:                  | [Increase] [Decrease] this Change Order:  |
| ے   |  | Substantial Completion:   |
| \$  | t Daire in comment in this Change Order            | Ready for final payment:  |
| Contrac   | t Price incorporating this Change Order:           | Contract Times with all approved Change Orders: Substantial Completion:   |
| \$  |  | Ready for final payment:  |
| By:   | Recommended by Engineer (if required)              | Accepted by Contractor  |
| Title:  |  |   |
| Date:   |  |   |
|   | Authorized by Owner                                | Approved by Funding Agency (if applicable)  |
| Ву:   |  |   |
| Title:  |  |   |
|   |  |   |
| Date:   |  |   |

EJCDC® C-941, Change Order.

This page intentionally left blank

# FIELD ORDER NO.: [Number of Field Order]

| Owner: Engineer: Contractor: Project:        | Owner's Project No.: Engineer's Project No.: Contractor's Project No.:   |
|--|--|
| Date Issued:                                 | Effective Date of Field Order:   |
| accordance with Paragraph 11.04 of the Gener | form the Work described in this Field Order, issued in ral Conditions, for minor changes in the Work without Contractor considers that a change in Contract Price or roposal before proceeding with this Work. |
| Reference:                                   |  |
| Specification Section(s):                    |  |
| Drawing(s) / Details (s):                    |  |
| Description:                                 |  |
| [Description of the change to the Work]      |  |
|  |  |
| Attachments:                                 |  |
| [List documents supporting change]           |  |
| Issued by Engineer                           |  |
| Ву:  |  |
| Title:                                       |  |
| Date:  |  |
|  |  |

EJCDC® C-942, Field Order.

This page intentionally left blank

# **CERTIFICATE OF SUBSTANTIAL COMPLETION**

| Owner:  |  | Owner's Project No.:   |
|---|--|--|
| Engineer:   | Meyer & Associates, Inc.   | Engineer's Project No.:  |
| Contractor:   |  | Contractor's Project No.:  |
| Project:  |  |  |
| This $\square$ Preliminary  | ☐ Final Certificate of Substantial Comp  | letion applies to:   |
| $\square$ All Work $\square$  | The following specified portions of the \  | Nork:  |
| Date of Substantial   | Completion:  |  |
| Contractor, and Eng<br>the Work or portion<br>Contract pertaining<br>of Substantial Com | gineer, and found to be substantially con<br>n thereof designated above is hereby es   | red by authorized representatives of Owner, implete. The Date of Substantial Completion of tablished, subject to the provisions of the Substantial Completion in the final Certificate e contractual correction period and |
| inclusive, and the fa   | s to be completed or corrected is attach<br>ailure to include any items on such list d<br>blete all Work in accordance with the Co | • •  |
|   | ntractual responsibilities recorded in thi<br>er and Contractor; see Paragraph 15.03.  | s Certificate should be the product of mutual D of the General Conditions.   |
| utilities, insurance,   |  | urity, operation, safety, maintenance, heat, cupancy of the Work must be as provided in  |
| Amendments to Ov  | vner's Responsibilities: $oxtimes$ None $oxtimes$ As fo  | ollows:  |
| Amendments to Co  | ntractor's Responsibilities: $oxtimes$ None $oxtimes$ ,  | As follows:  |
| The following docu  | ments are attached to and made a part  | of this Certificate:   |
| This Certificate doe  | _  | not in accordance with the Contract complete the Work in accordance with the   |
| Recommended by  | Engineer:  |  |
| By (signature):   |  |  |
| Name (printed):   |  |  |
| Title:  |  |  |

| Accepted by Cont | actor: |  |
|------------------|--------|--|
| By (signature):  |        |  |
| Name (printed):  |        |  |
| Title:           |        |  |
|                  |        |  |
| Accepted by Own  | er:    |  |
| By (signature):  |        |  |
| Name (printed):  |        |  |
| Title:           |        |  |
|                  |        |  |

## NOTICE OF ACCEPTABILITY OF WORK

|                                |  | NOTICE OF ACCEPTABILITY OF WORK   |
|--------------------------------|--|---|
| Ow                             | ner:   | Owner's Project No.:  |
| Eng                            | gineer:  | Engineer's Project No.:   |
| Cor                            | ntractor:  | Contractor's Project No.:   |
| Pro                            | ject:  |   |
| Not                            | tice Date:   | Effective Date of the Construction Contract:  |
| to Co<br>is ac<br>("Co<br>date | ontractor, and that the Work for<br>cceptable, expressly subject to<br>ntract Documents") and of to<br>d ("Owner-Engineer of<br>essly subject to the following | to the Owner and Contractor that Engineer recommends final payment furnished and performed by Contractor under the Construction Contract to the provisions of the Construction Contract's Contract Documents he Agreement between Owner and Engineer for Professional Services Agreement"). This Notice of Acceptability of Work (Notice) is made a terms and conditions to which all who receive and rely on said Notice |
| 1.                             |  | red with the skill and care ordinarily used by members of the engineering similar conditions at the same time and in the same locality.   |
| 2.                             | This Notice reflects and is a  | n expression of the Engineer's professional opinion.  |
| 3.                             | This Notice has been prepa<br>the Notice Date.   | ared to the best of Engineer's knowledge, information, and belief as of   |
| 4.                             | employed by Owner to perf<br>of the Contractor's Work) u<br>within Engineer's knowledg   | y on and expressly limited by the scope of services Engineer has been<br>form or furnish during construction of the Project (including observation<br>inder the Owner-Engineer Agreement, and applies only to facts that are<br>ge or could reasonably have been ascertained by Engineer as a result of<br>polities specifically assigned to Engineer under such Owner-Engineer   |
| 5.                             | Contract, an acceptance of<br>but not limited to defect<br>responsibility for any failu  | thee or warranty of Contractor's performance under the Construction Work that is not in accordance with the Contract Documents, including live Work discovered after final inspection, nor an assumption of ure of Contractor to furnish and perform the Work thereunder in act Documents, or to otherwise comply with the Contract Documents or arantees specified therein.  |
| 6.                             |  | eve Contractor of any surviving obligations under the Construction<br>Owner's reservations of rights with respect to completion and final   |
| Engi                           | neer   |   |
|                                | By (signature):  |   |
|                                |  |   |
|                                | Title:   |   |

This page intentionally left blank



# Meyer & Associates, Inc. Consulting Engineers

| Project:         Pay Application No.:         Pay Application No.:         To:           Owner:         Contractor:         Project No.:         Adjusted Contract Amount:         Adjusted Contract Amount:         Approved Change Orders: | _                    |                        |     |
|--|----------------------|------------------------|-----|
| From:  Adjusted Contract Amount:  Project No.: Approved Change Orders  | Project:             | Pay Application No.:   |     |
| Adjusted Contract Amount:  | Owner:               | From:                  | To: |
| Adjusted Contract Amount:  | Contractor:          | Project No.:           |     |
|  | Original Contract Ar | Approved Change Orders | ::  |

|  | This Period | Total to Date |
|--|-------------|---------------|
| Total Value of Work Completed:   | \$0.00      | \$0.00        |
| Retainage: 10%   | \$0.00      | \$0.00        |
| Materials On Hand:   | \$0.00      | \$0.00        |
| Deductions:  | \$0.00      | \$0.00        |
| Net Earned to Date:  | \$0.00      | \$0.00        |
| Less Amount Paid to Date:  |             |               |
| Net Amount Due This Application:   | \$0.00      | \$0.00        |
| THE PROPERTY OF THE PROPERTY O |             |               |

applied to discharge in full all obligations of CONTRACTOR incurred in connection with Work covered by all prior Applications for Payment inclusive; and (2) Title to all materials and equipment incorporated in said Work or otherwise listed in or covered by this Application for Payment will pass to OWNER at time of payment free and clear of all liens, claims, security interests and encumbrances (except such as covered by Bond acceptable to OWNER). (3) Partial pay amounts and associated quantities, cash allowances and stated allowance work are subject to reconciliation by Engineer in accordance with controlling provisions of the contract. (4) Approval of interim payment application does not in anyway constitute approval of time (days) and schedule. The undersigned CONTRACTOR certifies that (1) all previous progress payments received from OWNER on account of Work done under the Contract referred to above have been

| ŏ     | Certified Correct for Payment | Y     | Recommended for Payment                  |       | Approved for Payment |   |
|-------|-------------------------------|-------|--|-------|----------------------|---|
|       | Contractor                    |       | ivieyei & Associates, III.c.<br>Engineer |       | Owner                |   |
| By:   |                               | By:   |  | By:   |                      | 1 |
| Date: |                               | Date: |  | Date: |                      |   |

This page intentionally left blank

Meyer & Associates, Inc. Consulting Engineers

|                     |  | Completed To Date       | Quantity Total Cost or % (9) (8) | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 |  |
|---------------------|--|-------------------------|----------------------------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|--|
| to                  |  | Completed This Period C | Total Cost Qu                    | \$0.00 | \$0.00 | \$0.00 | 00.0\$ | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | 9  |
|                     |  | Completed               | Quantity<br>or %<br>(6)          |        |        |        |        |        |        |        |        |        |  |
| for Period          | ONLY<br>act.   | nnt                     | Total Cost<br>Per Unit<br>(5)    | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | \$0.00 | 0  |
|                     | CONTRACT (<br>ntract only.<br>er original contr<br>riginal contract.   | Contract Amount         | Cost per<br>Unit<br>(4)          |        |        |        |        |        |        |        |        |        |  |
|                     | DER ORIGINAL the original cor orm. this period unde  |                         | Quantity<br>(3)                  |        |        |        |        |        |        |        |        |        |  |
| Pay Application No. | COST OF WORK COMPLETED TO DATE UNDER ORIGINAL CONTRACT ONLY Entries must be limited to work and costs under the original contract only. Columns (1) through (5). Enter data from Bid Form. Columns (6) and (7). Show all work completed this period under original contract. Columns (8) and (9). Show all work completed to date under original contract. |                         | Description of Items<br>(2)      |        |        |        |        |        |        |        |        |        | i di |
|                     | <del>L.</del>  |                         | Item No.<br>(1)                  |        |        |        |        |        |        |        |        |        |  |

Meyer & Associates, Inc. Consulting Engineers

|          | Pay Application No.  | 0   | <u> </u>   | for Period   | 1/0/00  | t   | 1/0/00                        |  |  |                          |                                 |
|----------|--|---|--|--|---|---|-------------------------------|--|--|--------------------------|---------------------------------|
| 7        | INSTRUCTIONS FOR MATERIAL INVENTORY Entries must be limited to materials and costs under the original contract only. Column (3): Price of materials per invoices. Column (4): Quantity received the Column (5): Cumulative quantity received. Column (6): Column 3 x 4. Column Column (9): Cumulative quantity used. Column (10): Column 3 x 8. Column | IATERIAL INV<br>o materials ar<br>terials per inv<br>quantity rece<br>quantity usec | /ENTORY Id costs unde oices. Colum ived. Columr Colum!. Colum! | r the original nn (4): Quan (6): Column 3): Column 3): | DRY s under the original contract only. Column (4): Quantity received this month with corresponding invoices attached. Column (6): Column 3 x 4. Column (7): Column 3 x 5. Column (8): Quantity used in current month. Imm (10): Column 3 x 8. Column (11): Column 3 x 9. Column (12): Column 7 - 11. | his month wir<br>nn (7): Colun<br>1 (11): Colun | th correspond<br>mn 3 x 5. Co | ding invoices<br>blumn (8): Q<br>umn (12): C | attached.<br>uantity used i<br>olumn 7 - 11. | n current mo             | nth.                            |
| Item No. | Description of Items<br>(2)  | Material<br>Unit Price<br>(3)   | Received Current (Quantity)                                    | Received Total (Quantity)                              | Received<br>Current<br>(Value)  | Received<br>Total<br>(Value)                    | Used<br>Current<br>(Quantity) | Used<br>Total<br>(Quantity)                  | Used<br>Current<br>(Value)                   | Used<br>Total<br>(Value) | Current In-<br>Stock<br>(Value) |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          |  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          | SUBTOTAL   |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          | Sales Tax  | 9.00%   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          | Retainage  | 10.00%  |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |
|          | TOTAL  |   |  |  | \$0.00  | \$0.00  |                               |  | \$0.00                                       | \$0.00                   | \$0.00                          |

**GENERAL CONDITIONS** 

This page intentionally left blank

# STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

# **Prepared By**









# **Endorsed By**





# Copyright<sup>©</sup> 2018

National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882

www.nspe.org

American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474

www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723

www.asce.org

The copyright for this EJCDC document is owned jointly by the three sponsoring organizations listed above. The National Society of Professional Engineers is the Copyright Administrator for the EJCDC documents; please direct all inquiries regarding EJCDC copyrights to NSPE.

NOTE: EJCDC publications may be purchased at <a href="www.ejcdc.org">www.ejcdc.org</a>, or from any of the sponsoring organizations above.

# STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

#### **TABLE OF CONTENTS**

|            |   | Page |
|------------|---|------|
| Article 1- | – Definitions and Terminology   | 1    |
| 1.01       | Defined Terms   | 1    |
| 1.02       | Terminology   | 6    |
| Article 2- | Preliminary Matters   | 7    |
| 2.01       | Delivery of Performance and Payment Bonds; Evidence of Insurance              | 7    |
| 2.02       | Copies of Documents   | 7    |
| 2.03       | Before Starting Construction  | 7    |
| 2.04       | Preconstruction Conference; Designation of Authorized Representatives         | 8    |
| 2.05       | Acceptance of Schedules   | 8    |
| 2.06       | Electronic Transmittals   | 8    |
| Article 3- | -Contract Documents: Intent, Requirements, Reuse                              | 9    |
| 3.01       | Intent  | 9    |
| 3.02       | Reference Standards   | 9    |
| 3.03       | Reporting and Resolving Discrepancies   | 10   |
| 3.04       | Requirements of the Contract Documents  | 10   |
| 3.05       | Reuse of Documents  | 11   |
| Article 4- | -Commencement and Progress of the Work  | 11   |
| 4.01       | Commencement of Contract Times; Notice to Proceed                             | 11   |
| 4.02       | Starting the Work   | 11   |
| 4.03       | Reference Points  | 11   |
| 4.04       | Progress Schedule   | 12   |
| 4.05       | Delays in Contractor's Progress   | 12   |
| Article 5- | -Site; Subsurface and Physical Conditions; Hazardous Environmental Conditions | 13   |
| 5.01       | Availability of Lands   | 13   |
| 5.02       | Use of Site and Other Areas   | 14   |
| 5.03       | Subsurface and Physical Conditions  | 15   |
|            |   |      |

| 5.04      | Differing Subsurface or Physical Conditions            | 16 |
|-----------|--|----|
| 5.05      | Underground Facilities                                 | 17 |
| 5.06      | Hazardous Environmental Conditions at Site             | 19 |
| Article 6 | —Bonds and Insurance                                   | 21 |
| 6.01      | Performance, Payment, and Other Bonds                  | 21 |
| 6.02      | Insurance—General Provisions                           | 22 |
| 6.03      | Contractor's Insurance                                 | 24 |
| 6.04      | Builder's Risk and Other Property Insurance            | 25 |
| 6.05      | Property Losses; Subrogation                           | 25 |
| 6.06      | Receipt and Application of Property Insurance Proceeds | 27 |
| Article 7 | —Contractor's Responsibilities                         | 27 |
| 7.01      | Contractor's Means and Methods of Construction         | 27 |
| 7.02      | Supervision and Superintendence                        | 27 |
| 7.03      | Labor; Working Hours                                   | 27 |
| 7.04      | Services, Materials, and Equipment                     | 28 |
| 7.05      | "Or Equals"  | 28 |
| 7.06      | Substitutes  | 29 |
| 7.07      | Concerning Subcontractors and Suppliers                | 31 |
| 7.08      | Patent Fees and Royalties                              | 32 |
| 7.09      | Permits  | 33 |
| 7.10      | Taxes  | 33 |
| 7.11      | Laws and Regulations                                   | 33 |
| 7.12      | Record Documents                                       | 33 |
| 7.13      | Safety and Protection                                  | 34 |
| 7.14      | Hazard Communication Programs                          | 35 |
| 7.15      | Emergencies  | 35 |
| 7.16      | Submittals   | 35 |
| 7.17      | Contractor's General Warranty and Guarantee            | 38 |
| 7.18      | Indemnification  | 39 |
| 7.19      | Delegation of Professional Design Services             | 39 |
| Article 8 | —Other Work at the Site                                | 40 |
| 8.01      | Other Work   | 40 |
| 8.02      | Coordination   | 41 |

| 8.03       | Legal Relationships   | 41 |
|------------|---|----|
| Article 9  | —Owner's Responsibilities   | 42 |
| 9.01       | Communications to Contractor  | 42 |
| 9.02       | Replacement of Engineer   | 42 |
| 9.03       | Furnish Data  | 42 |
| 9.04       | Pay When Due  | 42 |
| 9.05       | Lands and Easements; Reports, Tests, and Drawings                         | 43 |
| 9.06       | Insurance   | 43 |
| 9.07       | Change Orders   | 43 |
| 9.08       | Inspections, Tests, and Approvals   | 43 |
| 9.09       | Limitations on Owner's Responsibilities                                   | 43 |
| 9.10       | Undisclosed Hazardous Environmental Condition                             | 43 |
| 9.11       | Evidence of Financial Arrangements  | 43 |
| 9.12       | Safety Programs   | 43 |
| Article 10 | O—Engineer's Status During Construction                                   | 44 |
| 10.01      | Owner's Representative  | 44 |
| 10.02      | Visits to Site  | 44 |
| 10.03      | Resident Project Representative   | 44 |
| 10.04      | Engineer's Authority  | 44 |
| 10.05      | Determinations for Unit Price Work  | 45 |
| 10.06      | Decisions on Requirements of Contract Documents and Acceptability of Work | 45 |
| 10.07      | Limitations on Engineer's Authority and Responsibilities                  | 45 |
| 10.08      | Compliance with Safety Program  | 45 |
| Article 1  | 1—Changes to the Contract   | 46 |
| 11.01      | Amending and Supplementing the Contract                                   | 46 |
| 11.02      | Change Orders   | 46 |
| 11.03      | Work Change Directives  | 46 |
| 11.04      | Field Orders  | 47 |
| 11.05      | Owner-Authorized Changes in the Work                                      | 47 |
| 11.06      | Unauthorized Changes in the Work  | 47 |
| 11.07      | Change of Contract Price  | 47 |
| 11.08      | Change of Contract Times  | 49 |
| 11.09      | Change Proposals  | 49 |

| 11.10       | Notification to Surety   | 50 |
|-------------|--|----|
| Article 12- | -Claims  | 50 |
| 12.01       | Claims   | 50 |
| Article 13- | -Cost of the Work; Allowances; Unit Price Work                               | 51 |
| 13.01       | Cost of the Work   | 51 |
| 13.02       | Allowances   | 55 |
| 13.03       | Unit Price Work  | 55 |
| Article 14- | -Tests and Inspections; Correction, Removal, or Acceptance of Defective Work | 56 |
| 14.01       | Access to Work   | 56 |
| 14.02       | Tests, Inspections, and Approvals  | 56 |
| 14.03       | Defective Work   | 57 |
| 14.04       | Acceptance of Defective Work   | 58 |
| 14.05       | Uncovering Work  | 58 |
| 14.06       | Owner May Stop the Work  | 58 |
| 14.07       | Owner May Correct Defective Work   | 59 |
| Article 15- | -Payments to Contractor; Set-Offs; Completion; Correction Period             | 59 |
| 15.01       | Progress Payments  | 59 |
| 15.02       | Contractor's Warranty of Title   | 62 |
| 15.03       | Substantial Completion   | 62 |
| 15.04       | Partial Use or Occupancy   | 63 |
| 15.05       | Final Inspection   | 64 |
| 15.06       | Final Payment  | 64 |
| 15.07       | Waiver of Claims   | 65 |
| 15.08       | Correction Period  | 66 |
| Article 16- | -Suspension of Work and Termination  | 67 |
| 16.01       | Owner May Suspend Work   | 67 |
| 16.02       | Owner May Terminate for Cause  | 67 |
| 16.03       | Owner May Terminate for Convenience  | 68 |
| 16.04       | Contractor May Stop Work or Terminate  | 68 |
| Article 17- | -Final Resolution of Disputes  | 69 |
| 17.01       | Methods and Procedures   | 69 |
| Article 18- | –Miscellaneous   | 69 |
| 18.01       | Giving Notice  | 69 |

| 18.02 | Computation of Times    | 69 |
|-------|-------------------------|----|
| 18.03 | Cumulative Remedies     | 70 |
| 18.04 | Limitation of Damages   | 70 |
| 18.05 | No Waiver               | 70 |
| 18.06 | Survival of Obligations | 70 |
| 18.07 | Controlling Law         | 70 |
| 18.08 | Assignment of Contract  | 70 |
| 18.09 | Successors and Assigns  | 70 |
| 18.10 | Headings                | 70 |

| EJCDC® C-700, Standard General Conditions of the Construction Contract. |  |
|---|--|

# STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

#### ARTICLE 1—DEFINITIONS AND TERMINOLOGY

#### 1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents, a term printed with initial capital letters, including the term's singular and plural forms, will have the meaning indicated in the definitions below. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
  - Addenda—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
  - Agreement—The written instrument, executed by Owner and Contractor, that sets forth
    the Contract Price and Contract Times, identifies the parties and the Engineer, and
    designates the specific items that are Contract Documents.
  - 3. Application for Payment—The document prepared by Contractor, in a form acceptable to Engineer, to request progress or final payments, and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
  - 4. *Bid*—The offer of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
  - 5. *Bidder*—An individual or entity that submits a Bid to Owner.
  - 6. *Bidding Documents*—The Bidding Requirements, the proposed Contract Documents, and all Addenda.
  - 7. *Bidding Requirements*—The Advertisement or invitation to bid, Instructions to Bidders, Bid Bond or other Bid security, if any, the Bid Form, and the Bid with any attachments.
  - 8. Change Order—A document which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, or other revision to the Contract, issued on or after the Effective Date of the Contract.
  - 9. Change Proposal—A written request by Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment in Contract Price or Contract Times; contesting an initial decision by Engineer concerning the requirements of the Contract Documents or the acceptability of Work under the Contract Documents; challenging a set-off against payments due; or seeking other relief with respect to the terms of the Contract.

#### 10. Claim

 a. A demand or assertion by Owner directly to Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment of Contract Price or Contract Times; contesting an initial decision by Engineer concerning the

- requirements of the Contract Documents or the acceptability of Work under the Contract Documents; contesting Engineer's decision regarding a Change Proposal; seeking resolution of a contractual issue that Engineer has declined to address; or seeking other relief with respect to the terms of the Contract.
- b. A demand or assertion by Contractor directly to Owner, duly submitted in compliance with the procedural requirements set forth herein, contesting Engineer's decision regarding a Change Proposal, or seeking resolution of a contractual issue that Engineer has declined to address.
- c. A demand or assertion by Owner or Contractor, duly submitted in compliance with the procedural requirements set forth herein, made pursuant to Paragraph 12.01.A.4, concerning disputes arising after Engineer has issued a recommendation of final payment.
- d. A demand for money or services by a third party is not a Claim.
- 11. Constituent of Concern—Asbestos, petroleum, radioactive materials, polychlorinated biphenyls (PCBs), lead-based paint (as defined by the HUD/EPA standard), hazardous waste, and any substance, product, waste, or other material of any nature whatsoever that is or becomes listed, regulated, or addressed pursuant to Laws and Regulations regulating, relating to, or imposing liability or standards of conduct concerning, any hazardous, toxic, or dangerous waste, substance, or material.
- 12. *Contract*—The entire and integrated written contract between Owner and Contractor concerning the Work.
- 13. *Contract Documents*—Those items so designated in the Agreement, and which together comprise the Contract.
- 14. *Contract Price*—The money that Owner has agreed to pay Contractor for completion of the Work in accordance with the Contract Documents.
- 15. *Contract Times*—The number of days or the dates by which Contractor shall: (a) achieve Milestones, if any; (b) achieve Substantial Completion; and (c) complete the Work.
- 16. *Contractor*—The individual or entity with which Owner has contracted for performance of the Work.
- 17. Cost of the Work—See Paragraph 13.01 for definition.
- 18. *Drawings*—The part of the Contract that graphically shows the scope, extent, and character of the Work to be performed by Contractor.
- 19. *Effective Date of the Contract*—The date, indicated in the Agreement, on which the Contract becomes effective.
- 20. *Electronic Document*—Any Project-related correspondence, attachments to correspondence, data, documents, drawings, information, or graphics, including but not limited to Shop Drawings and other Submittals, that are in an electronic or digital format.
- 21. Electronic Means—Electronic mail (email), upload/download from a secure Project website, or other communications methods that allow: (a) the transmission or communication of Electronic Documents; (b) the documentation of transmissions, including sending and receipt; (c) printing of the transmitted Electronic Document by the

recipient; (d) the storage and archiving of the Electronic Document by sender and recipient; and (e) the use by recipient of the Electronic Document for purposes permitted by this Contract. Electronic Means does not include the use of text messaging, or of Facebook, Twitter, Instagram, or similar social media services for transmission of Electronic Documents.

- 22. Engineer—The individual or entity named as such in the Agreement.
- 23. Field Order—A written order issued by Engineer which requires minor changes in the Work but does not change the Contract Price or the Contract Times.
- 24. Hazardous Environmental Condition—The presence at the Site of Constituents of Concern in such quantities or circumstances that may present a danger to persons or property exposed thereto.
  - a. The presence at the Site of materials that are necessary for the execution of the Work, or that are to be incorporated into the Work, and that are controlled and contained pursuant to industry practices, Laws and Regulations, and the requirements of the Contract, is not a Hazardous Environmental Condition.
  - b. The presence of Constituents of Concern that are to be removed or remediated as part of the Work is not a Hazardous Environmental Condition.
  - c. The presence of Constituents of Concern as part of the routine, anticipated, and obvious working conditions at the Site, is not a Hazardous Environmental Condition.
- 25. Laws and Regulations; Laws or Regulations—Any and all applicable laws, statutes, rules, regulations, ordinances, codes, and binding decrees, resolutions, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- 26. *Liens*—Charges, security interests, or encumbrances upon Contract-related funds, real property, or personal property.
- 27. *Milestone*—A principal event in the performance of the Work that the Contract requires Contractor to achieve by an intermediate completion date, or by a time prior to Substantial Completion of all the Work.
- 28. Notice of Award—The written notice by Owner to a Bidder of Owner's acceptance of the Bid.
- 29. *Notice to Proceed*—A written notice by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work.
- 30. Owner—The individual or entity with which Contractor has contracted regarding the Work, and which has agreed to pay Contractor for the performance of the Work, pursuant to the terms of the Contract.
- 31. *Progress Schedule*—A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising Contractor's plan to accomplish the Work within the Contract Times.
- 32. *Project*—The total undertaking to be accomplished for Owner by engineers, contractors, and others, including planning, study, design, construction, testing, commissioning, and start-up, and of which the Work to be performed under the Contract Documents is a part.

- 33. Resident Project Representative—The authorized representative of Engineer assigned to assist Engineer at the Site. As used herein, the term Resident Project Representative (RPR) includes any assistants or field staff of Resident Project Representative.
- 34. Samples—Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and that establish the standards by which such portion of the Work will be judged.
- 35. *Schedule of Submittals*—A schedule, prepared and maintained by Contractor, of required submittals and the time requirements for Engineer's review of the submittals.
- 36. Schedule of Values—A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.
- 37. Shop Drawings—All drawings, diagrams, illustrations, schedules, and other data or information that are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work. Shop Drawings, whether approved or not, are not Drawings and are not Contract Documents.
- 38. Site—Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements, and such other lands or areas furnished by Owner which are designated for the use of Contractor.
- 39. Specifications—The part of the Contract that consists of written requirements for materials, equipment, systems, standards, and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable to the Work.
- 40. *Subcontractor*—An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work.
- 41. Submittal—A written or graphic document, prepared by or for Contractor, which the Contract Documents require Contractor to submit to Engineer, or that is indicated as a Submittal in the Schedule of Submittals accepted by Engineer. Submittals may include Shop Drawings and Samples; schedules; product data; Owner-delegated designs; sustainable design information; information on special procedures; testing plans; results of tests and evaluations, source quality-control testing and inspections, and field or Site quality-control testing and inspections; warranties and certifications; Suppliers' instructions and reports; records of delivery of spare parts and tools; operations and maintenance data; Project photographic documentation; record documents; and other such documents required by the Contract Documents. Submittals, whether or not approved or accepted by Engineer, are not Contract Documents. Change Proposals, Change Orders, Claims, notices, Applications for Payment, and requests for interpretation or clarification are not Submittals.
- 42. Substantial Completion—The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion of such Work.

- 43. Successful Bidder—The Bidder to which the Owner makes an award of contract.
- 44. *Supplementary Conditions*—The part of the Contract that amends or supplements these General Conditions.
- 45. Supplier—A manufacturer, fabricator, supplier, distributor, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or a Subcontractor.

#### 46. Technical Data

- a. Those items expressly identified as Technical Data in the Supplementary Conditions, with respect to either (1) existing subsurface conditions at or adjacent to the Site, or existing physical conditions at or adjacent to the Site including existing surface or subsurface structures (except Underground Facilities) or (2) Hazardous Environmental Conditions at the Site.
- b. If no such express identifications of Technical Data have been made with respect to conditions at the Site, then Technical Data is defined, with respect to conditions at the Site under Paragraphs 5.03, 5.04, and 5.06, as the data contained in boring logs, recorded measurements of subsurface water levels, assessments of the condition of subsurface facilities, laboratory test results, and other factual, objective information regarding conditions at the Site that are set forth in any geotechnical, environmental, or other Site or facilities conditions report prepared for the Project and made available to Contractor.
- c. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data, and instead Underground Facilities are shown or indicated on the Drawings.
- 47. *Underground Facilities*—All active or not-in-service underground lines, pipelines, conduits, ducts, encasements, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or systems at the Site, including but not limited to those facilities or systems that produce, transmit, distribute, or convey telephone or other communications, cable television, fiber optic transmissions, power, electricity, light, heat, gases, oil, crude oil products, liquid petroleum products, water, steam, waste, wastewater, storm water, other liquids or chemicals, or traffic or other control systems. An abandoned facility or system is not an Underground Facility.
- 48. *Unit Price Work*—Work to be paid for on the basis of unit prices.
- 49. Work—The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction; furnishing, installing, and incorporating all materials and equipment into such construction; and may include related services such as testing, start-up, and commissioning, all as required by the Contract Documents.
- 50. Work Change Directive—A written directive to Contractor issued on or after the Effective Date of the Contract, signed by Owner and recommended by Engineer, ordering an addition, deletion, or revision in the Work.

#### 1.02 Terminology

- A. The words and terms discussed in Paragraphs 1.02.B, C, D, and E are not defined terms that require initial capital letters, but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning.
- B. Intent of Certain Terms or Adjectives: The Contract Documents include the terms "as allowed," "as approved," "as ordered," "as directed" or terms of like effect or import to authorize an exercise of professional judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to describe an action or determination of Engineer as to the Work. It is intended that such exercise of professional judgment, action, or determination will be solely to evaluate, in general, the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a functioning whole as shown or indicated in the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective is not intended to and shall not be effective to assign to Engineer any duty or authority to undertake responsibility contrary to the provisions of Article 10 or any other provision of the Contract Documents.
- C. Day: The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.
- D. *Defective*: The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
  - 1. does not conform to the Contract Documents;
  - 2. does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
  - 3. has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 15.03 or Paragraph 15.04).

#### E. Furnish, Install, Perform, Provide

- 1. The word "furnish," when used in connection with services, materials, or equipment, means to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
- 2. The word "install," when used in connection with services, materials, or equipment, means to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
- 3. The words "perform" or "provide," when used in connection with services, materials, or equipment, means to furnish and install said services, materials, or equipment complete and ready for intended use.
- 4. If the Contract Documents establish an obligation of Contractor with respect to specific services, materials, or equipment, but do not expressly use any of the four words "furnish," "install," "perform," or "provide," then Contractor shall furnish and install said services, materials, or equipment complete and ready for intended use.

- F. Contract Price or Contract Times: References to a change in "Contract Price or Contract Times" or "Contract Times or Contract Price" or similar, indicate that such change applies to (1) Contract Price, (2) Contract Times, or (3) both Contract Price and Contract Times, as warranted, even if the term "or both" is not expressed.
- G. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

#### **ARTICLE 2—PRELIMINARY MATTERS**

#### 2.01 Delivery of Performance and Payment Bonds; Evidence of Insurance

- A. Performance and Payment Bonds: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner the performance bond and payment bond (if the Contract requires Contractor to furnish such bonds).
- B. Evidence of Contractor's Insurance: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner, with copies to each additional insured (as identified in the Contract), the certificates, endorsements, and other evidence of insurance required to be provided by Contractor in accordance with Article 6, except to the extent the Supplementary Conditions expressly establish other dates for delivery of specific insurance policies.
- C. Evidence of Owner's Insurance: After receipt of the signed counterparts of the Agreement and all required bonds and insurance documentation, Owner shall promptly deliver to Contractor, with copies to each additional insured (as identified in the Contract), the certificates and other evidence of insurance required to be provided by Owner under Article 6.

# 2.02 Copies of Documents

- A. Owner shall furnish to Contractor four printed copies of the Contract (including one fully signed counterpart of the Agreement), and one copy in electronic portable document format (PDF). Additional printed copies will be furnished upon request at the cost of reproduction.
- B. Owner shall maintain and safeguard at least one original printed record version of the Contract, including Drawings and Specifications signed and sealed by Engineer and other design professionals. Owner shall make such original printed record version of the Contract available to Contractor for review. Owner may delegate the responsibilities under this provision to Engineer.

#### 2.03 Before Starting Construction

- A. *Preliminary Schedules*: Within 10 days after the Effective Date of the Contract (or as otherwise required by the Contract Documents), Contractor shall submit to Engineer for timely review:
  - a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract;
  - 2. a preliminary Schedule of Submittals; and
  - 3. a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work

into component parts in sufficient detail to serve as the basis for progress payments during performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.

#### 2.04 Preconstruction Conference; Designation of Authorized Representatives

- A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work, and to discuss the schedules referred to in Paragraph 2.03.A, procedures for handling Shop Drawings, Samples, and other Submittals, processing Applications for Payment, electronic or digital transmittals, and maintaining required records.
- B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit and receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.

## 2.05 Acceptance of Schedules

- A. At least 10 days before submission of the first Application for Payment a conference, attended by Contractor, Engineer, and others as appropriate, will be held to review the schedules submitted in accordance with Paragraph 2.03.A. No progress payment will be made to Contractor until acceptable schedules are submitted to Engineer.
  - The Progress Schedule will be acceptable to Engineer if it provides an orderly progression
    of the Work to completion within the Contract Times. Such acceptance will not impose
    on Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or
    progress of the Work, nor interfere with or relieve Contractor from Contractor's full
    responsibility therefor.
  - 2. Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
  - Contractor's Schedule of Values will be acceptable to Engineer as to form and substance
    if it provides a reasonable allocation of the Contract Price to the component parts of the
    Work.
  - 4. If a schedule is not acceptable, Contractor will have an additional 10 days to revise and resubmit the schedule.

#### 2.06 Electronic Transmittals

- A. Except as otherwise stated elsewhere in the Contract, the Owner, Engineer, and Contractor may send, and shall accept, Electronic Documents transmitted by Electronic Means.
- B. If the Contract does not establish protocols for Electronic Means, then Owner, Engineer, and Contractor shall jointly develop such protocols.
- C. Subject to any governing protocols for Electronic Means, when transmitting Electronic Documents by Electronic Means, the transmitting party makes no representations as to long-term compatibility, usability, or readability of the Electronic Documents resulting from the recipient's use of software application packages, operating systems, or computer hardware differing from those used in the drafting or transmittal of the Electronic Documents.

#### ARTICLE 3—CONTRACT DOCUMENTS: INTENT, REQUIREMENTS, REUSE

#### 3.01 Intent

- A. The Contract Documents are complementary; what is required by one Contract Document is as binding as if required by all.
- B. It is the intent of the Contract Documents to describe a functionally complete Project (or part thereof) to be constructed in accordance with the Contract Documents.
- C. Unless otherwise stated in the Contract Documents, if there is a discrepancy between the electronic versions of the Contract Documents (including any printed copies derived from such electronic versions) and the printed record version, the printed record version will govern.
- D. The Contract supersedes prior negotiations, representations, and agreements, whether written or oral.
- E. Engineer will issue clarifications and interpretations of the Contract Documents as provided herein.
- F. Any provision or part of the Contract Documents held to be void or unenforceable under any Law or Regulation will be deemed stricken, and all remaining provisions will continue to be valid and binding upon Owner and Contractor, which agree that the Contract Documents will be reformed to replace such stricken provision or part thereof with a valid and enforceable provision that comes as close as possible to expressing the intention of the stricken provision.
- G. Nothing in the Contract Documents creates:
  - 1. any contractual relationship between Owner or Engineer and any Subcontractor, Supplier, or other individual or entity performing or furnishing any of the Work, for the benefit of such Subcontractor, Supplier, or other individual or entity; or
  - any obligation on the part of Owner or Engineer to pay or to see to the payment of any money due any such Subcontractor, Supplier, or other individual or entity, except as may otherwise be required by Laws and Regulations.

## 3.02 Reference Standards

- A. Standards Specifications, Codes, Laws and Regulations
  - Reference in the Contract Documents to standard specifications, manuals, reference standards, or codes of any technical society, organization, or association, or to Laws or Regulations, whether such reference be specific or by implication, means the standard specification, manual, reference standard, code, or Laws or Regulations in effect at the time of opening of Bids (or on the Effective Date of the Contract if there were no Bids), except as may be otherwise specifically stated in the Contract Documents.
  - 2. No provision of any such standard specification, manual, reference standard, or code, and no instruction of a Supplier, will be effective to change the duties or responsibilities of Owner, Contractor, or Engineer from those set forth in the part of the Contract Documents prepared by or for Engineer. No such provision or instruction shall be effective to assign to Owner or Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility

inconsistent with the provisions of the part of the Contract Documents prepared by or for Engineer.

## 3.03 Reporting and Resolving Discrepancies

#### A. Reporting Discrepancies

- 1. Contractor's Verification of Figures and Field Measurements: Before undertaking each part of the Work, Contractor shall carefully study the Contract Documents, and check and verify pertinent figures and dimensions therein, particularly with respect to applicable field measurements. Contractor shall promptly report in writing to Engineer any conflict, error, ambiguity, or discrepancy that Contractor discovers, or has actual knowledge of, and shall not proceed with any Work affected thereby until the conflict, error, ambiguity, or discrepancy is resolved by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
- 2. Contractor's Review of Contract Documents: If, before or during the performance of the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the Contract Documents, or between the Contract Documents and (a) any applicable Law or Regulation, (b) actual field conditions, (c) any standard specification, manual, reference standard, or code, or (d) any instruction of any Supplier, then Contractor shall promptly report it to Engineer in writing. Contractor shall not proceed with the Work affected thereby (except in an emergency as required by Paragraph 7.15) until the conflict, error, ambiguity, or discrepancy is resolved, by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
- Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof.

#### B. Resolving Discrepancies

- Except as may be otherwise specifically stated in the Contract Documents, the provisions
  of the part of the Contract Documents prepared by or for Engineer take precedence in
  resolving any conflict, error, ambiguity, or discrepancy between such provisions of the
  Contract Documents and:
  - a. the provisions of any standard specification, manual, reference standard, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference as a Contract Document); or
  - b. the provisions of any Laws or Regulations applicable to the performance of the Work (unless such an interpretation of the provisions of the Contract Documents would result in violation of such Law or Regulation).

#### 3.04 Requirements of the Contract Documents

A. During the performance of the Work and until final payment, Contractor and Owner shall submit to the Engineer in writing all matters in question concerning the requirements of the Contract Documents (sometimes referred to as requests for information or interpretation—RFIs), or relating to the acceptability of the Work under the Contract Documents, as soon as possible after such matters arise. Engineer will be the initial interpreter of the requirements of the Contract Documents, and judge of the acceptability of the Work.

- B. Engineer will, with reasonable promptness, render a written clarification, interpretation, or decision on the issue submitted, or initiate an amendment or supplement to the Contract Documents. Engineer's written clarification, interpretation, or decision will be final and binding on Contractor, unless it appeals by submitting a Change Proposal, and on Owner, unless it appeals by filing a Claim.
- C. If a submitted matter in question concerns terms and conditions of the Contract Documents that do not involve (1) the performance or acceptability of the Work under the Contract Documents, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, then Engineer will promptly notify Owner and Contractor in writing that Engineer is unable to provide a decision or interpretation. If Owner and Contractor are unable to agree on resolution of such a matter in question, either party may pursue resolution as provided in Article 12.

#### 3.05 Reuse of Documents

- A. Contractor and its Subcontractors and Suppliers shall not:
  - have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media versions, or reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project without written consent of Owner and Engineer and specific written verification or adaptation by Engineer; or
  - 2. have or acquire any title or ownership rights in any other Contract Documents, reuse any such Contract Documents for any purpose without Owner's express written consent, or violate any copyrights pertaining to such Contract Documents.
- B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein precludes Contractor from retaining copies of the Contract Documents for record purposes.

#### ARTICLE 4—COMMENCEMENT AND PROGRESS OF THE WORK

- 4.01 Commencement of Contract Times; Notice to Proceed
  - A. The Contract Times will commence to run on the 30th day after the Effective Date of the Contract or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Contract. In no event will the Contract Times commence to run later than the 60th day after the day of Bid opening or the 30th day after the Effective Date of the Contract, whichever date is earlier.

#### 4.02 Starting the Work

A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work may be done at the Site prior to such date.

#### 4.03 Reference Points

A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the

established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or requires relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or property monuments by professionally qualified personnel.

#### 4.04 Progress Schedule

- A. Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.05 as it may be adjusted from time to time as provided below.
  - Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.05) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times.
  - 2. Proposed adjustments in the Progress Schedule that will change the Contract Times must be submitted in accordance with the requirements of Article 11.
- B. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work will be delayed or postponed pending resolution of any disputes or disagreements, or during any appeal process, except as permitted by Paragraph 16.04, or as Owner and Contractor may otherwise agree in writing.

#### 4.05 Delays in Contractor's Progress

- A. If Owner, Engineer, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times.
- B. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delay, disruption, or interference caused by or within the control of Contractor. Delay, disruption, and interference attributable to and within the control of a Subcontractor or Supplier shall be deemed to be within the control of Contractor.
- C. If Contractor's performance or progress is delayed, disrupted, or interfered with by unanticipated causes not the fault of and beyond the control of Owner, Contractor, and those for which they are responsible, then Contractor shall be entitled to an equitable adjustment in Contract Times. Such an adjustment will be Contractor's sole and exclusive remedy for the delays, disruption, and interference described in this paragraph. Causes of delay, disruption, or interference that may give rise to an adjustment in Contract Times under this paragraph include but are not limited to the following:
  - Severe and unavoidable natural catastrophes such as fires, floods, epidemics, and earthquakes;
  - 2. Abnormal weather conditions;
  - 3. Acts or failures to act of third-party utility owners or other third-party entities (other than those third-party utility owners or other third-party entities performing other work at or adjacent to the Site as arranged by or under contract with Owner, as contemplated in Article 8); and
  - 4. Acts of war or terrorism.

- D. Contractor's entitlement to an adjustment of Contract Times or Contract Price is limited as follows:
  - 1. Contractor's entitlement to an adjustment of the Contract Times is conditioned on the delay, disruption, or interference adversely affecting an activity on the critical path to completion of the Work, as of the time of the delay, disruption, or interference.
  - Contractor shall not be entitled to an adjustment in Contract Price for any delay, disruption, or interference if such delay is concurrent with a delay, disruption, or interference caused by or within the control of Contractor. Such a concurrent delay by Contractor shall not preclude an adjustment of Contract Times to which Contractor is otherwise entitled.
  - 3. Adjustments of Contract Times or Contract Price are subject to the provisions of Article 11.
- E. Each Contractor request or Change Proposal seeking an increase in Contract Times or Contract Price must be supplemented by supporting data that sets forth in detail the following:
  - 1. The circumstances that form the basis for the requested adjustment;
  - 2. The date upon which each cause of delay, disruption, or interference began to affect the progress of the Work;
  - 3. The date upon which each cause of delay, disruption, or interference ceased to affect the progress of the Work;
  - 4. The number of days' increase in Contract Times claimed as a consequence of each such cause of delay, disruption, or interference; and
  - 5. The impact on Contract Price, in accordance with the provisions of Paragraph 11.07.

Contractor shall also furnish such additional supporting documentation as Owner or Engineer may require including, where appropriate, a revised progress schedule indicating all the activities affected by the delay, disruption, or interference, and an explanation of the effect of the delay, disruption, or interference on the critical path to completion of the Work.

- F. Delays, disruption, and interference to the performance or progress of the Work resulting from the existence of a differing subsurface or physical condition, an Underground Facility that was not shown or indicated by the Contract Documents, or not shown or indicated with reasonable accuracy, and those resulting from Hazardous Environmental Conditions, are governed by Article 5, together with the provisions of Paragraphs 4.05.D and 4.05.E.
- G. Paragraph 8.03 addresses delays, disruption, and interference to the performance or progress of the Work resulting from the performance of certain other work at or adjacent to the Site.

# ARTICLE 5—SITE; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS

#### 5.01 Availability of Lands

A. Owner shall furnish the Site. Owner shall notify Contractor in writing of any encumbrances or restrictions not of general application but specifically related to use of the Site with which Contractor must comply in performing the Work.

- B. Upon reasonable written request, Owner shall furnish Contractor with a current statement of record legal title and legal description of the lands upon which permanent improvements are to be made and Owner's interest therein as necessary for giving notice of or filing a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
- C. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment.

#### 5.02 Use of Site and Other Areas

#### A. Limitation on Use of Site and Other Areas

- 1. Contractor shall confine construction equipment, temporary construction facilities, the storage of materials and equipment, and the operations of workers to the Site, adjacent areas that Contractor has arranged to use through construction easements or otherwise, and other adjacent areas permitted by Laws and Regulations, and shall not unreasonably encumber the Site and such other adjacent areas with construction equipment or other materials or equipment. Contractor shall assume full responsibility for (a) damage to the Site; (b) damage to any such other adjacent areas used for Contractor's operations; (c) damage to any other adjacent land or areas, or to improvements, structures, utilities, or similar facilities located at such adjacent lands or areas; and (d) for injuries and losses sustained by the owners or occupants of any such land or areas; provided that such damage or injuries result from the performance of the Work or from other actions or conduct of the Contractor or those for which Contractor is responsible.
- 2. If a damage or injury claim is made by the owner or occupant of any such land or area because of the performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible, Contractor shall (a) take immediate corrective or remedial action as required by Paragraph 7.13, or otherwise; (b) promptly attempt to settle the claim as to all parties through negotiations with such owner or occupant, or otherwise resolve the claim by arbitration or other dispute resolution proceeding, or in a court of competent jurisdiction; and (c) to the fullest extent permitted by Laws and Regulations, indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against any such claim, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused directly or indirectly, in whole or in part by, or based upon, Contractor's performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible.
- B. Removal of Debris During Performance of the Work: During the progress of the Work the Contractor shall keep the Site and other adjacent areas free from accumulations of waste materials, rubbish, and other debris. Removal and disposal of such waste materials, rubbish, and other debris will conform to applicable Laws and Regulations.
- C. Cleaning: Prior to Substantial Completion of the Work Contractor shall clean the Site and the Work and make it ready for utilization by Owner. At the completion of the Work Contractor shall remove from the Site and adjacent areas all tools, appliances, construction equipment

- and machinery, and surplus materials and shall restore to original condition all property not designated for alteration by the Contract Documents.
- D. Loading of Structures: Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall Contractor subject any part of the Work or adjacent structures or land to stresses or pressures that will endanger them.

#### 5.03 Subsurface and Physical Conditions

- A. Reports and Drawings: The Supplementary Conditions identify:
  - 1. Those reports of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data;
  - 2. Those drawings of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to the Site (except Underground Facilities), that contain Technical Data; and
  - 3. Technical Data contained in such reports and drawings.
- B. *Underground Facilities*: Underground Facilities are shown or indicated on the Drawings, pursuant to Paragraph 5.05, and not in the drawings referred to in Paragraph 5.03.A. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data.
- C. Reliance by Contractor on Technical Data: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely upon the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b.
- D. Limitations of Other Data and Documents: Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
  - the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto;
  - 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings;
  - the contents of other Site-related documents made available to Contractor, such as record drawings from other projects at or adjacent to the Site, or Owner's archival documents concerning the Site; or
  - 4. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions, or information.

## 5.04 Differing Subsurface or Physical Conditions

- A. *Notice by Contractor*: If Contractor believes that any subsurface or physical condition that is uncovered or revealed at the Site:
  - 1. is of such a nature as to establish that any Technical Data on which Contractor is entitled to rely as provided in Paragraph 5.03 is materially inaccurate;
  - 2. is of such a nature as to require a change in the Drawings or Specifications;
  - 3. differs materially from that shown or indicated in the Contract Documents; or
  - 4. is of an unusual nature, and differs materially from conditions ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents;

then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such condition or perform any Work in connection therewith (except with respect to an emergency) until receipt of a written statement permitting Contractor to do so.

- B. Engineer's Review: After receipt of written notice as required by the preceding paragraph, Engineer will promptly review the subsurface or physical condition in question; determine whether it is necessary for Owner to obtain additional exploration or tests with respect to the condition; conclude whether the condition falls within any one or more of the differing site condition categories in Paragraph 5.04.A; obtain any pertinent cost or schedule information from Contractor; prepare recommendations to Owner regarding the Contractor's resumption of Work in connection with the subsurface or physical condition in question and the need for any change in the Drawings or Specifications; and advise Owner in writing of Engineer's findings, conclusions, and recommendations.
- C. Owner's Statement to Contractor Regarding Site Condition: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the subsurface or physical condition in question, addressing the resumption of Work in connection with such condition, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations, in whole or in part.
- D. Early Resumption of Work: If at any time Engineer determines that Work in connection with the subsurface or physical condition in question may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the condition in question has been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- E. Possible Price and Times Adjustments
  - Contractor shall be entitled to an equitable adjustment in Contract Price or Contract
    Times, to the extent that the existence of a differing subsurface or physical condition, or
    any related delay, disruption, or interference, causes an increase or decrease in

Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:

- a. Such condition must fall within any one or more of the categories described in Paragraph 5.04.A;
- b. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03; and,
- c. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E.
- 2. Contractor shall not be entitled to any adjustment in the Contract Price or Contract Times with respect to a subsurface or physical condition if:
  - a. Contractor knew of the existence of such condition at the time Contractor made a commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract, or otherwise;
  - b. The existence of such condition reasonably could have been discovered or revealed as a result of any examination, investigation, exploration, test, or study of the Site and contiguous areas expressly required by the Bidding Requirements or Contract Documents to be conducted by or for Contractor prior to Contractor's making such commitment; or
  - c. Contractor failed to give the written notice required by Paragraph 5.04.A.
- 3. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 4. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the subsurface or physical condition in question.
- F. Underground Facilities; Hazardous Environmental Conditions: Paragraph 5.05 governs rights and responsibilities regarding the presence or location of Underground Facilities. Paragraph 5.06 governs rights and responsibilities regarding Hazardous Environmental Conditions. The provisions of Paragraphs 5.03 and 5.04 are not applicable to the presence or location of Underground Facilities, or to Hazardous Environmental Conditions.

# 5.05 Underground Facilities

- A. Contractor's Responsibilities: Unless it is otherwise expressly provided in the Supplementary Conditions, the cost of all of the following are included in the Contract Price, and Contractor shall have full responsibility for:
  - 1. reviewing and checking all information and data regarding existing Underground Facilities at the Site;
  - complying with applicable state and local utility damage prevention Laws and Regulations;

- 3. verifying the actual location of those Underground Facilities shown or indicated in the Contract Documents as being within the area affected by the Work, by exposing such Underground Facilities during the course of construction;
- 4. coordination of the Work with the owners (including Owner) of such Underground Facilities, during construction; and
- 5. the safety and protection of all existing Underground Facilities at the Site, and repairing any damage thereto resulting from the Work.
- B. Notice by Contractor: If Contractor believes that an Underground Facility that is uncovered or revealed at the Site was not shown or indicated on the Drawings, or was not shown or indicated on the Drawings with reasonable accuracy, then Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing regarding such Underground Facility.
- C. *Engineer's Review*: Engineer will:
  - 1. promptly review the Underground Facility and conclude whether such Underground Facility was not shown or indicated on the Drawings, or was not shown or indicated with reasonable accuracy;
  - identify and communicate with the owner of the Underground Facility; prepare recommendations to Owner (and if necessary issue any preliminary instructions to Contractor) regarding the Contractor's resumption of Work in connection with the Underground Facility in question;
  - obtain any pertinent cost or schedule information from Contractor; determine the extent,
    if any, to which a change is required in the Drawings or Specifications to reflect and
    document the consequences of the existence or location of the Underground Facility; and
  - 4. advise Owner in writing of Engineer's findings, conclusions, and recommendations.
  - During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.
- D. Owner's Statement to Contractor Regarding Underground Facility: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the Underground Facility in question addressing the resumption of Work in connection with such Underground Facility, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations in whole or in part.
- E. Early Resumption of Work: If at any time Engineer determines that Work in connection with the Underground Facility may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the Underground Facility in question and conditions affected by its presence have been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- F. Possible Price and Times Adjustments
  - Contractor shall be entitled to an equitable adjustment in the Contract Price or Contract
    Times, to the extent that any existing Underground Facility at the Site that was not shown

or indicated on the Drawings, or was not shown or indicated with reasonable accuracy, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:

- a. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03;
- b. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E; and
- c. Contractor gave the notice required in Paragraph 5.05.B.
- 2. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 3. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the Underground Facility in question.
- 4. The information and data shown or indicated on the Drawings with respect to existing Underground Facilities at the Site is based on information and data (a) furnished by the owners of such Underground Facilities, or by others, (b) obtained from available records, or (c) gathered in an investigation conducted in accordance with the current edition of ASCE 38, Standard Guideline for the Collection and Depiction of Existing Subsurface Utility Data, by the American Society of Civil Engineers. If such information or data is incorrect or incomplete, Contractor's remedies are limited to those set forth in this Paragraph 5.05.F.

#### 5.06 Hazardous Environmental Conditions at Site

- A. *Reports and Drawings*: The Supplementary Conditions identify:
  - 1. those reports known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site;
  - 2. drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site; and
  - 3. Technical Data contained in such reports and drawings.
- B. Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely on the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b. Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
  - 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures

- of construction to be employed by Contractor, and safety precautions and programs incident thereto;
- 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings; or
- 3. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions or information.
- C. Contractor shall not be responsible for removing or remediating any Hazardous Environmental Condition encountered, uncovered, or revealed at the Site unless such removal or remediation is expressly identified in the Contract Documents to be within the scope of the Work.
- D. Contractor shall be responsible for controlling, containing, and duly removing all Constituents of Concern brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible, and for any associated costs; and for the costs of removing and remediating any Hazardous Environmental Condition created by the presence of any such Constituents of Concern.
- E. If Contractor encounters, uncovers, or reveals a Hazardous Environmental Condition whose removal or remediation is not expressly identified in the Contract Documents as being within the scope of the Work, or if Contractor or anyone for whom Contractor is responsible creates a Hazardous Environmental Condition, then Contractor shall immediately: (1) secure or otherwise isolate such condition; (2) stop all Work in connection with such condition and in any area affected thereby (except in an emergency as required by Paragraph 7.15); and (3) notify Owner and Engineer (and promptly thereafter confirm such notice in writing). Owner shall promptly consult with Engineer concerning the necessity for Owner to retain a qualified expert to evaluate such condition or take corrective action, if any. Promptly after consulting with Engineer, Owner shall take such actions as are necessary to permit Owner to timely obtain required permits and provide Contractor the written notice required by Paragraph 5.06.F. If Contractor or anyone for whom Contractor is responsible created the Hazardous Environmental Condition in question, then Owner may remove and remediate the Hazardous Environmental Condition, and impose a set-off against payments to account for the associated costs.
- F. Contractor shall not resume Work in connection with such Hazardous Environmental Condition or in any affected area until after Owner has obtained any required permits related thereto, and delivered written notice to Contractor either (1) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work, or (2) specifying any special conditions under which such Work may be resumed safely.
- G. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, as a result of such Work stoppage, such special conditions under which Work is agreed to be resumed by Contractor, or any costs or expenses incurred in response to the Hazardous Environmental Condition, then within 30 days of Owner's written notice regarding the resumption of Work, Contractor may submit a Change Proposal, or Owner may impose a set-off. Entitlement to any such adjustment is subject to the provisions of Paragraphs 4.05.D, 4.05.E, 11.07, and 11.08.
- H. If, after receipt of such written notice, Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special

conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work, following the contractual change procedures in Article 11. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 8.

- I. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court, arbitration, or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition (1) was not shown or indicated in the Drawings, Specifications, or other Contract Documents, identified as Technical Data entitled to limited reliance pursuant to Paragraph 5.06.B, or identified in the Contract Documents to be included within the scope of the Work, and (2) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.I obligates Owner to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- J. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the failure to control, contain, or remove a Constituent of Concern brought to the Site by Contractor or by anyone for whom Contractor is responsible, or to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.J obligates Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- K. The provisions of Paragraphs 5.03, 5.04, and 5.05 do not apply to the presence of Constituents of Concern or to a Hazardous Environmental Condition uncovered or revealed at the Site.

### **ARTICLE 6—BONDS AND INSURANCE**

# 6.01 Performance, Payment, and Other Bonds

- A. Contractor shall furnish a performance bond and a payment bond, each in an amount at least equal to the Contract Price, as security for the faithful performance and payment of Contractor's obligations under the Contract. These bonds must remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 15.08, whichever is later, except as provided otherwise by Laws or Regulations, the terms of a prescribed bond form, the Supplementary Conditions, or other provisions of the Contract.
- B. Contractor shall also furnish such other bonds (if any) as are required by the Supplementary Conditions or other provisions of the Contract.
- C. All bonds must be in the form included in the Bidding Documents or otherwise specified by Owner prior to execution of the Contract, except as provided otherwise by Laws or

Regulations, and must be issued and signed by a surety named in "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Department Circular 570 (as amended and supplemented) by the Bureau of the Fiscal Service, U.S. Department of the Treasury. A bond signed by an agent or attorney-in-fact must be accompanied by a certified copy of that individual's authority to bind the surety. The evidence of authority must show that it is effective on the date the agent or attorney-in-fact signed the accompanying bond.

- D. Contractor shall obtain the required bonds from surety companies that are duly licensed or authorized, in the state or jurisdiction in which the Project is located, to issue bonds in the required amounts.
- E. If the surety on a bond furnished by Contractor is declared bankrupt or becomes insolvent, or the surety ceases to meet the requirements above, then Contractor shall promptly notify Owner and Engineer in writing and shall, within 20 days after the event giving rise to such notification, provide another bond and surety, both of which must comply with the bond and surety requirements above.
- F. If Contractor has failed to obtain a required bond, Owner may exclude the Contractor from the Site and exercise Owner's termination rights under Article 16.
- G. Upon request to Owner from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Owner shall provide a copy of the payment bond to such person or entity.
- H. Upon request to Contractor from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Contractor shall provide a copy of the payment bond to such person or entity.

## 6.02 Insurance—General Provisions

- A. Owner and Contractor shall obtain and maintain insurance as required in this article and in the Supplementary Conditions.
- B. All insurance required by the Contract to be purchased and maintained by Owner or Contractor shall be obtained from insurance companies that are duly licensed or authorized in the state or jurisdiction in which the Project is located to issue insurance policies for the required limits and coverages. Unless a different standard is indicated in the Supplementary Conditions, all companies that provide insurance policies required under this Contract shall have an A.M. Best rating of A-VII or better.
- C. Alternative forms of insurance coverage, including but not limited to self-insurance and "Occupational Accident and Excess Employer's Indemnity Policies," are not sufficient to meet the insurance requirements of this Contract, unless expressly allowed in the Supplementary Conditions.
- D. Contractor shall deliver to Owner, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Contractor has obtained and is maintaining the policies and coverages required by the Contract. Upon request by Owner or any other insured, Contractor shall also furnish other evidence of such required insurance, including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, full disclosure of all relevant exclusions, and evidence of insurance required to be purchased and maintained by

- Subcontractors or Suppliers. In any documentation furnished under this provision, Contractor, Subcontractors, and Suppliers may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those applicable to this Contract.
- E. Owner shall deliver to Contractor, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Owner has obtained and is maintaining the policies and coverages required of Owner by the Contract (if any). Upon request by Contractor or any other insured, Owner shall also provide other evidence of such required insurance (if any), including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, and full disclosure of all relevant exclusions. In any documentation furnished under this provision, Owner may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those relevant to this Contract.
- F. Failure of Owner or Contractor to demand such certificates or other evidence of the other party's full compliance with these insurance requirements, or failure of Owner or Contractor to identify a deficiency in compliance from the evidence provided, will not be construed as a waiver of the other party's obligation to obtain and maintain such insurance.
- G. In addition to the liability insurance required to be provided by Contractor, the Owner, at Owner's option, may purchase and maintain Owner's own liability insurance. Owner's liability policies, if any, operate separately and independently from policies required to be provided by Contractor, and Contractor cannot rely upon Owner's liability policies for any of Contractor's obligations to the Owner, Engineer, or third parties.

#### H. Contractor shall require:

- Subcontractors to purchase and maintain worker's compensation, commercial general liability, and other insurance that is appropriate for their participation in the Project, and to name as additional insureds Owner and Engineer (and any other individuals or entities identified in the Supplementary Conditions as additional insureds on Contractor's liability policies) on each Subcontractor's commercial general liability insurance policy; and
- 2. Suppliers to purchase and maintain insurance that is appropriate for their participation in the Project.
- If either party does not purchase or maintain the insurance required of such party by the Contract, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage.
- J. If Contractor has failed to obtain and maintain required insurance, Contractor's entitlement to enter or remain at the Site will end immediately, and Owner may impose an appropriate set-off against payment for any associated costs (including but not limited to the cost of purchasing necessary insurance coverage), and exercise Owner's termination rights under Article 16.
- K. Without prejudice to any other right or remedy, if a party has failed to obtain required insurance, the other party may elect (but is in no way obligated) to obtain equivalent insurance to protect such other party's interests at the expense of the party who was required to provide such coverage, and the Contract Price will be adjusted accordingly.

- L. Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor or Contractor's interests. Contractor is responsible for determining whether such coverage and limits are adequate to protect its interests, and for obtaining and maintaining any additional insurance that Contractor deems necessary.
- M. The insurance and insurance limits required herein will not be deemed as a limitation on Contractor's liability, or that of its Subcontractors or Suppliers, under the indemnities granted to Owner and other individuals and entities in the Contract or otherwise.
- N. All the policies of insurance required to be purchased and maintained under this Contract will contain a provision or endorsement that the coverage afforded will not be canceled, or renewal refused, until at least 10 days prior written notice has been given to the purchasing policyholder. Within three days of receipt of any such written notice, the purchasing policyholder shall provide a copy of the notice to each other insured and Engineer.

#### 6.03 Contractor's Insurance

- A. *Required Insurance*: Contractor shall purchase and maintain Worker's Compensation, Commercial General Liability, and other insurance pursuant to the specific requirements of the Supplementary Conditions.
- B. *General Provisions*: The policies of insurance required by this Paragraph 6.03 as supplemented must:
  - 1. include at least the specific coverages required;
  - 2. be written for not less than the limits provided, or those required by Laws or Regulations, whichever is greater;
  - remain in effect at least until the Work is complete (as set forth in Paragraph 15.06.D), and longer if expressly required elsewhere in this Contract, and at all times thereafter when Contractor may be correcting, removing, or replacing defective Work as a warranty or correction obligation, or otherwise, or returning to the Site to conduct other tasks arising from the Contract;
  - 4. apply with respect to the performance of the Work, whether such performance is by Contractor, any Subcontractor or Supplier, or by anyone directly or indirectly employed by any of them to perform any of the Work, or by anyone for whose acts any of them may be liable; and
  - 5. include all necessary endorsements to support the stated requirements.
- C. Additional Insureds: The Contractor's commercial general liability, automobile liability, employer's liability, umbrella or excess, pollution liability, and unmanned aerial vehicle liability policies, if required by this Contract, must:
  - 1. include and list as additional insureds Owner and Engineer, and any individuals or entities identified as additional insureds in the Supplementary Conditions;
  - 2. include coverage for the respective officers, directors, members, partners, employees, and consultants of all such additional insureds;
  - 3. afford primary coverage to these additional insureds for all claims covered thereby (including as applicable those arising from both ongoing and completed operations);

- 4. not seek contribution from insurance maintained by the additional insured; and
- 5. as to commercial general liability insurance, apply to additional insureds with respect to liability caused in whole or in part by Contractor's acts or omissions, or the acts and omissions of those working on Contractor's behalf, in the performance of Contractor's operations.

# 6.04 Builder's Risk and Other Property Insurance

- A. Builder's Risk: Unless otherwise provided in the Supplementary Conditions, Contractor shall purchase and maintain builder's risk insurance upon the Work on a completed value basis, in the amount of the Work's full insurable replacement cost (subject to such deductible amounts as may be provided in the Supplementary Conditions or required by Laws and Regulations). The specific requirements applicable to the builder's risk insurance are set forth in the Supplementary Conditions.
- B. Property Insurance for Facilities of Owner Where Work Will Occur: Owner is responsible for obtaining and maintaining property insurance covering each existing structure, building, or facility in which any part of the Work will occur, or to which any part of the Work will attach or be adjoined. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, providing coverage consistent with that required for the builder's risk insurance, and will be maintained until the Work is complete, as set forth in Paragraph 15.06.D.
- C. Property Insurance for Substantially Complete Facilities: Promptly after Substantial Completion, and before actual occupancy or use of the substantially completed Work, Owner will obtain property insurance for such substantially completed Work, and maintain such property insurance at least until the Work is complete, as set forth in Paragraph 15.06.D. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, and provide coverage consistent with that required for the builder's risk insurance. The builder's risk insurance may terminate upon written confirmation of Owner's procurement of such property insurance.
- D. Partial Occupancy or Use by Owner: If Owner will occupy or use a portion or portions of the Work prior to Substantial Completion of all the Work, as provided in Paragraph 15.04, then Owner (directly, if it is the purchaser of the builder's risk policy, or through Contractor) will provide advance notice of such occupancy or use to the builder's risk insurer, and obtain an endorsement consenting to the continuation of coverage prior to commencing such partial occupancy or use.
- E. Insurance of Other Property; Additional Insurance: If the express insurance provisions of the Contract do not require or address the insurance of a property item or interest, then the entity or individual owning such property item will be responsible for insuring it. If Contractor elects to obtain other special insurance to be included in or supplement the builder's risk or property insurance policies provided under this Paragraph 6.04, it may do so at Contractor's expense.

# 6.05 Property Losses; Subrogation

A. The builder's risk insurance policy purchased and maintained in accordance with Paragraph 6.04 (or an installation floater policy if authorized by the Supplementary Conditions), will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against

Engineer or its consultants, or their officers, directors, members, partners, employees, agents, consultants, or subcontractors.

- 1. Owner and Contractor waive all rights against each other and the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from any of the perils, risks, or causes of loss covered by such policies and any other property insurance applicable to the Work; and, in addition, waive all such rights against Engineer, its consultants, all individuals or entities identified in the Supplementary Conditions as builder's risk or installation floater insureds, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, under such policies for losses and damages so caused.
- 2. None of the above waivers extends to the rights that any party making such waiver may have to the proceeds of insurance held by Owner or Contractor as trustee or fiduciary, or otherwise payable under any policy so issued.
- B. Any property insurance policy maintained by Owner covering any loss, damage, or consequential loss to Owner's existing structures, buildings, or facilities in which any part of the Work will occur, or to which any part of the Work will attach or adjoin; to adjacent structures, buildings, or facilities of Owner; or to part or all of the completed or substantially completed Work, during partial occupancy or use pursuant to Paragraph 15.04, after Substantial Completion pursuant to Paragraph 15.03, or after final payment pursuant to Paragraph 15.06, will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against Contractor, Subcontractors, or Engineer, or the officers, directors, members, partners, employees, agents, consultants, or subcontractors of each and any of them, and that the insured is allowed to waive the insurer's rights of subrogation in a written contract executed prior to the loss, damage, or consequential loss.
  - Owner waives all rights against Contractor, Subcontractors, and Engineer, and the
    officers, directors, members, partners, employees, agents, consultants and
    subcontractors of each and any of them, for all losses and damages caused by, arising out
    of, or resulting from fire or any of the perils, risks, or causes of loss covered by such
    policies.
- C. The waivers in this Paragraph 6.05 include the waiver of rights due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other insured peril, risk, or cause of loss.
- D. Contractor shall be responsible for assuring that each Subcontract contains provisions whereby the Subcontractor waives all rights against Owner, Contractor, all individuals or entities identified in the Supplementary Conditions as insureds, the Engineer and its consultants, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, relating to, or resulting from fire or other peril, risk, or cause of loss covered by builder's risk insurance, installation floater, and any other property insurance applicable to the Work.

## 6.06 Receipt and Application of Property Insurance Proceeds

- A. Any insured loss under the builder's risk and other policies of property insurance required by Paragraph 6.04 will be adjusted and settled with the named insured that purchased the policy. Such named insured shall act as fiduciary for the other insureds, and give notice to such other insureds that adjustment and settlement of a claim is in progress. Any other insured may state its position regarding a claim for insured loss in writing within 15 days after notice of such claim.
- B. Proceeds for such insured losses may be made payable by the insurer either jointly to multiple insureds, or to the named insured that purchased the policy in its own right and as fiduciary for other insureds, subject to the requirements of any applicable mortgage clause. A named insured receiving insurance proceeds under the builder's risk and other policies of insurance required by Paragraph 6.04 shall maintain such proceeds in a segregated account, and distribute such proceeds in accordance with such agreement as the parties in interest may reach, or as otherwise required under the dispute resolution provisions of this Contract or applicable Laws and Regulations.
- C. If no other special agreement is reached, Contractor shall repair or replace the damaged Work, using allocated insurance proceeds.

#### ARTICLE 7—CONTRACTOR'S RESPONSIBILITIES

# 7.01 Contractor's Means and Methods of Construction

- A. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction.
- B. If the Contract Documents note, or Contractor determines, that professional engineering or other design services are needed to carry out Contractor's responsibilities for construction means, methods, techniques, sequences, and procedures, or for Site safety, then Contractor shall cause such services to be provided by a properly licensed design professional, at Contractor's expense. Such services are not Owner-delegated professional design services under this Contract, and neither Owner nor Engineer has any responsibility with respect to (1) Contractor's determination of the need for such services, (2) the qualifications or licensing of the design professionals retained or employed by Contractor, (3) the performance of such services, or (4) any errors, omissions, or defects in such services.

### 7.02 Supervision and Superintendence

- A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work in accordance with the Contract Documents.
- B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who will not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.

# 7.03 Labor; Working Hours

A. Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall maintain good discipline and order at the Site.

- B. Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of Contractor's employees; of Suppliers and Subcontractors, and their employees; and of any other individuals or entities performing or furnishing any of the Work, just as Contractor is responsible for Contractor's own acts and omissions.
- C. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent thereto, and except as otherwise stated in the Contract Documents, all Work at the Site will be performed during regular working hours, Monday through Friday. Contractor will not perform Work on a Saturday, Sunday, or any legal holiday. Contractor may perform Work outside regular working hours or on Saturdays, Sundays, or legal holidays only with Owner's written consent, which will not be unreasonably withheld.

# 7.04 Services, Materials, and Equipment

- A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start up, and completion of the Work, whether or not such items are specifically called for in the Contract Documents.
- B. All materials and equipment incorporated into the Work must be new and of good quality, except as otherwise provided in the Contract Documents. All special warranties and guarantees required by the Specifications will expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.
- C. All materials and equipment must be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.

## 7.05 *"Or Equals"*

- A. Contractor's Request; Governing Criteria: Whenever an item of equipment or material is specified or described in the Contract Documents by using the names of one or more proprietary items or specific Suppliers, the Contract Price has been based upon Contractor furnishing such item as specified. The specification or description of such an item is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or equal" item is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material, or items from other proposed Suppliers, under the circumstances described below.
  - 1. If Engineer in its sole discretion determines that an item of equipment or material proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, Engineer will deem it an "or equal" item. For the purposes of this paragraph, a proposed item of equipment or material will be considered functionally equal to an item so named if:
    - a. in the exercise of reasonable judgment Engineer determines that the proposed item:
      - 1) is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;

- 2) will reliably perform at least equally well the function and achieve the results imposed by the design concept of the completed Project as a functioning whole;
- 3) has a proven record of performance and availability of responsive service; and
- 4) is not objectionable to Owner.
- b. Contractor certifies that, if the proposed item is approved and incorporated into the Work:
  - 1) there will be no increase in cost to the Owner or increase in Contract Times; and
  - 2) the item will conform substantially to the detailed requirements of the item named in the Contract Documents.
- B. *Contractor's Expense*: Contractor shall provide all data in support of any proposed "or equal" item at Contractor's expense.
- C. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each "or-equal" request. Engineer may require Contractor to furnish additional data about the proposed "or-equal" item. Engineer will be the sole judge of acceptability. No "or-equal" item will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an "or-equal," which will be evidenced by an approved Shop Drawing or other written communication. Engineer will advise Contractor in writing of any negative determination.
- D. Effect of Engineer's Determination: Neither approval nor denial of an "or-equal" request will result in any change in Contract Price. The Engineer's denial of an "or-equal" request will be final and binding, and may not be reversed through an appeal under any provision of the Contract.
- E. Treatment as a Substitution Request: If Engineer determines that an item of equipment or material proposed by Contractor does not qualify as an "or-equal" item, Contractor may request that Engineer consider the item a proposed substitute pursuant to Paragraph 7.06.

### 7.06 Substitutes

- A. Contractor's Request; Governing Criteria: Unless the specification or description of an item of equipment or material required to be furnished under the Contract Documents contains or is followed by words reading that no substitution is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material under the circumstances described below. To the extent possible such requests must be made before commencement of related construction at the Site.
  - Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is functionally equivalent to that named and an acceptable substitute therefor. Engineer will not accept requests for review of proposed substitute items of equipment or material from anyone other than Contractor.
  - 2. The requirements for review by Engineer will be as set forth in Paragraph 7.06.B, as supplemented by the Specifications, and as Engineer may decide is appropriate under the circumstances.

- 3. Contractor shall make written application to Engineer for review of a proposed substitute item of equipment or material that Contractor seeks to furnish or use. The application:
  - a. will certify that the proposed substitute item will:
    - 1) perform adequately the functions and achieve the results called for by the general design;
    - 2) be similar in substance to the item specified; and
    - 3) be suited to the same use as the item specified.

### b. will state:

- 1) the extent, if any, to which the use of the proposed substitute item will necessitate a change in Contract Times;
- 2) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item; and
- 3) whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty.
- c. will identify:
  - 1) all variations of the proposed substitute item from the item specified; and
  - 2) available engineering, sales, maintenance, repair, and replacement services.
- d. will contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including but not limited to changes in Contract Price, shared savings, costs of redesign, and claims of other contractors affected by any resulting change.
- B. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each substitute request, and to obtain comments and direction from Owner. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No substitute will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an acceptable substitute. Engineer's determination will be evidenced by a Field Order or a proposed Change Order accounting for the substitution itself and all related impacts, including changes in Contract Price or Contract Times. Engineer will advise Contractor in writing of any negative determination.
- C. *Special Guarantee*: Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- D. Reimbursement of Engineer's Cost: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for making changes in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.

- E. *Contractor's Expense*: Contractor shall provide all data in support of any proposed substitute at Contractor's expense.
- F. Effect of Engineer's Determination: If Engineer approves the substitution request, Contractor shall execute the proposed Change Order and proceed with the substitution. The Engineer's denial of a substitution request will be final and binding, and may not be reversed through an appeal under any provision of the Contract. Contractor may challenge the scope of reimbursement costs imposed under Paragraph 7.06.D, by timely submittal of a Change Proposal.

# 7.07 Concerning Subcontractors and Suppliers

- A. Contractor may retain Subcontractors and Suppliers for the performance of parts of the Work. Such Subcontractors and Suppliers must be acceptable to Owner. The Contractor's retention of a Subcontractor or Supplier for the performance of parts of the Work will not relieve Contractor's obligation to Owner to perform and complete the Work in accordance with the Contract Documents.
- B. Contractor shall retain specific Subcontractors and Suppliers for the performance of designated parts of the Work if required by the Contract to do so.
- C. Subsequent to the submittal of Contractor's Bid or final negotiation of the terms of the Contract, Owner may not require Contractor to retain any Subcontractor or Supplier to furnish or perform any of the Work against which Contractor has reasonable objection.
- D. Prior to entry into any binding subcontract or purchase order, Contractor shall submit to Owner the identity of the proposed Subcontractor or Supplier (unless Owner has already deemed such proposed Subcontractor or Supplier acceptable during the bidding process or otherwise). Such proposed Subcontractor or Supplier shall be deemed acceptable to Owner unless Owner raises a substantive, reasonable objection within 5 days.
- E. Owner may require the replacement of any Subcontractor or Supplier. Owner also may require Contractor to retain specific replacements; provided, however, that Owner may not require a replacement to which Contractor has a reasonable objection. If Contractor has submitted the identity of certain Subcontractors or Suppliers for acceptance by Owner, and Owner has accepted it (either in writing or by failing to make written objection thereto), then Owner may subsequently revoke the acceptance of any such Subcontractor or Supplier so identified solely on the basis of substantive, reasonable objection after due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor or Supplier.
- F. If Owner requires the replacement of any Subcontractor or Supplier retained by Contractor to perform any part of the Work, then Contractor shall be entitled to an adjustment in Contract Price or Contract Times, with respect to the replacement; and Contractor shall initiate a Change Proposal for such adjustment within 30 days of Owner's requirement of replacement.
- G. No acceptance by Owner of any such Subcontractor or Supplier, whether initially or as a replacement, will constitute a waiver of the right of Owner to the completion of the Work in accordance with the Contract Documents.

- H. On a monthly basis, Contractor shall submit to Engineer a complete list of all Subcontractors and Suppliers having a direct contract with Contractor, and of all other Subcontractors and Suppliers known to Contractor at the time of submittal.
- I. Contractor shall be solely responsible for scheduling and coordinating the work of Subcontractors and Suppliers.
- J. The divisions and sections of the Specifications and the identifications of any Drawings do not control Contractor in dividing the Work among Subcontractors or Suppliers, or in delineating the Work to be performed by any specific trade.
- K. All Work performed for Contractor by a Subcontractor or Supplier must be pursuant to an appropriate contractual agreement that specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract for the benefit of Owner and Engineer.
- L. Owner may furnish to any Subcontractor or Supplier, to the extent practicable, information about amounts paid to Contractor for Work performed for Contractor by the Subcontractor or Supplier.
- M. Contractor shall restrict all Subcontractors and Suppliers from communicating with Engineer or Owner, except through Contractor or in case of an emergency, or as otherwise expressly allowed in this Contract.

## 7.08 Patent Fees and Royalties

- A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If an invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Owner or Engineer, its use is subject to patent rights or copyrights calling for the payment of any license fee or royalty to others, the existence of such rights will be disclosed in the Contract Documents.
- B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights.
- C. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.

#### 7.09 Permits

A. Unless otherwise provided in the Contract Documents, Contractor shall obtain and pay for all construction permits, licenses, and certificates of occupancy. Owner shall assist Contractor, when necessary, in obtaining such permits and licenses. Contractor shall pay all governmental charges and inspection fees necessary for the prosecution of the Work which are applicable at the time of the submission of Contractor's Bid (or when Contractor became bound under a negotiated contract). Owner shall pay all charges of utility owners for connections for providing permanent service to the Work.

### 7.10 *Taxes*

A. Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

## 7.11 Laws and Regulations

- A. Contractor shall give all notices required by and shall comply with all Laws and Regulations applicable to the performance of the Work. Neither Owner nor Engineer shall be responsible for monitoring Contractor's compliance with any Laws or Regulations.
- B. If Contractor performs any Work or takes any other action knowing or having reason to know that it is contrary to Laws or Regulations, Contractor shall bear all resulting costs and losses, and shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such Work or other action. It is not Contractor's responsibility to make certain that the Work described in the Contract Documents is in accordance with Laws and Regulations, but this does not relieve Contractor of its obligations under Paragraph 3.03.
- C. Owner or Contractor may give written notice to the other party of any changes after the submission of Contractor's Bid (or after the date when Contractor became bound under a negotiated contract) in Laws or Regulations having an effect on the cost or time of performance of the Work, including but not limited to changes in Laws or Regulations having an effect on procuring permits and on sales, use, value-added, consumption, and other similar taxes. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times resulting from such changes, then within 30 days of such written notice Contractor may submit a Change Proposal, or Owner may initiate a Claim.

#### 7.12 Record Documents

A. Contractor shall maintain in a safe place at the Site one printed record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, written interpretations and clarifications, and approved Shop Drawings. Contractor shall keep such record documents in good order and annotate them to show changes made during construction. These record documents, together with all approved Samples, will be available to Engineer for reference. Upon completion of the Work, Contractor shall deliver these record documents to Engineer.

# 7.13 Safety and Protection

- A. Contractor shall be solely responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the Work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety Laws and Regulations.
- B. Contractor shall designate a qualified and experienced safety representative whose duties and responsibilities are the prevention of Work-related accidents and the maintenance and supervision of safety precautions and programs.
- C. Contractor shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury, or loss to:
  - 1. all persons on the Site or who may be affected by the Work;
  - 2. all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and
  - 3. other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, other work in progress, utilities, and Underground Facilities not designated for removal, relocation, or replacement in the course of construction.
- D. All damage, injury, or loss to any property referred to in Paragraph 7.13.C.2 or 7.13.C.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor at its expense (except damage or loss attributable to the fault of Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone employed by any of them, or anyone for whose acts any of them may be liable, and not attributable, directly or indirectly, in whole or in part, to the fault or negligence of Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly employed by any of them).
- E. Contractor shall comply with all applicable Laws and Regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and shall erect and maintain all necessary safeguards for such safety and protection.
- F. Contractor shall notify Owner; the owners of adjacent property; the owners of Underground Facilities and other utilities (if the identity of such owners is known to Contractor); and other contractors and utility owners performing work at or adjacent to the Site, in writing, when Contractor knows that prosecution of the Work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property or work in progress.
- G. Contractor shall comply with the applicable requirements of Owner's safety programs, if any. Any Owner's safety programs that are applicable to the Work are identified or included in the Supplementary Conditions or Specifications.
- H. Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.

- I. Contractor's duties and responsibilities for safety and protection will continue until all the Work is completed, Engineer has issued a written notice to Owner and Contractor in accordance with Paragraph 15.06.C that the Work is acceptable, and Contractor has left the Site (except as otherwise expressly provided in connection with Substantial Completion).
- J. Contractor's duties and responsibilities for safety and protection will resume whenever Contractor or any Subcontractor or Supplier returns to the Site to fulfill warranty or correction obligations, or to conduct other tasks arising from the Contract Documents.

## 7.14 Hazard Communication Programs

A. Contractor shall be responsible for coordinating any exchange of safety data sheets (formerly known as material safety data sheets) or other hazard communication information required to be made available to or exchanged between or among employers at the Site in accordance with Laws or Regulations.

# 7.15 *Emergencies*

A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused by an emergency, or are required as a result of Contractor's response to an emergency. If Engineer determines that a change in the Contract Documents is required because of an emergency or Contractor's response, a Work Change Directive or Change Order will be issued.

#### 7.16 Submittals

- A. Shop Drawing and Sample Requirements
  - 1. Before submitting a Shop Drawing or Sample, Contractor shall:
    - a. review and coordinate the Shop Drawing or Sample with other Shop Drawings and Samples and with the requirements of the Work and the Contract Documents;
    - b. determine and verify:
      - 1) all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect to the Submittal;
      - 2) the suitability of all materials and equipment offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and
      - all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto;
    - c. confirm that the Submittal is complete with respect to all related data included in the Submittal.
  - Each Shop Drawing or Sample must bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review of that Submittal, and that Contractor approves the Submittal.

- 3. With each Shop Drawing or Sample, Contractor shall give Engineer specific written notice of any variations that the Submittal may have from the requirements of the Contract Documents. This notice must be set forth in a written communication separate from the Submittal; and, in addition, in the case of a Shop Drawing by a specific notation made on the Shop Drawing itself.
- B. Submittal Procedures for Shop Drawings and Samples: Contractor shall label and submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals.

# 1. Shop Drawings

- a. Contractor shall submit the number of copies required in the Specifications.
- b. Data shown on the Shop Drawings must be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to provide, and to enable Engineer to review the information for the limited purposes required by Paragraph 7.16.C.

#### 2. Samples

- a. Contractor shall submit the number of Samples required in the Specifications.
- b. Contractor shall clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the Submittal for the limited purposes required by Paragraph 7.16.C.
- 3. Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole expense and responsibility of Contractor.

### C. Engineer's Review of Shop Drawings and Samples

- Engineer will provide timely review of Shop Drawings and Samples in accordance with the
  accepted Schedule of Submittals. Engineer's review and approval will be only to
  determine if the items covered by the Submittals will, after installation or incorporation
  in the Work, comply with the requirements of the Contract Documents, and be
  compatible with the design concept of the completed Project as a functioning whole as
  indicated by the Contract Documents.
- 2. Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction, or to safety precautions or programs incident thereto.
- 3. Engineer's review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.
- 4. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 7.16.A.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer will

- document any such approved variation from the requirements of the Contract Documents in a Field Order or other appropriate Contract modification.
- 5. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for complying with the requirements of Paragraphs 7.16.A and B.
- 6. Engineer's review and approval of a Shop Drawing or Sample, or of a variation from the requirements of the Contract Documents, will not, under any circumstances, change the Contract Times or Contract Price, unless such changes are included in a Change Order.
- 7. Neither Engineer's receipt, review, acceptance, or approval of a Shop Drawing or Sample will result in such item becoming a Contract Document.
- 8. Contractor shall perform the Work in compliance with the requirements and commitments set forth in approved Shop Drawings and Samples, subject to the provisions of Paragraph 7.16.C.4.

## D. Resubmittal Procedures for Shop Drawings and Samples

- Contractor shall make corrections required by Engineer and shall return the required number of corrected copies of Shop Drawings and submit, as required, new Samples for review and approval. Contractor shall direct specific attention in writing to revisions other than the corrections called for by Engineer on previous Submittals.
- 2. Contractor shall furnish required Shop Drawing and Sample submittals with sufficient information and accuracy to obtain required approval of an item with no more than two resubmittals. Engineer will record Engineer's time for reviewing a third or subsequent resubmittal of a Shop Drawing or Sample, and Contractor shall be responsible for Engineer's charges to Owner for such time. Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges.
- 3. If Contractor requests a change of a previously approved Shop Drawing or Sample, Contractor shall be responsible for Engineer's charges to Owner for its review time, and Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges, unless the need for such change is beyond the control of Contractor.

# E. Submittals Other than Shop Drawings, Samples, and Owner-Delegated Designs

- 1. The following provisions apply to all Submittals other than Shop Drawings, Samples, and Owner-delegated designs:
  - a. Contractor shall submit all such Submittals to the Engineer in accordance with the Schedule of Submittals and pursuant to the applicable terms of the Contract Documents.
  - b. Engineer will provide timely review of all such Submittals in accordance with the Schedule of Submittals and return such Submittals with a notation of either Accepted or Not Accepted. Any such Submittal that is not returned within the time established in the Schedule of Submittals will be deemed accepted.
  - c. Engineer's review will be only to determine if the Submittal is acceptable under the requirements of the Contract Documents as to general form and content of the Submittal.

- d. If any such Submittal is not accepted, Contractor shall confer with Engineer regarding the reason for the non-acceptance, and resubmit an acceptable document.
- 2. Procedures for the submittal and acceptance of the Progress Schedule, the Schedule of Submittals, and the Schedule of Values are set forth in Paragraphs 2.03. 2.04, and 2.05.
- F. Owner-delegated Designs: Submittals pursuant to Owner-delegated designs are governed by the provisions of Paragraph 7.19.

## 7.17 Contractor's General Warranty and Guarantee

- A. Contractor warrants and guarantees to Owner that all Work will be in accordance with the Contract Documents and will not be defective. Engineer is entitled to rely on Contractor's warranty and guarantee.
- B. Owner's rights under this warranty and guarantee are in addition to, and are not limited by, Owner's rights under the correction period provisions of Paragraph 15.08. The time in which Owner may enforce its warranty and guarantee rights under this Paragraph 7.17 is limited only by applicable Laws and Regulations restricting actions to enforce such rights; provided, however, that after the end of the correction period under Paragraph 15.08:
  - 1. Owner shall give Contractor written notice of any defective Work within 60 days of the discovery that such Work is defective; and
  - 2. Such notice will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the notice.
- C. Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
  - abuse, or improper modification, maintenance, or operation, by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
  - 2. normal wear and tear under normal usage.
- D. Contractor's obligation to perform and complete the Work in accordance with the Contract Documents is absolute. None of the following will constitute an acceptance of Work that is not in accordance with the Contract Documents, a release of Contractor's obligation to perform the Work in accordance with the Contract Documents, or a release of Owner's warranty and guarantee rights under this Paragraph 7.17:
  - 1. Observations by Engineer;
  - 2. Recommendation by Engineer or payment by Owner of any progress or final payment;
  - 3. The issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
  - 4. Use or occupancy of the Work or any part thereof by Owner;
  - 5. Any review and approval of a Shop Drawing or Sample submittal;
  - 6. The issuance of a notice of acceptability by Engineer;
  - 7. The end of the correction period established in Paragraph 15.08;
  - 8. Any inspection, test, or approval by others; or

- 9. Any correction of defective Work by Owner.
- E. If the Contract requires the Contractor to accept the assignment of a contract entered into by Owner, then the specific warranties, guarantees, and correction obligations contained in the assigned contract will govern with respect to Contractor's performance obligations to Owner for the Work described in the assigned contract.

## 7.18 *Indemnification*

- A. To the fullest extent permitted by Laws and Regulations, and in addition to any other obligations of Contractor under the Contract or otherwise, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from losses, damages, costs, and judgments (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising from third-party claims or actions relating to or resulting from the performance or furnishing of the Work, provided that any such claim, action, loss, cost, judgment or damage is attributable to bodily injury, sickness, disease, or death, or to damage to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom, but only to the extent caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable.
- B. In any and all claims against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, by any employee (or the survivor or personal representative of such employee) of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, the indemnification obligation under Paragraph 7.18.A will not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' compensation acts, disability benefit acts, or other employee benefit acts.

#### 7.19 Delegation of Professional Design Services

- A. Owner may require Contractor to provide professional design services for a portion of the Work by express delegation in the Contract Documents. Such delegation will specify the performance and design criteria that such services must satisfy, and the Submittals that Contractor must furnish to Engineer with respect to the Owner-delegated design.
- B. Contractor shall cause such Owner-delegated professional design services to be provided pursuant to the professional standard of care by a properly licensed design professional, whose signature and seal must appear on all drawings, calculations, specifications, certifications, and Submittals prepared by such design professional. Such design professional must issue all certifications of design required by Laws and Regulations.
- C. If a Shop Drawing or other Submittal related to the Owner-delegated design is prepared by Contractor, a Subcontractor, or others for submittal to Engineer, then such Shop Drawing or other Submittal must bear the written approval of Contractor's design professional when submitted by Contractor to Engineer.

- D. Owner and Engineer shall be entitled to rely upon the adequacy, accuracy, and completeness of the services, certifications, and approvals performed or provided by the design professionals retained or employed by Contractor under an Owner-delegated design, subject to the professional standard of care and the performance and design criteria stated in the Contract Documents.
- E. Pursuant to this Paragraph 7.19, Engineer's review, approval, and other determinations regarding design drawings, calculations, specifications, certifications, and other Submittals furnished by Contractor pursuant to an Owner-delegated design will be only for the following limited purposes:
  - 1. Checking for conformance with the requirements of this Paragraph 7.19;
  - 2. Confirming that Contractor (through its design professionals) has used the performance and design criteria specified in the Contract Documents; and
  - 3. Establishing that the design furnished by Contractor is consistent with the design concept expressed in the Contract Documents.
- F. Contractor shall not be responsible for the adequacy of performance or design criteria specified by Owner or Engineer.
- G. Contractor is not required to provide professional services in violation of applicable Laws and Regulations.

### ARTICLE 8—OTHER WORK AT THE SITE

#### 8.01 Other Work

- A. In addition to and apart from the Work under the Contract Documents, the Owner may perform other work at or adjacent to the Site. Such other work may be performed by Owner's employees, or through contracts between the Owner and third parties. Owner may also arrange to have third-party utility owners perform work on their utilities and facilities at or adjacent to the Site.
- B. If Owner performs other work at or adjacent to the Site with Owner's employees, or through contracts for such other work, then Owner shall give Contractor written notice thereof prior to starting any such other work. If Owner has advance information regarding the start of any third-party utility work that Owner has arranged to take place at or adjacent to the Site, Owner shall provide such information to Contractor.
- C. Contractor shall afford proper and safe access to the Site to each contractor that performs such other work, each utility owner performing other work, and Owner, if Owner is performing other work with Owner's employees, and provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work.
- D. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose work will be affected.

- E. If the proper execution or results of any part of Contractor's Work depends upon work performed by others, Contractor shall inspect such other work and promptly report to Engineer in writing any delays, defects, or deficiencies in such other work that render it unavailable or unsuitable for the proper execution and results of Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent defects and deficiencies in such other work.
- F. The provisions of this article are not applicable to work that is performed by third-party utilities or other third-party entities without a contract with Owner, or that is performed without having been arranged by Owner. If such work occurs, then any related delay, disruption, or interference incurred by Contractor is governed by the provisions of Paragraph 4.05.C.3.

#### 8.02 Coordination

- A. If Owner intends to contract with others for the performance of other work at or adjacent to the Site, to perform other work at or adjacent to the Site with Owner's employees, or to arrange to have utility owners perform work at or adjacent to the Site, the following will be set forth in the Supplementary Conditions or provided to Contractor prior to the start of any such other work:
  - 1. The identity of the individual or entity that will have authority and responsibility for coordination of the activities among the various contractors;
  - An itemization of the specific matters to be covered by such authority and responsibility;
  - 3. The extent of such authority and responsibilities.
- B. Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination.

# 8.03 Legal Relationships

A. If, in the course of performing other work for Owner at or adjacent to the Site, the Owner's employees, any other contractor working for Owner, or any utility owner that Owner has arranged to perform work, causes damage to the Work or to the property of Contractor or its Subcontractors, or delays, disrupts, interferes with, or increases the scope or cost of the performance of the Work, through actions or inaction, then Contractor shall be entitled to an equitable adjustment in the Contract Price or the Contract Times. Contractor must submit any Change Proposal seeking an equitable adjustment in the Contract Price or the Contract Times under this paragraph within 30 days of the damaging, delaying, disrupting, or interfering event. The entitlement to, and extent of, any such equitable adjustment will take into account information (if any) regarding such other work that was provided to Contractor in the Contract Documents prior to the submittal of the Bid or the final negotiation of the terms of the Contract, and any remedies available to Contractor under Laws or Regulations concerning utility action or inaction. When applicable, any such equitable adjustment in Contract Price will be conditioned on Contractor assigning to Owner all Contractor's rights against such other contractor or utility owner with respect to the damage, delay, disruption, or interference that is the subject of the adjustment. Contractor's entitlement to an adjustment of the Contract Times or Contract Price is subject to the provisions of Paragraphs 4.05.D and 4.05.E.

- B. Contractor shall take reasonable and customary measures to avoid damaging, delaying, disrupting, or interfering with the work of Owner, any other contractor, or any utility owner performing other work at or adjacent to the Site.
  - If Contractor fails to take such measures and as a result damages, delays, disrupts, or interferes with the work of any such other contractor or utility owner, then Owner may impose a set-off against payments due Contractor, and assign to such other contractor or utility owner the Owner's contractual rights against Contractor with respect to the breach of the obligations set forth in this Paragraph 8.03.B.
  - 2. When Owner is performing other work at or adjacent to the Site with Owner's employees, Contractor shall be liable to Owner for damage to such other work, and for the reasonable direct delay, disruption, and interference costs incurred by Owner as a result of Contractor's failure to take reasonable and customary measures with respect to Owner's other work. In response to such damage, delay, disruption, or interference, Owner may impose a set-off against payments due Contractor.
- C. If Contractor damages, delays, disrupts, or interferes with the work of any other contractor, or any utility owner performing other work at or adjacent to the Site, through Contractor's failure to take reasonable and customary measures to avoid such impacts, or if any claim arising out of Contractor's actions, inactions, or negligence in performance of the Work at or adjacent to the Site is made by any such other contractor or utility owner against Contractor, Owner, or Engineer, then Contractor shall (1) promptly attempt to settle the claim as to all parties through negotiations with such other contractor or utility owner, or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law, and (2) indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against any such claims, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such damage, delay, disruption, or interference.

# ARTICLE 9—OWNER'S RESPONSIBILITIES

- 9.01 Communications to Contractor
  - A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.
- 9.02 Replacement of Engineer
  - A. Owner may at its discretion appoint an engineer to replace Engineer, provided Contractor makes no reasonable objection to the replacement engineer. The replacement engineer's status under the Contract Documents will be that of the former Engineer.
- 9.03 Furnish Data
  - A. Owner shall promptly furnish the data required of Owner under the Contract Documents.
- 9.04 Pay When Due
  - A. Owner shall make payments to Contractor when they are due as provided in the Agreement.

- 9.05 Lands and Easements; Reports, Tests, and Drawings
  - A. Owner's duties with respect to providing lands and easements are set forth in Paragraph 5.01.
  - B. Owner's duties with respect to providing engineering surveys to establish reference points are set forth in Paragraph 4.03.
  - C. Article 5 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of conditions at the Site, and drawings of physical conditions relating to existing surface or subsurface structures at the Site.

## 9.06 Insurance

A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 6.

## 9.07 Change Orders

A. Owner's responsibilities with respect to Change Orders are set forth in Article 11.

## 9.08 Inspections, Tests, and Approvals

A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 14.02.B.

### 9.09 Limitations on Owner's Responsibilities

A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.

#### 9.10 Undisclosed Hazardous Environmental Condition

A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 5.06.

### 9.11 Evidence of Financial Arrangements

A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract (including obligations under proposed changes in the Work).

### 9.12 Safety Programs

- A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed.
- B. Owner shall furnish copies of any applicable Owner safety programs to Contractor.

#### ARTICLE 10—ENGINEER'S STATUS DURING CONSTRUCTION

# 10.01 Owner's Representative

A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract.

### 10.02 Visits to Site

- A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe, as an experienced and qualified design professional, the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform generally to the Contract Documents. On the basis of such visits and observations, Engineer will keep Owner informed of the progress of the Work and will endeavor to guard Owner against defective Work.
- B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 10.07. Particularly, but without limitation, during or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

#### 10.03 Resident Project Representative

- A. If Owner and Engineer have agreed that Engineer will furnish a Resident Project Representative to represent Engineer at the Site and assist Engineer in observing the progress and quality of the Work, then the authority and responsibilities of any such Resident Project Representative will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in the Supplementary Conditions and in Paragraph 10.07.
- B. If Owner designates an individual or entity who is not Engineer's consultant, agent, or employee to represent Owner at the Site, then the responsibilities and authority of such individual or entity will be as provided in the Supplementary Conditions.

# 10.04 Engineer's Authority

- A. Engineer has the authority to reject Work in accordance with Article 14.
- B. Engineer's authority as to Submittals is set forth in Paragraph 7.16.
- C. Engineer's authority as to design drawings, calculations, specifications, certifications and other Submittals from Contractor in response to Owner's delegation (if any) to Contractor of professional design services, is set forth in Paragraph 7.19.
- D. Engineer's authority as to changes in the Work is set forth in Article 11.

E. Engineer's authority as to Applications for Payment is set forth in Article 15.

# 10.05 Determinations for Unit Price Work

A. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor as set forth in Paragraph 13.03.

### 10.06 Decisions on Requirements of Contract Documents and Acceptability of Work

A. Engineer will render decisions regarding the requirements of the Contract Documents, and judge the acceptability of the Work, pursuant to the specific procedures set forth herein for initial interpretations, Change Proposals, and acceptance of the Work. In rendering such decisions and judgments, Engineer will not show partiality to Owner or Contractor, and will not be liable to Owner, Contractor, or others in connection with any proceedings, interpretations, decisions, or judgments conducted or rendered in good faith.

## 10.07 Limitations on Engineer's Authority and Responsibilities

- A. Neither Engineer's authority or responsibility under this Article 10 or under any other provision of the Contract, nor any decision made by Engineer in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer, will create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any surety for or employee or agent of any of them.
- B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
- D. Engineer's review of the final Application for Payment and accompanying documentation, and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Contractor under Paragraph 15.06.A, will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals, that the results certified indicate compliance with the Contract Documents.
- E. The limitations upon authority and responsibility set forth in this Paragraph 10.07 also apply to the Resident Project Representative, if any.

## 10.08 Compliance with Safety Program

A. While at the Site, Engineer's employees and representatives will comply with the specific applicable requirements of Owner's and Contractor's safety programs of which Engineer has been informed.

#### ARTICLE 11—CHANGES TO THE CONTRACT

# 11.01 Amending and Supplementing the Contract

- A. The Contract may be amended or supplemented by a Change Order, a Work Change Directive, or a Field Order.
- B. If an amendment or supplement to the Contract includes a change in the Contract Price or the Contract Times, such amendment or supplement must be set forth in a Change Order.
- C. All changes to the Contract that involve (1) the performance or acceptability of the Work, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, must be supported by Engineer's recommendation. Owner and Contractor may amend other terms and conditions of the Contract without the recommendation of the Engineer.

## 11.02 Change Orders

- A. Owner and Contractor shall execute appropriate Change Orders covering:
  - Changes in Contract Price or Contract Times which are agreed to by the parties, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive;
  - Changes in Contract Price resulting from an Owner set-off, unless Contractor has duly contested such set-off;
  - 3. Changes in the Work which are: (a) ordered by Owner pursuant to Paragraph 11.05, (b) required because of Owner's acceptance of defective Work under Paragraph 14.04 or Owner's correction of defective Work under Paragraph 14.07, or (c) agreed to by the parties, subject to the need for Engineer's recommendation if the change in the Work involves the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters; and
  - 4. Changes that embody the substance of any final and binding results under: Paragraph 11.03.B, resolving the impact of a Work Change Directive; Paragraph 11.09, concerning Change Proposals; Article 12, Claims; Paragraph 13.02.D, final adjustments resulting from allowances; Paragraph 13.03.D, final adjustments relating to determination of quantities for Unit Price Work; and similar provisions.
- B. If Owner or Contractor refuses to execute a Change Order that is required to be executed under the terms of Paragraph 11.02.A, it will be deemed to be of full force and effect, as if fully executed.

# 11.03 Work Change Directives

A. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the modification ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order, following negotiations by the parties as to the Work Change Directive's effect, if any, on the Contract Price and Contract Times; or, if negotiations are unsuccessful, by a determination under the terms of the Contract Documents governing adjustments, expressly including Paragraph 11.07 regarding change of Contract Price.

- B. If Owner has issued a Work Change Directive and:
  - 1. Contractor believes that an adjustment in Contract Times or Contract Price is necessary, then Contractor shall submit any Change Proposal seeking such an adjustment no later than 30 days after the completion of the Work set out in the Work Change Directive.
  - Owner believes that an adjustment in Contract Times or Contract Price is necessary, then
    Owner shall submit any Claim seeking such an adjustment no later than 60 days after
    issuance of the Work Change Directive.

#### 11.04 Field Orders

- A. Engineer may authorize minor changes in the Work if the changes do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Such changes will be accomplished by a Field Order and will be binding on Owner and also on Contractor, which shall perform the Work involved promptly.
- B. If Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, then before proceeding with the Work at issue, Contractor shall submit a Change Proposal as provided herein.

## 11.05 Owner-Authorized Changes in the Work

- A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work. Changes involving the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters will be supported by Engineer's recommendation.
- B. Such changes in the Work may be accomplished by a Change Order, if Owner and Contractor have agreed as to the effect, if any, of the changes on Contract Times or Contract Price; or by a Work Change Directive. Upon receipt of any such document, Contractor shall promptly proceed with the Work involved; or, in the case of a deletion in the Work, promptly cease construction activities with respect to such deleted Work. Added or revised Work must be performed under the applicable conditions of the Contract Documents.
- C. Nothing in this Paragraph 11.05 obligates Contractor to undertake work that Contractor reasonably concludes cannot be performed in a manner consistent with Contractor's safety obligations under the Contract Documents or Laws and Regulations.

### 11.06 Unauthorized Changes in the Work

A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents, as amended, modified, or supplemented, except in the case of an emergency as provided in Paragraph 7.15 or in the case of uncovering Work as provided in Paragraph 14.05.C.2.

### 11.07 Change of Contract Price

- A. The Contract Price may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Price must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment of Contract Price must comply with the provisions of Article 12.
- B. An adjustment in the Contract Price will be determined as follows:

- 1. Where the Work involved is covered by unit prices contained in the Contract Documents, then by application of such unit prices to the quantities of the items involved (subject to the provisions of Paragraph 13.03);
- Where the Work involved is not covered by unit prices contained in the Contract Documents, then by a mutually agreed lump sum (which may include an allowance for overhead and profit not necessarily in accordance with Paragraph 11.07.C.2); or
- 3. Where the Work involved is not covered by unit prices contained in the Contract Documents and the parties do not reach mutual agreement to a lump sum, then on the basis of the Cost of the Work (determined as provided in Paragraph 13.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 11.07.C).
- C. *Contractor's Fee*: When applicable, the Contractor's fee for overhead and profit will be determined as follows:
  - 1. A mutually acceptable fixed fee; or
  - 2. If a fixed fee is not agreed upon, then a fee based on the following percentages of the various portions of the Cost of the Work:
    - a. For costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2, the Contractor's fee will be 15 percent;
    - b. For costs incurred under Paragraph 13.01.B.3, the Contractor's fee will be 5 percent;
    - c. Where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 11.07.C.2.a and 11.07.C.2.b is that the Contractor's fee will be based on: (1) a fee of 15 percent of the costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2 by the Subcontractor that actually performs the Work, at whatever tier, and (2) with respect to Contractor itself and to any Subcontractors of a tier higher than that of the Subcontractor that actually performs the Work, a fee of 5 percent of the amount (fee plus underlying costs incurred) attributable to the next lower tier Subcontractor; provided, however, that for any such subcontracted Work the maximum total fee to be paid by Owner will be no greater than 27 percent of the costs incurred by the Subcontractor that actually performs the Work;
    - d. No fee will be payable on the basis of costs itemized under Paragraphs 13.01.B.4, 13.01.B.5, and 13.01.C;
    - e. The amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in Cost of the Work will be the amount of the actual net decrease in Cost of the Work and a deduction of an additional amount equal to 5 percent of such actual net decrease in Cost of the Work; and
    - f. When both additions and credits are involved in any one change or Change Proposal, the adjustment in Contractor's fee will be computed by determining the sum of the costs in each of the cost categories in Paragraph 13.01.B (specifically, payroll costs, Paragraph 13.01.B.1; incorporated materials and equipment costs, Paragraph 13.01.B.2; Subcontract costs, Paragraph 13.01.B.3; special consultants costs, Paragraph 13.01.B.4; and other costs, Paragraph 13.01.B.5) and applying to each such cost category sum the appropriate fee from Paragraphs 11.07.C.2.a through 11.07.C.2.e, inclusive.

#### 11.08 Change of Contract Times

- A. The Contract Times may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Times must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment in the Contract Times must comply with the provisions of Article 12.
- B. Delay, disruption, and interference in the Work, and any related changes in Contract Times, are addressed in and governed by Paragraph 4.05.

# 11.09 Change Proposals

A. Purpose and Content: Contractor shall submit a Change Proposal to Engineer to request an adjustment in the Contract Times or Contract Price; contest an initial decision by Engineer concerning the requirements of the Contract Documents or relating to the acceptability of the Work under the Contract Documents; challenge a set-off against payment due; or seek other relief under the Contract. The Change Proposal will specify any proposed change in Contract Times or Contract Price, or other proposed relief, and explain the reason for the proposed change, with citations to any governing or applicable provisions of the Contract Documents. Each Change Proposal will address only one issue, or a set of closely related issues.

# B. Change Proposal Procedures

- 1. *Submittal*: Contractor shall submit each Change Proposal to Engineer within 30 days after the start of the event giving rise thereto, or after such initial decision.
- 2. Supporting Data: The Contractor shall submit supporting data, including the proposed change in Contract Price or Contract Time (if any), to the Engineer and Owner within 15 days after the submittal of the Change Proposal.
  - a. Change Proposals based on or related to delay, interruption, or interference must comply with the provisions of Paragraphs 4.05.D and 4.05.E.
  - b. Change proposals related to a change of Contract Price must include full and detailed accounts of materials incorporated into the Work and labor and equipment used for the subject Work.

The supporting data must be accompanied by a written statement that the supporting data are accurate and complete, and that any requested time or price adjustment is the entire adjustment to which Contractor believes it is entitled as a result of said event.

- 3. Engineer's Initial Review: Engineer will advise Owner regarding the Change Proposal, and consider any comments or response from Owner regarding the Change Proposal. If in its discretion Engineer concludes that additional supporting data is needed before conducting a full review and making a decision regarding the Change Proposal, then Engineer may request that Contractor submit such additional supporting data by a date specified by Engineer, prior to Engineer beginning its full review of the Change Proposal.
- 4. Engineer's Full Review and Action on the Change Proposal: Upon receipt of Contractor's supporting data (including any additional data requested by Engineer), Engineer will conduct a full review of each Change Proposal and, within 30 days after such receipt of the Contractor's supporting data, either approve the Change Proposal in whole, deny it in whole, or approve it in part and deny it in part. Such actions must be in writing, with a copy provided to Owner and Contractor. If Engineer does not take action on the Change

Proposal within 30 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of Engineer's inaction the Change Proposal is deemed denied, thereby commencing the time for appeal of the denial under Article 12.

- 5. *Binding Decision*: Engineer's decision is final and binding upon Owner and Contractor, unless Owner or Contractor appeals the decision by filing a Claim under Article 12.
- C. Resolution of Certain Change Proposals: If the Change Proposal does not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters, then Engineer will notify the parties in writing that the Engineer is unable to resolve the Change Proposal. For purposes of further resolution of such a Change Proposal, such notice will be deemed a denial, and Contractor may choose to seek resolution under the terms of Article 12.
- D. *Post-Completion*: Contractor shall not submit any Change Proposals after Engineer issues a written recommendation of final payment pursuant to Paragraph 15.06.B.

### 11.10 Notification to Surety

A. If the provisions of any bond require notice to be given to a surety of any change affecting the general scope of the Work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Times), the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

### **ARTICLE 12—CLAIMS**

#### 12.01 *Claims*

- A. *Claims Process*: The following disputes between Owner and Contractor are subject to the Claims process set forth in this article:
  - 1. Appeals by Owner or Contractor of Engineer's decisions regarding Change Proposals;
  - 2. Owner demands for adjustments in the Contract Price or Contract Times, or other relief under the Contract Documents;
  - Disputes that Engineer has been unable to address because they do not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters; and
  - 4. Subject to the waiver provisions of Paragraph 15.07, any dispute arising after Engineer has issued a written recommendation of final payment pursuant to Paragraph 15.06.B.
- B. Submittal of Claim: The party submitting a Claim shall deliver it directly to the other party to the Contract promptly (but in no event later than 30 days) after the start of the event giving rise thereto; in the case of appeals regarding Change Proposals within 30 days of the decision under appeal. The party submitting the Claim shall also furnish a copy to the Engineer, for its information only. The responsibility to substantiate a Claim rests with the party making the Claim. In the case of a Claim by Contractor seeking an increase in the Contract Times or Contract Price, Contractor shall certify that the Claim is made in good faith, that the supporting data are accurate and complete, and that to the best of Contractor's knowledge

- and belief the amount of time or money requested accurately reflects the full amount to which Contractor is entitled.
- C. Review and Resolution: The party receiving a Claim shall review it thoroughly, giving full consideration to its merits. The two parties shall seek to resolve the Claim through the exchange of information and direct negotiations. The parties may extend the time for resolving the Claim by mutual agreement. All actions taken on a Claim will be stated in writing and submitted to the other party, with a copy to Engineer.

#### D. Mediation

- 1. At any time after initiation of a Claim, Owner and Contractor may mutually agree to mediation of the underlying dispute. The agreement to mediate will stay the Claim submittal and response process.
- 2. If Owner and Contractor agree to mediation, then after 60 days from such agreement, either Owner or Contractor may unilaterally terminate the mediation process, and the Claim submittal and decision process will resume as of the date of the termination. If the mediation proceeds but is unsuccessful in resolving the dispute, the Claim submittal and decision process will resume as of the date of the conclusion of the mediation, as determined by the mediator.
- 3. Owner and Contractor shall each pay one-half of the mediator's fees and costs.
- E. *Partial Approval*: If the party receiving a Claim approves the Claim in part and denies it in part, such action will be final and binding unless within 30 days of such action the other party invokes the procedure set forth in Article 17 for final resolution of disputes.
- F. Denial of Claim: If efforts to resolve a Claim are not successful, the party receiving the Claim may deny it by giving written notice of denial to the other party. If the receiving party does not take action on the Claim within 90 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of the inaction, the Claim is deemed denied, thereby commencing the time for appeal of the denial. A denial of the Claim will be final and binding unless within 30 days of the denial the other party invokes the procedure set forth in Article 17 for the final resolution of disputes.
- G. Final and Binding Results: If the parties reach a mutual agreement regarding a Claim, whether through approval of the Claim, direct negotiations, mediation, or otherwise; or if a Claim is approved in part and denied in part, or denied in full, and such actions become final and binding; then the results of the agreement or action on the Claim will be incorporated in a Change Order or other written document to the extent they affect the Contract, including the Work, the Contract Times, or the Contract Price.

## ARTICLE 13—COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

# 13.01 Cost of the Work

- A. Purposes for Determination of Cost of the Work: The term Cost of the Work means the sum of all costs necessary for the proper performance of the Work at issue, as further defined below. The provisions of this Paragraph 13.01 are used for two distinct purposes:
  - 1. To determine Cost of the Work when Cost of the Work is a component of the Contract Price, under cost-plus-fee, time-and-materials, or other cost-based terms; or

- 2. When needed to determine the value of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price. When the value of any such adjustment is determined on the basis of Cost of the Work, Contractor is entitled only to those additional or incremental costs required because of the change in the Work or because of the event giving rise to the adjustment.
- B. Costs Included: Except as otherwise may be agreed to in writing by Owner, costs included in the Cost of the Work will be in amounts no higher than those commonly incurred in the locality of the Project, will not include any of the costs itemized in Paragraph 13.01.C, and will include only the following items:
  - 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor in advance of the subject Work. Such employees include, without limitation, superintendents, foremen, safety managers, safety representatives, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work will be apportioned on the basis of their time spent on the Work. Payroll costs include, but are not limited to, salaries and wages plus the cost of fringe benefits, which include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, sick leave, and vacation and holiday pay applicable thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, will be included in the above to the extent authorized by Owner.
  - 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts will accrue to Owner. All trade discounts, rebates, and refunds and returns from sale of surplus materials and equipment will accrue to Owner, and Contractor shall make provisions so that they may be obtained.
  - 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, which will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee will be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 13.01.
  - 4. Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed or retained for services specifically related to the Work.
  - 5. Other costs consisting of the following:
    - a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
    - b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, which are

consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.

 In establishing included costs for materials such as scaffolding, plating, or sheeting, consideration will be given to the actual or the estimated life of the material for use on other projects; or rental rates may be established on the basis of purchase or salvage value of such items, whichever is less. Contractor will not be eligible for compensation for such items in an amount that exceeds the purchase cost of such item.

#### c. Construction Equipment Rental

- 1) Rentals of all construction equipment and machinery, and the parts thereof, in accordance with rental agreements approved by Owner as to price (including any surcharge or special rates applicable to overtime use of the construction equipment or machinery), and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs will be in accordance with the terms of said rental agreements. The rental of any such equipment, machinery, or parts must cease when the use thereof is no longer necessary for the Work.
- 2) Costs for equipment and machinery owned by Contractor or a Contractor-related entity will be paid at a rate shown for such equipment in the equipment rental rate book specified in the Supplementary Conditions. An hourly rate will be computed by dividing the monthly rates by 176. These computed rates will include all operating costs.
- 3) With respect to Work that is the result of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price ("changed Work"), included costs will be based on the time the equipment or machinery is in use on the changed Work and the costs of transportation, loading, unloading, assembly, dismantling, and removal when directly attributable to the changed Work. The cost of any such equipment or machinery, or parts thereof, must cease to accrue when the use thereof is no longer necessary for the changed Work.
- d. Sales, consumer, use, and other similar taxes related to the Work, and for which Contractor is liable, as imposed by Laws and Regulations.
- e. Deposits lost for causes other than negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
- f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of builder's risk or other property insurance established in accordance with Paragraph 6.04), provided such losses and damages have resulted from causes other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable. Such losses include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses will be included in the Cost of the Work for the purpose of determining Contractor's fee.

- g. The cost of utilities, fuel, and sanitary facilities at the Site.
- h. Minor expenses such as communication service at the Site, express and courier services, and similar petty cash items in connection with the Work.
- i. The costs of premiums for all bonds and insurance that Contractor is required by the Contract Documents to purchase and maintain.
- C. Costs Excluded: The term Cost of the Work does not include any of the following items:
  - 1. Payroll costs and other compensation of Contractor's officers, executives, principals, general managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 13.01.B.1 or specifically covered by Paragraph 13.01.B.4. The payroll costs and other compensation excluded here are to be considered administrative costs covered by the Contractor's fee.
  - 2. The cost of purchasing, renting, or furnishing small tools and hand tools.
  - 3. Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.
  - 4. Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.
  - 5. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.
  - 6. Expenses incurred in preparing and advancing Claims.
  - 7. Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraph 13.01.B.

#### D. Contractor's Fee

- 1. When the Work as a whole is performed on the basis of cost-plus-a-fee, then:
  - a. Contractor's fee for the Work set forth in the Contract Documents as of the Effective Date of the Contract will be determined as set forth in the Agreement.
  - b. for any Work covered by a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work, Contractor's fee will be determined as follows:
    - 1) When the fee for the Work as a whole is a percentage of the Cost of the Work, the fee will automatically adjust as the Cost of the Work changes.
    - 2) When the fee for the Work as a whole is a fixed fee, the fee for any additions or deletions will be determined in accordance with Paragraph 11.07.C.2.
- 2. When the Work as a whole is performed on the basis of a stipulated sum, or any other basis other than cost-plus-a-fee, then Contractor's fee for any Work covered by a Change

Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work will be determined in accordance with Paragraph 11.07.C.2.

E. Documentation and Audit: Whenever the Cost of the Work for any purpose is to be determined pursuant to this Article 13, Contractor and pertinent Subcontractors will establish and maintain records of the costs in accordance with generally accepted accounting practices. Subject to prior written notice, Owner will be afforded reasonable access, during normal business hours, to all Contractor's accounts, records, books, correspondence, instructions, drawings, receipts, vouchers, memoranda, and similar data relating to the Cost of the Work and Contractor's fee. Contractor shall preserve all such documents for a period of three years after the final payment by Owner. Pertinent Subcontractors will afford such access to Owner, and preserve such documents, to the same extent required of Contractor.

#### 13.02 Allowances

- A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.
- B. Cash Allowances: Contractor agrees that:
  - the cash allowances include the cost to Contractor (less any applicable trade discounts)
    of materials and equipment required by the allowances to be delivered at the Site, and
    all applicable taxes; and
  - 2. Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment for any of the foregoing will be valid.
- C. *Owner's Contingency Allowance*: Contractor agrees that an Owner's contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.
- D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor for Work covered by allowances, and the Contract Price will be correspondingly adjusted.

#### 13.03 Unit Price Work

- A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.
- B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Payments to Contractor for Unit Price Work will be based on actual quantities.
- C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.
- D. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision

thereon will be final and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, and the final adjustment of Contract Price will be set forth in a Change Order, subject to the provisions of the following paragraph.

# E. Adjustments in Unit Price

- 1. Contractor or Owner shall be entitled to an adjustment in the unit price with respect to an item of Unit Price Work if:
  - a. the quantity of the item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement; and
  - b. Contractor's unit costs to perform the item of Unit Price Work have changed materially and significantly as a result of the quantity change.
- 2. The adjustment in unit price will account for and be coordinated with any related changes in quantities of other items of Work, and in Contractor's costs to perform such other Work, such that the resulting overall change in Contract Price is equitable to Owner and Contractor.
- 3. Adjusted unit prices will apply to all units of that item.

#### ARTICLE 14—TESTS AND INSPECTIONS; CORRECTION, REMOVAL, OR ACCEPTANCE OF DEFECTIVE WORK

#### 14.01 Access to Work

A. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and authorities having jurisdiction have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply with such procedures and programs as applicable.

#### 14.02 Tests, Inspections, and Approvals

- A. Contractor shall give Engineer timely notice of readiness of the Work (or specific parts thereof) for all required inspections and tests, and shall cooperate with inspection and testing personnel to facilitate required inspections and tests.
- B. Owner shall retain and pay for the services of an independent inspector, testing laboratory, or other qualified individual or entity to perform all inspections and tests expressly required by the Contract Documents to be furnished and paid for by Owner, except that costs incurred in connection with tests or inspections of covered Work will be governed by the provisions of Paragraph 14.05.
- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.

- D. Contractor shall be responsible for arranging, obtaining, and paying for all inspections and tests required:
  - 1. by the Contract Documents, unless the Contract Documents expressly allocate responsibility for a specific inspection or test to Owner;
  - 2. to attain Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work;
  - 3. by manufacturers of equipment furnished under the Contract Documents;
  - 4. for testing, adjusting, and balancing of mechanical, electrical, and other equipment to be incorporated into the Work; and
  - 5. for acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work.

Such inspections and tests will be performed by independent inspectors, testing laboratories, or other qualified individuals or entities acceptable to Owner and Engineer.

- E. If the Contract Documents require the Work (or part thereof) to be approved by Owner, Engineer, or another designated individual or entity, then Contractor shall assume full responsibility for arranging and obtaining such approvals.
- F. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation. Such uncovering will be at Contractor's expense unless Contractor had given Engineer timely notice of Contractor's intention to cover the same and Engineer had not acted with reasonable promptness in response to such notice.

#### 14.03 Defective Work

- A. Contractor's Obligation: It is Contractor's obligation to assure that the Work is not defective.
- B. *Engineer's Authority*: Engineer has the authority to determine whether Work is defective, and to reject defective Work.
- C. *Notice of Defects*: Prompt written notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor.
- D. Correction, or Removal and Replacement: Promptly after receipt of written notice of defective Work, Contractor shall correct all such defective Work, whether or not fabricated, installed, or completed, or, if Engineer has rejected the defective Work, remove it from the Project and replace it with Work that is not defective.
- E. *Preservation of Warranties*: When correcting defective Work, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.
- F. Costs and Damages: In addition to its correction, removal, and replacement obligations with respect to defective Work, Contractor shall pay all claims, costs, losses, and damages arising out of or relating to defective Work, including but not limited to the cost of the inspection, testing, correction, removal, replacement, or reconstruction of such defective Work, fines levied against Owner by governmental authorities because the Work is defective, and the costs of repair or replacement of work of others resulting from defective Work. Prior to final payment, if Owner and Contractor are unable to agree as to the measure of such claims, costs,

losses, and damages resulting from defective Work, then Owner may impose a reasonable set-off against payments due under Article 15.

# 14.04 Acceptance of Defective Work

A. If, instead of requiring correction or removal and replacement of defective Work, Owner prefers to accept it, Owner may do so (subject, if such acceptance occurs prior to final payment, to Engineer's confirmation that such acceptance is in general accord with the design intent and applicable engineering principles, and will not endanger public safety). Contractor shall pay all claims, costs, losses, and damages attributable to Owner's evaluation of and determination to accept such defective Work (such costs to be approved by Engineer as to reasonableness), and for the diminished value of the Work to the extent not otherwise paid by Contractor. If any such acceptance occurs prior to final payment, the necessary revisions in the Contract Documents with respect to the Work will be incorporated in a Change Order. If the parties are unable to agree as to the decrease in the Contract Price, reflecting the diminished value of Work so accepted, then Owner may impose a reasonable set-off against payments due under Article 15. If the acceptance of defective Work occurs after final payment, Contractor shall pay an appropriate amount to Owner.

#### 14.05 Uncovering Work

- A. Engineer has the authority to require additional inspection or testing of the Work, whether or not the Work is fabricated, installed, or completed.
- B. If any Work is covered contrary to the written request of Engineer, then Contractor shall, if requested by Engineer, uncover such Work for Engineer's observation, and then replace the covering, all at Contractor's expense.
- C. If Engineer considers it necessary or advisable that covered Work be observed by Engineer or inspected or tested by others, then Contractor, at Engineer's request, shall uncover, expose, or otherwise make available for observation, inspection, or testing as Engineer may require, that portion of the Work in question, and provide all necessary labor, material, and equipment.
  - 1. If it is found that the uncovered Work is defective, Contractor shall be responsible for all claims, costs, losses, and damages arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or reconstruction (including but not limited to all costs of repair or replacement of work of others); and pending Contractor's full discharge of this responsibility the Owner shall be entitled to impose a reasonable set-off against payments due under Article 15.
  - 2. If the uncovered Work is not found to be defective, Contractor shall be allowed an increase in the Contract Price or an extension of the Contract Times, directly attributable to such uncovering, exposure, observation, inspection, testing, replacement, and reconstruction. If the parties are unable to agree as to the amount or extent thereof, then Contractor may submit a Change Proposal within 30 days of the determination that the Work is not defective.

#### 14.06 Owner May Stop the Work

A. If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed Work will conform to the Contract Documents, then Owner may order Contractor to stop the Work,

or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work will not give rise to any duty on the part of Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or entity, or any surety for, or employee or agent of any of them.

# 14.07 Owner May Correct Defective Work

- A. If Contractor fails within a reasonable time after written notice from Engineer to correct defective Work, or to remove and replace defective Work as required by Engineer, then Owner may, after 7 days' written notice to Contractor, correct or remedy any such deficiency.
- B. In exercising the rights and remedies under this Paragraph 14.07, Owner shall proceed expeditiously. In connection with such corrective or remedial action, Owner may exclude Contractor from all or part of the Site, take possession of all or part of the Work and suspend Contractor's services related thereto, and incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere. Contractor shall allow Owner, Owner's representatives, agents and employees, Owner's other contractors, and Engineer and Engineer's consultants access to the Site to enable Owner to exercise the rights and remedies under this paragraph.
- C. All claims, costs, losses, and damages incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 14.07 will be charged against Contractor as set-offs against payments due under Article 15. Such claims, costs, losses and damages will include but not be limited to all costs of repair, or replacement of work of others destroyed or damaged by correction, removal, or replacement of Contractor's defective Work.
- D. Contractor shall not be allowed an extension of the Contract Times because of any delay in the performance of the Work attributable to the exercise by Owner of Owner's rights and remedies under this Paragraph 14.07.

#### ARTICLE 15—PAYMENTS TO CONTRACTOR; SET-OFFS; COMPLETION; CORRECTION PERIOD

#### 15.01 Progress Payments

A. Basis for Progress Payments: The Schedule of Values established as provided in Article 2 will serve as the basis for progress payments and will be incorporated into a form of Application for Payment acceptable to Engineer. Progress payments for Unit Price Work will be based on the number of units completed during the pay period, as determined under the provisions of Paragraph 13.03. Progress payments for cost-based Work will be based on Cost of the Work completed by Contractor during the pay period.

#### B. Applications for Payments

- At least 20 days before the date established in the Agreement for each progress payment (but not more often than once a month), Contractor shall submit to Engineer for review an Application for Payment filled out and signed by Contractor covering the Work completed as of the date of the Application and accompanied by such supporting documentation as is required by the Contract Documents.
- 2. If payment is requested on the basis of materials and equipment not incorporated in the Work but delivered and suitably stored at the Site or at another location agreed to in writing, the Application for Payment must also be accompanied by: (a) a bill of sale, invoice, copies of subcontract or purchase order payments, or other documentation

establishing full payment by Contractor for the materials and equipment; (b) at Owner's request, documentation warranting that Owner has received the materials and equipment free and clear of all Liens; and (c) evidence that the materials and equipment are covered by appropriate property insurance, a warehouse bond, or other arrangements to protect Owner's interest therein, all of which must be satisfactory to Owner.

- Beginning with the second Application for Payment, each Application must include an
  affidavit of Contractor stating that all previous progress payments received by Contractor
  have been applied to discharge Contractor's legitimate obligations associated with prior
  Applications for Payment.
- 4. The amount of retainage with respect to progress payments will be as stipulated in the Agreement.

# C. Review of Applications

- Engineer will, within 10 days after receipt of each Application for Payment, including each
  resubmittal, either indicate in writing a recommendation of payment and present the
  Application to Owner, or return the Application to Contractor indicating in writing
  Engineer's reasons for refusing to recommend payment. In the latter case, Contractor
  may make the necessary corrections and resubmit the Application.
- 2. Engineer's recommendation of any payment requested in an Application for Payment will constitute a representation by Engineer to Owner, based on Engineer's observations of the executed Work as an experienced and qualified design professional, and on Engineer's review of the Application for Payment and the accompanying data and schedules, that to the best of Engineer's knowledge, information and belief:
  - a. the Work has progressed to the point indicated;
  - the quality of the Work is generally in accordance with the Contract Documents (subject to an evaluation of the Work as a functioning whole prior to or upon Substantial Completion, the results of any subsequent tests called for in the Contract Documents, a final determination of quantities and classifications for Unit Price Work under Paragraph 13.03, and any other qualifications stated in the recommendation); and
  - c. the conditions precedent to Contractor's being entitled to such payment appear to have been fulfilled in so far as it is Engineer's responsibility to observe the Work.
- 3. By recommending any such payment Engineer will not thereby be deemed to have represented that:
  - a. inspections made to check the quality or the quantity of the Work as it has been performed have been exhaustive, extended to every aspect of the Work in progress, or involved detailed inspections of the Work beyond the responsibilities specifically assigned to Engineer in the Contract; or
  - b. there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold payment to Contractor.

- 4. Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment, will impose responsibility on Engineer:
  - a. to supervise, direct, or control the Work;
  - b. for the means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto;
  - c. for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work;
  - d. to make any examination to ascertain how or for what purposes Contractor has used the money paid by Owner; or
  - e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens.
- 5. Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated in Paragraph 15.01.C.2.
- 6. Engineer will recommend reductions in payment (set-offs) necessary in Engineer's opinion to protect Owner from loss because:
  - a. the Work is defective, requiring correction or replacement;
  - b. the Contract Price has been reduced by Change Orders;
  - c. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
  - d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible; or
  - e. Engineer has actual knowledge of the occurrence of any of the events that would constitute a default by Contractor and therefore justify termination for cause under the Contract Documents.

#### D. Payment Becomes Due

1. Ten days after presentation of the Application for Payment to Owner with Engineer's recommendation, the amount recommended (subject to any Owner set-offs) will become due, and when due will be paid by Owner to Contractor.

#### E. Reductions in Payment by Owner

- 1. In addition to any reductions in payment (set-offs) recommended by Engineer, Owner is entitled to impose a set-off against payment based on any of the following:
  - a. Claims have been made against Owner based on Contractor's conduct in the performance or furnishing of the Work, or Owner has incurred costs, losses, or damages resulting from Contractor's conduct in the performance or furnishing of the Work, including but not limited to claims, costs, losses, or damages from workplace injuries, adjacent property damage, non-compliance with Laws and Regulations, and patent infringement;

- b. Contractor has failed to take reasonable and customary measures to avoid damage, delay, disruption, and interference with other work at or adjacent to the Site;
- c. Contractor has failed to provide and maintain required bonds or insurance;
- d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible;
- e. Owner has incurred extra charges or engineering costs related to submittal reviews, evaluations of proposed substitutes, tests and inspections, or return visits to manufacturing or assembly facilities;
- f. The Work is defective, requiring correction or replacement;
- g. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
- h. The Contract Price has been reduced by Change Orders;
- i. An event has occurred that would constitute a default by Contractor and therefore justify a termination for cause;
- j. Liquidated or other damages have accrued as a result of Contractor's failure to achieve Milestones, Substantial Completion, or final completion of the Work;
- k. Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and discharge of such Liens; or
- I. Other items entitle Owner to a set-off against the amount recommended.
- 2. If Owner imposes any set-off against payment, whether based on its own knowledge or on the written recommendations of Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and the specific amount of the reduction, and promptly pay Contractor any amount remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, if Contractor remedies the reasons for such action. The reduction imposed will be binding on Contractor unless it duly submits a Change Proposal contesting the reduction.
- 3. Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld will be treated as an amount due as determined by Paragraph 15.01.D.1 and subject to interest as provided in the Agreement.

#### 15.02 Contractor's Warranty of Title

A. Contractor warrants and guarantees that title to all Work, materials, and equipment furnished under the Contract will pass to Owner free and clear of (1) all Liens and other title defects, and (2) all patent, licensing, copyright, or royalty obligations, no later than 7 days after the time of payment by Owner.

# 15.03 Substantial Completion

A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete and request that Engineer issue a certificate of Substantial Completion. Contractor shall at the same time

- submit to Owner and Engineer an initial draft of punch list items to be completed or corrected before final payment.
- B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.
- C. If Engineer considers the Work substantially complete, Engineer will deliver to Owner a preliminary certificate of Substantial Completion which will fix the date of Substantial Completion. Engineer shall attach to the certificate a punch list of items to be completed or corrected before final payment. Owner shall have 7 days after receipt of the preliminary certificate during which to make written objection to Engineer as to any provisions of the certificate or attached punch list. If, after considering the objections to the provisions of the preliminary certificate, Engineer concludes that the Work is not substantially complete, Engineer will, within 14 days after submission of the preliminary certificate to Owner, notify Contractor in writing that the Work is not substantially complete, stating the reasons therefor. If Owner does not object to the provisions of the certificate, or if despite consideration of Owner's objections Engineer concludes that the Work is substantially complete, then Engineer will, within said 14 days, execute and deliver to Owner and Contractor a final certificate of Substantial Completion (with a revised punch list of items to be completed or corrected) reflecting such changes from the preliminary certificate as Engineer believes justified after consideration of any objections from Owner.
- D. At the time of receipt of the preliminary certificate of Substantial Completion, Owner and Contractor will confer regarding Owner's use or occupancy of the Work following Substantial Completion, review the builder's risk insurance policy with respect to the end of the builder's risk coverage, and confirm the transition to coverage of the Work under a permanent property insurance policy held by Owner. Unless Owner and Contractor agree otherwise in writing, Owner shall bear responsibility for security, operation, protection of the Work, property insurance, maintenance, heat, and utilities upon Owner's use or occupancy of the Work.
- E. After Substantial Completion the Contractor shall promptly begin work on the punch list of items to be completed or corrected prior to final payment. In appropriate cases Contractor may submit monthly Applications for Payment for completed punch list items, following the progress payment procedures set forth above.
- F. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the punch list.

#### 15.04 Partial Use or Occupancy

A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without

significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:

- At any time, Owner may request in writing that Contractor permit Owner to use or occupy any such part of the Work that Owner believes to be substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 15.03.A through 15.03.E for that part of the Work.
- At any time, Contractor may notify Owner and Engineer in writing that Contractor considers any such part of the Work substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
- 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of Paragraph 15.03 will apply with respect to certification of Substantial Completion of that part of the Work and the division of responsibility in respect thereof and access thereto.
- 4. No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 6.04 regarding builder's risk or other property insurance.

#### 15.05 Final Inspection

A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work, or agreed portion thereof, is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.

#### 15.06 Final Payment

#### A. Application for Payment

- After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, annotated record documents (as provided in Paragraph 7.12), and other documents, Contractor may make application for final payment.
- 2. The final Application for Payment must be accompanied (except as previously delivered) by:
  - a. all documentation called for in the Contract Documents;
  - b. consent of the surety, if any, to final payment;
  - c. satisfactory evidence that all title issues have been resolved such that title to all Work, materials, and equipment has passed to Owner free and clear of any Liens or other title defects, or will so pass upon final payment.

- d. a list of all duly pending Change Proposals and Claims; and
- e. complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of the Work, and of Liens filed in connection with the Work.
- 3. In lieu of the releases or waivers of Liens specified in Paragraph 15.06.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (a) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (b) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien, or Owner at its option may issue joint checks payable to Contractor and specified Subcontractors and Suppliers.
- B. Engineer's Review of Final Application and Recommendation of Payment: If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's review of the final Application for Payment and accompanying documentation as required by the Contract Documents, Engineer is satisfied that the Work has been completed and Contractor's other obligations under the Contract have been fulfilled, Engineer will, within 10 days after receipt of the final Application for Payment, indicate in writing Engineer's recommendation of final payment and present the final Application for Payment to Owner for payment. Such recommendation will account for any set-offs against payment that are necessary in Engineer's opinion to protect Owner from loss for the reasons stated above with respect to progress payments. Otherwise, Engineer will return the Application for Payment to Contractor, indicating in writing the reasons for refusing to recommend final payment, in which case Contractor shall make the necessary corrections and resubmit the Application for Payment.
- C. Notice of Acceptability: In support of its recommendation of payment of the final Application for Payment, Engineer will also give written notice to Owner and Contractor that the Work is acceptable, subject to stated limitations in the notice and to the provisions of Paragraph 15.07.
- D. Completion of Work: The Work is complete (subject to surviving obligations) when it is ready for final payment as established by the Engineer's written recommendation of final payment and issuance of notice of the acceptability of the Work.
- E. Final Payment Becomes Due: Upon receipt from Engineer of the final Application for Payment and accompanying documentation, Owner shall set off against the amount recommended by Engineer for final payment any further sum to which Owner is entitled, including but not limited to set-offs for liquidated damages and set-offs allowed under the provisions of this Contract with respect to progress payments. Owner shall pay the resulting balance due to Contractor within 30 days of Owner's receipt of the final Application for Payment from Engineer.

#### 15.07 Waiver of Claims

A. By making final payment, Owner waives its claim or right to liquidated damages or other damages for late completion by Contractor, except as set forth in an outstanding Claim,

- appeal under the provisions of Article 17, set-off, or express reservation of rights by Owner. Owner reserves all other claims or rights after final payment.
- B. The acceptance of final payment by Contractor will constitute a waiver by Contractor of all claims and rights against Owner other than those pending matters that have been duly submitted as a Claim, or appealed under the provisions of Article 17.

#### 15.08 Correction Period

- A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the Supplementary Conditions or the terms of any applicable special guarantee required by the Contract Documents), Owner gives Contractor written notice that any Work has been found to be defective, or that Contractor's repair of any damages to the Site or adjacent areas has been found to be defective, then after receipt of such notice of defect Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:
  - 1. correct the defective repairs to the Site or such adjacent areas;
  - 2. correct such defective Work;
  - 3. remove the defective Work from the Project and replace it with Work that is not defective, if the defective Work has been rejected by Owner, and
  - 4. satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others, or to other land or areas resulting from the corrective measures.
- B. Owner shall give any such notice of defect within 60 days of the discovery that such Work or repairs is defective. If such notice is given within such 60 days but after the end of the correction period, the notice will be deemed a notice of defective Work under Paragraph 7.17.B.
- C. If, after receipt of a notice of defect within 60 days and within the correction period, Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. Contractor shall pay all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others). Contractor's failure to pay such costs, losses, and damages within 10 days of invoice from Owner will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the failure to pay.
- D. In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that item may start to run from an earlier date if so provided in the Specifications.
- E. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this paragraph, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.

F. Contractor's obligations under this paragraph are in addition to all other obligations and warranties. The provisions of this paragraph are not to be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.

#### ARTICLE 16—SUSPENSION OF WORK AND TERMINATION

#### 16.01 Owner May Suspend Work

A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by written notice to Contractor and Engineer. Such notice will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be entitled to an adjustment in the Contract Price or an extension of the Contract Times directly attributable to any such suspension. Any Change Proposal seeking such adjustments must be submitted no later than 30 days after the date fixed for resumption of Work.

#### 16.02 Owner May Terminate for Cause

- A. The occurrence of any one or more of the following events will constitute a default by Contractor and justify termination for cause:
  - Contractor's persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment, or failure to adhere to the Progress Schedule);
  - 2. Failure of Contractor to perform or otherwise to comply with a material term of the Contract Documents;
  - 3. Contractor's disregard of Laws or Regulations of any public body having jurisdiction; or
  - 4. Contractor's repeated disregard of the authority of Owner or Engineer.
- B. If one or more of the events identified in Paragraph 16.02.A occurs, then after giving Contractor (and any surety) 10 days' written notice that Owner is considering a declaration that Contractor is in default and termination of the Contract, Owner may proceed to:
  - 1. declare Contractor to be in default, and give Contractor (and any surety) written notice that the Contract is terminated; and
  - 2. enforce the rights available to Owner under any applicable performance bond.
- C. Subject to the terms and operation of any applicable performance bond, if Owner has terminated the Contract for cause, Owner may exclude Contractor from the Site, take possession of the Work, incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere, and complete the Work as Owner may deem expedient.
- D. Owner may not proceed with termination of the Contract under Paragraph 16.02.B if Contractor within 7 days of receipt of notice of intent to terminate begins to correct its failure to perform and proceeds diligently to cure such failure.
- E. If Owner proceeds as provided in Paragraph 16.02.B, Contractor shall not be entitled to receive any further payment until the Work is completed. If the unpaid balance of the Contract Price exceeds the cost to complete the Work, including all related claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects,

attorneys, and other professionals) sustained by Owner, such excess will be paid to Contractor. If the cost to complete the Work including such related claims, costs, losses, and damages exceeds such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when so approved by Engineer, incorporated in a Change Order. When exercising any rights or remedies under this paragraph, Owner shall not be required to obtain the lowest price for the Work performed.

- F. Where Contractor's services have been so terminated by Owner, the termination will not affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue, or any rights or remedies of Owner against Contractor or any surety under any payment bond or performance bond. Any retention or payment of money due Contractor by Owner will not release Contractor from liability.
- G. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 6.01.A, the provisions of that bond will govern over any inconsistent provisions of Paragraphs 16.02.B and 16.02.D.

#### 16.03 Owner May Terminate for Convenience

- A. Upon 7 days' written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such case, Contractor shall be paid for (without duplication of any items):
  - completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination, including fair and reasonable sums for overhead and profit on such Work;
  - expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses; and
  - 3. other reasonable expenses directly attributable to termination, including costs incurred to prepare a termination for convenience cost proposal.
- B. Contractor shall not be paid for any loss of anticipated profits or revenue, post-termination overhead costs, or other economic loss arising out of or resulting from such termination.

#### 16.04 Contractor May Stop Work or Terminate

- A. If, through no act or fault of Contractor, (1) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority, or (2) Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (3) Owner fails for 30 days to pay Contractor any sum finally determined to be due, then Contractor may, upon 7 days' written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate the contract and recover from Owner payment on the same terms as provided in Paragraph 16.03.
- B. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 30 days to pay Contractor any sum finally determined to be due, Contractor may, 7 days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The

provisions of this paragraph are not intended to preclude Contractor from submitting a Change Proposal for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this paragraph.

#### **ARTICLE 17—FINAL RESOLUTION OF DISPUTES**

#### 17.01 Methods and Procedures

- A. *Disputes Subject to Final Resolution*: The following disputed matters are subject to final resolution under the provisions of this article:
  - 1. A timely appeal of an approval in part and denial in part of a Claim, or of a denial in full, pursuant to Article 12; and
  - 2. Disputes between Owner and Contractor concerning the Work, or obligations under the Contract Documents, that arise after final payment has been made.
- B. *Final Resolution of Disputes*: For any dispute subject to resolution under this article, Owner or Contractor may:
  - 1. elect in writing to invoke the dispute resolution process provided for in the Supplementary Conditions;
  - agree with the other party to submit the dispute to another dispute resolution process;
  - 3. if no dispute resolution process is provided for in the Supplementary Conditions or mutually agreed to, give written notice to the other party of the intent to submit the dispute to a court of competent jurisdiction.

#### **ARTICLE 18—MISCELLANEOUS**

#### 18.01 Giving Notice

- A. Whenever any provision of the Contract requires the giving of written notice to Owner, Engineer, or Contractor, it will be deemed to have been validly given only if delivered:
  - 1. in person, by a commercial courier service or otherwise, to the recipient's place of business;
  - 2. by registered or certified mail, postage prepaid, to the recipient's place of business; or
  - 3. by e-mail to the recipient, with the words "Formal Notice" or similar in the e-mail's subject line.

#### 18.02 Computation of Times

A. When any period of time is referred to in the Contract by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.

#### 18.03 Cumulative Remedies

A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract. The provisions of this paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.

# 18.04 Limitation of Damages

A. With respect to any and all Change Proposals, Claims, disputes subject to final resolution, and other matters at issue, neither Owner nor Engineer, nor any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, shall be liable to Contractor for any claims, costs, losses, or damages sustained by Contractor on or in connection with any other project or anticipated project.

#### 18.05 No Waiver

A. A party's non-enforcement of any provision will not constitute a waiver of that provision, nor will it affect the enforceability of that provision or of the remainder of this Contract.

#### 18.06 Survival of Obligations

A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract, as well as all continuing obligations indicated in the Contract, will survive final payment, completion, and acceptance of the Work or termination of the Contract or of the services of Contractor.

# 18.07 Controlling Law

A. This Contract is to be governed by the law of the state in which the Project is located.

#### 18.08 Assignment of Contract

A. Unless expressly agreed to elsewhere in the Contract, no assignment by a party to this Contract of any rights under or interests in the Contract will be binding on the other party without the written consent of the party sought to be bound; and, specifically but without limitation, money that may become due and money that is due may not be assigned without such consent (except to the extent that the effect of this restriction may be limited by law), and unless specifically stated to the contrary in any written consent to an assignment, no assignment will release or discharge the assignor from any duty or responsibility under the Contract.

#### 18.09 Successors and Assigns

A. Owner and Contractor each binds itself, its successors, assigns, and legal representatives to the other party hereto, its successors, assigns, and legal representatives in respect to all covenants, agreements, and obligations contained in the Contract Documents.

#### 18.10 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

**SUPPLEMENTARY CONDITIONS** 

This page intentionally left blank

# SUPPLEMENTARY CONDITIONS OF THE CONSTRUCTION CONTRACT

# **Prepared By**









# **Endorsed By**





# Copyright<sup>©</sup> 2018

National Society of Professional Engineers 1420 King Street, Alexandria, VA 22314-2794 (703) 684-2882

www.nspe.org

American Council of Engineering Companies 1015 15th Street N.W., Washington, DC 20005 (202) 347-7474

www.acec.org

American Society of Civil Engineers 1801 Alexander Bell Drive, Reston, VA 20191-4400 (800) 548-2723

www.asce.org

The copyright for this EJCDC document is owned jointly by the three sponsoring organizations listed above. The National Society of Professional Engineers is the Copyright Administrator for the EJCDC documents; please direct all inquiries regarding EJCDC copyrights to NSPE.

NOTE: EJCDC publications may be purchased at <a href="www.ejcdc.org">www.ejcdc.org</a>, or from any of the sponsoring organizations above.

# SUPPLEMENTARY CONDITIONS OF THE CONSTRUCTION CONTRACT

# **TABLE OF CONTENTS**

|  | Page |
|--|------|
| Article 1— Definitions and Terminology   | 1    |
| Article 2— Preliminary Matters   | 1    |
| Article 3— Contract Documents: Intent, Requirements, Reuse                               | 1    |
| Article 4— Commencement and Progress of the Work   | 2    |
| Article 5— Site, Subsurface and Physical Conditions, Hazardous Environmental Conditions  | 3    |
| Article 6— Bonds and Insurance   | 5    |
| Article 7— Contractor's Responsibilities   | 8    |
| Article 8— Other Work at the Site  | 9    |
| Article 9— Owner's Responsibilities  | 9    |
| Article 10— Engineer's Status During Construction  | 10   |
| Article 11— Changes to the Contract  | 11   |
| Article 12— Claims   | 11   |
| Article 13— Cost of Work; Allowances, Unit Price Work                                    | 11   |
| Article 14— Tests and Inspections; Correction, Removal, or Accceptance of Defective Work | 12   |
| Article 15— Payments to Contractor, Set Offs; Completions; Correction Period             | 12   |
| Article 16— Suspension of Work and Termination   | 13   |
| Article 17— Final Resolutions of Disputes  | 14   |
| Article 18— Miscellaneous  | 14   |

| This page intentionally left blank |   |     |
|------------------------------------|---|-----|
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
|                                    |   |     |
| EJCDC® C-800,                      | Supplementary Conditions of the Construction Contract | ct. |

### SUPPLEMENTARY CONDITIONS OF THE CONSTRUCTION CONTRACT

These Supplementary Conditions amend or supplement EJCDC® C-700, Standard General Conditions of the Construction Contract (2018). The General Conditions remain in full force and effect except as amended.

The terms used in these Supplementary Conditions have the meanings stated in the General Conditions. Additional terms used in these Supplementary Conditions have the meanings stated below, which are applicable to both the singular and plural thereof.

The address system used in these Supplementary Conditions is the same as the address system used in the General Conditions, with the prefix "SC" added—for example, "Paragraph SC-4.05."

#### ARTICLE 1—DEFINITIONS AND TERMINOLOGY

- 1.01 Defined Terms
- SC-1.01 Delete Paragraph 3 in its entirety and insert the following in its place:
  - 3. Application for Payment—The document prepared by Contractor, using the form included in the project manual, to request progress or final payments, and which is to be accompanied by such supporting documentation as is required by the Contract Documents.

#### **ARTICLE 2—PRELIMINARY MATTERS**

- 2.02 Copies of Documents
- SC-2.02 Delete Paragraph 2.02.A in its entirety and insert the following in its place:
  - A. Owner shall furnish to Contractor one fully signed counterpart of the Agreement and one USB drive containing PDF files of conformed Contract Documents incorporating and integrating all Addenda and any amendments negotiated prior to the Effective Date of the Contract. Printed copies of the conformed Contract Documents will be furnished upon request at the cost of reproduction.

#### ARTICLE 3—CONTRACT DOCUMENTS: INTENT, REQUIREMENTS, REUSE

- 3.01 Intent
- SC-3.01 Add the following new paragraphs immediately after Paragraph 5.01.G:
  - H. The Drawings do not purport to show all the details of the Work. The Drawings, Specifications and Contract Documents are intended to describe a functionally complete project, any work that may reasonably be inferred as being required to produce the intended result is to be supplied by the Contractor even though not specifically called for in the Contract Documents (either the Drawings or Specifications).

#### ARTICLE 4—COMMENCEMENT AND PROGRESS OF THE WORK

- 4.01 Commencement of Contract Times; Notice to Proceed
- SC-4.01 Delete Paragraph 4.01.A in its entirety and insert the following in its place:
  - A. The Contract Times will commence to run on the day indicated in the Notice to Proceed. The Notice to Proceed will be given within 30 days after the Effective Date of the Contract unless the Owner and Contractor, upon mutual consent, agree to extend the deadline to issue the Notice to Proceed.
- 4.05 Delays in Contractor's Progress
- SC-4.05 Amend Paragraph 4.05.C by adding the following subparagraphs:
  - 2. Abnormal weather conditions;
    - a. The existence of abnormal weather conditions will be determined on a month-bymonth basis in accordance with the following:
      - Every workday on which total precipitation occurring between 7:00 p.m. on the preceding day (regardless of whether such preceding day is a workday) through 7:00 p.m. on the workday in question equals or exceeds <u>0.25</u>" of precipitation will be considered a "bad weather day", and
      - Every workday on which site conditions are still too wet as a result of rain on preceding days. Such "mud" days must be thoroughly documented by the Contractor and approved by the Resident Project Representative.
      - 3) Contractor shall anticipate the following number of foreseeable bad weather days per month:

| January 11  | May 5    | September 4 |
|-------------|----------|-------------|
| February 10 | June 6   | October 3   |
| March 8     | July 6   | November 5  |
| April 7     | August 5 | December 8  |

4) In each month, every bad weather day exceeding the number of foreseeable bad weather days established in the preceding list will be considered as "abnormal weather conditions." The existence of abnormal weather conditions will not relieve Contractor of the obligation to demonstrate and document that delays caused by abnormal weather are specific to the planned work activities or that such activities thus delayed were on Contractor's thencurrent Progress Schedule's critical path for the Project.

# ARTICLE 5—SITE, SUBSURFACE AND PHYSICAL CONDITIONS, HAZARDOUS ENVIRONMENTAL CONDITIONS

#### 5.01 Availability of Lands

- SC-5.01.A Add the following new paragraph immediately after Paragraph 5.01.A:
  - 1. In the event all lands, rights-of way, easements, or servitudes are not obtained as herein contemplated prior to the Notice to Proceed, the Contractor shall begin the Work upon such land and rights-of-way as Owner has previously acquired.
  - 2. Permanent improvements on private property are limited to those areas having an executed easement, servitude agreement, or Right of Entry prior to installation. Contractor is to comply with all terms, restrictions, special conditions, etc., as described in the easements or servitudes. Costs of such compliance are to be reflected in the related work. No separate compensation will be made for this compliance.
- SC-5.01 Add the following new paragraphs immediately after Paragraph 5.01.C:
  - D. The Contractor shall be responsible for coordinating his construction activities through the Owner and Engineer to ensure that all improvements are placed in the proper location. Permanent improvements on private property are limited to those areas having an executed easement agreement prior to installation. Contractor is to comply with all terms, restrictions, special conditions, etc., as described in the easements. Costs of such compliance is to be reflected in related work. No separate compensation will be made for this compliance.
- 5.03 Subsurface and Physical Conditions
- SC-5.03 Delete Paragraphs 5.03.A and 5.03.B in their entirety and insert the following:
  - 3. No reports of explorations or tests of subsurface conditions at or adjacent to the Site, or drawings of physical conditions relating to existing surface or subsurface structures at the Site, are known to Owner.
- 5.05 Underground Facilities
- SC-5.05 Delete subparagraph 5.05.A.3 in its entirety and insert the following in its place:
  - verifying the actual location of those Underground Facilities shown or not shown, indicated or not indicated, in the Contract Documents as being within the area affected by the Work, by exposing such Underground Facilities during the course of construction;
- SC-5.05 Delete Paragraph 5.05B in its entirety and insert the following in its place:
  - B. Notice by Contractor: If Contractor encounters what he believes to be an unforeseen conflict with an Underground Facility, then Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing regarding such Underground Facility. An Underground Facility may be considered an unforeseen conflict only if all of the following provisions apply:

- 1. Underground Facility was not shown or indicated on the Drawings.
- 2. Underground Facility was not shown or indicated on the Drawings with reasonable accuracy.
- Presence of Underground Facility was not visually apparent based on aboveground indicators such as signage, markers, piping, meters, pedestals, etc. as could reasonably expect to be seen during Contractor's pre-bid site visit.
- SC-5.05 Delete Paragraph 5.05C.1 in its entirety and insert the following in its place:
  - 4. Promptly review the Underground Facility and conclude whether such Underground Facility meets the criteria of an unforeseen conflict as described in SC5.05A.
- SC-5.05 Delete Paragraph 5.05F in its entirety and insert the following in its place:
  - Contractor shall be entitled to an equitable adjustment in the Contract Price or Contract
    Times, to the extent that any existing Underground Facility at the Site meets the criteria
    of an unforeseen conflict, and any related delay, disruption, or interference, causes an
    increase or decrease in Contractor's cost of, or time required for, performance of the
    Work; subject, however, to the following:
    - a. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03;
    - b. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E; and
    - c. Contractor gave the notice required in Paragraph 5.05.B.
  - If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
  - 3. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the Underground Facility in question.
- 5.06 Hazardous Environmental Conditions
- SC-5.06 Add the following new paragraphs immediately after Paragraph 5.06.A.3:
  - 4. The following table lists the reports known to Owner relating to Hazardous Environmental Conditions at or adjacent to the Site, and the Technical Data (if any) upon which Contractor may rely:

| Report Title | Date of Report | Technical Data |
|--------------|----------------|----------------|
| None         | N/A            | N/A            |
|              |                |                |
|              |                |                |

5. The following table lists the drawings known to Owner relating to Hazardous Environmental Conditions at or adjacent to the Site, and Technical Data (if any) contained in such Drawings upon which Contractor may rely:

| Drawings Title | Date of Drawings | Technical Data |
|----------------|------------------|----------------|
| None           | N/A              | N/A            |
|                |                  |                |
|                |                  |                |

#### ARTICLE 6—BONDS AND INSURANCE

- 6.01 Performance, Payment, and Other Bonds
- SC-6.01 Add the following new paragraphs immediately after Paragraph 6.01.H:
  - In exercising any of the provisions of the Contract Documents, neither the Owner nor the Engineer shall be responsible for notifying the surety. It shall be the Contractor's responsibility to notify the surety of changes or assignments and progress of the Work. Nothing in these documents shall in any way indicate or incur a responsibility of the Owner or Engineer to safeguard the surety's financial interest relative to the Work.
- 6.02 Insurance General Provisions
- SC-6.02 Add the following new paragraphs immediately after Paragraph 6.02.N:
  - O. The following must appear verbatim in the Insurance Certificate: "Blanket additional insureds and blanket waiver of subrogation as required by contract."
- 6.03 Contractor's Insurance
- SC-6.03 Supplement Paragraph 6.03 with the following provisions after Paragraph 6.03.C:
  - D. Workers' Compensation and Employer's Liability: Contractor shall purchase and maintain workers' compensation and employer's liability insurance, including, as applicable, United States Longshoreman and Harbor Workers' Compensation Act, and foreign voluntary workers' compensation (from available sources, notwithstanding the jurisdictional requirement of Paragraph 6.02.B of the General Conditions).

| Workers' Compensation and Related Policies | Policy limits of not less than: |
|--|---------------------------------|
| Workers' Compensation                      |                                 |
| State                                      | Statutory                       |
| Applicable Federal (e.g., Longshoreman's)  | Statutory                       |
| Employer's Liability                       |                                 |
| Each accident                              | \$1,000,000                     |
| Each employee                              | \$1,000,000                     |
| Policy limit                               | \$1,000,000                     |

E. Commercial General Liability—Claims Covered: Contractor shall purchase and maintain commercial general liability insurance, covering all operations by or on behalf of Contractor, on an occurrence basis, against claims for:

- 1. damages because of bodily injury, sickness or disease, or death of any person other than Contractor's employees,
- 2. damages insured by reasonably available personal injury liability coverage, and
- 3. damages because of injury to or destruction of tangible property wherever located, including loss of use resulting therefrom.
- F. Commercial General Liability—Form and Content: Contractor's commercial liability policy must be written on a 1996 (or later) Insurance Services Organization, Inc. (ISO) commercial general liability form (occurrence form) and include the following coverages and endorsements:
  - 1. Products and completed operations coverage.
    - a. Such insurance must be maintained for three years after final payment.
    - b. Contractor shall furnish Owner and each other additional insured (as identified in the Supplementary Conditions or elsewhere in the Contract) evidence of continuation of such insurance at final payment and three years thereafter.
  - 2. Blanket contractual liability coverage, including but not limited to coverage of Contractor's contractual indemnity obligations in Paragraph 7.18.
  - 3. Severability of interests and no insured-versus-insured or cross-liability exclusions.
  - 4. Underground, explosion, and collapse coverage.
  - 5. Personal injury coverage.
  - 6. Additional insured endorsements that include both ongoing operations and products and completed operations coverage through ISO Endorsements CG 20 10 10 01 and CG 20 37 10 01 (together). If Contractor demonstrates to Owner that the specified ISO endorsements are not commercially available, then Contractor may satisfy this requirement by providing equivalent endorsements.
  - 7. For design professional additional insureds, ISO Endorsement CG 20 32 07 04 "Additional Insured—Engineers, Architects or Surveyors Not Engaged by the Named Insured" or its equivalent.
- G. Commercial General Liability—Excluded Content: The commercial general liability insurance policy, including its coverages, endorsements, and incorporated provisions, must not include any of the following:
  - Any modification of the standard definition of "insured contract" (except to delete the railroad protective liability exclusion if Contractor is required to indemnify a railroad or others with respect to Work within 50 feet of railroad property).
  - 2. Any exclusion for water intrusion or water damage.
  - 3. Any provisions resulting in the erosion of insurance limits by defense costs other than those already incorporated in ISO form CG 00 01.
  - 4. Any exclusion of coverage relating to earth subsidence or movement.
  - 5. Any exclusion for the insured's vicarious liability, strict liability, or statutory liability (other than worker's compensation).

- 6. Any limitation or exclusion based on the nature of Contractor's work.
- 7. Any professional liability exclusion broader in effect than the most recent edition of ISO form CG 22 79.
- H. Commercial General Liability—Minimum Policy Limits

| Commercial General Liability                      | Policy limits of not |
|---|----------------------|
|   | less than:           |
| General Aggregate                                 | \$2,000,000          |
| Products—Completed Operations Aggregate           | \$1,000,000          |
| Personal and Advertising Injury                   | \$1,000,000          |
| Bodily Injury and Property Damage—Each Occurrence | \$1,000,000          |

I. Automobile Liability: Contractor shall purchase and maintain automobile liability insurance for damages because of bodily injury or death of any person or property damage arising out of the ownership, maintenance, or use of any motor vehicle. The automobile liability policy must be written on an occurrence basis.

| Automobile Liability                                      | Policy limits of not less than: |
|---|---------------------------------|
| Combined Single Limit                                     |                                 |
| Combined Single Limit (Bodily Injury and Property Damage) | \$1,000,000                     |

J. Umbrella or Excess Liability: Contractor shall purchase and maintain umbrella or excess liability insurance written over the underlying employer's liability, commercial general liability, and automobile liability insurance described in the Paragraphs above. The coverage afforded must be at least as broad as that of each and every one of the underlying policies.

| Excess or Umbrella Liability | Policy limits of not less than: |
|------------------------------|---------------------------------|
| Each Occurrence              | \$2,000,000                     |
| General Aggregate            | \$2,000,000                     |

- K. Using Umbrella or Excess Liability Insurance to Meet CGL and Other Policy Limit Requirements: Contractor may meet the policy limits specified for employer's liability, commercial general liability, and automobile liability through the primary policies alone, or through combinations of the primary insurance policy's policy limits and partial attribution of the policy limits of an umbrella or excess liability policy that is at least as broad in coverage as that of the underlying policy, as specified herein. If such umbrella or excess liability policy was required under this Contract, at a specified minimum policy limit, such umbrella or excess policy must retain a minimum limit of \$2,000,000 after accounting for partial attribution of its limits to underlying policies, as allowed above.
- L. Contractor's Pollution Liability Insurance: Contractor shall purchase and maintain a policy covering third-party injury and property damage, including cleanup costs, as a result of pollution conditions arising from Contractor's operations and completed operations. This insurance must be maintained for no less than three years after final completion.

| Contractor's Pollution Liability | Policy limits of not less than: |
|----------------------------------|---------------------------------|
| Each Occurrence/Claim            | Not Required                    |
| General Aggregate                | Not Required                    |

#### 6.04 Builder's Risk and Other Property Insurance

- SC-6.04 Delete Paragraph 6.04.A and insert the following in its place:
  - A. Builder's Risk: Owner will not be responsible for purchasing and maintaining property insurance upon the Work prior to Substantial Completion; as a result, it is recommended that Contractor purchase and maintain property insurance upon the Work at the Site in the amount of the full replacement cost thereof.

#### ARTICLE 7—CONTRACTOR'S RESPONSIBILITIES

- 7.03 Labor; Working Hours
- SC-7.03 Add the following new subparagraphs immediately after Paragraph 7.03.C:
  - 1. Regular working hours will be 6 AM 7 PM.
  - 2. Contractor's requests for working outside of regular hours and/or on a Saturday, Sunday, or legal holiday shall be made at least 48 hours in advance.
- 7.04 Services, Materials, and Equipment
- SC-7.04 Add a new paragraph immediately after Paragraph 7.04.C:
  - D. Suppliers of all materials or equipment furnished shall supply manufacturer's printed warranties on all materials or equipment furnished. Duration of warranties shall be as specified in the appropriate section of the Specifications. The Contractor shall furnish these warranties with the shop drawings and record documents. Limitations contained in the warranties from suppliers or manufacturers shall in no way limit or reduce the Contractor's obligations during the specified correction period.
- 7.07 Concerning Subcontractors and Suppliers
- SC-7.07 Add a new paragraph immediately after Paragraph 7.07.M:
  - N. The Contractor shall not award work valued at more than 50% of the Contract Price to a single Subcontractor without prior written approval of the Owner.
- 7.10 *Taxes*
- SC-7.10 Add a new paragraph immediately after Paragraph 7.10.A:
  - 3. Owner is exempt from payment of sales and compensating use taxes of the State of Louisiana and of cities and parishes thereof on all materials to be incorporated into the Work. In accordance with applicable rules adopted and promulgated by the Louisiana Department of Revenue, the Owner shall designate the Contractor and all subcontractors as its agents for the purchase and lease of materials, supplies or equipment for the project. The Contractor and all subcontractors shall accept the agency designation. The designation and acceptance thereof shall be made on the form prescribed by the Louisiana State Department of Revenue

which form shall be part of the contract between the Owner, and the Contractor. A copy of this form is included in the Project Manual. The agency relationship between the Owner and the Contractor and all subcontractors shall relieve the Contractor and subcontractors (1) from paying any state or local sales or state or local use taxes on materials, supplies or equipment which is affixed to and/or made a part of the real estate of the Project or Work or which is permanently incorporated into the Project or Work and, (2) from paying any state or local use taxes on any materials, supplies or equipment which is leased and used exclusively for the Project or Work. Accordingly, in preparing their bids and computing costs, the Contractor and subcontractors shall not consider sales and/or use taxes which would otherwise be due.

#### 7.11 Laws and Regulations

- SC-7.11 Add a new paragraph immediately after Paragraph 7.11.C:
  - D. Prior to commencing work for the City of Sulphur, Resident Louisiana Contractors shall complete and submit to the City a "Contractor Application". Non-Resident Contractors shall complete and submit the "Non-Resident Contractor Certification Checklist" and "Non-Resident Contractor Application". Both the Resident Louisiana Contractor and Non-Resident Contractor requirements and applications are as provided in Exhibit "7.11".

#### 7.18 *Indemnification*

- SC-7.18 Add a new paragraph immediately after Paragraph 7.18.B:
  - C. If through the acts of neglect on the part of Contractor, any other contractor or any Subcontractor shall suffer loss or damage on the Work, Contractor shall settle with such other contractor or Subcontractor by agreement or arbitration if such other contractor or Subcontractor will so settle. If such other contractor or Subcontractor shall assert any claim against Owner and/or Engineer, or the officers, directors, members, partners, employees, agents, consultants and Subcontractors of each on account of any damage alleged to have been sustained, Contractor shall indemnify and save harmless Owner, Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each against any such claims.

#### ARTICLE 8—OTHER WORK AT THE SITE

There are no Supplementary Conditions in this Article.

### **ARTICLE 9—OWNER'S RESPONSIBILITIES**

There are no Supplementary Conditions in this Article.

#### ARTICLE 10—ENGINEER'S STATUS DURING CONSTRUCTION

#### 10.03 Resident Project Representative

SC-10.03 Add the following new paragraphs immediately after Paragraph 10.03.B:

- C. The Resident Project Representative (RPR) will be Engineer's representative at the Site. RPR's dealings in matters pertaining to the Work in general will be with Engineer and Contractor. RPR's dealings with Subcontractors will only be through or with the full knowledge or approval of Contractor. The RPR will:
  - 1. Conferences and Meetings: Attend meetings with Contractor, such as preconstruction conferences, progress meetings, job conferences, and other Project-related meetings (but not including Contractor's safety meetings), and as appropriate prepare and circulate copies of minutes thereof.
  - Safety Compliance: Comply with Site safety programs, as they apply to RPR, and if
    required to do so by such safety programs, receive safety training specifically related to
    RPR's own personal safety while at the Site.

#### 3. Liaison

- a. Serve as Engineer's liaison with Contractor. Working principally through Contractor's authorized representative or designee, assist in providing information regarding the provisions and intent of the Contract Documents.
- b. Assist Engineer in serving as Owner's liaison with Contractor when Contractor's operations affect Owner's on-Site operations.
- c. Assist in obtaining from Owner additional details or information, when required for Contractor's proper execution of the Work.

#### 4. Review of Work; Defective Work

- a. Conduct on-Site observations of the Work to assist Engineer in determining, to the extent set forth in Paragraph 10.02, if the Work is in general proceeding in accordance with the Contract Documents.
- b. Observe whether any Work in place appears to be defective.
- c. Observe whether any Work in place should be uncovered for observation, or requires special testing, inspection or approval.

#### 5. Inspections and Tests

- a. Observe Contractor-arranged inspections required by Laws and Regulations, including but not limited to those performed by public or other agencies having jurisdiction over the Work.
- b. Accompany visiting inspectors representing public or other agencies having jurisdiction over the Work.
- 6. Payment Requests: Review Applications for Payment with Contractor.

#### 7. Completion

a. Participate in Engineer's visits regarding Substantial Completion.

- b. Assist in the preparation of a punch list of items to be completed or corrected.
- c. Participate in Engineer's visit to the Site in the company of Owner and Contractor regarding completion of the Work, and prepare a final punch list of items to be completed or corrected by Contractor.
- d. Observe whether items on the final punch list have been completed or corrected.

#### D. The RPR will not:

- 1. Authorize any deviation from the Contract Documents or substitution of materials or equipment (including "or-equal" items).
- 2. Exceed limitations of Engineer's authority as set forth in the Contract Documents.
- 3. Undertake any of the responsibilities of Contractor, Subcontractors, or Suppliers.
- 4. Advise on, issue directions relative to, or assume control over any aspect of the means, methods, techniques, sequences or procedures of construction.
- Advise on, issue directions regarding, or assume control over security or safety practices, precautions, and programs in connection with the activities or operations of Owner or Contractor.
- 6. Participate in specialized field or laboratory tests or inspections conducted off-site by others except as specifically authorized by Engineer.
- 7. Authorize Owner to occupy the Project in whole or in part.

#### ARTICLE 11—CHANGES TO THE CONTRACT

#### 11.09 Change Proposals

SC-11.09 Supplement Paragraph 11.09.B.2.b by adding the following sentence:

Proposal must include a detailed, itemized list of labor, material, and equipment costs for the Contractor's work including quantities and unit costs for each item of labor, material, and equipment. Similar itemized breakdowns shall be provided for all Subcontractor work.

#### **ARTICLE 12—CLAIMS**

No suggested Supplementary Conditions in this Article.

#### ARTICLE 13—COST OF WORK; ALLOWANCES, UNIT PRICE WORK

#### 13.01 *Cost of the Work*

SC-13.01 Supplement Paragraph 13.01.B.5.c.(2) by adding the following sentence:

The equipment rental rate book that governs the included costs for the rental of machinery and equipment owned by Contractor (or a related entity) under the Cost of the Work provisions of this Contract is the most current edition of the Rental Rate Blue Book for Construction Equipment.

- SC-13.01 Supplement Paragraph 13.01.C.2 by adding the following definition of small tools and hand tools:
  - a. For purposes of this paragraph, "small tools and hand tools" means any tool or equipment whose current price if it were purchased new at retail would be less than \$500.

#### 13.03 Unit Price Work

SC-13.03 Delete Paragraph 13.03.E in its entirety and insert the following in its place:

- E. Adjustments in Unit Price
  - 1. Contractor or Owner shall be entitled to an adjustment in the unit price with respect to an item of Unit Price Work if:
    - a. the extended price of a particular item of Unit Price Work amounts to 20 percent or more of the Contract Price (based on estimated quantities at the time of Contract formation) and the variation in the quantity of that particular item of Unit Price Work actually furnished or performed by Contractor differs by more than 30 percent from the estimated quantity of such item indicated in the Agreement; and
    - b. Contractor's unit costs to perform the item of Unit Price Work have changed materially and significantly as a result of the quantity change.
  - The adjustment in unit price will account for and be coordinated with any related changes in quantities of other items of Work, and in Contractor's costs to perform such other Work, such that the resulting overall change in Contract Price is equitable to Owner and Contractor.
  - 3. Adjusted unit prices will apply to all units of that item.
  - 4. Where unit prices are contained in the initial Contract, no deviations shall be allowed in computing negotiated Change Order costs.

## ARTICLE 14—TESTS AND INSPECTIONS; CORRECTION, REMOVAL, OR ACCCEPTANCE OF DEFECTIVE WORK

- 14.02 Tests, Inspections, and Approvals
- SC-14.02 Add new paragraphs immediately after Paragraph 14.02.F:
  - G. Contractor shall provide at no cost to the Owner any required specimens for testing purposes.
  - H. Contractor is responsible for all costs for follow-up testing required as a result of failed tests.

#### ARTICLE 15—PAYMENTS TO CONTRACTOR, SET OFFS; COMPLETIONS; CORRECTION PERIOD

- 15.01 *Progress Payments*
- SC-15.01 Add the following new subparagraphs following subparagraph15.01.B.4:
  - 5. Applications for Payment must be executed by an authorized representative of the Contractor. An applicable document shall accompany the first Application for Payment

Page 12 of 14

- that provides evidence of authorization to sign such as a corporate resolution for a corporation.
- 6. To ensure that processing milestones are complied with, the following schedule applies:
  - a. Work included for payment shall consist of work performed prior to the monthly cutoff date. The cutoff date shall be established at the preconstruction conference in a manner to be consistent with the Owner's meeting schedule.
  - b. Contractor shall review monthly quantities and work performed with the Resident Project Representative and confirm agreement on said quantities and work performed prior to submitting the Application for Payment.
  - Contractor shall submit the Application for Payment to the Engineer not later than 5 days following the cutoff date.

#### 15.03 Substantial Completion

- SC-15.03 Add the following new subparagraphs to Paragraph 15.03.C:
  - 1. Any punch list generated during the Project will include the cost estimates for each item based on the mobilization, labor, material, and equipment costs of correcting each punch list item.
  - 2. Contractor is responsible for recording the certificate of Substantial Completion at the Clerk of Court in the parish where the Work is located.

#### 15.06 Final Payment

- SC-15.06 Add the following new subparagraph following subparagraph15.06.A.2.e:
  - f. A certified clear lien certificate on the Contract obtained from the Clerk of Court at least 45 days after the recordation date of the certificate of Substantial Completion.

#### **ARTICLE 16—SUSPENSION OF WORK AND TERMINATION**

- 16.04 Contractor May Stop Work or Terminate
- SC-16.04 Delete Paragraph 16.04B of the General Conditions and substitute the following in its place:
  - 3. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 45 days to pay Contractor any sum finally determined to be due, Contractor may, 7 days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The provisions of this paragraph are not intended to preclude Contractor from submitting a Change Proposal for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this paragraph.

#### ARTICLE 17—FINAL RESOLUTIONS OF DISPUTES

17.02 Attorneys' Fees

SC-17.02 Add the following new paragraph immediately after Paragraph 17.01.

A. For any matter subject to final resolution under this Article, the prevailing party shall be entitled to an award of its attorneys' fees incurred in the final resolution proceedings, in an equitable amount to be determined in the discretion of the court, arbitrator, arbitration panel, or other arbiter of the matter subject to final resolution, taking into account the parties' initial demand or defense positions in comparison with the final result.

#### **ARTICLE 18—MISCELLANEOUS**

18.01 Giving Notice

SC-18.01 Delete Paragraph 18.01.A and insert the following in its place:

- A. Whenever any provision of the Contract requires the giving of written notice to Owner, Engineer, or Contractor, it will be deemed to have been validly given only if delivered by one of the following means and copied and delivered in a similar manner to the Engineer:
  - 1. in person, by a commercial courier service or otherwise, to the recipient's place of business;
  - 2. by registered or certified mail, postage prepaid, to the recipient's place of business; or
  - by e-mail to the recipient, with the words "Formal Notice" or similar in the e-mail's subject line.

**TECHNICAL PROVISIONS** 

This page intentionally left blank

## **Section H-100 – General Requirements**

### Beauregard Parish Police Jury Parish Annex Building Renovations (Phase I)

## M.A. Project No. H1-22029-DA

## Table of Contents

| H-100-1    | <b>Project Identification</b>                       |
|------------|---|
| H-100-1.1  | Service of Notice                                   |
| H-100-1.2  | Scope of Work                                       |
| H-100-1.3  | Drawings and Specifications Intent                  |
| H-100-2    | Contract Considerations                             |
| H-100-2.1  | Reference Specifications                            |
| H-100-2.2  | Liquidated Damages                                  |
| H-100-2.3  | Local Conditions                                    |
| H-100-2.4  | Materials Storage                                   |
| H-100-2.5  | Procedure for Record Drawing                        |
| H-100-2.6  | Procedures for Shop Drawings                        |
| H-100-2.7  | Clarification of One Year Correction Period         |
| H-100-2.8  | Clarification of Warranties on Mechanical Equipment |
| H-100-2.9  | Contractor's Understanding                          |
| H-100-2.10 | Assignment and Contract Changes                     |
| H-100-2.11 | Sureties Provision                                  |
| H-100-2.12 | Substantial Completion                              |
| H-100-3    | Preliminary Matters                                 |
| H-100-3.1  | Preconstruction Conference                          |
| H-100-3.2  | Construction Schedule                               |
| H-100-3.3  | Project Field Office                                |
| H-100-3.4  | Project Sign  |
| H-100-4    | General Construction                                |
| H-100-4.1  | Maintenance of Traffic                              |
| H-100-4.2  | Existing Driveways and Driveway Culverts            |
| H-100-4.3  | Existing Fences and Fence Structures                |
| H-100-4.4  | Protection of Existing Eater Facilities             |
| H-100-4.5  | Temporary Utilities                                 |
| H-100-4.6  | Protection of the Site                              |
| H-100-4.7  | Existing Utilities                                  |
| H-100-4.8  | Photographs   |
| H-100-4.9  | Water Supply  |
| H-100-4.10 | Preconstruction Documentation                       |

| H-100-4.11 | Standard of Manufacturer               |
|------------|--|
| H-100-4.12 | Safety Fencing                         |
| H-100-4.13 | Salvaged Material                      |
| H-100-4.14 | Minimize Negative Impact on the Public |
| H-100-4.15 | School/Business Direction Signs        |
| H-100-4.16 | Final Cleanup                          |

### TECHNICAL PROVISIONS SECTION H-100 GENERAL REQUIREMENTS

#### H-100-1 PROJECT IDENTIFICATION

#### H-100-1.1 SERVICE OF NOTICE:

All notices required to be given shall be mailed or delivered to the Owner, care of the identified contact person and copied and mailed or delivered to the Engineer.

#### H-100-1.2 SCOPE OF WORK

Contract work includes furnishing all materials, labor, equipment, supervision, tools and other necessary items for the complete and satisfactory <u>site civil</u> renovation improvements to the Parish Annex Building as detailed on the plans.

#### H-100-1.3 DRAWINGS AND SPECIFICATIONS INTENT

The Drawings are described previously in the Supplementary Conditions and do not purport to show all the details of the Work. The Drawings, Specifications and Contract Documents are intended to describe a functionally complete project, any work that may reasonably be inferred as being required to produce the intended result is to be supplied by the Contractor even though not specifically called for in the Contract Documents (either the Drawings or Specifications). The Engineer shall issue clarifications and interpretations, which may include supplemental Drawings and Specifications, in accordance with the provisions of General Conditions Article 3.1, 3.2 and 9.4 and other applicable provisions.

#### H-100-2 CONTRACT CONSIDERATIONS

#### H-100-2.1 REFERENCE SPECIFICATIONS

The Louisiana Department of Transportation & Development Specifications for Roads and Bridges, 2016 Edition Part Nos. 2 thru 10 are hereby designated to be the reference specifications and are made a part of these contract documents by reference.

The Technical Specifications contained herein only highlight and supplement certain areas of the reference Specifications; therefore, it shall be the Contractor's responsibility to thoroughly familiarize himself with the referenced specifications. In case of conflict the Technical Specifications contained herein will govern.

#### H-100-2.2 LIQUIDATED DAMAGES

The Owner shall have the right to deduct the liquidated damages from any monies in its hands, otherwise due, or to become due, to the Contractor, or to sue for and recover compensation for damages for nonperformance of this contract within the time stipulated.

Liquidated damages are set at \$500.00 per calendar day.

#### H-100-2.3 LOCAL CONDITIONS

The Contractor shall satisfy himself regarding all local conditions affecting his work by personal investigations, and neither the information contained in this Section nor that derived from maps or plans or from the Owner or his agents or employees, shall act to relieve the Contractor from any responsibility hereunder or from fulfilling any and all of the terms and requirements of this Contract.

#### H-100-2.4 MATERIALS STORAGE

All material shall be stored in strict accordance with the manufacturer instructions.

The Contractor shall provide a <u>fenced storage site</u> capable of storage of all material ordered if the Contractor is requesting payment for materials stockpiled.

#### H-100-2.5 PROCEDURE FOR RECORD DRAWING

As a supplement to General Conditions Section 6.19 requirement of the Contractor to maintain record drawings, the Contractor shall maintain and provide the Engineer with record documents as specified herein.

#### 1. Maintenance of Documents

- A. Maintain in Contractor's field office in clean, dry legible condition the following:
  - (1) Contract Drawings
  - (2) Specifications
  - (3) Addenda
  - (4) Approved Shop Drawings
  - (5) Change Orders
  - (6) Other modifications of Contract, Test Records, Survey Data, Field Orders, and
  - (7) All other documents pertinent to Contractor's work.
- B. Provide files and racks for proper storage and easy access.

- C. Make documents available at all times for inspection by the Engineer and the Owner.
- D. Record documents shall not be used for any other purpose and shall not be removed from the office without the Engineer's approval.

#### 2. Marking System

A. Use colored pencils for marking changes, revisions, additions, deletions, etc., to the record set of Contract Drawings.

#### 3. Recording

- A. Label each document "PROJECT RECORD" in two inch (2O) high printed letters.
- B. Keep record documents current.
- C. Do not permanently conceal any work until required information has been recorded.
- D. Contract Drawing Legibly mark to record actual construction including:
  - 1. Depths of various elements in relation to datum.
  - 2. Horizontal and vertical location of underground utilities and appurtenances referenced to permanent surface improvements.
  - 3. Location of internal appurtenances concealed in construction referenced to visible and accessible features of the work.
  - 4. Field changes of dimension and detail.
  - 5. Changes made by Change Order or Field Order.
  - 6. Details not on original Contract Drawings.
- E. Specifications and Addenda Legibly mark up each Section to record:
  - 1. Manufacturer, trade name, catalog number and supplier of each product and item of equipment actually installed.
  - 2. Changes made by Change Order or Field Order.
  - 3. Other matters not originally specified.

F. Shop Drawings - Maintain as record documents and legibly annotate drawings to record changes made after review.

#### 4. Submittal

- A. At completion of project, deliver record documents to the Engineer.
- B. Accompany submittal with transmittal letter containing:
  - 1. Date:
  - 2. Project title and number;
  - 3. Contractor's name and address;
  - 4. Title and number of each record document;
  - 5. Certification that each document as submitted is complete and accurate; and
  - 6. Signature of Contractor, or his authorized representative.

#### 5. Part of Other Work

Record drawings are considered a part of the related work and as such payment for work may be denied or reduced appropriately should the record drawings not be current during any periodic pay estimate period.

#### H-100-2.5.1 Record Drawings for Water Distribution Systems

In addition to other record drawings and document requirements as shown above in H-100-2.5, the Contractor shall prepare, maintain and provide the Engineer with record drawings of any water distribution facilities constructed by him. The purpose of these record drawings is to create a permanent record of the water system that will enable the future location of water mains for repair, extensions, taps, etc. The record drawings shall be utilized to identify the size and type of the mains and the area served, to locate the valves, hydrants, meters, etc., and to identify the locations of water plants, tanks and booster stations. To accomplish this, all of these items must be referenced by measurements and direction from a permanent, easily identifiable structure such as the center of roads, fences, utility poles, buildings, and the like.

The record drawing information obtained by the Contractor shall be marked on the record set of the Contract Drawings. Recording the information shall be neat, orderly and legible. Records shall be kept current. Measurements shall be made as the work proceeds and recorded before the work is covered.

All marks on the record set showing the record information shall be made in pencil or red pencil color. Any deletions or change to the original set should be indicated by tracing over the change with a light green pencil color, not by

scratching out. Tracing the changes in green will help keep the plans neat and will identify what is to be deleted from the plans.

The use of circle templates and a straight edge in drawing the information will also keep the record set neat.

If roads and/or road names not shown on the plans are located in the field, they should be added to the record set.

To clarify locations and dimensions, details of intersections, etc., may be needed as a supplement to the record set. These details should be drawn on the same sheet next to the intersection or off to the side of bottom of the sheet. If room is not available to draw the details on the sheet, they may be drawn on a separate page of paper and kept in a binder. All detail pages must be referenced identifying on which plan sheet the intersection appears. Also, intersections on the plan sheet must refer to the page on which the detail appears. Street names shall be noted on all details and a north arrow included if north is not up the page.

Measurements must be made with accurate methods. Long distances, over a few hundred feet, can be measured with a measuring wheel but shorter distances should be measured with a tape or chain. Do not guess at any measurements. Keep all ties (measurements) perpendicular to the main. Use triangular measurements if ties cannot be perpendicular. Tie the water mains and appurtenances to a permanent structure and identify the structure. Record all measurements while they are being made. Do not rely on memory.

Use stationing, such as "6+35" (this means six hundred feet plus 35 feet or 635 ft.) for recording the long measurements along water lines. Start the zero station (0+00) at an intersection of a water line, or other likely beginning point. Identify on the plans where stationing begins and which direction the stations continue. Use as many beginning points as required but equate the zero station with any other station referencing the same point (example 17+42 W. = 0+00 S.).

All water mains installed shall be shown on the record set and the size and type of pipe identified. The actual location of the main shall be shown; that is, which side of the road and if the main was installed on public or private right-of-way. A tie (measurement) to the center of the roadway or other structure should be made at intervals frequent enough to allow future location of the main. If the main stays parallel to the road, at a constant distance, few ties will be required, but if the main meanders, ties every 100 feet or closer may be needed. The station of each tie must be recorded along with the measurement. Stations and ties shall also be made for any change in direction of the main or locations of tees, reducers, etc.

All valves will be located by measurements from the center of roads and/or other structures. The side of the road that the valve is on shall be recorded also.

Any meter installed will also require recording. The stationing used to measure the water mains shall be used to tie down the locations of the meters. The side of the main on which the meter is installed shall be noted. Also note the size of the meter, if not standard, the distance from the meter to the main and the size and type of the service pipe.

The meter number, location and name and address of the customer shall be recorded on a separate list and given to the Owner upon completion of the project.

All fire hydrants installed shall be recorded. Note whether or not auxiliary valves were installed. Hydrants may be located by stations along mains and/or by measurements from roads, etc. The distance and direction from the hydrant to the main shall be recorded also.

Canal crossings, if a special method of crossing is used, shall be shown. The pipe material and method, along with length, and beginning and ending stations shall be recorded. A measurement from a bridge or headwall to the beginning or end of the crossing should be included. The depth below the streambed shall be given or if the main is above the creek, on a bridge or suspended, an elevation sketch should be drawn to show details.

At the end of mains, show if a gate valve, plug, flush valve or other item was installed. The end shall be stationed from an intersection and tied to the center of the road or other structure.

Casings shall be stationed at beginning and end and dimensioned from the structure that they are crossing. If vents are installed, they should be noted also.

All other miscellaneous appurtenances and water main related work shall be identified and tied-down to a structure and/or stationed.

Once the project is completed, all record drawing information obtained during construction of the project must be turned in to the Engineer as set forth elsewhere in these specifications. These records must be complete and in an orderly format.

The following pages include examples of how record drawings should appear. A legend of symbols and a list of abbreviations to be used on the record drawings also follows:

#### ABBREVIATIONS TO BE USED ON RECORD DRAWINGS

| ITEM              | SYMBOL    | ITEM                 | SYMBOL   |
|-------------------|-----------|----------------------|----------|
| Aggregate -       | Aggr.     | Headwall -           | Hdwl.    |
| Angle, Angles -   |           | Inch -               | In.      |
| Approximate -     | Approx.   | Junction -           | Jct.     |
| Asbestos Cement - | A.C.      | Left -               | Lt.      |
| Asphalt -         | Asp.      | Linear Foot -        | Lin. Ft. |
| Auxiliary -       | Aux.      | Number -             | No. or # |
| Avenue -          | Ave.      | Manhole -            | M.H.     |
| Barbed Wire -     | B.W.      | Meter -              | M.       |
| Begin -           | Beg.      | North -              | N.       |
| Bridge -          | Br.       | Polyethylene -       | P.E.     |
| Building -        | Bldg.     | Pavement -           | Pvmt.    |
| Casing -          | Csg.      | Polyvinyl Chloride - | P.V.C.   |
| Cast Iron -       | C.I.      | Power Pole -         | P.P.     |
| Cattle Guard -    | C.G.      | Post Hydrant -       | P.H.     |
| Class -           | Cl.       | Railroad -           | R.R.     |
| Centerline -      | c or C.L. | Reducer -            | Red.     |
| Chain Link -      | C.L.      | Residence -          | Res.     |
| Corner -          | Cor.      | Right -              | Rt.      |
| Culvert -         | Cul.      | Right-of-Way -       | R/W      |
| Crossing -        | X-ing     | Road -               | Rd.      |
| Creek Crossing -  | Cr. X-ing | Roadway -            | Rdwy.    |
| East -            | Е.        | Schedule -           | Sch.     |
| Ductile Iron -    | D.I.      | South -              | S.       |
| Existing -        | Exist.    | Street -             | St.      |
| Detail -          | Det'l     | Surface -            | Surf.    |
| Feet -            | Ft.       | Steel -              | Stl.     |
| Fence -           | Fc.       | Telephone Pole -     | T.P.     |
| Fire Hydrant -    | F.H.      | Typical -            | Typ.     |
| Flush Valve -     | F.V.      | Valve -              | Va.      |
| Galvanized Iron - | G.I.      | Water Meter -        | W.M.     |
| Gate Valve -      | G.V.      | West -               | W.       |
| Guy Pole -        | G.P.      | With -               | w/       |
| Highway -         | Hwy.      | Without -            | w/o      |

#### H-100-2.6 PROCEDURES FOR SHOP DRAWINGS

As a supplement to the General Conditions, the following also applies to the shop drawings and their submittal procedure.

To meet the qualifications necessary to ensure approval of a submittal as a Shop Drawing, the submittal must:

- A. Include drawings of the item in a sufficiently large scale so as to provide clarity to allow adequate evaluation of the complete product as it is proposed to be furnished at the job site.
- B. Include specifications and descriptions of the item as it is proposed to be furnished at the job site.
- C. Include drawings of the item showing the physical dimensions of the item, the clearances necessary to install the item.
- D. Include drawings and specifications showing materials of construction, their strengths, the surface preparation, and paint used upon them. The specific grade of surface preparation shall be specified if it differs from the requirements called for in the Technical Provisions.
- E. Include copies of sample warranties on each item of equipment.
- F. Include specifications describing operation and maintenance requirements for the item.
- G. Include drawings and specifications of special equipment needed for proper installation of the item and whether or not these items are to be furnished by the Supplier/Vendor. A complete set of field assembly and installation instructions for the installation contractor shall be provided as a part of the Certified Shop Drawings.

The Contractor shall submit six (6) copies each of the shop drawings submittal on respective materials, certified and stamped. Uncertified or unstamped submittals may be treated as being incomplete. All submittals must be accompanied with written letter or transmittal signed by an authorized representative of the contractor. Submittal from other parties will not be accepted. Incomplete submittals will be returned as incomplete with no action taken.

#### H-100-2.7 CLARIFICATION OF ONE YEAR CORRECTION PERIOD

Provisions of General Conditions are not intended to waive the Contractor's obligation to perform the Work as Specified after the elapsing of the twelve month period. Work which is required by these documents as amended (if applicable) but is found:

- 1) not to have been performed
- 2) not to have been performed per Specifications

must be corrected as provided in the General Conditions irrespective of the twelve month period as provided by applicable law.

#### H-100-2.8 CLARIFICATION OF WARRANTIES ON MECHANICAL EQUIPMENT

All warranties are between the Contractor and Owner. Additional manufacturer's warranties may be specifically required herein and definitely should be required by the Contractor. The initiation of the Contractor's warranty is as defined in the General Conditions Section 29 (as modified if applicable) irrespective of any third party's limitation to the Contractor.

#### H-100-2.9 CONTRACTOR'S UNDERSTANDING

By submitting a proposal the Contractor acknowledges that he has carefully examined all documents pertaining to the work, the location, accessibility and general character of the site of the work and all existing structures within and adjacent to the site, and has satisfied himself as to the nature of the work, the condition of existing structures, the conformation of the ground, the character, quality and quantity of the material to be encountered, the character of the equipment, machinery, plant, any other facilities needed preliminary to and during prosecution of the work, the general and local conditions, the construction hazards, and all other matters which can in any way affect the work under the Contract. It is further mutually agreed that by submitting a proposal the Contractor acknowledges that he has satisfied himself as the correctness of the plans, drawings, specifications and other contract documents for the construction of the work and that he accepts all the terms and stipulations contained therein; and that he is prepared to work in peace and harmony with other contractors performing work on the site.

No verbal agreement or conversation with any officer, agent or employee of the Owner or of the Engineer, or with the Owner himself, either before or after the execution of the Contract, shall affect or modify any of the terms, conditions, or other obligations set forth in any of the contract documents.

#### H-100-2.9.1 TIME CONSTRAINTS

By submitting a proposal the contractor is certifying that he understands the required time constraints on the construction period of this contract and has incorporated the level of effort required to complete all specified work within the required time frame.

The contract time can be adjusted as provided in the Contract Documents. In computing extensions due to abnormal weather conditions, the following chart will be used to determine normal anticipated days loss due to rain:

January 10 May 5 September 4 February 9 June 6 October 3

H-100 - 11

March 8 July 6 November 7 April 7 August 5 December 7

Days in excess of the above on a cumulative basis shall be considered "abnormal" per the General Conditions.

#### H-100-2.10 ASSIGNMENT AND CONTRACT CHANGES

In exercising any contract provisions for assignments and other contract changes, the Owner shall not be require to notify the surety or sureties on the Contractor's bond or bonds. it shall be the Contractor's responsibility to advise the surety or sureties of any changes or assignments, and the progress of his work.

#### H-100-2.11 SURETIES PROVISIONS

Sureties shall meet all specified qualifications in the Contract Documents. Nothing in these documents shall in any way indicate or incur a responsibility of the Owner or Engineer to safeguard the surety's financial interest relative to this work.

#### H-100-2.12 SUBSTANTIAL COMPLETION

The procedure for securing the status of "substantially complete" shall be as defined in the General Conditions as modified in this section.

#### H-100-2.12.1 EQUIPMENT START UP

All equipment shall be given an initial start up under the manufacturer's representative as specified. This start-up by itself does not constitute the commencement of any warranty or operational period.

#### H-100-2.12.2 OPERATIONAL PERIOD

The Contractor, as part of the work, shall start up the facility. Requirements relating to specific equipment appear elsewhere in these specifications.

Generally, the Contractor must have all equipment and controls, capable of full operation prior to performing a facility start up. Partial start ups pertaining to isolated equipment shall not constitute a facility start-up.

An Operation Inspection of the project will be required before the facility start up. Therefore, when the work is sufficiently complete, the Contractor shall notify the Engineer in writing and request an operational inspection. The notification shall include a list of items remaining to be completed or corrected.

Within ten (10) working days after the notification, the Engineer will inspect the work and verify the list of items remaining to be completed or corrected.

If the Engineer does not concur that the work is sufficiently complete, he will notify the Contractor in writing and state his reasons. Upon Engineer's concurrence that the work is sufficiently complete, the Contractor shall start up the facility.

After successfully starting the station up and demonstrating it to the Owner and Engineer, the Contractor shall operate and maintain the facility for a period of ten (10) consecutive days. These days shall run without interruption due to failures associated with the new equipment. If failures occur, they shall be promptly remedied and the facility put back on line. The ten (10) days period will then begin anew.

Upon completion of the ten (10) day operational period, the Contractor may turn the daily maintenance of the facility over to the Owner, under the Contractor's supervision. This will not begin any warranty under this contract. At this time, with Owner's concurrence, the Contractor may demolish/abandon, as specified elsewhere, any existing facility to be abandoned which has been replaced by the newly started facility.

At the time that the operation of the station is turned over to the Owner, the Owner will begin paying for electrical power and lubricants and other daily maintenance items.

#### H-100-2.12.3 RECORD DRAWINGS

As required elsewhere, records shall be current at the time of the Contractor's request.

#### H-100-2.12.4 WORK STATUS

Prior to requesting a substantial completion, all pay item work shall be completed. Minor corrective work shall be acceptable to remain undone.

#### H-100-2.12.5 INSPECTION FOR SUBSTANTIAL COMPLETION

As a supplement to the General Conditions, an inspection for substantial completion is required.

After completion or correction of all items noted during the operational inspection (including subsequent deficiencies discovered) and completion of stated prerequisites, the Contractor shall notify the Engineer in writing that the work is substantially complete and request an inspection for issuance of substantial completion certificate.

The Contractor shall request the inspection in writing to the Engineer.

The Contractor shall submit the following items to the Engineer when requesting the inspection: (1) manufacturer's inspection reports when required in the Technical Specifications; (2) certificate of inspection indicating compliance with the laws, ordinances, codes and regulations of Federal, State and local governmental agencies; and (3) a certificate that the Contractor has carefully inspected the work and found it to be entirely complete and in full compliance with the Contract Documents.

Within ten (10) working days after the substantial completion notification, the Engineer will inspect the work for verification.

Upon substantial completion of the work, the Engineer will issue the Contractor a "Certificate of Substantial Completion". This certificate shall be executed by the Owner, the Engineer, and the Contractor. This certificate may enumerate any conditions to be met by the Contractor for final acceptance of the work by the Engineer.

Once executed by all parties, the Contractor shall have the document recorded by the Clerk of Court in the parish where the work is located. Failure to have the certificate recorded promptly may delay payment of the retainage. After 45 days from the record date of the Certificate of Final Acceptance, the Contractor shall obtain from the Clerk of Court a certified lien certificate on his contract. If the contract is found to be free of liens, the final payment of retainage will be processed upon receipt of the lien free certificate and a request for payment of the retainage from the Contractor, subject to the provisions of Article 14 in the General Conditions.

#### H-100-3 PRELIMINARY MATTERS

#### H-100-3.1 PRECONSTRUCTION CONFERENCE

Prior to beginning construction, a pre-construction conference will be held between the Contractor(s), the Owner, and the Consulting Engineers to reach agreements relating to responsibilities and procedures of each interested party to see that the project is installed according to the approved plans and specifications and the conditions under which disbursements for construction costs are authorized and will be paid. This meeting will be prearranged by the Owner.

#### H-100-3.2 CONSTRUCTION SCHEDULE

Prior to the Preconstruction Conference, the Contractor shall prepare and submit to the Engineer a proposed progress schedule in the form of a bar chart showing:

- 1. Appropriate items of work.
- 2. The percent of total work for each item of work.
- 3. Time of beginning and time of completion of each item by calendar period.
- 4. Other data pertinent to each item (such as total quantity of each, etc.)
- 5. The bar chart shall be arranged so that actual progress of the work can be charted beside proposed schedule of work for each item.

The form and arrangement of the progress schedule shall be as approved by the Engineer. The breakdown of the work into various items shall be as approved by the Engineer.

The bar chart progress schedule shall be updated monthly by the contractor to show actual work progress and submitted to the Engineer in quadruplicate, monthly.

When monthly requests for payment are made, the progress schedule shall be attached to the payment request. No requests for payment will be approved unless the monthly submittals are completed.

#### H-100-3.3 PROJECT FIELD OFFICE

A project field office for use by others is not required.

#### H-100-3.4 PROJECT SIGN

No Project Sign is required.

#### H-100-4 GENERAL CONSTRUCTION

#### H-100-4.1 MAINTENANCE OF TRAFFIC

The Contractor shall at all times maintain all streets, trenches, driveways, walkways, and ditches in such a condition as to minimize inconvenience to the public and to the property owner.

It shall be the sole responsibility of the Contractor to provide adequate protection for his workers and adequate warning for motorists, such as signs, warning lights and barricades. The cost of these warning devices will be measured for payment when listed separately in the Bid Proposal under Pay Item 713(01), Temporary Signs and Barricades, Lump Sum. If there is no separate lighting in the Proposal, payment will be included in other items and no direct payment will be made. The LA DOTD Manual on Uniform Traffic Control Devices (MUCTD) shall be used as a guide for construction signing.

#### H-100-4.2 EXISTING DRIVEWAYS & DRIVEWAY CULVERTS

Under this contract, construction work may be required for driveways to be excavated, utilized as excavation storage area, etc. Irrespective of the construction use, under all circumstances the Contractor shall restore the surface of the driveway to its original condition including any necessary removal and replacement of existing driveway culverts. The Contractor shall replace existing culverts at no separate pay; he must include the cost of this work in other items.

#### H-100-4.3 EXISTING FENCES & FENCE STRUCTURES

Under this Contract, construction work conflicting with fences and fence structures may be encountered. It will be the responsibility of the Contractors to remove and replace any and all fences encountered. The fences shall be restored to their original condition.

There will be no separate pay item for this work, the Contractor must include the cost of this in other work.

#### H-100-4.4 PROTECTION OF EXISTING WATER FACILITIES

#### 1. Water Supply Interconnections

There shall be no physical connection between a public or private potable water supply system and a sewer, or appurtenance thereto which would permit the passage of any sewage or polluted water into the potable supply.

#### 2. Relation to Water Mains

#### (a) Horizontal Separation

Whenever possible, sewers should be laid at least 10 feet, horizontally, from any existing or proposed water main. Should local conditions prevent a lateral separation of 10 feet, a sewer may be laid closer than 10 feet to a water main if:

a.It is laid in a separate trench.

b.It is laid in the same trench with the water mains located at one side on a bench of undisturbed earth.

c.In either case the elevation of the crown of the sewer is at least 18 inches below the invert of the water main.

#### (b) Vertical Separation

Whenever sewers must cross under water mains, the sewer shall be laid at such an elevation that the top of the sewer is at least 18 inches below the bottom of the water main. When the elevation of the sewer cannot be buried to meet the above requirement, the water main shall e relocated to provide this separation or reconstructed with slip-on or mechanical-joint cast-iron pipe, asbestos-cement pressure pipe or prestressed concrete cylinder pipe for a distance of 10 feet on each side of the sewer. One full length of water main should be centered over the sewer so that both joints will be as far from the sewer as possible. No separate item will be made for such work. This work shall be included in the price per linear foot bid for sewer line construction.

#### H-100-4.5 TEMPORARY UTILITIES

The Contractor shall provide all temporary utilities such as electricity and water necessary to test and complete his various items. Should more than one contractor be involved in a project, the contractors shall coordinate with each other for the necessary utilities.

The Owner will pay for any permanent utility meter deposits; however, the Contractor shall be responsible for any utility bills incurred until the facility is turned over to the Owner for use.

#### H-100-4.6 PROTECTION OF THE SITE

The Contractor shall protect adjacent utilities, equipment, driveways, fences, cargo, automobiles, etc. The Contractor shall replace or compensate the Owner, for any damage he causes to physical property.

#### H-100-4.7 EXISTING UTILITIES

The work may be located in the vicinity of existing gas, telephone, sewer, water and/or electric lines, and the Contractor shall handle his work in a manner so as not to damage the existing utilities. He shall repair or pay for any repairs for damages done to existing utilities.

#### H-100-4.8 PHOTOGRAPHS

Contractor shall take photographs indicating the state of progress on a monthly basis. These photographs shall be mounted on 8-1/2" x 11", dated, identified and submitted as an attachment to the monthly request for payment.

#### H-100-4.9 WATER SUPPLY

Upon completion of Work requiring testing/disinfection, the Owner shall supply the water for testing/disinfection.

#### H-100-4.10 PRECONSTRUCTION DOCUMENTATION

The Contractor is responsible for documenting the condition of property and facilities adjacent to the work prior to construction. Contractor shall video tape and photograph the project area and provide copies of documentation to Engineer. In the event of damage claims to adjacent improvements the submitted documentation will be used for an initial interpretation of events.

#### H-100-4.11 STANDARD OF MANUFACTURER

Not applicable.

#### H-100-4.12 SAFETY FENCING

Safety fencing is required around the perimeter of unattended stationary excavations. Stationary excavations are those construction excavations intended to remain open in excess of 4 hours (such as boring entrance and exit pits).

Safety fencing may be a flexible fabric such as geo grid soil reinforcement, etc. Delineating ribbon or surveying tape will not satisfy this requirement. There is no separate pay for this work.

#### H-100-4.13 SALVAGED MATERIAL

Designated salvageable material shall be removed, without unnecessary damage, in sections which may be readily transported and shall be transported and stacked at a storage facility within ten miles of the project.

#### H-100-4.14 MINIMIZE NEGATIVE IMPACT ON THE PUBLIC

To help minimize the negative impact on the adjacent schools, business, and neighborhood the following shall be required by the contractor:

- a. Before any road is closed or detoured, the Contractor shall coordinate with the Owner and the Engineer.
- b. Maintenance of traffic shall be required in areas adjacent to the construction. In addition to required temporary construction signing, attractively painted signs shall be constructed and strategically placed to direct the public to schools or businesses directly affected by the construction process. These signs shall be maintained and updated as required for each sequence in the construction process. The Project Engineer shall inspect these "direction signs" and have final determination as to whether they are adequate for their intended purpose. Reference General Requirements, Section H-100-4.15 School/Business Direction Signs.

c. The Contractor shall minimize the sprawl of the construction process. Before the Contractor moves to the next sequence in the construction process, the site of the completed construction shall be cleaned up and graded to the satisfaction of the Project Engineer.

At the end of the project, a final cleanup and grading of the full length of the project shall be done as required to assure all areas affected by the construction process are properly graded, seeded and otherwise suitable to be returned to the general public. Final cleanup requirements are presented in the General Requirements, Section H-100-4.16 Final Cleanup.

d. In the case where traffic is to be maintained at an intersection (no 24 hour detour in place), intersection shall be paved one quarter at a time.

In the case when traffic is detoured for the duration of the reconstruction of an intersection, paving one quarter at a time shall not be required.

#### H-100-4.15 SCHOOL/BUSINESS DIRECTION SIGNS

Direction signs shall be provided to direct the public to the respective business or school, etc. adjacent to area affected by the construction process. These signs shall be attractively painted and made of 1/4O exterior plywood, or equal, with a 2Ox 4O wooden post, or equal.

The 1/4O exterior plywood panel shall be approximately 24O wide and 18O high. The painted lettering shall be approximately 4O to 5O tall with a 3/4O to 1O line thickness. Information such as name of the place being directed to and a direction arrow shall be provided.

A minimum of two signs per establishment shall be required. More than two (2) signs per establishment may be required, at no additional cost.

As a rule of thumb, any establishment whose driveway is directly affected by the construction process shall require direction signs. The Project Engineer shall make the final determination of which establishments will require direction signs and their location in the field. As the construction area varies or the sequence of construction changes, additional signs may be required and/or relocation of the signs needed.

All direction signing shall be paid for under Bid Item - 713(01) Temporary Signs and Barricades, when so included on the Bid Proposal. In the absence of Item 713, the cost shall be included in other related work.

#### H-100-4.16 FINAL CLEANUP

The contractor will be responsible for cleanup of the work specifically including removal of construction materials and debris; backfilling of trenches, depressions and ruts resulting from the work and general site grading and dress up to restore to the original condition.

The Contractor shall be fully responsible for the overall Final Project Cleanup, including removal of all construction debris and material, final project grading and dressup, seeding, etc. It is the intent of the final cleanup requirement that all areas subjected to construction will be fine graded and dressed so that they can be easily mowed with a push type residential mower and not pond water.

## **TECHNICAL PROVISIONS**

## **SECTION 202**

## REMOVING OR RELOCATING STRUCTURES AND OBSTRUCTIONS $\underline{ \text{TABLE OF CONTENTS} }$

| 202.01 | DESCRIPTION                                  |
|--------|--|
| 202.02 | GENERAL CONSTRUCTION REQUIREMENTS            |
| 202.03 | REMOVING STRUCTURES                          |
| 202.04 | RELOCATING STRUCTURES                        |
| 202.05 | REMOVING ENVIRONMENTALLY SENSITIVE MATERIALS |
| 202.06 | PLUGGING OR RELOCATING EXISTING WATER WELLS  |
| 202.07 | MEASUREMENT                                  |
| 202.08 | PAYMENT                                      |

This page intentionally left blank

# TECHNICAL PROVISIONS SECTION 202 REMOVING OR RELOCATING STRUCTURES AND OBSTRUCTIONS

**202.01 DESCRIPTION.** This work consists of the removal or the relocation of structures, facilities or obstructions, hereinafter referred to as "structures" from the project right-of-way unless specified otherwise.

The removal of a structure from the project right-of-way is the razing, demolishing, and disposal of the structure after salvageable parts, components, and materials, as designated on the plans, have been recovered by the contractor.

The relocation of a structure from the project right-of-way is its movement, reassembly, restoration, reconstruction, or equivalent replacement at a new location outside of, and adjacent to, the project right-of-way including all service connections, appurtenances, and accessories as directed.

For the purposes of this section, remove structures and obstructions visible at the time of bid, including all related structures or as designated in the plans. Structures may include buildings, floor slabs, foundations, fuel tanks, septic tanks, fences, pipes, bridges, drainage structures, pavements, walks, curbs, abandoned pipelines and other similar facilities, or obstructions not designated or permitted to remain within the project right-of-way. This work also includes backfilling of resulting trenches, holes, and pits. If structures or obstructions are encountered which differ materially from those ordinarily encountered, the provisions of 105.18 shall apply.

Quality assurance requirements shall be as specified in the latest edition of the Department's publication titled *Application of Quality Assurance Specifications for Embankment and Base Course.* 

Erosion control shall be in accordance with Section 204.

**202.02 GENERAL CONSTRUCTION REQUIREMENTS.** Remove and dispose of all portions of structures or obstructions on the right-of-way, except items for which other provisions have been made for removal or relocation. When specified, remove structures and appurtenances that extend beyond the right-of-way or that are entirely on private property. Remove specified salvageable material in sections which may be readily transported without unnecessary damage. Stack salvageable material at specified storage areas. When no storage sites are specified, deliver salvaged materials to the nearest DOTD maintenance unit. Dispose of materials not specified to be salvaged off the project right-of-way outside the view of the traveling public with written permission of the property owner on whose property the material is placed. Furnish copies of agreements (including rights of entry, etc.) With property owners to the engineer prior to beginning of work. The agreement must contain language holding the department harmless regarding any liabilities of the contractor or property owners. A certificate of release from the property owner will be required before final acceptance. Fill holes left by structure removal or the removal of materials associated with contaminated soils or sites by blading the area with surrounding soil or backfilling with soil complying with 203.06.1. Compact to a condition similar to the surrounding soils or as directed.

If any fuel storage tanks or other environmentally sensitive or contaminated sites are located during construction, stop construction activity in the immediate vicinity of the environmentally sensitive or contaminated site and notify the project engineer who in turn will notify the Department's Materials and Testing Section immediately for guidance. Testing and clean-up by the contractor shall be coordinated with the Materials and Testing Section.

The Department reserves the right to eliminate work items in accordance with 104.02.4.

**202.03 REMOVING STRUCTURES.** Unsalvageable materials in a structure designated for removal shall become the property of the contractor and shall be removed and disposed of by the contractor.

Demolish and remove appurtenances forming a part of a structure to be demolished, whether integral or not integral to the structure. Demolish and remove washhouses, garages, cisterns, and other buildings and appurtenances used in conjunction with a structure in the same manner as the structure. Remove existing yard fences, drives, walks, and shrubbery. The above are all considered part of the structure to be demolished and removed.

Plug and seal all abandoned water wells in accordance with 202.06. Demolishing of a structure, any part of which is used as a service station, shall include the removal of gasoline pumps, tanks, pipes, signs and other appurtenances. Remove underground fuel tanks in accordance with 202.05.2. Existing underground fuel tanks shall not be reused or used for other purposes.

Remove and dispose of material in existing foundations, concrete or masonry floors, chimneys, and other appurtenances.

Remove and dispose of cattle pens, cane derricks, cattle guards, or other such structures.

**202.03.1** Pavement, Base Courses, Walks, and Curbs: Dispose of pavements, stabilized or treated base courses, walks, curbs, and gutters, designated for removal as shown on the plans and as directed.

When the existing shoulder underdrain at the pavement edge is to remain in place and in service but removal of the shoulder surfacing and base is required, do not damage the existing shoulder underdrains. Damaged shoulder underdrains shall be satisfactorily repaired at no direct pay.

**202.03.2 Pipe:** Remove and store pipe that is to be re-laid so that there will be no loss or undue damage before relaying. Replace sections lost from storage or unduly damaged at no direct pay. When specified, pipe not to be re-laid and considered usable shall be salvaged, cleaned of soils or other materials, stored and stacked.

**202.03.3 Bridges and Drainage Structures:** Bridges, including approach slabs, and drainage structures in use by traffic shall not be removed until satisfactory arrangements have been made to accommodate traffic.

Unless otherwise directed or shown on the plans, remove substructures to natural stream bottom and those parts outside the stream to one foot below natural ground surface. Remove existing structures within the limits of a new structure as necessary to accommodate construction of the new structure.

Dismantle steel or wood bridges to be salvaged without unnecessary damage. Dismantling shall include stripping all hardware. Match-mark structural members before dismantling.

Explosives, when used, shall be in accordance with 107.11. Complete blasting or other operations necessary for removal of an existing structure or obstruction, which may damage new construction, prior to placing the new work.

**202.04 RELOCATING STRUCTURES.** Place structures to be relocated in their new locations as directed and restore to their original condition. Place structures to be relocated on foundations of the same type and character as the original foundations.

Relocate appurtenances forming a part of a structure to be relocated, whether integral or not integral to the structure, in the same manner as the structure. Relocate or replace appurtenances associated with the structure as directed with appurtenances of the same size, type, and character as existed before the structure was relocated.

Disconnect sanitary sewers, water, gas, electric, television cable, and telephone service lines connected to structures being relocated and reconnect as quickly as possible. The contractor shall be responsible for all notices to public utility companies and for all fees charged by them. Relocate existing yard fences, drives, and walks; extend same as necessary. Remove and replant existing shrubbery at new locations as designated. All of the above shall be considered as appurtenances not integral to the structures to be removed and relocated.

Remove material in existing foundations, concrete or masonry floors, chimneys and other appurtenances, when not used in reconstruction of appurtenances, and dispose of in accordance with 202.02. Furnish new material required in performing any of these operations at no direct pay.

Relocate contents of structures with the structure to its new site. When not feasible to relocate structures with contents therein, remove the contents from the structure at its original location and properly store and replace in the relocated structure without damage or loss to contents.

Relocate cattle pens, cane derricks, cattle guards, or other such structures on or beyond the right-of-way line as directed. Use materials in structures suitable for reuse in their reconstruction. Furnish new materials similar in kind to that in place at no direct pay, including foundations.

Prior to removal of butane or propane gas tanks, obtain the written approval of the Louisiana Liquefied Petroleum Gas Commission. Do not use or reuse existing underground butane or propane gas tanks for other purposes. The Department will reimburse the contractor for the cost of the new tank when the contractor presents the original receipted bill.

Furnish the engineer a Certificate of Release from each property owner; in case of separate ownerships of structure and property, furnish a Certificate of Release from each owner. This certificate shall state that the relocated structures are in an acceptable condition and that said owner waives all claims for damages to the property and structures relocated. When a Certificate of Release cannot be secured from the property owner, submit to the engineer a notarized letter documenting the inability to obtain the release.

202.05 REMOVING ENVIRONMENTALLY SENSITIVE MATERIALS. When removal or remediation of any environmentally sensitive or contaminated sites is required during construction, coordinate construction operations through the materials and testing section. If failure to follow the guidelines of the materials and testing section subsequently causes or

increases harm or damage to the environment, then all resulting fines and clean-up costs shall be the responsibility of the contractor.

**202.05.1 Asbestos:** When information is available, the Department will indicate on the plans which structures contain friable or non-friable asbestos. When a structure is identified on the plans or discovered on the project to contain asbestos and will be demolished or renovated, contact the Materials and Testing Section to coordinate disposal prior to commencing asbestos removal. Use a certified asbestos abatement contractor for proper removal and disposal. All applicable requirements for proper handling of asbestos material shall then be followed by the contractor for the continued removal of the asbestos containing material. Notify the Department of Environmental Quality (DEQ), Air Quality Division through the use of the proper notification form, DEQ AAC2, at least 10 calendar days prior to initiation of demolition or renovation of structure(s). The contractor shall maintain and furnish to the engineer, all records pertaining to the disposal of the asbestos containing material, either as non-friable or friable asbestos, within 21 calendar days of the material being removed from the site for disposal.

Asbestos-containing materials in structures that are removed or relocated without disturbing asbestos will not be abated. Provide a Certificate of Release to the engineer.

**202.05.1.1 Non-Friable Asbestos:** When a structure contains non-friable asbestos, carefully remove the asbestos without excessive breakage or crushing before demolition or renovation of the structure. Dispose of the non-friable asbestos material at an approved industrial landfill.

**202.05.1.2 Friable Asbestos:** When a structure contains friable asbestos, request that DEQ provide a confirmation letter with an Asbestos Disposal Verification Form (ADVF). Complete the ADVF within 90 calendar days from the date of issue. Only contractors or subcontractors certified by DEQ as Asbestos Abatement Entities shall remove friable asbestos from structures. Remove the asbestos before structure demolition or renovation. Perform friable asbestos removal, handling, and disposal in accordance with the latest requirements for asbestos abatement of the DEQ Air Quality Division.

Maintain, and furnish to the engineer within 21 calendar days, Chain of Custody verification records for the friable asbestos from the work site to the disposal site. These records will become part of the permanent project records.

202.05.2 Underground Fuel Tanks: Before removal, underground fuel tanks will be registered with the DEQ by the DOTD Materials and Testing Section as abandoned underground storage tanks. The contractor shall notify the project engineer in writing at least 45 calendar days prior to removal of tanks. Upon receipt of the contractor's notification, the engineer will immediately notify the Materials and Testing Section. All site activities, including the collection of closure samples and tank removal, as defined in the latest DEQ Underground Storage Tank (UST) regulations, shall be performed by a DEQ approved contractor. Submit closure test results, all documentation, and all necessary forms to the Materials and Testing Section to be approved and forwarded to DEQ. The contractor and/or the certified UST subcontractor shall note that all contact and/or coordination with the DEQ is to be the responsibility of the Materials and Testing Section.

Take all necessary precautions to prevent the infiltration of water into tanks and tank excavations during the work.

During routine site closure, the removal, transportation, and disposal of tanks, and the handling of contaminated soil and contaminated fluid, shall be in accordance with all local, state,

and federal laws and regulations. Limits of excavation and quantities of contaminated soil and contaminated fluid to be removed, transported, and disposed shall be as specified or as directed.

When underground storage tanks (UST) have been filled with concrete, sand, or other such material and are designated on the plans for removal, the contractor or certified UST subcontractor shall remove, transport and dispose of such tanks in accordance with the recommendations of the American Petroleum Institute (API) and the requirements of the Louisiana Department of Environmental Quality (DEQ) or other regulatory agency of jurisdiction. When such UST are discovered during construction, stop construction activity in the immediate vicinity of the UST and notify the project engineer in accordance with this subsection. The DOTD Materials and Testing Section will verify the closure status of such filled UST discovered during construction prior to any UST site activity by the contractor or certified UST subcontractor. The contractor or certified UST subcontractor shall collect and submit for laboratory analysis a representative sample of non-solidified fill material within the storage tank for landfill acceptance. The results of the laboratory analysis shall be used to determine the disposition of the UST fill material. Provide a copy of all laboratory analyses to the Department's Materials and Testing Section for verification prior to profiling materials for landfill acceptance.

**202.05.3 Contaminated Soils:** Soil in areas of underground fuel tanks or other areas contaminated with petroleum products or other identified toxic materials at levels above the regulatory limits and is non-protective of groundwater shall be excavated as shown on the plans or as directed. Determination of groundwater protection shall be through the use of the Synthetic Precipitation Leachate Procedure (SPLP) or as directed by the Materials and Testing Section.

Remove the overburden above the contaminated soil to the dimensions shown on the plans or as directed. Also, excavate the contaminated soil at the locations shown on the plans or as directed. Contaminated soil determined to be protective of groundwater, through the use of the SPLP, shall be excavated by the contractor and placed in the roadbed when the soil is determined to be "suitable soil" by the engineer, and when the volume of soil is within quantities specified on the plans. No additional cover of the contaminated soil, other than the specified paved surfaces courses, will be required in the roadbed.

All remaining contaminated soil determined to be protective of groundwater, but not used in the roadbed, shall be placed in other embankment areas within the limits of the project.

Contaminated soil placed in other embankment areas shall be covered with 2 feet of compacted soil by the contractor in accordance with Section 203. Final grade shall be maintained in accordance with the plans. Load the contaminated soil determined not to be protective of groundwater into approved hauling vehicles and dispose of in a site approved by the DEQ. Furnish the engineer, within 21 calendar days, Chain of Custody verification records for the contaminated soil. The Materials and Testing Section will verify that all contaminated soil has been removed.

While the excavation is open, construct and maintain a soil berm around the excavation to prevent surface water runoff from entering the excavation. The removed overburden may be used to construct the berm and backfill the excavation.

Removal and disposal of contaminated soils will be in accordance with all local, state, and federal laws and regulations.

**202.05.4 Contaminated Fluids:** Remove and dispose of contaminated fluid, in underground fuel tanks, in areas of underground fuel tanks, or other areas as shown on the plans or as directed.

The Department will determine the quantity of contaminated fluid to be removed.

Pump the contaminated fluid into approved hauling vehicles. Remove contaminated fluid from underground fuel tanks before tank removal.

Dispose or recycle of contaminated fluid in a site approved by the Department of Environmental Quality. Furnish the engineer, within 21 calendar days, Chain of Custody verification records for the contaminated fluid.

The Department will verify the removal of the contaminated fluid. Removal and disposal of contaminated fluids will be in accordance with all local, state, and federal laws and regulations.

202.05.5 Paint Containing Lead or Other Hazardous Materials on Metal Surfaces: Remove steel members of structures protected by paint containing lead or other hazardous materials as shown on the plans or as discovered in the field and prepare for transport in accordance with Section 107.

Prior to removal, transport, treatment, or disposal of any steel members, submit the following to the engineer:

- 1. Plan of removal or treatment of steel members.
- 2. Plan for transport of steel members and any hazardous materials.
- 3. Name and address of the licensed recycling center.

Deliver such steel members to a licensed recycling center capable of processing steel members coated with paint identified as hazardous by the Resource Conservation and Recovery Act (RCRA).

The DOTD or the Owner will be the Generator and obtain the generator number. The contractor will be responsible for obtaining an approved disposal site, arranging for transporting the material and/all testing required. The manifest for transportation will have the DOTD Generator number on it and should be signed by the contractor, DOTD Inspector, and the Disposal Operator with copies to each upon completion.

Unless otherwise directed or shown on the plans, the contractor will be allowed to retain any steel member once the lead paint has been removed and disposed of prior to steel leaving the jobsite in accordance with procedure above at no cost to the Department.

Transport all steel members or hazardous material in accordance with all federal, state, and local laws. Provide Certificates of Disposal, Chain of Custody forms, or other applicable documents within 21 calendar days following each shipment.

**202.05.6 Treated Timber:** Remove creosoted and other treated timber or lumber shown on the plans or discovered in the field; and prepare for transport by methods approved by the Department. Dispose of all materials that are not designated to be salvaged by the Department or salvaged by the contractor in an appropriate landfill. Provide Certificates of Disposal, Chain of Custody forms, or other applicable documents within 21 calendar days following each shipment.

**202.05.7** Universal Wastes: Universal wastes are hazardous wastes defined in LAC Title 33, Part V, Chapter 38, Section 3813 to include batteries, pesticides, thermostats, lamps and antifreeze. Universal wastes shall be removed in accordance with the plans and shall be stored and prepared for transport as specified in LAC Title 33, Part V, Chapter 38 and herein.

Inform all employees who handle universal wastes of the proper handling and emergency procedures appropriate to the type of waste.

**202.05.8** Other Regulated Materials (ORM): Items for removal under this subsection are defined as any material not considered in the above subsections and may be disposed of as a

solid waste in the appropriate solid waste landfill. Such materials may include asphalt shingles, noninfectious medical waste, etc. not covered in other items.

**202.06 PLUGGING OR RELOCATING EXISTING WATER WELLS.** Plug and seal all abandoned water wells at the locations shown on the plans, or as directed by the engineer, in accordance with the Water Well Rules, Regulations, and Standards, State of Louisiana. This document is available at the department of transportation and development, water resources section. Water well abandonment must be accomplished by a DOTD licensed water well contractor. Relocated water wells shall conform to the Sanitary Code of the State of Louisiana as prepared and promulgated by the Louisiana State Board of Health.

**202.07 MEASUREMENT.** Removing structures and obstructions will be measured on a lump sum basis or by the unit as stipulated in the contract and shall include appurtenances, foundations, etc.

Hauling salvaged materials to storage sites will not be measured for payment.

When an item is included for removal of bridges, the removal of the approach slabs, superstructure, and substructure will be considered part of the work unless otherwise specified or shown on the plans.

Removing or relocating structures will be measured by the unit stipulated in the contract. Each principal structure with its associated appurtenances, whether integral or not integral to the structure being removed or relocated, will be considered as a separate unit including its associated appurtenances.

Plugging of existing abandoned water wells or relocating water wells will be measured per each well plugged and accepted or relocated.

Measurement for removal of contaminated soil and non-contaminated overburden will be by the cubic yard using the in-place quantities as determined by cross-sections.

Measurement for contaminated fluid will be by the gallon.

Removal, transportation, and related fees for disposal of steel members of structures protected by paint containing lead or other hazardous materials, or creosoted timbers or lumber, will be considered part of the work when shown on the plans and will not be measured for payment.

When a structure to be removed or relocated is shown on the plans to contain universal wastes, the removal, storage and transport of the universal waste to an approved disposal site or destination facility will not be measured for payment but will be included in the structure to be removed or relocated.

Measurement for removal of Other Regulated Materials (ORM) will be as designated on the plans.

**202.08 PAYMENT.** Payment for removal of structures or specific obstruction items stipulated for removal and disposal under unit price or lump sum pay items will be made at the contract price per unit or lump sum as specified. This will include demolishing, removing and disposing of such items as well as the excavation and backfill incidental to their removal when required. When the removal is in an area to be excavated and payment for excavation is made

under other items, no deduction will be made from the excavation quantities. The price shall also include salvage of materials, their custody, preservation, storage on the right-of-way or as designated on the plans, and disposal.

Payment for the removal of bridges will include removal of the approach slabs, superstructure, and substructure.

Payment for the relocation of structures will be made at the contract unit price which will include all costs for moving, reassembly, restoration, reconstruction, or equivalent replacement of the structures.

Payment for plugging and sealing existing abandoned water wells or relocating existing water wells will be made at the contract unit price which will include all labor, material, tools, equipment, and incidentals necessary to complete the work.

If a structure is to be removed or relocated as a unit under Pay Items 202-01, 202-02, or 202-03 and;

- 1. The contractor enters into an agreement with a property owner for disposition of the structure other than as shown on the plans; or if it is subsequently determined that said structure can remain in place, in whole or in part, with or without minor adjustments, and;
- 2. The contractor enters into an agreement with the property owner incorporating such revised determination and any accompanying adjustments regarding said structure, including any damages for leaving the structure in place, then;
- 3. The contractor shall furnish such agreement to the engineer for approval. If approval is given by the engineer, the contractor shall furnish the Department with a Certificate of Release from the property owner for the unit.

In case of multiple ownership interest in the structure and/or property, a Certificate of Release from each owner shall be furnished. This certificate shall state that said owner waives all claims for damages to the property and structure to be removed, relocated, left in place, or otherwise handled to the owner's satisfaction.

Except as provided hereinafter, the contractor will be paid for removing, relocating, or other handling of the structure at the contract unit price as listed under Pay Items 202-01, 202-02, or 202-03. However, a deduction will be made in such amount if:

- 1. A determination to allow the structure to remain in place results in a decrease in cost to the contractor;
- 2. An allowance is made to the owner for damages to the property or structure caused by the contractor, or its subcontractor, agent, or assign; or
- 3. Any other adjustment of the contract amount for removal, relocation, or other handling of said unit under Pay Items 202-01, 202-02, or 202-03, is deemed justified.

When a structure has been identified on the plans as containing friable or non-friable asbestos, the price for asbestos removal and disposal will be included in the bid price for removal, relocation, or demolition of the structure. When a structure is found to contain friable or non-friable asbestos and it has not been identified on the plans as containing asbestos, payment for the removal and disposal of the asbestos will be made in accordance with 109.04, including the cost of all testing.

Payment for removal, transportation, and disposal of contaminated soils and fluids will be in accordance with rates specified in applicable appendices (currently Appendices A and B) of the *Louisiana Motor Fuels Underground Storage Tank Trust Fund Cost Control* guidance document as maintained and updated by the Louisiana Department of Environmental Quality (DEQ). The DEQ cost control guidance document can be obtained at DEQ website. All payments under this

item will be in accordance with industry standards, which include all equipment, labor, and materials necessary to complete the work, including backfilling any excavation. Payment for work not covered in the cost control guidance document, or any disputed payments, will be negotiated and resolved prior to performance of work. The Department will reimburse the contractor monthly for the actual incurred cost for such services. The contractor shall furnish documentation with the request for reimbursement.

Payment for removing steel members of a structure identified on the plans as being protected by paint containing lead or other hazardous materials, or creosoted timbers or lumber, and transporting them to the recycling center or landfill, will be included in the bid price for removal or relocation of the structure. When a structure is determined to have steel members protected by paint containing lead or other hazardous materials, or creosoted timber or lumber, and it has not been identified on the plans as such, payment for removal and transport of the members to a licensed recycling center or landfill will be made in accordance with 109.04. Unless otherwise directed or shown on the plans, the contractor will be allowed to retain any steel member once the lead paint has been removed and disposed of prior to steel leaving the jobsite and a Chain of Custody form or other applicable documentation is submitted to the engineer within 21 calendar days.

When the plans show that a structure to be removed or relocated contains a universal waste, payment for the removal of the universal waste will be included in the contract unit price for the removal or relocation of the structure, which will also include all equipment, labor, and materials required for the removal, storage, and transport of the universal waste in accordance with LAC Title 33, Part V, Chapter 38. When a structure to be removed or relocated is found to contain a universal waste and it is not identified as such on the plans, payment for the removal, storage, and transport of such universal waste in accordance with LAC Title 33, Part V, Chapter 38 will be made in accordance with 109.04.

Payment for removal of other regulated materials (ORM) will be as designated on the plans.

#### Payment will be made under:

| Item No. | Pay Item                                | Pay Unit    |
|----------|---|-------------|
| 202-01   | Removal of Structures and Obstructions  | Lump Sum    |
| 202-02   | Removal of                              | As Required |
| 202-03   | Relocation of                           | As Required |
| 202-04   | Excavation, Disposal and Backfilling of |             |
|          | Non-Contaminated Overburden             | Cubic Yard  |
| 202-05   | Excavation, Disposal and Backfilling of |             |
|          | Contaminated Soil                       | Cubic Yard  |
| 202-06   | Removal and Disposal of Contaminated    |             |
|          | Fluid                                   | Gallon      |
| 202-07   | Plugging Existing Water Wells           |             |

This page intentionally left blank

## **TECHNICAL PROVISIONS**

## **SECTION 203**

## EXCAVATION AND EMBANKMENT

## **TABLE OF CONTENTS**

| 203.01 | DESCRIPTION           |
|--------|-----------------------|
| 203.02 | GENERAL EXCAVATION    |
| 203.03 | DRAINAGE EXCAVATION   |
| 203.04 | UNSUITABLE MATERIAL   |
| 203.05 | BORROW                |
| 203.06 | SOIL USAGE            |
| 203.07 | GENERAL REQUIREMENTS  |
| 203.08 | CUT AREA PREPARATION  |
| 203.09 | NONPLASTIC EMBANKMENT |
| 203.10 | PLASTIC SOIL BLANKET  |
| 203.11 | GEOTEXTILE FABRICS    |
| 203.12 | QUALITY CONTROL       |
| 203.13 | ACCEPTANCE            |
| 203.14 | MEASUREMENT           |
| 203.15 | PAYMENT               |

This page intentionally left blank

# TECHNICAL PROVISIONS SECTION 203 EXCAVATION AND EMBANKMENT

**203.01 DESCRIPTION.** This work consists of excavation, disposal, placement, and compaction of materials for which provisions have not been made under other sections of these specifications. This work shall include excavation and embankment construction for roadways and other structures, excavation for ditches and channels, and other grading operations necessary for the work in accordance with these specifications and in conformity with the lines, grades, thicknesses, and typical sections shown on the plans or established. When specified, supply, install, and monitor settlement plates. When contaminated soils or underground tanks are encountered, handling shall be in accordance with Section 202. Disposal of material shall be in accordance with 202.02.

The plans may include data regarding the boring and classification of existing materials. The Department does not guarantee that individual samples are representative of the entire project, and bidders are required to study, make interpretations and additional investigations, as necessary, at no direct pay. The bidder shall determine the suitability of the on-site soils to meet specifications of Section 203.

The contractor shall comply with 107.09 for work in, over or adjacent to navigable waters and wetlands, and shall comply with 107.27 when cultural artifacts, historical sites, or archaeological sites are encountered.

Quality assurance requirements shall be as specified in the latest edition of the Department's publication titled *Application of Quality Assurance Specifications for Embankment and Base Course*.

Excavated material may be used in accordance with 203.06. Temporary erosion control shall be in accordance with Section 204.

**203.02 GENERAL EXCAVATION.** General excavation consists of the excavation of materials, as required by the plans, except drainage excavation and structural excavation. General excavation also includes unsuitable material in accordance with 203.04.

#### 203.03 HYDRAULIC EXCAVATION.

**203.03.1 Drainage Excavation:** Drainage excavation includes the excavation for drainage beyond the limits of the roadway section. Drainage excavation also includes inlet and outlet ditches to structures or roadways; changes in or deepening of channels of streams, berm ditches, ditches parallel or adjacent to the roadway beyond the limits of the roadway section; and material excavated from areas under bridges.

**203.03.2 Cleaning Existing Ditches:** This item consists of excavating and disposing of sediment and vegetative materials from existing ditches in order to reestablish flow lines on the existing alignment in accordance with plan details and the following. Establishment of ditch grades, if necessary for this item, will be the responsibility of the project engineer.

Unless otherwise directed, dispose of material excavated from existing ditches in accordance with 202.02.

**203.04 UNSUITABLE MATERIAL.** Unsuitable materials are soils containing significant amounts of debris or organic matter including stumps, roots, logs, and humus, or other materials which will decay or produce subsidence, including highly saturated soils, which the engineer determines are not satisfactory for use in the embankment or other construction purposes. Remove unsuitable materials and dispose of as general excavation. Remove unsuitable materials determined to be environmentally sensitive and dispose of in accordance with 202.05.

**203.05 BORROW**. Borrow is defined as soils required for construction of embankments or other portions of the work in excess of soils obtained from excavation. Obtain borrow from an approved source and use in accordance with 203.06. Make arrangements for obtaining borrow at no direct pay.

Securing of an exclusive option by a bidder or contractor on borrow areas or materials for the work will be considered a violation of Louisiana law and will be a basis for rejection of bids or such other action the Department deems advisable.

Notify the engineer in writing a minimum of 30 calendar days in advance of contractors borrow operations so that samples may be taken and soil tests completed prior to beginning borrow operations. Include in notification: boring requests, quality control test results, property owner agreements, and other information as required by the Materials Sampling Manual and the *Application of Quality Assurance Specifications for Embankment and Base Course* to the District Lab.

Unless otherwise authorized in writing, locate borrow pits, gravel pits, and quarry sites at least 300 feet from the right-of-way.

When sources of borrow are located adjacent to a stream or river listed on the National System of Wild and Scenic Rivers or the Louisiana Natural and Scenic Rivers System, locate borrow pits and any stockpiled materials at least 300 feet from the natural stream or river bank.

Clear the borrow pit and access to allow access for DOTD boring equipment. Survey the borrow area with a base line staked. Furnish both the engineer and laboratory with a location plat and borrow pit plat. Do not begin borrow operations until materials are approved for use.

Sampling of soils from open excavations in lieu of borings will be allowed provided the open excavations display and allow sampling of each soil strata and the excavation is at no cost to the Department.

**203.06 SOIL USAGE.** The laboratory will test and classify soil in accordance with DOTD TR 423 from samples taken in the original location or from designated stockpiles. Soil shall be classified and approved prior to its being placed in embankments or other final positions on the project. Blending in the pit by approved methods to adjust percent silt or sand will be permitted. Do not blend soils that do not meet Liquid Limit or PI requirements in order to modify the Liquid Limit or PI. Soils may be treated with lime to reduce PI in accordance with 203.06.5.

Soil properties will be determined by the test methods shown in Table 203-1, "Soil Properties."

Table 203-1 Soil Properties

| Property              | Test Method |
|-----------------------|-------------|
| Plasticity Index (PI) | DOTD TR 428 |
| Liquid Limit (LL)     | DOTD TR 428 |
| % Organic             | DOTD TR 413 |
| % Silt                | DOTD TR 407 |
| рН                    | DOTD TR 430 |

- **203.06.1 Usable Soils:** Usable soils shall have a maximum PI of 25 and a maximum organic content of 5 percent. Soils with a silt content of 50 percent or greater and also a PI of 10 or less will not be allowed.
- **203.06.2** Nonplastic Embankment: Nonplastic embankment shall be as specified in 203.09.
- **203.06.3 Headers:** Headers are that portion of the embankment within 500 feet of a bridge end. Construct headers for their full height with usable soils meeting the requirements of 203.06.1, and having a minimum PI of 11, a maximum PI of 25, and a maximum silt content of 65 percent. No lime treatment to the soil to meet the PI requirements will be permitted. Compact headers to 98 percent of maximum dry density in accordance with 203.07.
- **203.06.4** Embankments other than Headers: Construct embankments with usable soils meeting the requirements of 203.06.1. Soil with a PI greater than 25 and less than 35 will be permitted when treated with a minimum of 6 percent lime, by volume, provided the organic content and silt requirements given in 203.06.1 are met. If lime treatment is used, it will be at no direct pay. Lime treatment shall be Type E Treatment conforming to Section 304.

The contractor may request in writing that usable soils for temporary detour roads have a PI not to exceed 45 and a maximum silt content of 75 percent provided:

- 1. This material will be removed and not become part of the permanent embankment, and
- 2. The contractor agrees to take responsibility for any additional maintenance required.

#### 203.06.5 Plastic Soil for Slopes

- 203.06.5.1 Use Topsoil in Accordance with Section 715 Embankment Material: The outside layer of embankment (fill sections) will consist of a plastic soil blanket in accordance with 203.10.
- **203.06.5.2** Cut Slopes, PI Less than 10: When soils having a PI less than 10 exist on cut slopes, undercut 12 inches and place a plastic soil blanket conforming to 203.10.
- 203.06.5.3 Cut Slopes, PI 10 or Greater: When soils having a PI of 10 or greater but with a pH less than 5.5 or greater than 8.5 exist on cut slopes, undercut and place a plastic soil blanket complying with 203.10. In lieu of furnishing a plastic soil blanket, the soil may be modified in place so that the pH of the soil complies with the requirements of 203.10, at the option of the engineer and concurrence of the contractor. In such case, payment will be in accordance with existing items or 109.04, as applicable, not to exceed the cost of undercut and replacement.

203.06.6 Usable Soils for Slope Adjustments and Shoulder Widening: When the thickness of embankment material used for slope adjustment or unpaved shoulder is less than 12 inches, a plastic soil complying with 203.10 will be required. If the thickness is greater than 12 inches, the contractor will be allowed to substitute plastic soil for usable soil, provided the widening is not directly below a paved shoulder.

#### 203.07 GENERAL REQUIREMENTS

203.07.1 General: Excavation and embankment work consists of constructing roadway embankments, including preparation of surfaces on which they are to be placed; constructing drainage excavation; constructing backslopes; constructing dikes, when required; placing and compacting approved material in areas where unusable material has been undercut and removed; placing and compacting embankment material in holes, pits, and other depressions.

Do not place or spread embankment materials on portland cement concrete or asphalt concrete pavements. Do not damage pavement surfaces, edges and joints during embankment operations.

**203.07.2 Surface Layer Preparation:** Complete all necessary clearing and grubbing in an area, prior to beginning excavation, grading, or embankment operations in that area. Prior to any embankment operations in an area, cut ditches as required to facilitate drainage in that area unless otherwise noted on the plans.

When preparing surface layers on which the embankment or base is to be placed, attempt all normal earthwork construction methods before undercutting or modifying the soil with additives. Such construction methods may include, but are not limited to, the following and will be at no direct pay:

- 1. Draining and drying of the surface until the material is within the limits of optimum moisture before compaction is attempted.
- 2. Using lighter weight construction equipment for manipulating, disking, drying, and compacting the material.
- 3. Placing successive loads of approved material in a uniformly distributed layer of a thickness necessary to support equipment while placing subsequent layers.
- 4. Rerouting heavy construction equipment around the area until the embankment can support the equipment without damage to foundation soils.

Remove unstable materials by undercutting, unless otherwise directed, and backfill to required section with usable soils as directed.

When undercutting is required, conduct the operations in such manner that the engineer can make necessary measurements before backfill is placed.

When a new roadway is to be constructed on an existing roadbed, remove existing surface courses. When the surface of the existing roadbed is within 2 foot of finished sub-grade, scarify the existing roadbed full width to a depth of not less than 9 inches and re-compact to at least 95.0 percent of maximum dry density.

**203.07.3 Excavation:** Excavated material shall become the property of the contractor. Soils from excavation areas may be used in embankments or other finished sections when approved. Dispose of surplus or unusable excavated material in accordance with 202.02 or as provided in this subsection.

When obliteration of old roadways is required, include grading operations necessary to satisfactorily incorporate the old roadway into the new roadway and surroundings to the satisfaction of the engineer and to allow drainage.

**203.07.4 Settlement Plate Installation and Monitoring:** Furnish and install settlement plate as shown on the plans.

Install settlement plate prior to placement of any fill. Place settlement plate on top of the geotextile fabric if shown on the plans, otherwise, place the settlement plate on natural ground. Maintain a vertical and undamaged riser pipe during embankment placement and compaction. Replace any settlement plate damaged during construction at no cost to the Department.

Establish the elevation of the settlement plate. Take initial settlement plate elevation readings immediately after installation. Unless specified on the plans, monitor settlement plate as follows: immediately after the final lift placement, twice weekly for the first month, and weekly for five months thereafter. Repeat this sequence if any additional fill placement or surcharge is required. Record the embankment elevation and surcharge height for each settlement plate reading. Submit settlement plate readings to the Project Engineer. The Project Engineer will verify the initial and final elevations for acceptance.

Remove riser pipe and casing three feet below base course upon acceptance.

203.07.5 Embankment: Embankment material shall be in accordance with 203.06. Place in uniform layers not exceeding 12 inches of un-compacted thickness. Place each layer for the full width of embankment, blend as necessary to obtain a uniform material, bring to a uniform moisture content, and compact to a minimum of 95.0 percent of maximum dry density before the next layer is placed. Determine maximum dry density in accordance with DOTD TR 415 or TR 418 and percent in-place density in accordance with DOTD TR 401. The density of the embankment shall be such that the density of the type of base course being constructed shall be met. The moisture content at the time of compaction, tested in accordance with DOTD TR 403, shall be within a range of  $\pm 2.0$  percent of optimum moisture established in accordance with DOTD TR 415 or TR 418. If not, reprocess and re-compact the lifts until these requirements are met.

Topsoil shall be placed and compacted in accordance with 715.03.

Ensure that final embankment slope lines are uniform in appearance. Measure as necessary to assure that the elevations at the top, bottom, and intermediate breaks in the slope are such that minimum acceptable slopes are achieved. Visually inspect the slopes and ensure the slopes are straight without valleys or humps. If an apparent discrepancy is discovered upon visual inspection, take measurements a minimum of every 10 feet measured along the slope between theoretical break points in the embankment. Allowable tolerances for slope grade will not be less than by 0.03 foot/foot nor greater than 0.15 foot/foot. The slopes shall be reworked until these criteria are met. The top of embankment shall not vary from the established grade by more than  $\pm 0.1$  foot.

Conduct operations to prevent lamination between lifts. Correct all laminations between lifts prior to placing additional lifts. Assure that surfaces of excavated areas and embankments are smooth and uniform. Do not disturb material outside the construction limits.

When excavation and embankment construction results in surface soils having a PI less than 10, or pH less than 5.5 or greater than 8.5, place a plastic soil blanket complying with 203.10.

The contractor shall be responsible for the stability of embankments until final acceptance. Construction activities which may lead to subsequent embankment damage, will not be permitted.

When embankments are constructed on a surface sloping more than 6:1 from the horizontal, cut the slope of the ground on which the embankment is to be placed into steps, as directed, before fill is placed.

When an embankment is to be constructed to a height of less than 5 feet, remove heavy sod and objectionable vegetation from the area on which the embankment is to be placed. Scarify the area to a depth of approximately 9 inches. Re-compact this area to at least 95.0 percent of maximum dry density in accordance with DOTD TR 415 or TR 418 and percent in-place density in accordance with DOTD TR 401. When height of fill is 5 feet or more, removal of sod will not be required, but disk the area on which embankment is to be placed to the satisfaction of the engineer and re-compact before construction of embankment.

When embankment material is to be deposited only on one side of abutments, wing walls, piers, or culvert head walls, do not compact the area immediately adjacent to the structure to the extent that it will cause excessive pressure against the structure. When the embankment is to be deposited on both sides of a concrete wall or similar structure, conduct operations so that the embankment is always at approximately the same elevation on both sides of the structure. Backfill structures in accordance with Section 802.

When embankments are constructed in lakes, streams, swamps, or other unstable areas and unstable material cannot be removed or the area drained, the requirement for placing material in layers as outlined above may be waived. When this requirement is waived, place the embankment by end dump or other approved methods to an elevation where normal construction methods can begin. Construct embankments placed above this elevation in layers as specified above. When a wave of unsuitable material is forced up in front of the end dumping operation, it shall become the property of the contractor and be removed as necessary. In addition, do not allow this material to be trapped and incorporated in the embankment except as part of plastic soil for slopes.

**203.08 CUT AREA PREPARATION**. Scarify and compact the top 12 inches of the cut area to such density that the compaction requirements of the type base course being constructed shall be met. Construction, compaction, and testing requirements shall comply with 203.07.

When unstable soils are encountered, the engineer will determine the limits to be undercut. Excavate to a stable foundation or to the depth required by the engineer and backfill to existing grade. Undercut shall be constructed and tested in accordance with 203.07. When a stable foundation cannot be reached, "bridge-in" the embankment materials and construct the remaining embankment to existing grade in accordance with 203.07.

#### 203.09 NONPLASTIC EMBANKMENT

**203.09.1 Materials:** Non-plastic embankment material shall comply with 1003.09 or the following, unless otherwise specified on the plans.

**203.09.2 General Requirements:** Do not entrap unsuitable material defined in 203.04 in the embankment. Remove any such material at no direct pay.

Leave surcharge materials on the embankment for at least the specified number of days after approval of the increment. Damage to embankment increments due to the contractor's operations shall be satisfactorily repaired by the contractor at no direct pay. Remove excess surcharge materials after the surcharge period. Verification cross-sections of the final embankment will be taken after removal of the surcharge. Material required due to additional subsidence after cross-sections are taken will be paid under the appropriate item.

After all embankment increments have been surcharged, satisfactorily dispose of excess surcharge material in accordance with 202.02 at no direct pay.

Except for stone embankments, furnish and place a plastic soil blanket complying with 203.10.

**203.09.3 Nonplastic Embankment Construction:** Construct nonplastic embankments by mechanical methods.

Unless otherwise shown on the plans, place material in lifts not exceeding 15 inches of uncompacted thickness after establishing a working table as directed. Compact each lift and test in accordance with 203.07.

203.09.4 Blended Calcium Sulfate Embankment Construction: Add water or use other suitable means to prevent dust resulting from the transport and placement of dry material. Place blended embankment material in lifts not exceeding 12 inches in thickness (loose) after establishing a working table as directed. Compact each lift to at least 95 percent of maximum dry density prior to placement of subsequent lifts. Determine the maximum density in accordance with DOTD TR 418 modified to include a drying temperature not to exceed 140°F. Perform field density testing in accordance with 203.07. Determine moisture content for density corrections by oven drying the material at 140°F for a minimum of 24 hours. Provide a forced draft type oven capable of maintaining this temperature. Also, furnish and place a plastic soil blanket complying with 203.10.

Do not place blended calcium sulfate within 10 feet of metal drainage structures. The contractor will be allowed to substitute natural stone, flowable fill under Section 710, or other material in 1003.08 as approved by the Department.

203.10 PLASTIC SOIL BLANKET. Plastic soil blanket shall consist of soils having a minimum PI of 11, maximum PI of 35, a maximum silt content of 65 percent, and a pH not less than 5.5 or greater than 8.5, and a minimum organic content of 3 percent. The contractor will be allowed to blend organic materials to achieve the minimum 3 percent organic content. The plastic soil blanket shall support a satisfactory stand of grass in accordance with Sections 714 or 717. Construct the soil blanket to a minimum thickness of 12 inches. Areas requiring a plastic soil blanket shall be approved prior to placement of the plastic soil blanket. After materials are placed and spread, remove lumps, stones, roots and other foreign matter from the area. Spread and roll soil blanket material in a manner that leaves a uniform surface. Ensure that any remaining ridges or grooves, including cleat tracks from the dozer, will be parallel to the roadway during the period of time between placement and seeding.

Place plastic soil blanket in a timely manner to prevent erosion.

#### 203.11 GEOTEXTILE FABRICS

- **203.11.1 General:** Furnish and place geotextile fabric in accordance with these specifications and in conformance with the details shown on the plans.
  - **203.11.2 Materials:** The geotextile fabric shall comply with Section 1019.
- 203.11.3 Construction Requirements: Keep rolls of geotextile fabric covered and protected from ultraviolet degradation at all times until use. Cover geotextile fabric that has been installed with embankment material within seven calendar days. When ultraviolet damage occurs, remove and replace the geotextile fabric. Place the geotextile fabric at the locations shown on the plans or as directed. Overlap or sew adjacent rolls of geotextile fabric. When rolls are overlapped, overlap a minimum of 18 inches or as specified in the plans, including the ends of the rolls. Place the top layer of the geotextile fabric parallel with adjacent rolls and in the direction of embankment placement. When rolls are sewn, join adjacent rolls by sewing

with polyester or kevlar thread. When field sewing, employ the J-seam or "Butterfly" seam with the two pieces of geotextile fabric mated together, turned inwards so as to sew through four layers of fabric. Sew with two rows of Type 401, two-thread chain stitch. Factory seams other than specified shall be submitted to the Materials and Testing Section for approval. Where the ground is covered with water or soil is saturated, sewing of the geotextile fabric will be required.

Place the geotextile fabric as smooth as possible with no wrinkles or folds, except in curved road sections. For curved road sections, fold the geotextile fabric to accommodate the curve. The fold shall be in the direction of construction and pinned or stapled. Fill and compact ruts that occur during construction prior to placement of geotextile fabric.

Remove damaged geotextile fabric and replace with new geotextile fabric or cover with a second layer of geotextile fabric extending 2 feet in each direction from the damaged area.

**203.12 QUALITY CONTROL.** Locate, select, and place material conforming to specification requirements. Control processes, including performing tests and making adjustments as necessary, to result in a uniform quality product meeting all the requirements of the plans and specifications. Perform tests for in-place moisture content in accordance with DOTD TR 403, at a frequency that will ensure that the material is within the tolerances of optimum moisture. Perform tests for in-place density in accordance with DOTD TR 401 at a frequency that will ensure that the compactive effort is producing a uniform product that conforms to specification requirements. Control placement and finishing to ensure conformance with the lines, grades, thickness, and typical cross-sections shown on the plans or established.

Sections will be inspected prior to acceptance testing. Correct obviously deficient areas prior to acceptance testing.

**203.13 ACCEPTANCE.** The Department will perform inspection, sampling, and testing for acceptance. Correct any area that is deficient whether identified by inspection or testing.

The embankment (with surcharge, if required) will be approved in increments of 1000 feet, except terminal increments which may be less than 1000 feet.

Maximum density for earthwork will be determined in accordance with DOTD TR 415 or DOTD TR 418; in-place density will be determined in accordance with DOTD TR 401.

#### **203.14 MEASUREMENT**

**203.14.1 General:** Unless otherwise specified, borrow material in accordance with 203.05, topsoil, and plastic soil for slopes in accordance with 203.06.6 will be considered incidental to the embankment and will not be measured separately, but will be measured as embankment. Removal and stockpiling of existing topsoil will be measured by the in-situ square yard.

Measurement of undercut will be from subgrade or original ground, whichever is lower.

No measurement will be made for excavation for culverts or culvert headwalls.

When the grade line of a pipe or box culvert is raised or lowered more than 2 feet from the grade line shown on the plans or is relocated to a site requiring an equivalent change in excavation, payment will be increased or decreased accordingly at the rate of three times the contract unit price for General Excavation (or Embankment if General Excavation is not a contract pay item). The volume to be used in the increase or decrease will be a rectangular solid the length of the pipe or box culvert, the outside width of the pipe or box culvert plus 3 feet, and the average change in invert elevation minus 2 feet.

203.14.2 General Excavation, Embankment and Nonplastic Embankment: The measurement of quantities will be computed by the average end area method and will be that area bound by (1) the original ground line established by location (plan) cross-sections or new original cross-sections obtained by the contractor, and (2) the final theoretical pay line as shown on the plans, or established by the engineer, adjusted for field changes. New original cross-sections will be taken after clearing, and prior to grubbing.

The final theoretical pay line shall be derived from the profile grade, typical section and ditch grades shown in the plans, along with approved plan changes and other field changes made by the engineer. No increase in quantities will be authorized for overbuilding unless directed by the engineer.

Pay lines for surcharged embankments will be the theoretical surcharge lines shown on the plans. No measurement will be made for removing and disposing of excess surcharge materials.

When payment is made for embankment in its final position, no additional quantity will be measured due to settlement, compaction, erosion or other cause.

Excavation and embankment for crossovers, turnouts, driveway approaches or other minor installations will not be included in the measurement.

A depth and width tolerance of  $\pm 1.5$  feet will be allowed for excavation of unsuitable material. Overdepth and overwidth will be waived at no direct pay; however, no measurement for payment will be made for additional embankment material required to backfill areas beyond theoretical unsuitable material lines.

Measurement will be made by one or more of the following methods:

**203.14.2.1 Plan Quantity:** The quantities of excavation and embankment will be those shown in the plans, provided the project is constructed essentially to the theoretical pay line.

When the plans have been revised or when disagreement exists between the contractor and the engineer as to the accuracy of the plan quantities for the entire project, or any substantial portion thereof, either party may require that quantities be revised. The party requesting the revision will be responsible for isolating and detailing the error in an easily understood format which may include cross-sections, sketches, and computations. The revision will be verified and agreed to by the other party. Quantity revisions will not be considered without advanced notice to both parties and unless the original cross-sections have been taken.

No payment will be made to the contractor to re-compute new plan quantities.

**203.14.2.2 Field Cross-Sections:** When payment lines are not shown on the plans and cannot be established, in lieu of final theoretical pay lines, field cross-sections will be used to determine pay quantities for excavation and embankment.

After clearing operations, the contractor shall take original cross-sections for the entire length of the project. Take all original cross-sections in the presence of the Department. Take cross-sections at sufficient intervals to accurately determine earthwork quantities, not to exceed 100 linear feet. Take the cross-sections in accordance with DOTD procedures, and furnish results to the Department immediately in a format satisfactory to the engineer. The Department reserves the right to take additional cross-sections as needed to verify the contractor's cross-sections. In the event the cross-sections do not verify, the contractor shall investigate and reconcile any differences.

The original cross-sections will be used to determine the accuracy of the location cross-sections by using random sections not farther apart than 1000 linear feet and centerline elevations at intervals of 100 linear feet. The location cross-sections will be considered to be

usable if the average of the differentials does not exceed  $\pm 0.3$  foot. For significant portions of the project with obvious errors between location and original cross-sections, the contractor's original field cross-sections will be used, and will not be part of the verification process. In all cases where location sections are unavailable, new originals are to be taken and used.

**203.14.3 Hydraulic Excavation:** After completion of drainage excavation operations at each individual location, measurement will be made in accordance with 203.14.2.1 or 203.14.2.2. Elevations for underwater excavation will be determined in accordance with DOTD TR 426.

Cleaning existing ditches will be measured by the linear foot along the center line of each ditch.

203.14.4 Settlement Plate Installation and Monitoring: Settlement plate installation and monitoring will be measured per each, which includes furnishing, installing, monitoring and removing; and includes all labor, materials, equipment, tools, and incidentals necessary to complete the work.

#### 203.14.5 Excavation and Embankment

- 203.14.5.1 Linear Measurement: When excavation or embankment is to be measured on a linear basis, measure the length along the centerline or the baseline used in the plans and include performing the excavation, embankment and grading work necessary for construction of the project. It is the contractor's responsibility to determine quantities of earthwork necessary to complete this item.
- **203.14.5.2** Lump Sum Measurement: When excavation and embankment is to be measured by the lump sum, this item includes performing the excavation, embankment, and grading work necessary for construction of the project. It is the contractor's responsibility to determine the correct quantities of earthwork required to complete this item. No adjustment in contract price will be made.
- **203.14.6 Borrow** (Vehicular Measurement): The material will be measured by the cubic yard in approved hauling vehicles at the point of delivery in accordance with 109.01.
- **203.14.7 Geotextile Fabric:** Geotextile fabric will be measured by the square yard of covered area in place.
- **203.15 PAYMENT.** Payment for the accepted quantities will be made at the contract unit prices, which includes furnishing the equipment, labor and materials necessary to complete the items.

Payment for roadway obliteration will be made under appropriate roadway removal and excavation items. Removal of existing asphalt pavement asphalt will be paid for under Section 202. Blading and shaping to drain will be considered incidental and will not be measured for pay. Excavation, other than blading and shaping, generally over 1 foot in depth over a substantial area, will be paid as general excavation for the full depth of cut.

Payment for undercut will be as general excavation, and payment for required backfill will be made as embankment.

Plastic soil blanket and topsoil will be included in the pay volume for the embankment. Payment for the removal and stockpiling of existing topsoil will be by the in-situ square yard.

No direct payment will be made for acquisition of borrow materials outside the right-ofway; acquisition of right-of-way and constructing haul roads; stockpiling and re-handling of materials; precautionary measures to protect private property and utilities; or furnishing necessary water and watering equipment. Excavation for plastic soil blanket in cut sections, when required, will be made as general excavation and payment for the required plastic soil blanket will be made as embankment.

Payment for cleaning existing ditches will be made at the contract unit price per linear foot, which includes removal of obstructions, furnishing and placing required backfill material, and disposing of removed material.

Payment for settlement plate installation and monitoring will be made at the contract unit price per each. If additional surcharge material is required it will be paid for as embankment. Compensation for extension of the monitoring period may be allowed in accordance with 109.04.

Payment will be made under:

| Item No. | Pay Item                                     | Pay Unit    |
|----------|--|-------------|
| 203-01   | General Excavation                           | Cubic Yard  |
| 203-02   | Drainage Excavation                          | Cubic Yard  |
| 203-03   | Embankment                                   | Cubic Yard  |
| 203-04   | Nonplastic Embankment                        | Cubic Yard  |
| 203-05   | Excavation and Embankment                    | Lump Sum    |
| 203-06   | Excavation and Embankment                    | Linear Foot |
| 203-07   | Borrow (Vehicular Measurement)               | Cubic Yard  |
| 203-08   | Geotextile Fabric                            | Square Yard |
| 203-09   | Removal and Stockpiling of Existing          |             |
|          | Topsoil                                      | Square Yard |
| 203-10   | Cleaning Existing Ditches                    | Linear Foot |
| 203-11   | Settlement Plate Installation and Monitoring | Each        |

This page intentionally left blank

## **TECHNICAL PROVISIONS**

## **SECTION 302**

## **CLASS II BASE COURSE**

## TABLE OF CONTENTS

| 302.01 | DESCRIPTION                          |
|--------|--------------------------------------|
| 302.02 | MATERIALS                            |
| 302.03 | EQUIPMENT                            |
| 302.04 | GENERAL CONSTRUCTION REQUIREMENTS    |
| 302.05 | MIXING                               |
| 302.06 | TRANSPORTING AND PLACING ON SUBGRADE |
| 302.07 | COMPACTING AND FINISHING             |
| 302.08 | QUALITY CONTROL ROADWAY OPERATIONS   |
| 302.09 | PROTECTION AND CURING                |
| 302.10 | MAINTENANCE OF BASE COURSE           |
| 302.11 | WEATHER LIMITATIONS                  |
| 302.12 | ACCEPTANCE REQUIREMENTS              |
| 302.13 | MEASUREMENT                          |
| 302.14 | PAYMENT                              |

This page intentionally left blank

## TECHNICAL PROVISIONS SECTION 302 CLASS II BASE COURSE

**302.01 DESCRIPTION.** Furnish and place Class II roadway and shoulder base course on a prepared surface in accordance with these specifications, in conformity with the lines, grades, thickness, and typical sections shown on the plans or established. Control the selection, placement, mixing and compaction of materials so that the completed base course is uniform and conforms to plan dimensions and other acceptance requirements.

Quality assurance requirements shall be as specified in the latest edition of the Department's publication titled *Application of Quality Assurance Specifications for Embankment and Base Course.* 

When not specified, any of the following types may be used:

- 1. Soil Cement
- 2. Crushed Stone
- 3. Asphalt Concrete Base Course on Embankment Layer
- 4. Recycled Portland Cement Concrete
- 5. Blended Calcium Sulfate

Unless approved otherwise in writing, use the same base course material throughout the project in accordance with these specifications.

In areas that are inaccessible for mixing and compacting, in turnouts, crossovers, and in other isolated or irregular areas, portland cement concrete complying with Section 901 or asphalt concrete base course complying with Section 502 may be used in lieu of the specified Class II base course material with approval. If using asphalt or portland cement concrete, the top half of the base course thickness shall be asphalt or portland cement concrete. If used, portland cement concrete shall be a minimum thickness of 6 inches. The remaining thickness shall be the same type and construction as the top layer of embankment, treated layer, or subgrade. Do not place raw, untreated material between a treated layer and the concrete. Place, consolidate, finish, and cure concrete as directed in accordance with Section 706.

Submit a dust control plan to address weather, sight clearance, operational procedures, traffic control, and any other project specific concerns. Failure to maintain sight clearance will result in the engineer stopping contractor operations.

The Department will identify dust sensitive areas in the plans. In these specific areas, the dust control plan must also include environmental requirements. In order to meet air quality standards, the contractor may be required to use central plant mixing of cement treated mixtures in dust sensitive areas at no direct pay. The contractor may use other types of Class II base course in dust sensitive areas at no direct pay.

**302.02 MATERIALS.** Materials shall comply with the following sections or subsections and requirements.

Geotextile Fabric 203.11 & 1019

Asphalt Concrete 502
Portland Cement Concrete 901
Portland Cement 1001.01
Blended Hydraulic Cement 1001.02
Asphalt Materials 1002

 Stone
 1003.01 & 1003.03.1

 Recycled Portland Cement Concrete
 1003.01 & 1003.03.2

 Blended Calcium Sulfate
 1003.01 & 1003.03.3

Water 1018.01

**302.02.1 Soils for Soil Cement:** Soils for soil cement base course shall consist of materials that will stabilize with cement in accordance with DOTD TR 432. Such materials are those soils classified as A-1-a, A-1-b, A-2-4, A-2-6, A-4, and A-6 in accordance with DOTD TR 423. Do not use soil with a Liquid Limit greater than 35, a Plasticity Index (PI) greater than 15, or an organic content greater than 2 percent.

Determine Liquid Limit and Plasticity Index in accordance with DOTD TR 428. Determine organic content in accordance with DOTD TR 413. Do not use soil with over 79 percent sand or 60 percent silt when tested in accordance with DOTD TR 407. Soils may be blended to adjust the percentages of sand or silt to meet specification requirements; however, in-place blending is not allowed. Do not blend or treat soils that do not meet Liquid Limit or PI requirements to reduce Liquid Limit or PI. Do not use topsoil. Obtain the material to be stabilized from outside right-of-way limits except as provided in 106.02.3. The engineer will take samples from the roadway or stockpile in accordance with the Material Sampling Manual. The District Laboratory Engineer will approve materials prior to blending and the final product.

Acceptance of soils with organic contents between 2 to 5 percent may be allowed based on determination of increased cement percentages in accordance with DOTD TR 432 Method B or C, whichever is applicable, using the design compressive strength criteria listed for stabilization. Maximum cement rate allowed will be 14 percent by volume. Perform the laboratory testing specified above at no expense to the Department. The laboratory used must be approved by the Materials Engineer Administrator.

Take samples in the presence of the engineer in accordance with the Material Sampling Manual. Obtain sufficient material to provide the District laboratory with approximately 200 pounds of the base material to be treated for verification testing. The engineer will take immediate possession of the verification samples. Also provide approximately 10 pounds of the selected cementitious material to the District laboratory. Provide materials for verification testing at no cost to the Department.

Submit all design data used to determine the recommended cement rate to the District Laboratory Engineer for approval. Prior to approval of the design, the District Laboratory

will perform verification testing. Verification testing by the District Laboratory will consist of molding, curing and testing a minimum of three specimens in accordance with TR 432, at the percentage of cementitious material and at the optimum moisture determined by the contractor's laboratory. The recommended cement rate will be considered verified if test results indicate that the minimum strength criteria have been met and that the optimum moisture are within 2 percent of that submitted by the contractor. Normal testing time for verification testing may require up to 21 calendar days. Do not begin construction operations until the design is approved.

- **302.02.2 Portland Cement**: Use Type I or II portland cement. The quantity of cement used shall be supported by proof of delivery.
- **302.02.3 Blended Hydraulic Cement**: The cement shall be Type IP. The quantity of cement used shall be supported by proof of delivery.
- **302.02.4 Asphalt Concrete Base Course:** The material requirements for asphalt concrete base course shall be as described in Section 502. The top half of the base thickness shall be asphalt concrete and the remaining thickness shall be the same type and construction as the top layer of embankment, treated layer, or subgrade. Do not place raw, untreated material between a treated layer and the asphalt concrete.
- **302.02.5 Blended Calcium Sulfate:** Take gradation samples in accordance with 1003.03.3 from the dedicated stockpiles at the point of material origin.
- **302.03 EQUIPMENT.** Obtain approval of equipment prior to use. When using in-place mixing, the equipment shall conform to 303.03. When using central mixing, the equipment shall conform to 301.03.1. Compaction equipment shall conform to 301.03.1.5.
- **302.04 GENERAL CONSTRUCTION REQUIREMENTS.** Place base course material on a subgrade prepared in accordance with Sections 203, 304, 305, 306 as specified. Construct asphalt concrete base course in accordance with Section 502. Do not use blended calcium sulfate in areas needed to facilitate traffic control. Do not place blended calcium sulfate within 10 feet of metal drainage structures. The contractor will be allowed to substitute any untreated Class II base course material listed in 302.01.

Finished lift thickness shall be 9 inches maximum. The Department may allow single lift construction for depths exceeding 9 inches and up to 12 inches based on a rolling pattern that obtains the required density.

Use a Class D geotextile separator fabric if an aggregate base course is to be placed on untreated or lime-treated soils.

#### 302.05 MIXING.

**302.05.1 Soil Cement:** Combine soil with cement and water by in-place mixing or in a central plant and shape on the subgrade. When in-place mixing is done, spread and mix the cement prior to adding any more water.

A minimum of 70 percent of the pulverized soil, as determined by DOTD TR 431, shall pass the No. 4 sieve after mixing. Determine the optimum moisture of the mixture in accordance with DOTD TR 415 or TR 418. The percentage of moisture in the mixture, by dry weight, shall not vary from the optimum moisture by more than  $\pm 2.0$  percent at the time of compaction when tested in accordance with DOTD TR 403.

**302.05.1.1 In-Place Mixing:** Samples to determine optimum moisture and maximum dry density will be taken by the project engineer.

Determine maximum dry density in accordance with DOTD TR 415 or TR 418 and inplace density in accordance with DOTD TR 401. After placement of soil and prior to mixing with cement, shape the soil to required section and compact to at least 93.0 percent of maximum dry density at the required grade.

From materials sampled in-place on the project, the engineer will determine the percentage of cement in accordance with DOTD TR 432 prior to mixing. Depending on the type of cement and soil to be used, normal testing time to determine required cement content may require 21 calendar days. Add water as needed to bring the moisture content of the mixture within the tolerance and uniformly mix with the materials. During the mixing process, add water only through the spray bar of the in-place mixer which is adjusted to provide uniform coverage across the completed width of the roadway for the full depth of the base. Do not allow wet streaks or spots.

The method of cement distribution shall be such that the amount of cement used can be readily determined. Determine the spread rate of cement in accordance with DOTD TR 436.

When the moisture content is not within  $\pm 2.0$  percent of optimum, discontinue operations and do not resume until the moisture content is controlled within this tolerance. Do not place and pulverize more than one transport until moisture content is within  $\pm 2.0$  percent of optimum.

**302.05.1.2** Central Plant Mixing: Mixing in a central mix plant shall conform to Section 301. When using central plant mixing, a reduction of 1.0 percent in the volume of cement required will be permitted.

**302.05.2** Crushed Stone and Recycled Portland Cement Concrete: Do not allow crushed stone or recycled portland cement concrete base courses to segregate during construction. Take gradation samples in accordance with 1003.03 from the dedicated stockpiles at the point of material origin.

**302.05.3 Blended Calcium Sulfate:** Do not use for crossovers, drives, or in areas needed to facilitate traffic control. In lieu of blended calcium sulfate, substitute any untreated Class II base course material listed in 302.01. Do not place blended calcium sulfate within 10 feet of metal drainage structures. Use approved backfill material in Section 701.

**302.06 TRANSPORTING AND PLACING ON SUBGRADE.** Use only transportation and spreading methods that do not damage the subgrade. Place and spread sufficient base course material to obtain required width and compacted thickness within the tolerances set forth in 302.12. Do not allow subgrade material to contaminate the base course. Any contamination will require retesting and correction of deficiencies. Do not place, spread, or mix base course material on portland cement concrete or asphalt concrete pavements. Do not allow base course construction operations to damage adjacent pavement surfaces, edges and joints. Add water or use other suitable means to prevent dust during the transporting and placing of materials.

#### 302.07 COMPACTING AND FINISHING.

**302.07.1 General:** The finished base course shall have a smooth, uniform, closely knit surface, free from ridges, waves, laminations or loose material. Thoroughly roll the surface and finish to grade. The cross-slope shall not vary by more than  $\pm 0.003$  foot/foot. Density requirement shall be in accordance with 302.12. Do not damage the subgrade layer during compaction operations.

**302.07.2 Soil Cement:** When using central plant mixing, compact the material and finish in accordance with 301.10, except that the automatic grade machine will not be required. When using in-place mixing, compact the material and finish in accordance with 303.06.

Begin mixing operations within one hour of placement. Complete compaction and finishing operations within three hours after initial mixing of cement with base course materials. Upon expiration of the three-hour period after initial mixing, only intermediate finishing (tight blading) of the base course surface will be allowed. Dispose of excess bladed material. Do not drift bladed material along the base. Use stabilized material in the base course except for that small amount necessary for tight blading. Excessive blading, exceeding 10 percent of the base thickness, to achieve plan depth will not be allowed. Complete operations, including tight blading, within 24 hours of mixing. The finished base course shall have a smooth, uniform, closely-knit surface, free from ridges, waves, laminations, or loose materials. Do not spread cement within 2 hours of sunset, unless otherwise approved by the project engineer.

**302.07.3 Crushed Stone and Recycled Portland Cement Concrete:** Compact these materials using an approved sheepsfoot-type roller and finish-roll with an approved pneumatic tire roller or a smooth steel wheel roller. Keep the surface uniformly moist during compaction and final finishing.

**302.07.4 Asphalt Concrete:** Compact and finish asphalt concrete in accordance with Section 502. The soil layer shall be compacted and finished in accordance with the top layer of embankment or subgrade.

**302.07.5 Blended Calcium Sulfate:** During placement of blended calcium sulfate, the percentage of moisture in the mixture, by dry weight, shall not vary from the optimum moisture by more than ±2.0 percent. After application of water, allow the moisture to reach equilibrium in the base before applying rolling techniques. Roll blended calcium sulfate to the edge of the embankment or subgrade. Compact each layer to at least 95 percent of maximum dry density.

Determine optimum moisture and maximum density in accordance with DOTD TR 418 Method G modified to include a maximum drying temperature of 140°F.

Proof roll by using a load of 25 tons in a 12 to 14 cubic yard tandem dump truck with ten wheels or approved loaded truck determined by the project engineer. Proof rolling shall be a minimum of 5 passes in each direction at the same locations and at a maximum vehicle speed of 3 mph.

Test all blended calcium sulfate base by proof rolling immediately prior to placement of surfacing material, including asphalt binder. Correct any irregularities or soft spots prior to placement of the surfacing material. Any rain event on the project site between the proof rolling and placement of the surfacing will require an additional proof rolling as noted above.

**302.08 QUALITY CONTROL OF ROADWAY OPERATIONS.** Control the selection, placement, compaction, cement spread, mixing, moisture content, density, thickness, width, surface finish, cross-slope, and grade to produce a completed base course that is uniform and conforms to plan dimensions and other acceptance requirements as provided herein. Control operations to prevent contamination, segregation, soft spots, wet spots, laminations, and other deficiencies. Perform tests necessary to adequately control the work.

#### 302.09 PROTECTION AND CURING.

**302.09.1 Soil Cement:** Upon completion of intermediate finishing, immediately protect the base course against drying by applying an asphalt curing membrane in accordance with Section 506. Place asphalt curing membrane on the same day as treatment. Maintain complete coverage of the curing membrane from the initial application until the placement of the next course. When allowing traffic, including construction equipment, on the base course, place at least the first lift of surfacing within 30 calendar days unless otherwise directed.

302.09.2 Crushed Stone, Recycled Portland Cement Concrete, Soil Layer Under Asphalt Concrete, and Blended Calcium Sulfate: Cover the base course with asphalt prime coat in accordance with Section 505 as soon as practical to avoid water infiltration due to rainfall. Maintain complete coverage of asphalt prime coat from the initial application until the placement of the next course.

302.10 MAINTENANCE OF BASE COURSE. Protect the base course from damage from public traffic or the contractor's operations and satisfactorily maintain the base course, including the asphalt curing membrane or prime coat. Repair damaged base course at no direct pay. When requiring patching of the base course, in addition to removing damaged or unsound base course, remove a sufficient width and depth of sound base course to ensure satisfactory placement of patching material. The engineer's approval of the type of patching material will be required before use. Patching or other base course repair shall restore a uniform surface, shall conform to the requirements of the material being used, and shall be completed before paving operations begin. Patch any failures detected during paving.

Do not allow public traffic or construction traffic on the completed base course during the 72-hour curing period. If conditions permit, route both public traffic and construction traffic off the completed base course onto shoulders or other suitable areas during the 72-hour curing period. Traffic may be permitted on the base course during the curing period if conditions warrant and approved by the engineer. When permitting traffic to use the completed base course subsequent to the 72-hour curing period and prior to construction of the surface course, further protect the base by additional applications of asphalt curing membrane or prime coat in accordance with 301.12 at no direct pay.

Prior to surface course construction, correct deficiencies and weak spots, clean the base course surface, repair any damages caused by traffic, and keep the surface true to grade and cross section at no direct pay. Apply and maintain additional asphalt curing membrane or prime coat as directed at no direct pay.

When surfacing with asphalt concrete, place the first lift of surfacing within 30 calendar days.

**302.11 WEATHER LIMITATIONS.** Do not construct base course when the subgrade or stockpiles are frozen, when raining, when the ambient air temperature is below 35°F, in the case of cement treated bases, or the temperature forecasted by the U.S. Weather Service is to be 25°f or less within the 24-hour period following placement.

**302.12 ACCEPTANCE REQUIREMENTS.** Soils and aggregates will be sampled for acceptance by the Department in accordance with the Materials Sampling Manual.

For central plant mixing, determine the cement content in accordance with 301.16. For in-place mixing, determine the cement content in accordance with 302.05. Test the moisture content of the soil cement or cement treated mixtures for conformance to optimum moisture content in accordance with DOTD TR 403.

Test the pulverization of the soil cement or cement treated mixtures in accordance with DOTD TR 431. At least 70 percent shall pass the No. 4 sieve.

Check base course, except asphalt concrete, for acceptance in increments of 1000 linear feet per roadway or 2000 linear feet per shoulder constructed separately. Asphalt concrete acceptance will be in accordance with Section 502.

**302.12.1 Density Requirements:** Upon completion of compaction operations, determine base course in-place density, except asphalt concrete, in accordance with DOTD TR 401. Determine density requirements for asphalt concrete base course in accordance with Section 502.

The density requirements for Class II base course materials other than asphalt concrete shall be a minimum of 95.0 percent maximum dry density in accordance with DOTD TR 418.

**302.12.1.1 Soil Cement and Treated Layer Under Asphalt Concrete:** When the density test value for the section falls below 95.0 percent, a payment adjustment will be applied in accordance with Table 302-1 as follows.

Table 302-1
Density Acceptance and Payment Schedule

| Density Test Value (percent) | Percent of Contract Unit Price |
|------------------------------|--------------------------------|
| 95.0 & Above                 | 100                            |
| 94.0 to 94.9                 | 90                             |
| 93.0 to 93.9                 | 75                             |
| Below 93.0                   | 50 or Remove <sup>1</sup>      |

At the option of the Chief Engineer.

302.12.1.2 Crushed Stone, Recycled Portland Cement Concrete, Blended Calcium Sulfate, and Soil Layer under Asphalt Concrete Base Course: When any test value is less than the required density, continue compaction until obtaining the specified density.

The acceptance requirements for blended calcium sulfate base course shall be the same as stone base course with the following modifications. Upon completion of compaction operations, determine the density in accordance with DOTD TR 401 except that all moisture content determinations for density calculations shall be conducted by oven drying the material for 24 hours at 140°F. A forced draft type oven capable of maintaining the temperature shall be provided by the contractor for field moisture content determination for density control.

**302.12.2** Thickness Requirements: Determine the thickness of the completed base course in accordance with DOTD TR 602.

Do not allow the completed base course to vary from plan thickness in excess of the tolerances in Table 302-2 below. Correct base course thickness deficiencies in excess of these tolerances at no direct pay.

Table 302-2
Base Course Thickness Tolerance

| (All Bases Except Asphalt Concrete) Under- | (Stabilized & Treated Bases) Over- |  |
|--|------------------------------------|--|
| Thickness, Inches                          | Thickness, Inches                  |  |
| 3/4  | $1^{1}/2$                          |  |

When using crushed stone base or recycled concrete base over soil cement base, the individual base layer tolerances shall be in accordance with Table 302-2, and the total base course under-thickness shall not exceed 1 inch.

Any failing area will be isolated for purposes of correction.

Determine asphalt concrete base thickness in accordance with Section 502.

When using central plant mixing, over-thickness may be waived at no direct pay.

**302.12.2.1 Soil Cement, and Treated Layer Under Asphalt Concrete:** When not permitting grade adjustments, correct under-thickness deficiencies in excess of tolerance by

removing and replacing the full depth of base course in deficient areas with one of the following materials:

- 1. The same type of base course.
- 2. Asphalt concrete complying with Section 502.
- 3. Concrete complying with Section 901.

When permitting grade adjustments, correct thickness deficiencies either by furnishing and placing a supplemental layer of asphalt concrete complying with Section 502 for the full width of base course or by removing and replacing deficient base course. When approved, corrections may be made by re-stabilizing the existing material in accordance with this section, and the cement content may be reduced from design contents with approval of the District Laboratory Engineer.

Thickness of the supplemental layer of asphalt concrete shall be in accordance with Table 302-3 as follows.

Table 302-3
Supplemental Asphalt Concrete Layer Thickness

| Under-Thickness, Inch   | In-Place Mixing Over-Thickness, | Minimum Thickness of Supplemental Asphalt |  |
|-------------------------|---------------------------------|---|--|
| mon                     | Inches                          | Concrete, Inch <sup>1</sup>               |  |
| 1 to 1 <sup>1</sup> /4  | $1^{3}/4$ to 2                  | $1^{1}/4$                                 |  |
| $1^{1}/2$ to $1^{3}/4$  | $2^{1}/4$ to $2^{1}/2$          | 1 <sup>1</sup> /2                         |  |
| $2 \text{ to } 2^{1}/2$ | $2^{3}/4$ to 3                  | 2   |  |
| Over 2 <sup>1</sup> /2  | Over 3 Remove and Repl          |   |  |

May be included in the subsequent lift.

When using reconstruction as the method of correction, the above tolerances shall apply.

**302.12.2.2** Crushed Stone, Blended Calcium Sulfate, and Recycled Portland Cement Concrete: When allowing grade adjustments, correct under-thickness in excess of 3/4 inch to plan thickness by furnishing, placing, reworking, shaping, and compacting additional base course material as required. When not allowing grade adjustments, remove the material and replace at no direct pay.

**302.12.2.3 Asphalt Concrete Base Course:** When not allowing grade adjustments, correct under-thickness in excess of the tolerances given in 502.12 to plan thickness by removing and replacing the full depth of base course. When allowing grade adjustments, correct under-thickness in excess of the tolerances given in 502.12 to plan thickness by placing and compacting an  $1^{1}/4$ -inch thick minimum supplemental layer of asphalt concrete complying with Section 502 at no direct pay.

**302.12.3 Width Requirements:** Determine the width of the completed base course in accordance with DOTD TR 602. Do not allow roadway base course width to vary from plan width in excess of +6 inches. Do not allow shoulder base course width to vary from plan width in excess of +3 inches. No under-widths are allowed for shoulder or roadway

At the option of the Department after investigation.

bases. When the base course for both roadway and shoulders are constructed at the same time, the 6-inch tolerance will be applied. Correct base course width deficiencies in excess of the above tolerances as follows at no expense to the Department:

### 302.12.3.1 Soil Cement and Asphalt Concrete Base Course

- **302.12.3.1.1 Over-Width:** Over-widths of asphalt concrete and treated base courses mixed in a central plant may be waived at no additional cost to the Department. When not allowing grade adjustments, remove the full depth and width of base course in areas having over-widths in excess of the foregoing tolerances and replace to the plan width with one of the following materials:
  - 1. The same type of base course.
  - 2. Asphalt concrete complying with Section 502.
  - 3. Concrete complying with Section 901.

In lieu of removing and replacing the over-width areas of base course, at the Department's option, any base course less than 12 inches over-width will be allowed to remain in place at an adjusted payment of 90 percent of the contract unit price for the complete section. Remove over-width in excess of 12 inches and replace as indicated above. When approved, corrections may be made by restabilizing the existing material in accordance with this subsection, and the cement content may be reduced from design contents with approval of the District Laboratory Engineer.

When permitting grade adjustments, correct base course width deficiencies by removing and replacing as specified above, or by furnishing and placing a 1<sup>1</sup>/4-inch thick supplemental layer of asphalt concrete complying with Section 502 on the 1000-foot section for the full width of the base course.

- 302.12.3.1.2 Under-Width: Correct under-widths of base course in excess of the foregoing tolerances to plan width and thickness by furnishing and placing additional materials; however, the width of widening materials shall be not less than 12 inches. When approved, corrections may be made by restabilizing the existing material in accordance with this section, and the cement content may be reduced from design contents with approval of the District Laboratory Engineer. Materials for widening deficient base course may be asphalt concrete complying with Section 502 or concrete complying with Section 901.
- 302.12.3.2 Crushed Stone, Blended Calcium Sulfate, and Recycled Portland Cement Concrete: Over-widths will be waived at no additional cost to the Department. Correct under-widths in excess of the foregoing tolerances to plan widths by furnishing, placing, reworking, shaping, and compacting additional base course material as required.
- **302.12.4 Grade and Cross-Slope**: The finished grade shall be within  $\pm 1/2$  inch of the established grade. Do not allow the cross-slope to vary by more than  $\pm 0.003$  foot/foot.
- **302.12.5** Correction Deficiencies: Correct deficiencies in surface finish, cross-slope, grade, contamination, segregation, soft spots, wet spots, laminations, and other deficiencies at no direct pay. Correct deficiencies by removing and replacing or as directed.
- **302.13 MEASUREMENT.** The quantities of Class II base course for payment will be the design volumes or areas specified in the plans and adjustments thereto. Design quantities are based on the horizontal dimensions and compacted thickness of the completed base course shown

on the plans. Design quantities will be adjusted if the engineer makes changes to adjust to field conditions, if plan errors are proven, or if design changes are necessary.

Geotextile fabric used beneath the base course will not be measured for payment.

**302.14 PAYMENT.** Payment for Class II base course will be made at the contract unit price, adjusted as specified in 302.12 and the following provisions, which includes furnishing and placing required base course materials, portland cement, blended hydraulic cement, water, asphalt curing membrane, and prime coat.

Any payment adjustment in asphalt concrete shall be in accordance with Section 502 and shall apply to the cubic yard total quantity of base course when payment is by cubic yard. For other materials, when making payment adjustments for more than one deficiency, they shall be cumulative.

Payment for geotextile fabric will be included in the contract unit price for base course. Payment will be made under:

| Item No. | em No. Pay Item      |          | Pay Unit    |
|----------|----------------------|----------|-------------|
| 302-01   | Class II Base Course |          | Cubic Yard  |
| 302-02   | Class II Base Course | in Thick | Square Yard |

This page intentionally left blank

## **TECHNICAL PROVISIONS**

## **SECTION 502**

## ASPHALT CONCRETE MIXTURES

## TABLE OF CONTENTS

| 502.01 | DESCRIPTION  |
|--------|--|
| 502.02 | MATERIALS  |
| 502.03 | DESIGN OF ASPHALT MIXTURES AND JOB MIX FORMULA (JMF) APPROVAL                                |
| 502.04 | JOB MIX FORMULA VALIDATION AND APPROVAL  |
| 502.05 | QUALITY CONTROL AND PLANT ACCEPTANCE   |
| 502.06 | PLANT INSPECTION AND AUDITS  |
| 502.07 | ROADWAY OPERATIONS   |
| 502.08 | HAULING, PAVE AND FINISHING  |
| 502.09 | ROLLING AND COMPACTION   |
| 502.10 | ROADWAY LOT SIZES  |
| 502.11 | ROADWAY ACCEPTANCE   |
| 502.12 | SURFACE TOLERANCE EQUIPMENT, QUALITY CONTROL, ACCEPTANCE, MEASUREMENT AND PAYMENT ADJUSTMENT |
| 502.13 | DIMENSIONAL REQUIREMENTS   |
| 502.14 | MEASUREMENT  |
| 502.15 | PAYMENT  |

This page intentionally left blank

## TECHNICAL PROVISONS SECTION 502 ASPHALT CONCRETE MIXTURES

#### **502.01 DESCRIPTION.**

**502.01.1 General:** Furnish and construct asphalt concrete mixtures in accordance with Table 502-6 and in conformance with the lines, grades, thicknesses, and typical sections in the plans.

Comply with Section 503, Equipment and Processes and the Application of Quality Assurance Specifications for Asphalt Concrete Mixtures (QA Manual).

Use a DOTD certified laboratory accredited by AMRL, CMEC, or other accreditation agency approved by DOTD.

**502.01.2 Lift Description and Mixture Types:** The wearing course is defined as the final lift placed. The binder course is defined as the lift placed prior to the final lift as defined in the plans. When a Section 501 thin lift mix is used in conjunction with construction of 502 mixtures, it is a finish course.

Mainline mixtures include wearing, binder and base courses for travel lane, ramps greater than 300 feet, interstate acceleration/deceleration lanes, turn lanes, and the two center lanes for airports.

Minor mixes include mixture used for bike paths, crossovers, curbs, detour roads, driveways, guardrail widening, islands, joint repair, leveling, medians, parking lots, shoulders, turnouts, ramps less than or equal to 300 feet, patching, widening, miscellaneous handwork, and any other mixture that is not mainline.

**502.02 MATERIALS.** Comply with applicable Part X subsections listed herein. Sample in accordance with the Materials Sampling Manual and ensure testing in accordance with the procedures listed in Part X and Table 502-1. Keep accurate records, including proof of deliveries of all materials used in asphalt concrete mixtures. Furnish copies of these records to the engineer upon request.

| Aggregates                       | 1003.01 & 1003.06   |
|----------------------------------|---------------------|
| Anti-Strip Additives             | 1002.02             |
| Asphalt                          | 1002                |
| Crumb Rubber                     | 1002.02.2           |
| Hydrated Lime                    | 1018.02             |
| Mineral Fiber                    | 1002.02.5           |
| Mineral Filler                   | 1003.06.6           |
| Mix Release Agent                | 1018.10             |
| Reclaimed Asphalt Pavement (RAP) | 1003.01 & 1003.06.5 |
| Warm Mix Additives               | 1002.02.4           |

Table 502-1
Test Procedures for Asphalt Concrete

| Description  | Test Method       |  |
|--|-------------------|--|
| Specific Gravity and Density of Compressed Asphalt | DOTD TR 304       |  |
| Mixtures   |                   |  |
| Theoretical Maximum Specific Gravity, Gmm          | DOTD TR 327       |  |
| Asphalt Cement Content, Pb                         | DOTD TR 323       |  |
| Mechanical Analysis of Extracted Aggregate         | DOTD TR 309       |  |
| Moisture Content of Loose HMA                      | DOTD TR 319       |  |
| Degree of Particle Coating (plant requirement)     | DOTD TR 328       |  |
| Bulk Specific Gravity and Absorption               | AASHTO T 84, T 85 |  |
| Coarse Aggregate Angularity, % Crushed             | DOTD TR 306       |  |
| (Double Faced)                                     |                   |  |
| Fine Aggregate Angularity                          | DOTD TR 121       |  |
| Flat and Elongated Particles                       | ASTM D 4791       |  |
| Sand Equivalent                                    | DOTD TR 120       |  |
| Mixture Conditioning (Aging) of HMA Mixtures       | AASHTO R 30       |  |
| Superpave Volumetric Mix Design                    | AASHTO M 323      |  |
| Preparing Gyratory Samples                         | AASHTO T 312      |  |
| Asphalt Cement Draindown                           | ASTM D 6390       |  |
| Longitudinal Profile Using Automated Profilers     | DOTD TR 644       |  |
| Thickness and Width of Base and Subbase            | DOTD TR 602       |  |
| Loaded Wheel Tester (LWT)                          | AASHTO T 324      |  |
| Semicircular Bend Test (SCB)                       | TR 330            |  |

**502.02.1 Asphalt Cement:** Comply with Table 502-2. If the asphalt cement does not comply with the requirements of Section 1002, cease mix production until proper asphalt material is supplied.

**Table 502-2** 

**Asphalt Cement Usage** 

| Location  | Mix<br>Level | Asphalt<br>Grade<br>Required | Substitutions Allowed  |                               |  |
|---|--------------|------------------------------|--|-------------------------------|--|
|   |              |                              | Lower  | Grade <sup>1</sup>            | Higher Grade                           |
| Mainline<br>Wearing<br>& Binder <sup>2,3</sup>      | 1            | PG 70-22m                    | PG 67-22<br>(Binder only)<br>with traffic volume<br>< 3500 ADT |                               | PG 82-22rm,<br>and PG 76-<br>22m       |
| Mainline<br>Wearing<br>& Binder <sup>2,3,4</sup>    | 2 and<br>SMA | PG 76-22m                    | PG 70-22m<br>with<br>Hydrated<br>Lime                          | PG 70-22m<br>(Binder<br>Only) | PG 82-22rm                             |
| Base  | 1            | PG 67-22                     | PG 58-28 <sup>5</sup>  |                               | PG 82-22rm,<br>PG 76-22m,<br>PG 70-22m |
| Minor Mixes<br>including<br>Leveling <sup>2,3</sup> | ALL          | PG 67-22                     |  |                               | PG 82-22rm,<br>PG 76-22m,<br>PG 70-22m |

Lower grade substitutions are only allowed if LWT rut depths < 6mm for the design level.

#### **502.02.2** Additives.

**502.02.2.1 Anti-Strip (AS):** Add anti-strip additive at the minimum rate of 0.6 percent by weight of asphalt cement and thoroughly mix in-line with the virgin asphalt cement at the plant. Increase the anti-strip additive or change to different additive as needed to meet Loaded Wheel Test, LWT, requirements. Discontinue production until satisfactory adjustments are made when the amount of anti-strip additive is not in accordance with the approved JMF.

**502.02.2.2 Hydrated Lime:** When used, specify rate of hydrated lime additive on the Job Mix Formula. Add hydrated lime additive at a minimum of 1.5 percent and thoroughly mix with aggregates in conformance with 503.05.5 as required to meet LWT requirements.

**502.02.2.3 Waste Tire Rubber Additive:** When used, crumb rubber may be pre-blended or, with approval by the Materials Laboratory, may be blended at the plant. The maximum rubber replacement is 10 percent by weight of asphalt.

When blending crumb rubber at the contractor's plant, add crumb rubber to a PG 67-22 material on the Approved Materials List. Add 30 mesh (or finer) crumb rubber as required to meet grade PG 82-22rm. Comply with 1002.02.2.

<sup>&</sup>lt;sup>2</sup>For single lift overlay match grade of overlay.

Semicircular bend test (SCB), minimum, Jc=0.5 KJ/m2 required for all substitutions.

<sup>&</sup>lt;sup>4</sup>Semicircular bend test (SCB), minimum, Jc=0.6 KJ/m2 required for all substitutions.

<sup>&</sup>lt;sup>5</sup>When 21-30% RAP is used, PG 58-28 is required.

- **502.02.2.4 Latex Additive:** When added at the contractor's plant, blend a minimum of 1.0 percent residual latex by weight of asphalt cement to a PG 67-22 material on the Approved Material List, and in accordance with Section 503. Meet PG 70-22m requirement using pre-qualified asphalt material and latex.
- **502.02.2.5 Warm Mix Asphalt Additives:** When used, add only approved warm mix chemical additives. Foaming is allowed.
- **502.02.3 Aggregates:** Use aggregates from approved sources. Blend aggregates to meet Sections 502 and 1003.
- **502.02.3.1 Friction Ratings:** Friction ratings for aggregates are determined in accordance with 1003.01.2.4. Table 502-3 describes the friction ratings and corresponding usage allowed for the current average daily traffic (ADT) shown on the plans. Friction rating requirements apply to the mainline wearing course only, unless a finish course is applied. If a finish course is applied, then the friction rating requirements do not apply to wearing course.

All binder and base mixes and minor mixes do not have aggregate friction rating requirements.

Table 502-3
Aggregate Friction Rating

| Friction Rating | Allowable Usage  |
|-----------------|--|
| I               | All mixtures   |
| II              | All mixtures   |
| III             | All mixtures, except mainline wearing courses with plan Average Daily Traffic (ADT) greater than 7000 <sup>1</sup> |
| IV              | All mixtures, except mainline wearing courses <sup>2</sup>   |

When plan current average daily traffic (ADT) is greater than 7000, blending of Friction Rating III aggregates and Friction Rating I and/or II aggregates will be allowed for travel lane wearing courses at the following percentages. At least 30 percent by weight (mass) of the total aggregates shall have a Friction Rating of I, or at least 50 percent by weight (mass) of the total aggregate shall have a Friction Rating of II. The frictional aggregates used to obtain the required percentages shall not have more than 10 percent passing the No. 8 (2.36 mm) sieve.

**502.02.3.2 Reclaimed Asphalt Pavement (RAP):** Keep reclaimed asphalt pavement separate from other materials at the plant in such a manner that will allow for Department inspection and acceptance. Keep stockpiles uniform and free of soil, debris, foreign matter and other contaminants. Allowable RAP percentages are defined in Table 502-6. Screen or crush RAP to pass a maximum of 2 inch sieve prior to use. Additional RAP is allowed in all mixes except for Airports and SMA when RAP stockpile is prescreened on a 1 inch scalping screen.

When the average daily traffic (ADT) is less than 2500, blending of Friction Rating IV aggregates with Friction Rating I and/or II aggregates will be allowed for travel lane wearing courses at the following percentages. At least 50 percent by weight (mass) of the total aggregate in the mixture shall have a Friction Rating of I or II. The frictional aggregates used to obtain the required percentages shall not have more than 10 percent passing the No. 8 (2.36 mm) sieve.

**502.02.3.3 Mineral Filler:** When used, comply with the requirements of 1003.06.6.

**502.02.3.4 Natural Sand:** When used, meet the requirements of Table 502-6 and 1003.06.3.

**502.02.3.5 Fibers:** When required to prevent draindown, use cellulose or mineral fiber, meeting the requirements of 1002.02.5 When used, add fibers at a rate sufficient to prevent draindown.

**502.03 DESIGN OF ASPHALT MIXTURES AND JOB MIX FORMULA** (JMF) APPROVAL. Design all asphalt mixtures for optimum asphalt content in compliance with the mix design in accordance with AASHTO M323, AASHTO M325 for SMA, and the requirements of Table 502-6 and Table 1003-14.

At minimum, all design submittals must include the recommended materials proportions, extracted gradation, recommended mix and compaction temperatures, and supporting design data. Submit the recommended JMF electronically through Site Manager Materials (SMM) or other data system as designated by the Department for District Laboratory Engineer acceptance with all supporting design data. No mixture shall be produced until the proposed JMF has been accepted.

Indicate the optimum mixing and compaction temperatures as suggested by the asphalt binder supplier on the JMF. Mix temperatures are recommended by the asphalt supplier as determined by rotational viscosity or other means. Warm Mix Asphalt technology may be used to reduce this temperature and must be noted on the JMF. Warm mix asphalt may be substituted with a minimum production temperature of 275°F.

Once a plant is producing an acceptable JMF, keep JMF production within the specified tolerances. Changes will be reviewed and accepted by the District Laboratory Engineer as necessary.

The engineer may require a new mix design when roadway acceptance requirements are not being met or plant quality data indicates non-compliance.

**502.03.1 Mixtures Design Substitutions:** Use only Warm Mix Asphalt (WMA) additives that are listed on the Approved Material List.

The 3/4-inch Nominal Maximum Size (NMS) wearing course may be substituted for binder course but not substituted for base course. The 1-inch NMS binder course may be substituted for base course.

The 1/2-inch NMS wearing course may be substituted for incidental paving, Level A. Shoulders may be any mixture type shown in Table 502-4 regardless of design level except that shoulder wearing must be a 1/2-inch or 3/4-inch NMS mixture.

Apply all specification requirements for the substituted mixture with the following exceptions: When wearing course is substituted for binder course, Table 502-3 does not apply. When wearing or binder are substituted for binder or base, the allowable RAP percentage shall meet the intended use specified in Table 502-6.

When a 501 finish course and a 502 wearing course are required on a project, allowable RAP percentage for wearing may meet binder course requirement.

## 502.04 JOB MIX FORMULA VALIDATION AND APPROVAL.

The Department and contractor will jointly test plant mix to validate each JMF for mainline mixture and accept each JMF whenever a plant begins initial operations for the Department in a specific plant location, or whenever a plant experiences a change in materials or change in source of materials, or when there are significant changes in equipment, such as the introduction of a new crusher, drum mixer, burner, foaming device, etc. Evaluate each JMF at least once every two years. Meet LWT requirements and all applicable requirements of Table 502-6.

For Minor Mixes, validation is not required for mixture designs, but the mixture must meet specification requirements. In order to validate minor mixes, the plant Gmm must be determined. The average of the first five (5) plant Gmms will become the new JMF target.

For all mixes, validation is not required when the asphalt grade has changed or asphalt source has changed, but must meet LWT requirements and all applicable requirements of Table 502-6.

- **502.04.1 Validation Plant Lot:** The validation plant lot ("VP-lot"), is a maximum of 2000 tons of plant produced mix. Divide into 5 equal parts for validation sampling and testing.
- **502.04.2 Validation:** Report the mean, standard deviation, Quality Index and percent within limits (PWL) of the test results in accordance with the QA manual. The JMF is considered conditionally validated if the following parameters are 71 percent within limits of the JMF and meet the specifications.
  - 1. Theoretical Maximum Specific Gravity (Gmm),
  - 2. Percent Gmm at Ninitial,
  - 3. Percent passing the No. 8 and No. 200 sieves,
  - 4. Percent Air Voids at Ndesign, and
  - 5. VFA.

The average of all other validation tests shall meet the specifications limits in Table 502-6. The production can continue during conditional validation. The JMF is considered validated with passing LWT results. If the LWT fails twice, cease production and re-design. Upon validation of the JMF, the average of the validated results will become the JMF targets.

**502.04.3 Payment for Plant Validation:** Payment will be in accordance with 502.15.

The validation mixture is not paid separately, but is considered part of the roadway lot.

**502.05 QUALITY CONTROL AND PLANT ACCEPTANCE.** All quality control information, plant records, etc. will be considered part of the Department's acceptance decision. Exercise quality control over all materials and their assembly, design, processing, production, hauling, laydown and associated equipment to ensure compliance with Table 502-4 and all other specifications herein. At the end of each production day, notify the District Lab Engineer (DLE) and the DOTD Asphalt District Inspector (ADI) of the next scheduled mix production run and placement.

For plant quality control, a plant lot, or "P-Lot' is defined as 1000 tons of continuously produced mixture from one JMF. Obtain a sample of plant mixture and test the mixture once every 1000 tons using a random sampling approach. Minimum quality control testing for each P-Lot is as follows:

## Loose Mix

- 1. Theoretical Maximum Specific Gravity, Gmm
- 2. % Asphalt Cement Content
- 3. Gradation
- 4. % Crushed
- 5. Temperature, and
- 6. % Moisture content

## Compacted Specimen, Ndesign

- 1. % Gmm at Ninitial
- 2. % Air Voids, V
- 3. % VMA
- 4. % VFA, and
- 5. % Gmm at Nmax (1 per 5 P-Lots)

Age all loose mix tested for G<sub>mm</sub> or volumetrics for one hour in accordance with AASHTO R30 prior to testing. Age warm mix for two hours.

Determine the rolling five test results average and standard deviation for aggregate gradation, asphalt content, air voids, and G<sub>mm</sub>. Take corrective action or cease production when the latest rolling five test results show:

- 1. Air voids or Gmm fall below 71 PWL (based on the lastest rolling five test results); or
- 2. Average VFA is outside of specification limits; or
- 3. Gradation for the No. 8 and No. 200 sieve is outside of specification limits; or
- 4. Asphalt content is  $\pm 0.2\%$  the JMF target.

Enter all plant quality control data into the Department's approved data management system. The full range of gradation mix tolerances will be allowed even if they fall outside the control points. The District Laboratory Engineer may require re-validation of the mix when the average of the Quality Control data indicates non-compliance with the specified limits or tolerances.

Measure the moisture content of the cold feed aggregates daily in accordance with DOTD TR 403. The moisture content of the final mixture, measured daily, shall not exceed 0.3 percent by weight (mass) when tested in accordance with DOTD TR 319.

**502.06 PLANT INSPECTION AND AUDITS.** All Department inspection procedures, including sampling and testing, and the contractor's quality control data form the basis for acceptance of the asphalt. The Department's Certified

Asphalt Plant Inspector will randomly visit and inspect asphalt plants, sample and test material, and review documentation to ensure conformance to specification requirements. In particular, the inspector will take a minimum of the following samples which may be tested for verification:

## Loose Mix

- 1. Theoretical Maximum Specific Gravity, G<sub>mm</sub>,
- 2. % Asphalt Cement Content,
- 3. Gradation, and
- 4. % Crushed

## Compacted Specimen, N<sub>design</sub> (Using contractor's equipment)

- 1. % G<sub>mm</sub> at N<sub>initial</sub>,
- 2. % Air Voids, Va,
- 3. % VMA, and
- 4. % VFA.

## Compacted Specimen, $7.0 \pm 0.5\%$ AV (Using contractor's equipment)

1. Loaded Wheel Testing (LWT) as needed.

The inspector will review contractor data and documentation. The inspector will check the plant equipment, lab equipment and plant operations. The inspector will sample asphalt cement working tank and or transport during random plant visits and will obtain random asphalt cement transport samples as requested by the Materials Lab.

Lack of conformance after 5 P-lots to specification requirements may result in increased sampling, reduced pay, removal and replacement of the asphalt mixture, decertification of the technician, and/or decertification of the plant. Correct deficiencies or cease operations.

## **502.07 ROADWAY OPERATIONS.**

**502.07.1 Weather Limitations:** Apply asphalt concrete mixtures on a dry surface when the ambient temperature is above 50°F for wearing courses and 40°F for base and binder courses. Material in transit, or a maximum of 100 tons in a surge bin or silo used as a surge bin, at the time plant operation is discontinued may be placed. All mixture placed is expected to perform satisfactorily and meet specification requirements. Inclement weather will be sufficient reason to terminate or not begin production.

When base course mixtures are placed in plan thicknesses of 2 3/4 inches or greater, disregard temperature limitations provided all other specification requirements are met. When a wearing course is substituted for a binder course mixture, apply the temperature limitation for binder course.

**502.07.2 Surface Preparation:** Maintain the surface being covered. Acceptance is required for each surface prior to placement of subsequent surface.

Roadway slope shall be established at the base course level unless otherwise authorized by the engineer. The absolute minimum lift thickness placed shall be 1/4 inch

greater than the nominal maximum aggregate size as shown on Table 502-6. Failure to meet minimum thickness is subject to removal.

**502.07.2.1 Cleaning:** Sweep the surface to be covered clean of dust, dirt, caked clay, caked material, vegetation, and loose material by revolving brooms or other mechanical sweepers supplemented with hand equipment as directed. Remove excess joint filler from the surface by an approved method when mixtures are to be placed on portland cement concrete pavement or previously overlaid portland cement concrete. Remove any existing raised pavement markers prior to asphalt concrete overlay operations. Payment for removal of pavement markings will be in accordance with the applicable item.

Wash the surface with water in addition to brooming when brooming alone does not adequately clean the surface.

When tack coat is exposed to traffic for more than one (1) calendar day, becomes contaminated, or degrades due to inclement weather, reapply the tack coat at the initial recommended rate at no direct pay.

## 502.07.2.2 Applying Tack Coat:

**502.07.2.2.1** Existing Pavement Surfaces: Before constructing each course, apply an approved asphalt tack coat in accordance with Section 504. Protect the tack coat and spot patch as required.

502.07.2.2.2 Raw Aggregate Base Course and Raw Embankment Surfaces: Apply an approved asphalt prime coat to unprimed surfaces, or protect inplace prime coat and spot apply prime coat as required, in accordance with Section 505.

**502.07.2.2.3** Cement and Lime Stabilized or Treated Embankment and Base Course Surfaces: Apply an approved asphalt curing membrane when none is in place, or protect the in-place curing membrane and spot apply, as required, with asphalt material in accordance with Section 506.

**502.07.2.2.4 Other Surfaces:** Cover contact surfaces of curbs, gutters, manholes, edges of longitudinal and transverse joints, and other structures with a uniform coating of an approved asphalt tack coat complying with Section 504 before placing asphalt mixtures.

## **502.07.3 Joint Construction:**

**502.07.3.1 Longitudinal Joints:** When constructing longitudinal joints, set the screed to allow approximately 2 inches onto the adjacent pass. Use approved 10-foot static straight edge to maintain no greater than 1/8-inch deviation in grade. Make necessary correction in joint before continuing operations. Offset longitudinal joints in one layer over those in the layer below by a minimum of 3 inches; however, keep the top layer joint 6 inches to 9 inches from the centerline of two lane highways. Offset 6 inches to 9 inches from lane lines when the roadway is more than two lanes. Construct the narrow strip first.

**502.07.3.2 Transverse Joints:** Construct transverse joints by milling or hand forming paper butt joints. Use an approved 10-foot static straightedge to identify the location to be cut back to maintain no greater than a 1/8-inch deviation in grade. Lightly tack the cut face of the previously placed mat before fresh material is placed. Rest the screed on shims that are approximately 25 percent of lift thickness placed on the compacted

mat. Provide an adequate crew to form transverse joints. Additionally, meet the transverse joint surface tolerance requirements of Table 502-5. Make necessary corrections to the joint before continuing placement operations.

Offset transverse joints in succeeding lifts by at least 3 feet.

**502.08 HAULING, PAVING AND FINISHING.** Transport mixtures from the plant and deliver to the paver at a temperature no cooler than 25°F below the lower limit of the approved job mix formula, maintaining a temperature of the WMA mix not cooler than 245°F going through the paver. Send no loads so late in the day that completion of spreading and compaction of the mixture cannot be completed during daylight, unless artificial lighting has been approved and is on site.

Load haul trucks to minimize segregation.

Place each course of asphalt mixture in accordance with the specified lift thickness shown in Table 502-6.

With the engineer's approval, motor patrols may be used to level isolated depressions in the initial layer, provided this construction does not result in unsatisfactory subsequent lifts.

**502.08.1** Coordination of Production: Coordinate and manage plant production, transportation of mix and placement operations to achieve a high quality pavement. Provide sufficient hauling vehicles to ensure continuous plant and roadway operations. The engineer will order a halt to operations when sufficient hauling vehicles are not available.

On final wearing course construction under traffic with pavement layers of 2 inches compacted thickness or less, the contractor will be permitted to pave one travel lane for a full day and the adjacent travel lane the next work day. When the adjacent travel lane is not paved the next work day and the longitudinal joint is exposed to traffic for more than 3 calendar days, and it has been determined that the roadway edge is not true to line and grade as previously constructed, cut back the entire length of exposed longitudinal joint to lift thickness to a vertical edge and heavily tack unless a notch wedge device is used. When pavement layers are greater than 2 inches compacted thickness, place approximately 1/2 of each day's production in one lane and the remainder in the adjacent lane unless an approved notched wedge device is used.

Protect pavement from traffic until it has sufficiently hardened to the extent the surface is not damaged.

502.08.2 Paving Operations: When placing the final two lifts of asphalt concrete on the roadway travel lanes, use a material transfer vehicle (MTV) as described in 503.14. During continuous paving, maintain temperature of the mixture constant. At no time shall there be more than 50°F difference in temperature as measured in 300 linear feet of paving or 25°F across the full paved width. All mixtures shall flow through the paver hopper. Lift into the hopper any mixture dropped in front of the paver or reject such material and cast it aside. Deliver material to the paver at a uniform rate and in an amount within the capacity of paving and compacting equipment. Adjust the paver speed and number of trucks to maintain one truck waiting in addition to the one at

the paver in order to maintain continuous paving operations. Maintain a uniform height of material in front of the screed.

Keep the paver steady and in constant alignment during mix transfer. Maintain a level of mix higher than the paver hopper feed slats at all times. Use pavers and operators capable of placing mixtures to required line, grade and surface tolerance without resorting to hand finishing.

Construct longitudinal joints and edges along established lines. Utilize some form of longitudinal control for the paver to follow, preferably a string line. Position and operate the paver to closely follow the established line. Correct irregularities in alignment by trimming or filling directly behind the paver. Check the texture for uniformity after each load of material has been placed. Check the adjustment of screed, feed screws, hopper feed, etc., frequently and adjust as required to assure uniform spreading of the mix to proper line and grade and adequate compaction. When segregation of materials or other deficiencies occur, suspend paving operations until the cause is determined and corrected.

Correct surface irregularities directly behind the paver. Hand placement will be allowed in accordance with 502.08.3 for surface repair, taking care never to cast material over the fresh surface.

Discontinue paving operations when any screed control device malfunctions during binder or wearing course placement operations. When malfunctions occur, limit material through the paver to that which is in transit. Assume responsibility of meeting all specifications and yield requirements, and bear the cost of any overrun during malfunctions. Do not resume paving operations until the malfunction is fixed.

When paving operations are interrupted, remove and replace at no direct pay, mixture that has cooled below the point that it cannot be finished, or compacted to meet specifications. When additional mix is required to increase superelevation in curves, the use of automatic slope control is optional. However, ensure slope by measuring with a slope board. Allow the engineer use of the slope board upon request.

Use the traveling reference plane method of construction for airport runways unless designated otherwise on the plans. Unless the erected string line is required or directed, use the 27-foot (minimum) traveling reference plane method of construction for roadway travel lanes. The requirements of 502.08.2.1, 502.08.2.2, and 502.08.2.3 shall apply for mechanical pavers.

**502.08.2.1 Traveling Reference Plane:** Obtain approval of the traveling reference plane method before use. After the initial paving strip of each lift is finished and compacted, place adjacent paving strips to the grade of the initial paving strip using the traveling reference plane or shoe device to control grade and a slope control device to control cross slope.

On multilane pavements, the initial paving strip and the sequence of lane construction will be subject to approval.

When both outside edges of the paving strip being placed are flush with previously placed material, do not use the slope control device. A grade sensor is required for each side of the paver.

In superelevated curves, the cross slope shall be changed from that specified for tangents to that specified for superelevation in gradual increments while the paver is in motion so a smooth transition in grade is obtained. This change in cross slope shall be accomplished within the transition distance specified.

This is the minimum acceptable method and the contractor must meet or exceed current surface tolerance specifications.

**502.08.2.2 Erected Stringline:** Use the erected stringline method in isolated areas as directed by the engineer. This method may be used on the first lift of asphalt when the underlying new or reconstructed bases do not have grade control requirements. Equip pavers for roadway travel lanes with automatic screed and slope control devices when used with an erected stringline.

An erected stringline shall consist of a piano wire or approved equal stretched between stakes set at no greater than 25-foot intervals. Tension the stringline between supports so that there is less than 1/8 inch variance between supports when the sensor is in place. If required, place the initial paving strip of the first lift constructed using an erected stringline referenced to established grade. When permitted, mixtures required to level isolated depressions may be placed without automatic screed control. Subsequent lifts may be constructed by use of the traveling reference plane, provided surface and grade tolerances are met on the previous lift.

Only one grade sensor and the slope control device are necessary for roadways with a normal crown on tangent alignment. Superelevated curves will require the use of two grade sensors and two erected stringlines to obtain proper grade and slope; however, when the automatic screed control device is equipped with a dial or other device which can be conveniently used to change the cross slope in small increments, superelevated curves may be constructed using this device and one erected stringline.

After the initial paving strip of the first lift is finished and compacted, lay adjacent paving strips using an approved traveling reference plane.

**502.08.2.3** Without Automatic Screed Control: When permitted, pavers without automatic screed control may be used for pavement patching, pavement widening, paved drives and turnouts.

**502.08.3 Hand Placement:** When the use of mechanical finishing equipment is not practical, the mix may be placed and finished by hand to the satisfaction of the engineer. During paving operations, material shall be thoroughly loosened and uniformly distributed. Material that has formed into lumps and does not break down readily will be rejected. Check the surface before rolling and correct irregularities.

## 502.09 ROLLING AND COMPACTION.

**502.09.1 General:** After placement, uniformly compact mixture by rolling while still hot, to a density that complies with Table 502-5. If continuous roller operation

is discontinued, move rollers to cooler areas of the mat where they will not leave surface indentations. The use of steel wheel rollers in the vibratory mode, which result in excessive crushing of aggregate, will not be permitted.

Utilize experienced operators when rolling the mixture using consistent rolling sequences and uniform methods to achieve specified density and smoothness. Uniformly overlap preceding passes of individual roller passes to ensure complete coverage of the paving area. Do not tear or crack the mat by varying the roller speed, amplitude, vibration frequency or other roller operation. Operate non-vibrating steel wheel rollers with drive wheels toward the paver. Correct any operation causing displacement, tearing or cracking of the mat.

Prohibit use of equipment, which leaves tracks or indented areas that cannot be corrected in normal operations or fails to produce a satisfactory surface. Stop use of equipment resulting in accumulation of material and subsequent shedding of accumulated material into the mixture or onto the mat.

Keep rollers of steel wheel rollers properly moistened without excess water to prevent adhesion of mixture to rollers.

Maintain adequate heat for pneumatic tire rollers to prevent mix from adhering to tires. Operate the pneumatic tire roller at a contact pressure which will result in a uniform, tightly knit surface. Keep the pneumatic tire roller approximately 6 inches from unsupported edges of the paving strip; however, when an adjacent paving strip is down, overlap the adjacent paving strip approximately 6 inches.

Vibratory rollers may be used provided they do not impair the stability of the pavement structure or underlying layers. Vibratory rollers shall not be used on the first lift of asphalt concrete placed over the asphalt treated drainage blanket. When mix is placed on newly constructed cement or lime stabilized or treated layers, do not use vibratory rollers until base is approved by the engineer and not for at least 5 days after such stabilization or treatment.

It is the responsibility of the contractor to determine the number, size, and type of rollers to sufficiently compact the mixture to the specified density and surface smoothness. Ensure that the rolling equipment is capable of maintaining the pace of the paver and conforms to 503.16.

The surface of mixtures after compaction shall be smooth and true to cross slope and grade within the tolerances specified. Remove mixtures that become loose, broken, contaminated or otherwise defective and replace with fresh hot mixture compacted to conform to the surrounding mixture.

Excessive rippling of the mat surface will not be accepted. Ripples are small bumps in the pavement surface which usually appear in groups in a frequent and regular manner. No more than 12 ripples or peaks will be allowed in any 100-foot section. Rippling indicates a problem with the paving operation or mix that requires immediate corrective action by the contractor; otherwise cease operations. Correct unacceptable areas at no direct pay.

After rolling, ensure that newly finished pavements have a uniform, tightly knit surface free of cracks, tears, roller marks or other deficiencies. Correct deficiencies at no direct pay and adjust operations to correct the problem. This may require the contractor to adjust the mix or furnish additional or different equipment.

**502.09.2 Hand Compaction:** Along forms, curbs, headers, walls and at other places inaccessible to rollers, compact the mixture uniformly to the satisfaction of the engineer with approved hand tampers or mechanical tampers, conforming to 503.17.

**502.10 ROADWAY LOT SIZES.** A roadway lot is determined as mix placed consecutively on the project from a specific JMF.

## 502.10.1 Mainline Mix Lot Sizes:

The mainline sublot size is 7500 linear lane feet; the mainline lot is five sublots or 37,500 linear lane feet. Any project with less than 37,500 linear lane feet for any mix type is also defined as a lot. The final mainline lot size may be extended one sublot with the approval of the engineer.

## **502.10.2** Minor Mix:

Minor mix lots will be defined as 1000 tons delivered to the project by mix type. The following types should be kept in separate lots.

**502.10.2.2 Minor Lots with Density Requirement:** Minor mix lots with density requirements are 1000 tons. These include bike paths, crossovers, detour roads, parking lots, patching, widening, uniform leveling thicker than 1.5 inches, tapers, and shoulders paved independently which are less than 8 feet wide.

**502.10.2.3 Minor Lots without Density Requirement:** Minor mix lots such as curbs, driveways, guardrail widening, islands, joint repair, spot leveling, medians, turnouts and  $\leq 4$  feet shoulder paved with the mainline do not have density requirements. Make compaction effort to the satisfaction of the engineer. Lots are 1000 tons.

For projects, or separate locations within a project, requiring less than 250 tons, the JMF, materials, and plant and paving operations shall be satisfactory to the engineer. Sampling and testing requirements may be modified by the engineer and the payment adjustment for deviations waived.

**502.11 ROADWAY ACCEPTANCE.** Acceptance testing for pavement density and dimensional tolerances will be conducted on that portion of the lot placed on each contract. Acceptance testing for surface tolerance will be conducted upon completion of mainline paving.

Do not place asphalt concrete mixture exhibiting deficiencies such as segregation, contamination, lumps, non-uniform coating, excessive temperature variations, or other deficiencies apparent on visual inspection.

Correct and/or replace at no direct pay any asphalt concrete mix exhibiting deficiencies, such as segregation, contamination, alignment deviations, variations in surface texture and appearance or other deficiencies, apparent on visual inspection. Poor construction practices such as handwork, improper truck exchanges, improper joint

construction, or other deficiencies, apparent on visual inspection, will be corrected at no direct pay.

**502.11.1 Density:** Obtain pavement samples from each sublot within 24 hours after placement. When this falls on a day the contractor is not working, sampling will be done within 3 calendar days. Sample at locations determined by the PE using random number tables shown in DOTD S605.

When the sampling location determined by random sampling falls within areas that are to be replaced or within 18 inches of the unsupported pavement edge, another random sampling location will be used.

Take cores, approximately 6 inches in diameter, with an approved core drill. Furnish samples cut from the completed work. Replace removed pavement with hot or cold mixture and refinished during the work day coring is performed at no additional pay. Sample in the presence of the engineer's representative. Do not use cores less than 1 3/8 inches thick for payment determination. For transport by parties other than DOTD representatives, ensure that the cores are individually wrapped, sealed, signed, and dated by the DOTD inspector or representative using an approved method. Any evidence of tampering with the core will result in the cores being rejected and additional pavement samples being required.

The engineer or his representative will transport cores in approved transport containers. When allowed, the contractor or third party will transport in an approved, locked transport container.

Divide the 7500-linear-lane-foot sublot into three segments of 2500 linear feet each. Obtain one acceptance core from each segment for a total of three cores. Take a verification core randomly from the 7500-linear-foot sublot. Take a resolution core randomly from the 7500-linear-foot sublot. There are five 7500-foot sublots for each 37,500 linear foot lot. For each lot, there are a total of 15 acceptance cores, 5 verification cores and 5 resolution cores.

For project lots between 2500 and 5000 linear feet, take two acceptance cores per sublot. Projects having less than 2500 linear feet will require 3 cores. Sampling for projects with less than 250 tons may be modified by the Project Engineer.

**502.11.1.1 Testing of Roadway Cores (Method 1):** The District Laboratory will calculate the density of each acceptance roadway core using the G<sub>mb</sub> of the core and the representative maximum specific gravity, G<sub>mm</sub>, in accordance with 502.05.

The density requirement for each lot is shown in Table 502-5. Cores will be retained for a period of 10 days after density is reported.

**502.11.1.2 Testing of Roadway Cores (Method 2) Contractor's Testing of Roadway Cores in Acceptance Decision:** With proven plant production consistency, and when recommended by the District Laboratory Engineer and approved by the Materials Engineer, contractor may request to be allowed to sample and test roadway cores for acceptance at no cost to the Department in lieu of District Laboratory acceptance testing. Density calculations for each acceptance roadway and verification core will utilize the Gmb of the core and the representative maximum specific gravity, Gmm as determined in accordance with 502.05. Refer to 502.11.1 for core responsibility. The District Laboratory

roadway lot verification will be based on a means comparison between the District Laboratory verification average and the contractor acceptance average for each lot. If the means comparison produces a difference, use the resolution cores for pay determination. The Department will send the resolution cores to a certified Independent Assurance (IA) laboratory to determine pay in accordance with 502.11.1.5.

For Method 2: The plant production consistency will be determined as follows: The Department will continuously monitor plant data and roadway data by JMF, by plant, by contractor. Plant data will be monitored in accordance with 502.06. Roadway data will be monitored using statistical methods comparing means and variances (F and t) tests. Continued use of Method 2 is allowed unless the plant or roadway data fail to verify with data set of a minimum of 45 contractor acceptance tests and 15 DOTD verification tests results, and it is determined by the DLE and Independent Assurance team that the contractor's production data meets requirements.

If the F and t test fail an investigation shall be conducted by the IA team. If the contractor data after F and t analysis is performed and is found to be error, DOTD acceptance testing of roadway cores will resume and independent accredited laboratory could be required for plant testing at no cost to the Department until such time as the problem is identified and resolved.

- 502.11.1.3 "Minor with Density" Requirements: For Method 1: When density is specified in Table 502-5, the roadway inspector will identify core locations to be cut by the contractor. The District Laboratory will test three cores for density every 1000 tons per mix type placed per roadway sampling procedure mentioned above and pay in accordance with Table 502-7. The District Laboratory will calculate the density of each roadway core using the Gmb of the core and the representative maximum specific gravity, Gmm, in accordance with 502.05. For Method 2: The contractor will perform acceptance test per above method. Table 502-7 is used to compute pay.
- **502.11.1.4 Minor Mix without Density:** This minor mix shall have a neat, uniform appearance and be compacted by methods to the satisfaction of the engineer. Collect one loose mix specimen, from roadway, per project, for G<sub>mm</sub> verification.
- **502.11.1.5 Verification:** One core will be selected every 7500 linear lane feet and will be evaluated by either Method 1 or Method 2 in accordance with 502.11.1.1.
- **502.11.1.6 Resolution.** One core from each 7500 linear lane feet of placed mix will be chosen at random and will be double sealed, signed by both contractor and Department's certified inspectors in accordance with the Quality Assurance Manual as required or for documentation. The resolution core will be tested at a certified IA laboratory as described in the QA manual.
- 502.12 SURFACE TOLERANCE EQUIPMENT, QUALITY CONTROL, ACCEPTANCE, MEASUREMENT AND PAYMENT ADJUSTMENT. Measure the top two lifts of the mainline travel lanes with an approved inertial profiler. Maintain record of intermediate measures of smoothness quality as described herein. Final acceptance will

be based on the last measurement taken on the final wearing course of the travel lanes. Measurement of the center two lanes will be required for airports.

Constantly monitor equipment, materials, and processes to ensure that surface tolerance requirements are met.

502.12.1 Equipment: For longitudinal surface tolerance quality control testing and acceptance testing on mainline wearing and binder courses, furnish and use a DOTD certified inertial profiler. Certified profilers will have a DOTD decal indicating the date of certification and profiler system parameter settings. Measure longitudinal surface profile in inches per mile in accordance with DOTD TR 644 and report as the International Roughness Index (IRI). Verify the profiler system parameter settings before each run. Demonstrate the daily set up procedure and pre-operation tests in accordance with the manufacturer's procedures and DOTD TR 644. Ensure that a copy of the manufacturer's setup, pre-operation, and general operating procedures for measuring surface tolerance are available at all times during measurement.

For transverse quality control testing and for longitudinal quality control testing for wearing course on bike paths, detour roads, parking lots, and shoulders; furnish and use an approved 10-foot metal static straight-edge and electronic or static level.

Profiler system parameter settings shall be verified before and during each run by the DOTD inspector. For transverse, cross slope and grade testing, furnish a 10-foot metal static straightedge and electronic or static level for Department use.

- **502.12.2 Longitudinal Smoothness Quality Control:** Within 7 calendar days of placement, for mainline wearing and binder courses, run the certified profiler. View the raw data with ProVAL to determine IRI and to view Profilograph Simulation for each wheelpath. Make corrections to operation and/or mixture to ensure that the overall ride and individual bump requirements are met. Correct all individual bumps which are more than 1/4 inch as identified on Profilograph Simulation or when tested with a 10-foot metal static straightedge. Ensure that the following quality requirements are met:
- 1. Produce IRI which meets the requirements for 100 percent pay in accordance with Table 502-8. Continued surface tolerance penalties are not allowed.
- 2. Correct all individual bumps which are more than 1/4 inch when tested with a 10-foot metal static straightedge. Utilize the Profilograph Simulation on ProVAL to help identify these bumps.
- 3. Correct ripples to the satisfaction of the engineer. Report Profilograph Simulation for areas with 12 or more small, regular bumps in a 100-foot section or for any areas in question.

Minor mixes shall comply with Table 502-5. For minor mixes, use the 10-foot metal static straightedge to check for conformance to specifications.

**502.12.3 Transverse Smoothness, Cross Slope, and Grade:** The Department will test the surface of the binder and wearing courses at selected locations for conformance to the surface tolerance requirements of this subsection and Table 502-5. Make corrections as directed in accordance with 502.12.4.

- **502.12.3.1 Transverse Smoothness:** Areas with surface deviations in excess of specification limits shall be isolated and corrected in accordance with 502.12.4. Control the transverse surface finish.
- **502.12.3.2** Cross Slope: When the plans require the section to be constructed to a specified cross slope, take measurements at selected locations using a stringline, a slope board, an electronic or static level mounted on a 10-foot metal static straightedge, or other comparable device. Control the cross slope for each lane to comply with the tolerances shown in Table 502-5. Make corrections in accordance with 502.12.4.
- **502.12.3.3 Grade:** When the plans require the pavement to be constructed to a specified profile grade, test for conformance at selected locations, using a stringline or other comparable device. Control grade variations so that the tolerances shown in Table 502-5 are not exceeded. Grade tolerances shall apply to only one longitudinal line, such as the centerline or outside edge of pavement. Make corrections in accordance with 502.12.4.
- **502.12.4 Correction of Deficient Areas:** Correct areas as required in 502.12.2 and those not meeting Table 502-5, and Table 502-8. Correct wearing and binder courses by grinding. In lieu of grinding, the Project Engineer may penalize the contractor \$800 per area of small individual bumps, and/or per "Ripple" as defined in 502.12.2.
- **502.12.4.1 Deficiencies in Mainline Wearing Course:** Correct deficiencies in the final wearing course by removing and replacing mixture, or by diamond grinding or other approved device across the lane and applying a light tack coat, or by furnishing and placing a supplemental layer of wearing course mixture at least 1 1/2 inches compacted thickness for the full width of the roadway meeting specification requirements at no direct pay. If the supplemental layer does not meet specification requirements to the satisfaction of the engineer, remove and replace or correct it by other methods approved by the engineer.

For areas that will not be improved by grinding such as minor dips, extreme vertical curves, areas with < 1/4 inch bump as measured with a 10 feet metal static straight edge, the engineer may waive the requirement to grind.

- **502.12.4.2 Deficiencies in Mainline Binder Courses:** Correct deficiencies in binder course: longitudinal, transverse, cross slope, and grade to meet specification requirements at no direct pay. Make corrections before subsequent courses are constructed.
- **502.12.4.3 Deficiencies in Minor Mixes:** Correct deficiencies in minor mixes by diamond grinding or approved method at the project engineer's direction.
- **502.12.5 Surface Tolerance Acceptance:** Measure the top two lifts of the mainline travel lanes with an approved inertial profiler. Final acceptance will be based on the last measurement taken on the final wearing course of the travel lanes. Measurement of the center two lanes will be required for airports.
- 502.12.5.1 Longitudinal Surface Tolerance Acceptance: Measure surface tolerance at the completion of the project and after all corrections have been made or at an approved portion of the project in accordance with 502.12.2. Measure the mainline wearing course continuously from start to finish in the direction of travel. The

measurement shall be performed by the contractor in the presence of a Department representative. The measurement may also be made by the Materials and Testing Section, or by a private company approved by the Department. Report one IRI measurement in inches per mile for the entire project. A stand- alone pay adjustment factor will be determined in accordance with 502.15.

Place a start and stop mark at the beginning and end of each travel lane so that measurements can be rerun by the Department if needed. Interim measurements of a portion may be allowed, with approval of the engineer, as follows:

- 1. For partial acceptance in accordance with 105.17.1.
- 2. Due to phasing or sequence of construction, this measurement may result in 100 percent pay or less. However, payment exceeding 100 percent for this section of roadway will only be allowed if the smoothness re- measured at the completion of the project meets the requirements of Table 502-8.
- 3. For an unavoidable lengthy delay, apply the same payment criteria as No. 2 above.

The mainline longitudinal surface tolerance IRI specification requirements are shown in Table 502-8. Perform profiler testing and submit data to the engineer before starting paving operations. To ensure that the contractor has corrected deficiencies, the Department will spot check for 1/4 inch bumps in accordance with 502.12.2. Although grinding may be waived by the engineer, the measured roughness will still contribute to the total IRI for the project.

A DOTD inspector will be present for the final test run and will immediately receive a copy of the raw data, the ".erd file" and any files with information about the project, the operator, the equipment, the settings, daily pre-operation results, and a copy of the IRI results via USB flash drive provided by the contractor. In addition to the data transferred by USB storage device, provide to the engineer a paper copy of the IRI report. Acceptance for the project will be in accordance with Tables 502-8, based on the data. The Department may elect to perform and utilize independent ride quality test results for acceptance at any time.

- **502.12.5.2 Exclusions:** The final IRI measurement shall be taken in entirety, without exclusions. The Department will then review the profile report obtained for each lane of the mainline wearing course. In special cases or extenuating circumstances, the engineer may isolate or exclude sections of the profile. These include the following:
  - 1. Bridge ends, and sections that are within 150 feet of bridge ends.
- 2. Outside wheelpath of curb and gutter sections that require adjustment in order to maintain adequate drainage.
  - 3. Manholes, catch basins, valve and junction boxes.
  - 4. Street intersections or rail road crossings of a different grade.
  - 5. Structures located in the roadway which cause abrupt deviations in the profile.
- 6.Transitions to and from ramps and turn lanes and sections within 200ft of the limits of the project if the limits begin or end at an intersection.
- 7. Sections where the project engineer determines that attaining smoothness is beyond the contractor's reasonable control.

Exclusions will not be used to simply isolate sections of road that are in poor condition when the project is let. The roughness in excluded areas will not be included in the total IRI used for payment purposes, but shall meet the requirements of 502.12.2. The quantity of asphalt represented by the length excluded will not receive a pay adjustment for surface tolerance.

- **502.12.6 Surface Tolerance Measurement:** Measure and report the average IRI of each wheel path of each mainline lane in inches per mile and reach mainline lane prorated for the entire project. The theoretical quantity is computed by using the total length of lanes, the plan thickness, and the plan width, excluding shoulders and minor mixes. Adjust the tons as necessary affected represented for each mainline travel lane.
- **502.12.7 Payment Adjustment for Surface Tolerance:** Apply a percent payment adjustment for the quantity of tons represented in each land of the mainline wearing course. This pay adjustment is in addition to pay adjustments for density as described in 502.15.2. For mainline wearing course, a separate pay adjustment for surface tolerance measured on the mainline wearing course based on Table 502-8 shall apply. Apply the adjustment to the theoretical lane quantity and contract unit price.
- **502.13 DIMENSIONAL REQUIREMENTS.** Ensure that mixtures conform to the following dimensional requirements only. No other acceptance tests will be required for these mixtures. Over-thickness and over-width will be accepted at no direct pay.
- **502.13.1 Thickness:** For mixture specified for payment on cubic yard or square yard basis, thickness of mixtures will be determined by the Department in accordance with DOTD TR 602. Under-thickness shall not exceed 1/4 inch.

Correct area under-thickness in excess of 1/4 inch to plan thickness at no direct pay. Furnishing and placing additional mixture in accordance with 502.12.4.1. Correct excesses of 1/2 inch for category D, Table 502-8. When grade adjustments do not permit placing additional mixture, remove the deficient under-thickness area and replace at no additional pay.

For mixtures specified for payment on a per ton basis, thickness of mixtures will be determined by the plans, Table 502-6, and that agreed to with the Project Engineer. Under thickness shall not exceed 1/2 inch. Removal and replacement of deficient under-thickness area(s) or other approved remediation agreed to by the Project Engineer will be at no direct pay.

- **502.13.2** Width: The width of completed courses will be determined in accordance with DOTD TR 602. Correct under-widths by furnishing and placing additional mixture to a minimum width of 1 foot and plan thickness at no direct pay.
- **502.14 MEASUREMENT.** Measure asphalt concrete by the ton of 2,000 pounds from printed weights as provided in Section 503. Provide stamped printer tickets with each truckload of material delivered denoting JMF number and plant tonnage.

Material lost, wasted, rejected or applied contrary to specifications will not be measured for payment.

Estimated quantities of asphalt concrete shown on the plans are based on 110 lb/sq yd/inch thickness. The measured quantity of asphalt mixtures will be multiplied by the following adjustment factors to obtain the pay quantity.

| Theoretical Maximum Specific              | <u>Adjustment</u> |
|---|-------------------|
| Gravity, (G <sub>mm</sub> ) (DOTD TR 327) | <u>Factor</u>     |
|   |                   |
| 2.340 - 2.360                             | 1.02              |
| 2.361 - 2.399                             | 1.01              |
| 2.400 - 2.540                             | 1.00              |
| 2.541 - 2.570                             | 0.99              |
| 2.571 - 2.590                             | 0.98              |

The adjustment factor for mixtures with theoretical maximum specific gravities less than 2.340 or more than 2.590 will be determined by the following formulas:

Theoretical maximum specific gravity less than 2.340:

$$F = \frac{2.400}{S}$$

Theoretical maximum specific gravity more than 2.590:

$$F = \frac{2.540}{S}$$

where,

F = quantity adjustment factor

S = theoretical maximum specific gravity of mixture from approved job mix formula.

**502.14.1 Volume or Area Measurement:** The quantities for payment will be the design quantities specified in the plans and adjustments thereto. Design quantities will be adjusted when the engineer makes changes to adjust the field conditions or when design changes are necessary. Design quantities are based on the horizontal dimensions and compacted thickness of the completed course shown on the plans.

## **502.15 PAYMENT.**

- of asphalt mixture accepted on the roadway. Payment for all mixes will be at the contract unit price of asphalt mixture accepted on the roadway. Payment for asphalt concrete will include furnishing all required materials, producing the mixtures, preparing the surfaces on which the mixtures are placed, hauling the mixtures to the work site, and placing and compacting the mixtures. When the mix does not meet requirements, payment adjustments shall be assessed. Production of mix that is not eligible for 100 percent payment will not be allowed on a continuous basis. When test results demonstrate that payment adjustments are necessary, satisfactory mixture and compaction adjustments shall be made, or production shall be discontinued. All calculations for percent payment adjustments will be rounded to the nearest one (1) percent. Payment for removal of pavement markings will be in accordance with the applicable item. Payment adjustments will be determined in accordance with 502.14 and the QA Manual.
- **502.15.2 Mainline Mixtures.** For all mainline mixtures, adjustments in contract unit price for roadway density as required by Table 502-5 and will be based on PWL using Tables 502-9 and 502-10 for all acceptance cores in the lot. This payment adjustment will be applied to the theoretical mainline lane quantity and contract unit price.

In addition, for mainline wearing course, a separate pay adjustment for surface tolerance based on Table 502-8 shall apply for all travel lanes based on the theoretical mainline lane quantity and contract unit price.

The theoretical quantity is computed by using the plan width, the plan thickness, and the total length of travel lanes, without exclusion areas.

#### 502.15.3 Minor Mixtures.

- **502.15.3.1 Minor Shoulder Lots, > 4 Feet Wide.** Adjustments in contract price for shoulder density will be based on the average density for all cores in the lot and Table 502-5.
- **502.15.3.2 Minor Lots with Density.** Adjustments in contract price will be based on the core density for each lot in accordance with Table 502-7.
- 502.15.4 Payment for Tack. Tack coat as required in 502.07.2.2 "Applying Tack Coat" will be considered incidental to the 502 item. If the engineer adjusts the application rate of tack coat from that specified by the contract document, payment for the asphalt mixture will be increased or decreased based on the difference in the applied quantity of asphalt emulsion shown on paid invoices (total of charges). The contractor shall provide copies of paid invoices for this determination. Apply 95 percent payment to the 502 item when the tack coat rates do not meet the application rate as allow by the engineer.
- **502.15.5 Payment Adjustment for Asphalt Cement.** A minimum payment adjustments of 10 percent of the 502 item will apply to mixtures that do not meet specification but are within

one grade of the specification. Asphalt that exceeds one lower grade difference in specification will be subject to 50 percent payment reduction or removal at the discretion of the Chief Engineer.

**502.15.6 Payment Adjustment for Surface Tolerance.** Payment adjustment will be in accordance with 502.12.7. Apply a percent payment adjustment for quantity of tons represented in each lane of the mainline wearing course. This pay adjustment is in addition to the pay adjustments for density as described in 502.15.2. For mainline wearing course, a separate pay adjustment for surface tolerance measured on the mainline wearing course based on Table 502-8 shall apply. Apply the adjustment to the theoretical lane quantity and contract unit price.

**502.15.7 Payment for Erected Stringline.** When the use of an erected stringline is not specified, but directed by the engineer, an additional payment of \$3500 per contract plus \$0.25 per linear foot will be made for mixtures placed by the erected stringline method. When the use of an erected stringline is specified, no additional payment will be made.

## Payment will be made under:

| Item No. | Pay Item              | Pay Unit    |
|----------|-----------------------|-------------|
| 502-01   | Asphalt Concrete      | Ton         |
| 502-02   | Asphalt Concrete      | Cubic Yard  |
| 502-03   | Asphalt Concrete,     |             |
|          | (Inches Thick)        | Square Yard |
| 502-04   | Asphalt Concrete (SMA | A) Ton      |
|          | Wearing Course        |             |

502-26

Table 502-4
Display Produced Asphalt Mixture Bequirements and Toler

| rialit rio           | riailt Floduced Aspliait Mixture Neduireilleits and Tolerailces   | MIXIME     | i e dan e ii | IEIIIS AIIU | Oleiaiic   | 200                        |
|----------------------|---|------------|--------------|-------------|------------|----------------------------|
|                      | REQUIREMENTS FOR EXTRACTED ASPHALT CEMENT AND AGGREGATE GRADATION | R EXTRACTI | ED ASPHALT   | CEMENT ANI  | O AGGREGAT | <b>TE GRADATION</b>        |
| U.S. (Metric) Sieve  | ½ inch  | ½ inch     | % inch       | 1 inch      | 1.5 inch   | Mix Tolerance <sup>1</sup> |
| % Passing            | SMA   | Nominal    | Nominal      | Nominal     | Nominal    |                            |
| 2 inch               | I   | 1          | l            | 1           | 100        | ∓4                         |
| 1 1/2 inch           |   | 1          | l            | 100         | 90-100     | ±4                         |
| 1 inch               | I   | 1          | 100          | 90-100      | 89 Max.    | ±4                         |
| 3/4 inch             | 100   | 100        | 90-100       | 89 Max      | I          | ±4                         |
| 1/2 inch             | 90-100  | 90-100     | 89 Max       | I           | I          | <del>+</del> 4             |
| 3/8 inch             | 75 Max.   | 89 Max.    |              | I           | I          | ±4                         |
| No. 4                | 24-34   | l          |              | 1           | 1          | ±4                         |
| No. 8                | 16-28   | 29-58      | 26-49        | 23-45       | 19-41      | +3                         |
| No. 16               |   | 1          |              | 1           | I          | ±2                         |
| No. 30               | 12-25   | 1          |              | 1           | I          | ±2                         |
| No. 50               | 11-22   | 1          |              | 1           | I          | ±2                         |
| No 100               |   | 1          | l            | 1           | I          | ±2                         |
| No. 200              | 7-13  | 4.0-10.0   | 3.0-8.0      | 2.0-7.0     | 1.0-6.0    | ±0.7                       |
| Extracted Asphalt, % | 6.0 min.  | 1          |              | I           | l          | ±0.2                       |
| Mix Temperature      |   | 1          | [            |             |            | ±25°F                      |
|                      |   |            |              |             |            |                            |

<sup>1</sup>Upon validation of the JMF, the validation averages will be used for JMF target values.

| Та  | <b>Table 502-5</b>                  |                                     |                                       |                                |
|---|-------------------------------------|-------------------------------------|---------------------------------------|--------------------------------|
| Asphalt Pave  | Asphalt Pavement Requirements       | ments                               |                                       |                                |
| Density, Min. % of Theoretical Maximum Specific Gravity, AASHTO T 209 Method C      | T 209 Method C                      |                                     |                                       |                                |
| Mainline, SMA<br>Mainline   |                                     | 93.5<br>92.0                        |                                       |                                |
| Minor with density ref, 502.10.2.2 "Roadway lot," patching, and widening > 2.5-feet | idening > 2.5-feet                  | 0.06                                |                                       |                                |
| Surface Tolerance Variation   | Longitudinal <sup>1</sup><br>inches | Transverse <sup>2,3</sup><br>inches | Cross Slope <sup>2,3</sup> inches [%] | Grade <sup>3,4</sup><br>inches |
| Mainline Wearing Courses, Category A, B   | N/A <sup>5</sup>                    | 1/8                                 | 3/8 [0.3]                             | 1/2                            |
| Mainline Wearing Courses, Category C  | N/A <sup>5</sup>                    | 1/4                                 | 1/2 [0.4]                             | 1/2                            |
| Mainline Wearing Courses, Category D  | 1/2                                 | 1/2                                 | 3/4 [0.6]                             | 3/4                            |
| Mainline Binder Courses   | 1/4                                 | 1/2                                 | 3/4 [0.6]                             | 3/4                            |
| Minor Mixes <sup>6</sup>  | 3/8                                 | 3/8                                 | 3/4 [0.6]                             | 3/4                            |
| Bike Paths, Detour Roads and Parking Lots<br>Shoulder, Ramps < 300'                 | 1/2                                 |                                     |                                       |                                |
|   |                                     |                                     |                                       |                                |

<sup>1</sup> See 502.12.2.
<sup>2</sup> Based on 10 feet, using 10-foot static straightedge and static or electronic level.
<sup>3</sup> See 502.12.3.
<sup>4</sup> Applicable only when profile grade is specified.
<sup>5</sup> Mainline wearing and binder are measured with inertial profiler, see 502.12.
<sup>6</sup> Except bike paths, detour roads, parking lots, and shoulders.

Table 502-6<sup>1</sup>
Asphalt Concrete General Criteria

| Nominal Max., Size<br>Agg.                                      | _                                   | ).5 inch<br>2.5 mm) |  | _                 | 1.75 inch<br>19 mm) |               |                |               | 0 inch<br>5 mm)             |                    | 1.5 inch<br>(37.5<br>mm)    | SMA           |
|---|-------------------------------------|---------------------|--|-------------------|---------------------|---------------|----------------|---------------|-----------------------------|--------------------|-----------------------------|---------------|
| Type of Mix   | Incidental<br>Paving <sup>2,9</sup> | Wea<br>Cou          |  | Wearing<br>Course | Binder              | Course        | Binder         | Course        | Base<br>Course <sup>9</sup> | ATB <sup>8,9</sup> | Base<br>Course <sup>9</sup> | Wearing       |
| Level <sup>3</sup>  | Α                                   | 1                   | 2  | 2                 | 1                   | 2             | 1              | 2             | 1                           | 1                  | 1                           | 2             |
| Coarse Agg. Angularity,<br>% Crushed, (Double<br>Faced) + No. 4 | 55                                  | 75                  | 95   | 95                | 75                  | 95            | 75             | 95            | 75                          | 75                 | 75                          | 98            |
| Fine Agg. Angularity,<br>Min. % - No. 8                         | 40                                  | 40                  | 45   | 45                | 40                  | 45            | 40             | 45            | 40                          | 40                 | 40                          | 45            |
| Flat and Elongated<br>Particles, % Max. (5:1)                   |                                     |                     |  |                   |                     |               | 10             |               |                             |                    |                             |               |
| Sand Equivalent, Min. %<br>(Fine Agg.) - No. 4                  | 40                                  | 40                  | 45   | 45                | 40                  | 45            | 40             | 45            | 40                          | 40                 | 40                          | NA            |
| Natural Sand - Max. %   | NA                                  | 1                   | 5  |                   | 15                  |               |                | 15            |                             | 25                 | 25                          | 0             |
| Asphalt Binder  |                                     |                     | Table 502-2, (3% minimum for Asphalt Treated base (ATB), 6% min for SMA) |                   |                     |               |                |               |                             |                    |                             |               |
| Friction Rating <sup>3</sup>                                    |                                     | Table 502-3         |  |                   |                     |               |                |               |                             |                    |                             |               |
| RAP, Max. % of Mix <sup>4</sup>                                 | 20                                  | 15                  | 15   | 15                | 20                  | 20            | 20             | 20            | 30                          | 30                 | 30                          | 0             |
|   |                                     |                     | Compacted Mix Volumetrics <sup>4</sup>                                   |                   |                     |               |                |               |                             |                    |                             |               |
| VMA, Min. % <sup>5</sup>  | 13.5                                | 13.5                |  |                   |                     |               |                |               |                             | n/a                | 10.5                        | 16.0          |
| Air Voids, % <sup>6</sup>                                       |                                     |                     |  |                   |                     | (2.5-4.       | 5); (no lii    | mit for A     | TB)                         |                    |                             |               |
| VFA, % <sup>6</sup>   |                                     |                     |  |                   |                     | (69-8         | 0); no lin     | nit for A     | ГВ                          |                    |                             |               |
| N <sub>initial</sub> 90% max. <sup>7</sup><br>(Gyrations)       | 7                                   | 7                   | 7  | 7                 | 7                   | 7             | 7              | 7             | 7                           | n/a                | 7                           | 7             |
| N <sub>design</sub> 96.5±1 %<br>(Gyrations)                     | 55                                  | 55                  | 65   | 65                | 55                  | 65            | 55             | 65            | 55                          | 30                 | 55                          | 65            |
| N <sub>max</sub> 98 % max.<br>(Gyrations)                       | 90                                  | 90                  | 105  | 105               | 90                  | 105           | 90             | 105           | 90                          | n/a                | 90                          | 65            |
| LWT, max. rut-design, mm<br>@ # passes,<br>@ 50°C               | 10 @<br>10,000                      | 10 @<br>20,000      | 6 @<br>20,000  | 6 @<br>20,000     | 10 @<br>20,000      | 6 @<br>20,000 | 10 @<br>20,000 | 6 @<br>20,000 | 12 @<br>20,000              | 10<br>@10,000      | 12 @<br>20,000              | 6 @<br>20,000 |
| Dust/Effective Asphalt Ratio, %                                 |                                     |                     |  |                   |                     |               | 0.6 –          | 1.6           |                             |                    |                             |               |
| SCB, min, Jc,<br>KJ/m2 @ 25°C                                   | All mix des                         |                     |  |                   |                     |               |                |               |                             |                    |                             |               |
|   | All mix des                         | sign level          | 2 must   |                   | mum 0.6             | S Jc.         |                |               |                             |                    |                             |               |
| Design Lift<br>Thickness, inch <sup>10</sup>                    | 2.0-                                | 1.5-                | -2.0   | 1.5–2.0           | 2.0-                | -3.0          | 2.5-           | -4.0          | 2.5+                        | 3.0+               | 4.0+                        | 1.5-2.0       |

<sup>&</sup>lt;sup>1</sup>See also Table 1003-1 Asphalt Aggregate Properties.

<sup>&</sup>lt;sup>2</sup>May be used for airports, bike paths, crossovers, curbs, driveways, guardrail widening, islands, joint repair, leveling, parking lots, shoulders, turnouts, and other incidental items approved by the engineer. (May be used as a standard roadway mix for local governments.)

Table 502-7
Payment Adjustment Schedule for Minor Mixture<sup>1</sup>

| Parameter <sup>2</sup>                        | Perce  | nt of Contract Unit Pri | ce/Lot                    |
|---|--------|-------------------------|---------------------------|
| Parameter                                     | 100    | 90                      | 50 or Remove <sup>3</sup> |
| Average Roadway<br>Density, % G <sub>mm</sub> | ≥ 90.0 | 89.9 to 88.1            | ≤ 88.0                    |

<sup>&</sup>lt;sup>1</sup>See 502.11.1.3.

502-29

<sup>&</sup>lt;sup>3</sup>Mixtures shall meet the friction rating requirements in Table 502-3 for travel lane wearing courses with ADT >7000, <sup>4</sup>Maximum 20% RAP will be allowed in all shoulder wearing course mixtures. RAP will not be allowed for airports. Five (5) additional RAP will be allowed in all mixes except for airports and SMA when RAP stockpile is pre-screened on a 1-inch scalping screen.

 $<sup>^5</sup>$ Air voids, VMA, VFA,  $^{\%}$  G<sub>mm</sub>  $^{@}$  N<sub>initial</sub>, and  $^{\%}$  G<sub>mm</sub>  $^{@}$  N<sub>design</sub> are determined on samples compacted to N<sub>design</sub>; The parameter of percent G<sub>mm</sub>  $^{@}$  N<sub>max</sub> is determined on a sample compacted to N<sub>max</sub>.

<sup>&</sup>lt;sup>6</sup>Air voids mix design target is a 3.5%, Mix design minimum VFA is 72.0%; Mix design minimum VFA for PG82-22rm is 75.0% and 71% for 25 mm NMS mixtures

<sup>&</sup>lt;sup>7</sup>For Level 1 mixtures, Ninitial shall be 91.0% max. For Level A mixes, N<sub>initial</sub> shall be 92.0% max.

<sup>&</sup>lt;sup>8</sup>Asphalt Treated Base (ATB) may be used for patching of base material, for shoulder <3500 ADT and maintenance widening; when used achieve average density of 90% G<sub>mm</sub> as measured per minor mix table.

 $<sup>^9\</sup>mathrm{Semicircular\,Bend\,Test}$  (SCB) is not required for Level A, Base Course, or ATB mixtures.

Absolute minimum of lift thickness across width equal to 1/2 inch lower than minimum lift thickness.

<sup>&</sup>lt;sup>2</sup>Of the total number of cores per lot. Determine surface tolerance in accordance with Table 502-9, if required.

<sup>&</sup>lt;sup>3</sup>At the option of the Chief Engineer.

# Table 502-8 Payment Adjustment Schedules for Longitudinal

Surface Tolerance, Maximum International Roughness Index, Inches per Mile

| Percent of Contract Unit Price <sup>1</sup>   | 102% | 100%   | 95%  | 80%  | 50% or<br>Remove <sup>2</sup> |
|---|------|--|--|--|-------------------------------|
| Category A <sup>5</sup> All Interstates, Three or more lift construction  | <45  | <65  | 65-85  | 86-149   | 150                           |
| Category B <sup>5</sup> Two Lift Overlays over cold plane surface and two lift overlay over improved base.  | <55  | <75  | 75-95  | 96 <149  | 150                           |
| Category C Two lift overlay over existing surface, Single- Lift Overlays with surface prep. Single Lift Overlays Over Cold Planed Surfaces or improved base | <55  | <85  | 85-110   | 111- 149   | 150                           |
| Category D<br>Single-Lift Overlays Over<br>Unimproved Surfaces <sup>3,4</sup>   | N/A  | Improvement,<br>or less than or<br>equal to 65<br>for all other<br>pavements | 0% - 19%<br>Improvement<br>when initial is<br>greater than<br>95 | IRI Greater<br>than Existing<br>when initial is<br>greater than<br>110 |                               |

Based on total theoretical quantity.

502-30 273

At the option of the Chief Engineer.

<sup>&</sup>lt;sup>3</sup> A project with an unimproved surface has no surface preparation item.

<sup>&</sup>lt;sup>4</sup> Contractor shall take IRI measurements before and after construction and shall show a minimum of 20% improvement.

Remove and replace any individual 0.05-mile segment having greater than an average of 150 in/mile. Removal and replacement will be at the direction of the Chief Engineer. This note does not apply to excluded areas.

Table 502-9
Quality Index Values for Estimating Percent Within Limits (PWL)

|                |  |       | (PWL)                |  |                        |              |
|----------------|--|-------|----------------------|--|------------------------|--------------|
| PWL            | n = 3  | n = 4 | n = 5 - 6            | n = 7 - 9                                    | n = 10 - 12            | n = 13 - 15  |
| 99             | 1.16   | 1.47  | 1.68                 | 1.89   | 2.04                   | 2.14         |
| 98             | 1.15   | 1.44  | 1.61                 | 1.77   | 1.86                   | 1.93         |
| 97             | 1.15<br>1.15   | 1.41  | 1.61<br>1.55         | 1.77<br>1.67                                 | 1.74                   | 1.80         |
| 96             | 1 15   | 1.38  | 1.49                 | 1.59<br>1.52<br>1.46                         | 1.64                   | 1.69         |
| 95             | 1 14   | 1.35  | 1.45                 | 1.52   | 1.56                   | 1.59         |
| 96<br>95<br>94 | 1.14<br>1.13   | 1.32  | 1.40                 | 1.46   | 1.49                   | 1.51         |
| 93             | 1.12   | 1.29  | 1.36                 | 1.40<br>1.35<br>1.30<br>1.25<br>1.21<br>1.17 | 1.43                   | 1.44         |
| 92             | 1.11<br>1.10   | 1.26  | 1.31                 | 1.35   | 1.37                   | 1.38         |
| 91             | 1.10   | 1.23  | 1.31<br>1.27         | 1.30   | 1.32                   | 1.38<br>1.32 |
| 90             | 1.09   | 1.20  | 1.23                 | 1.25   | 1.26                   | 1.27         |
| 89             | 1.09<br>1.08<br>1.07                                 | 1.17  | 1.23<br>1.20         | 1.21   | 1.21                   | 1.22         |
| 89<br>88       | 1.07   | 1.14  | 1.16                 | 1.17   | 1.17                   | 1.17         |
| 87             | 1.06   | 1.11  | 1.12                 | I 1.12                                       | 1.13                   | 1.13         |
| 86             | 1.05   | 1.08  | 1.08                 | 1.08   | 1.08                   | 1.08         |
| 86<br>85       | 1.03   | 1.05  | 1.08<br>1.05         | 1.08<br>1.05                                 | 1.04                   | 1.08<br>1.04 |
| 84             | 1.06<br>1.05<br>1.03<br>1.02<br>1.00<br>0.98<br>0.96 | 1.02  | 1.02                 | 1.01   | 1.00                   | 1.00         |
| 83<br>82       | 1.00   | 0.99  | 0.98<br>0.95<br>0.92 | 1.01<br>0.97<br>0.94                         | 0.96                   | 0.96<br>0.92 |
| 82             | 0.98   | 0.96  | 0.95                 | 0.94   | 0.93                   | 0.92         |
| 81             | 0.96   | 0.93  | 0.92                 | 0.90   | 0.89                   | 0.89         |
| 80<br>79<br>78 | 0.94<br>0.92<br>0.89                                 | 0.90  | 0.88<br>0.85         | 0.87<br>0.83<br>0.80                         | 0.85                   | 0.85         |
| 79             | 0.92   | 0.87  | 0.85                 | 0.83   | 0.82                   | 0.82         |
| 78             | 0.89   | 0.84  | 0.82                 | 0.80   | 0.79                   | 0.78         |
| 77             | 0.87   | 0.81  | 0.79                 | 0.77   | 0.76                   | 0.75         |
| 77<br>76       | 0.84   | 0.78  | 0.76                 | 0.74   | 0.72                   | 0.72         |
| 75             | 0.84<br>0.82   | 0.75  | 0.79<br>0.76<br>0.73 | 0.71   | 0.69                   | 0.69         |
| 74             | 0.79<br>0.77<br>0.74                                 | 0.72  | 0.70                 | 0.67<br>0.64<br>0.61                         | 0.66                   | 0.66         |
| 73             | 0.77   | 0.69  | 0.67                 | 0.64   | 0.63                   | 0.62         |
| 73<br>72       | 0.74   | 0.66  | 0.64                 | 0.61   | 0.60                   | 0.62<br>0.59 |
| 71             | 0.71   | 0.63  | 0.60                 | 0.58<br>0.55                                 | 0.57                   | 0.56         |
| 70             | 0.71<br>0.68<br>0.65<br>0.62<br>0.59<br>0.56         | 0.60  | 0.58                 | 0.55   | 0.54                   | 0.54         |
| 69             | 0.65   | 0.57  | 0.55                 | 0.53   | 0.51                   | 0.51         |
| 68<br>67<br>66 | 0.62   | 0.54  | 0.52                 | 0.50<br>0.47                                 | 0.48                   | 0.48         |
| 67             | 0.59   | 0.51  | 0.49                 | 0.47   | 0.46                   | 0.45         |
| 66             | 0.56   | 0.48  | 0.46                 | 0.44   | 0.43                   | 0.42         |
| 65             | 0.53   | 0.45  | 0.43                 | 0.41   | 0.40                   | 0.40         |
| 64             | 0.49   | 0.42  | 0.40                 | 0.38   | 0.37                   | 0.37         |
| 63             | 0.46   | 0.39  | 0.37                 | 0.35   | 0.35                   | 0.34         |
| 62             | 0.43<br>0.39<br>0.36                                 | 0.36  | 0.34                 | 0.33<br>0.30<br>0.27                         | 0.32                   | 0.31         |
| 61             | 0.39   | 0.33  | 0.31<br>0.28         | 0.30   | 0.30                   | 0.29         |
| 60             | 0.36   | 0.30  | 0.28                 | 0.27   | 0.26                   | 0.26         |
| 59<br>58<br>57 | 0.32<br>0.29<br>0.25                                 | 0.27  | 0.25                 | 0.24<br>0.21<br>0.19                         | 0.24                   | 0.23         |
| 58             | 0.29   | 0.24  | 0.23<br>0.20         | 0.21   | 0.21                   | 0.21         |
| 57             | 0.25   | 0.21  | 0.20                 | 0.19   | 0.18                   | 0.18         |
| 56             | 0.22<br>0.18<br>0.14                                 | 0.18  | 0.17                 | 0.16   | 0.16                   | 0.15         |
| 55<br>54       | 0.18   | 0.15  | 0.14                 | 0.13<br>0.11                                 | 0.13                   | 0.13         |
| 54             | 0.14   | 0.12  | 0.11<br>0.08         | 0.11   | 0.10                   | 0.10         |
| 53             | 0.11   | 0.09  | 0.08                 | 0.08   | 0.08                   | 0.08         |
| 52             | 0.07<br>0.03<br>0.00                                 | 0.06  | 0.06                 | 0.05   | 0.05                   | 0.05         |
| 51             | 0.03   | 0.03  | 0.03                 | 0.03   | 0.03                   | 0.03         |
| Note 1: For no | 1 (1 (1))  | 0.00  | 0.00                 | 1 to 100 minus the                           | 0 00<br>stabular PWI o | 0.00         |

Note 1: For negative values of Q<sub>U</sub> or Q<sub>L</sub>. PWL<sub>U</sub> or PWL<sub>L</sub> is equal to 100 minus the tabular PWL<sub>U</sub> or PWL<sub>L</sub>.

Note 2: If the value of Q<sub>U</sub> or Q<sub>L</sub> does not correspond exactly to a value in the table, use the next higher value.

502-31

Table 502-10
Payment Adjustment for Mainline Pavement Density

(PWL)

|           |       |     |     | ]   | Percen |       | ent - % |     |       |         |
|-----------|-------|-----|-----|-----|--------|-------|---------|-----|-------|---------|
| Estimated | n = 3 | n=4 | n = | n = | n =    | n = 8 | n = 10  | N = | n     | n = 18  |
| PWL       |       |     | 5   | 6   | 7      | to 9  | to 12   | 13  | =14   | and     |
|           |       | 100 |     |     | 1      |       | 1       | 1   | to 17 | greater |
| 100 to 81 | 100   | 100 | 100 | 100 | 100    | 100   | 100     | 100 | 100   | 100     |
| 80        | 100   | 100 | 100 | 100 | 100    | 100   | 100     | 100 | 100   | 99      |
| 79        | 100   | 100 | 100 | 100 | 100    | 100   | 100     | 100 | 99    | 98      |
| 78        | 100   | 100 | 100 | 100 | 100    | 100   | 100     | 99  | 99    | 98      |
| 77        | 100   | 100 | 100 | 100 | 100    | 100   | 99      | 98  | 98    | 97      |
| 76        | 100   | 100 | 100 | 100 | 100    | 99    | 99      | 98  | 97    | 96      |
| 75        | 100   | 100 | 100 | 100 | 100    | 99    | 98      | 97  | 97    | 95      |
| 74        | 100   | 100 | 100 | 100 | 100    | 98    | 98      | 96  | 96    | 94      |
| 73        | 100   | 100 | 100 | 100 | 99     | 98    | 97      | 96  | 95    | 93      |
| 72        | 100   | 100 | 100 | 99  | 99     | 97    | 97      | 95  | 94    | 92      |
| 71        | 100   | 100 | 100 | 99  | 98     | 97    | 96      | 94  | 93    | 92      |
| 70        | 100   | 100 | 99  | 98  | 98     | 96    | 96      | 94  | 93    | 91      |
| 69        | 100   | 100 | 98  | 98  | 97     | 95    | 95      | 93  | 92    | 90      |
| 68        | 100   | 100 | 98  | 97  | 96     | 94    | 94      | 92  | 91    | 89      |
| 67        | 100   | 100 | 97  | 96  | 96     | 94    | 94      | 91  | 90    | 88      |
| 66        | 100   | 99  | 97  | 96  | 95     | 93    | 93      | 90  | 89    | 87      |
| 65        | 100   | 99  | 96  | 95  | 94     | 92    | 92      | 90  | 88    | 86      |
| 64        | 99    | 98  | 96  | 94  | 94     | 92    | 91      | 89  | 88    | 85      |
| 63        | 99    | 98  | 95  | 94  | 93     | 91    | 90      | 88  | 87    | 84      |
| 62        | 99    | 97  | 95  | 93  | 92     | 90    | 89      | 87  | 86    | 83      |
| 61        | 98    | 96  | 94  | 92  | 91     | 89    | 89      | 86  | 85    | 82      |
| 60        | 98    | 95  | 94  | 92  | 91     | 89    | 88      | 85  | 84    | 81      |
| 59        | 97    | 95  | 93  | 91  | 90     | 88    | 87      | 84  | 83    | 80      |
| 58        | 97    | 94  | 92  | 90  | 89     | 87    | 86      | 83  | 82    | 79      |
| 57        | 96    | 93  | 91  | 89  | 88     | 86    | 85      | 82  | 81    | 78      |
| 56        | 95    | 92  | 90  | 89  | 87     | 85    | 84      | 81  | 80    | 77      |
| 55        | 95    | 92  | 90  | 88  | 86     | 84    | 83      | 79  | 79    | 76      |
| 54        | 94    | 91  | 89  | 87  | 85     | 83    | 82      | 78  | 77    | 75      |
| 53        | 93    | 90  | 88  | 86  | 85     | 82    | 80      | 77  | 76    | 74      |
| 52        | 92    | 89  | 87  | 85  | 84     | 81    | 79      | 76  | 75    | 72      |
| 51        | 91    | 88  | 85  | 84  | 83     | 80    | 78      | 74  | 74    | 71      |
| 50        | 90    | 88  | 84  | 83  | 82     | 79    | 77      | 74  | 73    | 70      |
| 49        | 90    | 87  | 83  | 82  | 81     | 77    | 76      | 72  | 71    | 69      |
| 48        | 89    | 86  | 82  | 81  | 80     | 76    | 74      | 71  | 70    | 67      |

502-32 275

| 47 | 88 | 85 | 81 | 80 | П       | 79    | 75       |         | 73   | П       | 70 | ( | 58 | 66 |
|----|----|----|----|----|---------|-------|----------|---------|------|---------|----|---|----|----|
| 46 | 87 | 84 | 80 | 79 | Ħ       | 77    | 74       | 1       | 72   | Ħ       | 68 | ( | 67 | 64 |
| 45 | 86 | 83 | 79 | 78 | I       | 76    | 73       | T       | 71   | Ħ       | 67 | ( | 66 | 63 |
| 44 | 85 | 82 | 78 | 77 | П       | 75    | 71       |         | 69   | Ħ       | 65 | ( | 54 | 62 |
| 43 | 85 | 81 | 77 | 76 | П       | 74    | 70       |         | 68   | Ħ       | 64 | ( | 63 | 60 |
| 42 | 84 | 80 | 76 | 75 | П       | 73    | 69       |         | 67   | I       | 63 | ( | 52 | 59 |
| 41 | 83 | 79 | 75 | 73 | П       | 71    | 68       |         | 65   |         | 62 | ( | 60 | 58 |
| 40 | 82 | 77 | 74 | 72 | П       | 70    | 66       |         | 64   | П       | 61 | 4 | 59 | 57 |
| 39 | 81 | 76 | 72 | 71 | П       | 69    | 65       |         | 63   |         | 59 | 4 | 57 | 55 |
| 38 | 80 | 75 | 71 | 70 |         | 67    | 63       |         | 61   |         | 58 | 4 | 56 | 54 |
| 37 | 79 | 74 | 70 | 68 |         | 66    | 62       |         | 60   |         | 56 | 4 | 55 | 52 |
| 36 | 78 | 73 | 68 | 67 |         | 65    | 61       |         | 58   |         | 55 | 4 | 53 | 51 |
| 35 | 77 | 72 | 67 | 66 |         | 63    | 60       |         | 57   |         | 53 | 4 | 52 | 50 |
| 34 | 76 | 71 | 66 | 65 |         | 62    | 58       |         | 55   |         | 52 | 4 | 50 |    |
| 33 | 75 | 70 | 65 | 63 |         | 61    | 57       |         | 54   |         | 50 |   |    |    |
| 32 | 74 | 69 | 63 | 61 |         | 60    | 55       |         | 52   |         |    |   |    |    |
| 31 | 73 | 67 | 62 | 60 | П       | 59    | 54       |         | 51   | Π       |    |   |    |    |
| 30 | 72 | 66 | 61 | 58 | П       | 57    | 52       |         | 50   |         |    |   |    |    |
| 29 | 71 | 65 | 59 | 57 | П       | 56    | 51       |         |      |         |    |   |    |    |
| 28 | 70 | 64 | 58 | 55 |         | 54    | 50       |         |      |         |    |   |    |    |
| 27 | 69 | 62 | 57 | 54 |         | 53    |          |         |      |         |    |   |    |    |
| 26 | 68 | 61 | 55 | 52 |         | 52    |          |         |      |         |    |   |    |    |
| 25 | 67 | 60 | 52 | 51 | П       | 50    |          |         |      |         |    |   |    |    |
| 24 | 66 | 59 | 50 | 50 |         |       |          |         |      |         |    |   |    |    |
| 23 | 64 | 58 |    |    |         |       |          |         |      |         |    |   |    |    |
| 22 | 63 | 56 |    | 50 | P       | ercei | nt or Re | en      | nove |         |    |   |    |    |
| 21 | 62 | 54 |    |    |         |       |          |         |      |         |    |   |    |    |
| 20 | 61 | 53 |    |    |         |       |          |         |      |         |    |   |    |    |
| 19 | 60 | 53 |    |    | $\prod$ |       |          |         |      |         |    |   |    |    |
| 18 | 59 | 52 |    |    |         |       |          |         |      |         |    |   |    |    |
| 17 | 58 | 50 |    |    |         |       |          |         |      |         |    |   |    |    |
| 16 | 57 |    |    |    |         |       |          |         |      |         |    |   |    |    |
| 15 | 56 |    |    |    | $\prod$ |       |          |         |      | $\prod$ |    |   |    |    |
| 14 | 54 |    |    |    |         |       |          |         |      |         |    |   |    |    |
| 13 | 53 |    |    |    |         |       |          |         |      |         |    |   |    |    |
| 12 | 52 |    |    |    |         |       |          |         |      |         |    |   |    |    |
| 11 | 51 |    |    |    | $\prod$ |       |          | $\prod$ |      |         |    |   |    |    |
| 10 | 50 |    |    |    |         |       |          |         |      | $\prod$ |    |   |    |    |

This page intentionally left blank

## **TECHNICAL PROVISIONS**

## **SECTION 706**

## CONCRETE WALKS, DRIVES AND INCIDENTAL PAVING

## TABLE OF CONTENTS

| 706.01 | DESCRIPTION               |
|--------|---------------------------|
| 706.02 | MATERIALS                 |
| 706.03 | CONSTRUCTION REQUIREMENTS |
| 706.04 | MEASUREMENT               |
| 706.05 | PAYMENT                   |

This page intentionally left blank

#### TECHNICAL PROVISIONS

## **SECTION 706**

## CONCRETE WALKS, DRIVES, AND

## INCIDENTAL PAVING

**706.01 DESCRIPTION.** Furnish and construct portland cement concrete walks, handicapped curb ramps, drives and incidental paving slabs in accordance with these specifications and in conformity with lines, grades, and dimensions shown on the plans or established.

**706.02 MATERIALS.** Materials shall comply with the following sections or subsections:

| Portland Cement Concrete (Class M) | 901       |
|------------------------------------|-----------|
| Joint Filler                       | 1005.01.3 |
| Reinforcing Steel                  | 1009.01   |
| Curing Materials                   | 1011.01   |

## 706.03 CONSTRUCTION REQUIREMENTS.

- **706.03.1 Excavation:** Excavate to required depth and width. Shape the top of the subgrade and compact to a firm, even surface conforming to the section shown on the plans. Remove unsuitable material and dispose of in accordance with 202.02 and replace with approved material at no direct pay.
- **706.03.2 Forms:** Forms shall be of wood or metal and shall extend the full depth of concrete. Forms shall be straight, clean, and of sufficient strength to resist the pressure of concrete. Brace forms to ensure that forms remain in horizontal and vertical alignment until their removal.

Concrete may be placed by slip-form methods. Place slip-formed concrete with an approved machine designed to spread, vibrate, consolidate, and finish concrete in one pass of the machine with a minimum of hand finishing. Rigidly hold sliding forms together to prevent spreading of forms. After the passing of the side forms, there shall be no noticeable slumping of concrete.

- **706.03.3 Subgrade:** Thoroughly moisten the subgrade immediately prior to placing concrete.
- **706.03.4 Placing and Finishing:** Place the concrete on the subgrade, strike-off to required thickness, and tamp sufficiently to bring the mortar to the surface. Finish the surface with a wood float or steel trowel followed by brushing to a slightly rough finish. Round joints and edges with an edging tool having a 1/4 inch radius.

## 706.03.5 Joints:

**706.03.5.1 Expansion Joints:** Fill expansion joints with 1/2 inch thick preformed expansion joint filler. Install expansion joints at maximum 100-foot intervals, and between

intersecting paving and any fixed structure, such as a building, bridge, or curbing, and between intersecting paving and the handicapped curb ramps. Extend expansion joint material for the full width and depth of paving.

**706.03.5.2 Weakened Plane:** Form weakened planes by a jointing tool or other acceptable means. Extend weakened planes into concrete for at least one-quarter of the depth. Weakened planes shall be approximately 1/8 inch wide.

**706.03.5.2.1 Walks:** Space weakened planes for walks equal to the width of the walk.

**706.03.5.2.2 Drives:** Form a longitudinal weakened plane along the centerline of drives more than 16 feet wide. Form transverse weakened planes at not more than 16-foot intervals.

706.03.5.2.3 Incidental Paving: Form weakened planes for incidental paving at intervals not exceeding 30 times the thickness of the concrete in length or width. Construct joints in incidental paving placed adjacent to jointed concrete to match existing joints, with intermediate joints formed as necessary not to exceed the maximum joint spacing.

**706.03.5.3 Construction Joints:** Form construction joints around manholes, utility poles, etc., extending into paving. Install 1/4 inch thick preformed expansion joint filler into these joints.

**706.03.5.4 Tie-ins:** Make tie-ins of existing concrete by full depth sawing at no direct pay.

**706.03.6** Curing: Cure concrete in accordance with 601.03.10.

**706.03.7 Detectable Warning Surface for Handicap Ramps and At-Grade Sidewalk Intersections:** When sidewalks intersect with roadways, equip the sidewalk with a detectable warning system consisting of raised truncated domes as a transition between the sidewalk and the street as required by the Americans With Disabilities Act, 28 CFR § Part 36, ADA Standards for Accessible Design. Install detectable warnings (truncated domes) on the ramp surface over the full width of the ramp throat for a distance of 24 inches in the direction of travel from the back of the curb. Also install detectable warnings (truncated domes) on at-grade sidewalks intersecting with roadways for a distance of 36 inches in the direction of travel from the end of the sidewalk. Detectable warning surfaces may be added to at-grade sidewalks intersecting with driveways at the discretion of the design section or Project Engineer. Lay out truncated domes on a square grid in order to allow enough space for wheelchairs to roll between the domes.

Light reflectance of the truncated domes and the underlying surface shall meet the 70 percent contrast requirement of ADAAG.

**706.04 MEASUREMENT.** Quantities of concrete walks, drives, and incidental paving slabs for payment will be the design quantities as specified on the plans and adjustments thereto. Design quantities will be adjusted if the engineer makes changes to adjust to field conditions, if design errors are proven or if design changes are made. Design areas are based on the horizontal dimensions shown on the plans. Excavation, backfill, reinforcing steel, and joint materials will not be measured for payment.

Handicapped curb ramps, including the detectable surface warning system, will be measured per each.

Detectable surface warning systems for at-grade sidewalk intersection will not be measured for payment.

**706.05 PAYMENT.** Payment for concrete walks, drives, and incidental paving will be made on a lot basis at the contract unit price per square yard, adjusted in accordance with the following provisions. Payment for each lot will be made in accordance with table 901-5. Size, sampling, and testing of each concrete lot shall be in accordance with the materials sampling manual.

Payment for handicapped curb ramps, including the detectable surface warning system, will be made at the contract unit price per each and shall include, but not be limited to, curb transitions, detectable warning system, gutter, landing, and base.

Payment will be made under:

| Item No. | Pay Item                   | Pay Unit    |
|----------|----------------------------|-------------|
| 706-01   | Concrete Walk              | Square Yard |
| 706-02   | Concrete Drive             | Square Yard |
| 706-03   | Incidental Concrete Paving | Square Yard |
| 706-04   | Handicapped Curb Ramps     | Each        |

This page intentionally left blank

## **TECHNICAL PROVISIONS**

## **SECTION 712**

## **REVETMENTS**

## TABLE OF CONTENTS

| 712.01 | DESCRIPTION               |
|--------|---------------------------|
| 712.02 | MATERIALS                 |
| 712.03 | CONSTRUCTION REQUIREMENTS |
| 712.04 | MEASUREMENT               |
| 712.05 | PAYMENT                   |

This page intentionally left blank

## TECHNICAL PROVISIONS

## **SECTION 712**

## REVETMENTS

**712.01 DESCRIPTION.** Furnish and construct revetments for protection of embankment slopes, stream channels, culvert end treatments, and other areas. Revetments shall be constructed in accordance with these specifications and in conformity with the details shown on the plans or as directed.

When including an item for Flexible Revetments in the contract, the contractor has the option of furnishing revetments of stone, recycled portland cement concrete, wet-batched sacked concrete, dry-batched prepackaged sacked concrete, or cabled articulated concrete block mattress.

When including an item for Stone Revetment in the contract, the contractor has the option of furnishing revetments of either stone or recycled portland cement concrete.

When including an item for Sacked Concrete Revetment in the contract, the contractor has the option of furnishing revetments of either wet-batched sacked concrete or dry-batched prepackaged sacked concrete.

Use the same type revetment at each location. Except for cast-in-place revetment, place revetments on geotextile fabric.

## 712.02 MATERIALS.

- **712.02.1 Concrete:** Concrete for cast-in-place revetment shall be Class R complying with Section 901.
  - **712.02.2 Geotextile Fabric:** Geotextile fabric shall comply with 1019.01.
- 712.02.3 Wet-Batched Sacked Concrete: Sacks shall comply with 1018.07. Concrete shall be Class R concrete complying with Section 901. Concrete shall be wet-batched prior to placement in sacks. Add mixing water to concrete as required to produce a slump of 4 inches to 6 inches. Immediately place the sacked concrete in the revetment after batching.
- 712.02.4 Stone and Recycled Portland Cement Concrete: Stone and recycled portland cement concrete shall comply with Section 711 Riprap Class 30 lb.
- 712.02.5 Dry-Batched Prepackaged Sacked Concrete: Use a prepackaged an Approved Materials List concrete product consisting of one part cement and a maximum of 5 parts sand by weight or other approved mix with the same cement content. Dry mix until uniform in color.
  - **712.02.5.1** Cement: Cement shall be in accordance with 901.08.1 and 1001.01.
- **712.02.5.2 Aggregate:** Aggregates shall comply with the gradation requirements of 1003.08.
- **712.02.5.3 Sacks:** Sacks shall comply with 1018.07 and shall be capable of holding the concrete mixture without leakage during handling.
  - 712.02.5.4 Water: Water shall be from an approved source in accordance with 1018.01.

- **712.02.6 Usable Soil:** Usable soil shall be in accordance with 203.06.1.
- 712.02.7 Cabled Articulated Concrete Block Mattress: Cabled articulated concrete block mattresses shall consist of interlocking cellular concrete blocks connected with cables to bind the individual blocks into mattresses. The dimensions of the finished mattress shall be as shown on the plans or as directed by the engineer. Size and arrange holes for cable penetrations to minimize exposure of cables to potential environmental degradation. Do not pass cables through open areas within the dimensions of individual blocks. Geotextile fabric, as specified in 712.02.2, may be glued to the bottom of the mattress in lieu of placing the geotextile in the trench prior to placement.
- 712.02.7.1 Cellular Concrete Blocks: The cellular concrete blocks shall be interlocking, and capable of articulation when formed into mattresses. Furnish open or closed cell blocks as shown on the plans. Ensure that concrete used to make the blocks meets the requirements of Section 901 and has a compressive strength of 4000 psi at 28 days. Use concrete aggregates meeting the requirements of 1003.08 and listed on the Approved Materials List.
- 712.02.7.2 Cable: Use galvanized steel or continuous filament polyester fiber cable to connect the blocks to form a mattress. Ensure that the cable has adequate tensile strength to lift and handle the mattress safely. Permanent deformation of the cable and mattress due to elongation and elasticity of the cable during handling and placement shall be negligible.
- 712.03 CONSTRUCTION REQUIREMENTS. Construct revetments in dry or dewatered areas, unless otherwise directed. Remove logs, stumps, and other undesirable material from areas on which revetments are to be placed. Provide usable soil to bring areas to grade and compact to the density of surrounding ground to the engineer's satisfaction before final grading. Grade the revetment areas to required sections.
- 712.03.1 Geotextile Fabric Placement: Bury ends of geotextile fabric for anchorage as shown on the plans. Lap adjacent strips of geotextile fabric at least 18 inches. Pin the laps at maximum 5-foot intervals. Do not damage geotextile fabric during revetment placement.

Repair damaged fabric in accordance with 203.11.3 or replace at no direct pay.

712.03.2 Concrete Cast-in-Place Revetment: Before concrete placement, place preformed 1/4 inch thick expansion joint filler complying with 1005.01.1 around piles, columns, or other structural elements as directed.

Commence placing concrete revetment for slope protection at the toe of revetment and progress upslope. Place revetment for stream channels and other relatively level areas as directed.

After placement, cure the revetment surface in accordance with 601.03.10.

- 712.03.3 Wet-Batched Sacked Concrete Placement: Uniformly fill sacks to approximately 3/4 cubic foot. Fold the open end under the bag during placement. Place sacks of wet-batched concrete in one layer in contact with adjacent sacks and tamp into position by approved methods. Begin placement of sacked concrete at the revetment toe and progress upslope. Place sacked concrete revetment for stream channels and other relatively level areas as directed.
- 712.03.4 Dry-Batched Prepackaged Concrete Placement: Uniformly fill sacks to approximately 3/4 cubic foot and seal the ends by tying, stitching, or other approved methods. Tightly pack the filled sacks against each other. Begin placement at the revetment toe and

progress upslope with staggered joints. At the end of each day's operations and upon completion at a location, saturate the sacks and contents with water. The quantity of water required shall be as directed at no direct pay. The compressive strength shall comply with Section 901 for Class R concrete.

712.03.5 Stone Placement: Construct toe and end walls by placing stone in the trench lined with geotextile fabric. Begin placement of stone at the bottom of the slope in a layer having the specified average thickness. Place stone by approved methods. A tolerance of 2 inches above or below the specified thickness will be allowed. Openings between stones exposing more than 4 square inches of geotextile fabric will not be permitted.

712.03.6 Cabled Articulated Concrete Block Mattress Placement: Sufficiently excavate the area in which the mattress is to be placed to ensure that the mattress is resting on the bottom of the trench. Excavate the trench to an elevation 6 inches below the grade line shown on the plans. The mattress placement direction shall be as shown on the plans or as directed by the engineer. Where requiring more than one width or length of mattress, bind adjacent mattresses together according to plan details to the satisfaction of the engineer.

**712.04 MEASUREMENT.** Revetments will be measured by the square yard of surface area to be re-vetted as shown on the plans and as directed. Site preparation, geotextile fabric, and expansion joint filler will not be measured for payment. Excavation, cables, and ties for cabled articulated concrete block mattress will not be measured for payment.

712.05 PAYMENT. Payment for all revetments will be made at the contract unit price per square yard, which includes furnishing and installing all materials as shown on the plan details and in accordance with this section. Payment for concrete cast-in-place revetment, wet, and dry batched sacked concrete revetment, will be made on a lot basis at the contract unit price per square yard, adjusted in accordance with table 901-5. Size, sampling, and testing of each concrete lot shall be in accordance with the materials sampling manual. Payment for cabled articulated concrete block mattress will be made per square yard.

Payment will be made under:

| Item No. | Pay Item                                   | Pay Unit    |
|----------|--|-------------|
| 712-01   | Concrete Cast-in-Place Revetment           | Square Yard |
| 712-02   | Sacked Concrete Revetment                  | Square Yard |
| 712-03   | Stone Revetment                            | Square Yard |
| 712-04   | Flexible Revetment                         | Square Yard |
| 712-05   | Cabled Articulated Concrete Block Mattress | Square Yard |

This page intentionally left blank

# **TECHNICAL PROVISIONS**

# **SECTION 713**

# TEMPORARY TRAFFIC CONTROL

# TABLE OF CONTENTS

| 713.01 | DESCRIPTION                                |
|--------|--|
| 713.02 | MATERIALS                                  |
| 713.03 | FABRICATION                                |
| 713.04 | TEMPORARY SIGNS AND BARRICADES             |
| 713.05 | PORTABLE CHANGEABLE MESSAGE SIGNS          |
| 713.06 | TEMPORARY PRECAST CONCRETE BARRIERS        |
| 713.07 | PAVEMENT MARKINGS                          |
| 713.08 | PORTABLE WORK ZONE TRAFFIC CONTROL DEVICES |
| 713.09 | TRAFFIC CONTROL MANAGEMENT                 |
| 713.10 | NIGHTTIME CONSTRUCTION OPERATIONS          |
| 713.11 | MEASUREMENT                                |
| 713.12 | PAYMENT                                    |

This page intentionally left blank

#### **TECHNICAL PROVISIONS**

#### **SECTION 713**

#### TEMPORARY TRAFFIC CONTROL

713.01 DESCRIPTION. Furnish, install, maintain, and remove temporary construction barricades, precast concrete barriers, lights, signals, pavement markings, and signs; provide flaggers; and comply with all other requirements regarding the protection of the work, workers, and safety of the public. Unless otherwise noted in the plans or special provisions, this work also provides for traffic control management in compliance with the *Manual On Uniform Traffic Control Devices* (MUTCD), including the installation, inspection, maintenance, and removal of all traffic control devices relative to work on the project. Signs, barricades, barriers, channelizing devices, pavement markings, or any other temporary traffic control measures shall comply with plan details, *Temporary Traffic Control Standards*, the MUTCD, and these specifications.

Signs, barricades, barriers, channelizing devices, pavement markings, and arrangements thereof, as shown on the plans, are minimum requirements. Furnish and install appropriate signs for special conditions as directed. Requirements for proper signs, barricades, barriers, channelizing devices, or other safety precautions promulgated by the contractor's insurers are not negated by these specifications. These specifications shall not be construed to relieve the contractor of responsibilities for the safety of the public, for liability in connection therewith, or compliance with state and local laws or ordinances.

Assign one or more authorized Traffic Control Supervisors (TCS) to provide traffic control management for the project. If assigning more than one TCS, then submit a weekly schedule identifying who will be in charge of providing traffic control management on a daily basis to the engineer. The TCS shall have a set of all contract documents relating to traffic control (and traffic staging if applicable), a current copy of the MUTCD, and a current copy of Temporary Traffic Control Standards readily available at all times.

If a subcontractor provides traffic control management, the subcontractor's TCS shall meet all the requirements set forth herein.

The contractor may assign one or more Traffic Control Technicians (TCT) to assist the TCS in inspection and maintenance of Traffic Control Devices.

**713.02 MATERIALS.** Materials for temporary signs, barricades, barriers, and related devices shall comply with the following sections and subsections:

| Portland Cement Concrete    | 901       |
|-----------------------------|-----------|
| Reinforcing Steel           | 1009.01   |
| Backing Material            | 1015.04.2 |
| Reflective Sheeting         | 1015.05   |
| Temporary Pavement Markings | 1015.08   |

| Raised Pavement Markers & Adhesive | 1015.09 |
|------------------------------------|---------|
| Thermoplastic Pavement Markings    | 1015.10 |
| Traffic Paint                      | 1015.12 |
| Barricade Warning Lights           | 1018.13 |

- **713.02.1 Temporary Pavement Markings:** Temporary pavement markings shall be a minimum of 4 inches wide.
- 713.02.2 Reflective Sheeting: Reflective sheeting requirements for temporary signs, barricades, channelizing devices, drums, and cones shall comply with the following:
- **713.02.2.1 Temporary Signs and Barricades:** On the mainline of freeways and expressways, fabricate the initial advanced warning construction sign using DOTD Type X (Fluorescent Orange) reflective sheeting. Reflective sheeting for all other temporary signs and barricades shall comply with the requirements of ASTM D 4956, Type III.
- **713.02.2.2 Vertical Panels:** Reflective sheeting for vertical panels used to channelize or divide traffic shall meet the requirements of ASTM D 4956, Type III.
- 713.02.2.3 Drums and Supercones: Reflective sheeting for drums and supercones shall meet the requirements of ASTM D 4956, Type III, and the Supplementary Requirement S2 for Reboundable Sheeting as specified in 1015.05.6.
- **713.02.2.4 Traffic Cones:** Reflective sheeting for traffic cones shall meet the requirements of ASTM D4956, Type III or VI.
- **713.03 FABRICATION.** Fabricate temporary signs, barricades, and related devices according to 729.04. Label back of signs with name of fabricator, date fabricated and the department project number. Fabricate precast concrete barriers according to Section 805.

#### 713.04 TEMPORARY SIGNS AND BARRICADES.

713.04.1 General: When work is in progress, furnish and install temporary signs, barricades, and related devices on portions of the work covered by the Notice to Proceed or when operations are suspended. During such times that temporary signs, barricades, and related devices are not in place, maintain required traffic control devices. Do not begin construction work until signs, barricades, and other traffic control devices have been erected and approved.

The contractor's Traffic Control Supervisor (TCS) shall coordinate with the engineer before removing or covering any signs that conflict with temporary traffic control signs.

When placing signs, coordinate with the engineer in removing Departmental signs, so that appropriate signs are in place at all times.

Signing shall remain in place and be maintained by the contractor, supplemented by additional signs as required, throughout the period of work. When erecting previously used signs on a project, the engineer must inspect and approve these signs before erection. Remove all signs with reduced retroreflectivity or excessive color fading from the work zone. In case of a dispute over a rejected used sign, the engineer may take such measurements or review

retroreflectivity and color data obtained by the contractor to determine if the sign meets minimum standards for new materials. Replace signs that do not meet the minimum standards for new materials at no direct pay.

Rejected signs will be marked on the back "NOT FOR USE ON STATE PROJECTS."

Remove signs placed by the contractor according to the Traffic Control Plan. The Department will ensure that all permanent highway signs are in place upon completion and prior to final acceptance of the project.

On projects where constructing the surface course with asphalt concrete or portland cement concrete, install permanent striping and raised pavement markers (when required) prior to removal of barricades.

Upon removal, signs, barricades, and related devices furnished and placed by the contractor shall remain the contractor's property.

**713.04.2 Advance Warning Area:** When specified, provide Type C arrow boards for temporary traffic control at locations shown on the plans or as directed.

**713.04.3 Sign Supports:** Mount signs a minimum of 5 feet above the higher of the roadway or the ground clearance. In urban areas, mount the signs a minimum of 7 feet above the roadway.

**713.05 PORTABLE CHANGEABLE MESSAGE SIGNS.** Furnish, operate, and maintain solar powered portable changeable message signs at all locations designated on the plans or as directed.

The portable changeable message sign shall be in good operational condition when delivered to the job site. The engineer will inspect the signs; if they are found to be in good operational condition with all working parts functioning, the signs will be approved for use on the project.

The message sign shall consist of three separate lines. Each line shall consist of eight characters. Each character shall nominally be 18 inches in height. The width shall be adequate to meet the below legibility requirements. Each character shall be a 5 x 7 LED module or hybrid LED disk. Characters shall be separated at a distance such that the legibility requirements are maintained.

All internally illuminated portions of the sign shall be amber in color. Externally illuminated surfaces meant for message display shall be fluorescent yellow. Non-illuminated surfaces on the front panel shall be flat black in color.

The sign shall be clearly visible under all conditions and all lanes of travel from a distance of 1000 feet perpendicular to the sign center. The sign shall maintain this legibility throughout the entire project. The contractor shall be responsible for maintaining this minimum legibility. Determination of legibility distance shall rest solely with the engineer.

Use the portable changeable message sign in conjunction with other traffic signs and devices in accordance with the plans, project specifications, and as directed by the engineer. Messages shall be approved by the engineer.

Store the signs in an approved secure storage area when not in use. Perform all maintenance operations recommended by the manufacturer and keep adequate records of such operations.

Keep the signs clean and in good repair at all times.

**713.06 TEMPORARY PRECAST CONCRETE BARRIERS.** Barrier units shall be furnished by the contractor unless specified otherwise. Each barrier unit shall be 15 feet in length.

When the barrier units are furnished by the Department, the units will be furnished at no cost to the contractor. The contractor shall load the barrier units at the location specified, deliver the units to the construction site, and place them as required.

The contractor shall relocate barrier units as required during construction.

Furnish connecting pins and plastic reflectors at no additional cost to the Department. Reflectors shall have 7.0 square inches minimum reflective area. Install a maximum of 15 feet apart (each side) and in accordance with the manufacturer's recommendations. Replace damaged pins or reflectors as directed by the engineer.

After completion of the work, barrier units furnished by the Department shall be removed, transported by the contractor to the location specified, and unloaded as directed. All costs of loading, transporting, and unloading the barrier units shall be included in the contract price for this item. Barrier units furnished by the contractor shall, upon removal, remain the contractor's property. Satisfactorily repair or replace damaged barrier units at no direct pay.

**713.07 PAVEMENT MARKINGS.** Color, width, and type of temporary pavement markings shall be in accordance with Table 713-1 and the MUTCD. Temporary pavement markings shall be in place by the end of each day's operation.

Apply temporary striping tape by approved methods to the satisfaction of the engineer. Apply thermoplastic pavement markings in accordance with 732.03. Apply painted traffic striping in accordance with Section 737.

Table 713-1
Temporary Pavement Marking 12,3

|                                 | remporary ravement warking               |  |   |  |
|---------------------------------|--|--|---|--|
|                                 |  | Two-lane<br>Highways   | Undivided<br>Multilane<br>Highways  | Divided<br>Multilane<br>Highways                             |
| S<br>H<br>O<br>R<br>T           | All ADT's with time <7 days              | Lane lines 4-ft tape on<br>40-ft centers; with<br>no passing zone<br>markings. "Do Not<br>Pass" and "Pass With<br>Care" signs as required  | Lane lines 4-ft tape<br>on 40-ft centers;<br>dbl yellow<br>centerline               | Lane lines 4-ft<br>tape on 40-ft<br>centers                  |
| T<br>E<br>R<br>M                | All ADT's with time >7 days and ≤30 days | Lane lines 4-ft tape on 40-ft centers with no passing zone markings and no edgelines. "Do Not Pass" and "Pass With Care" signs as required | Lane lines 4-ft tape<br>on 40-ft centers; dbl<br>yellow centerline<br>and edgelines | Lane lines 4-ft<br>tape on 40-ft<br>centers and<br>edgelines |
| L<br>O<br>N<br>G<br>T<br>E<br>R | All ADT's with time >30 days             | Standard 10-ft lane lines, no-passing zone markings; when pavement width is ≥22 ft, edge lines   | Standard 10-ft<br>lane lines,<br>centerlines, edge<br>lines                         | Standard 10-ft<br>lane lines,<br>centerlines, edge<br>lines  |

No-passing zones shall be delineated as indicated whenever a project is open to traffic.

# **713.07.1 Short-term Pavement Markings:** Provide short-term pavement markings on all pavement surfaces under traffic.

Install temporary striping tape a minimum of 4 feet long on a maximum of 40-foot centers on centerlines of two-lane highways and lane lines of multilane highways. When short-term pavement markings require no-passing zone markings or double yellow centerlines on undivided multilane highways, use any of the temporary pavement markings listed in 713.02.

Removal of short-term pavement markings only required on the final surface.

On all Asphalt Surface Treatments that are open to traffic and used as a final wearing course or as an interlayer, temporary pavement markings (tabs) on 20-foot centers shall be used in lieu of the 4-foot tape on 40-foot centers.

A \$150 per day penalty will be assessed the contractor if Table 713-1 is not adhered to.

713.07.2 Long-term Pavement Markings: Provide long-term pavement markings on all surfaces not covered by an additional surface within two weeks. Long-term pavement markings shall include, but are not limited to, standard lane and centerline markings, edge lines, no passing zone markings on two-lane highways, stop bars, and legend and symbol markings as shown on the permanent pavement marking details. Layout work for exact location of markings will only be required on the final wearing surface.

These markings include all of the pavement markings listed in 713.02. Long-term markings do not include the installation of raised pavement markers.

**713.07.3 Final Surface:** On the final surface of portland cement concrete pavement or asphalt concrete pavement, place temporary markings with sufficient accuracy to avoid conflict with permanent striping. Temporary pavement markings on the final surface shall be any of the pavement markings listed in 713.02.

Place permanent markings over traffic paint on final surfaces provided the temporary markings have been placed in the final configuration and the painted lines are not flaking or showing signs of deterioration.

When required, remove temporary pavement markings in accordance with the requirements for the type of permanent marking being used. No objectionable staining of pavement surface as a result of the removal procedure will be allowed.

- 713.07.4 Temporary Reflectorized Raised Pavement Markers: When required, install temporary reflectorized raised pavement markers in accordance with Section 731.
- 713.07.5 Pavement Markings for Asphalt Surface Treatment: The type of markings shall be in accordance with Table 713-1. Put short-term temporary pavement markings in place at the end of each day's operation. Put long-term temporary pavement markings in place as soon as practical after expiration of the four-day maintenance period following the asphalt surface treatment operation. On the final wearing course, place permanent markings within two weeks following completion of the long-term temporary pavement markings.

When used on the final wearing course, painted traffic striping shall be in accordance with Section 737.

Install temporary raised markers on centerlines of two-lane highways and lane lines of multilane highways in accordance with 1015.08.3. "No passing zone" markings shall be any of the temporary pavement markings listed in 713.02.

Install the temporary raised pavement markers in accordance with the manufacturers' recommendations or as directed by the engineer. Place temporary raised markers consisting of flexible reflective tabs at 20-foot intervals on the centerline of the roadway. Install the markers so that the reflective faces of the markers are perpendicular to a line parallel to the roadway centerline.

If directed by the engineer, remove the temporary raised pavement markers after permanent striping has been accomplished. Repair damage to the pavement surface at no direct pay.

**713.08 PORTABLE WORK ZONE TRAFFIC CONTROL DEVICES.** All Category I, II, and III portable work zone traffic control devices, as described below, shall be crashworthy as determined by evaluations through the National Cooperative Highway Research Program

(NCHRP) 350 for Test Level 3 (TL-3) or the American Association of State Transportation Officials (AASHTO) Manual for Assessing Safety Hardware (MASH).

713.08.1 Category I Devices: Category I devices are low-mass, single-piece traffic cones, tubular markers, single-piece drums, and flexible delineators. By definition, they are considered crashworthy devices meeting NCHRP Report 350 or MASH criteria. Drum and light combinations with Type A or C warning lights and vandal resistant fastener hardware are included as Category I devices. In lieu of testing for crashworthiness, acceptance of Category I devices for compliance with NCHRP 350 or MASH will be allowed based on self-certification by the supplier. The supplier shall certify that the product is crashworthy in accordance with the evaluation criteria of NCHRP 350 or MASH. This certification may be a one-page affidavit signed by the supplier, with supporting documentation kept on file to be furnished if requested.

713.08.2 Category II Devices: Category II devices include other low mass traffic control devices such as portable barricades either with or without lights or signs, portable sign stands, temporary sign posts, portable vertical panel assemblies, and drums with lights not meeting the drum and light combination requirements for Category I. Individual crash testing is required for Category II devices. FHWA letters of approval shall serve as verification that these devices comply with the crash testing requirements of NCHRP Report 350 or MASH. Provide for the engineer a listing of all the Category II devices to be used on the project prior to installation including a reference to the FHWA Work Zone letter number for each device. Also certify that each device has been crash tested and meets the NCHRP 350 or MASH requirements.

713.08.3 Category III Devices: Category III devices include massive devices such as temporary concrete barriers, water filled barriers, and temporary attenuators. Individual crash testing is required for Category III devices. FHWA letters of approval shall serve as verification that these devices comply with the crash testing requirements of NCHRP Report 350 or MASH. Provide for the engineer a listing of all the Category III devices to be used on the project prior to installation including a reference to the FHWA Work Zone letter number for each device. Also certify that each device has been crash tested and meets the NCHRP 350 or MASH requirements.

#### 713.09 TRAFFIC CONTROL MANAGEMENT.

**713.09.1 Authorization:** Prior to commencing work requiring traffic control management, submit to the engineer proof of the Traffic Control Supervisor's (TCS) and Traffic Control Technician's (TCT) current authorizations.

713.09.1.1 Traffic Control Supervisor (TCS) Authorization: The Department will accept the TCS authorization of other approved agencies or firms only if all of the following minimum TCS requirements are met:

- 1. Successful completion of a work zone traffic control supervisor course approved by the Department.
  - 2. Passing a written examination on the work zone traffic control supervisor course.
- 3. A minimum of one year full-time field experience, verified by the agency or firm, in work zone traffic control. This experience may be verified by the Department at its discretion.
  - 4. A TCS refresher course is required every 4 years.

- **713.09.1.2 Traffic Control Technician (TCT) Authorization:** The Department will accept the TCT authorization of other approved agencies or firms only if all of the following minimum requirements are met:
- 1. Successful completion of a work zone traffic control technician course approved by the Department.
  - 2. Passing a written examination on the work zone traffic control technician course.
  - 3. A TCT refresher course is required every 4 years.
- 713.09.2 Traffic Control Supervisor (TCS) Duties: The TCS shall be responsible for traffic control management. The TCS shall be available to the engineer to address traffic control management issues as needed. The following is a listing of the TCS's primary duties:
- 1. Personally provide traffic control management and supervision services at the project site. The TCS may have other assigned duties, but shall be readily available at all times to perform TCS duties as required in the contract. A minimum of one TCT or TCS shall be required on site during working hours, except the following where a TCS shall be onsite at all times during working hours:
  - \* freeways, expressways, and interstates
  - \* multilane roads with posted speeds of 45 mph and greater
  - \* other roadways with ADT equal to and greater than 25,000.
- 2. Be responsible for observing and evaluating both the daytime and nighttime performance of all traffic control devices installed on the project, in accordance with the Traffic Control Plan (TCP). Ensure that the devices are performing effectively as planned for both safety and traffic operations purposes. Do this inspection upon the initial installation of the devices and when any changes are made. This is in addition to the inspection of traffic control required in 713.09.5.
- 3. Be responsible for revisions requested by the contractor to the traffic control plan established in the contract and submit the new traffic control plan in accordance with 713.09.3.
- 4. Be responsible for the training of flagging personnel. This training will ensure that all flagging done on the project is in compliance with the MUTCD Part VI and Louisiana Work Zone Traffic Control Details. Flaggers shall be re-qualified every 4 years.
- 5. Coordinate all traffic control operations for the duration of the contract, including those of subcontractors, utility companies, and suppliers, to ensure that all traffic control is in place and fully operational prior to the commencement of any work. The Department recognizes that the contractor does not have direct control over the traffic control operations of the utility companies. The coordination provided by the TCS when dealing with utility companies is specifically for the purpose of coordinating concurrent utility traffic control with any other construction traffic control to avoid conflicts.
- 6. Coordinate, in writing, all project activities with the appropriate law enforcement, fire control agencies, and other appropriate public agencies as determined at the pre-construction conference by the engineer. Also invite the above agencies to the pre-construction conference.

- 7. The Department, in collaboration with the TCS, will prepare and submit statements concerning road closures, delays, and other project activities to the news media on a weekly basis or more often as needed. Submit news releases to the engineer for review and approval prior to the Department's submittal to the news media.
- 8. Notify the engineer, or designee, immediately of all vehicular accidents and/or incidents related to the project traffic control. Document the time and date of notification in the traffic control diary. Also monitor and document queues that occur as necessary.
  - 9. Attend the pre-construction conference and all project meetings.
- 10. Assume responsibility for the maintenance, cleanliness, replacement, and removal of traffic control devices of the existing traffic control plan during working and non-working hours.
- 713.09.3 Traffic Control Plan Revisions: Make requests for revision in the traffic control plan in writing to the engineer a minimum of 14 calendar days in advance of the needed revision. If the requested revision falls within the scope of the existing contract drawings, the engineer may approve the revision. If the engineer determines that the requested revision is outside the scope of the contract drawings, the contractor will be required to submit a change order request. The change order drawings shall conform to the following:
- 1. Letter size original contract drawings. Submit change order drawings on high quality, white  $8\,1/2\,x\,11$  inch letter size paper. The drawings may be hand drafted or computer drafted and arranged in landscape format on the page. The text and drawings must be legible after reproduction on standard reproduction equipment. Left, bottom, and right hand margins shall be at least 1/2 inch and the top margin shall be 1 inch.
- 2. Full size original contract drawings. Submit change order drawings on high quality, 4-mil, double-matte film using a plotting or reproduction process that fuses the graphics to ensure durability. Repeated handling and friction due to stacking of plans shall not smear, flake or rub off the graphics. Improper plotter settings and plotter wear may cause inconsistent durability of the drawings. Test samples of the submitted drawings for durability. Advance samples of matte films may be submitted for approval; however, the contract plans will be tested separately. Failures will result in rejection of the submittal. Drawing sizes shall comply with 801.05.2.2.1.
- 3. Lettering on change order drawings shall be of adequate size to facilitate a 50 percent reduction of plans. Make additions or changes with a permanent type of waterproof ink made for this purpose. If revised cross-sections are required, plot the cross-sections on standard cross-section sheets. As a minimum, draw the ground line, centerline elevation, and station numbers in ink; the remaining information may be drawn in pencil.

Regardless of size, identify all required change order drawings and documents with the DOTD project title and project number. Sign and seal all plans and calculations by a Louisiana licensed civil engineer.

All plans submitted by the contractor shall conform to these specifications and standards. The DOTD Chief Engineer may reject any plans not conforming to these standards.

Revisions to the TCP that are determined to be outside the scope of the original contract drawings must be approved by the DOTD District Traffic Engineering Division prior to implementation of the requested revision. In some cases on high traffic routes or high priority projects, the revisions must be approved by the HQ Traffic Operations Engineer.

713.09.4 Traffic Control Diary: The TCS shall maintain a project traffic control diary using the Department's Site Manager Program. As a requirement of 713.09.2.8, keep the traffic control diary current on a daily basis and electronically sign each daily entry. A date stamp is required on each diary, so it is imperative to complete these diaries in a timely manner. Completion and maintaining of the daily diaries in accordance with the plans and specifications is subject to the LA R.S. 14:133 "Filing or Maintaining False Public Records." Photographs and videotapes may be used to supplement the written text.

Make the traffic control diary available at all times for inspection by the engineer. Review the diary with the engineer on a weekly basis and submit a copy to the engineer on a monthly basis. Failure to complete the diary on a daily basis or make the diary available for review shall result in a deduction from payments for the work of \$150 per calendar day as stipulated damages for each day the diary is not completed or maintained. On days when the Department's Site Manager Program is unavailable, either due to location or operation, the TCS shall make arrangements with the approval of the Project Engineer to submit the traffic control diaries daily. Failure to submit the monthly copy of the diary to the engineer shall result in the withholding of the next partial payment until the past due diaries are submitted. Submitted diaries that indicate that contemporary daily record keeping has not been maintained, as determined by the engineer, the Department's Work Zone Engineer, or the Department's Statewide Traffic Control Specialist shall result in a deduction of \$150 for each such deficiency as stipulated damages from payments for the work. The lack of a weekly review by the engineer shall not relieve the contractor from the assessment of stipulated damages for its failure to maintain a daily traffic control diary. The traffic control diary is part of pay item 713 and shall become the property of the Department at the completion of the project.

The contractor, with the approval of the engineer, the Department's Work Zone Engineer, or the Department's Statewide Traffic Control Specialist, may cease the requirement for a traffic control diary when:

- 1. The project has been partially accepted and/or no remaining work exists on the project site that impacts the travelling public or,
  - 2. When all signs and barricades are removed at the conclusion of the project.

When referring to the daily completion of the dairy, it is meant that the TCS shall complete the diary by the end of the following day except as follows:

- 1. If the contractor does not work on Saturday, the Friday and Saturday diaries shall be entered into SiteManager no later than Monday morning at 9:00 am.
- 2. If the contractor works on Saturday, the Friday diary shall be entered by the end of the day Saturday, but the Saturday diary shall be entered no later than Monday at 9:00 am. The Sunday diary shall still be entered by the end of the day Monday.

713.09.5 Inspection of Traffic Control: The TCS shall be responsible for the inspection of all traffic control devices every calendar day that traffic control devices are in use. This inspection may be delegated to the TCT, except for the conditions described in 713.09.2.1 above, where the TCS shall conduct the inspections himself. Regardless, the TCS shall be stationed within one hour of the jobsite. Use the *Quality Guidelines for Work Zone Traffic Control Devices* standard by the American Traffic Safety Services Association (ATSSA) to evaluate the condition of the traffic control devices to determine if they are acceptable for use. Provide for the immediate repair, cleaning, or replacement of any traffic control devices not functioning as required to ensure the safety of the motorist and construction personnel and/or not meeting the ATSSA standard.

Conduct inspection of the traffic control devices by the TCS at the beginning and end of each workday, and as scheduled or directed by the engineer during the workday. Inspect the traffic control devices by the TCS on weekends, holidays, or other non-work days at least once per day. Inspect traffic control devices by the TCS at least once a week during nighttime periods and the same night after any modifications or changes have been made in the traffic control devices.

713.09.6 Failure to Comply: The engineer, the Department's Work Zone Engineer, or the Department's Statewide Traffic Control Specialist may suspend all or part of the contractor's operation(s) for failure to comply with the approved "Traffic Control Plan" or failure to correct unsafe traffic conditions within a reasonable period of time after such notification is given to the contractor in writing. If major traffic control deficiencies require immediate corrective action for the safety of the travelling public, the engineer, the Department's Work Zone Engineer, or the Department's Statewide Traffic Control Specialist may completely suspend the contractor's operations. This suspension can either be spoken or written, but if spoken, shall be followed up in writing as soon as practical. The Department reserves the right to revoke or de-certify the TCS for gross neglect of these duties. At this point, the TCS shall retake a Department approved TCS course and shall be subject to a 90-day probationary period at the discretion of the Department.

In the event that the contractor does not take appropriate action to bring the deficient traffic control into compliance with the approved traffic control plan or to correct the unsafe traffic conditions, the Department may proceed with the corrective action using its own forces, and such costs will be deducted from payments owed to the contractor.

If the contractor's operations are suspended, the normal assessment of contract time will not cease for the period required to correct these unsafe conditions and traffic control deficiencies. The contractor will not be relieved of the responsibility to provide traffic control safety to the traveling public when a project is under full or partial project suspension. When a project is under suspension due to the contractor's failure to comply with this section, or when the contract is under stipulated damages, continue to provide traffic control management. No additional measurement or payment will be made. If suspensions or partial suspensions are requested by the contractor, the additional traffic control management costs will be at no expense to the Department.

**713.09.7 Engineer Modifications:** The provisions included in the plans and specifications for handling and controlling traffic during construction may be changed by the engineer, with the approval of the DOTD District Traffic Operations Engineer, due to actual field conditions

encountered. Such changes will be made by written instruction to the contractor and be considered an amendment of the plans and specifications as of the date of the change.

#### 713.10 NIGHTTIME CONSTRUCTION OPERATIONS.

**713.10.1 Description:** This work consists of furnishing, installing, operating, maintaining, moving, and removing portable light towers and equipment-mounted fixtures for nighttime construction operations. Nighttime construction operations are defined as work performed after sunset and before sunrise.

**713.10.2 Equipment Requirements:** Materials and equipment shall be in good operating condition and in compliance with applicable OSHA, NEC, and NEMA codes.

The contractor shall furnish, to the engineer, two light meters capable of measuring the level of illuminance. These light meters will be used by the engineer to check the adequacy of illumination throughout the nighttime construction operations. The light meters will become the property of the contractor after final acceptance.

Suitable brackets and hardware shall be provided to mount lighting fixtures on equipment and machinery. Mountings shall be designed so that light fixtures can be positioned as necessary to reduce glare and provide the required illumination. Mounting brackets and fixtures shall not interfere with the equipment operator or any overhead structures and shall be securely connected to the fixtures to ensure minimum vibration.

Equipment-mounted systems shall be attached to construction equipment to provide Level II and Level III illuminance. Equipment mounted lighting shall be designed and positioned to be operated independently of general illumination.

Portable systems may consist of ground-mounted, trailer-mounted, or equipment mounted light towers. Portable light towers shall be sturdy and free-standing without the aid of guy wires or bracing. Towers shall be capable of being moved as necessary to keep pace with the construction operation. Extreme caution shall be used when moving portable light towers in the vicinity of overhead utilities. Portable lighting systems shall be positioned to minimize the risk of being impacted by traffic on the roadway or by construction equipment.

Conventional vehicle headlights shall not be permitted as the sole means of illumination while working. All motorized vehicles shall be equipped with conventional vehicle headlights to permit safe movement in non-illuminated areas. Use of strobe lights on vehicles and equipment is prohibited. Use of flashing lights shall be kept to a minimum to prevent motorist distraction. Flashing lights shall not be used behind barrier protection systems.

Switches shall be provided to adequately control the various lights. All wiring shall be weatherproof and installed according to local, state, federal, and OSHA requirements. Ground fault circuit interrupters shall be provided for electrical outlets used for electrical tools and extension cords. The contractor shall provide sufficient fuel, spare lamps, generators, and qualified personnel to ensure that all required lights operate continuously during nighttime construction operations. In the event of any failure of the lighting system, the construction operation shall be discontinued until the required level of illumination is restored. In

residential areas, generator systems shall be selected to comply with local noise ordinances. A supply of emergency flares shall be maintained by the contractor for use in the event of emergency or unanticipated situations.

**713.10.3 Illumination Requirements:** All operations that are performed during nighttime hours shall be properly illuminated to allow for the safe performance and inspection of the work.

Work area is defined as a minimum of 50 feet ahead and behind the employee, where work is to be performed. A minimum of 5 foot-candles (54 lux) shall be maintained throughout the work area during nighttime construction operations, and during the setup and removal of lane or roadway closures.

Lighting shall be adequate to meet the required level of illuminance and uniformity over the work area as follows:

713.10.3.1 Level I (5 foot-candles, 54 lux): This level of illuminance shall be provided for all work areas of general construction operations, such as excavation and embankment; cleaning and sweeping; landscaping; planting and seeding. Stockpiles shall also be illuminated to Level I to enhance safety and improve work efficiency.

713.10.3.2 Level II (10 foot-candles, 108 lux): This level of illuminance is required for areas on or around construction equipment such as that used for drainage installations, striping, base course construction, milling, asphalt paving operations, and concrete placement and removal. This level is necessary for safe operation of equipment and for obtaining an acceptable level of accuracy.

713.10.3.3 Level III (20 foot-candles, 215 lux): This level of illuminance is required for tasks requiring a higher level of visual performance or for tasks with a higher level of difficulty. Such tasks include, pavement or structural crack filling, joint repair, joint cleaning, joint sealing, pavement patching and repairs, saw-cutting, installation of signal equipment or other electrical/mechanical equipment, and other tasks involving fine details or intricate parts and equipment.

713.10.4 Glare Control: All lighting provided under this item shall be designed, installed, and operated to avoid glare interference with roadway traffic or discomfort for residences adjoining the roadway. The contractor shall locate, aim, and adjust the lights to provide the required level of illuminance and uniformity in the work area without the creation of objectionable glare. The engineer shall determine when glare exceeds acceptable levels, either for traffic or adjoining residences. The contractor shall provide shields, visors, or louvers on luminaries as necessary to reduce objectionable levels of glare.

At a minimum, the following requirements shall be met to avoid objectionable glare to oncoming traffic:

- 1. Tower-mounted luminaries shall generally be aimed either parallel or perpendicular to the roadway.
- 2. All luminaries shall be aimed such that the center of the beam axis is no greater than 60 degrees from the vertical.

- 3. Luminous intensity of any luminary shall not exceed 20,000 candelas at an angle of 72 degrees from the vertical.
- 713.10.5 Operational Requirements: Thirty days prior to the start of night time operations, the contractor shall submit a lighting plan to the engineer for acceptance. The contractor shall select appropriate lighting systems and design a lighting plan to achieve the required illuminance levels.

The lighting plan shall include location of lights necessary for every aspect of work; description of light equipment to be used; description of power source; attachment and mounting details for lights to be attached to equipment; technical details pertaining to the lighting fixtures; details on hoods, louvers, shields, or other glare control methods; and lighting calculations confirming that the illumination requirements will be met by the layout plan.

Lighting inspection will include (1) light meter measurements to determine illumination levels, (2) subjective observation of the lighting setup to evaluate glare potential for drivers and workers, and (3) a physical check of the lighting equipment to ensure that it complies with the specification requirements included in the contractor's lighting plan.

Prior to the first night of operation, the engineer will check the adequacy of the installed lighting using a light meter. A summary of these measurements will be noted in the inspection records to provide a basis for comparing subsequent measurements. If the required illuminance levels are not met, the contractor shall make the necessary adjustments before any work proceeds.

Operational checks shall be made when construction phasing changes and lighting plan changes are required to accommodate different phases of construction. Periodic checks will be made throughout the duration of nighttime operations. If the required illuminance levels are not met, the contractor shall make the necessary adjustments to the lighting plan before work continues.

During construction operations, in the event of any failure of the lighting system, the operations shall be discontinued until the required level of illumination is restored.

#### 713.11 MEASUREMENT.

713.11.1 Temporary Signs and Barricades: When the contract does not include a pay item for "Temporary Signs and Barricades," the provision of temporary construction signs, barricades, and related devices will be considered by the Department to be for the convenience of the contractor and will not be measured for payment.

When including a pay item for "Temporary Signs and Barricades" in the contract, the furnishing, erecting, maintaining, and subsequent removing of temporary construction signs, barricades, and related devices will be measured on a lump sum basis.

Flashing arrow boards will not be measured for separate payment, but will be included in the contract lump sum price for Temporary Signs and Barricades.

713.11.2 Temporary Pavement Markings: When the contract does not include a pay item for Temporary Pavement Markings, provision of these markings will be considered by the Department to be for the convenience of the contractor and will not be measured for payment.

When the contract includes an item for Temporary Pavement Markings, these markings' acceptable furnishing, placing, maintenance, and subsequent removal will be measured by the linear foot or mile, as specified.

When measuring by the linear foot of striping, measurement will be made for the material placed. Gaps will be excluded.

When measuring by the mile of single strip per roadway per application, no deduction will be made for the standard design gaps in broken line striping; however, deductions will be made for the length of other gaps or omitted sections.

Temporary pavement legends and symbols will be measured per each legend or symbol.

Temporary reflectorized raised pavement markers will be measured per each marker furnished, placed, and accepted. Removal of temporary reflectorized raised pavement markers will not be measured for payment.

713.11.3 Temporary Precast Concrete Barriers: When the contract does not include a pay item for Temporary Precast Concrete Barriers, the provision of these barriers will be considered by the Department to be for the convenience of the contractor and will not be measured for payment.

Temporary Precast Concrete Barriers furnished by the contractor will be measured per each unit installed, which includes construction, delivery, furnishing, installing, maintaining, and removing each unit from the jobsite a single time. Temporary Precast Concrete Barriers (Department furnished) will be measured per each unit installed which includes collecting from the location specified, transporting, and delivering to the project site, and all costs of handling, maintaining, and returning each unit to the location specified or as directed.

Further movements of barriers for subsequent construction phases will be measured per movement of each barrier.

- 713.11.4 Traffic Control Management: Traffic control management will not be measured for payment.
- **713.11.5 Nighttime Construction Operations:** Nighttime construction operations will not be measured for payment.

**713.12 PAYMENT.** Payment for temporary construction signs, barricades, and related devices will be at the contract lump sum price in accordance with the payment schedule of Table 713-2.

Table 713-2 Payment Schedule

**Temporary Signs, Barricades and Related Devices** 

| Percent of            | Allowable Percent of           |
|-----------------------|--------------------------------|
| Total Contract Amount | Lump Sum Price for             |
| Earned                | Temporary Signs and Barricades |
| Initial Erection      | 40                             |
| 25                    | 60                             |
| 50                    | 80                             |
| 75                    | 95                             |
| 100                   | 100                            |

Payment for temporary pavement markings will be made at the respective contract unit prices. Payment for temporary precast concrete barriers will be made at the contract unit price per each. The concrete in temporary precast barriers furnished by the contractor will be identified by lots and shall be subject to pay adjustments in accordance with Table 901-6 and Note 1 therein. Size, sampling, and testing of each concrete lot shall be in accordance with the Materials Sampling Manual.

Payment for additional movements of temporary concrete barriers will be made per movement of each barrier when required in the plans and directed by the engineer.

Payment for portable changeable message signs will be made at the contract unit price per each during the life of the contract.

#### Payment will be made under:

| Item No. | Pay Item  | Pay Unit    |
|----------|---|-------------|
| 713-01   | Temporary Signs and Barricades                  | Lump Sum    |
| 713-02   | Temporary Pavement Markings                     | Linear Foot |
| 713-03   | Temporary Pavement Markings (Broken Line)       | Mile        |
| 713-04   | Temporary Pavement Markings (Solid Line)        | Mile        |
| 713-05   | Temporary Pavement Legends and Symbols          | Each        |
| 713-06   | Temporary Reflectorized Raised Pavement Markers | Each        |
| 713-07   | Temporary Precast Concrete Barrier              |             |
|          | (Contractor Furnished)                          | Each        |
| 713-08   | Temporary Precast Concrete Barrier              |             |
|          | (Department Furnished)                          | Each        |
| 713-09   | Temporary Portable Barrier                      | Each        |
| 713-10   | Temporary Precast Concrete Barrier Movement     | Each        |
| 713-11   | Portable Changeable Message Signs               | Each        |
|          |   |             |

# TECHNICAL PROVISIONS

### **SECTION 727**

#### **MOBILIZATION**

# TABLE OF CONTENTS

| 727.01 | DESCRIPTION               |
|--------|---------------------------|
| 727.02 | MATERIAL                  |
| 727.03 | CONSTRUCTION REQUIREMENTS |
| 727.04 | MEASUREMENT               |
| 727.05 | PAYMENT                   |

This page intentionally left blank

#### **TECHNICAL PROVISIONS**

#### **SECTION 727**

#### **MOBILIZATION**

#### 727.01 DESCRIPTION.

Mobilization consists of preparatory work and operations, including those necessary for movement of personnel, equipment, supplies, and incidentals to the project site; the establishment of offices, buildings, and other facilities necessary for work on the project; the cost of bonds and any required insurance; and other preconstruction expenses necessary for start of the work, excluding the cost of construction materials.

#### 727.02 MATERIALS. Vacant

#### 727.03 CONSTRUCTION REQUIREMENTS. Vacant

**MEASUREMENT.** Mobilization will be measured for payment as a lump sum.

**727.05 PAYMENT.** When the contract does not include a pay item for mobilization, no direct payment will be made for mobilization.

When the contract contains a pay item for mobilization, payment will be made at the contract lump sum price, subject to the following provisions: Partial payments for mobilization will be made in accordance with the schedule of Table 727-1 up to a maximum of 10 percent of the original total contract amount, including this item. Payment of any remaining amount will be made upon completion of all work under the contract.

Table 727-1
Mobilization Payment Schedule

| Percent of Total Contract Amount Earned | Allowable Percent of the Lump Sum Price for Mobilization |
|---|--|
| 1st Partial Estimate                    | 25   |
| 10                                      | 50   |
| 25                                      | 75   |
| 50                                      | 100  |

No payment adjustments will be made for this item due to changes in the work in accordance with Section 109.

When the contract includes a pay item for field laboratories under Section 722, payment for mobilization will exclude those facilities.

Payment will be made under:

| Item No. | Pay Item     | Pay Unit |
|----------|--------------|----------|
| 727-01   | Mobilization | Lump Sum |

# TECHNICAL PROVISIONS SECTION 740 CONSTRUCTION LAYOUT

# TABLE OF CONTENTS

| 740.01 | DESCRIPTION               |
|--------|---------------------------|
| 740.02 | CONSTRUCTION REQUIREMENTS |
| 740.03 | MEASUREMENT               |
| 740.04 | PAYMENT                   |

This page intentionally left blank

#### **TECHNICAL PROVISIONS**

#### **SECTION 740**

#### **CONSTRUCTION LAYOUT**

**740.01 DESCRIPTION.** This section sets forth requirements for all construction layouts. Establish lines and grades, take all cross-sections, and stake out the construction work in accordance with these specifications, plan details, and as directed. Construction layout also includes, but is not limited to, the layout of pavement striping and raised pavement markers, setting of line and grade for construction of superelevated curves or other applicable work items, and assisting in the coordination of utility relocation activities to ensure that the placement of relocated facilities will not conflict with required construction.

**740.02 CONSTRUCTION REQUIREMENTS.** Establish all lines and grades and stake out all project work, including sufficient vertical and horizontal control points for utility relocations for use by the Department and others.

The project survey control and horizontal alignment are based on the Louisiana State Plane Coordinate System. The construction plans and/or right-of-way map depicts the coordinates and datum of sufficient survey control points to establish or re-establish horizontal control throughout the length of the project. Employ such methods as approved by the project engineer for the location of the project alignment and other necessary survey control points in accordance with currently acceptable surveying standards and practices. When required, the Department will also provide one bench mark on or near the project for vertical control. Verify the values of any intermediate bench marks shown on the plans, by checking against the bench mark established by the Department for vertical control.

Employ qualified engineering and surveying personnel experienced in layout and construction of highways and bridges to correctly establish and keep complete and comprehensive records (field books or approved electronic files) of all lines and grades necessary from initial layout to final acceptance. Provide sufficient qualified staff, of at least one employee, on site during utility relocation periods. Provide any necessary survey work to ensure there are no utility conflicts with required construction. Provide daily documentation of utility relocation activities for incorporation into the project diaries.

The contractor shall be liable for the accuracy of the initial layout and all subsequent alignment and elevations and shall, at no additional pay, rebuild, repair or make good any portion of the work found to be incorrectly positioned either horizontally or vertically at any time before final acceptance. Notify the engineer immediately of any apparent errors in the plans. Compute and provide template grades to the engineer. In order to obtain pipe order lengths, provide the appropriate grades to the engineer two weeks in advance of the work.

Numbered notebooks for recording of all lines and grades will be provided by the Department and shall be properly indexed and cross referenced by the contractor before return to the engineer for submittal with the final estimate. Computer generated printouts will be allowed when approved.

Set stationing for overlay projects using an approved measuring device that is accurate to 0.1 percent. Place stakes every 100 linear feet and maintain throughout construction.

For pavement preservation type projects, the contractor will be responsible for recording the location of all existing pavement markings and laying out the required final markings subject to the approval of the Project Engineer.

Perform the layout of striping, raised pavement markers, and signs by methods approved by the engineer prior to placement.

#### 740.03 MEASUREMENT.

Measurements for determination of pay quantities will be made by the Department. Construction layout and utility oversight and coordination will be measured per lump sum, which will include all labor, materials, tools, equipment, and incidentals required to complete the work.

No changes in the lump sum contract price will be made for minor additions or deletions to the scope of work.

#### **740.04 PAYMENT.**

Payment for construction layout, and utility oversight and coordination will be made at the contract lump sum price in accordance with Table 740-1 and Table 740-2, respectively.

Table 740-1 Construction Layout Payment Schedule

| Percent of Total Contract Amount | Allowable Percent of Lump Sum |
|----------------------------------|-------------------------------|
| Earned                           | Price for Construction Layout |
| Staffed                          | 25                            |
| 25                               | 50                            |
| 50                               | 80                            |
| 75                               | 95                            |
| 100                              | 100                           |

Table 740-2 Utility Oversight and Coordination Payment Schedule

| Percent of Utility Relocation Complete | Allowable Percent of Lump Sum Price for Construction Layout |
|--|---|
| Staffed                                | 25  |
| 25                                     | 50  |
| 50                                     | 65  |
| 75                                     | 80  |
| 100                                    | 85  |
| Project Completion                     | 100   |

Payment will be made under:

| Item No. | Pay Item                           | Pay Unit |
|----------|------------------------------------|----------|
| 740-01   | Construction Layout                | Lump Sum |
| 740-02   | Utility Oversight and Coordination | Lump Sum |

This page intentionally left blank

#### SECTION 01 10 00 - SUMMARY

#### PART 1 - GENERAL

#### 1.1 SUMMARY

#### A. Section Includes:

- 1. Project information.
- 2. Work covered by Contract Documents.
- 3. Phased construction.
- 4. Work under Owner's separate contracts.
- 5. Owner-furnished/Contractor-installed (OFCI) products.
- 6. Contractor's use of site and premises.
- 7. Coordination with occupants.
- 8. Work restrictions.
- 9. Specification and Drawing conventions.

#### B. Related Requirements:

1. Section 01 50 00 "Temporary Facilities and Controls" for limitations and procedures governing temporary use of Owner's facilities.

#### 1.2 PROJECT INFORMATION

- A. Project Identification: Activities Building Improvements
  - 1. Project Location: 219 W. 2<sup>nd</sup> St, DeRidder, LA 70634
- B. Owner: Beauregard Parish Police Jury
  - 1. Owner's Representative: Bryan McReynolds
- C. Engineer: Meyer & Associates, Inc.
  - 1. Engineer's Representative: Byron Racca
- D. Web-Based Project Software: Project software will be used for purposes of managing communication and documents during the construction stage.
  - 1. See Section 01 31 00 "Project Management and Coordination." for requirements for using web-based Project software.

#### 1.3 WORK COVERED BY CONTRACT DOCUMENTS

A. The Work of Project is defined by the Contract Documents and consists of the following:

SUMMARY 01 10 00 - 1

318

- M.A. No. H1-22029-DA RMG - 22029
- 1. Repair to existing facility including doors and windows and over pour of portion of concrete floor.
- 2. Build out of new offices and restrooms including: metal studs, gypsum board, painting, suspended acoustic ceilings, LVT flooring, ceramic tile floor and walls, s.c. wood doors with HM frames. New fire alarm, lighting, plumbing and HVAC.

#### B. Type of Contract:

1. Project will be constructed under a single prime contract.

#### 1.4 OWNER-FURNISHED/CONTRACTOR-INSTALLED (OFCI) PRODUCTS

- A. Owner's Responsibilities: Owner will furnish products indicated and perform the following, as applicable:
  - 1. Provide to Contractor Owner-reviewed Product Data, Shop Drawings, and Samples.
  - 2. Provide for delivery of Owner-furnished products to Project site.
  - 3. Upon delivery, inspect, with Contractor present, delivered items.
    - a. If Owner-furnished products are damaged, defective, or missing, arrange for replacement.
  - 4. Obtain manufacturer's inspections, service, and warranties.
  - 5. Inform Contractor of earliest available delivery date for Owner-furnished products.
- B. Contractor's Responsibilities: The Work includes the following, as applicable:
  - 1. Designate delivery dates of Owner-furnished products in Contractor's construction schedule, utilizing Owner-furnished earliest available delivery dates.
  - 2. Review Owner-reviewed Product Data, Shop Drawings, and Samples, noting discrepancies and other issues in providing for Owner-furnished products in the Work.
  - 3. Receive, unload, handle, store, protect, and install Owner-furnished products.
  - 4. Make building services connections for Owner-furnished products.
  - 5. Protect Owner-furnished products from damage during storage, handling, and installation and prior to Substantial Completion.
  - 6. Repair or replace Owner-furnished products damaged following receipt.

#### 1.5 CONTRACTOR'S USE OF SITE AND PREMISES

- A. Restricted Use of Site: Contractor shall have limited use of Project site for construction operations as indicated on Drawings by the Contract limits and as indicated by requirements of this Section.
- B. Limits on Use of Site: Limit use of Project site to areas within the Contract limits indicated. Do not disturb portions of Project site beyond areas in which the Work is indicated.
  - 1. Limits on Use of Site: Confine construction operations to area indicated on site plan including the Construction Safety and Phasing Plan.

SUMMARY 01 10 00 - 2

319

- M.A. No. H1-22029-DA RMG - 22029
- 2. Driveways, Walkways, and Entrances: Keep driveways loading areas, and entrances serving premises clear and available to Owner, Owner's employees, and emergency vehicles at all times. Do not use these areas for parking or for storage of materials.
- C. Condition of Existing Building: Maintain portions of existing building affected by construction operations in a weathertight condition throughout construction period. Repair damage caused by construction operations.
- D. Condition of Existing Grounds: Maintain portions of existing grounds, landscaping, and hardscaping affected by construction operations throughout construction period. Repair damage caused by construction operations.

#### 1.6 COORDINATION WITH OCCUPANTS

A. Partial Owner Occupancy: Owner will occupy the premises during entire construction period, with the exception of areas under construction. Cooperate with Owner during construction operations to minimize conflicts and facilitate Owner usage. Perform the Work so as not to interfere with Owner's operations. Maintain existing exits unless otherwise indicated.

#### 1.7 WORK RESTRICTIONS

- A. Comply with restrictions on construction operations.
  - 1. Comply with limitations on use of public streets, work on public streets, rights of way, and other requirements of authorities having jurisdiction.
- B. On-Site Work Hours: Limit work in the existing building to normal business working hours of 7:00 a.m. to 5:00 p.m., Monday through Friday, unless otherwise indicated.
- C. Existing Utility Interruptions: Do not interrupt utilities serving facilities occupied by Owner or others unless permitted under the following conditions and then only after providing temporary utility services according to requirements indicated:
  - 1. Notify Owner not less than two days in advance of proposed utility interruptions.
  - 2. Obtain Owner's written permission before proceeding with utility interruptions.
- D. Noise, Vibration, Dust, and Odors: Coordinate operations that may result in high levels of noise and vibration, dust, odors, or other disruption to Owner occupancy with Owner.
  - 1. Notify Owner not less than two days in advance of proposed disruptive operations.
  - 2. Obtain Owner's written permission before proceeding with disruptive operations.
- E. Smoking and Controlled Substance Restrictions: Use of tobacco products, alcoholic beverages, and other controlled substances on Owner's property is not permitted.
- F. Employee Identification: Owner will provide identification tags for Contractor personnel working on Project site. Require personnel to use identification tags at all times.
- G. Employee Screening: Comply with Owner's requirements for drug and background screening of Contractor personnel working on Project site.

SUMMARY 01 10 00 - 3

M.A. No. H1-22029-DA RMG - 22029

1. Maintain list of approved screened personnel with Owner's representative.

#### 1.8 SPECIFICATION AND DRAWING CONVENTIONS

- A. Specification Content: The Specifications use certain conventions for the style of language and the intended meaning of certain terms, words, and phrases when used in particular situations. These conventions are as follows:
  - 1. Imperative mood and streamlined language are generally used in the Specifications. The words "shall," "shall be," or "shall comply with," depending on the context, are implied where a colon (:) is used within a sentence or phrase.
  - 2. Text Color: Text used in the Specifications, including units of measure, manufacturer and product names, and other text may appear in multiple colors or underlined as part of a hyperlink; no emphasis is implied by text with these characteristics.
  - 3. Hypertext: Text used in the Specifications may contain hyperlinks. Hyperlinks may allow for access to linked information that is not residing in the Specifications. Unless otherwise indicated, linked information is not part of the Contract Documents.
  - 4. Specification requirements are to be performed by Contractor unless specifically stated otherwise.
- B. Division 00 Contracting Requirements: General provisions of the Contract, including General and Supplementary Conditions, apply to all Sections of the Specifications.
- C. Division 01 General Requirements: Requirements of Sections in Division 01 apply to the Work of all Sections in the Specifications.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 10 00

SUMMARY 01 10 00 - 4

321

#### SECTION 01 23 00 - ALTERNATES

#### PART 1 - GENERAL

#### 1.1 SUMMARY

A. Section includes administrative and procedural requirements for alternates.

#### 1.2 DEFINITIONS

- A. Alternate: An amount proposed by bidders and stated on the Bid Form for certain work defined in the bidding requirements that may be added to the base bid amount if Owner decides to accept a corresponding change either in the amount of construction to be completed or in the products, materials, equipment, systems, or installation methods described in the Contract Documents.
  - 1. Alternates described in this Section are part of the Work only if enumerated in the Agreement.
  - 2. The cost for each alternate is the net addition to the Contract Sum to incorporate alternate into the Work. No other adjustments are made to the Contract Sum.

#### 1.3 PROCEDURES

- A. Coordination: Revise or adjust affected adjacent work as necessary to completely integrate work of the alternate into Project.
  - 1. Include as part of each alternate, miscellaneous devices, accessory objects, and similar items incidental to or required for a complete installation whether or not indicated as part of alternate.
- B. Notification: Immediately following award of the Contract, notify each party involved, in writing, of the status of each alternate. Indicate if alternates have been accepted, rejected, or deferred for later consideration. Include a complete description of negotiated revisions to alternates.
- C. Execute accepted alternates under the same conditions as other work of the Contract.
- D. Schedule: A schedule of alternates is included at the end of this Section. Specification Sections referenced in schedule contain requirements for materials necessary to achieve the work described under each alternate.

#### PART 2 - PRODUCTS (Not Used)

#### PART 3 - EXECUTION

#### 3.1 SCHEDULE OF ALTERNATES

- A. Add Alternate No. 1: Interior Build-out.
  - 1. Base Bid: All site work and exterior work to the building envelope is to be included in base bid.
  - 2. Alternate No. 1: Provide and install all interior build-out per plans and specs.

END OF SECTION 01 23 00

ALTERNATES 01 23 00 - 1

322

This page intentionally left blank

#### SECTION 01 26 00 - CONTRACT MODIFICATION PROCEDURES

### PART 1 - GENERAL

#### 1.1 SUMMARY

A. This Section specifies administrative and procedural requirements for handling and processing Contract modifications.

### 1.2 MINOR CHANGES IN THE WORK

A. Architect will issue supplemental instructions authorizing Minor Changes in the Work, not involving adjustment to the Contract Sum or the Contract Time, on AIA Document G710, "Architect's Supplemental Instructions."

# 1.3 PROPOSAL REQUESTS

- A. Owner-Initiated Proposal Requests: Architect will issue a detailed description of proposed changes in the Work that may require adjustment to the Contract Sum or the Contract Time. If necessary, the description will include supplemental or revised Drawings and Specifications.
  - 1. Proposal Requests issued by Architect are for information only. Do not consider them instructions either to stop work in progress or to execute the proposed change.
  - 2. Within 20 days after receipt of Proposal Request, submit a quotation estimating cost adjustments to the Contract Sum and the Contract Time necessary to execute the change.
    - a. Include a list of quantities of products required or eliminated and unit costs, with total amount of purchases and credits to be made. If requested, furnish survey data to substantiate quantities.
    - b. Indicate applicable taxes, delivery charges, equipment rental, and amounts of trade discounts.
    - c. Include costs of labor and supervision directly attributable to the change.
    - d. Include an updated Contractor's Construction Schedule that indicates the effect of the change, including, but not limited to, changes in activity duration, start and finish times, and activity relationship. Use available total float before requesting an extension of the Contract Time.
- B. Contractor-Initiated Proposals: If latent or unforeseen conditions require modifications to the Contract, Contractor may propose changes by submitting a request for a change to Architect.
  - 1. Substitution of materials or manufacturers not contained in original bid may be ONLY for the following reasons:
    - a. Substantial advantage for the Owner.
    - b. Manufacturer has made changes related to conditions beyond the control of the Architect or Contractor that have been changed since the Project was bid.
    - c. Some aspect of the Manufacturer can no longer be provided since the time the Project was bid.

- M.A. No. H1-22029-DA RMG - 22029
- 2. Include a statement outlining reasons for the change and the effect of the change on the Work. Provide a complete description of the proposed change. Indicate the effect of the proposed change on the Contract Sum and the Contract Time.
- 3. Include a list of quantities of products required or eliminated and unit costs, with total amount of purchases and credits to be made. If requested, furnish survey data to substantiate quantities.
- 4. Indicate applicable taxes, delivery charges, equipment rental, and amounts of trade discounts
- 5. Include costs of labor and supervision directly attributable to the change.
- 6. Include an updated Contractor's Construction Schedule that indicates the effect of the change, including, but not limited to, changes in activity duration, start and finish times, and activity relationship. Use available total float before requesting an extension of the Contract Time.
- C. Proposal Request Form: Use AIA Document G709 for Proposal Requests.

#### 1.4 CHANGE ORDER PROCEDURES

A. On Owner's approval of a Proposal Request, Architect will issue a Change Order for signatures of Owner and Contractor.

#### 1.5 CONSTRUCTION CHANGE DIRECTIVE

- A. Construction Change Directive: Architect may issue a Construction Change Directive on AIA Document G714. Construction Change Directive instructs Contractor to proceed with a change in the Work, for subsequent inclusion in a Change Order.
  - 1. Construction Change Directive contains a complete description of change in the Work. It also designates method to be followed to determine change in the Contract Sum or the Contract Time.
- B. Documentation: Maintain detailed records on a time and material basis of work required by the Construction Change Directive.
  - 1. After completion of change, submit an itemized account and supporting data necessary to substantiate cost and time adjustments to the Contract.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 26 00

#### SECTION 01 29 00 - PAYMENT PROCEDURES

# PART 1 - GENERAL

#### 1.1 SUMMARY

A. This Section specifies administrative and procedural requirements necessary to prepare and process Applications for Payment.

#### 1.2 RELATED SECTIONS

- A. Section 01 31 00 Project Management and Coordination.
- B. Coordinate with requirements of General Conditions & Supplemental Conditions and submit the following Items at the Pre-Construction Conference and prior to initial Application for Payment:
  - 1. List of subcontractors.
  - 2. Schedule of Values.
  - 3. Contractor's Construction Schedule.
  - 4. Schedule of unit prices.

# 1.3 SCHEDULE OF VALUES

- A. Coordination: Coordinate preparation of the Schedule of Values with preparation of Contractor's Construction Schedule. Cost-loaded CPM Schedule may supplant the Schedule of Values.
  - 1. Correlate line items in the Schedule of Values with other required administrative forms and schedules, including Application for Payment forms with Continuation Sheets Submittals Schedule and Contractor's Construction Schedule.
  - 2. Submit the Schedule of Values to Architect at earliest possible date but no later than seven days before the date scheduled for submittal of initial Applications for Payment.
- B. Format and Content: Use the Project Manual table of contents as a guide to establish line items for the Schedule of Values. Provide at least one line item for each Specification Section.
  - 1. Identification: Include the following Project identification on the Schedule of Values:
    - a. Project name and location.
    - b. Name of Architect.
    - c. Architect's Project Number.
    - d. Contractor's name and address.
    - e. Date of submittal.
  - 2. Submit draft of AIA Document G703 Continuation Sheets.
  - 3. Provide a breakdown of the Contract Sum in enough detail to facilitate continued evaluation of Applications for Payment and progress reports. Coordinate with the Project Manual table of contents. Provide several line items for principal subcontract amounts, where appropriate. Include separate line items under required principal subcontracts for operation and maintenance manuals, punch list activities, Project Record Documents, and demonstration and training in the amount of 5 percent of the Contract Sum.
  - 4. Round amounts to nearest whole dollar; total shall equal the Contract Sum.

- 5. Provide a separate line item in the Schedule of Values for each part of the Work where Applications for Payment may include materials or equipment purchased or fabricated and stored, but not yet installed.
- 6. Each item in the Schedule of Values and Applications for Payment shall be complete. Include total cost and proportionate share of general overhead and profit for each item.
  - a. Temporary facilities and other major cost items that are not direct cost of actual work-in-place may be shown either as separate line items in the Schedule of Values or distributed as general overhead expense, at Contractor's option.
- 7. Schedule Updating: Update and resubmit the Schedule of Values before the next Applications for Payment when Change Orders or Construction Change Directives result in a change in the Contract Sum.

# 1.4 APPLICATIONS FOR PAYMENT

- A. Each Application for Payment shall be consistent with previous applications and payments as certified by Architect and paid for by Owner.
  - 1. Initial Application for Payment, Application for Payment at time of Recommendation of Acceptance, and final Application for Payment involve additional requirements.
- B. Payment Application Times: Application for Payment shall be submitted on or about the first of each month for the value of labor and materials incorporated into the Work and of materials, suitably stored, at the site as of the twenty-fifth day of the preceding month, less normal retainage as follows, per R.S. 38:2248
  - 1. Projects with Contract price of \$500,000.00, or more 5% of the Contract price.
- C. Payment Application Forms: Use AIA Document G702 and AIA Document G703 Continuation Sheets as form for Applications for Payment.
- D. Application Preparation: Complete every entry on form. Architect will return incomplete applications without action.
  - 1. Entries shall match data on the Schedule of Values and Contractor's Construction Schedule. Use updated schedules if revisions were made.
  - 2. Include amounts of Change Orders and Construction Change Directives issued before last day of construction period covered by application.
  - 3. Periodic Construction Photographs may be sent under separate cover. Email is preferred. Send to the Project Architect.
- E. Transmittal: Submit 4 signed original copies of each Application for Payment to Architect by a method ensuring receipt within 24 hours. One copy shall include waivers of lien and similar attachments if required.
  - 1. Transmit each copy with a transmittal form listing attachments and recording appropriate information about application.
- F. Initial Application for Payment: Administrative actions and submittals that must precede or coincide with submittal of first Application for Payment include the following:
  - 1. Submittals Schedule (preliminary if not final).
  - 2. List of Contractor's staff assignments.
  - 3. List of Contractor's principal consultants.
  - 4. Copies of authorizations and licenses from authorities having jurisdiction for performance of the Work.

- 5. Initial progress report.
- 6. Report of preconstruction conference.
- G. Application for Payment at Recommendation of Acceptance: After the Recommendation of Acceptance has been issued, submit an Application for Payment showing 100 percent completion for portion of the Work claimed as substantially complete.
  - 1. Include documentation supporting claim that the Work is complete and a statement showing an accounting of changes to the Contract Sum.
- H. Final Payment Application: Submit final Application for Payment with releases and supporting documentation not previously submitted and accepted, including, but not limited, to the following:
  - 1. Evidence of completion of Project closeout requirements.
  - 2. Clear Lien Certificate.
  - 3. Consent of Surety to Final Payment.
  - 4. Final, liquidated damages settlement statement.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 29 00

This page intentionally left blank

#### SECTION 01 31 00 - PROJECT MANAGEMENT AND COORDINATION

#### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. This Section includes administrative provisions for coordinating construction operations on Project including, but not limited to, the following:
  - 1. Coordination Drawings.
  - 2. Project meetings.
  - 3. Requests for Interpretation (RFIs).
- B. See Division 01 Section "Execution" for procedures for coordinating general installation and field-engineering services, including establishment of benchmarks and control points.

#### 1.2 DEFINITIONS

A. RFI: Request from Contractor seeking interpretation or clarification of the Contract Documents.

#### 1.3 COORDINATION

- A. Coordination: Coordinate construction operations included in different Sections of the Specifications to ensure efficient and orderly installation of each part of the Work. Coordinate construction operations, included in different Sections, that depend on each other for proper installation, connection, and operation.
  - 1. Schedule construction operations in sequence required to obtain the best results where installation of one part of the Work depends on installation of other components, before or after its own installation.
  - 2. Coordinate installation of different components with other contractors to ensure maximum accessibility for required maintenance, service, and repair.
  - 3. Make adequate provisions to accommodate items scheduled for later installation.
  - 4. Where availability of space is limited, coordinate installation of different components to ensure maximum performance and accessibility for required maintenance, service, and repair of all components, including mechanical and electrical.
- B. Prepare memoranda for distribution to each party involved, outlining special procedures required for coordination. Include such items as required notices, reports, and list of attendees at meetings.
  - 1. Prepare similar memoranda for Owner and separate contractors if coordination of their Work is required.
- C. Administrative Procedures: Coordinate scheduling and timing of required administrative procedures with other construction activities and activities of other contractors to avoid conflicts and to ensure orderly progress of the Work. Such administrative activities include, but are not limited to, the following:
  - 1. Preparation of Contractor's Construction Schedule.

M.A. No. H1-22029-DA RMG - 22029

- 2. Preparation of the Schedule of Values.
- 3. Installation and removal of temporary facilities and controls.
- 4. Delivery and processing of submittals.
- 5. Progress meetings.
- 6. Preinstallation conferences.
- 7. Project closeout activities.
- 8. Startup and adjustment of systems.
- 9. Project closeout activities.

#### 1.4 SUBMITTALS

- A. Coordination Drawings: Prepare Coordination Drawings if limited space availability necessitates maximum utilization of space for efficient installation of different components or if coordination is required for installation of products and materials fabricated by separate entities.
  - 1. Content: Project-specific information, drawn accurately to scale. Do not base Coordination Drawings on reproductions of the Contract Documents or standard printed data. Include the following information, as applicable:
    - a. Indicate functional and spatial relationships of components of architectural, structural, civil, mechanical, and electrical systems.
    - b. Indicate dimensions shown on the Contract Drawings and make specific note of dimensions that appear to be in conflict with submitted equipment and minimum clearance requirements. Provide alternate sketches to Architect for resolution of such conflicts. Minor dimension changes and difficult installations will not be considered changes to the Contract.
  - 2. Sheet Size: At least 8-1/2 by 11 inches (215 by 280 mm) but no larger than 30 by 40 inches (750 by 1000 mm).
  - 3. Number of Copies: Submit three opaque copies of each submittal. Architect will return one copy.
  - 4. Refer to individual Sections for Coordination Drawing requirements for Work in those Sections.

## B. Electronic Submittals

1. Reference Section 01 33 00 Submittal Procedures for requirement of electronic submittals.

# 1.5 PROJECT MEETINGS

- A. General: Schedule and coordinate all major sub-contractors to be present at meetings and conferences at Project site, unless otherwise indicated.
  - 1. Attendees: Inform participants and others involved, and individuals whose presence is required, of date and time of each meeting. Coordinate with Architect for timing of scheduled meeting dates and times with milestones of construction.
  - 2. Agenda: Contractor will prepare the meeting agenda. Distribute to all invited attendees

- 3. Minutes: Designer will record significant discussions and agreements achieved. Distribute the meeting minutes to everyone concerned, including Owner and Contractor, within three days of the meeting.
- B. Preconstruction Conference: Attend a preconstruction conference before starting construction, at a time convenient to Owner and Architect, but no later than 15 days after execution of the Agreement. Hold the conference at Project site or another convenient location. Conduct the meeting to review responsibilities and personnel assignments.
  - 1. Attendees: Authorized representatives of Owner, Architect, and their consultants; Contractor and its superintendent; major subcontractors; suppliers; and other concerned parties shall attend the conference. All participants at the conference shall be familiar with Project and authorized to conclude matters relating to the Work.
  - 2. Submit items required by Supplemental Conditions Section 7.1 including the following:
    - a. Schedule of Values, in standard CSI format.
    - b. Construction Schedule.
    - c. List of Sub-contractors with contact information.
    - d. List of Major Suppliers with contact information.
    - e. Fixed job site overhead cost itemized with documentation to support daily rates.
    - f. Bond Premium Rate (Performance & Payment) with supporting Letter from GC Insurance Carrier.
    - g. Labor burden by trade for both Subcontractors and General Contractors (notarized and including W/C information sheet).
      - \*\*Labor burden for applicable payroll taxes shall include WC, Medicare, FICA, FUTA, and SUTA only.
    - h. Internal Rate Charges for all significant company owned equipment.
  - 3. Agenda: Discuss items of significance that could affect progress, including the following:
    - a. Tentative construction schedule.
    - b. Phasing.
    - c. Critical work sequencing and long-lead items.
    - d. Designation of key personnel and their duties.
    - e. Procedures for processing field decisions and Change Orders.
    - f. Procedures for RFIs.
    - g. Procedures for testing and inspecting.
    - h. Procedures for processing Applications for Payment.
    - i. Distribution of the Contract Documents.
    - i. Submittal procedures.
    - k. Preparation of Record Documents.
    - 1. Use of the premises.
    - m. Work restrictions.
    - n. Owner's occupancy requirements.
    - o. Responsibility for temporary facilities and controls.
    - p. Construction waste management and recycling.
    - q. Parking availability.
    - r. Office, work, and storage areas.
    - s. Equipment deliveries and priorities.
    - t. First aid.
    - u. Security.
    - v. Progress cleaning.

M.A. No. H1-22029-DA RMG - 22029

- w. Working hours.
- 4. Minutes: Review minutes as recorded by Architect and submit proposed meeting minutes changes promptly to the Architect.
- C. Preinstallation Conferences: Conduct a preinstallation conference at Project site before each construction activity that requires coordination with other construction.
  - 1. Attendees: Installer and representatives of manufacturers and fabricators involved in or affected by the installation and its coordination or integration with other materials and installations that have preceded or will follow, shall attend the meeting. Advise Architect of scheduled meeting dates.
  - 2. Agenda: Review progress of other construction activities and preparations for the particular activity under consideration, including requirements for the following:
    - a. The Contract Documents.
    - b. Options.
    - c. Related RFIs.
    - d. Related Change Orders.
    - e. Purchases.
    - f. Deliveries.
    - g. Submittals.
    - h. Review of mockups.
    - i. Possible conflicts.
    - j. Compatibility problems.
    - k. Time schedules.
    - l. Weather limitations.
    - m. Manufacturer's written recommendations.
    - n. Warranty requirements.
    - o. Compatibility of materials.
    - p. Acceptability of substrates.
    - q. Temporary facilities and controls.
    - r. Space and access limitations.
    - s. Regulations of authorities having jurisdiction.
    - t. Testing and inspecting requirements.
    - u. Installation procedures.
    - v. Coordination with other work.
    - w. Required performance results.
    - x. Protection of adjacent work.
    - y. Protection of construction and personnel.
  - 3. Record significant conference discussions, agreements, and disagreements, including required corrective measures and actions.
  - 4. Reporting: Distribute minutes of the meeting to each party present and to parties who should have been present.
  - 5. Do not proceed with installation if the conference cannot be successfully concluded. Initiate whatever actions are necessary to resolve impediments to performance of the Work and reconvene the conference at earliest feasible date.
- D. Progress Meetings: Conduct progress meetings with Owner, Architect, General Contractor and Subcontractors at monthly intervals. Conduct weekly progress meetings with Architect, General

Contractor and Subcontractors. Coordinate dates of monthly meetings with preparation of payment requests.

- 1. Attendees: In addition to representatives of Owner and Architect, each contractor, subcontractor, supplier, and other entity concerned with current progress or involved in planning, coordination, or performance of future activities shall be represented at these meetings. All participants at the conference shall be familiar with Project and authorized to conclude matters relating to the Work.
- 2. Agenda: Review and correct or approve minutes of previous progress meeting. Review other items of significance that could affect progress. Include topics for discussion as appropriate to status of Project.
  - a. Contractor's Construction Schedule: Review progress since the last meeting. Determine whether each activity is on time, ahead of schedule, or behind schedule, in relation to Contractor's Construction Schedule. Determine how construction behind schedule will be expedited; secure commitments from parties involved to do so. Discuss whether schedule revisions are required to ensure that current and subsequent activities will be completed within the Contract Time.
    - 1) Review schedule for next period.
  - b. Review present and future needs of each entity present, including the following:
    - 1) Interface requirements.
    - 2) Sequence of operations.
    - 3) Status of submittals.
    - 4) Deliveries.
    - 5) Off-site fabrication.
    - 6) Access.
    - 7) Site utilization.
    - 8) Temporary facilities and controls.
    - 9) Work hours.
    - 10) Hazards and risks.
    - 11) Progress cleaning.
    - 12) Quality and work standards.
    - 13) Status of correction of deficient items.
    - 14) Field observations.
    - 15) RFIs.
    - 16) Status of proposal requests.
    - 17) Pending changes.
    - 18) Status of Change Orders.
    - 19) Pending claims and disputes.
    - 20) Documentation of information for payment requests.
- 3. Minutes: Record the meeting minutes.
- 4. Reporting: Distribute minutes of the meeting to each party present and to parties who should have been present.
  - a. Schedule Updating: Revise Contractor's Construction Schedule after each progress meeting where revisions to the schedule have been made or recognized. Issue revised schedule concurrently with the report of each meeting.

# 1.6 REQUESTS FOR INTERPRETATION (RFIs)

- A. Procedure: Immediately on discovery of the need for interpretation of the Contract Documents, and if not possible to request interpretation at Project meeting, prepare and submit an RFI in the form specified.
  - 1. RFIs shall originate with Contractor. RFIs submitted by entities other than Contractor will be returned with no response.
  - 2. Coordinate and submit RFIs in a prompt manner so as to avoid delays in Contractor's work or work of subcontractors.
- B. Content of the RFI: Include a detailed, legible description of item needing interpretation and the following:
  - 1. Project name.
  - 2. Date.
  - 3. Name of Contractor.
  - 4. Name of Architect.
  - 5. RFI number, numbered sequentially.
  - 6. Specification Section number and title and related paragraphs, as appropriate.
  - 7. Drawing number and detail references, as appropriate.
  - 8. Field dimensions and conditions, as appropriate.
  - 9. Contractor's suggested solution(s). If Contractor's solution(s) impact the Contract Time or the Contract Sum, Contractor shall state impact in the RFI.
  - 10. Contractor's signature.
  - 11. Attachments: Include drawings, descriptions, measurements, photos, Product Data, Shop Drawings, and other information necessary to fully describe items needing interpretation.
- C. Hard-Copy RFIs: CSI Form 13.2A.
  - 1. Identify each page of attachments with the RFI number and sequential page number.
- D. Architect's Action: Architect will review each RFI, determine action required, and return it. Allow seven working days for Architect's response for each RFI. RFIs received after 1:00 p.m. will be considered as received the following working day.
  - 1. The following RFIs will be returned without action:
    - a. Requests for approval of submittals.
    - b. Requests for approval of substitutions.
    - c. Requests for coordination information already indicated in the Contract Documents.
    - d. Requests for adjustments in the Contract Time or the Contract Sum.
    - e. Requests for interpretation of Architect's actions on submittals.
    - f. Incomplete RFIs or RFIs with numerous errors.
  - 2. Architect's action may include a request for additional information, in which case Architect's time for response will start again.

- M.A. No. H1-22029-DA RMG - 22029
- 3. Architect's action on RFIs that may result in a change to the Contract Time or the Contract Sum may be eligible for Contractor to submit Change Proposal according to Division 01 Section "Contract Modification Procedures."
  - a. If Contractor believes the RFI response warrants change in the Contract Time or the Contract Sum, notify Architect in writing within 10 days of receipt of the RFI response.
- E. On receipt of Architect's action, update the RFI log and immediately distribute the RFI response to affected parties. Review response and notify Architect within seven days if Contractor disagrees with response.
- F. RFI Log: Prepare, maintain, and submit a tabular log of RFIs organized by the RFI number. Submit log weekly.
  - 1. Project name.
  - 2. Name and address of Contractor.
  - 3. Name and address of Architect.
  - 4. RFI number including RFIs that were dropped and not submitted.
  - 5. RFI description.
  - 6. Date the RFI was submitted.
  - 7. Date Architect's response was received.
  - 8. Identification of related Minor Change in the Work, Construction Change Directive, and Proposal Request, as appropriate.
  - 9. Identification of related Field Order, Work Change Directive, and Proposal Request, as appropriate.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 31 00

This page intentionally left blank

#### SECTION 01 32 00 - CONSTRUCTION PROGRESS DOCUMENTATION

#### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. Section includes administrative and procedural requirements for documenting the progress of construction during performance of the Work, including the following:
  - 1. Contractor's Construction Schedule.
  - 2. Construction schedule updating reports.
  - 3. Weekly construction reports.
  - 4. Site condition reports.

## 1.2 DEFINITIONS

- A. Activity: A discrete part of a project that can be identified for planning, scheduling, monitoring, and controlling the construction Project. Activities included in a construction schedule consume time and resources.
  - 1. Critical Activity: An activity on the critical path that must start and finish on the planned early start and finish times.
  - 2. Predecessor Activity: An activity that precedes another activity in the network.
  - 3. Successor Activity: An activity that follows another activity in the network.
- B. CPM: Critical path method, which is a method of planning and scheduling a construction project where activities are arranged based on activity relationships. Network calculations determine when activities can be performed and the critical path of Project.
- C. Critical Path: The longest connected chain of interdependent activities through the network schedule that establishes the minimum overall Project duration and contains no float.
- D. Event: The starting or ending point of an activity.
- E. Float: The measure of leeway in starting and completing an activity.
  - 1. Float time belongs to the contractor.
  - 2. Free float is the amount of time an activity can be delayed without adversely affecting the early start of the successor activity.
  - 3. Total float is the measure of leeway in starting or completing an activity without adversely affecting the planned Project completion date.

## 1.3 INFORMATIONAL SUBMITTALS

- A. Format for Submittals: Submit required submittals in the following format:
  - 1. Working electronic/native copy of schedule file.

M.A. No. H1-22029-DA RMG - 22029

- 2. PDF file.
- 3. Two paper copies, of sufficient size to display entire period or schedule, as required.
- B. Startup Network Diagram: Of size required to display entire network for entire construction period. Show logic ties for activities.
- C. Contractor's Construction Schedule: Initial schedule, of size required to display entire schedule for entire construction period.
  - 1. Submit a working digital copy of schedule, using software indicated, and labeled to comply with requirements for submittals.
- D. CPM Reports: Concurrent with CPM schedule, submit each of the following reports. Format for each activity in reports shall contain activity number, activity description, original duration, remaining duration, early start date, early finish date, late start date, late finish date, and total float in calendar days.
  - 1. Activity Report: List of activities sorted by activity number and then early start date, or actual start date if known.
  - 2. Logic Report: List of preceding and succeeding activities for each activity, sorted in ascending order by activity number and then by early start date, or actual start date if known.
  - 3. Total Float Report: List of activities sorted in ascending order of total float.
- E. Construction Schedule Updating Reports: Submit with Applications for Payment.
- F. Weekly Construction Reports: Submit to Architect at weekly intervals, have available for review at all progress meetings.
- G. Site Condition Reports: Submit at time of discovery of differing conditions.

#### 1.4 COORDINATION

- A. Coordinate Contractor's Construction Schedule with the schedule of values, list of subcontracts, submittal schedule, progress reports, payment requests, and other required schedules and reports.
  - 1. Secure time commitments for performing critical elements of the Work from entities involved.
  - 2. Coordinate each construction activity in the network with other activities and schedule them in proper sequence.

#### 1.5 CONTRACTOR'S CONSTRUCTION SCHEDULE, GENERAL

- A. Computer Scheduling Software: Prepare schedules using current version of a program that has been developed specifically to manage construction schedules.
  - 1. Use Microsoft Project or Primavera, for current Windows operating system.

- M.A. No. H1-22029-DA RMG - 22029
- B. Time Frame: Extend schedule from date established for the Notice to Proceed to date of final completion.
  - 1. Contract completion date shall not be changed by submission of a schedule that shows an early completion date, unless specifically authorized by Change Order.
- C. Activities: Treat each floor or separate area as a separate numbered activity for each main element of the Work. Comply with the following:
  - 1. Activity Duration: Define activities so no activity is longer than 20 days, unless specifically allowed by Architect.
  - 2. Procurement Activities: Include procurement process activities for the following long lead items and major items, requiring a cycle of more than 60 days, as separate activities in schedule. Procurement cycle activities include, but are not limited to, submittals, approvals, purchasing, fabrication, and delivery.
    - a. List of major items or pieces of equipment critical to overall schedule.
  - 3. Submittal Review Time: Include review and resubmittal times indicated in Section 013300 "Submittal Procedures" in schedule. Coordinate submittal review times in Contractor's Construction Schedule with submittal schedule.
  - 4. Startup and Testing Time: Include no fewer than 15 days for startup and testing.
  - 5. Commissioning Time: Include no fewer than 15 days for commissioning.
  - 6. Substantial Completion: Indicate completion in advance of date established for Substantial Completion, and allow time for Architect's administrative procedures necessary for certification of Substantial Completion.
  - 7. Punch List and Final Completion: Include not more than 45 days for completion of punch list items and final completion.
- D. Constraints: Include constraints and work restrictions indicated in the Contract Documents and as follows in schedule, and show how the sequence of the Work is affected.
  - 1. Phasing: Arrange list of activities on schedule by phase.
  - 2. Owner-Furnished Products: Include a separate activity for each product. Include delivery date indicated in Section 01 10 00 "Summary." Delivery dates indicated stipulate the earliest possible delivery date.
  - 3. Work Restrictions: Show the effect of the following items on the schedule:
    - a. Coordination with existing construction.
    - b. Limitations of continued occupancies.
    - c. Uninterruptible services.
    - d. Partial occupancy before Substantial Completion.
    - e. Use-of-premises restrictions.
    - f. Provisions for future construction.
    - g. Seasonal variations.
    - h. Environmental control.
- E. Milestones: Include milestones indicated in the Contract Documents in schedule, including, but not limited to, the Notice to Proceed, Substantial Completion, and final completion and the following interim milestones:
  - 1. Start and completion of work for all major trades.

- M.A. No. H1-22029-DA RMG - 22029
- F. Upcoming Work Summary: Prepare summary report indicating activities scheduled to occur or commence prior to submittal of next schedule update. Summarize the following issues:
  - 1. Unresolved issues.
  - 2. Unanswered Requests for Information.
  - 3. Rejected or unreturned submittals.
  - 4. Notations on returned submittals.
  - 5. Pending modifications affecting the Work and the Contract Time.
- G. Contractor's Construction Schedule Updating: At monthly intervals, update schedule to reflect actual construction progress and activities. Issue schedule one week before each regularly scheduled progress meeting.
  - 1. Revise schedule immediately after each meeting or other activity where revisions have been recognized or made. Issue updated schedule concurrently with the report of each such meeting.
  - 2. Include a report with updated schedule that indicates every change, including, but not limited to, changes in logic, durations, actual starts and finishes, and activity durations.
  - 3. As the Work progresses, indicate final completion percentage for each activity.
- H. Recovery Schedule: When periodic update indicates the Work is 14 or more calendar days behind the current approved schedule, submit a separate recovery schedule indicating means by which Contractor intends to regain compliance with the schedule. Indicate changes to working hours, working days, crew sizes, equipment required to achieve compliance, and date by which recovery will be accomplished.
- I. Distribution: Distribute copies of approved schedule to Architect Owner, separate contractors, testing and inspecting agencies, and other parties identified by Contractor with a need-to-know schedule responsibility.
  - 1. Post copies in Project meeting rooms and temporary field offices.
  - 2. When revisions are made, distribute updated schedules to the same parties and post in the same locations. Delete parties from distribution when they have completed their assigned portion of the Work and are no longer involved in performance of construction activities.

# 1.6 GANTT-CHART SCHEDULE REQUIREMENTS

- A. Gantt-Chart Schedule: Submit a comprehensive, fully developed, horizontal, Gantt-chart-type, Contractor's Construction Schedule within 30 days of date established for the Notice to Proceed.
- B. Preparation: Indicate each significant construction activity separately. Identify first workday of each week with a continuous vertical line.
  - 1. For construction activities that require three months or longer to complete, indicate an estimated completion percentage in 10 percent increments within time bar.

# 1.7 CPM SCHEDULE REQUIREMENTS

A. General: Prepare network diagrams using AON (activity-on-node) format.

- B. Startup Network Diagram: Submit diagram within 14 days of date established for the Notice to Proceed. Outline significant construction activities for the first 90 days of construction. Include skeleton diagram for the remainder of the Work and a cash requirement prediction based on indicated activities.
- C. CPM Schedule: Prepare Contractor's Construction Schedule using a time-scaled CPM network analysis diagram for the Work.
  - 1. Develop network diagram in sufficient time to submit CPM schedule so it can be accepted for use no later than 60 days after date established for the Notice to Proceed.
    - a. Failure to include any work item required for performance of this Contract shall not excuse Contractor from completing all work within applicable completion dates.
  - 2. Conduct educational workshops to train and inform key Project personnel, including subcontractors' personnel, in proper methods of providing data and using CPM schedule information
  - 3. Establish procedures for monitoring and updating CPM schedule and for reporting progress. Coordinate procedures with progress meeting and payment request dates.
  - 4. Use "one workday" as the unit of time for individual activities. Indicate nonworking days and holidays incorporated into the schedule to coordinate with the Contract Time.
- D. CPM Schedule Preparation: Prepare a list of all activities required to complete the Work. Using the startup network diagram, prepare a skeleton network to identify probable critical paths.
  - 1. Activities: Indicate the estimated time duration, sequence requirements, and relationship of each activity in relation to other activities. Include estimated time frames for the following activities:
    - a. Preparation and processing of submittals.
    - b. Mobilization and demobilization.
    - c. Purchase of materials.
    - d. Delivery.
    - e. Fabrication.
    - f. Utility interruptions.
    - g. Installation.
    - h. Work by Owner that may affect or be affected by Contractor's activities.
    - i. Testing and inspection.
    - i. Commissioning.
    - k. Punch list and final completion.
    - 1. Activities occurring following final completion.
  - 2. Critical Path Activities: Identify critical path activities, including those for interim completion dates. Scheduled start and completion dates shall be consistent with Contract milestone dates.
  - 3. Processing: Process data to produce output data on a computer-drawn, time-scaled network. Revise data, reorganize activity sequences, and reproduce as often as necessary to produce the CPM schedule within the limitations of the Contract Time.
  - 4. Format: Mark the critical path. Locate the critical path near center of network; locate paths with most float near the edges.

- M.A. No. H1-22029-DA RMG - 22029
- a. Subnetworks on separate sheets are permissible for activities clearly off the critical path.
- E. Contract Modifications: For each proposed contract modification and concurrent with its submission, prepare a time-impact analysis using a network fragment to demonstrate the effect of the proposed change on the overall Project schedule.
- F. Initial Issue of Schedule: Prepare initial network diagram from a sorted activity list indicating straight "early start-total float." Identify critical activities. Prepare tabulated reports showing the following:
  - 1. Contractor or subcontractor and the Work or activity.
  - 2. Description of activity.
  - 3. Main events of activity.
  - 4. Immediate preceding and succeeding activities.
  - 5. Early and late start dates.
  - 6. Early and late finish dates.
  - 7. Activity duration in workdays.
  - 8. Total float or slack time.
  - 9. Average size of workforce.
  - 10. Dollar value of activity (coordinated with the schedule of values).
- G. Schedule Updating: Concurrent with making revisions to schedule, prepare tabulated reports showing the following:
  - 1. Identification of activities that have changed.
  - 2. Changes in early and late start dates.
  - 3. Changes in early and late finish dates.
  - 4. Changes in activity durations in workdays.
  - 5. Changes in the critical path.
  - 6. Changes in total float or slack time.
  - 7. Changes in the Contract Time.
- H. Value Summaries: Prepare two cumulative value lists, sorted by finish dates.
  - 1. In first list, tabulate activity number, early finish date, dollar value, and cumulative dollar value.
  - 2. In second list, tabulate activity number, late finish date, dollar value, and cumulative dollar value.
  - 3. In subsequent issues of both lists, substitute actual finish dates for activities completed as of list date.
  - 4. Prepare list for ease of comparison with payment requests; coordinate timing with progress meetings.
    - a. In both value summary lists, tabulate "actual percent complete" and "cumulative value completed" with total at bottom.
    - b. Submit value summary printouts one week before each regularly scheduled progress meeting.

## 1.8 REPORTS

- A. Weekly Construction Reports: Prepare a weekly construction report recording the following information concerning events at Project site:
  - 1. List of subcontractors at Project site.
  - 2. List of separate contractors at Project site.
  - 3. Approximate count of personnel at Project site.
  - 4. Equipment at Project site.
  - 5. Material deliveries.
  - 6. High and low temperatures and general weather conditions, including presence of rain or snow.
  - 7. Testing and inspection.
  - 8. Accidents.
  - 9. Meetings and significant decisions.
  - 10. Stoppages, delays, shortages, and losses.
  - 11. Meter readings and similar recordings.
  - 12. Emergency procedures.
  - 13. Orders and requests of authorities having jurisdiction.
  - 14. Change Orders received and implemented.
  - 15. Construction Change Directives received and implemented.
  - 16. Services connected and disconnected.
  - 17. Equipment or system tests and startups.
  - 18. Partial completions and occupancies.
  - 19. Substantial Completions authorized.
- B. Site Condition Reports: Immediately on discovery of a difference between site conditions and the Contract Documents, prepare and submit a detailed report. Submit with a Request for Information. Include a detailed description of the differing conditions, together with recommendations for changing the Contract Documents.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 32 00

This page intentionally left blank

### SECTION 01 32 33 - PHOTOGRAPHIC DOCUMENTATION

### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. This Section includes administrative and procedural requirements for the following:
  - 1. Preconstruction photographs.
  - 2. Periodic construction photographs.
- B. See Division 01 Section "Closeout Procedures" for submitting digital media as Project Record Documents at Project closeout.
- C. See Division 01 Section "Demonstration and Training" for submitting videotapes of demonstration of equipment and training of Owner's personnel.

#### 1.2 SUBMITTALS

- A. Key Plan: Submit key plan of Project site and building with notation of vantage points marked for location and direction of each photograph. Indicate elevation or story of construction. Include same label information as corresponding set of photographs.
- B. Construction Photographs: Submit two prints of each photographic view within seven days of taking photographs.
  - 1. Digital Images: Submit a complete set of digital image electronic files as a Project Record Document on CD-ROM. Identify electronic media with date photographs were taken. Submit images that have same aspect ratio as the sensor, uncropped.

#### 1.3 QUALITY ASSURANCE

### 1.4 COORDINATION

A. Auxiliary Services: Cooperate with photographer and provide auxiliary services requested, including access to Project site and use of temporary facilities, including temporary lighting required to produce clear, well-lit photographs without obscuring shadows.

## PART 2 - PRODUCTS

## 2.1 PHOTOGRAPHIC MEDIA

A. Digital Images: Provide images in JPG format, produced by a digital camera with minimum sensor size of 12.0 megapixels, and at an image resolution of not less than 1024 by 768 pixels.

#### PART 3 - EXECUTION

## 3.1 CONSTRUCTION PHOTOGRAPHS

- A. General: Take photographs using the maximum range of depth of field, and that are in focus, to clearly show the Work. Photographs with blurry or out-of-focus areas will not be accepted.
  - 1. Maintain key plan with each set of construction photographs that identifies each photographic location.
- B. Digital Images: Submit digital images exactly as originally recorded in the digital camera, without alteration, manipulation, editing, or modifications using image-editing software.
  - 1. Date and Time: Include date and time in filename for each image.
  - 2. Field Office Images: Maintain one set of images on CD-ROM in the field office at Project site, available at all times for reference. Identify images same as for those submitted to Architect.
- C. Preconstruction Photographs: Before commencement of excavation starting construction, take color, digital photographs of Project site and surrounding properties, including existing items to remain during construction, from different vantage points, as directed by Architect.
  - 1. Flag excavation areas construction limits before taking construction photographs.
  - 2. Take eight photographs to show existing conditions adjacent to property before starting the Work.
  - 3. Take eight photographs of existing buildings either on or adjoining property to accurately record physical conditions at start of construction.
- D. Periodic Construction Photographs: Take 12 color, digital photographs weekly, with timing each month adjusted to coincide with the cutoff date associated with each Application for Payment. Select vantage points to show status of construction and progress since last photographs were taken.
- E. Additional Photographs: Architect may issue requests for additional photographs, in addition to periodic photographs specified. Additional photographs will be paid for by Change Order and are not included in the Contract Sum.
  - 1. Three days' notice will be given, where feasible.
  - 2. In emergency situations, take additional photographs within 24 hours of request.
  - 3. Circumstances that could require additional photographs include, but are not limited to, the following:
    - a. Special events planned at Project site.
    - b. Immediate follow-up when on-site events result in construction damage or losses.
    - c. Photographs to be taken at fabrication locations away from Project site. These photographs are not subject to unit prices or unit-cost allowances.
    - d. Substantial Completion of a major phase or component of the Work.
    - e. Extra record photographs at time of final acceptance.
    - f. Owner's request for special publicity photographs.

M.A. No. H1-22029-DA RMG - 22029

END OF SECTION 01 32 33

This page intentionally left blank

### SECTION 01 33 00 - SUBMITTAL PROCEDURES

#### PART 1 - GENERAL

### 1.1 SUMMARY

#### A. Section Includes:

- 1. Submittal schedule requirements.
- 2. Administrative and procedural requirements for submittals.

#### 1.2 DEFINITIONS

- A. Action Submittals: Written and graphic information and physical samples that require Architect's and Construction Manager's responsive action. Action submittals are those submittals indicated in individual Specification Sections as "action submittals."
- B. Informational Submittals: Written and graphic information and physical samples that do not require Architect's responsive action. Submittals may be rejected for not complying with requirements. Informational submittals are those submittals indicated in individual Specification Sections as "informational submittals."

## 1.3 SUBMITTAL SCHEDULE

A. Submittal Schedule: Submit, as an action submittal, a list of submittals, arranged in chronological order by dates required by construction schedule. Include time required for review, ordering, manufacturing, fabrication, and delivery when establishing dates. Include additional time required for making corrections or revisions to submittals noted by Architect and additional time for handling and reviewing submittals required by those corrections.

#### 1.4 SUBMITTAL FORMATS

- A. Submittal Information: Include the following information in each submittal:
  - 1. Project name.
  - 2. Date.
  - 3. Name of Architect.
  - 4. Name of Contractor.
  - 5. Name of firm or entity that prepared submittal.
  - 6. Names of subcontractor, manufacturer, and supplier.
  - 7. Unique submittal number, including revision identifier. Include Specification Section number with sequential alphanumeric identifier; and alphanumeric suffix for resubmittals.
  - 8. Category and type of submittal.
  - 9. Submittal purpose and description.
  - 10. Number and title of Specification Section, with paragraph number and generic name for each of multiple items.

- M.A. No. H1-22029-DA RMG - 22029
- 11. Drawing number and detail references, as appropriate.
- 12. Indication of full or partial submittal.
- 13. Location(s) where product is to be installed, as appropriate.
- 14. Other necessary identification.
- 15. Remarks.
- 16. Signature of transmitter.
- B. Options: Identify options requiring selection by Architect.
- C. Deviations and Additional Information: On each submittal, clearly indicate deviations from requirements in the Contract Documents, including minor variations and limitations; include relevant additional information and revisions, other than those requested by Architect on previous submittals. Indicate by highlighting on each submittal or noting on attached separate sheet.
- D. Submittals for Utilizing Web-Based Project Management Software: Prepare submittals as PDF files, or other format indicated by Project management software.

# 1.5 SUBMITTAL PROCEDURES

- A. Prepare and submit submittals required by individual Specification Sections. Types of submittals are indicated in individual Specification Sections.
  - 1. Web-Based Project Management Software: Prepare submittals in PDF form, and upload to web-based Project management software website. Enter required data in web-based software site to fully identify submittal.
- B. Coordination: Coordinate preparation and processing of submittals with performance of construction activities.
  - 1. Coordinate each submittal with fabrication, purchasing, testing, delivery, other submittals, and related activities that require sequential activity.
  - 2. Submit all submittal items required for each Specification Section concurrently unless partial submittals for portions of the Work are indicated on approved submittal schedule.
  - 3. Submit action submittals and informational submittals required by the same Specification Section as separate packages under separate transmittals.
- C. Processing Time: Allow time for submittal review, including time for resubmittals, as follows. Time for review shall commence on Architect's receipt of submittal. No extension of the Contract Time will be authorized because of failure to transmit submittals enough in advance of the Work to permit processing, including resubmittals.
  - 1. Initial Review: Allow 15 days for initial review of each submittal. Allow additional time if coordination with subsequent submittals is required. Architect will advise Contractor when a submittal being processed must be delayed for coordination.
  - 2. Resubmittal Review: Allow 15 days for review of each resubmittal.
- D. Resubmittals: Make resubmittals in same form as initial submittal.
- E. Distribution: Furnish copies of final submittals to manufacturers, subcontractors, suppliers, fabricators, installers, authorities having jurisdiction, and others as necessary for performance of construction activities. Show distribution on transmittal forms.

M.A. No. H1-22029-DA RMG - 22029

F. Use for Construction: Retain complete copies of submittals on Project site. Use only final action submittals that are marked with approval notation from Architect's action stamp.

# 1.6 SUBMITTAL REQUIREMENTS

- A. Product Data: Collect information into a single submittal for each element of construction and type of product or equipment.
  - 1. If information must be specially prepared for submittal because standard published data are unsuitable for use, submit as Shop Drawings, not as Product Data.
  - 2. Mark each copy of each submittal to show which products and options are applicable.
  - 3. Include the following information, as applicable:
    - a. Manufacturer's catalog cuts.
    - b. Manufacturer's product specifications.
    - c. Standard color charts.
    - d. Statement of compliance with specified referenced standards.
    - e. Testing by recognized testing agency.
    - f. Application of testing agency labels and seals.
    - g. Notation of coordination requirements.
    - h. Availability and delivery time information.
  - 4. For equipment, include the following in addition to the above, as applicable:
    - a. Wiring diagrams that show factory-installed wiring.
    - b. Printed performance curves.
    - c. Operational range diagrams.
    - d. Clearances required to other construction, if not indicated on accompanying Shop Drawings.
  - 5. Submit Product Data before Shop Drawings, and before or concurrent with Samples.
- B. Shop Drawings: Prepare Project-specific information, drawn accurately to scale. Do not base Shop Drawings on reproductions of the Contract Documents or standard printed data.
  - 1. Preparation: Fully illustrate requirements in the Contract Documents. Include the following information, as applicable:
    - a. Identification of products.
    - b. Schedules.
    - c. Compliance with specified standards.
    - d. Notation of coordination requirements.
    - e. Notation of dimensions established by field measurement.
    - f. Relationship and attachment to adjoining construction clearly indicated.
    - g. Seal and signature of professional engineer if specified.
- C. Samples: Submit Samples for review of kind, color, pattern, and texture for a check of these characteristics with other materials.

- 1. Transmit Samples that contain multiple, related components such as accessories together in one submittal package.
- 2. Identification: Permanently attach label on unexposed side of Samples that includes the following:
  - a. Project name and submittal number.
  - b. Generic description of Sample.
  - c. Product name and name of manufacturer.
  - d. Sample source.
  - e. Number and title of applicable Specification Section.
  - f. Specification paragraph number and generic name of each item.
- 3. Web-Based Project Management Software: Prepare submittals in PDF form, and upload to web-based Project software website. Enter required data in web-based software site to fully identify submittal.
- 4. Disposition: Maintain sets of approved Samples at Project site, available for quality-control comparisons throughout the course of construction activity. Sample sets may be used to determine final acceptance of construction associated with each set.
  - a. Samples that may be incorporated into the Work are indicated in individual Specification Sections. Such Samples must be in an undamaged condition at time of use.
  - b. Samples not incorporated into the Work, or otherwise designated as Owner's property, are the property of Contractor.
- 5. Samples for Initial Selection: Submit manufacturer's color charts consisting of units or sections of units showing the full range of colors, textures, and patterns available.
  - a. Number of Samples: Submit one full set(s) of available choices where color, pattern, texture, or similar characteristics are required to be selected from manufacturer's product line. Architect will return submittal with options selected.
- 6. Samples for Verification: Submit full-size units or Samples of size indicated, prepared from same material to be used for the Work, cured and finished in manner specified, and physically identical with material or product proposed for use, and that show full range of color and texture variations expected. Samples include, but are not limited to, the following: partial sections of manufactured or fabricated components; small cuts or containers of materials; complete units of repetitively used materials; swatches showing color, texture, and pattern; color range sets; and components used for independent testing and inspection.
  - a. Number of Samples: Submit three sets of Samples. Architect will retain two Sample sets; remainder will be returned.
    - 1) Submit a single Sample where assembly details, workmanship, fabrication techniques, connections, operation, and other similar characteristics are to be demonstrated.
    - 2) If variation in color, pattern, texture, or other characteristic is inherent in material or product represented by a Sample, submit at least three sets of paired units that show approximate limits of variations.

- D. Product Schedule: As required in individual Specification Sections, prepare a written summary indicating types of products required for the Work and their intended location. Include the following information in tabular form:
- E. Qualification Data: Prepare written information that demonstrates capabilities and experience of firm or person. Include lists of completed projects with project names and addresses, contact information of architects and owners, and other information specified.
- F. Design Data: Prepare and submit written and graphic information indicating compliance with indicated performance and design criteria in individual Specification Sections. Include list of assumptions and summary of loads. Include load diagrams if applicable. Provide name and version of software, if any, used for calculations. Number each page of submittal.

#### G. Certificates:

- 1. Certificates and Certifications Submittals: Submit a statement that includes signature of entity responsible for preparing certification. Certificates and certifications shall be signed by an officer or other individual authorized to sign documents on behalf of that entity. Provide a notarized signature where indicated.
- 2. Installer Certificates: Submit written statements on manufacturer's letterhead certifying that Installer complies with requirements in the Contract Documents and, where required, is authorized by manufacturer for this specific Project.
- 3. Manufacturer Certificates: Submit written statements on manufacturer's letterhead certifying that manufacturer complies with requirements in the Contract Documents. Include evidence of manufacturing experience where required.
- 4. Material Certificates: Submit written statements on manufacturer's letterhead certifying that material complies with requirements in the Contract Documents.
- 5. Product Certificates: Submit written statements on manufacturer's letterhead certifying that product complies with requirements in the Contract Documents.
- 6. Welding Certificates: Prepare written certification that welding procedures and personnel comply with requirements in the Contract Documents. Submit record of Welding Procedure Specification and Procedure Qualification Record on AWS forms. Include names of firms and personnel certified.

## H. Test and Research Reports:

- 1. Compatibility Test Reports: Submit reports written by a qualified testing agency, on testing agency's standard form, indicating and interpreting results of compatibility tests performed before installation of product. Include written recommendations for substrate preparation and primers required.
- 2. Field Test Reports: Submit written reports indicating and interpreting results of field tests performed either during installation of product or after product is installed in its final location, for compliance with requirements in the Contract Documents.
- 3. Material Test Reports: Submit reports written by a qualified testing agency, on testing agency's standard form, indicating and interpreting test results of material for compliance with requirements in the Contract Documents.
- 4. Preconstruction Test Reports: Submit reports written by a qualified testing agency, on testing agency's standard form, indicating and interpreting results of tests performed before installation of product, for compliance with performance requirements in the Contract Documents.

- 5. Product Test Reports: Submit written reports indicating that current product produced by manufacturer complies with requirements in the Contract Documents. Base reports on evaluation of tests performed by manufacturer and witnessed by a qualified testing agency, or on comprehensive tests performed by a qualified testing agency.
- 6. Research Reports: Submit written evidence, from a model code organization acceptable to authorities having jurisdiction, that product complies with building code in effect for Project. Include the following information:
  - a. Name of evaluation organization.
  - b. Date of evaluation.
  - c. Time period when report is in effect.
  - d. Product and manufacturers' names.
  - e. Description of product.
  - f. Test procedures and results.
  - g. Limitations of use.

## 1.7 DELEGATED DESIGN SERVICES

- A. Performance and Design Criteria: Where professional design services or certifications by a design professional are specifically required of Contractor by the Contract Documents, provide products and systems complying with specific performance and design criteria indicated.
  - 1. If criteria indicated are insufficient to perform services or certification required, submit a written request for additional information to Architect.
- B. Delegated Design Services Certification: In addition to Shop Drawings, Product Data, and other required submittals, submit digitally signed PDF file, signed and sealed by the responsible design professional, for each product and system specifically assigned to Contractor to be designed or certified by a design professional.
  - 1. Indicate that products and systems comply with performance and design criteria in the Contract Documents. Include list of codes, loads, and other factors used in performing these services.

### 1.8 CONTRACTOR'S REVIEW

- A. Action Submittals and Informational Submittals: Review each submittal and check for coordination with other Work of the Contract and for compliance with the Contract Documents. Note corrections and field dimensions. Mark with approval stamp before submitting to Architect.
- B. Contractor's Approval: Indicate Contractor's approval for each submittal with a uniform approval stamp. Include name of reviewer, date of Contractor's approval, and statement certifying that submittal has been reviewed, checked, and approved for compliance with the Contract Documents.
  - 1. Architect will not review submittals received from Contractor that do not have Contractor's review and approval.

## 1.9 ARCHITECT'S REVIEW

- A. Action Submittals: Architect will review each submittal, indicate corrections or revisions required, and return it.
  - 1. Submittals by Web-Based Project Management Software: Architect will add a review stamp to the pdf file on Project management software website.
    - a. Actions taken by indication on the Architectural Review Stamp have the following meanings:
      - 1) No Exception Taken item is as specified and any/all option selected by contractor are approved.
      - 2) Make Corrections Noted indicates comments or selections included by Architect are to be implemented/coordinated in final product by contractor.
      - 3) Revise and Resubmit indicates comments made are to be implemented and resubmitted for approval.
      - 4) Submit Specified Item item submitted does not meet the specifications and requires resubmittal.
      - 5) Rejected item submitted is not acceptable.
- B. Informational Submittals: Architect will review each submittal and will return it if it does not comply with requirements. Architect will forward each submittal to appropriate party.
- C. Partial submittals prepared for a portion of the Work will be reviewed when use of partial submittals has received prior approval from Architect.
- D. Incomplete submittals are unacceptable, will be considered nonresponsive, and will be returned for resubmittal without review.
- E. Architect will discard submittals received from sources other than Contractor.

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 33 00

This page intentionally left blank

# SECTION 01 40 00 - QUALITY REQUIREMENTS

#### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. This Section includes administrative and procedural requirements for quality assurance and quality control of specified materials and systems.
- B. Mockups required to verify quality of workmanship and products included in multiple-component assemblies
- C. Testing and inspecting services are required to verify compliance with requirements specified or indicated. These services do not relieve Contractor of responsibility for compliance with the Contract Document requirements.
  - 1. Refer to Division 03, Division 05, Division 31 and Division 32 for specific testing required under those Sections.
  - 2. Specified tests, inspections, and related actions do not limit Contractor's other quality-assurance and -control procedures that facilitate compliance with the Contract Document requirements.
  - 3. Requirements for Contractor to provide quality-assurance and -control services required by Architect, Owner, or authorities having jurisdiction are not limited by provisions of this Section.
- D. See Divisions 02 through 49 Sections for specific tests, inspections, and quality control requirements.

# 1.2 DEFINITIONS

- A. Quality-Assurance Services: Activities, actions, and procedures performed before and during execution of the Work to guard against defects and deficiencies and substantiate that proposed construction will comply with requirements.
- B. Quality-Control Services: Tests, inspections, procedures, and related actions during and after execution of the Work to evaluate that actual products incorporated into the Work and completed construction comply with requirements. Services do not include contract enforcement activities performed by Architect.
- C. Mockups: Full-size, physical assemblies that are constructed on-site. Mockups are used to verify selections made under sample submittals, to demonstrate aesthetic effects and, where indicated, qualities of materials and execution, and to review construction, coordination, testing, or operation; they are not Samples. Approved mockups establish the standard by which the Work will be judged.
- D. Field Quality-Control Testing: Tests and inspections that are performed on-site for installation of the Work and for completed Work.
- E. Testing Agency: An entity engaged to perform specific tests, inspections, or both. Testing laboratory shall mean the same as testing agency.
- F. Experienced: When used with an entity, "experienced" means having successfully completed a minimum of five previous projects similar in size and scope to this Project; being familiar with

special requirements indicated; and having complied with requirements of authorities having jurisdiction.

### 1.3 MOCKUPS

- A. General: Review individual Specification Sections for required mockups to be constructed.
- B. Mockups: Before installing portions of the Work requiring mockups, build mockups for each form of construction and finish required to comply with the following requirements, using materials indicated for the completed Work:
  - 1. Build mockups in location and of size indicated or, if not indicated, as directed by Architect.
  - 2. Notify Architect seven days in advance of dates and times when mockups will be constructed.
  - 3. Demonstrate the proposed range of aesthetic effects and workmanship.
  - 4. Obtain Architect's approval of mockups before starting work, fabrication, or construction.
  - 5. Maintain mockups during construction in an undisturbed condition as a standard for judging the completed Work.
  - 6. Demolish and remove mockups when directed, unless otherwise indicated.

#### 1.4 RESPONSIBILITIES

- A. Owner Responsibilities: Owner will engage and pay for a qualified testing agency, accredited by AASHTO to perform required testing services.
  - 1. Owner will furnish Contractor with names, addresses, and telephone numbers of testing agency engaged and a description of types of testing and inspecting they are engaged to perform.
  - 2. Costs for retesting and reinspecting construction that replaces or is necessitated by work that failed to comply with the Contract Documents will be charged to Contractor, and the Contract Sum will be adjusted by Change Order.

### B. Contractor's responsibilities.

- 1. Notify testing agencies sufficiently in advance of operations to allow for completion of tests and assignment of inspection personnel; notify at least 48 hours in advance of time when Work that requires testing or inspecting will be performed.
- 2. Notify the laboratory sufficiently in advance of cancellation of required or scheduled testing operations. The Contractor shall be responsible for charges by testing agency due to failure to notify testing agency if requirements for testing are canceled.
- 3. Furnish nominal labor and sheltered working space as is necessary to obtain samples at the Project.
- 4. Advise testing agency identity of materials sources and instruct suppliers to allow tests or inspections by the laboratory.
- 5. In the event of construction which failed initial testing, the cost of re-testing and reinspecting replacement Work is Contractor's responsibility.

- 6. In the event of construction which failed quality standards-of-construction set forth in the Specification Section, the cost of re-testing and re-inspecting replacement Work is Contractor's responsibility
- 7. Re-testing and re-inspections are to be performed by Testing Agency that has been engaged by Owner.
- 8. Allow testing agency access to the Work, including providing lifts, ladders, etc. which may be required for testing agency to access area where the Work is being performed.
- C. Testing Agency Responsibilities: Cooperate with Architect and Contractor in performance of duties. Provide qualified personnel to perform required tests and inspections.
  - 1. Notify Architect and Contractor promptly of irregularities or deficiencies observed in the Work during performance of its services.
  - 2. Notify Architect and Contractor promptly if any hazardous materials are discovered.
  - 3. Determine the location from which test samples will be taken and in which in-situ tests are conducted.
  - 4. Conduct and interpret tests and inspections and state in each report whether tested and inspected work complies with or deviates from requirements.
  - 5. Submit a certified written report to Architect, Owner, and Contractor of each test, inspection, and similar quality-control service performed.
  - 6. Do not release, revoke, alter, or increase the Contract Document requirements nor approve or accept any portion of the Work.
  - 7. Do not perform any duties of Contractor.
- D. Related Services by Contractor and/or Subcontractor: Cooperate with agencies performing required tests, inspections, and similar quality-control services, and provide reasonable auxiliary services as requested. Notify agency sufficiently in advance of operations to permit assignment of personnel. Provide the following:
  - 1. Access to the Work.
  - 2. Incidental labor and facilities necessary to facilitate tests and inspections.
  - 3. Adequate quantities of representative samples of materials that require testing and inspecting. Assist agency in obtaining samples.
  - 4. Facilities for storage and field curing of test samples.
  - 5. Delivery of samples to testing agencies.
  - 6. Preliminary design mix proposed for use for material mixes that require control by testing agency.
  - 7. Security and protection for samples and for testing and inspecting equipment at Project site.
- E. Coordination: Coordinate sequence of activities to accommodate required quality-assurance and -control services with a minimum of delay and to avoid necessity of removing and replacing construction to accommodate testing and inspecting.
  - 1. Schedule times for tests, inspections, obtaining samples, and similar activities.

### 1.5 SPECIAL TESTS AND INSPECTIONS

- A. Special Tests and Inspections: Owner, at their option, may engage a qualified testing agency to conduct special tests and inspections for quality of products and materials.
- B. Minimum of three tests for Hurricane force Winds converted to pressure per ASTM.org/Standards/E1105.htm of Metal Panels, Insulated Concrete Panels, Glass Fiber Reinforced Panels, Windows, Storefront/Curtainwall per the following:
  - 1. Performance shall be evaluated in accordance with ASTM E1105-15 Standard Test Method for Field Determination of Water Penetration on exterior windows, skylight, doors and curtain walls by Uniform or Cyclic Static Air Pressure Difference.
  - 2. Replace materials, products or assemblies which fail to comply with specified quality of products and materials.
  - 3. Cost of re-testing and re-inspecting any failed and corrected work is responsibility of Contractor.

# PART 2 - PRODUCTS (Not Used)

### PART 3 - EXECUTION

### 3.1 REPAIR AND PROTECTION

- A. General: On completion of testing, inspecting, sample taking, and similar services, repair damaged construction and restore substrates and finishes.
  - 1. Provide materials and comply with installation requirements specified in other Specification Sections. Restore patched areas and extend restoration into adjoining areas with durable seams that are as invisible as possible and acceptable to Owner and Architect.
- B. Protect construction exposed by or for quality-control service activities.
- C. Repair and protection are Contractor's responsibility, regardless of the assignment of responsibility for quality-control services.

# END OF SECTION 01 40 00

#### SECTION 01 50 00 - TEMPORARY FACILITIES AND CONTROLS

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes requirements for temporary utilities, support facilities, and security and protection facilities.
- B. See Division 01 Section "Execution" for progress cleaning requirements.
- C. See Divisions 02 through 49 Sections for temporary heat, ventilation, and humidity requirements for products in those Sections.

### 1.2 DEFINITIONS

A. Permanent Enclosure: As determined by Architect, permanent or temporary roofing is complete, insulated, and weathertight; exterior walls are insulated and weathertight; and all openings are closed with permanent construction or substantial temporary closures.

### 1.3 USE CHARGES

- A. General: Cost or use charges for temporary facilities shall be included in the Contract Sum. Allow other entities to use temporary services and facilities without cost, including, but not limited to, Owner's construction forces, Architect, occupants of Project, testing agencies, and authorities having jurisdiction.
- B. Water Service: Provide connections and extensions of services as required for construction operations.
- C. Electric Power Service: Provide connections and extensions of services as required for construction operations.

## 1.4 SUBMITTALS

A. Site Plan: Show temporary facilities, utility hookups, staging areas, and parking areas for construction personnel.

### 1.5 OUALITY ASSURANCE

- A. Electric Service: Comply with NECA, NEMA, and UL standards and regulations for temporary electric service. Install service to comply with NFPA 70.
- B. Tests and Inspections: Arrange for authorities having jurisdiction to test and inspect each temporary utility before use. Obtain required certifications and permits.

### 1.6 PROJECT CONDITIONS

A. Temporary Use of Permanent Facilities: Installer of each permanent service shall assume responsibility for operation, maintenance, and protection of each permanent service during its use as a construction facility before Owner's acceptance, regardless of previously assigned responsibilities.

# TEMPORARY FACILITIES AND CONTROLS

## PART 2 - PRODUCTS

#### 2.1 MATERIALS for TEMPORARY CONSTRUCTIONS

- A. Pavement: Comply with Division 32 Section "Concrete Paving." pavement Sections.
- B. Wood Enclosure Fence: Plywood, 6 feet (1.8 m) high, framed with four 2-by-4-inch (50-by-100-mm) rails, with preservative-treated wood posts spaced not more than 8 feet (2.4 m) apart.
- C. Lumber and Plywood: Comply with requirements in Division 06 Section "Rough Carpentry Miscellaneous Rough Carpentry."
- D. Gypsum Board: Minimum 5/8" thick by 48 inches (1219 mm) wide by maximum available lengths; regular-type panels with tapered edges. Comply with ASTM C 36/C 36M.
- E. Insulation: Unfaced mineral-fiber blanket, manufactured from glass, slag wool, or rock wool; with maximum flame-spread and smoke-developed indexes of 25 and 50, respectively.

## 2.2 TEMPORARY FACILITIES

- A. Field Offices, General: Prefabricated or mobile units with serviceable finishes, temperature controls, and foundations adequate for normal loading.
- B. Storage and Fabrication Sheds: Provide sheds sized, furnished, and equipped to accommodate materials and equipment for construction operations.

## 2.3 EQUIPMENT

- A. Fire Extinguishers: Portable, UL rated; with class and extinguishing agent as required by locations and classes of fire exposures.
- B. HVAC Equipment: Unless Owner authorizes use of permanent HVAC system, provide vented, self-contained, liquid-propane-gas or fuel-oil heaters with individual space thermostatic control.
  - 1. Use of gasoline-burning space heaters, open-flame heaters, or salamander-type heating units is prohibited.
  - 2. Heating Units: Listed and labeled for type of fuel being consumed, by a testing agency acceptable to authorities having jurisdiction, and marked for intended use.
  - 3. Permanent HVAC System: If Owner authorizes use of permanent HVAC system for temporary use during construction, provide filter with MERV of 8 at each return air grille in system and remove at end of construction.

### PART 3 - EXECUTION

### 3.1 INSTALLATION, GENERAL

- A. Locate facilities where they will serve Project adequately and result in minimum interference with performance of the Work. Relocate and modify facilities as required by progress of the Work.
- B. Provide each facility ready for use when needed to avoid delay. Do not remove until facilities are no longer needed or are replaced by authorized use of completed permanent facilities.

### 3.2 TEMPORARY UTILITY INSTALLATION

- A. General: Install temporary service or connect to existing service.
  - 1. Arrange with utility company, Owner, and existing users for time when service can be interrupted, if necessary, to make connections for temporary services.
- B. Sewers and Drainage: Provide temporary utilities to remove effluent lawfully.
  - 1. Connect temporary sewers to municipal system as directed by authorities having jurisdiction.
- C. Water Service: Install water service and distribution piping in sizes and pressures adequate for construction.
- D. Sanitary Facilities: Provide temporary toilets, wash facilities, and drinking water for use of construction personnel. Comply with authorities having jurisdiction for type, number, location, operation, and maintenance of fixtures and facilities.
- E. Heating and Cooling: Provide temporary heating and cooling required by construction activities for curing or drying of completed installations or for protecting installed construction from adverse effects of low temperatures or high humidity. Select equipment that will not have a harmful effect on completed installations or elements being installed.
- F. Ventilation and Humidity Control: Provide temporary ventilation required by construction activities for curing or drying of completed installations or for protecting installed construction from adverse effects of high humidity. Select equipment that will not have a harmful effect on completed installations or elements being installed. Coordinate ventilation requirements to produce ambient condition required and minimize energy consumption.
- G. Electric Power Service: Provide electric power service and distribution system of sufficient size, capacity, and power characteristics required for construction operations.
  - 1. Install electric power service overhead or underground, unless otherwise indicated.
- H. Lighting: Provide temporary lighting with local switching that provides adequate illumination for construction operations, observations, inspections, and traffic conditions.
  - 1. Install and operate temporary lighting that fulfills security and protection requirements without operating entire system.
- I. Telephone Service: Provide temporary telephone service in common-use facilities for use by all construction personnel. Install one telephone line(s) for each field office.
  - 1. Provide additional telephone lines for the following:
    - a. Provide a dedicated telephone line for each facsimile machine and computer in each field office.
  - 2. At each telephone, post a list of important telephone numbers including police and fire departments Contractor's home office Architect's office Owner's office Principal subcontractors' field and home offices.
  - 3. Provide superintendent with cellular telephone or portable two-way radio for use when away from field office.
- J. Electronic Communication Service: Provide temporary electronic communication service, including electronic mail in field office.

### 3.3 SUPPORT FACILITIES INSTALLATION

- A. General: Comply with the following:
  - 1. Provide incombustible construction for offices, shops, and sheds located within construction area or within 30 feet (9 m) of building lines. Comply with NFPA 241.
  - 2. Maintain support facilities until near Recommendation of Acceptance. Remove before Recommendation of Acceptance. Personnel remaining after Recommendation of Acceptance will be permitted to use permanent facilities, under conditions acceptable to Owner.
- B. Temporary Roads and Paved Areas: Construct and maintain temporary roads and paved areas adequate for construction operations. Locate temporary roads and paved areas within construction limits indicated on Drawings.
  - 1. Provide dust-control treatment that is nonpolluting and nontracking. Reapply treatment as required to minimize dust.
- C. Temporary Roads and Paved Areas: Construct and maintain temporary roads and paved areas adequate for construction operations. Locate temporary roads and paved areas in same location as permanent roads and paved areas. Extend temporary roads and paved areas, within construction limits indicated, as necessary for construction operations.
  - 1. Coordinate elevations of temporary roads and paved areas with permanent roads and paved areas.
  - 2. Prepare subgrade and install subbase and base for temporary roads and paved areas according to Division 31 Section "Earth Moving."
  - 3. Recondition base after temporary use, including removing contaminated material, regrading, proofrolling, compacting, and testing.
  - 4. Delay installation of final course of permanent hot-mix asphalt pavement until immediately before Recommendation of Acceptance. Repair hot-mix asphalt base-course pavement before installation of final course according to Division 32 Section "Asphalt Paving."
- D. Traffic Controls: Comply with requirements of authorities having jurisdiction.
  - 1. Protect existing site improvements to remain including curbs, pavement, and utilities.
  - 2. Maintain access for fire-fighting equipment and access to fire hydrants.
- E. Parking: Provide temporary parking areas for construction personnel.
- F. Dewatering Facilities and Drains: Comply with requirements of authorities having jurisdiction. Maintain Project site, excavations, and construction free of water.
  - 1. Dispose of rainwater in a lawful manner that will not result in flooding Project or adjoining properties nor endanger permanent Work or temporary facilities.
  - 2. Remove snow and ice as required to minimize accumulations.
- G. Project Identification and Temporary Signs: Provide Project identification and other signs as indicated on Drawings. Install signs where indicated to inform public and individuals seeking entrance to Project. Unauthorized signs are not permitted.
  - 1. Provide temporary, directional signs for construction personnel and visitors.
  - 2. Maintain and touchup signs so they are legible at all times.

M.A. No. H1-22029-DA RMG - 22029

- H. Waste Disposal Facilities: Provide waste-collection containers in sizes adequate to handle waste from construction operations. Comply with requirements of authorities having jurisdiction. Comply with Division 01 Section "Execution" for progress cleaning requirements.
- I. Lifts and Hoists: Provide facilities necessary for hoisting materials and personnel.
  - 1. Truck cranes and similar devices used for hoisting materials are considered "tools and equipment" and not temporary facilities.
- J. Temporary Elevator Use: Refer to Division 14 Sections for temporary use of new elevators.
- K. Temporary Stairs: Until permanent stairs are available, provide temporary stairs where ladders are not adequate.
- L. Temporary Use of Permanent Stairs: Cover finished, permanent stairs with protective covering of plywood or similar material so finishes will be undamaged at time of acceptance.

## 3.4 SECURITY AND PROTECTION FACILITIES INSTALLATION

- A. Environmental Protection: Provide protection, operate temporary facilities, and conduct construction in ways and by methods that comply with environmental regulations and that minimize possible air, waterway, and subsoil contamination or pollution or other undesirable effects.
- B. Temporary Erosion and Sedimentation Control: Provide measures to prevent soil erosion and discharge of soil-bearing water runoff and airborne dust to adjacent properties and walkways, according to requirements of authorities having jurisdiction.
- C. Stormwater Control: Comply with authorities having jurisdiction. Provide barriers in and around excavations and subgrade construction to prevent flooding by runoff of stormwater from heavy rains.
- D. Tree and Plant Protection: Install temporary fencing located as indicated or outside the drip line of trees to protect vegetation from damage from construction operations. Use orange construction fence fabric, adequately supported by wood or metal uprights to prevent sag; maintain fencing upright throughout Project. Protect tree root systems from damage, flooding, and erosion.
- E. Pest Control: Engage pest-control service to recommend practices to minimize attraction and harboring of rodents, roaches, and other pests and to perform extermination and control procedures at regular intervals so Project will be free of pests and their residues at Recommendation of Acceptance. Obtain extended warranty for Owner. Perform control operations lawfully, using environmentally safe materials.
- F. Security Enclosure and Lockup: Install substantial temporary enclosure around partially completed areas of construction. Provide lockable entrances to prevent unauthorized entrance, vandalism, theft, and similar violations of security.
- G. Barricades, Warning Signs, and Lights: Comply with requirements of authorities having jurisdiction for erecting structurally adequate barricades, including warning signs and lighting.
- H. Temporary fencing: Metal hogwire supported by 2 x 4 posts or portable 6'-0" chain link panels supported by metal posts affixed to ground to prevent movement or overturning.
- I. Temporary Enclosures: Provide temporary enclosures for protection of construction, in progress and completed, from exposure, foul weather, other construction operations, and similar activities. Provide temporary weathertight enclosure for building exterior.
  - 1. Where heating or cooling is needed and permanent enclosure is not complete, insulate temporary enclosures.

- J. Temporary Fire Protection: Install and maintain temporary fire-protection facilities of types needed to protect against reasonably predictable and controllable fire losses. Comply with NFPA 241.
  - 1. Prohibit smoking in construction areas.
  - 2. Supervise welding operations, combustion-type temporary heating units, and similar sources of fire ignition according to requirements of authorities having jurisdiction.
  - 3. Develop and supervise an overall fire-prevention and -protection program for personnel at Project site. Review needs with local fire department and establish procedures to be followed. Instruct personnel in methods and procedures. Post warnings and information.
  - 4. Provide temporary standpipes and hoses for fire protection. Hang hoses with a warning sign stating that hoses are for fire-protection purposes only and are not to be removed. Match hose size with outlet size and equip with suitable nozzles.

# 3.5 OPERATION, TERMINATION, AND REMOVAL

- A. Supervision: Enforce strict discipline in use of temporary facilities. To minimize waste and abuse, limit availability of temporary facilities to essential and intended uses.
- B. Maintenance: Maintain facilities in good operating condition until removal.
  - 1. Maintain operation of temporary enclosures, heating, cooling, humidity control, ventilation, and similar facilities on a 24-hour basis where required to achieve indicated results and to avoid possibility of damage.
- C. Temporary Facility Changeover: Do not change over from using temporary security and protection facilities to permanent facilities until Recommendation of Acceptance.
- D. Termination and Removal: Remove each temporary facility when need for its service has ended, when it has been replaced by authorized use of a permanent facility, or no later than Recommendation of Acceptance. Complete or, if necessary, restore permanent construction that may have been delayed because of interference with temporary facility. Repair damaged Work, clean exposed surfaces, and replace construction that cannot be satisfactorily repaired.
  - 1. Materials and facilities that constitute temporary facilities are property of Contractor. Owner reserves right to take possession of Project identification signs.
  - 2. At Recommendation of Acceptance, clean and renovate permanent facilities used during construction period. Comply with final cleaning requirements specified in Division 01 Section "Closeout Procedures."

END OF SECTION 01 50 00

# SECTION 01 60 00 - PRODUCT REQUIREMENTS

## PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes administrative and procedural requirements for selection of products for use in Project; product delivery, storage, and handling; manufacturers' standard warranties on products; special warranties; product substitutions; and comparable products.
- B. See Division 01 Section "Closeout Procedures" for submitting warranties for Contract closeout.
- C. See Divisions 02 through 49 Sections for specific requirements for warranties on products and installations specified to be warranted.

### 1.2 DEFINITIONS

- A. Products: Items purchased for incorporating into the Work, whether purchased for Project or taken from previously purchased stock. The term "product" includes the terms "material," "equipment," "system," and terms of similar intent.
  - 1. Named Products: Items identified by manufacturer's product name, including make or model number or other designation shown or listed in manufacturer's published product literature, that is current as of date of the Contract Documents.
  - New Products: Items that have not previously been incorporated into another project or facility. Products salvaged or recycled from other projects are not considered new products.
  - 3. Comparable Product: Product that is demonstrated and approved prior to bid through pre-approval process, or where indicated as a product substitution, to have the indicated qualities related to type, function, dimension, in-service performance, physical properties, appearance, and other characteristics that equal or exceed those of specified product.
- B. Basis-of-Design Product Specification: Where a specific manufacturer's product is named and accompanied by the words "basis of design," including make or model number or other designation, to establish the significant qualities related to type, function, dimension, in-service performance, physical properties, appearance, and other characteristics for purposes of evaluating comparable products of other named manufacturers.

### 1.3 SUBMITTALS

A. Basis-of-Design Product Specification Submittal: Comply with requirements in Division 01 Section "Submittal Procedures." Show compliance with requirements.

## 1.4 QUALITY ASSURANCE

A. Compatibility of Options: If Contractor is given option of selecting between two or more products for use on Project, product selected shall be compatible with products previously selected, even if previously selected products were also options.

B. The Contractor shall be responsible in ensuring that no asbestos containing products or work is included within the scope of the work. The Contractor shall take whatever measures it deems necessary to insure that all employees, suppliers, fabricators, material men, subcontractors, or their assigns, comply with this requirement.

## 1.5 PRODUCT DELIVERY, STORAGE, AND HANDLING

A. Deliver, store, and handle products using means and methods that will prevent damage, deterioration, and loss, including theft. Comply with manufacturer's written instructions.

# B. Delivery and Handling:

- 1. Schedule delivery to minimize long-term storage at Project site and to prevent overcrowding of construction spaces.
- 2. Coordinate delivery with installation time to ensure minimum holding time for items that are flammable, hazardous, easily damaged, or sensitive to deterioration, theft, and other losses.
- 3. Deliver products to Project site in an undamaged condition in manufacturer's original sealed container or other packaging system, complete with labels and instructions for handling, storing, unpacking, protecting, and installing.
- 4. Inspect products on delivery to ensure compliance with the Contract Documents and to ensure that products are undamaged and properly protected.

# C. Storage:

- 1. Store products to allow for inspection and measurement of quantity or counting of units.
- 2. Store materials in a manner that will not endanger Project structure.
- 3. Store products that are subject to damage by the elements, under cover in a weathertight enclosure above ground, with ventilation adequate to prevent condensation.
- 4. Store cementitious products and materials on elevated platforms.
- 5. Store foam plastic from exposure to sunlight, except to extent necessary for period of installation and concealment.
- 6. Comply with product manufacturer's written instructions for temperature, humidity, ventilation, and weather-protection requirements for storage.
- 7. Protect stored products from damage and liquids from freezing.

### 1.6 PRODUCT WARRANTIES

- A. Warranties specified in other Sections shall be in addition to, and run concurrent with, other warranties required by the Contract Documents. Manufacturer's disclaimers and limitations on product warranties do not relieve Contractor of obligations under requirements of the Contract Documents.
  - 1. Manufacturer's Warranty: Preprinted written warranty published by individual manufacturer for a particular product and specifically endorsed by manufacturer to Owner.
  - 2. Special Warranty: Written warranty required by or incorporated into the Contract Documents, either to extend time limit provided by manufacturer's warranty or to provide more rights for Owner.

- B. Special Warranties: Prepare a written document that contains appropriate terms and identification, ready for execution. Submit a draft for approval before final execution.
  - 1. Manufacturer's Standard Form: Modified to include Project-specific information and properly executed.
  - 2. Specified Form: When specified forms are included with the Specifications, prepare a written document using appropriate form properly executed.
  - 3. Refer to Divisions 2 through 16 Sections for specific content requirements and particular requirements for submitting special warranties.
- C. Warranty Period: to commence upon date of Substantial Completion.
- D. Submittal Time: Comply with requirements in Division 01 Section "Closeout Procedures."

### PART 2 - PRODUCTS

### 2.1 PRODUCT SELECTION PROCEDURES

- A. General Product Requirements: Provide products that comply with the Contract Documents, that are undamaged and, unless otherwise indicated, that are new at time of installation.
  - 1. Provide products complete with accessories, trim, finish, fasteners, and other items needed for a complete installation and indicated use and effect.
  - 2. Standard Products: If available, and unless custom products or nonstandard options are specified, provide standard products of types that have been produced and used successfully in similar situations on other projects.
  - 3. Owner reserves the right to limit selection to products with warranties not in conflict with requirements of the Contract Documents.
  - 4. Where products are accompanied by the term "as selected," Architect will make selection.
  - 5. Where products are accompanied by the term "match sample," sample to be matched is to be furnished by Architect.
  - 6. Descriptive, performance, and reference standard requirements in the Specifications establish "salient characteristics" of products.

## B. Product Selection Procedures:

- 1. Products: Where Specifications include a list of names of both products and manufacturers, provide one of the products listed that complies with requirements.
- 2. Manufacturers: Where Specifications include a list of manufacturers' names, provide a product by one of the manufacturers listed that complies with requirements.
- 3. Product Options: Where Specifications indicate that sizes, profiles, and dimensional requirements on Drawings are based on a specific product or system, provide the specified product or system.
- 4. Basis-of-Design Product: Where Specifications name a product and include a list of manufacturers, provide the specified product or a comparable product by one of the other named manufacturers. Drawings and Specifications indicate sizes, profiles, dimensions, and other characteristics that are based on the product named.

Parish Annex Building – Renovation (Phase I) BPPJ

M.A. No. H1-22029-DA RMG - 22029

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 60 00

#### SECTION 01 73 00 - EXECUTION

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes general procedural requirements governing execution of the Work including, but not limited to, the following:
  - 1. Construction layout.
  - 2. Field engineering and surveying.
  - 3. General installation of products.
  - 4. Progress cleaning.
  - 5. Starting-up and adjusting.
  - 6. Protection of installed construction.
  - 7. Correction of the Work.
- B. See Division 01 Section "Closeout Procedures" for submitting final property survey with Project Record Documents, recording of Owner-accepted deviations from indicated lines and levels, and final cleaning.

### 1.2 SUBMITTALS

- A. Certificates: Submit certificate signed by land surveyor certifying that location and elevation of improvements comply with requirements.
- B. Landfill Receipts: Submit copy of receipts issued by a landfill facility, licensed to accept hazardous materials, for hazardous waste disposal.
- C. Certified Surveys: Submit two copies signed by land surveyor.
- D. Final Property Survey: Submit 10 copies showing the Work performed and record survey data.

### 1.3 QUALITY ASSURANCE

A. Land Surveyor Qualifications: A professional land surveyor who is legally qualified to practice in jurisdiction where Project is located and who is experienced in providing land-surveying services of the kind indicated.

EXECUTION 01 73 00 - 1

# PART 2 - PRODUCTS (Not Used)

#### PART 3 - EXECUTION

### 3.1 EXAMINATION

- A. Existing Conditions: The existence and location of site improvements, utilities, and other construction indicated as existing are not guaranteed. Before beginning work, investigate and verify the existence and location of mechanical and electrical systems and other construction affecting the Work.
  - 1. Before construction, verify the location and points of connection of utility services.
- B. Existing Utilities: The existence and location of underground and other utilities and construction indicated as existing are not guaranteed. Before beginning sitework, investigate and verify the existence and location of underground utilities and other construction affecting the Work.
  - 1. Before construction, verify the location and invert elevation at points of connection of sanitary sewer, storm sewer, and water-service piping; and underground electrical services.
  - 2. Furnish location data for work related to Project that must be performed by public utilities serving Project site.
- C. Acceptance of Conditions: Examine substrates, areas, and conditions, with Installer or Applicator present where indicated, for compliance with requirements for installation tolerances and other conditions affecting performance. Record observations.
  - 1. Verify compatibility with and suitability of substrates, including compatibility with existing finishes or primers.
  - 2. Examine roughing-in for mechanical and electrical systems to verify actual locations of connections before equipment and fixture installation.
  - 3. Examine walls, floors, and roofs for suitable conditions where products and systems are to be installed.
  - 4. Proceed with installation only after unsatisfactory conditions have been corrected. Proceeding with the Work indicates acceptance of surfaces and conditions.

#### 3.2 PREPARATION

- A. Existing Utility Information: Furnish information to local utility Owner that is necessary to adjust, move, or relocate existing utility structures, utility poles, lines, services, or other utility appurtenances located in or affected by construction.
  - 1. Coordinate with Electrical Drawings and authorities having jurisdiction.
- B. Field Measurements: Take field measurements as required to fit the Work properly. Recheck measurements before installing each product. Where portions of the Work are indicated to fit to other construction, verify dimensions of other construction by field measurements before fabrication. Coordinate fabrication schedule with construction progress to avoid delaying the Work.

EXECUTION 01 73 00 - 2

- M.A. No. H1-22029-DA RMG - 22029
- C. Space Requirements: Verify space requirements and dimensions of items shown diagrammatically on Drawings.
- D. Review of Contract Documents and Field Conditions: Immediately on discovery of the need for clarification of the Contract Documents, submit a request for information to Architect. Include a detailed description of problem encountered, together with recommendations for changing the Contract Documents. Submit requests on CSI Form 13.2A, "Request for Interpretation."

## 3.3 CONSTRUCTION LAYOUT

- A. Verification: Before proceeding to lay out the Work, verify layout information shown on Drawings, in relation to the property survey and existing benchmarks. If discrepancies are discovered, notify Architect promptly.
- B. General: Engage a land surveyor to lay out the Work using accepted surveying practices.
  - 1. Establish benchmarks and control points to set lines and levels at each story of construction and elsewhere as needed to locate each element of Project.
  - 2. Establish dimensions within tolerances indicated. Do not scale Drawings to obtain required dimensions.
  - 3. Inform installers of lines and levels to which they must comply.
  - 4. Check the location, level and plumb, of every major element as the Work progresses.
  - 5. Notify Architect when deviations from required lines and levels exceed allowable tolerances.
  - 6. Close site surveys with an error of closure equal to or less than the standard established by authorities having jurisdiction.
- C. Site Improvements: Locate and lay out site improvements, including pavements, grading, fill and topsoil placement, utility slopes, and invert elevations.
- D. Building Lines and Levels: Locate and lay out control lines and levels for structures, building foundations, column grids, and floor levels, including those required for mechanical and electrical work. Transfer survey markings and elevations for use with control lines and levels. Level foundations and piers from two or more locations.
- E. Record Log: Maintain a log of layout control work. Record deviations from required lines and levels. Include beginning and ending dates and times of surveys, weather conditions, name and duty of each survey party member, and types of instruments and tapes used. Make the log available for reference by Architect.

# 3.4 FIELD ENGINEERING

- A. Reference Points: Owner will engage licensed surveyor to locate existing permanent benchmarks, control points, and similar reference points before beginning the Work. Contractor will preserve and protect permanent benchmarks and control points during construction operations.
- B. Benchmarks: Establish and maintain a minimum of two permanent benchmarks on Project site, referenced to data established by survey control points. Comply with authorities having jurisdiction for type and size of benchmark.

EXECUTION 01 73 00 - 3

- M.A. No. H1-22029-DA RMG - 22029
- 1. Record benchmark locations, with horizontal and vertical data, on Project Record Documents.
- C. Certified Survey: On completion of foundation walls, major site improvements, and other work requiring field-engineering services, prepare a certified survey showing dimensions, locations, angles, and elevations of construction and sitework.
- D. Elevation Certificate: Provide Elevation Certificate by licensed Engineer or Land Surveyor in accordance with FEMA National Flood Insurance Program.
- E. Final Property Survey: Prepare a final property survey showing significant features (real property) for Project. Include on the survey a certification, signed by Land Surveyor, that principal metes, bounds, lines, and levels of Project are accurately positioned as shown on the survey.
  - 1. Recording: At Recommendation of Acceptance, have the final property survey recorded by or with authorities having jurisdiction as the official "property survey."

## 3.5 INSTALLATION

- A. General: Locate the Work and components of the Work accurately, in correct alignment and elevation, as indicated.
  - 1. Make vertical work plumb and make horizontal work level.
  - 2. Where space is limited, install components to maximize space available for maintenance and ease of removal for replacement.
  - 3. Conceal pipes, ducts, and wiring in finished areas, unless otherwise indicated.
- B. Comply with manufacturer's written instructions and recommendations for installing products in applications indicated.
- C. Install products at the time and under conditions that will ensure the best possible results. Maintain conditions required for product performance until Recommendation of Acceptance.
- D. Conduct construction operations so no part of the Work is subjected to damaging operations or loading in excess of that expected during normal conditions of occupancy.
- E. Tools and Equipment: Do not use tools or equipment that produce harmful noise levels.
- F. Templates: Obtain and distribute to the parties involved templates for work specified to be factory prepared and field installed. Check Shop Drawings of other work to confirm that adequate provisions are made for locating and installing products to comply with indicated requirements.
- G. Anchors and Fasteners: Provide anchors and fasteners as required to anchor each component securely in place, accurately located and aligned with other portions of the Work.
  - 1. Mounting Heights: Where mounting heights are not indicated, mount components at heights directed by Architect.
  - 2. Allow for building movement, including thermal expansion and contraction.
  - 3. Coordinate installation of anchorages. Furnish setting drawings, templates, and directions for installing anchorages, including sleeves, concrete inserts, anchor bolts, and items with integral anchors, that are to be embedded in concrete or masonry. Deliver such items to Project site in time for installation.

EXECUTION 01 73 00 - 4

- M.A. No. H1-22029-DA RMG - 22029
- H. Joints: Make joints of uniform width. Where joint locations in exposed work are not indicated, arrange joints for the best visual effect. Fit exposed connections together to form hairline joints.
- I. Hazardous Materials: Use products, cleaners, and installation materials that are not considered hazardous.

### 3.6 PROGRESS CLEANING

- A. General: Clean Project site and work areas daily, including common areas. Coordinate progress cleaning for joint-use areas where more than one installer has worked. Enforce requirements strictly. Dispose of materials lawfully.
  - 1. Comply with requirements in NFPA 241 for removal of combustible waste materials and debris.
  - 2. Do not hold materials more than 7 days during normal weather or 3 days if the temperature is expected to rise above 80 deg F (27 deg C).
  - 3. Containerize hazardous and unsanitary waste materials separately from other waste. Mark containers appropriately and dispose of legally, according to regulations.
- B. Site: Maintain Project site free of waste materials and debris.
- C. Work Areas: Clean areas where work is in progress to the level of cleanliness necessary for proper execution of the Work.
  - 1. Remove liquid spills promptly.
  - 2. Where dust would impair proper execution of the Work, broom-clean or vacuum the entire work area, as appropriate.
- D. Installed Work: Keep installed work clean. Clean installed surfaces according to written instructions of manufacturer or fabricator of product installed, using only cleaning materials specifically recommended. If specific cleaning materials are not recommended, use cleaning materials that are not hazardous to health or property and that will not damage exposed surfaces.
- E. Concealed Spaces: Remove debris from concealed spaces before enclosing the space.
- F. Exposed Surfaces in Finished Areas: Clean exposed surfaces and protect as necessary to ensure freedom from damage and deterioration at time of Recommendation of Acceptance.
- G. Waste Disposal: Burying or burning waste materials on-site will not be permitted. Washing waste materials down sewers or into waterways will not be permitted.
- H. During handling and installation, clean and protect construction in progress and adjoining materials already in place. Apply protective covering where required to ensure protection from damage or deterioration at Recommendation of Acceptance.
- I. Clean and provide maintenance on completed construction as frequently as necessary through the remainder of the construction period. Adjust and lubricate operable components to ensure operability without damaging effects.
- J. Limiting Exposures: Supervise construction operations to assure that no part of the construction, completed or in progress, is subject to harmful, dangerous, damaging, or otherwise deleterious exposure during the construction period.

EXECUTION 01 73 00 - 5

#### M.A. No. H1-22029-DA RMG - 22029

### 3.7 STARTING-UP AND ADJUSTING

- A. Start-up equipment and operating components to confirm proper operation. Remove malfunctioning units, replace with new units, and retest.
- B. Adjust operating components for proper operation without binding. Adjust equipment for proper operation.
- C. Test each piece of equipment to verify proper operation. Test and adjust controls and safeties. Replace damaged and malfunctioning controls and equipment.
- D. Manufacturer's Field Service: If a factory-authorized service representative is required to inspect field-assembled components and equipment installation, comply with qualification requirements in Division 01 Section "Quality Requirements."

## 3.8 PROTECTION OF INSTALLED CONSTRUCTION

- A. Provide final protection and maintain conditions that ensure installed Work is without damage or deterioration at time of Recommendation of Acceptance.
- B. Comply with manufacturer's written instructions for temperature and relative humidity.

## 3.9 CORRECTION OF THE WORK

- A. Repair or remove and replace defective construction. Restore damaged substrates and finishes. Comply with requirements in Division 01 Section "Cutting and Patching."
  - 1. Repairing includes replacing defective parts, refinishing damaged surfaces, touching up with matching materials, and properly adjusting operating equipment.
- B. Restore permanent facilities used during construction to their specified condition.
- C. Remove and replace damaged surfaces that are exposed to view if surfaces cannot be repaired without visible evidence of repair.
- D. Repair components that do not operate properly. Remove and replace operating components that cannot be repaired.
- E. Remove and replace chipped, scratched, and broken glass or reflective surfaces.

END OF SECTION 01 73 00

EXECUTION 01 73 00 - 6

#### SECTION 01 77 00 - CLOSEOUT PROCEDURES

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes administrative and procedural requirements for contract closeout, including, but not limited to, the following:
  - 1. Inspection procedures.
  - 2. Warranties.
  - 3. Final cleaning.
- B. See Division 01 Section 01 29 00 "Payment Procedures" for requirements for Applications for Payment for Substantial and Final Completion.
- C. See Division 01 Section 01 78 39 "Project Record Documents" for submitting Record Drawings, Record Specifications, and Record Product Data.
- D. See Division 01 Section 01 78 23 "Operation and Maintenance Data" for operation and maintenance manual requirements.
- E. See Division 01 Section 01 79 00 "Demonstration and Training" for requirements for instructing Owner's personnel.
- F. See Divisions 02 through 49 Sections for specific closeout and special cleaning requirements for the Work in those Sections.

### 1.2 RECOMMENDATION OF ACCEPTANCE

- A. Preliminary Procedures: Before requesting inspection for determining date of Recommendation of Acceptance, complete the following. List items below that are incomplete in request.
  - 1. Prepare a list of items to be completed and corrected (punch list), the value of items on the list, and reasons why the Work is not complete.
  - 2. Advise Owner of pending insurance changeover requirements.
  - 3. Submit specific warranties, workmanship bonds, maintenance service agreements, final certifications, and similar documents.
  - 4. Obtain and submit releases permitting Owner unrestricted use of the Work and access to services and utilities. Include occupancy permits, operating certificates, and similar releases.
  - 5. Prepare and submit Project Record Documents, operation and maintenance manuals, Final Completion construction photographs and digital images, damage or settlement surveys, property surveys, and similar final record information.
  - 6. Deliver tools, spare parts, extra materials, and similar items to location designated by Owner. Label with manufacturer's name and model number where applicable.
  - 7. Make final changeover of permanent locks and deliver keys to Owner. Advise Owner's personnel of changeover in security provisions.

- 8. Complete startup testing of systems.
- 9. Submit test/adjust/balance records.
- 10. Terminate and remove temporary facilities from Project site, along with mockups, construction tools, and similar elements.
- 11. Advise Owner of changeover in heat and other utilities.
- 12. Submit changeover information related to Owner's occupancy, use, operation, and maintenance.
- 13. Complete final cleaning requirements, including touchup painting.
- 14. Touch up and otherwise repair and restore marred exposed finishes to eliminate visual defects.
- B. Inspection: Submit a written request for inspection for Recommendation of Acceptance. On receipt of request, Architect will either proceed with inspection or notify Contractor of unfulfilled requirements. Architect will prepare the Recommendation of Acceptance after inspection or will notify Contractor of items, either on Contractor's list or additional items identified by Architect, that must be completed or corrected before certificate will be issued.
  - 1. Reinspection: Request reinspection when the Work identified in previous inspections as incomplete is completed or corrected.
  - 2. Results of completed inspection will form the basis of requirements for Final Completion.

## 1.3 FINAL COMPLETION

- A. Preliminary Procedures: Before requesting final inspection for determining date of Final Completion, complete the following:
  - 1. Submit a final Application for Payment according to Division 01 Section "Payment Procedures."
  - 2. Submit certified copy of Architect's Recommendation of Acceptance inspection list of items to be completed or corrected (punch list), endorsed and dated by Architect. The certified copy of the list shall state that each item has been completed or otherwise resolved for acceptance.
  - 3. Submit evidence of final, continuing insurance coverage complying with insurance requirements.
  - 4. Submit pest-control final inspection report and warranty.
  - 5. Instruct Owner's personnel in operation, adjustment, and maintenance of products, equipment, and systems. Submit demonstration and training videotapes.
- B. Inspection: Submit a written request for final inspection for acceptance. On receipt of request, Architect will either proceed with inspection or notify Contractor of unfulfilled requirements. Architect will prepare a final Certificate for Payment after inspection or will notify Contractor of construction that must be completed or corrected before certificate will be issued.
  - 1. Reinspection: Request reinspection when the Work identified in previous inspections as incomplete is completed or corrected.

# 1.4 LIST OF INCOMPLETE ITEMS (PUNCH LIST)

- A. Preparation: Submit three copies of list. Include name and identification of each space and area affected by construction operations for incomplete items and items needing correction including, if necessary, areas disturbed by Contractor that are outside the limits of construction.
  - 1. Organize list of spaces in sequential order, starting with exterior areas first.
  - 2. Organize items applying to each space by major element, including categories for ceiling, individual walls, floors, equipment, and building systems.
  - 3. Include the following information at the top of each page:
    - a. Project name.
    - b. Date.
    - c. Name of Architect.
    - d. Name of Contractor.
    - e. Page number.
  - 4. Submit list of incomplete items in the following format:
    - a. MS Excel Electronic File: Architect will return annotated file.
    - b. PDF Electronic File: Architect will return annotated file.
    - c. Web-Based Project Software Upload: Utilize software feature for creating and updating list of incomplete items (punch list).

#### 1.5 WARRANTIES

- A. Submittal Time: Submit written warranties on request of Architect for designated portions of the Work where commencement of warranties other than date of Recommendation of Acceptance is indicated.
- B. Organize warranty documents into an orderly sequence based on the table of contents of the Project Manual.
- C. Warranty Electronic File: Provide warranties and bonds in PDF format. Assemble complete warranty and bond submittal package into a single electronic PDF file with bookmarks enabling navigation to each item. Provide bookmarked table of contents at beginning of document.
  - 1. Submit on digital media acceptable to Architect.
- D. Provide additional copies of each warranty to include in operation and maintenance manuals.

### **PART 2 - PRODUCTS**

#### 2.1 MATERIALS

A. Cleaning Agents: Use cleaning materials and agents recommended by manufacturer or fabricator of the surface to be cleaned. Do not use cleaning agents that are potentially hazardous to health or property or that might damage finished surfaces.

## PART 3 - EXECUTION

#### 3.1 FINAL CLEANING

- A. General: Provide final cleaning. Conduct cleaning and waste-removal operations to comply with local laws and ordinances and Federal and local environmental and antipollution regulations.
- B. Cleaning: Employ experienced workers or professional cleaners for final cleaning. Clean each surface or unit to condition expected in an average commercial building cleaning and maintenance program. Comply with manufacturer's written instructions.
  - 1. Complete the following cleaning operations before requesting inspection for Recommendation of Acceptance for entire Project or for a portion of Project:
    - a. Clean Project site, yard, and grounds, in areas disturbed by construction activities, including landscape development areas, of rubbish, waste material, litter, and other foreign substances.
    - b. Sweep paved areas broom clean. Remove petrochemical spills, stains, and other foreign deposits.
    - c. Rake grounds that are neither planted nor paved to a smooth, even-textured surface.
    - d. Remove tools, construction equipment, machinery, and surplus material from Project site.
    - e. Remove snow and ice to provide safe access to building.
    - f. Clean exposed exterior and interior hard-surfaced finishes to a dirt-free condition, free of stains, films, and similar foreign substances. Avoid disturbing natural weathering of exterior surfaces. Restore reflective surfaces to their original condition.
    - g. Remove debris and surface dust from limited access spaces, including roofs, plenums, shafts, trenches, equipment vaults, manholes, attics, and similar spaces.
    - h. Sweep concrete floors broom clean in unoccupied spaces.
    - i. Vacuum carpet and similar soft surfaces, removing debris and excess nap; shampoo if visible soil or stains remain.
    - j. Clean transparent materials, including mirrors and glass in doors and windows. Remove glazing compounds and other noticeable, vision-obscuring materials. Replace chipped or broken glass and other damaged transparent materials. Polish mirrors and glass, taking care not to scratch surfaces.
    - k. Remove labels that are not permanent.
    - 1. Touch up and otherwise repair and restore marred, exposed finishes and surfaces. Replace finishes and surfaces that cannot be satisfactorily repaired or restored or that already show evidence of repair or restoration.
      - 1) Do not paint over "UL" and similar labels, including mechanical and electrical nameplates.

- Wipe surfaces of mechanical and electrical equipment and similar equipment. m. Remove excess lubrication, paint and mortar droppings, and other foreign substances.
- Replace parts subject to unusual operating conditions. n.
- Clean plumbing fixtures to a sanitary condition, free of stains, including stains o. resulting from water exposure.
- Replace disposable air filters and clean permanent air filters. Clean exposed p. surfaces of diffusers, registers, and grills.
- Clean light fixtures, lamps, globes, and reflectors to function with full efficiency. q. Replace burned-out bulbs, and those noticeably dimmed by hours of use, and defective and noisy starters in fluorescent and mercury vapor fixtures to comply with requirements for new fixtures.
- Leave Project clean and ready for occupancy. r.
- C. Pest Control: Engage an experienced, licensed exterminator to make a final inspection and rid Project of rodents, insects, and other pests. Prepare a report.
- D. Comply with safety standards for cleaning. Do not burn waste materials. Do not bury debris or excess materials on Owner's property. Do not discharge volatile, harmful, or dangerous materials into drainage systems. Remove waste materials from Project site and dispose of lawfully.

END OF SECTION 01 77 00

This page intentionally left blank

#### SECTION 01 78 23 - OPERATION AND MAINTENANCE DATA

### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. Section includes administrative and procedural requirements for preparing operation and maintenance manuals, including the following:
  - 1. Operation and maintenance documentation directory manuals.
  - 2. Emergency manuals.
  - 3. Systems and equipment operation manuals.
  - 4. Systems and equipment maintenance manuals.
  - 5. Product maintenance manuals.

### 1.2 CLOSEOUT SUBMITTALS

- A. Submit operation and maintenance manuals indicated. Provide content for each manual as specified in individual Specification Sections, and as reviewed and approved at the time of Section submittals. Submit reviewed manual content formatted and organized as required by this Section.
  - 1. Architect will comment on whether content of operation and maintenance submittals is acceptable.
  - 2. Where applicable, clarify and update reviewed manual content to correspond to revisions and field conditions.
- B. Format: Submit operation and maintenance manuals in the following format:
  - 1. Submit on digital media acceptable to Architect. Enable reviewer comments on draft submittals.
  - 2. Submit three paper copies. Architect will return two copies.
- C. Final Manual Submittal: Submit each manual in final form prior to requesting inspection for Substantial Completion and at least 15 days before commencing demonstration and training. Architect will return copy with comments.
  - 1. Correct or revise each manual to comply with Architect's comments. Submit copies of each corrected manual within 15 days of receipt of Architect's comments and prior to commencing demonstration and training.
- D. Comply with Section 017700 "Closeout Procedures" for schedule for submitting operation and maintenance documentation.

## 1.3 FORMAT OF OPERATION AND MAINTENANCE MANUALS

- A. Manuals, Electronic Files: Submit manuals in the form of a multiple file composite electronic PDF file for each manual type required.
  - 1. Electronic Files: Use electronic files prepared by manufacturer where available. Where scanning of paper documents is required, configure scanned file for minimum readable file size.
  - 2. File Names and Bookmarks: Bookmark individual documents based on file names. Name document files to correspond to system, subsystem, and equipment names used in manual directory and table of contents. Group documents for each system and subsystem into individual composite bookmarked files, then create composite manual, so that resulting bookmarks reflect the system, subsystem, and equipment names in a readily navigated file tree. Configure electronic manual to display bookmark panel on opening file.
- B. Manuals, Paper Copy: Submit manuals in the form of hard-copy, bound and labeled volumes.
  - 1. Binders: Heavy-duty, three-ring, vinyl-covered, loose-leaf binders, in thickness necessary to accommodate contents, sized to hold 8-1/2-by-11-inch (215-by-280-mm) paper; with clear plastic sleeve on spine to hold label describing contents and with pockets inside covers to hold folded oversize sheets.
  - 2. Drawings: Attach reinforced, punched binder tabs on drawings and bind with text.
    - a. If oversize drawings are necessary, fold drawings to same size as text pages and use as foldouts.
    - b. If drawings are too large to be used as foldouts, fold and place drawings in labeled envelopes and bind envelopes in rear of manual. At appropriate locations in manual, insert typewritten pages indicating drawing titles, descriptions of contents, and drawing locations.

# 1.4 REQUIREMENTS FOR EMERGENCY, OPERATION, AND MAINTENANCE MANUALS

- A. Organization of Manuals: Unless otherwise indicated, organize each manual into a separate section for each system and subsystem, and a separate section for each piece of equipment not part of a system. Each manual shall contain the following materials, in the order listed:
  - 1. Title page.
  - 2. Table of contents.
  - 3. Manual contents.
- B. Title Page: Include the following information:
  - 1. Subject matter included in manual.
  - 2. Name and address of Project.
  - 3. Name and address of Owner.
  - 4. Date of submittal.
  - 5. Name and contact information for Contractor.
  - 6. Name and contact information for Construction Manager.
  - 7. Name and contact information for Architect.
  - 8. Name and contact information for Commissioning Authority.

- M.A. No. H1-22029-DA RMG - 22029
- 9. Names and contact information for major consultants to the Architect that designed the systems contained in the manuals.
- 10. Cross-reference to related systems in other operation and maintenance manuals.
- C. Table of Contents: List each product included in manual, identified by product name, indexed to the content of the volume, and cross-referenced to Specification Section number in Project Manual.
- D. Manual Contents: Organize into sets of manageable size. Arrange contents alphabetically by system, subsystem, and equipment. If possible, assemble instructions for subsystems, equipment, and components of one system into a single binder.
- E. Identification: In the documentation directory and in each operation and maintenance manual, identify each system, subsystem, and piece of equipment with same designation used in the Contract Documents. If no designation exists, assign a designation according to ASHRAE Guideline 4, "Preparation of Operating and Maintenance Documentation for Building Systems."

# 1.5 EMERGENCY MANUALS

- A. Emergency Manual: Assemble a complete set of emergency information indicating procedures for use by emergency personnel and by Owner's operating personnel for types of emergencies indicated.
- B. Content: Organize manual into a separate section for each of the following:
  - 1. Type of emergency.
  - 2. Emergency instructions.
  - 3. Emergency procedures.
- C. Type of Emergency: Where applicable for each type of emergency indicated below, include instructions and procedures for each system, subsystem, piece of equipment, and component:
  - 1. Fire.
  - 2. Flood.
  - 3. Gas leak.
  - 4. Water leak.
  - 5. Power failure.
  - 6. Water outage.
  - 7. System, subsystem, or equipment failure.
  - 8. Chemical release or spill.
- D. Emergency Instructions: Describe and explain warnings, trouble indications, error messages, and similar codes and signals. Include responsibilities of Owner's operating personnel for notification of Installer, supplier, and manufacturer to maintain warranties.
- E. Emergency Procedures: Include the following, as applicable:
  - 1. Instructions on stopping.
  - 2. Shutdown instructions for each type of emergency.

- 3. Operating instructions for conditions outside normal operating limits.
- 4. Required sequences for electric or electronic systems.
- 5. Special operating instructions and procedures.

# 1.6 SYSTEMS AND EQUIPMENT OPERATION MANUALS

- A. Systems and Equipment Operation Manual: Assemble a complete set of data indicating operation of each system, subsystem, and piece of equipment not part of a system. Include information required for daily operation and management, operating standards, and routine and special operating procedures.
- B. Content: In addition to requirements in this Section, include operation data required in individual Specification Sections and the following information:
  - 1. System, subsystem, and equipment descriptions. Use designations for systems and equipment indicated on Contract Documents.
  - 2. Performance and design criteria if Contractor has delegated design responsibility.
  - 3. Operating standards.
  - 4. Operating procedures.
  - 5. Operating logs.
  - 6. Wiring diagrams.
  - 7. Control diagrams.
  - 8. Piped system diagrams.
  - 9. Precautions against improper use.
  - 10. License requirements including inspection and renewal dates.

# C. Descriptions: Include the following:

- 1. Product name and model number. Use designations for products indicated on Contract Documents.
- 2. Manufacturer's name.
- 3. Equipment identification with serial number of each component.
- 4. Equipment function.
- 5. Operating characteristics.
- 6. Limiting conditions.
- 7. Performance curves.
- 8. Engineering data and tests.
- 9. Complete nomenclature and number of replacement parts.

## D. Operating Procedures: Include the following, as applicable:

- 1. Startup procedures.
- 2. Equipment or system break-in procedures.
- 3. Routine and normal operating instructions.
- 4. Regulation and control procedures.
- 5. Instructions on stopping.
- 6. Normal shutdown instructions.
- 7. Seasonal and weekend operating instructions.
- 8. Required sequences for electric or electronic systems.
- 9. Special operating instructions and procedures.

M.A. No. H1-22029-DA RMG - 22029

- E. Systems and Equipment Controls: Describe the sequence of operation, and diagram controls as installed.
- F. Piped Systems: Diagram piping as installed, and identify color coding where required for identification.

## 1.7 SYSTEMS AND EQUIPMENT MAINTENANCE MANUALS

- A. Systems and Equipment Maintenance Manuals: Assemble a complete set of data indicating maintenance of each system, subsystem, and piece of equipment not part of a system. Include manufacturers' maintenance documentation, preventive maintenance procedures and frequency, repair procedures, wiring and systems diagrams, lists of spare parts, and warranty information.
- B. Content: For each system, subsystem, and piece of equipment not part of a system, include source information, manufacturers' maintenance documentation, maintenance procedures, maintenance and service schedules, spare parts list and source information, maintenance service contracts, and warranties and bonds, as described below.
- C. Manufacturers' Maintenance Documentation: Include the following information for each component part or piece of equipment:
  - 1. Standard maintenance instructions and bulletins; include only sheets pertinent to product or component installed. Mark each sheet to identify each product or component incorporated into the Work. If data include more than one item in a tabular format, identify each item using appropriate references from the Contract Documents. Identify data applicable to the Work and delete references to information not applicable.
    - a. Prepare supplementary text if manufacturers' standard printed data are not available and where the information is necessary for proper operation and maintenance of equipment or systems.
  - 2. Drawings, diagrams, and instructions required for maintenance, including disassembly and component removal, replacement, and assembly.
  - 3. Identification and nomenclature of parts and components.
  - 4. List of items recommended to be stocked as spare parts.
- D. Maintenance Procedures: Include the following information and items that detail essential maintenance procedures:
  - 1. Test and inspection instructions.
  - 2. Troubleshooting guide.
  - 3. Precautions against improper maintenance.
  - 4. Disassembly; component removal, repair, and replacement; and reassembly instructions.
  - 5. Aligning, adjusting, and checking instructions.
  - 6. Demonstration and training video recording, if available.
- E. Maintenance and Service Schedules: Include service and lubrication requirements, list of required lubricants for equipment, and separate schedules for preventive and routine maintenance and service with standard time allotment.

- F. Spare Parts List and Source Information: Include lists of replacement and repair parts, with parts identified and cross-referenced to manufacturers' maintenance documentation and local sources of maintenance materials and related services.
- G. Warranties and Bonds: Include copies of warranties and bonds and lists of circumstances and conditions that would affect validity of warranties or bonds.
  - 1. Include procedures to follow and required notifications for warranty claims.
- H. Drawings: Prepare drawings supplementing manufacturers' printed data to illustrate the relationship of component parts of equipment and systems and to illustrate control sequence and flow diagrams. Coordinate these drawings with information contained in record Drawings to ensure correct illustration of completed installation.

### 1.8 PRODUCT MAINTENANCE MANUALS

- A. Product Maintenance Manual: Assemble a complete set of maintenance data indicating care and maintenance of each product, material, and finish incorporated into the Work.
- B. Content: Organize manual into a separate section for each product, material, and finish. Include source information, product information, maintenance procedures, repair materials and sources, and warranties and bonds, as described below.
- C. Product Information: Include the following, as applicable:
  - 1. Product name and model number.
  - 2. Manufacturer's name.
  - 3. Color, pattern, and texture.
  - 4. Material and chemical composition.
  - 5. Reordering information for specially manufactured products.
- D. Maintenance Procedures: Include manufacturer's written recommendations and the following:
  - 1. Inspection procedures.
  - 2. Types of cleaning agents to be used and methods of cleaning.
  - 3. List of cleaning agents and methods of cleaning detrimental to product.
  - 4. Schedule for routine cleaning and maintenance.
  - 5. Repair instructions.
- E. Repair Materials and Sources: Include lists of materials and local sources of materials and related services.
- F. Warranties and Bonds: Include copies of warranties and bonds and lists of circumstances and conditions that would affect validity of warranties or bonds.
  - 1. Include procedures to follow and required notifications for warranty claims.

| Parish Annex | Building – R | enovation ( | (Phase | I) |
|--------------|--------------|-------------|--------|----|
| RPPI         |              |             |        |    |

M.A. No. H1-22029-DA RMG - 22029

PART 2 - PRODUCTS (Not Used)

PART 3 - EXECUTION (Not Used)

END OF SECTION 01 78 23

This page intentionally left blank

#### SECTION 01 78 39 - PROJECT RECORD DOCUMENTS

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes administrative and procedural requirements for Project Record Documents, including the following:
  - 1. Record Drawings.
  - 2. Record Specifications.
  - 3. Record Product Data.
- B. See Division 01 Section 01 78 23 "Operation and Maintenance Data" for operation and maintenance manual requirements.
- C. See Divisions 02 through 49 Sections for specific requirements for Project Record Documents of the Work in those Sections.

### 1.2 SUBMITTALS

- A. Record Drawings: Comply with the following:
  - 1. Number of Copies: Submit copies of Record Drawings as follows:
    - a. Initial Submittal: Submit one set(s) of marked-up Record Prints. Architect will mark whether general scope of changes, additional information recorded, and completeness of changes are acceptable. Architect will compile final submittal.
- B. Record Specifications: Submit one copy of Project's Specifications, including addenda and mark any contract modifications made during construction.
- C. Record Product Data: Submit two (2) copies of each Product Data submittal.
- D. Photographic Documentation: submit two (2) sets of CD's bound in close out document binder.

### PART 2 - PRODUCTS

### 2.1 RECORD DRAWINGS

- A. Record Prints: Maintain one set of black-line white prints of the Contract Drawings (24"x36" or 30"x42" depending on the scale of the drawings) and Shop Drawings.
  - 1. Preparation: Mark Record Prints to show the actual installation where installation varies from that shown originally. Require individual or entity who obtained record data, whether individual or entity is Installer, subcontractor, or similar entity, to prepare the marked-up Record Prints.
    - a. Give particular attention to information on concealed elements that would be difficult to identify or measure and record later.
    - b. Record data as soon as possible after obtaining it. Record and check the markup before enclosing concealed installations.

#### SECTION 01 79 00 - DEMONSTRATION AND TRAINING

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes administrative and procedural requirements for instructing Owner's personnel, including the following:
  - 1. Demonstration of operation of systems, subsystems, and equipment.
  - 2. Training in operation and maintenance of systems, subsystems, and equipment.
- B. See Divisions 02 through 49 Sections for specific requirements for demonstration and training for products in those Sections.

#### 1.2 SUBMITTALS

A. Instruction Program: Submit two copies of outline of instructional program for demonstration and training, including a schedule of proposed dates, times, length of instruction time, and instructors' names for each training module. Include learning objective and outline for each training module.

### 1.3 QUALITY ASSURANCE

- A. Facilitator Qualifications: A firm or individual experienced in training or educating maintenance personnel in a training program similar in content and extent to that indicated for this Project, and whose work has resulted in training or education with a record of successful learning performance.
- B. Instructor Qualifications: A factory-authorized service representative, complying with requirements in Division 01 Section "Quality Requirements," experienced in operation and maintenance procedures and training.
- C. Preinstruction Conference: Conduct conference at Project site. Review methods and procedures related to demonstration and training.
- D. Coordinate content of training modules with content of approved emergency, operation, and maintenance manuals. Do not submit instruction program until operation and maintenance data has been reviewed and approved by Architect.

# PART 2 - PRODUCTS

# 2.1 INSTRUCTION PROGRAM

- A. Program Structure: Develop an instruction program that includes individual training modules for each system and equipment not part of a system, as required by individual Specification Sections, and as follows:
  - 1. HVAC Systems
  - 2. Electrical Equipment
  - 3. Access Control System

- 4. Audio & Video Communications Systems
- B. Training Modules: Develop a learning objective and teaching outline for each module. Include a description of specific skills and knowledge that participant is expected to master. For each module, include instruction for the following:
  - 1. Basis of System Design, Operational Requirements, and Criteria: Include system and equipment descriptions, operating standards, regulatory requirements, equipment function, operating characteristics, limiting conditions, and performance curves.
  - 2. Documentation: Review emergency, operations, and maintenance manuals; Project Record Documents; identification systems; warranties and bonds; and maintenance service agreements.
  - 3. Emergencies: Include instructions on stopping; shutdown instructions; operating instructions for conditions outside normal operating limits; instructions on meaning of warnings, trouble indications, and error messages; and required sequences for electric or electronic systems.
  - 4. Operations: Include startup, break-in, control, and safety procedures; stopping and normal shutdown instructions; routine, normal, seasonal, and weekend operating instructions; operating procedures for emergencies and equipment failure; and required sequences for electric or electronic systems.
  - 5. Adjustments: Include alignments and checking, noise, vibration, economy, and efficiency adjustments.
  - 6. Troubleshooting: Include diagnostic instructions and test and inspection procedures.
  - 7. Maintenance: Include inspection procedures, types of cleaning agents, methods of cleaning, procedures for preventive and routine maintenance, and instruction on use of special tools.
  - 8. Repairs: Include diagnosis, repair, and disassembly instructions; instructions for identifying parts; and review of spare parts needed for operation and maintenance.

# PART 3 - EXECUTION

## 3.1 INSTRUCTION

- A. Facilitator: Engage a qualified facilitator to prepare instruction program and training modules, to coordinate instructors, and to coordinate between Contractor and Owner for number of participants, instruction times, and location.
- B. Engage qualified instructors to instruct Owner's personnel to adjust, operate, and maintain systems, subsystems, and equipment not part of a system.
  - 1. Owner will furnish an instructor to describe Owner's operational philosophy.
- C. Scheduling: Provide instruction at mutually agreed on times. For equipment that requires seasonal operation, provide similar instruction at start of each season.
  - 1. Schedule training with Owner, through Architect, with at least seven days' advance notice.
- D. Evaluation: At conclusion of each training module, assess and document each participant's mastery of module by use of an oral performance-based test.

END OF SECTION 01 79 00

This page intentionally left blank

## SECTION 02 41 19 - SELECTIVE DEMOLITION

### PART 1 - GENERAL

### 1.1 SUMMARY

### A. Section Includes:

- 1. Demolition and removal of selected portions of building or structure.
- 2. Demolition and removal of selected site elements.

### 1.2 MATERIALS OWNERSHIP

- A. Unless otherwise indicated, demolition waste becomes property of Contractor.
- B. Historic items, relics, antiques, and similar objects including, but not limited to, cornerstones and their contents, commemorative plaques and tablets, and other items of interest or value to Owner that may be uncovered during demolition remain the property of Owner.
  - 1. Carefully salvage in a manner to prevent damage and promptly return to Owner.

## 1.3 PREINSTALLATION MEETINGS

A. Predemolition Conference: Conduct conference at Project site.

#### 1.4 INFORMATIONAL SUBMITTALS

- A. Engineering Survey: Submit engineering survey of condition of building.
- B. Proposed Protection Measures: Submit report, including Drawings, that indicates the measures proposed for protecting individuals and property, for environmental protection, for dust control and, for noise control. Indicate proposed locations and construction of barriers.
- C. Schedule of selective demolition activities with starting and ending dates for each activity.
- D. Predemolition photographs or video.
- E. Statement of Refrigerant Recovery: Signed by refrigerant recovery technician.

## 1.5 CLOSEOUT SUBMITTALS

A. Inventory of items that have been removed and salvaged.

# 1.6 QUALITY ASSURANCE

A. Refrigerant Recovery Technician Qualifications: Certified by an EPA-approved certification program.

## 1.7 FIELD CONDITIONS

- A. Owner will occupy portions of building immediately adjacent to selective demolition area. Conduct selective demolition so Owner's operations will not be disrupted.
- B. Conditions existing at time of inspection for bidding purpose will be maintained by Owner as far as practical.
- C. Notify Architect of discrepancies between existing conditions and Drawings before proceeding with selective demolition.
- D. Hazardous Materials: It is not expected that hazardous materials will be encountered in the Work.
  - 1. If suspected hazardous materials are encountered, do not disturb; immediately notify Architect and Owner. Hazardous materials will be removed by Owner under a separate contract.
- E. Storage or sale of removed items or materials on-site is not permitted.
- F. Utility Service: Maintain existing utilities indicated to remain in service and protect them against damage during selective demolition operations.
  - 1. Maintain fire-protection facilities in service during selective demolition operations.
- G. Arrange selective demolition schedule so as not to interfere with Owner's operations.

### 1.8 WARRANTY

A. Existing Warranties: Remove, replace, patch, and repair materials and surfaces cut or damaged during selective demolition, by methods and with materials and using approved contractors so as not to void existing warranties.

# PART 2 - PRODUCTS

# 2.1 PERFORMANCE REQUIREMENTS

- A. Regulatory Requirements: Comply with governing EPA notification regulations before beginning selective demolition. Comply with hauling and disposal regulations of authorities having jurisdiction.
- B. Standards: Comply with ANSI/ASSP A10.6 and NFPA 241.

### PART 3 - EXECUTION

# 3.1 EXAMINATION

- A. Verify that utilities have been disconnected and capped before starting selective demolition operations.
- B. Perform an engineering survey of condition of building to determine whether removing any element might result in structural deficiency or unplanned collapse of any portion of structure or adjacent structures during selective building demolition operations.
- C. Inventory and record the condition of items to be removed and salvaged.

## 3.2 PREPARATION

A. Refrigerant: Before starting demolition, remove refrigerant from mechanical equipment according to 40 CFR 82 and regulations of authorities having jurisdiction.

## 3.3 UTILITY SERVICES AND MECHANICAL/ELECTRICAL SYSTEMS

- A. Existing Services/Systems to Remain: Maintain services/systems indicated to remain and protect them against damage.
- B. Existing Services/Systems to Be Removed, Relocated, or Abandoned: Locate, identify, disconnect, and seal or cap off utility services and mechanical/electrical systems serving areas to be selectively demolished.
  - 1. Owner will arrange to shut off indicated services/systems when requested by Contractor.
  - 2. Arrange to shut off utilities with utility companies.
  - 3. If services/systems are required to be removed, relocated, or abandoned, provide temporary services/systems that bypass area of selective demolition and that maintain continuity of services/systems to other parts of building.
  - 4. Disconnect, demolish, and remove fire-suppression systems, plumbing, and HVAC systems, equipment, and components indicated on Drawings to be removed.
    - a. Piping to Be Removed: Remove portion of piping indicated to be removed and cap or plug remaining piping with same or compatible piping material.
    - b. Piping to Be Abandoned in Place: Drain piping and cap or plug piping with same or compatible piping material and leave in place.
    - c. Equipment to Be Removed: Disconnect and cap services and remove equipment.
    - d. Equipment to Be Removed and Reinstalled: Disconnect and cap services and remove, clean, and store equipment; when appropriate, reinstall, reconnect, and make equipment operational.
    - e. Equipment to Be Removed and Salvaged: Disconnect and cap services and remove equipment and deliver to Owner.
    - f. Ducts to Be Removed: Remove portion of ducts indicated to be removed and plug remaining ducts with same or compatible ductwork material.

g. Ducts to Be Abandoned in Place: Cap or plug ducts with same or compatible ductwork material and leave in place.

## 3.4 PROTECTION

- A. Temporary Protection: Provide temporary barricades and other protection required to prevent injury to people and damage to adjacent buildings and facilities to remain.
- B. Temporary Shoring: Design, provide, and maintain shoring, bracing, and structural supports as required to preserve stability and prevent movement, settlement, or collapse of construction and finishes to remain, and to prevent unexpected or uncontrolled movement or collapse of construction being demolished.
- C. Remove temporary barricades and protections where hazards no longer exist.

### 3.5 SELECTIVE DEMOLITION

- A. General: Demolish and remove existing construction only to the extent required by new construction and as indicated. Use methods required to complete the Work within limitations of governing regulations and as follows:
  - 1. Neatly cut openings and holes plumb, square, and true to dimensions required. Use cutting methods least likely to damage construction to remain or adjoining construction. Use hand tools or small power tools designed for sawing or grinding, not hammering and chopping. Temporarily cover openings to remain.
  - 2. Cut or drill from the exposed or finished side into concealed surfaces to avoid marring existing finished surfaces.
  - 3. Do not use cutting torches until work area is cleared of flammable materials. At concealed spaces, such as duct and pipe interiors, verify condition and contents of hidden space before starting flame-cutting operations. Maintain portable fire-suppression devices during flame-cutting operations.
  - 4. Maintain fire watch during and for at least 12 hours after flame-cutting operations.
  - 5. Locate selective demolition equipment and remove debris and materials so as not to impose excessive loads on supporting walls, floors, or framing.
  - 6. Dispose of demolished items and materials promptly.
- B. Site Access and Temporary Controls: Conduct selective demolition and debris-removal operations to ensure minimum interference with roads, streets, walks, walkways, and other adjacent occupied and used facilities.

## C. Removed and Reinstalled Items:

- 1. Clean and repair items to functional condition adequate for intended reuse.
- 2. Pack or crate items after cleaning and repairing. Identify contents of containers.
- 3. Protect items from damage during transport and storage.
- 4. Reinstall items in locations indicated. Comply with installation requirements for new materials and equipment. Provide connections, supports, and miscellaneous materials necessary to make item functional for use indicated.

D. Existing Items to Remain: Protect construction indicated to remain against damage and soiling during selective demolition. When permitted by Architect, items may be removed to a suitable, protected storage location during selective demolition and cleaned and reinstalled in their original locations after selective demolition operations are complete.

### 3.6 CLEANING

- A. Remove demolition waste materials from Project site and dispose of them in an EPA-approved construction and demolition waste landfill acceptable to authorities having jurisdiction.
  - 1. Do not allow demolished materials to accumulate on-site.
  - 2. Remove and transport debris in a manner that will prevent spillage on adjacent surfaces and areas.
  - 3. Remove debris from elevated portions of building by chute, hoist, or other device that will convey debris to grade level in a controlled descent.
- B. Burning: Do not burn demolished materials.
- C. Clean adjacent structures and improvements of dust, dirt, and debris caused by selective demolition operations. Return adjacent areas to condition existing before selective demolition operations began.

END OF SECTION 02 41 19

This page intentionally left blank

### SECTION 03 30 00 - CAST-IN-PLACE CONCRETE

## PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 1 Specification Sections, apply to this Section.

### 1.2 RELATED SECTIONS:

A. Section 03 31 00 Underslab Vapor Retarder.

### 1.3 DESCRIPTION OF WORK:

A. The extent of concrete work is shown on drawings.

## 1.4 QUALITY ASSURANCE:

- A. Codes and Standards: Comply with provisions of following codes, specifications and standards except where more stringent requirements are shown or specified:
- B. ACI 301 "Specifications for Structural Concrete for Buildings".
- C. ACI 318 "Building Code Requirements for Reinforced Concrete".
- D. "Concrete Reinforcing Steel Institute Manual of Standard Practice".

## 1.5 SUBMITTALS:

- A. Product Data: Submit data for proprietary materials and items, including reinforcement and forming accessories, admixtures, patching compounds, waterstops, joint systems, curing compounds, dry-shake finish materials, and others as requested by Engineer.
- B. Shop Drawings: Reinforcement: Submit shop drawings for fabrication bending, and placement of concrete reinforcement. Comply with ACI 315 "Manual of Standard Practice for Detailing Reinforced Concrete Structures showing bar schedules, stirrup spacing, diagrams of bent bars, arrangement of con-crete reinforcement. Include special reinforcement required and openings through concrete structures.
- C. Samples: Submit samples of materials as specified and as otherwise requested by Engineer including names, sources and descriptions.
- D. Laboratory Test Reports: Submit laboratory test reports for concrete materials and mix design test as specified.
- E. Material Certificates: Provide materials certificates in lieu of materials laboratory test reports when permitted by Engineer. Material certificates shall be signed by manufacturer and Contractor, certifying that each material item complies with, or exceeds, specified requirements.

#### **PART 2 - PRODUCTS**

## 2.1 FORM MATERIALS:

- A. Forms for Exposed Finish Concrete: Unless otherwise indicated, construct formwork for exposed concrete surfaces with plywood, metal, metal-framed plywood faced or other acceptable panel-type materials, to provide continuous, straight, smooth, exposed surfaces. Furnish in largest practicable sizes to minimize number of joints and to conform to joint system shown on drawings. Provide form material with sufficient thickness to withstand pressure of newly-placed concrete without bow or deflection.
- B. Forms for Unexposed Finish Concrete: Form concrete surfaces which will be unexposed in finished structure with plywood lumber metal or other acceptable material. Provide lumber dressed on at least 2 edges and one side for tight fit.
- C. Form Coatings: Provide commercial formulation form-coating compounds that will not bond with, stain nor adversely affect concrete surfaces, and will not impair subsequent treatments of concrete surfaces.

### 2.2 REINFORCING MATERIALS:

- A. Reinforcing Bars: ASTM A 615, Grade 60, deformed.
- B. Steel Wire: ASTM A 82 plain, cold-drawn, steel.
- C. Welded Wire Fabric: ASTM A 185, welded steel wire fabric, smooth.
- D. Welded Wire Fabric: ASTM A 497, welded steel wire fabric, deformed.
- E. Supports for Reinforcement: Provide supports for reinforcement including bolsters, chairs, spacers and other devices for spacing, supporting and fastening reinforcing bars and welded wire fabric in place. Use wire bar type supports complying with CRSI specifications, unless otherwise acceptable.
- F. For slabs-on-grade, use supports with sand plates or horizontal runners where base material will not support chair legs.
- G. For exposed-to-view concrete surfaces, where legs of supports are in contact with forms, provide supports with legs which are plastic protected (CRSI Class 1) or stainless steel protected (CRSI Class 2).

## 2.3 CONCRETE MATERIALS:

- A. Portland Cement: ASTM C 150, Type 1, unless otherwise acceptable to Engineer.
- B. Use one brand of cement throughout project, unless otherwise acceptable to Engineer.
- C. Fly Ash: ASTM C 618, Type C. Limit Fly Ash to 10% of total Portland cement volume (cement + fly ash).
- D. Normal Weight Aggregates: ASTM C 33, and as herein specified. Provide aggregates from a single source.
- E. Do not use fine or coarse aggregates containing spalling-causing deleterious substances.
- F. Local aggregates not complying with ASTM C 33 but which have shown by special test or actual service to produce concrete of adequate strength and durability may be used when acceptable to Engineer

## CAST-IN-PLACE CONCRETE

- G. Lightweight Aggregates: ASTM C 330.
- H. Water: Potable.
- I. Air-Entraining Admixture: ASTM C 260.
- J. Water-Reducing Admixture: ASTM C 494, Type A, and contain not more than 0.1% chloride ions.
  - 1. Available Products: Subject to compliance with requirements, products which may be incorporated in the work include, but are not limited to, the following
    - a. "Eucon WR-75"; Euclid Chemical Co.
    - b. "Pozzolith 344"; Master Builders.
    - c. "Plastocrete 160"; Sika Chemical Corp.
    - d. "Chemtard"; Chem-Masters Corp.
- K. High-Range Water-Reducing Admixture (Super Plasticizer): ASTM C 494, Type F or Type G and contain not more than 0.1% chloride ions.
  - 1. Available Products: Subject to compliance with requirements, products which may be incorporated in the work include, but are not limited to, the following
    - a. "WRDA 19"; W.R. Grace
    - b. "PSP"; Protex Industries Inc.
    - c. "Super P"; Anti-Hydro.
    - d. "Sikament; Sika Chemical Corp.
    - e. "Mighty 150"; ICI Americas Corp.
    - f. "Eucon 37"; Euclid Chemical Co.
    - g. "PSI Super"; Gifford-Hill.
    - h. "Pozzolith 400"; Master Builders.
- L. Water-Reducing Non-Chloride Accelerator Admixture: ASTM C 494, Type E, and containing not more than 0.1% chloride ions.
  - 1. Available Products: Subject to compliance with requirements, products which may be incorporated in the work include, but are not limited to, the following:
    - a. "Accelguard 80"; Euclid Chemical Co.
    - b. "Pozzolith 500"; Master Builders.
    - c. Water-Reducing Retarding Admixture: ASTM C 494, Type D, and contain not more than 0.1% chloride ions.
- M. Available Products: Subject to compliance with requirements, products which may be incorporated in the work include, but are not limited to, the following:
  - 1. "Edoco 20006"; Edoco Technical Products.
  - 2. "Pozzolith 300-R"; Master Builders.
  - 3. "Eucon Retarder 75"; Euclid Chemical Co.
  - 4. "Daratard"; W.R. Grace.
  - 5. "Plastiment"; Sika Chemical Co.
- N. Calcium chloride or admixtures containing more than 0.1% chloride ions are not permitted. Admixtures containing any chloride ions shall not be used for concrete supported by metal deck.

### 2.4 RELATED MATERIALS:

- A. Absorptive Cover: Burlap cloth made from jute or kenaf, weighing approximately 9 oz. per sq. yd., complying with AASHTO M 182, Class 2.
- B. Moisture-Retaining Cover: One of the following, complying with ASTM C 171.
  - 1. Waterproof paper.
  - 2. Polyethylene film.
  - 3. Polyethylene-coated burlap.
- C. Liquid Membrane Forming Curing Compound: Liquid type membrane-forming curing compound complying with ASTM C 309, Type 1, Class A unless other type acceptable to Architect. Moisture loss not more than 0.055 gr./sq. cm. when applied to 200 sq. ft./gal.
  - 1. Available Products: Subject to compliance with requirements, products which may be incorporated in the work include, the following:
    - a. "Masterseal"; Master Builders.
    - b. "A-H 3 Way" Sealer"; Anti-Hydro Waterproofing Co.
    - c. "Ecocure"; Euclid Chemical Co.
    - d. "Clear Seal"; W.R. Grace.
    - e. "J-20 Acrylic Cure"; Dayton Superior.
    - f. "Sure Cure"; Kaufman Products Inc.
    - g. "Spartan-Cote"; The Burke Co.
    - h. "Sealkure"; Toch Div. Carboline.
    - i. "Kure-N-Seal"; Sonneborn-Contech.
    - j. "Polyclear"; Upco Chemical/USM Corp.
    - k. "L&M Cure"; L & M Construction Chemicals.
    - 1. "Klearseal"; Setcon Industries.
    - m. "LR-151"; Protex Industries.
    - n. "Hardtop"; Gifford-Hill.
- D. Bonding Compound: Polyvinyl acetate or acrylic base, rewettable type.
- E. Epoxy Adhesive: ASTM C 881, two component material suitable for use on dry or damp surfaces. Provide material "Type", "Grade", and "Class" to suit project requirements.
  - 1. Available Products: Subject to compliance with requirements, products which may be incorporated in the work include, but are not limited to, the following:
    - a. "Epoxtite"; A. C. Horn.
    - b. "Edoco 2118 Epoxy Adhesive"; Edoco Technical Prod.
    - c. "Sikadur Hi-Mod"; Sika Chemical Corp.
    - d. "Euco Epoxy 463 or 615"; Euclid Chemical Co.
    - e. "Patch and Bond Epoxy"; The Burke Co.
    - f. "Sure-Poxy"; Kaufman Products Inc.

### 2.5 PROPORTIONING AND DESIGN OF MIXES:

A. Prepare design mixes for each type and strength of concrete by either laboratory trial batch or field experience methods as specified in ACI 301. If trial batch method used, use an independent testing facility acceptable to Engineer for preparing and reporting proposed mix designs. The testing facility shall not be the same as used for field quality control testing unless otherwise acceptable to Engineer.

- B. Submit written reports to Engineer of each proposed mix for each class of concrete at least 15 days prior to start of work. Do not begin concrete production until mixes have been reviewed by Engineer.
- C. Design mixes to provide normal weight concrete with the following properties, as indicated on drawings and schedules:
  - 1. 3000 psi 28-day compressive strength; W/C ratio, 0.58 maximum (non-air-entrained), 0.46 maximum (air-entrained).
- D. Adjustment to Concrete Mixes: Mix design adjustments may be requested by Contractor when characteristics of materials, job conditions, weather, test results, or other circumstances warrant; at no additional cost to Owner and as accepted by Engineer. Laboratory test data for revised mix design and strength results must be submitted to and accepted by Engineer before using in work.

## E. Admixtures:

- 1. Use water-reducing admixture or high range water-reducing admixture (super plasticizer) in concrete as required for placement and workability.
- 2. Use air-entraining admixture in exterior exposed concrete unless otherwise indicated. Add air-entraining admixture at manufacturer's prescribed rate to result in concrete at point of placement having total air content with a tolerance of plus-or- minus 1-1/2% within following limits:
  - a. Concrete structures and slabs exposed to freezing and thawing, deicer chemicals, or subjected to hydraulic pressure:
    - 1) 4.5% (moderate exposure); 5.5% (severe exposure) 1-1/2"max. aggregate.
    - 2) 4.5% (Moderate exposure); 6.0% (severe exposure) 1" max. aggregate.
    - 3) 5.0% (moderate exposure); 6.0% (severe exposure) 3/4" max. aggregate.
    - 4) 5.5% (moderate exposure); 7.0% (severe exposure) 1/2" max. aggregate.
  - b. Other Concrete: 2% to 4% air.
- 3. Use admixtures for water-reducing and set-control in strict compliance with manufacturer's directions.
- F. Water-Cement Ratio: Provide concrete for following conditions with maximum water-cement (WC) ratios as follows:
  - 1. Subjected to freezing and thawing at time of placement; WC 0.50 maximum.
  - 2. General concrete, non-air-entrained; 0.58 maximum.
  - 3. General concrete, air-entrained; 0.46 maximum.
- G. Slump Limits: Proportion and design mixes to result in concrete slump at point of placement as follows:
  - 1. Ramps, slabs, and sloping surfaces: Not more than 3".
  - 2. Reinforced foundation systems: Not more than 4".
  - 3. Concrete containing HRWR admixture (super plasticizer): Not more than 8" after addition of HRWR to verified 2R-3-slump concrete.
  - 4. Other concrete: Not more than 4".

## 2.6 CONCRETE MIXES:

- A. Job-Site Mixing: Mix materials for concrete in appropriate drum type batch machine mixer. For mixers of one cu. yd., or smaller capacity, continue mixing at least 1-1/2 minutes, but not more than 5 minutes after ingredients are in mixer, before any part of batch is released. For mixers of capacity larger than one cu. yd., increase minimum 1-1/2 minutes of mixing time by 15 seconds for each additional cu. yd., or fraction thereof.
- B. Provide batch ticket for each batch discharged and used in work, indicating project identification name and number, date, mix type, mix time, quantity and amount of water introduced.
- C. Ready-Mix Concrete: Comply with requirements of ASTM C 94, and as herein specified.
- D. During hot weather, or under conditions contributing to rapid setting of concrete, a shorter mixing time than specified in ASTM C 94 may be required.
- E. When air temperature is between 85EF (30EC) and 90EF (32EC), reduce mixing and delivery time from 1-1/2 hours to 75 minutes, and when air temperature is above 90EF (32 C), reduce mixing and delivery time to 60 minutes.

## PART 3 - EXECUTION

#### 3.1 FORMS:

- A. Design, erect support brace and maintain formwork to support vertical and lateral loads that might be applied until such loads can be supported by concrete structure. Construct formwork so concrete members and structures are of correct size, shape, alignment, elevation and position.
  - 1. Design formwork to be readily removable without impact, shock or damage to cast-inplace concrete surfaces and adjacent materials.
- B. Construct forms to sizes, shapes, lines and dimensions shown, and to obtain accurate alignment, location, grades, level and plumb work in finished structures. Provide for openings, offsets, sinkages, keyways, recesses, moldings, rustications, reglets, chamfers, blocking, screeds, bulkheads, anchorages and inserts, and other features required in work. Use selected materials to obtain required finishes. Solidly butt joints and provide backup at joints to prevent leakage of cement paste.
- C. Fabricate forms for easy removal without hammering or prying against concrete surfaces. Provide crush plates or wrecking plates where stripping may damage cast concrete surfaces. Provide top forms for inclined surfaces where slope is too steep to place concrete with bottom forms only. Kerf wood inserts for forming keyways, reglets, recesses, and the like, to prevent swelling and for easy removal.
- D. Provide temporary openings where interior area of formwork is inaccessible for cleanout, for inspection before concrete placement, and for placement of concrete. Securely brace temporary openings and set tightly to forms to prevent loss of concrete mortar. Locate temporary openings on forms at inconspicuous locations.
- E. Chamfer exposed corners and edges as indicated using wood, metal, PVC or rubber chamfer strips fabricated to produce uniform smooth lines and tight edge joints.

- F. Form Ties (if applicable): Factory-fabricated, adjustable-length, removable or snapoff metal form ties designed to prevent form deflection and to prevent spalling concrete surfaces upon removal.
  - 1. Unless otherwise indicated, provide ties so portion remaining within concrete after removal is 1" inside concrete and will not leave holes larger than 1" diameter in concrete surface.
- G. Provisions for Other Trades: Provide openings in concrete formwork to accommodate work of other trades. Determine size and location of openings, recesses and chases from trades providing such items. Accurately place and securely support items built into forms.
- H. Cleaning and Tightening Thoroughly clean forms and adjacent surfaces to receive concrete. Remove chips, wood, sawdust, dirt or other debris just before concrete is placed. Retighten forms and bracing after concrete placement is required to eliminate mortar leaks and maintain proper alignment.

### 3.2 PLACING REINFORCEMENT:

- A. Comply with Concrete Reinforcing Steel Institute's recommended practice for "Placing Reinforcing Bars", for details and methods of reinforcement placement and supports, and as herein specified.
- B. Clean reinforcement of loose rust and mill scale, earth, ice, and other materials which reduce or destroy bond with concrete.
- C. Accurately position, support and secure reinforcement against displacement by formwork, construction, or concrete placement operations. Locate and support reinforcing by metal chairs, runners, bolsters, spacers, and hangers, as required.
- D. Place reinforcement to obtain at least minimum coverages for concrete protection. Arrange, space and securely tie bars and bar supports to hold reinforcement in position during concrete placement operations. Set wire ties so ends are directed into concrete, not toward exposed concrete surfaces.
- E. Install welded wire fabric in as long lengths as practicable. Lap adjoining pieces at least two full meshes and lace splices with wire. Offset end laps in adjacent widths to prevent continuous laps in either direction. At areas other than slab on grade, provide mesh in flat sheets.

### 3.3 JOINTS:

- A. Control joints in Slabs-on-Ground: Place control joints in concrete slabs-on-grade where shown or indicated on drawings. Construction joints shall tounge and groove made by using preformed galvanized steel metal edge forms with a minimum depth of at least the depth of concrete slab at joint location. Floor slabs-on-grade shall be poured in a checkerboard pattern in areas not exceeding 500 square feet. Contractor is to provide plan of pour locations and pattern to engineer for approval prior to commencing concrete placement. In event that control joints are not shown on drawings, contractor shall provide his proposed layout of control joints to engineer for approval. The maximum spacing of joints will not exceed 25 feet in any direction.
- B. Form contraction joints by inserting premolded plastic, hardboard or fiberboard strip into fresh concrete until top surface of strip is flush with slab surface. Tool slab edges round on each side of insert. After concrete has cured, remove inserts and clean groove of loose debris.

- C. Contraction joints may be formed by saw cuts as soon as possible after slab finishing as may be safely done without dislodging aggregate, by means of ultra-early entry saw cutting.
  - 1. Contraction Joints: Early-entry sawcut initial-set green concrete into weakened-plane areas, sectioning concrete into areas not to exceed ACI recommendations.
  - 2. Sawcut joints using dry-cut saw within 4-to-5 hours after steel trowel operations and when sawing will not ravel edges, nor dislodge aggregate.
  - 3. Sawcut contraction joints for a minimum depth of at least 1".
- D. Joint sealant material is specified in Division-7 sections of these specifications.

#### 3.4 INSTALLATION OF EMBEDDED ITEMS:

- A. General: Set and build into work anchorage devices and other embedded items required for other work that is attached to, or supported by, cast-in-place concrete. Use setting drawings, diagrams, instructions and directions provided by suppliers of items to be attached thereto.
- B. Edge Forms and Screed Strips for Slabs: Set edge forms or bulkheads and intermediate screed strips for slabs to obtain required elevations and contours in finished slab surface. Provide and secure units sufficiently strong to support types of screed strips by use of strike-off templates or accepted compacting type screeds.

### 3.5 PREPARATION OF FORM SURFACES:

- A. Clean re-used forms of concrete matrix residue, repair and patch as required to return forms to acceptable surface condition.
- B. Coat contact surfaces of forms with a form-coating compound before reinforcement is placed.
- C. Thin form-coating compounds only with thinning agent of type, and in amount, and under conditions of form-coating compound manufacturer's directions. Do not allow excess formcoating material to accumulate in forms or to come into contact with in-place concrete surfaces against which fresh concrete will be placed. Apply in compliance with manufacturer's instructions.
- D. Coat steel forms with a non-staining rust-preventative form oil or otherwise protect against rusting. Rust -stained steel formwork is not acceptable.

# 3.6 CONCRETE PLACEMENT:

- A. Pre-placement Inspection: Before placing concrete, inspect and complete formwork installation, reinforcing steel, and items to be embedded or cast-in. Notify other crafts to permit installation of their work; cooperate with other trades in setting such work. Moisten wood forms immediately before placing concrete where form coatings are not used.
- B. Coordinate a pre-installation conference with General Contractor, Owner's Representative, and Stained Concrete Floor Subcontractor to review surface finishing and chemical admixtures requirements.
- C. Coordinate the installation of joint materials and moisture barriers with placement of forms and reinforcing steel.
  - 1. General: Comply with ACI 304 "Recommended Practice for Measuring, Mixing, Transporting, and Placing Concrete", and as herein specified.

### CAST-IN-PLACE CONCRETE

- 2. Deposit concrete continuously or in layers of such thickness that no concrete will be placed on concrete which has hardened sufficiently to cause the formation of seams or planes of weakness. If a section cannot be placed continuously, provide construction joints as herein specified. Deposit concrete as nearly as practicable to its final location to avoid segregation.
- D. Placing Concrete in Forms: Deposit concrete in forms in horizontal layers not deeper than 24" and in a manner to avoid inclined construction joints. Where placement consists of several layers, place each layer while preceding layer is still plastic to avoid cold joints.
- E. Consolidate placed concrete by mechanical vibrating equipment supplemented by hand-spading, rodding or tamping. Use equipment and procedures for consolidation of concrete in accordance with ACI recommended practices.
- F. Do not use vibrators to transport concrete inside forms. Insert and withdraw vibrators vertically at uniformly spaced locations not farther than visible effectiveness of machine. Place vibrators to rapidly penetrate placed layer and at least 6" into preceding layer. Do not insert vibrators into lower layers of concrete that have begun to set. At each insertion limit duration of vibration to time necessary to consolidate concrete and complete embedment of reinforcement and other embedded items without causing segregation of mix.
- G. Placing Concrete Slabs: Deposit and consolidate concrete slabs in a continuous operation, within limits of construction joints, until the placing of a panel or section is completed.
- H. Consolidate concrete during placing operations so that concrete is thoroughly worked around reinforcement and other embedded items and into corners.
- I. Bring slab surfaces to correct level with straightedge and strikeoff. Use bull floats or darbies to smooth surface, free of humps or hollows. Do not disturb slab surfaces prior to beginning finishing operations.
- J. Maintain reinforcing in proper position during concrete placement operations.
- K. Cold Weather Placing: Protect concrete work from physical damage or reduced strength which could be caused by frost, freezing actions, or low temperatures, in compliance with ACI 306 and as herein specified.
  - 1. When air temperature has fallen to or is expected to fall below 40 F (4 C) uniformly heat water and aggregates before mixing to obtain a concrete mixture temperature of not less than 50 F (10 C) and not more than 80 F (27 C) at point of placement.
- L. Do not use frozen materials or materials containing ice or snow. Do not place concrete on frozen subgrade or on subgrade containing frozen materials.
- M. Do not use calcium chloride, salt and other materials containing antifreeze agents or chemical accelerators, unless otherwise accepted in mix designs.
- N. Hot Weather Placing: When hot weather conditions exist that would seriously impair quality and strength of concrete, place concrete in compliance with ACI 305 and as herein specified.
- O. Cool ingredients before mixing to maintain concrete temperature at time of placement below 90 F (32 C). Mixing water may be chilled, or chopped ice may be used to control temperature provided water equivalent of ice is calculated to total amount of mixing water. Use of liquid nitrogen to cool concrete is Contractor's option.
- P. Cool steel with water-soaked burlap if it becomes too hot, so that steel temperature will not exceed the ambient air temperature immediately before embedment in concrete.

- Q. Fog spray forms, reinforcing steel and subgrade just before concrete is placed.
- R. Use water-reducing retarding admixture (Type D) when required by high temperatures, low humidity, or other adverse placing conditions.

### 3.7 FINISH OF FORMED SURFACES:

- A. Rough Form Finish: For formed concrete surfaces not exposed-to view in the finish work or by other construction, unless otherwise indicated. This is the concrete surface having texture imparted by form facing material used with the holes and defective areas repaired and patched and fins and other projections exceeding 1/4" in height rubbed down or chipped off.
- B. Smooth Form Finish: For formed concrete surfaces exposed-to-view, or that are to be covered with a coating material applied directly to concrete, or a covering material applied directly to concrete, such as waterproofing, dampproofing, painting, or other similar system. This is as-cast concrete surface obtained with selected form facing material, arranged orderly and symmetrically with a minimum of seams. Repair and patch defective areas with fins or other projections completely removed and smoothed.

### 3.8 MONOLITHIC SLAB FINISHES:

- A. Float Finish: Apply float finish to monolithic slab surfaces to receive trowel finish and other finishes as hereinafter specified, and slab surfaces which are to be covered with membrane or elastic waterproofing, membrane or elastic roofing, or sand-bed terrazzo, and as otherwise indicated.
  - 1. After screeding, consolidating, and leveling concrete slabs, do not work surface until ready for floating. Begin floating when surface water has disappeared or when concrete has stiffened sufficiently to permit operation of power-driven floats, or both. Consolidate surface with power-driven floats or by handfloating if area is small or inaccessible to power units. Check and level surface plane so that depressions between high spots do not exceed 5/16" under a 10' straightedge. Cut down high spots and fill low spots. Uniformly slope surfaces to drains. Immediately after leveling, refloat surface to a uniform, smooth, granular texture.
- B. Trowel Finish: Apply trowel finish to monolithic slab surfaces to be exposed-to-view, and slab surfaces to be covered with resilient flooring, carpet, ceramic or quarry tile, paint or other thinfilm finish coating system.
  - 1. After floating, begin first trowel finish operation using power-driven trowel. Begin final troweling when surface produces a ringing sound as trowel is moved over surface. Consolidate concrete surface by final hand-troweling operation, free of trowel marks, uniform in texture and appearance, and with a level surface plane so that depressions between high spots do not exceed 1/4" under a 10' straightedge. Grind smooth surface defects which would telegraph through applied floor covering system.

# 3.9 CONCRETE CURING AND PROTECTION:

- A. General: Protect freshly placed concrete from premature drying and excessive cold or hot temperatures.
  - 1. Start initial curing as soon as free water has disappeared from concrete surface after placing and finishing. Weather permitting, keep continuously moist for not less than 7 days.

## CAST-IN-PLACE CONCRETE

- 2. Begin final curing procedures immediately following initial curing and before concrete has dried. Continue final curing for at least 7 days in accordance with ACI 301 procedures. Avoid rapid drying at end of final curing period.
- B. Curing Methods: Perform curing of concrete by curing and sealing compound, by moist curing, by moisture-retaining cover curing, and by combinations thereof, as herein specified.
- C. Provide moisture curing by following methods.
  - 1. Covering concrete surface with specified absorptive cover, thoroughly saturating cover with water and keeping continuously wet. Place absorptive cover to provide coverage of concrete surfaces and edges with 4" lap over adjacent absorptive covers.
- D. Provide moisture-cover curing as follows:
  - 1. Cover concrete surfaces with moisture-retaining cover for curing concrete, placed in widest practicable width with sides and ends lapped at least 3" and sealed by waterproof tape or adhesive. Immediately repair any holes or tears during curing period using cover material and waterproof tape.
- E. Provide curing and sealing compound to interior slabs with resilient flooring, carpet over cushion, or left exposed; and to exterior slabs, walks, and curbs as follows:
  - 1. Apply specified curing and sealing compound to concrete slabs as soon as final finishing operations are complete (within 2 hours). Apply uniformly in continuous operation by powerspray or roller in accordance with manufacturer's directions. Recoat areas subjected to heavy rainfall within 3 hours after initial application. Maintain continuity of coating and repair damage during curing period.
  - Do not use membrane curing compounds on surfaces which are to be covered with coating material applied directly to concrete, liquid floor hardener, waterproofing, dampproofing, membrane roofing, flooring (such as ceramic or quarry tile, glue-down carpet) painting, and other coatings and finish materials, unless otherwise acceptable to Architect.
  - 3. <u>DO NOT</u> apply curing compound to areas noted to receive "Stained Concrete"; Review finish schedule to verify areas.
- F. Curing Formed Surfaces (if applicable): Cure formed concrete surfaces, including undersides of beams, supported slabs and other similar surfaces by moist curing with forms in place for full curing period or until forms are removed. If forms are removed, continue curing by methods specified above, as applicable.
- G. Curing Unformed Surfaces: Cure unformed surfaces, such as slabs, floor topping, and other flat surfaces by application of appropriate curing method.
  - 1. Final cure concrete surfaces to receive liquid floor hardener or finish flooring by use of moisture-retaining cover unless otherwise directed.
- H. Sealer and Dustproofer: Apply a second coat of specified curing and sealing compound only to surfaces given a first coat.

## 3.10 REMOVAL OF FORMS:

A. Formwork not supporting weight of concrete such as sides of beams, walls, columns, and similar parts of the work, may be removed after cumulatively curing at not less than 50 F (10 C) for 24 hours after placing concrete, provided concrete is sufficiently hard to not be damaged by form removal operations, and provided curing and protection operations are maintained.

CAST-IN-PLACE CONCRETE

03 30 00 - 11

#### 3.12 RE-USE OF FORMS:

- A. Clean and repair surfaces of forms to be re-used in work. Split, frayed, delaminated or otherwise damaged form facing material will not be acceptable for exposed surfaces. Apply new form coating compound as specified for new formwork.
- B. When forms are extended for successive concrete placement, thoroughly clean surfaces, remove fins and laitance, and tighten forms to close joints. Align and secure joint to avoid offsets. Do not use "patched" forms for exposed concrete surfaces, except as acceptable to Architect.

## 3.13 MISCELLANEOUS CONCRETE ITEMS:

- A. Filling-In: Fill-in holes and openings left in concrete structures for passage of work by other trades unless otherwise shown or directed, after work of other trades is in place. Mix place and cure concrete as herein specified, to blend with in-place construction. Provide other miscellaneous concrete filling shown or required to complete work.
- B. Equipment Bases and Foundations: Provide machine and equipment bases and foundations, as shown on drawings. Set anchor bolts for machines and equipment to template at correct elevations, complying with certified diagrams or templates of manufacturer furnishing machines and equipment.
- C. Grout base plates and foundations as indicated, using specified non-shrink grout. Use non-metallic grout.

#### 3.14 CONCRETE SURFACE REPAIRS:

- A. Patching Defective Areas: Repair and patch defective areas with cement mortar immediately after removal of forms, when acceptable to Architect.
  - 1. Cut out honeycomb, rock pockets, voids over 1/4" in any dimension, and holes left by tie rods and bolts, down to solid concrete but in no case to a depth of less than 1". Make edges of cuts perpendicular to the concrete surface. Thoroughly clean, dampen with water and brush-coat the area to be patched with specified bonding agent. Place patching mortar after bonding compound has dried.
- B. For exposed-to-view surfaces, blend white portland cement and standard portland cement so that, when dry, patching mortar will match color surrounding. Provide test areas at inconspicuous location to verify mixture and color match before proceeding with patching. Compact mortar in place and strike-off slightly higher than surrounding surface.
- C. Repair of Formed Surfaces: Remove and replace concrete having defective surfaces if defects cannot be repaired to satisfaction of Engineer. Surface defects, as such, include color and texture irregularities, cracks, spalls, air bubbles, honeycomb, rock pockets; fins and other projections on surface; and stains and other discolorations that cannot be removed by cleaning. Flush out form tie holes, fill with dry pack mortar, or precast cement cone plugs secured in place with bonding agent.
- D. Repair concealed formed surfaces. where possible, that contain defects that affect the durability of concrete. If defects cannot be repaired, remove and replace concrete.
- E. Repair of Unformed Surfaces: Test unformed surfaces, such as monolithic slabs, for smoothness and verify surface plane to tolerances specified for each surface and finish. Correct low and high areas as herein specified. Test unformed surfaces sloped to drain for trueness of slope, in addition to smoothness using a template having required slope.

## CAST-IN-PLACE CONCRETE

- F. Repair finished unformed surfaces that contain defects which affect durability of concrete. Surface defects, as such, include crazing, cracks in excess of 0.01" wide or which penetrate to reinforcement or completely through non-reinforced sections regardless of width, spalling, popouts, honeycomb, rock pockets and other objectionable conditions.
- G. Correct high areas in unformed surfaces by grinding after concrete has cured at least 14 days.
- H. Correct low areas in unformed surfaces during, or immediately after completion of surface finishing operations by cutting out low areas and replacing with fresh concrete. Finish repaired areas to blend into adjacent concrete. Proprietary patching compounds may be used when acceptable to Engineer.
- I. Repair defective areas, except random cracks and single holes not exceeding 1" diameter, by cutting out and replacing with fresh concrete. Remove defective areas to sound concrete with clean, square cuts and expose reinforcing steel with at least 3/4" clearance all around. Dampen concrete surfaces in contact with patching concrete and apply bonding compound. Mix patching concrete of same materials to provide concrete of same type or class as original concrete. Place, compact and finish to blend with adjacent finished concrete. Cure in same manner as adjacent concrete.
- J. Repair isolated random cracks and single holes not over 1" in diameter by dry-pack method. Groove top of cracks and cut-out holes to sound concrete and clean of dust din and loose panicles. Dampen cleaned concrete surfaces and apply bonding compound. Mix dry-pack, consisting of one pan portland cement to 2-1/2 pans fine aggregate passing a No. 16 mesh sieve using only enough water as required for handling and placing. Place dry pack after bonding compound has dried. Compact dry-pack mixture in place and finish to match adjacent concrete. Keep patched area continuously moist for not less than 72 hours.
- K. Perform structural repairs with prior approval of Engineer for method and procedure, using specified epoxy adhesive and mortar.
- L. Repair methods not specified above may be used, subject to acceptance of Engineer.

## 3.15 OUALITY CONTROL TESTING DURING CONSTRUCTION:

The Owner will employ a testing laboratory to perform tests and to submit test reports. Sampling and testing for quality control during placement of concrete may include the following, as directed by Engineer.

- A. Sampling Fresh Concrete: ASTM C 172, except modified for slump to comply with ASTM C 94
- B. Slump: ASTM C 143; one test at point of discharge for each day's pour of each type of concrete; additional tests when concrete consistency seems to have changed.
- C. Air Content: ASTM C 173, volumetric method for lightweight or normal weight concrete; ASTM C 231 pressure method for normal weight concrete; one for each day's pour of each type of airentrained concrete.
- D. Concrete Temperature: Test hourly when air temperature is 40F and below, and when 80 F and above; and each time a set of compression test specimens made.
- E. Compression Test Specimen: ASTM C 31; one set of 3 standard cylinders for each compressive strength test, unless otherwise directed. Mold and store cylinders for laboratory cured test specimens except when field-cure test specimens are required.

- F. Compressive Strength Tests: ASTM C 39; one set for each day's pour exceeding 5 cu. yds. plus additional sets for each 50 cu. yds. over and above the first 25 cu. yds. of each concrete class placed in any one day; one specimen tested at 7 days, one specimen tested at 28 days and one specimen retained in reserve for later testing if required.
  - 1. When frequency of testing will provide less than 5 strength tests for a given class of concrete, conduct testing from at least randomly selected batches or from each batch if fewer than 5 are used
  - 2. When total quantity of a given class of concrete is less than 50 cu. yds., strength test may be waived by Engineer if, in his judgement, adequate evidence of satisfactory strength is provided.
  - 3. When strength of field-cured cylinders is less than 85% of companion laboratory-cured cylinders, evaluate current operations and provide corrective procedures for protecting and curing the inplace concrete.
  - 4. Strength level of concrete will be considered satisfactory if averages of sets of three consecutive strength test results equal or exceed specified compressive strength, and no individual strength test result falls below specified compressive by more than 500 psi.
- G. Test results will be reported in writing to Engineer and Contractor within 24 hours that tests are made. Reports of compressive strength tests shall contain the project identification name and number, date of concrete placement, name of concrete testing service, concrete type and class, location of concrete batch in structure, design compressive strength at 28 days, concrete mix proportions and materials; compressive breaking strength and type of break for both 7-day tests and 28-day tests.
- H. Nondestructive Testing: Impact hammer, sonoscope, or other nondestructive device may be permitted but shall not be used as the sole basis for acceptance or rejection.
- I. Additional Tests: The testing service will make additional tests of in-place concrete when test results indicate specified concrete strengths and other characteristics have not been attained in the structure, as directed by Engineer. Testing service may conduct tests to determine adequacy of concrete by cored cylinders complying with ASTM C 42 or by other methods as directed. Contractor shall pay for such tests conducted, and any other additional testing as may be required, when unacceptable concrete is verified.

END OF SECTION 03 30 00

#### SECTION 04 26 13 - MASONRY VENEER

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. Section Includes:
  - 1. Clay face brick.
  - 2. Concrete face brick.
- B. Products Installed but Not Furnished under This Section:
  - 1. Steel lintels in masonry veneer.
  - 2. Steel shelf angles for supporting masonry veneer.

# 1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Samples for Verification: For each type and color of brick and colored mortar.

# 1.3 INFORMATIONAL SUBMITTALS

A. Material Certificates: For each type and size of product.

## 1.4 QUALITY ASSURANCE

- A. Sample Panels: Build sample panels to verify selections made under Sample submittals and to demonstrate aesthetic effects. Comply with requirements in Section 014000 "Quality Requirements" for mockups.
  - 1. Build sample panels for each type of exposed unit masonry construction in sizes approximately 48 inches (1200 mm) long by 48 inches (1200 mm) high by full thickness.

# 1.5 FIELD CONDITIONS

- A. Cold-Weather Requirements: Do not use frozen materials or materials mixed or coated with ice or frost. Do not build on frozen substrates. Remove and replace unit masonry damaged by frost or by freezing conditions. Comply with cold-weather construction requirements contained in TMS 602/ACI 530.1/ASCE 6.
  - 1. Cold-Weather Cleaning: Use liquid cleaning methods only when air temperature is 40 deg F (4 deg C) and higher and will remain so until masonry has dried, but not less than seven days after completing cleaning.

M.A. No. H1-22029-DA RMG - 22029

B. Hot-Weather Requirements: Comply with hot-weather construction requirements contained in TMS 602/ACI 530.1/ASCE 6.

## PART 2 - PRODUCTS

## 2.1 UNIT MASONRY, GENERAL

- A. Masonry Standard: Comply with TMS 602/ACI 530.1/ASCE 6, except as modified by requirements in the Contract Documents.
- B. Defective Units: Referenced masonry unit standards may allow a certain percentage of units to contain chips, cracks, or other defects exceeding limits stated. Do not use units where such defects will be exposed in the completed Work.

#### 2.2 BRICK

- A. Shapes: Provide shapes indicated and as follows, with exposed surfaces matching finish and color of exposed faces of adjacent units:
  - 1. For ends of sills and caps and for similar applications that would otherwise expose unfinished brick surfaces, provide units without cores or frogs and with exposed surfaces finished.
  - 2. Provide special shapes for applications where shapes produced by sawing would result in sawed surfaces being exposed to view.
- B. Clay Face Brick: Facing brick complying with ASTM C216 or hollow brick complying with ASTM C652, Class H40V (void areas between 25 and 40 percent of gross cross-sectional area).
  - 1. < Double click here to find, evaluate, and insert list of manufacturers and products. >
  - 2. [Grade SW] [Grade MW or SW].
  - 3. [Type FBX] [Type FBS] [Type FBA] [or] [Type HBX] [Type HBS] [Type HBA].
  - 4. Initial Rate of Absorption: Less than 30 g/30 sq. in. (30 g/194 sq. cm) per minute when tested according to ASTM C67.
  - 5. Efflorescence: Provide brick that has been tested according to ASTM C67 and is rated "not effloresced."
  - 6. Surface Coating: Brick with colors or textures produced by application of coatings shall withstand 50 cycles of freezing and thawing according to ASTM C67 with no observable difference in the applied finish when viewed from 10 feet (3 m)[ or shall have a history of successful use in Project's area].
  - 7. Size (Actual Dimensions): [3-1/2 inches (89 mm) wide by 2-1/4 inches (57 mm) high by 7-1/2 inches (190 mm) long] [or] [3-5/8 inches (92 mm) wide by 2-1/4 inches (57 mm) high by 7-5/8 inches (194 mm) long].
  - 8. Size (Actual Dimensions): [3-1/2 inches (89 mm)] [or] [3-5/8 inches (92 mm)] wide by 2-1/4 inches (57 mm) high by 8 inches (203 mm) long.
  - 9. Size (Actual Dimensions): [3-1/2 inches (89 mm) wide by 2-3/4 inches (70 mm) high by 7-1/2 inches (190 mm) long] [or] [3-5/8 inches (92 mm) wide by 2-13/16 inches (71 mm) high by 7-5/8 inches (194 mm) long].

- 10. Size (Actual Dimensions): [3-1/2 inches (89 mm) wide by 2-3/4 inches (70 mm) high by 8 inches (203 mm) long] [or] [3-5/8 inches (92 mm) wide by 2-13/16 inches (71 mm) high by 8 inches (203 mm) long].
- 11. Size (Actual Dimensions): [3-1/2 inches (89 mm) wide by 3-1/2 inches (89 mm) high by 7-1/2 inches (190 mm) long] [or] [3-5/8 inches (92 mm) wide by 3-5/8 inches (92 mm) high by 7-5/8 inches (194 mm) long].
- 12. Size (Actual Dimensions): [3-1/2 inches (89 mm) wide by 3-1/2 inches (89 mm) high by 11-1/2 inches (292 mm) long] [or] [3-5/8 inches (92 mm) wide by 3-5/8 inches (92 mm) high by 11-5/8 inches (295 mm) long].
- 13. Color and Texture: [Medium brown, wire cut] [Full-range red, sand molded] [Buff, velour] [Match Architect's samples] [As selected by Architect].

## 2.3 MORTAR MATERIALS

- A. Portland Cement: ASTM C150/C150M, Type I or II, except Type III may be used for cold-weather construction. Provide natural color or white cement as required to produce mortar color indicated.
- B. Hydrated Lime: ASTM C207, Type S. (STRUCTURAL ONLY)
- C. Portland Cement-Lime Mix: Packaged blend of portland cement and hydrated lime containing no other ingredients.
- D. Masonry Cement: ASTM C91/C91M.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. Lafarge North America Inc.
    - b. Lehigh Hanson; HeidelbergCement Group.
- E. Mortar Pigments: Natural and synthetic iron oxides and chromium oxides, compounded for use in mortar mixes and complying with ASTM C979/C979M. Use only pigments with a record of satisfactory performance in masonry mortar.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. <u>Euclid Chemical Company (The); an RPM company.</u>
    - b. <u>Solomon Colors, Inc</u>.
- F. Colored Cement Products: Packaged blend made from [portland cement and hydrated lime] [masonry cement] [or] [mortar cement] and mortar pigments, all complying with specified requirements, and containing no other ingredients.
  - 1. Colored Portland Cement-Lime Mix:
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:

- 1) <u>Lafarge North America Inc</u>.
- 2) <u>Lehigh Hanson; HeidelbergCement Group</u>.
- 2. Colored Masonry Cement:
  - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - 1) Lafarge North America Inc.
    - 2) <u>Lehigh Hanson; HeidelbergCement Group</u>.
- G. Aggregate for Mortar: ASTM C144.
  - 1. White-Mortar Aggregates: Natural white sand or crushed white stone.
  - 2. Colored-Mortar Aggregates: Natural sand or crushed stone of color necessary to produce required mortar color.
- H. Cold-Weather Admixture: Nonchloride, noncorrosive, accelerating admixture complying with ASTM C494/C494M, Type C, and recommended by manufacturer for use in masonry mortar of composition indicated.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. <u>BASF Corporation</u>.
    - b. Euclid Chemical Company (The); an RPM company.
- I. Water-Repellent Admixture: Liquid water-repellent mortar admixture intended for use with concrete bricks containing integral water repellent from same manufacturer.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. BASF Corporation.
    - b. Euclid Chemical Company (The); an RPM company.
- J. Water: Potable.

## 2.4 TIES AND ANCHORS

- A. General: Ties and anchors shall extend at least 1-1/2 inches (38 mm) into veneer but with at least a 5/8-inch (16-mm) cover on outside face.
- B. Materials: Provide ties and anchors specified in this article that are made from materials that comply with the following unless otherwise indicated:
  - 1. Hot-Dip Galvanized, Carbon-Steel Wire: ASTM A82/A82M, with ASTM A153/A153M, Class B-2 coating.
  - 2. Steel Sheet, Galvanized after Fabrication: ASTM A1008/A1008M, Commercial Steel, with ASTM A153/A153M, Class B coating.

- C. Adjustable Anchors for Connecting to Structural Steel Framing: Provide anchors that allow vertical or horizontal adjustment but resist tension and compression forces perpendicular to plane of wall.
  - 1. Anchor Section for Welding to Steel Frame: Crimped 1/4-inch- (6.35-mm-) diameter, hot-dip galvanized-steel wire.
  - 2. Tie Section: Triangular-shaped wire tie made from 0.25-inch- (6.35-mm-) diameter, hot-dip galvanized-steel wire.
- D. Adjustable Anchors for Connecting to Concrete: Provide anchors that allow vertical or horizontal adjustment but resist tension and compression forces perpendicular to plane of wall.
  - 1. Connector Section: Dovetail tabs for inserting into dovetail slots in concrete and attached to tie section; formed from 0.105-inch- (2.66-mm-) thick, steel sheet, galvanized after fabrication.
  - 2. Tie Section: Triangular-shaped wire tie made from 0.25-inch- (6.35-mm-) diameter, hot-dip galvanized-steel wire.
  - 3. Corrugated-Metal Ties: Metal strips not less than 7/8 inch (22 mm) wide with corrugations having a wavelength of 0.3 to 0.5 inch (7.6 to 12.7 mm) and an amplitude of 0.06 to 0.10 inch (1.5 to 2.5 mm) made from 0.105-inch- (2.66-mm-) thick steel sheet, galvanized after fabrication with dovetail tabs for inserting into dovetail slots in concrete.
- E. Adjustable Masonry-Veneer Anchors:
  - 1. General: Provide anchors that allow vertical adjustment but resist a 100-lbf (445-N) load in both tension and compression perpendicular to plane of wall without deforming or developing play in excess of 1/16 inch (1.5 mm).
  - 2. Fabricate sheet metal anchor sections and other sheet metal parts from 0.105-inch- (2.66-mm-) thick steel sheet, galvanized after fabrication.
  - 3. Fabricate wire ties from 0.25-inch- (6.35-mm-) diameter, hot-dip galvanized-steel wire unless otherwise indicated.
  - 4. Fabricate wire connector sections from 0.25-inch- (6.35-mm-) diameter, hot-dip galvanized, carbon-steel wire.
  - 5. Contractor's Option: Unless otherwise indicated, provide any of the adjustable masonry-veneer anchors specified.
  - 6. Screw-Attached, Masonry-Veneer Anchors: Wire tie and a rib-stiffened, sheet metal anchor section with screw holes top and bottom, with a projecting vertical tab having a slotted hole for inserting wire tie.
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
      - 1) <u>FERO Corporation</u>.
      - 2) Hohmann & Barnard, Inc.
  - 7. Screw-Attached, Masonry-Veneer Anchors: Wire tie and a rib-stiffened, sheet metal anchor section with screw holes top and bottom, with projecting tabs having holes for inserting vertical legs of wire tie formed to fit anchor section.

- a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1) Heckmann Building Products, Inc.
  - 2) <u>Hohmann & Barnard, Inc.</u>
  - 3) Wire-Bond.
- 8. Screw-Attached, Masonry-Veneer Anchors: Wire tie and a sheet metal anchor section, 1-1/4 inches (32 mm) wide by 9 inches (229 mm) long, with screw holes top and bottom and with raised rib-stiffened strap, 5/8 inch (16 mm) wide by 5-1/2 inches (140 mm) long, stamped into center to provide a slot between strap and base for inserting wire tie.
  - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by the following:
    - 1) Hohmann & Barnard, Inc.
- 9. Screw-Attached, Masonry-Veneer Anchors: Wire tie and a gasketed sheet metal anchor section, 1-1/4 inches (32 mm) wide by 6 inches (152 mm) long, with screw holes top and bottom; top and bottom ends bent to form pronged legs of length to match thickness of insulation or sheathing; and raised rib-stiffened strap, 5/8 inch (16 mm) wide by 6 inches (152 mm) long, stamped into center to provide a slot between strap and base for inserting wire tie.
  - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - 1) Hohmann & Barnard, Inc.
    - 2) <u>Wire-Bond</u>.

## 2.5 EMBEDDED FLASHING MATERIALS

- A. Metal Flashing: Provide metal flashing complying with SMACNA's "Architectural Sheet Metal Manual" and as follows:
  - 1. Stainless Steel: ASTM A240/A240M or ASTM A666, Type 304, 0.016 inch (0.40 mm) thick.
  - 2. Copper: ASTM B370, Temper H00, cold-rolled copper sheet, 16-oz./sq. ft. (4.9-kg/sq. m) weight or 0.0216 inch (0.55 mm) thick or ASTM B370, Temper H01, high-yield copper sheet, 12-oz./sq. ft. (3.7-kg/sq. m) weight or 0.0162 inch (0.41 mm) thick.
  - 3. Fabricate continuous flashings in sections 96 inches (2400 mm) long minimum, but not exceeding 12 feet (3.7 m). Provide splice plates at joints of formed, smooth metal flashing.
  - 4. Fabricate metal drip edges from stainless steel. Extend at least 3 inches (76 mm) into wall and 1/2 inch (13 mm) out from wall, with outer edge bent down 30 degrees and hemmed.
  - 5. Fabricate metal sealant stops from stainless steel. Extend at least 3 inches (76 mm) into wall and out to exterior face of wall. At exterior face of wall, bend metal back on itself for 3/4 inch (19 mm) and down into joint 1/4 inch (6 mm) to form a stop for retaining sealant backer rod.

- B. Flexible Flashing: Use one of the following unless otherwise indicated:
  - 1. Rubberized-Asphalt Flashing: Composite flashing product consisting of a pliable, adhesive rubberized-asphalt compound, bonded to a high-density, cross-laminated polyethylene film to produce an overall thickness of not less than [0.030 inch (0.76 mm)] [0.040 inch (1.02 mm)].
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
      - 1) Advanced Building Products Inc.
      - 2) <u>Carlisle Coatings & Waterproofing Inc.</u>
      - 3) GCP Applied Technologies Inc.
      - 4) Heckmann Building Products, Inc.
      - 5) Hohmann & Barnard, Inc.
      - 6) Wire-Bond.
  - 2. Butyl Rubber Flashing: Composite, self-adhesive, flashing product consisting of a pliable, butyl rubber compound, bonded to a high-density polyethylene film, aluminum foil, or spunbonded polyolefin to produce an overall thickness of not less than [0.030 inch (0.76 mm)] [0.040 inch (1.02 mm)].
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
      - 1) <u>DuPont Safety & Construction</u>.
      - 2) GCP Applied Technologies Inc.
      - 3) <u>Protecto Wrap Company</u>.
      - 4) Raven Industries, Inc.
  - 3. Elastomeric Thermoplastic Flashing: Composite flashing product consisting of a polyester-reinforced ethylene interpolymer alloy.
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
      - 1) <u>DuPont</u>.
      - 2) Hohmann & Barnard, Inc.
      - 3) Mortar Net Solutions.
      - 4) Wire-Bond.
  - 4. EPDM Flashing: Sheet flashing product made from ethylene-propylene-diene terpolymer, complying with ASTM D4637/D4637M, 0.040 inch (1.02 mm) thick.
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
      - 1) Carlisle Coatings & Waterproofing Inc.
      - 2) Firestone Specialty Products.
      - 3) Heckmann Building Products, Inc.
      - 4) <u>Hohmann & Barnard, Inc.</u>

# 5) <u>Wire-Bond</u>.

- C. Solder and Sealants for Sheet Metal Flashings: As specified in Section 076200 "Sheet Metal Flashing and Trim."
- D. Adhesives, Primers, and Seam Tapes for Flashings: Flashing manufacturer's standard products or products recommended by flashing manufacturer for bonding flashing sheets to each other and to substrates.

### 2.6 MISCELLANEOUS MASONRY ACCESSORIES

- A. Compressible Filler: Premolded filler strips complying with ASTM D1056, Grade 2A1; compressible up to 35 percent; of width and thickness indicated; formulated from neoprene or PVC.
- B. Weep/Vent Products: Use one of the following unless otherwise indicated:
  - 1. Cellular Plastic Weep/Vent: One-piece, flexible extrusion made from UV-resistant polypropylene copolymer, full height and width of head joint and depth 1/8 inch (3 mm) less than depth of outer wythe, in color selected from manufacturer's standard.
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
      - 1) Advanced Building Products Inc.
      - 2) Heckmann Building Products, Inc.
      - 3) Hohmann & Barnard, Inc.
      - 4) Wire-Bond.
  - 2. Mesh Weep/Vent: Free-draining mesh; made from polyethylene strands, full height and width of head joint and depth 1/8 inch (3 mm) less than depth of outer wythe; in color selected from manufacturer's standard.
    - a. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
      - 1) Advanced Building Products Inc.
      - 2) CavClear/Archovations, Inc.
      - 3) <u>Keene Building Products</u>.
      - 4) <u>Mortar Net Solutions</u>.
- C. Cavity Drainage Material: Free-draining mesh, made from polymer strands that will not degrade within the wall cavity.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. <u>Advanced Building Products Inc.</u>
    - b. <u>CavClear/Archovations, Inc.</u>
    - c. Heckmann Building Products, Inc.

M.A. No. H1-22029-DA RMG - 22029

- d. Hohmann & Barnard, Inc.
- e. Mortar Net Solutions.
- f. Wire-Bond.
- 2. Configuration: Provide one of the following:
  - a. Strips, full depth of cavity and 10 inches (250 mm) high, with dovetail-shaped notches 7 inches (175 mm) deep that prevent clogging with mortar droppings.
  - b. Strips, not less than 1-1/2 inches (38 mm) thick and 10 inches (250 mm) high, with dimpled surface designed to catch mortar droppings and prevent weep holes from clogging with mortar.
  - c. Sheets or strips, full depth of cavity and installed to full height of cavity.

#### 2.7 MASONRY CLEANERS

- A. Proprietary Acidic Cleaner: Manufacturer's standard-strength cleaner designed for removing mortar/grout stains, efflorescence, and other new construction stains from new masonry without discoloring or damaging masonry surfaces. Use product expressly approved for intended use by cleaner manufacturer and manufacturer of masonry units being cleaned.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. <u>Diedrich Technologies, Inc.</u>; a Hohmann & Barnard company.
    - b. <u>EaCo Chem, Inc</u>.
    - c. PROSOCO, Inc.

### 2.8 MORTAR MIXES

- A. General: Do not use admixtures, including pigments, air-entraining agents, accelerators, retarders, water-repellent agents, antifreeze compounds, or other admixtures unless otherwise indicated.
  - 1. Do not use calcium chloride in mortar or grout.
  - 2. Use portland cement-lime or masonry cement mortar unless otherwise indicated.
  - 3. Add cold-weather admixture (if used) at same rate for all mortar that will be exposed to view, regardless of weather conditions, to ensure that mortar color is consistent.
- B. Preblended, Dry Mortar Mix: Furnish dry mortar ingredients in form of a preblended mix. Measure quantities by weight to ensure accurate proportions, and thoroughly blend ingredients before delivering to Project site.
- C. Mortar for Unit Masonry: Comply with ASTM C270, Proportion Specification. Use Type N unless another type is indicated.
- D. Pigmented Mortar: Use colored cement product or select and proportion pigments with other ingredients to produce color required. Do not add pigments to colored cement products.
  - 1. Pigments shall not exceed 10 percent of portland cement by weight.
  - 2. Pigments shall not exceed 5 percent of masonry cement or mortar cement by weight.

3. Application: Use pigmented mortar for exposed mortar joints.

- M.A. No. H1-22029-DA RMG - 22029
- E. Colored-Aggregate Mortar: Produce required mortar color by using colored aggregates and natural color or white cement as necessary to produce required mortar color.
  - 1. Mix to match Architect's sample.
  - 2. Application: Use colored aggregate mortar for exposed mortar joints.

### PART 3 - EXECUTION

## 3.1 INSTALLATION, GENERAL

- A. Use full-size units without cutting if possible. If cutting is required to provide a continuous pattern or to fit adjoining construction, cut units with motor-driven saws; provide clean, sharp, unchipped edges. Allow units to dry before laying unless wetting of units is specified. Install cut units with cut surfaces and, where possible, cut edges concealed.
- B. Select and arrange units for exposed unit masonry to produce a uniform blend of colors and textures. Mix units from several pallets or cubes as they are placed.
- C. Wetting of Brick: Wet brick before laying if initial rate of absorption exceeds 30 g/30 sq. in. (30 g/194 sq. cm) per minute when tested according to ASTM C67. Allow units to absorb water so they are damp but not wet at time of laying.

### 3.2 TOLERANCES

### A. Dimensions and Locations of Elements:

- 1. For dimensions in cross section or elevation, do not vary by more than plus 1/2 inch (12 mm) or minus 1/4 inch (6 mm).
- 2. For location of elements in plan, do not vary from that indicated by more than plus or minus 1/2 inch (12 mm).
- 3. For location of elements in elevation, do not vary from that indicated by more than plus or minus 1/4 inch (6 mm) in a story height or 1/2 inch (12 mm) total.

## B. Lines and Levels:

- 1. For bed joints and top surfaces of bearing walls, do not vary from level by more than 1/4 inch in 10 feet (6 mm in 3 m), or 1/2 inch (12 mm) maximum.
- 2. For conspicuous horizontal lines, such as lintels, sills, parapets, and reveals, do not vary from level by more than 1/8 inch in 10 feet (3 mm in 3 m), 1/4 inch in 20 feet (6 mm in 6 m), or 1/2 inch (12 mm) maximum.
- 3. For vertical lines and surfaces, do not vary from plumb by more than 1/4 inch in 10 feet (6 mm in 3 m), 3/8 inch in 20 feet (9 mm in 6 m), or 1/2 inch (12 mm) maximum.
- 4. For conspicuous vertical lines, such as external corners, door jambs, reveals, and expansion and control joints, do not vary from plumb by more than 1/8 inch in 10 feet (3 mm in 3 m), 1/4 inch in 20 feet (6 mm in 6 m), or 1/2 inch (12 mm) maximum.
- 5. For lines and surfaces, do not vary from straight by more than 1/4 inch in 10 feet (6 mm in 3 m), 3/8 inch in 20 feet (9 mm in 6 m), or 1/2 inch (12 mm) maximum.

### M.A. No. H1-22029-DA RMG - 22029

## C. Joints:

- 1. For bed joints, do not vary from thickness indicated by more than plus or minus 1/8 inch (3 mm), with a maximum thickness limited to 1/2 inch (12 mm).
- 2. For exposed head joints, do not vary from thickness indicated by more than plus or minus 1/8 inch (3 mm). Do not vary from adjacent bed-joint and head-joint thicknesses by more than 1/8 inch (3 mm).

#### 3.3 LAYING MASONRY WALLS

- A. Lay out walls in advance for accurate spacing of surface bond patterns with uniform joint thicknesses and for accurate location of openings, movement-type joints, returns, and offsets. Avoid using less-than-half-size units, particularly at corners, jambs, and, where possible, at other locations.
- B. Bond Pattern for Exposed Masonry: Unless otherwise indicated, lay exposed masonry in running bond; do not use units with less-than-nominal 4-inch (100-mm) horizontal face dimensions at corners or jambs.
- C. Fill space between steel frames and masonry solidly with mortar unless otherwise indicated.

## 3.4 MORTAR BEDDING AND JOINTING

- A. Lay masonry units with completely filled bed and head joints; butter ends with sufficient mortar to fill head joints and shove into place. Do not deeply furrow bed joints or slush head joints.
- B. Tool exposed joints slightly concave when thumbprint hard, using a jointer larger than joint thickness unless otherwise indicated.

### 3.5 ANCHORED MASONRY VENEERS

- A. Anchor masonry veneers to wall framing with masonry-veneer anchors to comply with the following requirements:
  - 1. Fasten screw-attached anchors through sheathing to wall framing with metal fasteners of type indicated. Use two fasteners unless anchor design only uses one fastener.
  - 2. Embed tie sections in masonry joints.
  - 3. Locate anchor sections to allow maximum vertical differential movement of ties up and down.
  - 4. Space anchors as indicated, but not more than 18 inches (458 mm) o.c. vertically and 24 inches (610 mm) o.c. horizontally, with not less than one anchor for each 2 sq. ft. (0.2 sq. m) of wall area. Install additional anchors within 12 inches (305 mm) of openings and at intervals, not exceeding 8 inches (203 mm), around perimeter.
  - 5. Space anchors as indicated, but not more than 16 inches (406 mm) o.c. vertically and 25 inches (635 mm) o.c. horizontally, with not less than one anchor for each 2.67 sq. ft. (0.25 sq. m) of wall area. Install additional anchors within 12 inches (305 mm) of openings and at intervals, not exceeding 36 inches (914 mm), around perimeter.

- M.A. No. H1-22029-DA RMG - 22029
- 6. Space anchors as indicated, but not more than 18 inches (458 mm) o.c. vertically and horizontally. Install additional anchors within 12 inches (305 mm) of openings and at intervals, not exceeding 24 inches (610 mm), around perimeter.
- B. Provide not less than 1 inch (25 mm) of airspace between back of masonry veneer and face of sheathing.

### 3.6 ANCHORING MASONRY TO STRUCTURAL STEEL AND CONCRETE

- A. Anchor masonry to structural steel and concrete, where masonry abuts or faces structural steel or concrete to comply with the following:
  - 1. Provide an open space not less than 1 inch (25 mm) wide between masonry and structural steel or concrete unless otherwise indicated. Keep open space free of mortar and other rigid materials.
  - 2. Anchor masonry with anchors embedded in masonry joints and attached to structure.
  - 3. Space anchors as indicated, but not more than 24 inches (610 mm) o.c. vertically and 36 inches (915 mm) o.c. horizontally.

## 3.7 FLASHING, WEEP HOLES, AND VENTS

- A. General: Install embedded flashing and weep holes in masonry at shelf angles, lintels, ledges, other obstructions to downward flow of water in wall, and where indicated. Install vents at shelf angles, ledges, and other obstructions to upward flow of air in cavities, and where indicated.
- B. Install flashing as follows unless otherwise indicated:
  - 1. Prepare masonry surfaces so they are smooth and free from projections that could puncture flashing. Where flashing is within mortar joint, place through-wall flashing on sloping bed of mortar and cover with mortar. Before covering with mortar, seal penetrations in flashing with adhesive, sealant, or tape as recommended by flashing manufacturer.
  - 2. At lintels and shelf angles, extend flashing a minimum of 6 inches (150 mm) into masonry at each end. At heads and sills, extend flashing 6 inches (150 mm) at ends and turn up not less than 2 inches (50 mm) to form end dams.
  - 3. Install metal drip edges beneath flexible flashing at exterior face of wall. Stop flexible flashing 1/2 inch (13 mm) back from outside face of wall, and adhere flexible flashing to top of metal drip edge.
  - 4. Install metal flashing termination beneath flexible flashing at exterior face of wall. Stop flexible flashing 1/2 inch (13 mm) back from outside face of wall, and adhere flexible flashing to top of metal flashing termination.
- C. Install weep holes in veneers in head joints of first course of masonry immediately above embedded flashing.
  - 1. Use specified weep/vent products to form weep holes.
  - 2. Space weep holes 24 inches (600 mm) o.c. unless otherwise indicated.

- M.A. No. H1-22029-DA RMG - 22029
- D. Place cavity drainage material in airspace behind veneers to comply with configuration requirements for cavity drainage material in "Miscellaneous Masonry Accessories" Article.
- E. Install vents in head joints in exterior wythes at spacing indicated. Use specified weep/vent products to form vents.
  - 1. Close cavities off vertically and horizontally with blocking in manner indicated. Install through-wall flashing and weep holes above horizontal blocking.

# 3.8 FIELD QUALITY CONTROL

- A. Testing and Inspecting: Owner will engage special inspectors to perform tests and inspections and prepare reports. Allow inspectors access to scaffolding and work areas as needed to perform tests and inspections. Retesting of materials that fail to comply with specified requirements shall be done at Contractor's expense.
- B. Inspections: Special inspections according to Level B in TMS 402/ACI 530/ASCE 5.
  - 1. Begin masonry construction only after inspectors have verified proportions of siteprepared mortar.
- C. Testing Prior to Construction: One set of tests.
- D. Clay Masonry Unit Test: For each type of unit provided, according to ASTM C67 for compressive strength.
- E. Concrete Masonry Unit Test: For each type of unit provided, according to ASTM C140 for compressive strength.
- F. Mortar Aggregate Ratio Test (Proportion Specification): For each mix provided, according to ASTM C780.

# 3.9 REPAIRING, POINTING, AND CLEANING

- A. In-Progress Cleaning: Clean unit masonry as work progresses by dry brushing to remove mortar fins and smears before tooling joints.
- B. Final Cleaning: After mortar is thoroughly set and cured, clean exposed masonry as follows:
  - 1. Test cleaning methods on sample wall panel; leave one-half of panel uncleaned for comparison purposes.
  - 2. Protect adjacent stone and nonmasonry surfaces from contact with cleaner.
  - 3. Wet wall surfaces with water before applying cleaners; remove cleaners promptly by rinsing surfaces thoroughly with clear water.
  - 4. Clean brick by bucket-and-brush hand-cleaning method described in BIA Technical Notes 20.
  - 5. Clean masonry with a proprietary acidic cleaner applied according to manufacturer's written instructions.

# 3.10 MASONRY WASTE DISPOSAL

- A. Waste Disposal as Fill Material: Dispose of clean masonry waste, including excess or soil-contaminated sand, waste mortar, and broken masonry units, by crushing and mixing with fill material as fill is placed.
  - 1. Do not dispose of masonry waste as fill within 18 inches (450 mm) of finished grade.
- B. Excess Masonry Waste: Remove excess clean masonry waste that cannot be used as fill, as described above or recycled, and other masonry waste, and legally dispose of off Owner's property.

END OF SECTION 042613

#### SECTION 05 40 00 - COLD-FORMED METAL FRAMING

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. Section Includes:
  - 1. Load-bearing wall framing.
  - 2. Exterior non-load-bearing wall framing.

### 1.2 PREINSTALLATION MEETINGS

A. Preinstallation Conference: Conduct conference at Project site.

### 1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings:
  - 1. Include layout, spacings, sizes, thicknesses, and types of cold-formed steel framing; fabrication; and fastening and anchorage details, including mechanical fasteners.
  - 2. Indicate reinforcing channels, opening framing, supplemental framing, strapping, bracing, bridging, splices, accessories, connection details, and attachment to adjoining work.

# 1.4 INFORMATIONAL SUBMITTALS

- A. Welding certificates.
- B. Product certificates.
- C. Product test reports.
- D. Evaluation Reports: For post-installed anchors and power-actuated fasteners, from ICC-ES or other qualified testing agency acceptable to authorities having jurisdiction.

# 1.5 QUALITY ASSURANCE

- A. Testing Agency Qualifications: Qualified according to ASTM E329 for testing indicated.
- B. Product Tests: Mill certificates or data from a qualified independent testing agency.
- C. Welding Qualifications: Qualify procedures and personnel according to the following:

- M.A. No. H1-22029-DA RMG - 22029
- 1. AWS D1.1/D1.1M, "Structural Welding Code Steel."
- 2. AWS D1.3/D1.3M, "Structural Welding Code Sheet Steel."
- D. Comply with AISI S230 "Standard for Cold-Formed Steel Framing Prescriptive Method for One and Two Family Dwellings."

# PART 2 - PRODUCTS

### 2.1 MANUFACTURERS

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. AllSteel & Gypsum Products, Inc.
  - 2. CEMCO; California Expanded Metal Products Co.
  - 3. Consolidated Fabricators Corp.; Building Products Division.
  - 4. Craco Manufacturing, Inc.
  - 5. Jaimes Industries.
  - 6. MarinoWARE.
  - 7. <u>SCAFCO Steel Stud Company</u>.
  - 8. Southeastern Stud & Components, Inc.
  - 9. State Building Products, Inc.
  - 10. Steel Construction Systems.
  - 11. United Metal Products, Inc.
  - 12. United Steel Deck, Inc.

# 2.2 PERFORMANCE REQUIREMENTS

- A. Cold-Formed Steel Framing Standards: Unless more stringent requirements are indicated, framing shall comply with AISI S100, AISI S200, and the following:
  - 1. Wall Studs: AISI S211.
  - 2. Headers: AISI S212.
  - 3. Lateral Design: AISI S213.
- B. Fire-Resistance Ratings: Comply with ASTM E119; testing by a qualified testing agency. Identify products with appropriate markings of applicable testing agency.

# 2.3 COLD-FORMED STEEL FRAMING MATERIALS

- A. Steel Sheet: ASTM A1003/A1003M, Structural Grade, Type H, metallic coated, of grade and coating designation as follows:
  - 1. Grade: As required by structural performance.
  - 2. Coating: G90 (Z275) or equivalent.
- B. Steel Sheet for Vertical Deflection Clips: ASTM A653/A653M, structural steel, zinc coated, of grade and coating as follows:

- 1. Grade: As required by structural performance.
- 2. Coating: **G90** (Z275).

# 2.4 LOAD-BEARING WALL FRAMING

- A. Steel Studs: Manufacturer's standard C-shaped steel studs, of web depths indicated, punched, with stiffened flanges, and as follows:
  - 1. Minimum Base-Metal Thickness: 0.0428 inch (1.09 mm).
  - 2. Flange Width: 1-5/8 inches (41 mm).
  - 3. Section Properties: Insert minimum allowable calculated section modulus, moment of inertia, and allowable moment.
- B. Steel Track: Manufacturer's standard U-shaped steel track, of web depths indicated, unpunched, with straight flanges, and matching minimum base-metal thickness of steel studs.
- C. Steel Box or Back-to-Back Headers: Manufacturer's standard C-shapes used to form header beams, of web depths indicated, unpunched, with stiffened flanges, and as follows:
  - 1. Minimum Base-Metal Thickness: 0.0428 inch (1.09 mm).
  - 2. Flange Width: 1-5/8 inches (41 mm).
  - 3. Section Properties: Insert minimum allowable calculated section modulus, moment of inertia, and allowable moment.

# 2.5 FRAMING ACCESSORIES

- A. Fabricate steel-framing accessories from ASTM A1003/A1003M, Structural Grade, Type H, metallic coated steel sheet, of same grade and coating designation used for framing members.
- B. Provide accessories of manufacturer's standard thickness and configuration, unless otherwise indicated.

# 2.6 ANCHORS, CLIPS, AND FASTENERS

- A. Steel Shapes and Clips: ASTM A36/A36M, zinc coated by hot-dip process according to ASTM A123/A123M.
- B. Anchor Bolts: ASTM F1554, Grade 36, threaded carbon-steel hex-headed bolts, carbon-steel nuts, and flat, hardened-steel washers; zinc coated by hot-dip process according to ASTM A153/A153M, Class C.
- C. Post-Installed Anchors: Fastener systems with bolts of same basic metal as fastened metal, if visible, unless otherwise indicated; with working capacity greater than or equal to the design load, according to an evaluation report acceptable to authorities having jurisdiction, based on ICC-ES AC193 or ICC-ES AC308 as appropriate for the substrate.
  - 1. Uses: Securing cold-formed steel framing to structure.
  - 2. Type: Torque-controlled expansion anchor or adhesive anchor.

- M.A. No. H1-22029-DA RMG - 22029
- 3. Material for Interior Locations: Carbon-steel components zinc plated to comply with ASTM B633 or ASTM F1941 (ASTM F1941M), Class Fe/Zn 5, unless otherwise indicated.
- D. Power-Actuated Anchors: Fastener systems with working capacity greater than or equal to the design load, according to an evaluation report acceptable to authorities having jurisdiction, based on ICC-ES AC70.
- E. Mechanical Fasteners: ASTM C1513, corrosion-resistant-coated, self-drilling, self-tapping, steel drill screws.
  - 1. Head Type: Low-profile head beneath sheathing; manufacturer's standard elsewhere.

# 2.7 MISCELLANEOUS MATERIALS

- A. Galvanizing Repair Paint: ASTM A780/A780M.
- B. Cement Grout: Portland cement, ASTM C150/C150M, Type I; and clean, natural sand, ASTM C404. Mix at ratio of 1 part cement to 2-1/2 parts sand, by volume, with minimum water required for placement and hydration.
- C. Nonmetallic, Nonshrink Grout: Factory-packaged, nonmetallic, noncorrosive, nonstaining grout, complying with ASTM C1107/C1107M, and with a fluid consistency and 30-minute working time.
- D. Shims: Load-bearing, high-density, multimonomer, nonleaching plastic; or cold-formed steel of same grade and metallic coating as framing members supported by shims.
- E. Sealer Gaskets: Closed-cell neoprene foam, 1/4 inch (6 mm) thick, selected from manufacturer's standard widths to match width of bottom track or rim track members as required.

# PART 3 - EXECUTION

### 3.1 PREPARATION

- A. Install load-bearing shims or grout between the underside of load-bearing wall bottom track and the top of foundation wall or slab at locations with a gap larger than 1/4 inch (6 mm) to ensure a uniform bearing surface on supporting concrete or masonry construction.
- B. Install sealer gaskets at the underside of wall bottom track or rim track and at the top of foundation wall or slab at stud or joist locations.

# 3.2 INSTALLATION, GENERAL

A. Cold-formed steel framing may be shop or field fabricated for installation, or it may be field assembled.

- B. Install cold-formed steel framing according to AISI S200, AISI S202, and manufacturer's written instructions unless more stringent requirements are indicated.
- C. Install cold-formed steel framing and accessories plumb, square, and true to line, and with connections securely fastened.
- D. Install framing members in one-piece lengths unless splice connections are indicated for track or tension members.
- E. Install temporary bracing and supports to secure framing and support loads equal to those for which structure was designed. Maintain braces and supports in place, undisturbed, until entire integrated supporting structure has been completed and permanent connections to framing are secured.
- F. Do not bridge building expansion joints with cold-formed steel framing. Independently frame both sides of joints.
- G. Install insulation, specified in Section 072100 "Thermal Insulation," in framing-assembly members, such as headers, sills, boxed joists, and multiple studs at openings, that are inaccessible on completion of framing work.
- H. Fasten hole-reinforcing plate over web penetrations that exceed size of manufacturer's approved or standard punched openings.

### 3.3 LOAD-BEARING WALL INSTALLATION

- A. Install continuous top and bottom tracks sized to match studs. Align tracks accurately and securely anchor at corners and ends, and at spacings as follows:
  - 1. Anchor Spacing: 32 inches (813 mm).
- B. Squarely seat study against top and bottom tracks, with gap not exceeding 1/8 inch (3 mm) between the end of wall-framing member and the web of track. Fasten both flanges of study to top and bottom tracks. Space study as follows:
  - 1. Stud Spacing: 16 inches (406 mm).
- C. Set studs plumb, except as needed for diagonal bracing or required for nonplumb walls or warped surfaces and similar configurations.
- D. Align studs vertically where floor framing interrupts wall-framing continuity. Where studs cannot be aligned, continuously reinforce track to transfer loads.
- E. Align floor and roof framing over studs according to AISI S200, Section C1. Where framing cannot be aligned, continuously reinforce track to transfer loads.
- F. Anchor studs abutting structural columns or walls, including masonry walls, to supporting structure.

- G. Install headers over wall openings wider than stud spacing. Locate headers above openings. Fabricate headers of compound shapes indicated or required to transfer load to supporting studs, complete with clip-angle connectors, web stiffeners, or gusset plates.
  - 1. Frame wall openings with not less than a double stud at each jamb of frame. Fasten jamb members together to uniformly distribute loads.
  - 2. Install tracks and jack studs above and below wall openings. Anchor tracks to jamb studs with clip angles or by welding, and space jack studs same as full-height wall studs.
- H. Install supplementary framing, blocking, and bracing in stud framing indicated to support fixtures, equipment, services, casework, heavy trim, furnishings, and similar work requiring attachment to framing.
  - 1. If type of supplementary support is not indicated, comply with stud manufacturer's written recommendations and industry standards in each case, considering weight or load resulting from item supported.
- I. Install horizontal bridging in stud system, spaced vertically 48 inches (1220 mm). Fasten at each stud intersection.
  - 1. Strap Bridging: Combination of flat, taut, steel sheet straps of width and thickness indicated and stud-track solid blocking of width and thickness to match studs. Fasten flat straps to stud flanges, and secure solid blocking to stud webs or flanges.
- J. Install steel sheet diagonal bracing straps to both stud flanges; terminate at and fasten to reinforced top and bottom tracks. Fasten clip-angle connectors to multiple studs at ends of bracing and anchor to structure.
- K. Install miscellaneous framing and connections, including supplementary framing, web stiffeners, clip angles, continuous angles, anchors, and fasteners, to provide a complete and stable wall-framing system.

#### 3.4 ERECTION TOLERANCES

- A. Install cold-formed steel framing level, plumb, and true to line to a maximum allowable tolerance variation of 1/8 inch in 10 feet (1:960) and as follows:
  - 1. Space individual framing members no more than plus or minus 1/8 inch (3 mm) from plan location. Cumulative error shall not exceed minimum fastening requirements of sheathing or other finishing materials.

### 3.5 REPAIRS AND PROTECTION

A. Galvanizing Repairs: Prepare and repair damaged galvanized coatings on fabricated and installed cold-formed steel framing with galvanized repair paint according to ASTM A780/A780M and manufacturer's written instructions.

END OF SECTION 05 40 00

#### SECTION 06 10 00 - ROUGH CARPENTRY

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes the following:
  - 1. Framing with dimension lumber.
  - 2. Wood blocking, cants, and nailers.
  - 3. Wood furring and grounds.
  - 4. Wood sleepers.
  - 5. Plywood utilitarian backing/mounting panels.

# 1.2 SUBMITTALS

- A. Product Data: For each type of process and factory-fabricated product.
  - 1. Include data for wood-preservative and fire-retardant treatment from chemical treatment manufacturer and certification by treating plant that treated materials comply with requirements.
- B. Material Certificates: For dimension lumber specified to comply with minimum allowable unit stresses. Indicate species and grade selected for each use and design values approved by the American Lumber Standards Committee Board of Review.
- C. Research/Evaluation Reports: For the following, showing compliance with building code in effect for Project:
  - 1. Wood-preservative-treated wood.
  - 2. Fire-retardant-treated wood.
  - 3. Power-driven fasteners.
  - 4. Powder-actuated fasteners.
  - 5. Expansion anchors.
  - 6. Metal framing anchors.

### PART 2 - PRODUCTS

# 2.1 WOOD PRODUCTS, GENERAL

- A. Lumber: DOC PS 20 and applicable rules of grading agencies indicated. If no grading agency is indicated, provide lumber that complies with the applicable rules of any rules-writing agency certified by the ALSC Board of Review. Provide lumber graded by an agency certified by the ALSC Board of Review to inspect and grade lumber under the rules indicated.
  - 1. Factory mark each piece of lumber with grade stamp of grading agency.

- M.A. No. H1-22029-DA RMG - 22029
- 2. For exposed lumber indicated to receive a stained or natural finish, mark grade stamp on end or back of each piece or omit grade stamp and provide certificates of grade compliance issued by grading agency.
- 3. Provide dressed lumber, S4S, unless otherwise indicated.

# 2.2 WOOD-PRESERVATIVE-TREATED LUMBER

- A. Preservative Treatment by Pressure Process: AWPA C2, except that lumber that is not in contact with the ground and is continuously protected from liquid water may be treated according to AWPA C31 with inorganic boron (SBX).
  - 1. Preservative Chemicals: Acceptable to authorities having jurisdiction and containing no arsenic or chromium.
- B. Kiln-dry lumber after treatment to a maximum moisture content of 19 percent.
- C. Mark lumber with treatment quality mark of an inspection agency approved by the ALSC Board of Review.
- D. Application: Treat items indicated on Drawings, and the following:
  - 1. Wood cants, nailers, curbs, blocking, stripping, and similar members in connection with roofing, flashing, vapor barriers, and waterproofing.
  - 2. Wood sills, sleepers, blocking, furring, stripping, and similar concealed members in contact with masonry or concrete.

### 2.3 FIRE-RETARDANT-TREATED MATERIALS

- A. General: Comply with performance requirements in AWPA C20 (lumber) and AWPA C27 (plywood).
  - 1. Use Exterior type for exterior locations and interior wet areas such as plumbing chases, Janitor Rooms, Toilet Rooms, Shower areas, etc.
  - 2. Use Interior Type A, High Temperature (HT) for enclosed roof areas, framing in attic spaces, and where indicated.
  - 3. Use Interior Type A, unless otherwise indicated.
- B. Identify fire-retardant-treated wood with appropriate classification marking of testing and inspecting agency acceptable to authorities having jurisdiction.
- C. Application: Treat items indicated on Drawings, and the following:
  - 1. Concealed blocking.
  - 2. Plywood backing panels.

### 2.4 MISCELLANEOUS LUMBER

- A. General: Provide miscellaneous lumber indicated and lumber for support or attachment of other construction, including the following:
  - 1. Blocking and nailers.
  - 2. Cants.
  - 3. Furring.

M.A. No. H1-22029-DA RMG - 22029

- 4. Grounds for finish materials.
- B. For items of dimension lumber size, provide Standard, Stud, or No. 3 grade lumber with 19 percent maximum moisture content of any species.
- C. For concealed boards, provide lumber with 19 percent maximum moisture content and any of the following species and grades:
  - 1. Mixed southern pine, No. 3 grade; SPIB.
  - 2. Eastern softwoods, No. 3 Common grade; NeLMA.
  - 3. Northern species, No. 2 3 Common grade; NLGA.
  - 4. Western woods, Construction or No. 2 Common grade; WCLIB or WWPA.

#### 2.5 PLYWOOD BACKING PANELS

A. Telephone and Electrical Equipment Backing Panels: DOC PS 1, Exposure 1, A-C Plugged, fire-retardant treated, in thickness indicated or, if not indicated, not less than 1/2-inch (13-mm) nominal thickness.

#### 2.6 FASTENERS

- A. General: Provide fasteners of size and type indicated that comply with requirements specified.
  - 1. Where rough carpentry is exposed to weather, in ground contact, or in areas of high relative humidity, provide fasteners with hot-dip zinc coating complying with ASTM A 153/A 153M.
  - 2. Where pressure-preservative treated wood is used, provide 300 stainless steel fasteners only.
- B. Power-Driven Fasteners: NES NER-272.
- C. Bolts: Steel bolts complying with ASTM A 307, Grade A (ASTM F 568M, Property Class 4.6); with ASTM A 563 (ASTM A 563M) hex nuts and, where indicated, flat washers.

## 2.7 METAL FRAMING ANCHORS

- A. Manufacturers: Subject to compliance with requirements, provide products as required by one of the following:
  - 1. Cleveland Steel Specialty Co.
  - 2. Harlen Metal Products, Inc.
  - 3. KC Metals Products, Inc.
  - 4. Simpson Strong-Tie Co., Inc.
  - 5. USP Structural Connectors.
- B. Allowable Design Loads: Provide products with allowable design loads, as published by manufacturer, that meet or exceed those of basis-of-design products. Manufacturer's published values shall be determined from empirical data or by rational engineering analysis and demonstrated by comprehensive testing performed by a qualified independent testing agency.
- C. Submit information regarding locations and conditions where hangers, anchors, etc. are proposed to be used to Architect for approval.

M.A. No. H1-22029-DA RMG - 22029

D. Galvanized Steel Sheet: Hot-dip, zinc-coated steel sheet complying with ASTM A 653/A 653M, G60 (Z180) coating designation.

### **PART 3 - EXECUTION**

# 3.1 INSTALLATION

- A. Set rough carpentry to required levels and lines, with members plumb, true to line, cut, and fitted. Fit rough carpentry to other construction; scribe and cope as needed for accurate fit. Locate furring, nailers, blocking, grounds, and similar supports to comply with requirements for attaching other construction.
- B. Framing Standard: Comply with AF&PA's "Details for Conventional Wood Frame Construction," unless otherwise indicated.
- C. Metal Framing Anchors: Install metal framing to comply with manufacturer's written instructions.
- D. Comply with AWPA M4 for applying field treatment to cut surfaces of preservative-treated lumber.
- E. Securely attach rough carpentry work to substrate by anchoring and fastening as indicated, complying with the following:
  - 1. NES NER-272 for power-driven fasteners.
  - 2. Table 2304.9.1, "Fastening Schedule," in ICC's International Building Code.
  - 3. Table 23-II-B-1, "Nailing Schedule," and Table 23-II-B-2, "Wood Structural Panel Roof Sheathing Nailing Schedule," in ICBO's Uniform Building Code.
  - 4. Table 2305.2, "Fastening Schedule," in BOCA's BOCA National Building Code.
  - 5. Table 2306.1, "Fastening Schedule," in SBCCI's Standard Building Code.
  - 6. Table 602.3(1), "Fastener Schedule for Structural Members," and Table 602.3(2), "Alternate Attachments," in ICC's International One- and Two-Family Dwelling Code.

### 3.2 PROTECTION

A. Protect wood that has been treated with inorganic boron (SBX) from weather. If, despite protection, inorganic boron-treated wood becomes wet, apply EPA-registered borate treatment. Apply borate solution by spraying to comply with EPA-registered label.

END OF SECTION 06 10 00

### SECTION 06 20 23 - INTERIOR FINISH CARPENTRY

# PART 1 - GENERAL

### 1.1 SUMMARY

- A. This Section includes the following:
  - 1. Interior trim.
  - 2. Shelving.
- B. See Section 06 41 16 "Plastic Laminate Faced Cabinets" for related interior woodwork not specified in this Section.

### 1.2 SUBMITTALS

- A. Product Data: For each type of process and factory-fabricated product.
- B. Samples: For each type of trim indicated.

# 1.3 QUALITY ASSURANCE

#### **PART 2 - PRODUCTS**

# 2.1 MATERIALS, GENERAL

- A. Lumber: DOC PS 20 and applicable grading rules of inspection agencies certified by ALSC's Board of Review.
- B. Softwood Plywood: DOC PS 1.
- C. Hardboard: AHA A135.4.
- D. MDF: ANSI A208.2, Grade 130, made with binder containing no urea-formaldehyde resin.

# 2.2 INTERIOR TRIM

- A. Softwood Lumber Trim:
  - 1. Species and Grade: Douglas fir-larch or Douglas fir south, Prime or D finish; NLGA, WCLIB, or WWPA.
  - 2. Species and Grade: Southern pine, B & B finish; SPIB.
  - 3. Maximum Moisture Content: 19 percent.

- M.A. No. H1-22029-DA RMG - 22029
- B. Softwood Moldings for Transparent Finish (Stain or Clear Finish): WMMPA WM 4, N-grade wood moldings. Made to patterns included in WMMPA WM 12.
  - 1. Species: White pine, Shortleaf, or Loblolly pine.
  - 2. Maximum Moisture Content: 15 percent.
- C. Hardwood Moldings for Transparent Finish (Stain or Clear Finish): WMMPA HWM 2, N-grade wood moldings made to patterns included in WMMPA WM 12.
  - 1. Species: Clear Walnut.
  - 2. Maximum Moisture Content: 9 percent.
- D. Moldings for Opaque Finish (Painted): Made to patterns included in WMMPA WM 12.
  - 1. Hardwood Moldings: WMMPA HWM 2, P-grade.
    - a. Species and Grade: Douglas fir-larch or Douglas fir south; NLGA, WCLIB, or WWPA Superior or C & Btr finish.
    - b. Spruce-pine-fir; NeLMA, NLGA, WCLIB, or WWPA 1 Common.
    - c. Alder, aspen, basswood, cottonwood, gum, magnolia, soft maple, sycamore, tupelo, or yellow poplar; NHLA A Finish.
    - d. Maximum Moisture Content: 15 percent.
  - 2. Other Material as indicated: Primed MDF.
- E. Molding Patterns:
  - 1. Base Pattern: As Selected in sizes indicated

## 2.3 SHELVING

- A. Exposed Shelving: Made from one of the following materials, 3/4 inch (19 mm) thick:
  - 1. Wood boards as specified above for lumber trim for opaque finish.
  - 2. Softwood Boards:
    - a. Kiln-dried eastern white, Idaho white, lodgepole, ponderosa, radiata, or sugar pine; NeLMA, NLGA, or WWPA C Select (Choice).
    - b. Kiln-dried Douglas fir-larch, Douglas fir south, or hem-fir; SPIB Superior or C & Btr finish; NLGA, WCLIB, or WWPA; or southern pine; B & B finish.
- B. Standards for Adjustable Shelf Supports: BHMA A156.9, B04071; brass-finished steel.

# 2.4 MISCELLANEOUS MATERIALS

- A. Glue: Aliphatic-resin, polyurethane, or resorcinol wood glue.
  - 1. Use wood glue that has a VOC content of 30 g/L or less when calculated according to 40 CFR 59, Subpart D (EPA Method 24).

# **PART 3 - EXECUTION**

# 3.1 PREPARATION

A. Before installing interior finish carpentry, condition materials to average prevailing humidity in installation areas for a minimum of 24 hours.

# 3.2 INSTALLATION, GENERAL

- A. Install interior finish carpentry level, plumb, true, and aligned with adjacent materials. Use concealed shims where necessary for alignment.
  - 1. Scribe and cut interior finish carpentry to fit adjoining work.
  - 2. Countersink fasteners, fill surface flush, and sand where face fastening is unavoidable.
  - 3. Install to tolerance of 1/8 inch in 96 inches (3 mm in 2438 mm) for level and plumb. Install adjoining interior finish carpentry with 1/32-inch (0.8-mm) maximum offset.

### 3.3 INTERIOR TRIM INSTALLATION

A. Install with minimum number of joints practical, using full-length pieces from maximum lengths of lumber available. Cope at returns and miter at corners to produce tight-fitting joints. Use scarf joints for end-to-end joints.

## 3.4 SHELVING INSTALLATION

- A. Cut shelf cleats at ends of shelves about 1/2 inch (13 mm) less than width of shelves and sand exposed ends smooth.
- B. Install shelf cleats by fastening to framing or backing with finish nails or trim screws, set below face and filled. Space fasteners not more than 16 inches (400 mm) o.c.
- C. Cut shelves to neatly fit openings with only enough gap to allow shelves to be removed and reinstalled. Install shelves, fully seated on cleats, brackets, and supports.

END OF SECTION 06 20 23

This page intentionally left blank

#### SECTION 06 41 16 - PLASTIC-LAMINATE-FACED ARCHITECTURAL CABINETS

### PART 1 - GENERAL

#### 1.1 SUMMARY

#### A. Section Includes:

- 1. Plastic-laminate-faced architectural cabinets.
- 2. Wood furring, blocking, shims, and hanging strips for installing plastic-laminate-faced architectural cabinets unless concealed within other construction before cabinet installation.
- 3. Plastic laminate-faced countertops.
- 4. Cabinet/counter accessories and hardware.

### 1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product, including panel products high-pressure decorative laminate adhesive for bonding plastic laminate and cabinet hardware and accessories.
- B. Shop Drawings: Show location of each item, dimensioned plans and elevations, large-scale details, attachment devices, and other components.

# C. Samples:

- 1. Plastic laminates, for each color, pattern, and surface finish.
- 2. Thermoset decorative panels, for each color, pattern, and surface finish.

## 1.3 INFORMATIONAL SUBMITTALS

A. Product Data: For each type of cabinet hardware and accessories, including, but not limited to hinges, pulls, drawer glides and specialty hardware such as file hangers.

# 1.4 QUALITY ASSURANCE

- A. Fabricator Qualifications: Comply with standards of AWI's Quality Program.
- B. Installer Qualifications: Fabricator of products.

# 1.5 FIELD CONDITIONS

A. Environmental Limitations: Do not deliver or install cabinets until building is enclosed, wet work is complete, and HVAC system is operating and maintaining temperature and relative humidity at occupancy levels during the remainder of the construction period.

### PART 2 - PRODUCTS

# 2.1 PLASTIC-LAMINATE-FACED ARCHITECTURAL CABINETS

- A. Quality Standard: Unless otherwise indicated, comply with the "Architectural Woodwork Standards" for grades of architectural plastic-laminate cabinets indicated for construction, finishes, installation, and other requirements.
- B. Grade: AWI Premium Grade
- C. Type of Construction: Frameless.
- D. Cabinet, Door, and Drawer Front Interface Style: Flush overlay.
- E. Countertops: 1 ½" thick edges, self-edged plastic laminate
- F. High-Pressure Decorative Laminate: NEMA LD 3, grades as indicated or if not indicated, as required by woodwork quality standard.
  - 1. <u>Manufacturers</u>: Subject to compliance with requirements, provide products by one of the following:
    - a. <u>Formica Corporation</u>.
    - b. Wilsonart International; Div. of Premark International, Inc.
    - c. Pionite Decorative Surfaces; A Subsidiary of Panolam Industries International, Inc.
    - d. Nevamar Decorative Surfaces; A Subsidiary of Panolam Industries International, Inc.
- G. Laminate Cladding for Exposed Surfaces:
  - 1. Horizontal Surfaces: Grade HGS.
  - 2. Postformed Surfaces: Grade HGP.
  - 3. Vertical Surfaces: Grade VGS.
  - 4. Pattern Direction: Vertically for doors and fixed panels, woodgrain to run vertically, with matching grain across door & drawer fronts.
- H. Materials for Semiexposed Surfaces:
  - 1. Surfaces Other than drawer bodies: Thermoset decorative panels.
  - 2. Drawer Sides and Backs: Thermoset decorative panels.
  - 3. Drawer Bottoms: Thermoset decorative panels.
  - 4. Edges: PVC plastic matching adjoining surfaces
- I. Dust Panels: 1/4-inch (6.4-mm) plywood or tempered hardboard above compartments and drawers unless located directly under tops.
- J. Colors, Patterns, and Finishes: Provide materials and products that result in colors and textures of exposed laminate surfaces complying with the following requirements:
  - 1. As selected by Architect from laminate manufacturer's full range, including premium & HD laminate, in the following categories:
    - a. Solid colors, matte finish.

- b. Wood grains, matte finish.
- c. Patterns, matte finish.

# 2.2 PLASTIC-LAMINATE COUNTERTOPS

- A. Quality Standard: Unless otherwise indicated, comply with the "Architectural Woodwork Standards" for grades indicated for construction, installation, and other requirements.
- B. Grade: Custom.
- C. High-Pressure Decorative Laminate: NEMA LD 3, Grade HGS.
  - 1. <u>Manufacturers</u>: Subject to compliance with requirements, provide products by one of the following:
    - a. Formica Corporation.
    - b. Wilsonart International; Div. of Premark International, Inc.
    - c. Pionite Decorative Surfaces; A Subsidiary of Panolam Industries International, Inc.
    - d. Nevamar Decorative Surfaces; A Subsidiary of Panolam Industries International, Inc.
- D. Colors, Patterns, and Finishes: Provide materials and products that result in colors and textures of exposed laminate surfaces complying with the following requirements:
  - 1. As selected by Architect from manufacturer's full range in the following categories:
    - a. Solid colors, matte finish.
    - b. Wood grains, matte finish.
    - c. Patterns, matte finish.
    - d. Premium, all finishes.
- E. Edge Treatment: Same as laminate cladding on horizontal surfaces.
- F. Core Material at Sinks: exterior-grade plywood.
- G. Core Thickness: 3/4 inch (19 mm).
  - 1. Build up countertop thickness to 1-1/2 inches (38 mm) at front, back, and ends with additional layers of core material laminated to top.

# 2.3 WOOD MATERIALS

- A. Wood Products: Provide materials that comply with requirements of referenced quality standard for each type of woodwork and quality grade specified unless otherwise indicated.
  - 1. Wood Moisture Content: 5 to 10 percent.
- B. Composite Wood and Agrifiber Products: Provide materials that comply with requirements of referenced quality standard for each type of woodwork and quality grade specified unless otherwise indicated.
  - 1. Medium-Density Fiberboard: ANSI A208.2, Grade 130
  - 2. Softwood Plywood: DOC PS 1, medium-density overlay.

- M.A. No. H1-22029-DA RMG - 22029
- 3. Veneer-Faced Panel Products (Hardwood Plywood): HPVA HP-1
- 4. Thermoset Decorative Panels: Medium-density fiberboard finished with thermally fused, melamine-impregnated decorative paper and complying with requirements of NEMA LD 3, Grade VGL, for test methods 3.3, 3.4, 3.6, 3.8, and 3.10.
- 5. Underlayment for countertops with sinks: Type "M" moisture resistant.

# 2.4 CABINET HARDWARE AND ACCESSORIES

- A. Frameless Concealed Hinges (European Type): BHMA A156.9, B01602, 170 degrees of opening, self-closing.
- B. Wire Pulls: Back mounted, solid metal, 4 inches (100 mm) long, 5/16 inch (8 mm) in diameter.
- C. Catches: Magnetic catches, BHMA A156.9, B03141.
- D. Shelf Rests: BHMA A156.9, B04013; metal.
- E. Drawer Slides: BHMA A156.9.
  - 1. Grade 1 and Grade 2: Side mounted; full-extension type; zinc-plated steel with polymer rollers.
  - 2. Grade 1HD-100 and Grade 1HD-200: Side mounted; full-extension type; zinc-plated-steel ball-bearing slides.
  - 3. For drawers not more than 3 inches (75 mm) high and not more than 24 inches (600 mm) wide provide Grade 1.
  - 4. For drawers more than 3 inches (75 mm) high but not more than 6 inches (150 mm) high and not more than 24 inches (600 mm) wide, provide Grade 1HD-100.
  - 5. For drawers more than 6 inches (150 mm) high or more than 24 inches (600 mm) wide, provide Grade 1HD-200.
  - 6. Round plastic wire grommets at approx. 3'-0" o.c. in designated cabinets and counters, 2" diameter with top cover. Verify exact locations in field with Owner:
    - a. Series EDP by Doug Mockett & Company, Inc.
    - b. Standard 2" Plastic Grommet by Rockler Woodworking and Hardware.
- F. Door Locks: BHMA A156.11, E07121.
- G. Drawer Locks: BHMA A156.11, E07041. Master key locks within same room.
- H. Door and Drawer Silencers: BHMA A156.16, L03011.
- I. Exposed Hardware Finishes: For exposed hardware, provide finish that complies with BHMA A156.18 for BHMA finish number indicated.
  - 1. To match door hardware finish selected.
- J. Grommets for Cable Passage through Countertops: 1-7/8 inches inside diameter, molded-plastic grommets and matching plastic caps with slot for wire passage. Color as selected by Architect from manufacturer's full range of standard colors.
  - 1. <u>Doug Mockett & Company, Inc.</u>

# 2. Schwin Design.

# K. Counter Support:

- 1. Acceptable products:
  - a. Rakks EH Series, Flush Mount
  - b. Kitchensource
- 2. Finish: To be selected by Architect.

# 2.5 MISCELLANEOUS MATERIALS

- A. Furring, Blocking, Shims, and Hanging Strips: Softwood or hardwood lumber, kiln dried to less than 15 percent moisture content.
- B. Anchors: Select material, type, size, and finish required for each substrate for secure anchorage. Provide metal expansion sleeves or expansion bolts for post-installed anchors. Use nonferrousmetal or hot-dip galvanized anchors and inserts at inside face of exterior walls and at floors.
- C. Adhesive for Bonding Plastic Laminate: Unpigmented contact cement.
  - 1. Adhesive for Bonding Edges: Hot-melt adhesive or adhesive specified above for faces.

#### 2.6 FABRICATION

- A. Complete fabrication, including assembly and hardware application, to maximum extent possible before shipment to Project site. Disassemble components only as necessary for shipment and installation. Where necessary for fitting at site, provide ample allowance for scribing, trimming, and fitting.
- B. Fabricate countertops to dimensions, profiles, and details indicated. Provide front and end overhang of 1 inch (25 mm) over base cabinets. Ease edges to radius indicated for the following:
  - 1. Solid-Wood (Lumber) Members: 1/16 inch (1.5 mm) unless otherwise indicated.
- C. Shop-cut openings to maximum extent possible to receive hardware, appliances, plumbing items, electrical work, and similar items. Locate openings accurately and use templates or roughing-in diagrams to produce accurately sized and shaped openings. Sand edges of cutouts to remove splinters and burrs.
- D. Install glass (where required) to comply with applicable requirements in Section 088000 "Glazing" and in GANA's "Glazing Manual." For glass in wood frames, secure glass with removable stops.

# 3.1 PREPARATION

A. Before installation, condition cabinets to average prevailing humidity conditions in installation areas.

### 3.2 INSTALLATION

- A. Grade: Install cabinets to comply with same grade as item to be installed.
- B. Install cabinets level, plumb, true, and straight. Shim as required with concealed shims. Install level and plumb to a tolerance of 1/8 inch in 96 inches (3 mm in 2400 mm).
- C. Grade: Install countertops to comply with same grade as item to be installed.
- D. Assemble countertops and complete fabrication at Project site to the extent that it was not completed in the shop.
  - 1. Provide cutouts for appliances, plumbing fixtures, electrical work, and similar items.
  - 2. Seal edges of cutouts by saturating with varnish.
- E. Field Jointing: Prepare edges to be joined in shop so Project-site processing of top and edge surfaces is not required.
  - 1. Secure field joints in plastic-laminate countertops with concealed clamping devices located within 6 inches (150 mm) of front and back edges and at intervals not exceeding 24 inches (600 mm). Tighten according to manufacturer's written instructions to exert a constant, heavy-clamping pressure at joints.
- F. Install countertops level, plumb, true, and straight. Shim as required with concealed shims. Install level and plumb to a tolerance of 1/8 inch in 96 inches (3 mm in 2400 mm).
- G. Scribe and cut cabinets to fit adjoining work, refinish cut surfaces, and repair damaged finish at cuts.
- H. Anchor cabinets to anchors or blocking built in or directly attached to substrates. Secure with countersunk, concealed fasteners and blind nailing. Use fine finishing nails or finishing screws for exposed fastening, countersunk and filled flush with woodwork.
- I. Cabinets: Install without distortion so doors and drawers fit openings properly and are accurately aligned. Adjust hardware to center doors and drawers in openings and to provide unencumbered operation. Complete installation of hardware and accessory items as indicated.
  - 1. Install cabinets with no more than 1/8 inch in 96-inch (3 mm in 2400-mm) sag, bow, or other variation from a straight line.
  - 2. Fasten wall cabinets through back, near top and bottom, and at ends not more than 16 inches (400 mm) o.c. with No. 10 wafer-head sheet metal screws through metal backing or metal framing behind wall finish.

M.A. No. H1-22029-DA

RMG - 22029

- J. Countertops: Anchor securely by screwing through corner blocks of base cabinets or other supports into underside of countertop.
  - 1. Install countertops with no more than 1/8 inch in 96-inch (3 mm in 2400-mm) sag, bow, or other variation from a straight line.
  - 2. Secure backsplashes to tops with concealed metal brackets at 16 inches (400 mm) o.c. and to walls with adhesive.
  - 3. Seal junctures of tops, splashes, and walls with mildew-resistant silicone sealant or another permanently elastic sealing compound recommended by countertop material manufacturer.

END OF SECTION 06 41 16

This page intentionally left blank

#### SECTION 07 21 00 - THERMAL & ACOUSTICAL INSULATION

### PART 1 - GENERAL

### 1.1 SUMMARY

- A. Section Includes:
  - 1. Glass-fiber blanket insulation.
  - 2. Acoustical batt insulation.

# 1.2 ACTION SUBMITTALS

A. Product Data: For each type of product indicated.

### 1.3 INFORMATIONAL SUBMITTALS

- A. Product test reports.
- B. Research/evaluation reports.

#### PART 2 - PRODUCTS

### 2.1 GLASS-FIBER BLANKET INSULATION

- A. <u>Manufacturers</u>: Subject to compliance with requirements, provide products by one of the following:
  - 1. <u>CertainTeed Corporation</u>.
  - 2. Johns Manville.
  - 3. Knauf Insulation.
  - 4. Owens Corning.
- B. Glass-Fiber Blanket Insulation, Unfaced: ASTM C665, Type I; passing ASTM E136 for combustion characteristics.
  - 1. Flame-Spread Index: Not more than 25 when tested in accordance with ASTM E84.
  - 2. Smoke-Developed Index: Not more than 50 when tested in accordance with ASTM E84.
  - 3. Labeling: Provide identification of mark indicating R-value of each piece of insulation 12 inches (305 mm) and wider in width.
  - 4. Ceilings: R-38
  - 5. Walls:
    - a. 4": R-15
    - b. 6": R-19

# 2.2 GLASS-FIBER BATT ACOUSTICAL INSULATION

- A. <u>Manufacturers</u>: Subject to compliance with requirements, provide products by one of the following:
  - 1. <u>CertainTeed Corporation</u>.
  - 2. Johns Manville.
  - 3. Knauf Insulation.
  - 4. Owens Corning.
- B. Unfaced, glass-fiber batt insulation for walls and above ceiling. 3 ½" thick.
  - 1. Certainteed "NoiseReducer".
  - 2. Johns Manville "Sound Control Batts".
  - 3. Owens Corning "QuietZone"
- C. ASTM E90-1990; ASTM C665 Type I, & ASTM E 136 for acoustical batt insulation.

# 2.3 ACCESSORIES

- A. Insulation for Miscellaneous Voids:
  - 1. Glass-Fiber Insulation: ASTM C764, Type II, loose fill; with maximum flame-spread and smoke-developed indexes of 5, per ASTM E84.
- B. Mesh for install at bottom of truss.

### **PART 3 - EXECUTION**

### 3.1 INSTALLATION, GENERAL

- A. Comply with insulation manufacturer's written instructions applicable to products and applications indicated.
- B. Install insulation that is undamaged, dry, and unsoiled and that has not been left exposed to ice, rain, or snow at any time.
- C. Extend insulation to envelop entire area to be insulated. Cut and fit tightly around obstructions and fill voids with insulation. Remove projections that interfere with placement.
- D. Provide sizes to fit applications indicated and selected from manufacturer's standard thicknesses, widths, and lengths. Apply single layer of insulation units to produce thickness indicated unless multiple layers are otherwise shown or required to make up total thickness.

# 3.2 INSTALLATION OF INSULATION FOR FRAMED CONSTRUCTION

A. Apply insulation units to substrates by method indicated, complying with manufacturer's written instructions. If no specific method is indicated, bond units to substrate with adhesive or use mechanical anchorage to provide permanent placement and support of units.

- B. Glass-Fiber Blanket Insulation: Install in cavities formed by framing members according to the following requirements: (Similar for both thermal and acoustical)
  - 1. Use insulation widths and lengths that fill the cavities formed by framing members. If more than one length is required to fill the cavities, provide lengths that will produce a snug fit between ends.
  - 2. Place insulation in cavities formed by framing members to produce a friction fit between edges of insulation and adjoining framing members.
  - 3. Maintain 3-inch (76-mm) clearance of insulation around recessed lighting fixtures not rated for or protected from contact with insulation.
  - 4. For metal-framed wall cavities where cavity heights exceed 96 inches (2438 mm), support unfaced blankets mechanically and support faced blankets by taping flanges of insulation to flanges of metal studs.
- C. Acoustical Partitions: Where designated "Sound Walls" on drawings, in addition to above:
  - 1. Lay additional continuous strip on suspended ceiling, to cover 4'-0" wide each side of partition.
  - 2. Completely caulk the perimeter of the partition, including along floor track and ceiling perimeter angle with acoustical caulk as specified in Section 07 92 00, Joint Sealants.

END OF SECTION 07 21 00

This page intentionally left blank

- M.A. No. H1-22029-DA RMG - 22029
- 2. Mark the Contract Drawings or Shop Drawings, whichever is most capable of showing actual physical conditions, completely and accurately. If Shop Drawings are marked, show cross-reference on the Contract Drawings.
- 3. Mark record sets with erasable, red-colored pencil. Use other colors to distinguish between changes for different categories of the Work at same location.
- 4. Note Construction Change Directive numbers, alternate numbers, Change Order numbers, and similar identification, where applicable.
- B. Format: Identify and date each Record Drawing; include the designation "PROJECT RECORD DRAWING" in a prominent location.
  - 1. Record Prints: Organize to Record Prints and newly prepared Record Drawings into manageable sets. Bind each set with durable paper cover sheets. Include identification on cover sheets.
  - 2. Identification: As follows:
    - a. Project name.
    - b. Date.
    - c. Designation "PROJECT RECORD DRAWINGS."
    - d. Name of Architect.
    - e. Name of Contractor.

### 2.2 RECORD SPECIFICATIONS

- A. Preparation: Mark Specifications to indicate the actual product installation where installation varies from that indicated in Specifications, addenda, and contract modifications. Provide one full set of final approved submittals.
  - 1. Give particular attention to information on concealed products and installations that cannot be readily identified and recorded later.
  - 2. Mark copy with the proprietary name and model number of products, materials, and equipment furnished, including substitutions and product options selected.
  - 3. Record the name of manufacturer, supplier, Installer, and other information necessary to provide a record of selections made.
  - 4. Note related Change Orders, Record Product Data, and Record Drawings where applicable.

## 2.3 RECORD PRODUCT DATA

- A. Preparation: Mark Product Data to indicate the actual product installation where installation varies substantially from that indicated in Product Data submittal.
  - 1. Give particular attention to information on concealed products and installations that cannot be readily identified and recorded later.
  - 2. Include significant changes in the product delivered to Project site and changes in manufacturer's written instructions for installation.
  - 3. Note related Change Orders, Record Specifications, and Record Drawings where applicable.

# 2.4 MISCELLANEOUS RECORD SUBMITTALS

A. Assemble miscellaneous records required by other Specification Sections for miscellaneous record keeping and submittal in connection with actual performance of the Work. Bind or file miscellaneous records and identify each, ready for continued use and reference.

# **PART 3 - EXECUTION**

## 3.1 RECORDING AND MAINTENANCE

- A. Recording: Maintain one copy of each submittal during the construction period for Project Record Document purposes. Post changes and modifications to Project Record Documents as they occur; do not wait until the end of Project.
  - 1. Electronic Submittals: it is the contractor's responsibility to download all records from the service provider and provide an electronic copy to the owner as part of the Owner's Manual.
- B. Maintenance of Record Documents and Samples: Store Record Documents and Samples in the field office apart from the Contract Documents used for construction. Do not use Project Record Documents for construction purposes. Maintain Record Documents in good order and in a clean, dry, legible condition, protected from deterioration and loss. Provide access to Project Record Documents for Architect's reference during normal working hours.

END OF SECTION 01 78 39

This page intentionally left blank

#### SECTION 07 62 00 - SHEET METAL FLASHING AND TRIM

### PART 1 - GENERAL

### 1.1 SUMMARY

### A. Section Includes:

1. Formed miscellaneous sheet metal trim and flashing fabrications to cover gaps, joints and where dissimilar material adjoin.

#### 1.2 RELATED SECTIONS

1. Refer to Section 07 52 16 SBS Modified Bitumen Roofing for associated flashings, scuppers, downspouts and trims related to built-up roofing.

# 1.3 PREINSTALLATION MEETINGS

- A. Refer to front end documents for sample agendas for Roofing Conferences.
- B. Preliminary Conference: Conduct at Project Site at least six weeks prior to anticipated start of roofing activities.
- C. Pre-Installation Conference: Conduct at Project Site within one week of anticipated start of roofing activities.

## 1.4 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings: For sheet metal flashing and trim.
  - 1. Include plans, elevations, sections, and attachment details.
  - 2. Distinguish between shop- and field-assembled work.
  - 3. Include identification of finish for each item.
  - 4. Include pattern of seams and details of termination points, expansion joints and expansion-joint covers, direction of expansion, roof-penetration flashing, and connections to adjoining work.
- C. Samples: For each exposed product and for each color and texture specified.
- D. Full range of colors available.

# 1.5 INFORMATIONAL SUBMITTALS

- A. Product certificates.
- B. Product test reports.
- C. Sample warranty.

# 1.6 CLOSEOUT SUBMITTALS

A. Maintenance data.

# 1.7 QUALITY ASSURANCE

- A. Fabricator Qualifications: Employs skilled workers who custom fabricate sheet metal flashing and trim similar to that required for this Project and whose products have a record of successful in-service performance.
  - 1. For copings and roof edge flashings that are SPRI ES-1 tested, shop shall be listed as able to fabricate required details as tested and approved.
- B. Mockups: Build mockups of individual conditions to verify selections and secure architect's approval to demonstrate aesthetic effects and to set quality standards for fabrication and installation.

# 1.8 WARRANTY

- A. Special Warranty on Finishes: Manufacturer agrees to repair finish or replace sheet metal flashing and trim that shows evidence of deterioration of factory-applied finishes within specified warranty period.
  - 1. Finish Warranty Period: 20 years from date of Recommendation of Acceptance.

### PART 2 - PRODUCTS

# 2.1 PERFORMANCE REQUIREMENTS

- A. General: Sheet metal flashing and trim assemblies shall withstand wind loads, structural movement, thermally induced movement, and exposure to weather without failure due to defective manufacture, fabrication, installation, or other defects in construction. Completed sheet metal flashing and trim shall not rattle, leak, or loosen, and shall remain watertight.
- B. Sheet Metal Standard for Flashing and Trim: Comply with NRCA's "The NRCA Roofing Manual" and SMACNA's "Architectural Sheet Metal Manual" requirements for dimensions and profiles shown unless more stringent requirements are indicated.
- C. Thermal Movements: Allow for thermal movements from ambient and surface temperature changes.

## 2.2 SHEET METALS

- A. General: Protect mechanical and other finishes on exposed surfaces from damage by applying strippable, temporary protective film before shipping.
  - 1. Exposed Coil-Coated Finish:
    - a. Two-Coat Fluoropolymer: AAMA 2605. Fluoropolymer finish containing not less than 70 percent PVDF resin by weight in color coat. Prepare, pretreat, and apply coating to exposed metal surfaces to comply with coating and resin manufacturers' written instructions.
    - b. Kynar 500 or Hylar 5000.

- 2. Color: As selected by Architect from manufacturer's full range. User Agency shall sign and date all approved color selections of pre-finished metals prior to Contractor ordering materials and forward copy to Facility Planning & Control.
- B. Galvanized Sheet Steel: Concealed locations to be hot-dipped galvanized sheet steel.
  - 1. Minimum G-90, ASTM A653
  - 2. Approval is required prior to use of any galvanized metal.

# 2.3 MISCELLANEOUS MATERIALS

- A. General: Provide materials and types of fasteners, solder, protective coatings, sealants, and other miscellaneous items as required for complete sheet metal flashing and trim installation and as recommended by manufacturer of primary sheet metal unless otherwise indicated.
- B. Fasteners: Wood screws, annular threaded nails, self-tapping screws, self-locking rivets and bolts, and other suitable fasteners designed to withstand design loads and recommended by manufacturer of primary sheet metal.
  - 1. General: Blind fasteners or self-drilling screws, gasketed, with hex-washer head.
    - a. Exposed Fasteners: Heads matching color of sheet metal using plastic caps or factory-applied coating. Provide metal-backed EPDM or PVC sealing washers under heads of exposed fasteners bearing on weather side of metal.
    - b. Blind Fasteners: High-strength aluminum or stainless-steel rivets suitable for metal being fastened.
    - c. Spikes and Ferrules: Same material as gutter; with spike with ferrule matching internal gutter width.
  - 2. Fasteners for Zinc-Coated (Galvanized) or Aluminum-Zinc Alloy-Coated Steel Sheet: Series 300 stainless steel or hot-dip galvanized steel according to ASTM A 153/A 153M or ASTM F 2329.

#### C. Solder:

- 1. For Zinc-Coated (Galvanized) Steel: ASTM B 32, Grade Sn50, 50 percent tin and 50 percent lead or Grade Sn60, 60 percent tin and 40 percent lead.
- D. Sealant Tape: Pressure-sensitive, 100 percent solids, polyisobutylene compound sealant tape with release-paper backing. Provide permanently elastic, nonsag, nontoxic, nonstaining tape 1/2 inch (13 mm) wide and 1/8 inch (3 mm) thick.
- E. Elastomeric Sealant: ASTM C 920, elastomeric polyurethane or silicone polymer sealant; of type, grade, class, and use classifications required to seal joints in sheet metal flashing and trim and remain watertight.
- F. Butyl Sealant: ASTM C 1311, single-component, solvent-release butyl rubber sealant; polyisobutylene plasticized; heavy bodied for hooked-type expansion joints with limited movement.
- G. Epoxy Seam Sealer: Two-part, noncorrosive, aluminum seam-cementing compound, recommended by aluminum manufacturer for exterior nonmoving joints, including riveted joints.
- H. Bituminous Coating: Cold-applied asphalt emulsion according to ASTM D 1187 where dissimilar metals adjoin.

# 2.4 FABRICATION, GENERAL

- A. General: Custom fabricate sheet metal flashing and trim to comply with details shown and recommendations in cited sheet metal standard that apply to design, dimensions, geometry, metal thickness, and other characteristics of item required. Fabricate sheet metal flashing and trim in shop to greatest extent possible.
  - 1. Obtain field measurements for accurate fit before shop fabrication.
  - 2. Form sheet metal flashing and trim to fit substrates without excessive oil canning, buckling, and tool marks; true to line, levels, and slopes; and with exposed edges folded back to form hems.
  - 3. Conceal fasteners and expansion provisions where possible. Do not use exposed fasteners on faces exposed to view.
- B. Expansion Provisions: Form metal for thermal expansion of exposed flashing and trim.
  - 1. Form expansion joints of intermeshing hooked flanges, not less than 1 inch (25 mm) deep, filled with butyl sealant concealed within joints.
  - 2. Use lapped expansion joints only where indicated on Drawings.
- C. Sealant Joints: Where movable, nonexpansion-type joints are required, form metal to provide for proper installation of elastomeric sealant according to cited sheet metal standard.
- D. Fabricate cleats and attachment devices from same material as accessory being anchored or from compatible, noncorrosive metal.
- E. Fabricate cleats and attachment devices of sizes as recommended by cited sheet metal standard for application, but not less than thickness of metal being secured.
- F. Seams: Fabricate nonmoving seams with flat-lock seams. Tin edges to be seamed, form seams, and solder.
- G. Seams in metal copings and parapet caps to be standing seam.
- H. Seams: Fabricate nonmoving seams with flat-lock seams. Form seams and seal with elastomeric sealant unless otherwise recommended by sealant manufacturer for intended use. Rivet joints where necessary for strength.

# PART 3 - EXECUTION

# 3.1 INSTALLATION, GENERAL

- A. General: Anchor sheet metal flashing and trim and other components of the Work securely in place, with provisions for thermal and structural movement.
  - 1. Use fasteners, solder, protective coatings, separators, sealants, and other miscellaneous items as required to complete sheet metal flashing and trim system.
  - 2. Install sheet metal flashing and trim true to line, levels, and slopes. Provide uniform, neat seams with minimum exposure of solder, welds, and sealant.
  - 3. Install sheet metal flashing and trim to fit substrates and to result in weathertight performance. Verify shapes and dimensions of surfaces to be covered before fabricating sheet metal.
  - 4. Space individual cleats (where unable to use continuous cleats) not more than 12 inches (300 mm) apart. Attach each cleat with at least two fasteners. Bend tabs over fasteners.

- M.A. No. H1-22029-DA RMG - 22029
- 5. Install exposed sheet metal flashing and trim with limited oil canning, and free of buckling and tool marks.
- 6. Torch cutting of sheet metal flashing and trim is not permitted.
- B. Metal Protection: Where dissimilar metals contact each other, or where metal contacts pressuretreated wood or other corrosive substrates, protect against galvanic action or corrosion by painting contact surfaces with bituminous coating or by other permanent separation as recommended by sheet metal manufacturer or cited sheet metal standard.
  - 1. Coat concealed side of sheet metal flashing and trim with bituminous coating where flashing and trim contact cementitious construction.
- C. Expansion Provisions: Provide for thermal expansion of exposed flashing and trim. Space movement joints at maximum of 10 feet (3 m) with no joints within 24 inches (600 mm) of corner or intersection.
  - 1. Form expansion joints of intermeshing hooked flanges, not less than 1 inch (25 mm) deep, filled with sealant concealed within joints.
  - 2. Use lapped expansion joints only where indicated on Drawings.
- D. Fasteners: Use fastener sizes that penetrate wood blocking or sheathing not less than 1-1/4 inches (32 mm) for nails and not less than 3/4 inch (19 mm) for wood screws.
- E. Conceal fasteners and expansion provisions where possible in exposed work and locate to minimize possibility of leakage. Cover and seal fasteners and anchors as required for a tight installation.
- F. Seal joints as required for watertight construction. Prepare joints and apply sealants to comply with requirements in Section 079200 "Joint Sealants."

### 3.2 FLASHING INSTALLATION

- A. General: Install sheet metal flashing and trim to comply with cited sheet metal standards and provide weathertight installation. Provide concealed fasteners where possible, and set units true to line, levels, and slopes. Install work with laps, joints, and seams that are permanently watertight and weather resistant.
- B. Miscellaneous Flashing: Anchor to resist uplift and outward forces according to recommendations in cited sheet metal standard unless otherwise indicated.

#### 3.3 CLEANING AND PROTECTION

- A. Clean and neutralize flux materials. Clean off excess solder.
- B. Clean off excess sealants, leaving no residue as evidence of excess.
- C. Remove temporary protective coverings and/or strippable films as sheet metal flashing and trims as installation progresses unless otherwise indicated in manufacturer's written installation instructions.

END OF SECTION 07 62 00

This page intentionally left blank

#### SECTION 07 92 00 - JOINT SEALANTS

### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. This Section includes joint sealants for the following applications:
  - 1. Exterior joints in vertical surfaces and horizontal nontraffic surfaces.
  - 2. Interior joints in vertical surfaces and horizontal nontraffic surfaces
  - 3. Interior perimeter of partitions designated "Sound Wall" on drawings.
- B. See Section 32 13 73 "Concrete Paving Joint Sealants" for sealing joints in traffic areas such as pavements, walkways, curbing, etc.
- C. See Section 08 80 00 "Glazing" for sealants in conjunction with openings.
- D. See Section 07 21 00 "Thermal and Acoustical Insulation" for sealants in conjunction with acoustical treatment of sound walls noted on drawings.

# 1.2 PERFORMANCE REQUIREMENTS

- A. Provide elastomeric joint sealants that establish and maintain watertight and airtight continuous joint seals without staining or deteriorating joint substrates.
- B. Provide joint sealants for interior applications that establish and maintain airtight and water-resistant continuous joint seals without staining or deteriorating joint substrates.

### 1.3 SUBMITTALS

- A. Product Data: For each joint-sealant product indicated.
- B. Samples: For each type and color of joint sealant required, provide Samples with joint sealants in 1/2-inch- (13-mm-) wide joints formed between two 6-inch- (150-mm-) long strips of material matching the appearance of exposed surfaces adjacent to joint sealants.
- C. Preconstruction field test reports.
- D. Compatibility and adhesion test reports.
- E. Product certificates.

# 1.4 QUALITY ASSURANCE

- A. Preconstruction Compatibility and Adhesion Testing: Submit samples of materials that will contact or affect joint sealants to joint-sealant manufacturers for testing according to manufacturer's standard test method to determine whether priming and other specific joint preparation techniques are required to obtain rapid, optimum adhesion of joint sealants to joint substrates.
- B. Preconstruction Field-Adhesion Testing: Before installing elastomeric sealants, field test their adhesion to Project joint substrates according to the method in ASTM C 1193 that is appropriate for the types of Project joints.

JOINT SEALANTS 07 92 00 - 1

466

- M.A. No. H1-22029-DA RMG - 22029
- C. Mockups: Build mockups incorporating sealant joints, as follows, to verify selections made under sample submittals and to demonstrate aesthetic effects and set quality standards for materials and execution:
  - 1. Joints in mockups of assemblies specified in other Sections that are indicated to receive elastomeric joint sealants, which are specified by reference to this Section.

## 1.5 WARRANTY

- A. Special Installer's Warranty: Installer's standard form in which Installer agrees to repair or replace elastomeric joint sealants that do not comply with performance and other requirements specified in this Section within specified warranty period.
  - 1. Warranty Period: Two years from date of Recommendation of Acceptance.

### PART 2 - PRODUCTS

## 2.1 MANUFACTURERS

- A. Available Products: Subject to compliance with requirements, products that may be incorporated into the Work include, but are not limited to, products listed in other Part 2 articles.
- B. Products: Subject to compliance with requirements, provide one of the products listed in other Part 2 articles.

# 2.2 MATERIALS, GENERAL

- A. Compatibility: Provide joint sealants, backings, and other related materials that are compatible with one another and with joint substrates under conditions of service and application, as demonstrated by sealant manufacturer, based on testing and field experience.
- B. Colors of Exposed Joint Sealants: As selected by Architect from manufacturer's full range.

# 2.3 ELASTOMERIC JOINT SEALANTS

- A. Elastomeric Sealants: Comply with ASTM C 920 and other requirements indicated for each liquid-applied chemically curing sealant specified, including those referencing ASTM C 920 classifications for type, grade, class, and uses related to exposure and joint substrates.
- B. Stain-Test-Response Characteristics: Where elastomeric sealants are specified to be nonstaining to porous substrates, provide products that have undergone testing according to ASTM C 1248 and have not stained porous joint substrates indicated for Project.
- C. Suitability for Immersion in Liquids. Where elastomeric sealants are indicated for Use I for joints that will be continuously immersed in liquids, provide products that have undergone testing according to ASTM C 1247 and qualify for the length of exposure indicated by reference to ASTM C 920 for Class 1 or 2. Liquid used for testing sealants is deionized water, unless otherwise indicated.
- D. Suitability for Contact with Food: Where elastomeric sealants are indicated for joints that will come in repeated contact with food, provide products that comply with 21 CFR 177.2600.
- E. All sealants suitable for substrate types M, G, A, & O. Class 25, unless noted otherwise.

- 2.4 POLYSULFIDE JOINT SEALANTS: For use where could be in contact with fuels, solvents, or chemicals; exterior application such as trafficways and vehicle parking areas.
  - A. Multicomponent Non-sag Polysulfide Sealant:
    - 1. Products:
      - a. Pecora Corporation; Synthacalk GC-2+.
      - b. PolySpec Corp.; Thiokol 2P.
      - c. Sonneborn, Division of ChemRex Inc.; Sonolastic Polysulfide Sealant.
  - B. Multicomponent Nonsag Immersible Polysulfide Sealant:
    - Products:
      - a. Pecora Corporation; GC-2+
      - b. PolySpec Corp.; T-2235-M.
  - C. Multicomponent Pourable Polysulfide Sealant:
    - 1. Products:
      - a. Meadows, W. R., Inc.; Deck-O-Seal.
      - b. Pacific Polymers, Inc.; Elastoseal 227 Type I (Pourable).
  - D. Single-Component Nonsag Polysulfide Sealant:
    - 1. Products:
      - a. Pacific Polymers, Inc.; Elastoseal 230 Type I (Gun Grade).
      - b. Polymeric Systems Inc.; PSI-7000.
- 2.5 SILICONE JOINT SEALANTS: For use where high adhesive qualities and extensive movement is expected; interior or exterior application such as joints between sidewalks and building as well as building expansion joints.
  - A. Multicomponent Nonsag Neutral-Curing Silicone Sealant:
    - 1. Products:
      - a. Dow Corning Corporation; 756 H.P.
      - b. Tremco Spectrem 4-TS
    - 2. Class: 50.
  - B. Single-Component Pourable Neutral-Curing Silicone Sealant: For use where vehicle traffic is anticipated.
    - 1. Products:
      - a. Dow Corning Corporation; 890-SL.
      - b. Pecora Corporation; 300 Pavement Sealant (Self Leveling).
      - c. Dow Corning Corporation; SL Parking Structure Sealant.
    - 2. Class: 100/50.
- 2.6 URETHANE JOINT SEALANTS: For use where high adhesive qualities and moderate movement is expected; exterior applications such as around storefront and windows.
  - A. Multicomponent Nonsag Urethane Sealant:
    - 1. Products:
      - a. Pecora Corporation; Dynatrol II.
      - b. Tremco; Dymeric 511.

- c. Tremco; Vulkem 922.
- 2. Class: 50.
- B. Multicomponent Nonsag Urethane Sealant:
  - 1. Products:
    - a. Sika Corporation, Inc.; Sikaflex 2c NS TG.
    - b. Sonneborn, Division of ChemRex Inc.; NP 2.
    - c. Tremco; Vulkem 227.
- C. Multicomponent Nonsag Urethane Sealant:
  - 1. Products:
    - a. Bostik Findley; Chem-Calk 500.
    - b. Pacific Polymers, Inc.; Elasto-Thane 227 R Type II (Gun Grade).
    - c. Tremco; Dymeric.
  - 2. Additional Movement Capability: 40 percent movement in extension and 25 percent in compression for a total of 65 percent movement.
- D. Multicomponent Nonsag Urethane Sealant:
  - 1. Products:
    - a. Pacific Polymers, Inc.; Elasto-Thane 227 Type II (Gun Grade).
    - b. Pecora Corporation; Dynatred.
- E. Multicomponent Nonsag Immersible Urethane Sealant:
  - 1. Products:
    - a. Pecora Corporation; Dynatred.
    - b. Tremco; Vulkem 227.
    - c. Tremco; Vulkem 322 DS.
- F. Multicomponent Pourable Urethane Sealant:
  - 1. Products:
    - a. Bostik Findley; Chem-Calk 550.
    - b. Meadows, W. R., Inc.; POURTHANE.
    - c. Pecora Corporation; Urexpan NR-200.
    - d. Tremco; THC-901.
    - e. Pecora Corporation; Urexpan NR 300, Type M.
- G. Multicomponent Pourable Immersible Urethane Sealant:
  - 1. Products:
    - a. Pacific Polymers, Inc.; Elasto-Thane 227 R Type II (Self Leveling).
    - b. Tremco; Vulkem 245.
- H. Single-Component Nonsag Urethane Sealant:
  - 1. Products:
    - a. Sika Corporation, Inc.; Sikaflex 1a.
    - b. Sonneborn, Division of ChemRex Inc.; NP 1.
    - c. Tremco; Vulkem 116.
- I. Single-Component Nonsag Urethane Sealant:
  - 1. Products:
    - a. Bostik Findley; Chem-Calk 900.

- b. Sika Corporation, Inc.; Sikaflex 15LMg
- c. Tremco; DyMonic.
- 2. Class: 50.

## 2.7 ACOUSTICAL JOINT SEALANTS

- A. Acoustical Sealant: @ all walls designated "Sound Wall" on drawings.
  - 1. Products:
    - a. Owens Corning QuietZone, premium silicone acrylic sealant.
    - b. Tremco Acoustical Sealant synthetic rubber sealant.
    - c. Sheetrock Brand Acoustical Sealant, acrylic, latex-based.
- 2.8 SOLVENT-RELEASE JOINT SEALANTS: For use where high adhesive qualities and high weather resistance is required.
  - A. Acrylic-Based Solvent-Release Joint Sealant: Comply with ASTM C 1311 or FS TT-S-00230.
    - 1. Products:
      - a. Schnee-Moorehead, Inc.; Acryl-R Acrylic Sealant.
      - b. Tremco; Mono 555.
  - B. Butyl-Rubber-Based Solvent-Release Joint Sealant: Comply with ASTM C 1085.
    - 1. Products:
      - a. Bostik Findley; Bostik 300.
      - b. Pecora Corporation; BC-158.
      - c. Sonneborn, Division of ChemRex Inc.; Sonneborn Multi-Purpose Sealant.
      - d. Tremco; Tremco Butyl Sealant.
  - C. Pigmented Narrow-Joint Sealant: Manufacturer's standard, solvent-release-curing, pigmented, synthetic-rubber sealant complying with AAMA 803.3 and formulated for sealing joints 3/16 inch (5 mm) or smaller in width.
    - 1. Products:
      - a. Fuller, H. B. Company; SC-0289.
      - b. Schnee-Morehead, Inc.; SM 5504 Acryl-R Narrow Joint Sealant.
- 2.9 LATEX JOINT SEALANTS: For use where sealant is to be painted and minimum movement is expected; interior application only.
  - A. Latex Sealant: Comply with ASTM C 834, Type O P, Grade NF.
  - B. Products:
    - 1. Bostik Findley; Chem-Calk 600.
    - 2. Pecora Corporation; AC-20+.
    - 3. Sonneborn, Division of ChemRex Inc.; Sonolac.
    - 4. Tremco; Tremflex 834.

## 2.10 JOINT-SEALANT BACKING

A. General: Provide sealant backings of material and type that are nonstaining; are compatible with joint substrates, sealants, primers, and other joint fillers; and are approved for applications indicated by sealant manufacturer based on field experience and laboratory testing.

JOINT SEALANTS 07 92 00 - 5

470

- B. Cylindrical Sealant Backings: ASTM C 1330, Type B (bicellular material with a surface skin), and of size and density to control sealant depth and otherwise contribute to producing optimum sealant performance:
- C. Elastomeric Tubing Sealant Backings: Neoprene, butyl, EPDM, or silicone tubing complying with ASTM D 1056, nonabsorbent to water and gas, and capable of remaining resilient at temperatures down to minus 26 deg F (minus 32 deg C). Provide products with low compression set and of size and shape to provide a secondary seal, to control sealant depth, and to otherwise contribute to optimum sealant performance.
- D. Bond-Breaker Tape: Polyethylene tape or other plastic tape recommended by sealant manufacturer for preventing sealant from adhering to rigid, inflexible joint-filler materials or joint surfaces at back of joint where such adhesion would result in sealant failure. Provide self-adhesive tape where applicable.

## 2.11 MISCELLANEOUS MATERIALS

- A. Primer: Material recommended by joint-sealant manufacturer where required for adhesion of sealant to joint substrates indicated, as determined from preconstruction joint-sealant-substrate tests and field tests.
- B. Cleaners for Nonporous Surfaces: Chemical cleaners acceptable to manufacturers of sealants and sealant backing materials, free of oily residues or other substances capable of staining or harming joint substrates and adjacent nonporous surfaces in any way, and formulated to promote optimum adhesion of sealants to joint substrates.
- C. Masking Tape: Non-staining, nonabsorbent material compatible with joint sealants and surfaces adjacent to joints.

### **PART 3 - EXECUTION**

### 3.1 PREPARATION

- A. Surface Cleaning of Joints: Clean out joints immediately before installing joint sealants.
  - 1. Remove all foreign material from joint substrates that could interfere with adhesion of joint sealant.
    - a. Clean porous joint substrate surfaces by brushing, grinding, blast cleaning, mechanical abrading, or a combination of these methods to produce a clean, dry, sound substrate capable of developing optimum bond with joint sealants. Remove loose particles remaining after cleaning operations above by vacuuming or blowing out joints with oil-free compressed air.
  - 2. Remove laitance and form-release agents from concrete.
    - a. Clean nonporous surfaces with chemical cleaners or other means that do not stain, harm substrates, or leave residues capable of interfering with adhesion of joint sealants.
- B. Joint Priming: Prime joint substrates, where recommended in writing by joint-sealant manufacturer, based on preconstruction joint-sealant-substrate tests or prior experience. Apply primer to comply with joint-sealant manufacturer's written instructions. Confine primers to areas of joint-sealant bond; do not allow spillage or migration onto adjoining surfaces.

C. Masking Tape: Use masking tape where required to prevent contact of sealant with adjoining surfaces that otherwise would be permanently stained or damaged by such contact or by cleaning methods required to remove sealant smears. Remove tape immediately after tooling without disturbing joint seal.

### 3.2 INSTALLATION

- A. Sealant Installation Standard: Comply with recommendations in ASTM C 1193 for use of joint sealants as applicable to materials, applications, and conditions indicated.
- B. Install sealant backings of type indicated to support sealants during application and at position required to produce cross-sectional shapes and depths of installed sealants relative to joint widths that allow optimum sealant movement capability.
  - 1. Do not leave gaps between ends of sealant backings.
  - 2. Do not stretch, twist, puncture, or tear sealant backings.
  - 3. Remove absorbent sealant backings that have become wet before sealant application and replace them with dry materials.
- C. Install bond-breaker tape behind sealants where sealant backings are not used between sealants and backs of joints.
- D. Install sealants using proven techniques that comply with the following and at the same time backings are installed:
  - 1. Place sealants so they directly contact and fully wet joint substrates.
  - 2. Completely fill recesses in each joint configuration.
  - 3. Produce uniform, cross-sectional shapes and depths relative to joint widths that allow optimum sealant movement capability.
- E. Tooling of Non-sag Sealants: Immediately after sealant application and before skinning or curing begins, tool sealants according to requirements specified below to form smooth, uniform beads of configuration indicated; to eliminate air pockets; and to ensure contact and adhesion of sealant with sides of joint.
  - 1. Remove excess sealant from surfaces adjacent to joints.
  - 2. Use tooling agents that are approved in writing by sealant manufacturer and that do not discolor sealants or adjacent surfaces.
  - 3. Provide concave joint configuration per Figure 5A in ASTM C 1193, unless otherwise indicated.
- F. Installation of Preformed Silicone-Sealant System: Comply with manufacturer's written instructions.
- G. Installation of Preformed Foam Sealants: Install each length of sealant immediately after removing protective wrapping, taking care not to pull or stretch material, producing seal continuity at ends, turns, and intersections of joints. For applications at low ambient temperatures where expansion of sealant requires acceleration to produce seal, apply heat to sealant in compliance with sealant manufacturer's written instructions.
- H. Installation of acoustical sealants:
  - 1. Install continuous bead along edge of drywall track, each side.
- I. Clean off excess sealant or sealant smears adjacent to joints as the Work progresses by methods and with cleaning materials approved in writing by manufacturers of joint sealants and of products in which joints occur.

END OF SECTION 07 92 00

JOINT SEALANTS 07 92 00 - 8

473

#### SECTION 08 11 13 - HOLLOW METAL DOORS AND FRAMES

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

### 1.2 SUMMARY

### A. Section Includes:

- 1. Interior and exterior standard and custom hollow metal doors and frames.
- 2. Steel sidelight, borrowed lite and transom frames.
- 3. Lite frames for glazing to be installed in hollow metal doors.

### B. Related Sections:

- 1. Section 04 21 13 "Unit Masonry" for embedding anchors into masonry construction for hollow metal work.
- 2. Section 08 14 16 "Flush Wood Doors" for doors to be fitted in hollow metal frames.
- 3. Section 08 80 00 "Glazing" for glass view panels in hollow metal doors.
- 4. Section 08 71 00 "Door Hardware".
- 5. Sections 09 90 00 "Painting" for field painting of hollow metal doors and frames.
- 6. Division 26 "Electrical" Sections for electrical connections including conduit and wiring for door controls and operators installed on frames with factory installed electrical knock out boxes.
- 7. Division 28 Section "Access Control" for access control devices installed at door openings and provided as part of a security access control system.
- C. Codes and References: Comply with the version year adopted by the Authority Having Jurisdiction.
  - 1. ANSI/SDI A250.8 Recommended Specifications for Standard Steel Doors and Frames.
  - 2. ANSI/SDI A250.4 Test Procedure and Acceptance Criteria for Physical Endurance for Steel Doors, Frames, Frames Anchors and Hardware Reinforcing.
  - 3. ANSI/SDI A250.6 Recommended Practice for Hardware Reinforcing on Standard Steel Doors and Frames.
  - 4. ANSI/SDI A250.10 Test Procedure and Acceptance Criteria for Prime Painted Steel Surfaces for Steel Doors and Frames.
  - 5. ANSI/SDI A250.11 Recommended Erection Instructions for Steel Frames.
  - 6. ASTM A1008 Standard Specification for Steel Sheet, Cold-Rolled, Carbon, Structural, High-Strength Low-Alloy and High-Strength Low-Alloy with Improved Formability.
  - 7. ASTM A653 Standard Specification for Steel Sheet, Zinc-Coated (Galvanized) or Zinc-Iron Alloy-Coated (Galvannealed) by the Hot-Dip Process.
  - 8. ASTM A924 Standard Specification for General Requirements for Steel Sheet, Metallic-Coated by the Hot-Dip Process.

- M.A. No. H1-22029-DA RMG - 22029
- 9. ASTM C 1363 Standard Test Method for Thermal Performance of Building Assemblies by Means of a Hot Box Apparatus.
- 10. ANSI/BHMA A156.115 Hardware Preparation in Steel Doors and Frames.
- 11. ANSI/SDI 122 Installation and Troubleshooting Guide for Standard Steel Doors and Frames
- 12. ANSI/NFPA 80 Standard for Fire Doors and Fire Windows; National Fire Protection Association.
- 13. ANSI/NFPA 105: Standard for the Installation of Smoke Door Assemblies.
- 14. NFPA 252 Standard Methods of Fire Tests of Door Assemblies; National Fire Protection Association.
- 15. UL 10C Positive Pressure Fire Tests of Door Assemblies.
- 16. UL 1784 Standard for Air Leakage Tests of Door Assemblies.

## 1.3 SUBMITTALS

- A. Product Data: For each type of product indicated. Include construction details, material descriptions, core descriptions, hardware reinforcements, profiles, anchors, fire-resistance rating, and finishes.
- B. Door hardware supplier is to furnish templates, template reference number and/or physical hardware to the steel door and frame supplier in order to prepare the doors and frames to receive the finish hardware items.
- C. Shop Drawings: Include the following:
  - 1. Elevations of each door design.
  - 2. Details of doors, including vertical and horizontal edge details and metal thicknesses.
  - 3. Frame details for each frame type, including dimensioned profiles and metal thicknesses.
  - 4. Locations of reinforcement and preparations for hardware.
  - 5. Details of anchorages, joints, field splices, and connections.
  - 6. Details of accessories.
  - 7. Details of moldings, removable stops, and glazing.
  - 8. Details of conduit and preparations for power, signal, and control systems.

## D. Samples for Verification:

1. Samples are only required if requested by the architect and for manufacturers that are not current members of the Steel Door Institute.

### 1.4 OUALITY ASSURANCE

- A. Source Limitations: Obtain hollow metal doors and frames through one source from a single manufacturer.
- B. Quality Standard: In addition to requirements specified, comply with ANSI/SDI A250.8, latest edition, "Recommended Specifications for Standard Steel Doors and Frames".
- C. Rated Door Assemblies where indicated: Assemblies complying with NFPA 80 that are listed and labeled by a qualified testing agency, for fire-protection ratings indicated, based on testing at positive pressure according to NFPA 252 (neutral pressure at 40" above sill) or UL 10C.
  - 1. Smoke Control Door Assemblies: Comply with NFPA 105.
    - a. Smoke "S" Label: Doors to bear "S" label, and include smoke and draft control gasketing applied to frame and on meeting stiles of pair doors.

- D. Fire-Rated, Borrowed-Light Frame Assemblies: Assemblies complying with NFPA 80 that are listed and labeled, by a testing and inspecting agency acceptable to authorities having jurisdiction, for fire-protection ratings indicated, based on testing according to NFPA 257. Provide labeled glazing material.
- E. Hurricane Resistant Exterior Openings: Provide exterior hollow metal doors and frames as complete and tested assemblies, or component assemblies, including approved hardware specified under Section 087100 "Door Hardware", to meet the wind loads, design pressures, debris impact resistance, applicable to the Project and as required by IBC Wind Zone maps.
  - 1. Test units according to ASTM E330, ASTM E1886, ASTM E1996 standards, certified by a qualified independent third party testing agency acceptable to authority having jurisdiction, and bearing a third party certification agency permanent label indicting windstorm approved product.
- F. Pre-Submittal Conference: Conduct conference in compliance with requirements in Division 01 Section "Project Meetings" with attendance by representatives of Supplier, Installer, and Contractor to review proper methods and procedures for installing hollow metal doors and frames and to verify installation of electrical knockout boxes and conduit at frames with electrified or access control hardware.

### 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Deliver hollow metal work palletized, wrapped, or crated to provide protection during transit and Project site storage. Do not use non-vented plastic.
- B. Deliver welded frames with two removable spreader bars across bottom of frames, tack welded to jambs and mullions.
- C. Store hollow metal work under cover at Project site. Place in stacks of five units maximum in a vertical position with heads up, spaced by blocking, on minimum 4-inch high wood blocking. Do not store in a manner that traps excess humidity.
  - 1. Provide minimum 1/4-inch space between each stacked door to permit air circulation. Door and frames to be stacked in a vertical upright position.

# 1.6 PROJECT CONDITIONS

A. Field Measurements: Verify actual dimensions of openings by field measurements before fabrication.

### 1.7 COORDINATION

A. Coordinate installation of anchorages for hollow metal frames. Furnish setting drawings, templates, and directions for installing anchorages, including sleeves, concrete inserts, anchor bolts, and items with integral anchors. Deliver such items to Project site in time for installation.

## 1.8 WARRANTY

- A. Special Warranty: Manufacturer's standard form in which manufacturer agrees to repair or replace doors that fail in materials or workmanship within specified warranty period.
- B. Warranty includes installation and finishing that may be required due to repair or replacement of defective doors.

## PART 2 - PRODUCTS

### 2.1 MANUFACTURERS

- A. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
  - 1. CECO Door Products.
  - 2. Curries Company.
  - 3. Steelcraft.

### 2.2 MATERIALS

- A. Cold-Rolled Steel Sheet (Interior): ASTM A 1008/A 1008M, Commercial Steel (CS), Type B; suitable for exposed applications.
- B. Metallic-Coated Steel Sheet (Exterior): ASTM A 653/A 653M, Commercial Steel (CS), Type B; with minimum G60 (Z180) or A60 (ZF180) metallic coating.
- C. Frame Anchors: ASTM A 653/A 653M, Commercial Steel (CS), Commercial Steel (CS), Type B; with minimum G60 (Z180) or A60 (ZF180) metallic coating.

### 2.3 STANDARD HOLLOW METAL DOORS

- A. General: Provide 1-3/4 inch doors of design indicated, not less than thickness indicated; fabricated with smooth surfaces, without visible joints or seams on exposed faces unless otherwise indicated. Comply with ANSI/SDI A250.8.
- B. Exterior Doors: Face sheets fabricated of commercial quality hot-dipped zinc coated steel that complies with ASTM A 653/A 653M, Coating Designation A60. Provide doors complying with requirements indicated below by referencing ANSI/SDI A250.8 for level and model and ANSI/SDI A250.4 for physical performance level:
  - 1. Design: Flush panel.
  - 2. Core Construction: Manufacturer's standard polystyrene. Where indicated, provide doors fabricated as thermal-rated assemblies with a minimum R-value of 2.8 or better.
  - 3. Level/Model: Level 2 and Physical Performance Level B (Heavy Duty), Minimum 16 gauge (0.053-inch -1.3-mm) thick steel, Model 2.
  - 4. Top and Bottom Edges: Reinforce tops and bottoms of doors with a continuous steel channel not less than 16 gauge, (0.0625) extending the full width of the door and welded to the face sheet. Doors with an inverted top channel to include a steel closure channel, screw attached, with the web of the channel flush with the face sheets of the door. Plastic or composite channel fillers are not acceptable.
  - 5. Hinge Reinforcement: Minimum 7 gauge (3/16") plate 1-1/4" x 9" or minimum 14 gauge continuous channel with pierced holes, drilled and tapped.
  - 6. Hardware Reinforcements: Fabricate according to ANSI/SDI A250.6 with reinforcing plates from same material as door face sheets.
- C. Interior Doors: Face sheets fabricated of commercial quality cold rolled steel that complies with ASTM A 1008/A 1008M. Provide doors complying with requirements indicated below by referencing ANSI/SDI A250.8 for level and model and ANSI/SDI A250.4 for physical performance level:

- 1. Design: Flush panel.
- 2. Core Construction: Manufacturer's standard one-piece polystyrene core, securely bonded to both faces.
  - a. Fire Door Core: As required to provide fire-protection and temperature-rise ratings indicated
- 3. Level/Model: Level 2 and Physical Performance Level B (Heavy Duty), Minimum 16 gauge (0.053-inch -1.3-mm) thick steel, Model 2.
- 4. Top and Bottom Edges: Reinforce tops and bottoms of doors with a continuous steel channel not less than 16 gauge, extending the full width of the door and welded to the face sheet.
- 5. Hinge Reinforcement: Minimum 7 gauge (3/16") plate 1-1/4" x 9" or minimum 14 gauge continuous channel with pierced holes, drilled and tapped.
- 6. Hardware Reinforcements: Fabricate according to ANSI/SDI A250.6 with reinforcing plates from same material as door face sheets.

### 2.4 STANDARD HOLLOW METAL FRAMES

- A. General: Comply with ANSI/SDI A250.8 and with details indicated for type and profile.
- B. Exterior Frames: Fabricated of hot-dipped zinc coated steel that complies with ASTM A 653/A 653M, Coating Designation A60.
  - 1. Fabricate frames with mitered or coped corners.
  - 2. Fabricate frames, with "closed and tight" miter seams continuously welded on face, finished smooth with no visible seam unless otherwise indicated.
  - 3. Frames for Steel Doors: Minimum 14 gauge (0.067-inch -1.7-mm) thick steel sheet.
- C. Interior Frames: Fabricated from cold-rolled steel sheet that complies with ASTM A 1008/A 1008M.
  - 1. Fabricate frames with mitered or coped corners.
  - 2. Fabricate frames, with "closed and tight" miter seams continuously welded on face, finished smooth with no visible seam unless otherwise indicated.
  - 3. Frames for Steel Doors, Wood Doors and Borrowed Lites for openings up to and including g 48": Minimum 16 gauge (0.053-inch -1.3-mm) thick steel sheet.
  - 4. Frames for openings 48 inches and wider in width: Minimum 14 gauge (0.067-inch -1.7-mm) thick steel sheet.
- D. Fire rated frames: Fabricate frames in accordance with NFPA 80, listed and labeled by a qualified testing agency, for fire-protection ratings indicated.
- E. Hardware Reinforcement: Fabricate according to ANSI/SDI A250.6 Table 4 with reinforcement plates from same material as frames.

## 2.5 FRAME ANCHORS

A. Jamb Anchors:

- M.A. No. H1-22029-DA RMG - 22029
- 1. Masonry Type: Adjustable strap-and-stirrup or T-shaped anchors to suit frame size, formed from A60 metallic coated material, not less than 0.042 inch thick, with corrugated or perforated straps not less than 2 inches wide by 10 inches long; or wire anchors not less than 0.177 inch thick.
- 2. Stud Wall Type: Designed to engage stud and not less than 0.042 inch thick.
- 3. Windstorm Opening Anchors: Types as tested and required for indicated wall types to meet specified wind load design criteria.
- B. Floor Anchors: Floor anchors to be provided at each jamb, formed from A60 metallic coated material, not less than 0.042 inches thick.
- C. Mortar Guards: Formed from same material as frames, not less than 0.016 inches thick.

### 2.6 LITE OPENINGS AND GLAZING

- A. Stops and Moldings: Provide stops and moldings around glazed lites where indicated. Form corners of stops and moldings with butted or mitered hairline joints at fabricator's shop. Fixed and removable stops to allow multiple glazed lites each to be removed independently. Coordinate frame rabbet widths between fixed and removable stops with the type of glazing and installation indicated.
- B. Preformed Metal Frames for Lite Openings: Manufacturer's standard frame formed of 0.048-inch-thick, cold rolled steel sheet; with baked enamel or powder coated finish; and approved for use in doors of fire protection rating indicated. Match pre-finished door paint color where applicable.
- C. Glazing: Comply with requirements in Division 08 Section "Glazing" and with the hollow metal door manufacturer's written instructions.
- D. Moldings for Glazed Lites in Doors and Loose Stops for Glazed Lites in Frames: Minimum 20 gauge thick, fabricated from same material as door face sheet in which they are installed.
- E. Fixed Frame Moldings: Formed integral with hollow metal frames, a minimum of 5/8 inch (16 mm) high unless otherwise indicated. Provide fixed frame moldings and stops on outside of exterior and on secure side of interior doors and frames

### 2.7 ACCESSORIES

- A. Mullions and Transom Bars: Join to adjacent members by welding.
- B. Grout Guards: Formed from same material as frames, not less than 0.016 inches thick.

# 2.8 FABRICATION

- A. Fabricate hollow metal work to be rigid and free of defects, warp, or buckle. Accurately form metal to required sizes and profiles, with minimum radius for thickness of metal. Where practical, fit and assemble units in manufacturer's plant. When shipping limitations so dictate, frames for large openings are to be fabricated in sections for splicing or splining in the field by others.
- B. Tolerances: Fabricate hollow metal work to tolerances indicated in ANSI/SDI A250.8.
- C. Hollow Metal Doors:

- 1. Exterior Doors: Provide optional weep-hole openings in bottom of exterior doors to permit moisture to escape where specified.
- 2. Glazed Lites: Factory cut openings in doors with applied trim or kits to fit. Factory install glazing where indicated.
- 3. Astragals: Provide overlapping astragals as noted in door hardware sets in Division 08 Section "Door Hardware" on one leaf of pairs of doors where required by NFPA 80 for fire-performance rating or where indicated. Extend minimum 3/4 inch beyond edge of door on which astragal is mounted.

## D. Hollow Metal Frames:

- 1. Shipping Limitations: Where frames are fabricated in sections due to shipping or handling limitations, provide alignment plates or angles at each joint, fabricated of same thickness metal as frames.
- 2. Welded Frames: Weld flush face joints continuously; grind, fill, dress, and make smooth, flush, and invisible.
  - a. Welded frames are to be provided with two steel spreaders temporarily attached to the bottom of both jambs to serve as a brace during shipping and handling. Spreader bars are for bracing only and are not to be used to size the frame opening.
- 3. Sidelight and Transom Bar Frames: Provide closed tubular members with no visible face seams or joints, fabricated from same material as door frame. Fasten members at crossings and to jambs by butt welding.
- 4. High Frequency Hinge Reinforcement: Provide high frequency hinge reinforcements at all corridor door openings and other openings 48-inches and wider with mortise butt type hinges at top hinge locations.
- 5. Provide countersunk, flat- or oval-head exposed screws and bolts for exposed fasteners unless otherwise indicated for removable stops, provide security screws at exterior locations.
- 6. Mortar Guards: Provide guard boxes at back of hardware mortises in frames at all hinges and strike preps regardless of grouting requirements.
- 7. Floor Anchors: Weld anchors to bottom of jambs and mullions with at least four spot welds per anchor.
- 8. Jamb Anchors: Provide number and spacing of anchors as follows:
  - a. Masonry Type: Locate anchors not more than 18 inches from top and bottom of frame. Space anchors not more than 32 inches o.c. and as follows:
    - 1) Two anchors per jamb up to 60 inches high.
    - 2) Three anchors per jamb from 60 to 90 inches high.
    - 3) Four anchors per jamb from 90 to 120 inches high.
    - 4) Four anchors per jamb plus 1 additional anchor per jamb for each 24 inches or fraction thereof above 120 inches high.
  - b. Stud Wall Type: Locate anchors not more than 18 inches from top and bottom of frame. Space anchors not more than 32 inches o.c. and as follows:
    - 1) Three anchors per jamb up to 60 inches high.
    - 2) Four anchors per jamb from 60 to 90 inches high.
    - 3) Five anchors per jamb from 90 to 96 inches high.

- 4) Five anchors per jamb plus 1 additional anchor per jamb for each 24 inches or fraction thereof above 96 inches high.
- 5) Two anchors per head for frames above 42 inches wide and mounted in metal stud partitions.
- 9. Door Silencers: Except on weatherstripped or gasketed doors, drill stops to receive door silencers. Silencers to be supplied by frame manufacturer regardless if specified in Division 08 Section "Door Hardware".
- E. Hardware Preparation: Factory prepare hollow metal work to receive template mortised hardware; include cutouts, reinforcement, mortising, drilling, and tapping according to the Door Hardware Schedule and templates furnished as specified in Division 08 Section "Door Hardware."
  - 1. Locate hardware as indicated, or if not indicated, according to ANSI/SDI A250.8.
  - 2. Reinforce doors and frames to receive non-template, mortised and surface mounted door hardware.
  - 3. Comply with applicable requirements in ANSI/SDI A250.6 and ANSI/DHI A115 Series specifications for preparation of hollow metal work for hardware.
  - 4. Coordinate locations of conduit and wiring boxes for electrical connections with Division 26 Sections.

### 2.9 STEEL FINISHES

- A. Prime Finishes: Doors and frames to be cleaned, and chemically treated to insure maximum finish paint adhesion. Surfaces of the door and frame exposed to view to receive a factory applied coat of rust inhibiting shop primer.
  - 1. Shop Primer: Manufacturer's standard, fast-curing, lead and chromate free primer complying with ANSI/SDI A250.10 acceptance criteria; recommended by primer manufacturer for substrate; and compatible with substrate and field-applied coatings.

## PART 3 - EXECUTION

### 3.1 EXAMINATION

- A. Examine substrates, areas, and conditions, with Installer present, for compliance with requirements for installation tolerances and other conditions affecting performance of the Work.
- B. General Contractor to verify the accuracy of dimensions given to the steel door and frame manufacturer for existing openings or existing frames (strike height, hinge spacing, hinge back set, etc.).
- C. Proceed with installation only after unsatisfactory conditions have been corrected.

# 3.2 PREPARATION

- A. Remove welded-in shipping spreaders installed at factory. Restore exposed finish by grinding, filling, and dressing, as required to make repaired area smooth, flush, and invisible on exposed faces.
- B. Prior to installation, adjust and securely brace welded hollow metal frames for square, level, twist, and plumb conditions.

- M.A. No. H1-22029-DA RMG - 22029
- C. Tolerances shall comply with SDI-117 "Manufacturing Tolerances Standard Steel Doors and Frames."
- D. Drill and tap doors and frames to receive non-template, mortised, and surface-mounted door hardware.

### 3.3 INSTALLATION

- A. General: Install hollow metal work plumb, rigid, properly aligned, and securely fastened in place; comply with Drawings and manufacturer's written instructions.
- B. Hollow Metal Frames: Install hollow metal frames of size and profile indicated. Comply with ANSI/SDI A250.11 and NFPA 80 at fire rated openings.
  - 1. Set frames accurately in position, plumbed, leveled aligned, and braced securely until permanent anchors are set. After wall construction is complete and frames properly set and secured, remove temporary braces, leaving surfaces smooth and undamaged. Shim as necessary to comply with installation tolerances.
  - 2. Floor Anchors: Provide floor anchors for each jamb and mullion that extends to floor, and secure with post-installed expansion anchors.
  - 3. Masonry Walls: Coordinate installation of frames to allow for solidly filling space between frames and masonry with mortar.
  - 4. Grout Requirements: Do not grout head of frames unless reinforcing has been installed in head of frame. Do not grout vertical or horizontal closed mullion members.
- C. Hollow Metal Doors: Fit hollow metal doors accurately in frames, within clearances specified below. Shim as necessary.
  - 1. Non-Fire-Rated Standard Steel Doors:
    - a. Jambs and Head: 1/8 inch plus or minus 1/16 inch.
    - b. Between Edges of Pairs of Doors: 1/8 inch plus or minus 1/16 inch.
    - c. Between Bottom of Door and Top of Threshold: Maximum 3/8 inch.
    - d. Between Bottom of Door and Top of Finish Floor (No Threshold): Maximum 3/4 inch.
  - 2. Fire-Rated Doors: Install doors with clearances according to NFPA 80.
- D. Field Glazing: Comply with installation requirements in Division 08 Section "Glazing" and with hollow metal manufacturer's written instructions.

### 3.4 ADJUSTING AND CLEANING

- A. Final Adjustments: Check and readjust operating hardware items immediately before final inspection. Leave work in complete and proper operating condition. Remove and replace defective work, including hollow metal work that is warped, bowed, dented/dimpled or otherwise unacceptable.
- B. Remove grout and other bonding material from hollow metal work immediately after installation.

C. Prime-Coat and Painted Finish Touchup: Immediately after erection, sand smooth rusted and/or damaged areas of prime coat, or painted finishes, and apply touchup of compatible air drying, rust-inhibitive primer, zinc rich primer (exterior and galvanized openings) or finish paint.

END OF SECTION 08 11 13

## 08 14 16 FLUSH WOOD DOORS

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 1 Specification Sections, apply to this Section.

## 1.2 SUMMARY

- A. This Section includes the following:
  - 1. Solid-core doors with wood-veneer faces.
  - 2. Factory finishing flush wood doors.
  - 3. Factory fitting flush wood doors to frames and factory machining for hardware.
  - 4. Lite frames for glazing installed in wood doors.
- B. Related Sections include the following:
  - 1. Section 06 10 00 "Finish Carpentry"
  - 2. Section 08 11 13 "Hollow Metal Doors and Frames"
  - 3. Section 08 12 16 "Interior Aluminum Frames"
  - 4. Section 08 80 00 "Glazing" for glass view panels in metal frames in flush wood doors.
  - 5. Section 08 71 00 "Finish Hardware" (Including Access Control Hardware)
- C. Standards and References: Comply with the version year adopted by the Authority Having Jurisdiction.
  - 1. ANSI A117.1 Accessible and Usable Buildings and Facilities.
  - 2. ANSI A208.1 Wood Particleboard.
  - 3. Intertek Testing Service (ITS Warnock Hersey) Certification Listings for Fire Doors.
  - 4. NFPA 80 Standard for Fire Doors and Fire Windows; National Fire Protection Association.
  - 5. NFPA 252 Standard Methods of Fire Tests of Door Assemblies; National Fire Protection Association.
  - 6. UL 10C Positive Pressure Fire Tests of Door Assemblies; UL 1784 Standard for Air Leakage Tests of Door Assemblies.
  - 7. Window and Door Manufacturers Association WDMA I.S.1-A Architectural Wood Flush Doors.

### 1.3 SUBMITTALS

- A. Product Data: For each type of door. Include details of core and edge construction, and trim for openings. Include factory-finishing specifications.
- B. Shop Drawings: Indicate location, size, and hand of each door; elevation of each kind of door; construction details not covered in Product Data; location and extent of hardware blocking; and other pertinent data.
  - 1. Indicate location, size, and hand of each door.
  - 2. Indicate dimensions and locations of mortises and holes for hardware.
  - 3. Indicate dimensions and locations of cutouts.

- M.A. No. H1-22029-DA RMG - 22029
- 4. Indicate location and extent of hardware blocking.
- 5. Indicate construction details not covered in Product Data.
- 6. Indicate doors to be factory finished and finish requirements.
- 7. Indicate requirements for veneer matching.
- 8. Indicate fire ratings for fire doors.
- C. Samples for Initial Selection: Color charts consisting of actual materials (photographs not acceptable) in small sections for the following:
  - 1. Faces of Factory-Finished Doors: Show the full range of colors available for stained finishes.

## D. Samples for Verification:

- 1. Factory finishes applied to actual door face materials, approximately 8 by 10 inches, for each material and finish. For each wood species and transparent finish, provide set of three samples showing typical range of color and grain to be expected in the finished work.
- 2. Corner sections of doors, approximately 8 by 10 inches, with door faces and edgings representing typical range of color and grain for each species of veneer and solid lumber required. Finish sample with same materials proposed for factory-finished doors Frames for light openings, 6 inches long, for each material, type, and finish required.

# 1.4 QUALITY ASSURANCE

- A. Source Limitations: Obtain flush wood doors through one source from a single manufacturer.
- B. Quality Standard: Comply with AWI's "Architectural Woodwork Quality Standards Illustrated."
  - 1. Provide AWI Quality Certification Labels or an AWI letter of licensing for Project indicating that doors comply with requirements of grades specified.
- C. Fire Rated Wood Doors: Doors complying with NFPA 80 that are listed and labeled by a testing and inspecting agency acceptable to authorities having jurisdiction, for fire ratings indicated, based on testing at positive pressure according to NFPA 252 (neutral pressure at 40" above sill) or UL10C.
  - 1. Smoke Control Door Assemblies: Comply with NFPA 105.
    - a. Smoke "S" Label: Doors to bear "S" label, and include smoke and draft control gasketing applied to frame and on meeting stiles of pair doors.

# 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Comply with requirements of referenced standard and manufacturer's written instructions.
- B. Package doors individually in vented plastic bags or cardboard cartons.
- C. Mark each door on top and bottom rail with opening number used on Shop Drawings.

# 1.6 PROJECT CONDITIONS

A. Environmental Limitations: Do not deliver or install doors until spaces are enclosed and weather tight, wet work in spaces is complete and dry, and HVAC system is operating and maintaining ambient temperature and humidity conditions at occupancy levels during the remainder of the construction period.

### 1.7 WARRANTY

- A. Special Warranty: Manufacturer's standard form, signed by manufacturer, Installer, and Contractor, in which manufacturer agrees to repair or replace doors that are defective in materials or workmanship.
  - 1. Failures include, but are not limited to, the following:
    - a. Warping (bow, cup, or twist) more than 1/4 inch in (up to) a 42-by-84-inch section.
    - b. Telegraphing of core construction in wood face veneers exceeding 0.01 inch in a 3-inch span.
  - 2. Warranty shall also include installation and finishing that may be required due to repair or replacement of defective doors.
  - 3. Warranty shall be in effect during the following period of time from date of Recommendation of Acceptance:
    - a. Solid-Core Interior Doors: Life of installation.

### PART 2 - PRODUCTS

### 2.1 MANUFACTURERS

- A. Available Manufacturers: Subject to compliance with requirements, manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
  - 1. Algoma Hardwoods: Architectural Series.
  - 2. Eggers Industries: Premium Series.
  - 3. Graham: GPD Series.
  - 4. Marshfield: Signature Series.
  - 5. VT Industries: Artistry Series.
  - 6. Warnock Hersey Mark

# 2.2 DOOR CONSTRUCTION, GENERAL

- A. Doors for Transparent Finish:
  - 1. Grade: Premium, with Grade AA faces.
  - 2. Species and Cut: Red oak, plain sliced.
  - 3. Match between Veneer Leaves: Book match.
  - 4. Assembly of Veneer Leaves on Door Faces: Center balance match.
  - 5. Pair and Set Match: Provide for doors hung in same opening or separated only by mullions.
  - 6. Room Match: Match door faces within each separate room or area of building. Corridor door faces do not need to match where they are separated by 20 feet or more.
  - 7. Room Match: Provide door faces of compatible color and grain within each separate room or area of building.
  - 8. Transom Match: Continuous match.
  - 9. Stiles: Same species as faces.

### 2.3 SOLID-CORE DOORS

- A. WDMA I.S.1-A Performance Grade: Extra Heavy Duty; Aesthetic Grade: Premium.
- B. Particleboard Cores: Comply with the following requirements:
  - 1. Particleboard: Wood fiber based materials complying with ANSI A208.1 Particleboard standard. Grade LD-2.
  - 2. Adhesive: Fully bonded construction using Polyurethane (PUR) glue.
  - 3. Blocking: As indicated under article "Blocking".
- C. Interior Veneer-Faced Doors: where exit devices are scheduled.
  - 1. Core: Structural composite lumber.
  - 2. Construction: Five plies with stiles and rails bonded to core, then entire unit abrasive planed and/or sanded before veneering.

### D. Fire-Rated Doors:

- 1. Construction: Construction and core specified above for type of face indicated or manufacturer's standard mineral-core construction as needed to provide fire rating indicated.
- 2. Blocking: For mineral-core doors, provide composite blocking with improved screw-holding capability approved for use in doors of fire ratings indicated as needed to eliminate through-bolting hardware.
- 3. Edge Construction: At hinge stiles, provide manufacturer's standard laminated-edge construction with improved screw-holding capability and split resistance and with outer stile matching face veneer. Comply with requirements of WDMA section P-1, Performance Standards for Architectural Wood Flush Doors
- 4. Pairs: Provide fire-rated pairs with fire-retardant stiles matching face veneer that are labeled and listed for kinds of applications indicated without formed-steel edges and astragals.

### 2.4 BLOCKING

- A. Non-Fire-Rated Doors:
  - 1. Provide blocking as indicated below:
    - a. HB3: 5 inch top and bottom rail blocking, in doors indicated to have closers and/or kick plates.

### B. Fire Rated Doors:

- 1. Provide blocking as indicated below:
  - a. HB1: 5 inch in doors indicated to have closers and overhead stops.

## 2.5 LITE FRAMES

- A. Wood Beads for lite Openings in Wood Doors:
  - 1. Wood Species: Same species as door faces.
  - 2. Profile: Type M1 Flush Bead.
  - 3. At 20-minute, fire-rated, wood-core doors, provide wood beads and metal glazing clips approved for such use.

- M.A. No. H1-22029-DA RMG - 22029
- B. Metal Frames for Light Openings in Fire Doors: Manufacturer's standard frame formed of 0.048-inch thick, cold-rolled steel sheet; baked enamel or powder coated finish and approved for use in doors of fire rating indicated.
  - 1. Manufacturers:
    - a. All Metal Stamping.
    - b. Anemostat.
    - c. Pemko.

### 2.6 FABRICATION

- A. Factory fit doors to suit frame-opening sizes indicated, with the following uniform clearances and bevels, unless otherwise indicated:
  - 1. Comply with clearance requirements of referenced quality standard for fitting. Comply with requirements in NFPA 80 for fire-rated doors.
  - 2. Undercut: As required per manufacturer's templates and sill condition.
- B. Factory machine doors for hardware that is not surface applied. Locate hardware to comply with DHI-WDHS-3. Comply with final hardware schedules, door frame Shop Drawings, DHI A115-W series standards, and hardware templates.
  - 1. Coordinate measurements of hardware mortises in metal frames to verify dimensions and alignment before factory machining.
  - 2. Metal Astragals: Factory machine astragals and formed steel edges for hardware for pairs of fire rated doors.
- C. Openings: Cut and trim openings through doors to comply with applicable requirements of referenced standards for kind(s) of door(s) required.
  - 1. Lite Openings: Trim openings with moldings of material and profile indicated.
- D. Electrical Raceways: Provide flush wood doors receiving electrified hardware with concealed wiring harness and standardized Molex<sup>TM</sup> plug connectors on both ends to accommodate up to twelve wires. Coordinate connectors on end of the wiring harness to plug directly into the electrified hardware and the through wire transfer hardware or wiring harness specified in hardware sets in Division 08 "Door Hardware". Wire nut connections are not acceptable.

#### 2.7 FACTORY FINISHING

- A. General: Comply with referenced quality standard AWI's "Architectural Woodwork Quality Standards Illustrated" for factory finishing.
- B. Complete fabrication, including fitting doors for openings and machining for hardware that is not surface applied, before finishing.
  - 1. Finish faces, all four edges, edges of cutouts, and mortises. Stains and fillers may be omitted on top and bottom edges, edges of cutouts, and mortises.
- C. Finish doors at factory.
- D. Transparent Finish: Provide a clear protective coating over the wood veneer allowing the natural color and grain of the selected wood species to provide the appearance specified. Stain is applied to the wood surface underneath the transparent finish to add color and design flexibility.
  - 1. Grade: Premium

- M.A. No. H1-22029-DA RMG - 22029
- 2. Finish: AWI System TR-6 catalyzed polyurethane or WDMA I.S. 1A TR8 UV Cured Acrylated Polyester finish performance requirements
- 3. Staining: As selected by Architect from manufacturer's full range.
- 4. Effect: Open-grain finish.
- 5. Sheen: Satin

## PART 3 - EXECUTION

### 3.1 EXAMINATION

- A. Examine doors and installed door frames before hanging doors.
  - 1. Verify that frames comply with indicated requirements for type, size, location, and swing characteristics and have been installed with level heads and plumb jambs.
  - 2. Doors with face or edge defects or shipping damage are to be rejected.
- B. Proceed with installation only after unsatisfactory conditions have been corrected.

### 3.2 INSTALLATION

- A. Hardware: For installation, see Section 08 71 00 "Door Hardware."
- B. Manufacturer's Written Instructions: Install doors to comply with manufacturer's written instructions, referenced quality standard, and as indicated.
  - 1. Install fire-rated doors in corresponding fire-rated frames according to NFPA 80.
- C. Factory-Fitted Doors: Align in frames for uniform clearance at each edge.
- D. Factory-Finished Doors: Restore finish before installation if fitting or machining is required at Project site.
- E. Field modifications to doors shall not be permitted, except those specifically allowed by manufacturer or fire rating requirements.

### 3.3 ADJUSTING

- A. Operation: Rehang or replace doors that do not swing or operate freely.
- B. Door Finish: Replace doors that are damaged or do not comply with requirements. Doors may be repaired or refinished only if it consists of minor damage of finish and work otherwise complies with requirements and that such repair exhibits no evidence of repair or refinishing.

END OF SECTION 08 14 16

### SECTION 08 41 13 - ALUMINUM-FRAMED ENTRANCES AND STOREFRONTS

## PART 1 - GENERAL

### 1.1 SUMMARY

- A. Section Includes:
  - 1. Storefront framing.
  - 2. Manual-swing entrance doors.

### 1.2 PREINSTALLATION MEETINGS

A. Preinstallation Conference: Conduct conference at Project site.

### 1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings: For aluminum-framed entrances and storefronts. Include plans, elevations, sections, full-size details, and attachments to other work.
  - 1. Show connection to and continuity with adjacent thermal, weather, air, and vapor barriers.
  - 2. Include point-to-point wiring diagrams.
- C. Samples: For each type of exposed finish required.
- D. Entrance Door Hardware Schedule: Prepared by or under supervision of supplier, detailing fabrication and assembly of entrance door hardware, as well as procedures and diagrams.
- E. Delegated-Design Submittal: For aluminum-framed entrances and storefronts indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.

# 1.4 INFORMATIONAL SUBMITTALS

- A. Energy Performance Certificates: NFRC-certified energy performance values from manufacturer.
- B. Product test reports.
- C. Source quality-control reports.
- D. Field quality-control reports.

E. Sample warranties.

## 1.5 CLOSEOUT SUBMITTALS

A. Maintenance data.

# 1.6 QUALITY ASSURANCE

- A. Installer Qualifications: An entity that employs installers and supervisors who are trained and approved by manufacturer.
- B. Product Options: Information on Drawings and in Specifications establishes requirements for aesthetic effects and performance characteristics of assemblies. Aesthetic effects are indicated by dimensions, arrangements, alignment, and profiles of components and assemblies as they relate to sightlines, to one another, and to adjoining construction.

### 1.7 WARRANTY

- A. Special Warranty: Installer agrees to repair or replace components of aluminum-framed entrances and storefronts that do not comply with requirements or that fail in materials or workmanship within specified warranty period.
  - 1. Warranty Period: 10 years from date of Substantial Completion.
- B. Special Finish Warranty: Standard form in which manufacturer agrees to repair finishes or replace aluminum that shows evidence of deterioration of factory-applied finishes within specified warranty period.
  - 1. Warranty Period: 20 years from date of Substantial Completion.

### PART 2 - PRODUCTS

# 2.1 PERFORMANCE REOUIREMENTS

- A. Delegated Design: Engage a qualified professional engineer, as defined in Section 014000 "Quality Requirements," to design aluminum-framed entrances and storefronts.
- B. General Performance: Comply with performance requirements specified, as determined by testing of aluminum-framed entrances and storefronts representing those indicated for this Project without failure due to defective manufacture, fabrication, installation, or other defects in construction.
  - 1. Aluminum-framed entrances and storefronts shall withstand movements of supporting structure, including, but not limited to, twist, column shortening, long-term creep, and deflection from uniformly distributed and concentrated live loads.
  - 2. Failure also includes the following:

- M.A. No. H1-22029-DA RMG - 22029
- a. Thermal stresses transferring to building structure.
- b. Glass breakage.
- c. Noise or vibration created by wind and thermal and structural movements.
- d. Loosening or weakening of fasteners, attachments, and other components.
- e. Failure of operating units.

### C. Structural Loads:

- 1. Wind Loads: As indicated on Drawings.
- D. Deflection of Framing Members: At design wind pressure, as follows:
  - 1. Deflection Normal to Wall Plane: Limited to edge of glass in a direction perpendicular to glass plane not exceeding 1/175 of the glass edge length for each individual glazing lite or an amount that restricts edge deflection of individual glazing lites to 3/4 inch (19.1 mm), whichever is less.
  - 2. Deflection Parallel to Glazing Plane: Limited to 1/360 of clear span or 1/8 inch (3.2 mm), whichever is smaller.
  - 3. Cantilever Deflection: Where framing members overhang an anchor point, as follows:
    - a. Perpendicular to Plane of Wall: No greater than 1/240 of clear span plus 1/4 inch (6.35 mm) for spans greater than 11 feet 8-1/4 inches (3.6 m) or 1/175 times span, for spans of less than 11 feet 8-1/4 inches (3.6 m).
- E. Structural: Test according to ASTM E 330/E 330M as follows:
  - 1. When tested at positive and negative wind-load design pressures, storefront assemblies, including entrance doors, do not evidence deflection exceeding specified limits.
  - 2. Test Durations: As required by design wind velocity, but not less than 10 seconds.
- F. Air Infiltration: Test according to ASTM E 283 for infiltration as follows:
  - 1. Fixed Framing and Glass Area:
    - a. Maximum air leakage of 0.06 cfm/sq. ft. (0.30 L/s per sq. m) at a static-air-pressure differential of 6.24 lbf/sq. ft. (300 Pa).
  - 2. Entrance Doors:
    - a. Single Doors: Maximum air leakage of 0.5 cfm/sq. ft. (2.54 L/s per sq. m) at a static-air-pressure differential of 1.57 lbf/sq. ft. (75 Pa).
- G. Water Penetration under Static Pressure: Test according to ASTM E 331 as follows:
  - 1. No evidence of water penetration through fixed glazing and framing areas, including entrance doors, when tested according to a minimum static-air-pressure differential of 20 percent of positive wind-load design pressure, but not less than 10 lbf/sq. ft. (480 Pa).
- H. Energy Performance: Certify and label energy performance according to NFRC as follows:

- M.A. No. H1-22029-DA RMG - 22029
- 1. Thermal Transmittance (U-factor): Fixed glazing and framing areas as a system shall have U-factor of not more than 0.41 Btu/sq. ft. x h x deg F (2.33 W/sq. m x K) as determined according to NFRC 100.
- 2. Solar Heat Gain Coefficient (SHGC): Fixed glazing and framing areas as a system shall have SHGC of no greater than .060 as determined according to NFRC 200.
- 3. Condensation Resistance: Fixed glazing and framing areas as a system shall have an NFRC-certified condensation resistance rating of no less than 75 as determined according to NFRC 500.
- I. Thermal Movements: Allow for thermal movements resulting from ambient and surface temperature changes.
  - 1. Temperature Change: 120 deg F (67 deg C), ambient; 180 deg F (100 deg C), material surfaces.

### 2.2 STOREFRONT SYSTEMS

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. Kawneer North America, an Arconic company.
  - 2. <u>Manko Window Systems, Inc.</u>
  - 3. Tubelite Inc.
  - 4. U.S. Aluminum; a brand of C.R. Laurence.
- B. Framing Members: Manufacturer's extruded- or formed-aluminum framing members of thickness required and reinforced as required to support imposed loads.
  - 1. Exterior Framing Construction: Thermally broken.
  - 2. Interior Vestibule Framing Construction: Nonthermal.
  - 3. Glazing System: Retained mechanically with gaskets on four sides.
  - 4. Finish: Clear anodic finish Color anodic finish.
  - 5. Fabrication Method: Field-fabricated stick system.
  - 6. Aluminum: Alloy and temper recommended by manufacturer for type of use and finish indicated.
  - 7. Steel Reinforcement: As required by manufacturer.
- C. Backer Plates: Manufacturer's standard, continuous backer plates for framing members, if not integral, where framing abuts adjacent construction.
- D. Brackets and Reinforcements: Manufacturer's standard high-strength aluminum with nonstaining, nonferrous shims for aligning system components.

# 2.3 ENTRANCE DOOR SYSTEMS

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. Kawneer North America, an Arconic company.

- 2. Manko Window Systems, Inc.
- 3. U.S. Aluminum; a brand of C.R. Laurence.
- B. Entrance Doors: Manufacturer's standard glazed entrance doors for manual-swing or automatic operation.
  - 1. Door Construction: 1-3/4-inch (44.5-mm) overall thickness, with minimum 0.125-inch-(3.2-mm-) thick, extruded-aluminum tubular rail and stile members. Mechanically fasten corners with reinforcing brackets that are deeply penetrated and fillet welded or that incorporate concealed tie rods.
    - a. Thermal Construction: High-performance plastic connectors separate aluminum members exposed to the exterior from members exposed to the interior.
  - 2. Door Design: Wide stile; 5-inch (127-mm) nominal width.
  - 3. Glazing Stops and Gaskets: Square, snap-on, extruded-aluminum stops and preformed gaskets.
    - a. Provide nonremovable glazing stops on outside of door.

### 2.4 ENTRANCE DOOR HARDWARE

- A. Entrance Door Hardware: Hardware not specified in this Section is specified in Section 087100 "Door Hardware."
- B. General: Provide entrance door hardware and entrance door hardware sets indicated in door and frame schedule for each entrance door, to comply with requirements in this Section.
- C. Designations: Requirements for design, grade, function, finish, quantity, size, and other distinctive qualities of each type of entrance door hardware are indicated in "Entrance Door Hardware Sets" Article. Products are identified by using entrance door hardware designations as follows:
  - 1. Named Manufacturers' Products: Manufacturer and product designation are listed for each door hardware type required for the purpose of establishing minimum requirements. Manufacturers' names are abbreviated in "Entrance Door Hardware Sets" Article.
  - 2. References to BHMA Standards: Provide products complying with these standards and requirements for description, quality, and function.
- D. Cylinders: As specified in Section 087100 "Door Hardware."
- E. Pivot Hinges: BHMA A156.4, Grade 1.
  - 1. Offset-Pivot Hinges: Provide top, bottom, and intermediate offset pivots at each door leaf.
- F. Mortise Auxiliary Locks: BHMA A156.5, Grade 1.
- G. Panic Exit Devices: BHMA A156.3, Grade 1, listed and labeled by a testing and inspecting agency acceptable to authorities having jurisdiction, for panic protection, based on testing according to UL 305.

- M.A. No. H1-22029-DA RMG - 22029
- H. Strikes: Provide strike with black-plastic dust box for each latch or lock bolt; fabricated for aluminum framing.
- I. Operating Trim: BHMA A156.6.
- J. Closers: BHMA A156.4, Grade 1, with accessories required for a complete installation, sized as required by door size, exposure to weather, and anticipated frequency of use; adjustable to comply with field conditions and requirements for opening force.
- K. Concealed Overhead Holders and Stops: BHMA A156.8, Grade 1.
- L. Door Stops: BHMA A156.16, Grade 1, floor or wall mounted, as appropriate for door location indicated, with integral rubber bumper.
- M. Weather Stripping: Manufacturer's standard replaceable components.
  - Compression Type: Made of ASTM D 2000 molded neoprene or ASTM D 2287 molded PVC.
  - 2. Sliding Type: AAMA 701/702, made of wool, polypropylene, or nylon woven pile with nylon-fabric or aluminum-strip backing.
- N. Weather Sweeps: Manufacturer's standard exterior-door bottom sweep with concealed fasteners on mounting strip.
- O. Thresholds: BHMA A156.21 raised thresholds beveled with a slope of not more than 1:2, with maximum height of 1/2 inch (12.7 mm).

# 2.5 GLAZING

- A. Glazing: Comply with Section 088000 "Glazing."
- B. Glazing Gaskets: Manufacturer's standard sealed-corner pressure-glazing system of black, resilient elastomeric glazing gaskets, setting blocks, and shims or spacers.
- C. Glazing Sealants: As recommended by manufacturer.

# 2.6 MATERIALS

- A. Sheet and Plate: ASTM B 209 (ASTM B 209M).
- B. Extruded Bars, Rods, Profiles, and Tubes: ASTM B 221 (ASTM B 221M).
- C. Extruded Structural Pipe and Tubes: ASTM B 429/B 429M.
- D. Structural Profiles: ASTM B 308/B 308M.
- E. Steel Reinforcement:
  - 1. Structural Shapes, Plates, and Bars: ASTM A 36/A 36M.
  - 2. Cold-Rolled Sheet and Strip: ASTM A 1008/A 1008M.

- M.A. No. H1-22029-DA RMG - 22029
- 3. Hot-Rolled Sheet and Strip: ASTM A 1011/A 1011M.
- 4. Primer: Manufacturer's standard zinc-rich, corrosion-resistant primer complying with SSPC-PS Guide No. 12.00; applied immediately after surface preparation and pretreatment. Select surface preparation methods according to recommendations in SSPC-SP COM, and prepare surfaces according to applicable SSPC standard.

## 2.7 FABRICATION

- A. Form or extrude aluminum shapes before finishing.
- B. Weld in concealed locations to greatest extent possible to minimize distortion or discoloration of finish. Remove weld spatter and welding oxides from exposed surfaces by descaling or grinding.
- C. Fabricate components that, when assembled, have the following characteristics:
  - 1. Profiles that are sharp, straight, and free of defects or deformations.
  - 2. Accurately fitted joints with ends coped or mitered.
  - 3. Physical and thermal isolation of glazing from framing members.
  - 4. Accommodations for thermal and mechanical movements of glazing and framing to maintain required glazing edge clearances.
  - 5. Provisions for field replacement of glazing from interior.
  - 6. Fasteners, anchors, and connection devices that are concealed from view to greatest extent possible.
- D. Mechanically Glazed Framing Members: Fabricate for flush glazing without projecting stops.
- E. Entrance Door Frames: Reinforce as required to support loads imposed by door operation and for installing entrance door hardware.
- F. Entrance Doors: Reinforce doors as required for installing entrance door hardware.
- G. Entrance Door Hardware Installation: Factory install entrance door hardware to the greatest extent possible. Cut, drill, and tap for factory-installed entrance door hardware before applying finishes.
- H. After fabrication, clearly mark components to identify their locations in Project according to Shop Drawings.

## 2.8 ALUMINUM FINISHES

- A. Clear Anodic Finish: AAMA 611, AA-M12C22A31, Class II, 0.010 mm or thicker.
- B. Color Anodic Finish: AAMA 611, AA-M12C22A32/A34, Class II, 0.010 mm or thicker.
  - 1. Color: match existing.

### PART 3 - EXECUTION

# 3.1 INSTALLATION

### A. General:

- 1. Comply with manufacturer's written instructions.
- 2. Do not install damaged components.
- 3. Fit joints to produce hairline joints free of burrs and distortion.
- 4. Rigidly secure nonmovement joints.
- 5. Install anchors with separators and isolators to prevent metal corrosion and electrolytic deterioration and to prevent impeding movement of moving joints.
- 6. Seal perimeter and other joints watertight unless otherwise indicated.

## B. Metal Protection:

- 1. Where aluminum is in contact with dissimilar metals, protect against galvanic action by painting contact surfaces with materials recommended by manufacturer for this purpose or by installing nonconductive spacers.
- 2. Where aluminum is in contact with concrete or masonry, protect against corrosion by painting contact surfaces with bituminous paint.
- C. Set continuous sill members and flashing in full sealant bed, as specified in Section 079200 "Joint Sealants," to produce weathertight installation.
- D. Install components plumb and true in alignment with established lines and grades.
- E. Install operable units level and plumb, securely anchored, and without distortion. Adjust weather-stripping contact and hardware movement to produce proper operation.
- F. Install glazing as specified in Section 088000 "Glazing."
- G. Entrance Doors: Install doors to produce smooth operation and tight fit at contact points.
  - 1. Exterior Doors: Install to produce weathertight enclosure and tight fit at weather stripping.
  - 2. Field-Installed Entrance Door Hardware: Install surface-mounted entrance door hardware according to entrance door hardware manufacturers' written instructions using concealed fasteners to greatest extent possible.

## 3.2 FIELD QUALITY CONTROL

- A. Testing Agency: Owner will engage a qualified testing agency to perform tests and inspections.
- B. Field Quality-Control Testing: Perform the following test on representative areas of aluminum-framed entrances and storefronts.
  - 1. Water-Spray Test: Before installation of interior finishes has begun, areas designated by Architect shall be tested according to AAMA 501.2 and shall not evidence water penetration.

- a. Perform a minimum of two tests in areas as directed by Architect.
- 2. Air Infiltration: ASTM E 783 at 1.5 times the rate specified for laboratory testing in "Performance Requirements" Article but not more than 0.09 cfm/sq. ft. (0.45 L/s per sq. m) at a static-air-pressure differential of 1.57 lbf/sq. ft. (75 Pa).
  - a. Perform a minimum of two tests in areas as directed by Architect.
- 3. Water Penetration: ASTM E 1105 at a minimum uniform and cyclic static-air-pressure differential of 0.67 times the static-air-pressure differential specified for laboratory testing in "Performance Requirements" Article, but not less than 6.24 lbf/sq. ft. (300 Pa), and shall not evidence water penetration.
- C. Aluminum-framed entrances and storefronts will be considered defective if they do not pass tests and inspections.
- D. Prepare test and inspection reports.

END OF SECTION 08 41 13

This page intentionally left blank

### SECTION 08 71 00 - DOOR HARDWARE

### PART 1 - GENERAL

### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 1 Specification Sections, apply to this Section.

### 1.2 SUMMARY

- A. This Section includes commercial door hardware for the following:
  - 1. Swinging doors.
- B. Door hardware includes, but is not necessarily limited to, the following:
  - 1. Mechanical door hardware.
- C. Related Sections:
  - 1. Division 08 Section "Hollow Metal Doors and Frames".
  - 2. Division 08 Section "Flush Wood Doors".
  - 3. Division 08 Section "Aluminum-Framed Entrances and Storefronts".
- D. Codes and References: Comply with the version year adopted by the Authority Having Jurisdiction.
  - 1. ANSI A117.1 Accessible and Usable Buildings and Facilities.
  - 2. ICC/IBC International Building Code.
  - 3. NFPA 70 National Electrical Code.
  - 4. NFPA 80 Fire Doors and Windows.
  - 5. NFPA 101 Life Safety Code.
  - 6. NFPA 105 Installation of Smoke Door Assemblies.
  - 7. State Building Codes, Local Amendments.
- E. Standards: All hardware specified herein shall comply with the following industry standards as applicable. Any undated reference to a standard shall be interpreted as referring to the latest edition of that standard:
  - 1. ANSI/BHMA Certified Product Standards A156 Series.
  - 2. UL10C Positive Pressure Fire Tests of Door Assemblies.
  - 3. ANSI/UL 294 Access Control System Units.

DOOR HARDWARE 08 71 00 - 1

500

### 1.3 SUBMITTALS

- A. Product Data: Manufacturer's product data sheets including installation details, material descriptions, dimensions of individual components and profiles, operational descriptions and finishes.
- B. Door Hardware Schedule: Prepared by or under the supervision of supplier, detailing fabrication and assembly of door hardware, as well as procedures and diagrams. Coordinate the final Door Hardware Schedule with doors, frames, and related work to ensure proper size, thickness, hand, function, and finish of door hardware.
  - 1. Format: Comply with scheduling sequence and vertical format in DHI's "Sequence and Format for the Hardware Schedule."
  - 2. Organization: Organize the Door Hardware Schedule into door hardware sets indicating complete designations of every item required for each door or opening. Organize door hardware sets in same order as in the Door Hardware Sets at the end of Part 3. Submittals that do not follow the same format and order as the Door Hardware Sets will be rejected and subject to resubmission.
  - 3. Content: Include the following information:
    - a. Type, style, function, size, label, hand, and finish of each door hardware item.
    - b. Manufacturer of each item.
    - c. Fastenings and other pertinent information.
    - d. Location of door hardware set, cross-referenced to Drawings, both on floor plans and in door and frame schedule.
    - e. Explanation of abbreviations, symbols, and codes contained in schedule.
    - f. Mounting locations for door hardware.
    - g. Door and frame sizes and materials.
    - h. Warranty information for each product.
  - 4. Submittal Sequence: Submit the final Door Hardware Schedule at earliest possible date, particularly where approval of the Door Hardware Schedule must precede fabrication of other work that is critical in the Project construction schedule. Include Product Data, Samples, Shop Drawings of other work affected by door hardware, and other information essential to the coordinated review of the Door Hardware Schedule.
- C. Keying Schedule: After a keying meeting with the owner has taken place prepare a separate keying schedule detailing final instructions. Submit the keying schedule in electronic format. Include keying system explanation, door numbers, key set symbols, hardware set numbers and special instructions. Owner must approve submitted keying schedule prior to the ordering of permanent cylinders/cores.

#### D. Informational Submittals:

1. Product Test Reports: Indicating compliance with cycle testing requirements, based on evaluation of comprehensive tests performed by manufacturer and witnessed by a qualified independent testing agency.

DOOR HARDWARE 08 71 00 - 2

E. Operating and Maintenance Manuals: Provide manufacturers operating and maintenance manuals for each item comprising the complete door hardware installation in quantity as required in Division 01, Closeout Procedures.

## 1.4 QUALITY ASSURANCE

- A. Hardware Supplier and Hardware Installer must obtain a license with the Louisiana Office of State Fire Marshall in accordance to RS 40:1464 and RS 40:1664.
- B. Manufacturers Qualifications: Engage qualified manufacturers with a minimum 5 years of documented experience in producing hardware and equipment similar to that indicated for this Project and that have a proven record of successful in-service performance.
- C. Certified Products: Where specified, products must maintain a current listing in the Builders Hardware Manufacturers Association (BHMA) Certified Products Directory (CPD).
- D. Installer Qualifications: A minimum 3 years documented experience installing both standard and electrified door hardware similar in material, design, and extent to that indicated for this Project and whose work has resulted in construction with a record of successful in-service performance.
- E. Door Hardware Supplier Qualifications: Experienced commercial door hardware distributors with a minimum 5 years documented experience supplying both mechanical and electromechanical hardware installations comparable in material, design, and extent to that indicated for this Project. Supplier recognized as a factory direct distributor by the manufacturers of the primary materials with a warehousing facility in Project's vicinity. Supplier to have on staff a certified Architectural Hardware Consultant (AHC) available during the course of the Work to consult with Contractor, Architect, and Owner concerning both standard and electromechanical door hardware and keying.
- F. Source Limitations: Obtain each type and variety of door hardware specified in this section from a single source unless otherwise indicated.
  - 1. Electrified modifications or enhancements made to a source manufacturer's product line by a secondary or third party source will not be accepted.
- G. Each unit to bear third party permanent label demonstrating compliance with the referenced standards.
- H. Keying Conference: Conduct conference to comply with requirements in Division 01 Section "Project Meetings." Keying conference to incorporate the following criteria into the final keying schedule document:
  - 1. Function of building, purpose of each area and degree of security required.
  - 2. Plans for existing and future key system expansion.
  - 3. Requirements for key control storage and software.
  - 4. Installation of permanent keys, cylinder cores and software.
  - 5. Address and requirements for delivery of keys.

DOOR HARDWARE 08 71 00 - 3

- M.A. No. H1-22029-DA RMG - 22029
- I. Pre-Submittal Conference: Conduct coordination conference in compliance with requirements in Division 01 Section "Project Meetings" with attendance by representatives of Supplier(s), Installer(s), and Contractor(s) to review proper methods and the procedures for receiving, handling, and installing door hardware.
  - 1. Prior to installation of door hardware, conduct a project specific training meeting to instruct the installing contractors' personnel on the proper installation and adjustment of their respective products. Product training to be attended by installers of door hardware (including electromechanical hardware) for aluminum, hollow metal and wood doors. Training will include the use of installation manuals, hardware schedules, templates and physical product samples as required.
  - 2. Inspect and discuss electrical roughing-in, power supply connections, and other preparatory work performed by other trades.
  - 3. Review sequence of operation narratives for each unique access controlled opening.
  - 4. Review and finalize construction schedule and verify availability of materials.
  - 5. Review the required inspecting, testing, commissioning, and demonstration procedures
- J. At completion of installation, provide written documentation that components were applied to manufacturer's instructions and recommendations and according to approved schedule.

#### 1.5 DELIVERY, STORAGE, AND HANDLING

- A. Inventory door hardware on receipt and provide secure lock-up and shelving for door hardware delivered to Project site. Do not store electronic access control hardware, software or accessories at Project site without prior authorization.
- B. Tag each item or package separately with identification related to the final Door Hardware Schedule, and include basic installation instructions with each item or package.
- C. Deliver, as applicable, permanent keys, cylinders, cores, access control credentials, software and related accessories directly to Owner via registered mail or overnight package service. Instructions for delivery to the Owner shall be established at the "Keying Conference".

#### 1.6 COORDINATION

- A. Templates: Obtain and distribute to the parties involved templates for doors, frames, and other work specified to be factory prepared for installing standard and electrified hardware. Check Shop Drawings of other work to confirm that adequate provisions are made for locating and installing hardware to comply with indicated requirements.
- B. Door and Frame Preparation: Doors and corresponding frames are to be prepared, reinforced and pre-wired (if applicable) to receive the installation of the specified electrified, monitoring, signaling and access control system hardware without additional in-field modifications.

#### 1.7 WARRANTY

- A. General Warranty: Reference Division 01, General Requirements. Special warranties specified in this Article shall not deprive Owner of other rights Owner may have under other provisions of the Contract Documents and shall be in addition to, and run concurrent with, other warranties made by Contractor under requirements of the Contract Documents.
- B. Warranty Period: Written warranty, executed by manufacturer(s), agreeing to repair or replace components of standard and electrified door hardware that fails in materials or workmanship within specified warranty period after final acceptance by the Owner. Failures include, but are not limited to, the following:
  - 1. Structural failures including excessive deflection, cracking, or breakage.
  - 2. Faulty operation of the hardware.
  - 3. Deterioration of metals, metal finishes, and other materials beyond normal weathering.
  - 4. Electrical component defects and failures within the systems operation.
- C. Warranty Period: Unless otherwise indicated, warranty shall be one year from date of Substantial Completion.

### 1.8 MAINTENANCE SERVICE

A. Maintenance Tools and Instructions: Furnish a complete set of specialized tools and maintenance instructions as needed for Owner's continued adjustment, maintenance, and removal and replacement of door hardware.

### PART 2 - PRODUCTS

#### 2.1 SCHEDULED DOOR HARDWARE

- A. General: Provide door hardware for each door to comply with requirements in Door Hardware Sets and each referenced section that products are to be supplied under.
- B. Designations: Requirements for quantity, item, size, finish or color, grade, function, and other distinctive qualities of each type of door hardware are indicated in the Door Hardware Sets at the end of Part 3. Products are identified by using door hardware designations, as follows:
  - 1. Named Manufacturer's Products: Product designation and manufacturer are listed for each door hardware type required for the purpose of establishing requirements. Manufacturers' names are abbreviated in the Door Hardware Schedule.
- C. Substitutions: Requests for substitution and product approval for inclusive mechanical and electromechanical door hardware in compliance with the specifications must be submitted in writing and in accordance with the procedures and time frames outlined in Division 01, Substitution Procedures. Approval of requests is at the discretion of the architect, owner, and their designated consultants.

#### 2.2 HANGING DEVICES

- A. Hinges: ANSI/BHMA A156.1 butt hinges with number of hinge knuckles and other options as specified in the Door Hardware Sets.
  - 1. Quantity: Provide the following hinge quantity:
    - a. Two Hinges: For doors with heights up to 60 inches.
    - b. Three Hinges: For doors with heights 61 to 90 inches.
    - c. Four Hinges: For doors with heights 91 to 120 inches.
    - d. For doors with heights more than 120 inches, provide 4 hinges, plus 1 hinge for every 30 inches of door height greater than 120 inches.
  - 2. Hinge Size: Provide the following, unless otherwise indicated, with hinge widths sized for door thickness and clearances required:
    - a. Widths up to 3'0": 4-1/2" standard or heavy weight as specified.
    - b. Sizes from 3'1" to 4'0": 5" standard or heavy weight as specified.
  - 3. Hinge Weight and Base Material: Unless otherwise indicated, provide the following:
    - a. Exterior Doors: Heavy weight, non-ferrous, ball bearing or oil impregnated bearing hinges unless Hardware Sets indicate standard weight.
    - b. Interior Doors: Standard weight, steel, ball bearing or oil impregnated bearing hinges unless Hardware Sets indicate heavy weight.
  - 4. Hinge Options: Comply with the following:
    - a. Non-removable Pins: With the exception of electric through wire hinges, provide set screw in hinge barrel that, when tightened into a groove in hinge pin, prevents removal of pin while door is closed; for the all out-swinging lockable doors.
  - 5. Manufacturers:
    - a. Hager Companies (HA) BB Series, 5 knuckle.
    - b. McKinney (MK) TA/T4A Series, 5 knuckle.
- B. Continuous Geared Hinges: ANSI/BHMA A156.26 Grade 1-600 continuous geared hinge. with minimum 0.120-inch thick extruded 6063-T6 aluminum alloy hinge leaves and a minimum overall width of 4 inches. Hinges are non-handed, reversible and fabricated to template screw locations. Factory trim hinges to suit door height and prepare for electrical cut-outs.
  - 1. Manufacturers:.
    - a. Hager Companies (HA).
    - b. Pemko (PE).

#### 2.3 DOOR OPERATING TRIM

- A. Door Push Plates and Pulls: ANSI/BHMA A156.6 door pushes and pull units of type and design specified in the Hardware Sets. Coordinate and provide proper width and height as required where conflicting hardware dictates.
  - 1. Push/Pull Plates: Minimum .050 inch thick, size as indicated in hardware sets, with beveled edges, secured with exposed screws unless otherwise indicated.
  - 2. Door Pull and Push Bar Design: Size, shape, and material as indicated in the hardware sets. Minimum clearance of 2 1/2-inches from face of door unless otherwise indicated.
  - 3. Offset Pull Design: Size, shape, and material as indicated in the hardware sets. Minimum clearance of 2 1/2-inches from face of door and offset of 90 degrees unless otherwise indicated.
  - 4. Pulls, where applicable, shall be provided with a 10" clearance from the finished floor on the push side to accommodate wheelchair accessibility.
  - 5. Fasteners: Provide manufacturer's designated fastener type as indicated in Hardware Sets.
  - 6. Manufacturers:
    - a. Rockwood (RO).
    - b. Trimco (TC).

#### 2.4 CYLINDERS AND KEYING

- A. General: Cylinder manufacturer to have minimum (10) years experience designing secured master key systems and have on record a published security keying system policy.
- B. Cylinder Types: Original manufacturer cylinders able to supply the following cylinder formats and types:
  - 1. Threaded mortise cylinders with rings and cams to suit hardware application.
  - 2. Rim cylinders with back plate, flat-type vertical or horizontal tailpiece, and raised trim ring.
  - 3. Bored or cylindrical lock cylinders with tailpieces as required to suit locks.
  - 4. Tubular deadlocks and other auxiliary locks.
  - 5. Mortise and rim cylinder collars to be solid and recessed to allow the cylinder face to be flush and be free spinning with matching finishes.
  - 6. Keyway: Manufacturer's Standard.
- C. Patented Cylinders: ANSI/BHMA A156.5, Grade 1 Certified Products Directory (CPD) listed cylinders employing a utility patented and restricted keyway requiring the use of a patented key. Cylinders are to be protected from unauthorized manufacture and distribution by manufacturer's United States patents. Cylinders are to be factory keyed with owner having the ability for onsite original key cutting.
  - 1. Patented key systems shall not be established with products that have an expired patent. Expired systems shall only be specified and supplied to support existing systems.

2. Manufacturers:

- a. Corbin Russwin (RU) Access 3 AP.
- b. Sargent (SA) Degree DG1.
- c. Schlage (SC) Everest 29 SL.
- D. Keying System: Each type of lock and cylinders to be factory keyed.
  - 1. Supplier shall conduct a "Keying Conference" to define and document keying system instructions and requirements.
  - 2. Furnish factory cut, nickel-silver large bow permanently inscribed with a visual key control number as directed by Owner.
  - 3. New System: Key locks to a new key system as directed by the Owner.
- E. Key Quantity: Provide the following minimum number of keys:
  - 1. Change Keys per Cylinder: Two (2)
  - 2. Master Keys (per Master Key Level/Group): Five (5).
  - 3. Construction Keys (where required): Ten (10).
  - 4. Construction Control Keys (where required): Two (2).
  - 5. Permanent Control Keys (where required): Two (2).
- F. Construction Keying: Provide temporary keyed construction cores.
- G. Key Registration List (Bitting List):
  - 1. Provide keying transcript list to Owner's representative in the proper format for importing into key control software.
  - 2. Provide transcript list in writing or electronic file as directed by the Owner.

### 2.5 KEY CONTROL

- A. Key Control Cabinet: Provide a key control system including envelopes, labels, and tags with self-locking key clips, receipt forms, 3-way visible card index, temporary markers, permanent markers, and standard metal cabinet. Key control cabinet shall have expansion capacity of 150% of the number of locks required for the project.
  - 1. Manufacturers:
    - a. Lund Equipment (LU).
    - b. MMF Industries (MM).
    - c. Telkee (TK).

#### 2.6 MECHANICAL LOCKS AND LATCHING DEVICES

- A. Cylindrical Locksets, Grade 1 (Heavy Duty): ANSI/BHMA A156.2, Series 4000, Operational Grade 1 Certified Products Directory (CPD) listed.
  - 1. Heavy duty cylindrical locks shall have a seven-year warranty.

- M.A. No. H1-22029-DA RMG - 22029
- 2. Vertical Impact: Exceed 100 vertical impacts (20 times ANSI/BHMA A156.2 requirements).
- 3. Furnish with solid cast levers, standard 2 3/4" backset, and 1/2" (3/4" at rated paired openings) throw brass or stainless steel latchbolt.
- 4. Locks are to be non-handed and fully field reversible.
- 5. Manufacturers:
  - a. Corbin Russwin Hardware (RU) CLX3300 Series.
  - b. Sargent Manufacturing (SA) 10X Line.
  - c. Schlage (SC) ND Series.

#### 2.7 LOCK AND LATCH STRIKES

- A. Strikes: Provide manufacturer's standard strike with strike box for each latch or lock bolt, with curved lip extended to protect frame, finished to match door hardware set, unless otherwise indicated, and as follows:
  - 1. Flat-Lip Strikes: For locks with three-piece antifriction latchbolts, as recommended by manufacturer.
  - 2. Extra-Long-Lip Strikes: For locks used on frames with applied wood casing trim.
  - 3. Aluminum-Frame Strike Box: Provide manufacturer's special strike box fabricated for aluminum framing.
  - 4. Double-lipped strikes: For locks at double acting doors. Furnish with retractable stop for rescue hardware applications.
- B. Standards: Comply with the following:
  - 1. Strikes for Mortise Locks and Latches: BHMA A156.13.
  - 2. Strikes for Bored Locks and Latches: BHMA A156.2.
  - 3. Strikes for Auxiliary Deadlocks: BHMA A156.36.
  - 4. Dustproof Strikes: BHMA A156.16.

### 2.8 CONVENTIONAL EXIT DEVICES

- A. General Requirements: All exit devices specified herein shall meet or exceed the following criteria:
  - 1. Exit devices shall have a five-year warranty.
  - 2. At doors not requiring a fire rating, provide devices complying with NFPA 101 and listed and labeled for "Panic Hardware" according to UL305. Provide proper fasteners as required by manufacturer including sex nuts and bolts at openings specified in the Hardware Sets.

- M.A. No. H1-22029-DA RMG - 22029
- 3. Where exit devices are required on fire rated doors, provide devices complying with NFPA 80 and with UL labeling indicating "Fire Exit Hardware". Provide devices with the proper fasteners for installation as tested and listed by UL. Consult manufacturer's catalog and template book for specific requirements.
- 4. Except on fire rated doors, provide exit devices with hex key dogging device to hold the pushbar and latch in a retracted position. Provide optional keyed cylinder dogging on devices where specified in Hardware Sets.
- 5. Devices must fit flat against the door face with no gap that permits unauthorized dogging of the push bar. The addition of filler strips is required in any case where the door light extends behind the device as in a full glass configuration.
- 6. Lever Operating Trim: Where exit devices require lever trim, furnish manufacturer's heavy duty escutcheon trim with threaded studs for thru-bolts.
  - a. Lock Trim Design: As indicated in Hardware Sets, provide finishes and designs to match that of the specified locksets.
  - b. Where function of exit device requires a cylinder, provide a cylinder (Rim or Mortise) as specified in Hardware Sets.
- 7. Vertical Rod Exit Devices: Where surface or concealed vertical rod exit devices are used at interior openings, provide as less bottom rod (LBR) unless otherwise indicated. Provide dust proof strikes where thermal pins are required to project into the floor.
- 8. Narrow Stile Applications: At doors constructed with narrow stiles, or as specified in Hardware Sets, provide devices designed for maximum 2" wide stiles.
- 9. Dummy Push Bar: Nonfunctioning push bar matching functional push bar.
- 10. Rail Sizing: Provide exit device rails factory sized for proper door width application.
- 11. Through Bolt Installation: For exit devices and trim as indicated in Door Hardware Sets.
- B. Conventional Push Rail Exit Devices (Heavy Duty): ANSI/BHMA A156.3, Grade 1 Certified Products Directory (CPD) listed panic and fire exit hardware devices furnished in the functions specified in the Hardware Sets. Exit device latch to be stainless steel, pullman type, with deadlock feature.
  - 1. Manufacturers:
    - a. Corbin Russwin Hardware (RU) ED4000 / ED5000 Series.
    - b. Sargent Manufacturing (SA) 80 Series.
    - c. Von Duprin (VD) 35A/98 XP Series.

#### 2.9 DOOR CLOSERS

A. All door closers specified herein shall meet or exceed the following criteria:

- M.A. No. H1-22029-DA RMG - 22029
- 1. General: Door closers to be from one manufacturer, matching in design and style, with the same type door preparations and templates regardless of application or spring size. Closers to be non-handed with full sized covers.
- 2. Standards: Closers to comply with UL-10C for Positive Pressure Fire Test and be U.L. listed for use of fire rated doors.
- 3. Size of Units: Comply with manufacturer's written recommendations for sizing of door closers depending on size of door, exposure to weather, and anticipated frequency of use. Where closers are indicated for doors required to be accessible to the Americans with Disabilities Act, provide units complying with ANSI ICC/A117.1.
- 4. Closer Arms: Provide heavy duty, forged steel closer arms unless otherwise indicated in Hardware Sets.
- 5. Closers shall not be installed on exterior or corridor side of doors; where possible install closers on door for optimum aesthetics.
- 6. Closer Accessories: Provide door closer accessories including custom templates, special mounting brackets, spacers and drop plates as required for proper installation. Provide through-bolt and security type fasteners as specified in the hardware sets.
- B. Door Closers, Surface Mounted (Heavy Duty): ANSI/BHMA A156.4, Grade 1 Certified Products Directory (CPD) listed surface mounted, heavy duty door closers with complete spring power adjustment, sizes 1 thru 6; and fully operational adjustable according to door size, frequency of use, and opening force. Closers to be rack and pinion type, one piece cast iron or aluminum alloy body construction, with adjustable backcheck and separate non-critical valves for closing sweep and latch speed control. Provide non-handed units standard..
  - 1. Manufacturers:
    - a. Corbin Russwin Hardware (RU) DC6000 Series.
    - b. LCN Closers (LC) 4040 Series.
    - c. Sargent Manufacturing (SA) 351 Series.

### 2.10 ARCHITECTURAL TRIM

#### A. Door Protective Trim

- 1. General: Door protective trim units to be of type and design as specified below or in the Hardware Sets.
- 2. Size: Fabricate protection plates (kick, armor, or mop) not more than 2" less than door width (LDW) on stop side of single doors and 1" LDW on stop side of pairs of doors, and not more than 1" less than door width on pull side. Coordinate and provide proper width and height as required where conflicting hardware dictates. Height to be as specified in the Hardware Sets.

- M.A. No. H1-22029-DA RMG - 22029
- 3. Where plates are applied to fire rated doors with the top of the plate more than 16" above the bottom of the door, provide plates complying with NFPA 80. Consult manufacturer's catalog and template book for specific requirements for size and applications.
- 4. Protection Plates: ANSI/BHMA A156.6 protection plates (kick, armor, or mop), fabricated from the following:
  - a. Stainless Steel: 300 grade, 050-inch thick.
- 5. Options and fasteners: Provide manufacturer's designated fastener type as specified in the Hardware Sets. Provide countersunk screw holes.
- 6. Manufacturers:
  - a. Rockwood (RO).
  - b. Trimco (TC).

#### 2.11 DOOR STOPS AND HOLDERS

- A. General: Door stops and holders to be of type and design as specified below or in the Hardware Sets.
- B. Door Stops and Bumpers: ANSI/BHMA A156.16, Grade 1 door stops and wall bumpers. Provide wall bumpers, either convex or concave types with anchorage as indicated, unless floor or other types of door stops are specified in Hardware Sets. Do not mount floor stops where they will impede traffic. Where floor or wall bumpers are not appropriate, provide overhead type stops and holders.
  - 1. Manufacturers:
    - a. Rockwood (RO).
    - b. Trimco (TC).
- C. Overhead Door Stops and Holders: ANSI/BHMA A156.8, Grade 1 Certified Products Directory (CPD) listed overhead stops and holders to be surface or concealed types as indicated in Hardware Sets. Track, slide, arm and jamb bracket to be constructed of extruded bronze and shock absorber spring of heavy tempered steel. Provide non-handed design with mounting brackets as required for proper operation and function.
  - 1. Manufacturers:
    - a. Norton Rixson (RF).
    - b. Rockwood (RO).

### 2.12 ARCHITECTURAL SEALS

A. General: Thresholds, weatherstripping, and gasket seals to be of type and design as specified below or in the Hardware Sets. Provide continuous weatherstrip gasketing on exterior doors and

- M.A. No. H1-22029-DA RMG - 22029
- provide smoke, light, or sound gasketing on interior doors where indicated. At exterior applications provide non-corrosive fasteners and elsewhere where indicated.
- B. Smoke Labeled Gasketing: Assemblies complying with NFPA 105 that are listed and labeled by a testing and inspecting agency acceptable to authorities having jurisdiction, for smoke control ratings indicated, based on testing according to UL 1784.
  - 1. Provide smoke labeled perimeter gasketing at all smoke labeled openings.
- C. Fire Labeled Gasketing: Assemblies complying with NFPA 80 that are listed and labeled by a testing and inspecting agency acceptable to authorities having jurisdiction, for fire ratings indicated, based on testing according to UL-10C.
  - 1. Provide intumescent seals as indicated to meet UL10C Standard for Positive Pressure Fire Tests of Door Assemblies, and NPFA 252, Standard Methods of Fire Tests of Door Assemblies.
- D. Sound-Rated Gasketing: Assemblies that are listed and labeled by a testing and inspecting agency, for sound ratings indicated.
- E. Replaceable Seal Strips: Provide only those units where resilient or flexible seal strips are easily replaceable and readily available from stocks maintained by manufacturer.
- F. Manufacturers:
  - 1. National Guard Products (NG).
  - 2. Pemko (PE).

#### 2.13 FABRICATION

A. Fasteners: Provide door hardware manufactured to comply with published templates generally prepared for machine, wood, and sheet metal screws. Provide screws according to manufacturers recognized installation standards for application intended.

### 2.14 FINISHES

- A. Standard: Designations used in the Hardware Sets and elsewhere indicate hardware finishes complying with ANSI/BHMA A156.18, including coordination with traditional U.S. finishes indicated by certain manufacturers for their products.
- B. Provide quality of finish, including thickness of plating or coating (if any), composition, hardness, and other qualities complying with manufacturer's standards, but in no case less than specified by referenced standards for the applicable units of hardware
- C. Protect mechanical finishes on exposed surfaces from damage by applying a strippable, temporary protective covering before shipping.

#### PART 3 - EXECUTION

### 3.1 EXAMINATION

- A. Examine scheduled openings, with Installer present, for compliance with requirements for installation tolerances, labeled fire door assembly construction, wall and floor construction, and other conditions affecting performance.
- B. Notify architect of any discrepancies or conflicts between the door schedule, door types, drawings and scheduled hardware. Proceed only after such discrepancies or conflicts have been resolved in writing.

### 3.2 PREPARATION

- A. Hollow Metal Doors and Frames: Comply with ANSI/DHI A115 series.
- B. Wood Doors: Comply with ANSI/DHI A115-W series.

#### 3.3 INSTALLATION

- A. Install each item of mechanical and electromechanical hardware and access control equipment to comply with manufacturer's written instructions and according to specifications.
  - 1. Installers are to be trained and certified by the manufacturer on the proper installation and adjustment of fire, life safety, and security products including: hanging devices; locking devices; closing devices; and seals.
- B. Mounting Heights: Mount door hardware units at heights indicated in following applicable publications, unless specifically indicated or required to comply with governing regulations:
  - 1. Standard Steel Doors and Frames: DHI's "Recommended Locations for Architectural Hardware for Standard Steel Doors and Frames."
  - 2. DHI TDH-007-20: Installation Guide for Doors and Hardware.
  - 3. Where indicated to comply with accessibility requirements, comply with ANSI A117.1 "Accessibility Guidelines for Buildings and Facilities."
  - 4. Provide blocking in drywall partitions where wall stops or other wall mounted hardware is located.
- C. Retrofitting: Install door hardware to comply with manufacturer's published templates and written instructions. Where cutting and fitting are required to install door hardware onto or into surfaces that are later to be painted or finished in another way, coordinate removal, storage, and reinstallation of surface protective trim units with finishing work specified in Division 9 Sections. Do not install surface-mounted items until finishes have been completed on substrates involved.
- D. Thresholds: Set thresholds for exterior and acoustical doors in full bed of sealant complying with requirements specified in Division 7 Section "Joint Sealants."

DOOR HARDWARE 08 71 00 - 14

E. Storage: Provide a secure lock up for hardware delivered to the project but not yet installed. Control the handling and installation of hardware items so that the completion of the work will not be delayed by hardware losses before and after installation.

#### 3.4 FIELD QUALITY CONTROL

- A. Field Inspection (Punch Report): Reference Division 01 Sections "Closeout Procedures". Produce project punch report for each installed door opening indicating compliance with approved submittals and verification hardware is properly installed, operating and adjusted. Include list of items to be completed and corrected, indicating the reasons or deficiencies causing the Work to be incomplete or rejected.
  - 1. Organization of List: Include separate Door Opening and Deficiencies and Corrective Action Lists organized by Mark, Opening Remarks and Comments, and related Opening Images and Video Recordings.

#### 3.5 ADJUSTING

A. Initial Adjustment: Adjust and check each operating item of door hardware and each door to ensure proper operation or function of every unit. Replace units that cannot be adjusted to operate as intended. Adjust door control devices to compensate for final operation of heating and ventilating equipment and to comply with referenced accessibility requirements.

#### 3.6 CLEANING AND PROTECTION

- A. Protect all hardware stored on construction site in a covered and dry place. Protect exposed hardware installed on doors during the construction phase. Install any and all hardware at the latest possible time frame.
- B. Clean adjacent surfaces soiled by door hardware installation.
- C. Clean operating items as necessary to restore proper finish. Provide final protection and maintain conditions that ensure door hardware is without damage or deterioration at time of owner occupancy.

## 3.7 DEMONSTRATION

A. Instruct Owner's maintenance personnel to adjust, operate, and maintain mechanical and electromechanical door hardware.

### 3.8 DOOR HARDWARE SETS

A. The hardware sets represent the design intent and direction of the owner and architect. They are a guideline only and should not be considered a detailed hardware schedule. Discrepancies, conflicting hardware and missing items should be brought to the attention of the architect with

corrections made prior to the bidding process. Omitted items not included in a hardware set should be scheduled with the appropriate additional hardware required for proper application and functionality.

- 1. Quantities listed are for each pair of doors, or for each single door.
- 2. The supplier is responsible for handing and sizing all products.
- 3. Where multiple options for a piece of hardware are given in a single line item, the supplier shall provide the appropriate application for the opening.
- 4. At existing openings with new hardware the supplier shall field inspect existing conditions prior to the submittal stage to verify the specified hardware will work as required. Provide alternate solutions and proposals as needed.
- B. Manufacturer's Abbreviations:
  - 1. MK McKinney
  - 2. PE Pemko
  - 3. RU Corbin Russwin
  - 4. RO Rockwood
  - 5. RF Rixson

## **Hardware Sets**

### **Set: 1.0**

Doors: E03

| 2 Continuous Hinge               | CFM_SLF-HD1         |       | PE |
|----------------------------------|---------------------|-------|----|
| 1 CVR Exit, NL                   | ED4800 503F ACP     | 630   | RU |
| 1 CVR Exit, EO                   | ED4800 EO           | 630   | RU |
| 2 Offset Pull                    | RM3131-24           | US32D | RO |
| 2 Drop Plate & Mounting Hardware | as required         | 689   | RU |
| 2 Surface Closer w Stop          | DC6210 A11          | 689   | RU |
| 1 Threshold                      | 2005AT x Width      |       | PE |
| 1 Perimeter Seal                 | by door / frame mfg |       |    |
| 2 Sweep                          | 315CN x Width       |       | PE |

Notes: GC to field verify existing conditions for hardware compatibility with door & frame.

#### **Set: 2.0**

Doors: E06

DOOR HARDWARE 08 71 00 - 16

| T4A3386 NRP                 | US32D  | MK   |
|-----------------------------|--|--|
| ED5860 N959ET ACP           | 630  | RU   |
| ED5860 N950ET               | 630  | RU   |
| DC6210 A3                   | 689  | RU   |
| K1050 10" High x LDW CSK    | US32D  | RO   |
| 406 / 409 / 446 as required | US26D  | RO   |
| 2005AT x Width              |  | PE   |
| 2891APK (Head & Jambs)      |  | PE   |
| 346C x LAR                  |  | PE   |
| 3452AV x Width              |  | PE   |
| 303CPK x Height             |  | PE   |
|                             | ED5860 N959ET ACP<br>ED5860 N950ET<br>DC6210 A3<br>K1050 10" High x LDW CSK<br>406 / 409 / 446 as required<br>2005AT x Width<br>2891APK (Head & Jambs)<br>346C x LAR<br>3452AV x Width | ED5860 N959ET ACP ED5860 N950ET 630 DC6210 A3 689 K1050 10" High x LDW CSK 406 / 409 / 446 as required 2005AT x Width 2891APK (Head & Jambs) 346C x LAR 3452AV x Width |

Notes: GC to field verify existing conditions for hardware compatibility with door & frame.

# **Set: 3.0**

Doors: 111B

| 3 Hinge (heavy weight)         | T4A3386 NRP                 | US32D | MK |
|--------------------------------|-----------------------------|-------|----|
| 1 Rim Exit, ST                 | ED5200 N959ET ACP           | 630   | RU |
| 1 Surface Closer w Hvy Dty Arm | DC6210 A3                   | 689   | RU |
| 1 Kick Plate                   | K1050 10" High x LDW CSK    | US32D | RO |
| 1 Door Stop                    | 406 / 409 / 446 as required | US26D | RO |
| 1 Threshold                    | 2005AT x Width              |       | PE |
| 1 Gasketing                    | 2891APK (Head & Jambs)      |       | PE |
| 1 Sweep                        | 315CN x Width               |       | PE |

Notes: GC to field verify existing conditions for hardware compatibility with door & frame.

# **Set: 4.0**

Doors: 114A, 116A

| 3 Hinge (heavy weight)  | T4A3386 NRP              | US32D | MK |
|-------------------------|--------------------------|-------|----|
| 1 Rim Exit, ST          | ED5200 N959ET ACP        | 630   | RU |
| 1 Surface Closer w Stop | DC6210 A11               | 689   | RU |
| 1 Kick Plate            | K1050 10" High x LDW CSK | US32D | RO |
| 1 Threshold             | 2005AT x Width           |       | PE |
| 1 Gasketing             | 2891APK (Head & Jambs)   |       | PE |

| M.A. | No. | H1-2 | 2202 | 29-I | ЭA |
|------|-----|------|------|------|----|
|      |     | RM   | G-   | 220  | 29 |

| 1 Rain Guard | 346C x LAR     | PE |
|--------------|----------------|----|
| 1 Sweep      | 3452AV x Width | PE |

# **Set: 5.0**

Doors: 105A, 118A

| 3 Hinge, Full Mortise | TA2714                      | US26D | MK |
|-----------------------|-----------------------------|-------|----|
| 1 Storeroom Lock      | CLX3357 NZD ACP             | 626   | RU |
| 1 Surface Closer      | DC6200 Reg / PA             | 689   | RU |
| 1 Kick Plate          | K1050 10" High x LDW CSK    | US32D | RO |
| 1 Door Stop           | 406 / 409 / 446 as required | US26D | RO |
| 3 Silencer            | As req'd                    |       | RO |

### **Set: 6.0**

Doors: 103A, 104A, 106A, 108A, 110A, 112A

| 3 Hinge, Full Mortise | TA2714                      | US26D | MK |
|-----------------------|-----------------------------|-------|----|
| 1 Entrance Lock       | CLX3351 NZD ACP             | 626   | RU |
| 1 Door Stop           | 406 / 409 / 446 as required | US26D | RO |
| 3 Silencer            | As req'd                    |       | RO |

# **Set: 7.0**

Doors: E04

| 3 Hinge, Full Mortise | TA2714                      | US26D | MK |
|-----------------------|-----------------------------|-------|----|
| 1 Rim Exit, ST        | ED5200 N959ET ACP           | 630   | RU |
| 1 Surface Closer      | DC6200 Reg / PA             | 689   | RU |
| 1 Kick Plate          | K1050 10" High x LDW CSK    | US32D | RO |
| 1 Door Stop           | 406 / 409 / 446 as required | US26D | RO |
| 3 Silencer            | As req'd                    |       | RO |

Notes: GC to field verify existing conditions for hardware compatibility with door & frame.

# Set: 8.0

Doors: E01, E02, E05

| 3 Hinge, Full Mortise   | TA2714            | US26D | MK |
|-------------------------|-------------------|-------|----|
| 1 Rim Exit, ST          | ED5200 N959ET ACP | 630   | RU |
| 1 Surface Closer w Stop | DC6210 A11        | 689   | RU |

| 1 Kick Plate | K1050 10" High x LDW CSK | US32D | RO |
|--------------|--------------------------|-------|----|
| 3 Silencer   | As req'd                 |       | RO |

Notes: GC to field verify existing conditions for hardware compatibility with door & frame.

# Set: 9.0

| Doors: | 102B, | 114B |
|--------|-------|------|
|--------|-------|------|

| 3 Hinge, Full Mortise | TA2714                      | US26D | MK |
|-----------------------|-----------------------------|-------|----|
| 1 Rim Exit, CL        | ED5200 N955ET ACP           | 630   | RU |
| 1 Surface Closer      | DC6200 Reg / PA             | 689   | RU |
| 1 Kick Plate          | K1050 10" High x LDW CSK    | US32D | RO |
| 1 Door Stop           | 406 / 409 / 446 as required | US26D | RO |
| 3 Silencer            | As req'd                    |       | RO |
|                       |                             |       |    |

# **Set: 9.1**

### Doors: 102A

| 1 Continuous Hinge               | CFM_SLF-HD1                 |       | PE |
|----------------------------------|-----------------------------|-------|----|
| 1 Rim Exit, CL                   | ED5200 N955ET ACP           | 630   | RU |
| 1 Drop Plate & Mounting Hardware | as required                 | 689   | RU |
| 1 Surface Closer                 | DC6200 Reg / PA             | 689   | RU |
| 1 Door Stop                      | 406 / 409 / 446 as required | US26D | RO |

# **Set: 10.0**

| 3 Hinge, Full Mortise | TA2714                      | US26D | MK |
|-----------------------|-----------------------------|-------|----|
| 1 Rim Exit, PA        | ED5200 N910ET               | 630   | RU |
| 1 Surface Closer      | DC6200 Reg / PA             | 689   | RU |
| 1 Kick Plate          | K1050 10" High x LDW CSK    | US32D | RO |
| 1 Door Stop           | 406 / 409 / 446 as required | US26D | RO |
| 3 Silencer            | As req'd                    |       | RO |

# Set: 11.0

# Doors: 115A, 115B

| 3 Hinge, Full Mortise | TA2714          | US26D | MK |
|-----------------------|-----------------|-------|----|
| 1 Classroom Lock      | CLX3355 NZD ACP | 626   | RU |
| 1 Surface Closer      | DC6200 Reg / PA | 689   | RU |

| M.A. | No. | H1- | 220  | 29-I | DΑ |
|------|-----|-----|------|------|----|
|      |     | RN  | lG - | 220  | 29 |

| <ul><li>1 Kick Plate</li><li>1 Door Stop</li><li>3 Silencer</li></ul>  | K1050 10" High x LDW CSK<br>406 / 409 / 446 as required<br>As req'd                                    | US32D<br>US26D  | RO<br>RO<br>RO                         |  |
|--|--|---|--|--|
| Set: 11.1<br>Doors: 107A, 109A   |  |   |  |  |
| <ol> <li>Continuous Hinge</li> <li>Classroom Lock</li> <li>Drop Plate &amp; Mounting Hardware</li> <li>Surface Closer</li> <li>Door Stop</li> </ol>  | CFM_SLF-HD1<br>CLX3355 NZD ACP<br>as required<br>DC6200 Reg / PA<br>406 / 409 / 446 as required        | 626<br>689<br>689<br>US26D                            | PE<br>RU<br>RU<br>RU<br>RO             |  |
| <u>Set: 12.0</u><br>Doors: 119A  |  |   |  |  |
| <ul> <li>3 Hinge (heavy weight)</li> <li>1 Pull Plate</li> <li>1 Push Plate</li> <li>1 Surface Closer</li> <li>1 Kick Plate</li> <li>1 Door Stop</li> <li>3 Silencer</li> </ul>  | T4A3786 BF 110 x 70C 70E DC6200 Reg / PA K1050 10" High x LDW CSK 406 / 409 / 446 as required As req'd | US26D<br>US32D<br>US32D<br>689<br>US32D<br>US26D      | MK<br>RO<br>RO<br>RU<br>RO<br>RO       |  |
| <u>Set: 13.0</u><br>Doors: 120A  |  |   |  |  |
| <ul> <li>3 Hinge (heavy weight)</li> <li>1 Pull Plate</li> <li>1 Push Plate</li> <li>1 Conc Overhead Stop</li> <li>1 Drop Plate &amp; Mounting Hardware</li> <li>1 Surface Closer</li> <li>1 Kick Plate</li> <li>3 Silencer</li> </ul> | T4A3786 BF 110 x 70C 70E 1-X36 as required DC6200 Reg / PA K1050 10" High x LDW CSK As req'd           | US26D<br>US32D<br>US32D<br>630<br>689<br>689<br>US32D | MK<br>RO<br>RO<br>RF<br>RU<br>RO<br>RO |  |

END OF SECTION 087100

DOOR HARDWARE 08 71 00 - 21

This page intentionally left blank

#### SECTION 08 80 00 - GLAZING

#### PART 1 - GENERAL

#### 1.1 GENERAL

A. Manufacture of glass and glazing materials and fabrication and installation of glazing materials, shall meet the requirements of the Safety Standard for Architectural Glazing Materials (16 CFR Part 1201) dated January 6, 1977, effective July 6, 1977 issued by the Consumer Product Safety Commission, and any amendments thereto.

### 1.2 SUMMARY

- A. Section includes glazing for the following products and applications, including those specified in other Sections where glazing requirements are specified by reference to this Section:
  - 1. Door lites.
  - 2. Aluminum Storefront entrances, windows, transoms, etc.
  - 3. Interior aluminum borrowed lites, including butt-joint glazing applications.

#### 1.3 RELATED SECTIONS

- A. Section 08 11 13 "Hollow Metal Doors and Frames" for door lites.
- B. Section 08 14 16 "Flush Wood Doors" for door lites.
- C. Section 08 42 13 "Aluminum Framed Entrances and Storefront".

#### 1.4 PERFORMANCE REQUIREMENTS

- A. Delegated Design: Design glass, including comprehensive engineering analysis according to ASTM E 1300 by a qualified professional engineer, using the following design criteria:
  - 1. Design Wind Pressures: Design for wind load designated for area of Project according to International Building Code, latest edition, and Wind Load Map, Figure 1609B.
  - 2. Differential Shading: Design glass to resist thermal stresses induced by differential shading within individual glass lites.

#### 1.5 ACTION SUBMITTALS

- A. Product Data: For each glass product and glazing material indicated.
- B. Glass Samples: For each type of glass product other than clear monolithic vision glass; 12 inches (300 mm) square.
- C. Glazing Schedule: List glass types and thicknesses for each size opening and location. Use same designations indicated on Drawings.
- D. Delegated-Design Submittal: For glass indicated to comply with performance requirements and design criteria, including analysis data signed and sealed by the qualified professional engineer responsible for their preparation.

GLAZING 08 80 00 - 1

#### 1.6 INFORMATIONAL SUBMITTALS

A. Submit test results from Manufacturer certifying to structural characteristics as specified above...

### 1.7 QUALITY ASSURANCE

- A. Glazing Publications: Comply with published recommendations of glass product manufacturers and organizations below, unless more stringent requirements are indicated. Refer to these publications for glazing terms not otherwise defined in this Section or in referenced standards.
  - 1. GANA Publications: GANA's "Glazing Manual."
  - 2. AAMA Publications: AAMA GDSG-1, "Glass Design for Sloped Glazing," and AAMA TIR-A7, "Sloped Glazing Guidelines."
  - 3. IGMA Publication for Sloped Glazing: IGMA TB-3001, "Guidelines for Sloped Glazing."
  - 4. IGMA Publication for Insulating Glass: SIGMA TM-3000, "North American Glazing Guidelines for Sealed Insulating Glass Units for Commercial and Residential Use."
- B. Safety Glazing Labeling: Where safety glazing labeling is indicated, permanently mark glazing with certification label of the SGCC or the manufacturer. Label shall indicate manufacturer's name, type of glass, thickness, and safety glazing standard with which glass complies.
- C. Insulating-Glass Certification Program: Permanently marked either on spacers or on at least one component lite of units with appropriate certification label of IGCC.

#### 1.8 WARRANTY

- A. Manufacturer's Special Warranty for Coated-Glass Products: Manufacturer's standard form in which coated-glass manufacturer agrees to replace coated-glass units that deteriorate within specified warranty period. Deterioration of coated glass is defined as defects developed from normal use that are not attributed to glass breakage or to maintaining and cleaning coated glass contrary to manufacturer's written instructions. Defects include peeling, cracking, and other indications of deterioration in coating.
  - 1. Warranty Period: 10 years from date of Recommendation of Acceptance.
- B. Manufacturer's Special Warranty on Insulating Glass: Manufacturer's standard form in which insulating-glass manufacturer agrees to replace insulating-glass units that deteriorate within specified warranty period. Deterioration of insulating glass is defined as failure of hermetic seal under normal use that is not attributed to glass breakage or to maintaining and cleaning insulating glass contrary to manufacturer's written instructions. Evidence of failure is the obstruction of vision by dust, moisture, or film on interior surfaces of glass.
  - 1. Warranty Period: 10 years from date of Recommendation of Acceptance.

#### PART 2 - PRODUCTS

# 2.1 GLASS PRODUCTS, GENERAL

- A. Thickness: Where glass thickness is indicated, it is a minimum. Provide glass lites in thicknesses as needed to comply with requirements indicated.
- B. Strength: Where float glass is indicated, provide annealed float glass, Type HS heat-treated float glass, or Type FT heat-treated float glass as needed to comply with "Performance

GLAZING 08 80 00 - 2

Requirements" Article. Where fully tempered glass is indicated, provide Type FT heat-treated float glass.

- C. Thermal and Optical Performance Properties: Provide glass with performance properties specified, as indicated in manufacturer's published test data, based on procedures indicated below:
  - 1. U-Factors: Center-of-glazing values, according to NFRC 100 and based on LBL's WINDOW 5.2 computer program, expressed as Btu/sq. ft. x h x deg F (W/sq. m x K).
  - 2. Solar Heat-Gain Coefficient and Visible Transmittance: Center-of-glazing values, according to NFRC 200 and based on LBL's WINDOW 5.2 computer program.
  - 3. Visible Reflectance: Center-of-glazing values, according to NFRC 300.

### 2.2 SUPPLIER / INSTALLER RESPONSIBILITY

A. It shall be the responsibility of the glass supplier/installer to review the drawings and to provide safety glazing where required by Codes. All safety glazing will have a permanent identification affixed to unit stating that the glass unit meets the requirements of "safety glazing".

### 2.3 GLASS PRODUCTS

- A. Float Glass: ASTM C 1036, Type I, Quality-Q3, Class I (clear) unless otherwise indicated.
- B. Ultraclear Float Glass: ASTM C 1036, Type I, Quality-Q3, Class I, complying with other requirements specified and with visible light transmission not less than 91 percent.
  - 1. Products: Subject to compliance with requirements, provide one of the following:
    - a. AFG Industries, Inc.; Krystal Klear.
    - b. Guardian Industries Corp.; Ultrawhite.
    - c. PPG Industries, Inc.; Starphire.
- C. Heat-Treated Float Glass: ASTM C 1048; Type I; Quality-Q3; Class I (clear) unless otherwise indicated; of kind and condition indicated.
- D. Soft-Coated, Self-Cleaning, Low-Maintenance Glass, Typical at exterior locations: Clear float glass Noted GL-1 with a coating on first surface having both photocatalytic and hydrophilic properties that act to loosen dirt and to cause water to sheet evenly over the glass instead of beading.
  - 1. Products: Subject to compliance with requirements, provide one of the following:
    - a. AFG Industries, Inc.; Spotless Ti-AC.
    - b. Guardian; SunGuard SuperNatrual 68
    - c. PPG Industries, Inc.; Solar Ban 60.
- E. Uncoated Tinted Float Glass: Class 2, complying with other requirements specified.
  - 1. Tint Color: as selected by Architect.

## 2.4 INSULATING GLASS

- A. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
  - a. AFG Industries, Inc.
  - b. Guardian

GLAZING 08 80 00 - 3

- c. PPG Industries, Inc.
- B. Insulating-Glass Units: Factory-assembled units consisting of sealed lites of glass separated by a dehydrated interspace, qualified according to ASTM E 2190, and complying with other requirements specified.
  - 1. All insulated glass units: Low-E coating on Surface #2; thickness of each glass lite =  $\frac{1}{4}$ ".
  - 2. Sealing System: Dual seal; overall thickness = 1"
  - 3. Spacer: Manufacturer's standard spacer material and construction; interspace content = Argon gas.
  - 4. Provide permanent Safety Glazing labeling in locations where required.

#### 2.5 GLAZING GASKETS

- A. Dense Compression Gaskets: Molded or extruded gaskets of profile and hardness required to maintain watertight seal, made from one of the following:
  - 1. Neoprene complying with ASTM C 864.
  - 2. EPDM complying with ASTM C 864.
  - 3. Silicone complying with ASTM C 1115.
  - 4. Thermoplastic polyolefin rubber complying with ASTM C 1115.

### 2.6 GLAZING SEALANTS

#### A. General:

- 1. Compatibility: Provide glazing sealants that are compatible with one another and with other materials they will contact, including glass products, seals of insulating-glass units, and glazing channel substrates, under conditions of service and application, as demonstrated by sealant manufacturer based on testing and field experience.
- 2. Suitability: Comply with sealant and glass manufacturers' written instructions for selecting glazing sealants suitable for applications indicated and for conditions existing at time of installation.
- 3. Sealants used inside the weatherproofing system, shall have a VOC content of not more than 250 g/L when calculated according to 40 CFR 59, Subpart D (EPA Method 24).
- 4. Colors of Exposed Glazing Sealants: As selected by Architect from manufacturer's full range.
- B. Glazing Sealant: Neutral-curing silicone glazing sealant complying with ASTM C 920, Type S, Grade NS, Class 50, Use NT.
  - 1. Product: Dow Corning 795 Silicone Building Sealant or prior approved equal.
    - a. Applications: all.

### 2.7 GLAZING TAPES

A. Back-Bedding Mastic Glazing Tapes: Preformed, butyl-based, 100 percent solids elastomeric tape; non-staining and non-migrating in contact with nonporous surfaces; with or without spacer rod as recommended in writing by tape and glass manufacturers for application indicated; and complying with ASTM C 1281 and AAMA 800 for products indicated below:

GLAZING 08 80 00 - 4

- 1. AAMA 804.3 tape, where indicated.
- 2. AAMA 806.3 tape, for glazing applications in which tape is subject to continuous pressure.
- 3. AAMA 807.3 tape, for glazing applications in which tape is not subject to continuous pressure.
- B. Expanded Cellular Glazing Tapes: Closed-cell, PVC foam tapes; factory coated with adhesive on both surfaces; and complying with AAMA 800 for the following types:
  - 1. AAMA 810.1, Type 1, for glazing applications in which tape acts as the primary sealant.
  - 2. AAMA 810.1, Type 2, for glazing applications in which tape is used in combination with a full bead of liquid sealant.

#### 2.8 MISCELLANEOUS GLAZING MATERIALS

- A. Cleaners, Primers, and Sealers: Types recommended by sealant or gasket manufacturer.
- B. Setting Blocks: Elastomeric material with a Shore, Type A durometer hardness of 85, plus or minus 5.
- C. Spacers: Elastomeric blocks or continuous extrusions of hardness required by glass manufacturer to maintain glass lites in place for installation indicated.
- D. Edge Blocks: Elastomeric material of hardness needed to limit glass lateral movement (side walking).
- E. Cylindrical Glazing Sealant Backing: ASTM C 1330, Type O (open-cell material), of size and density to control glazing sealant depth and otherwise produce optimum glazing sealant performance.

### 2.9 MONOLITHIC-GLASS TYPES (Reference Glass Types on Drawings)

- A. Clear heat-strengthened float glass (interior glass). Noted GL-1 on Drawings.
  - 1. Thickness: ½".
- B. Clear tempered glass (where required by Code for interior lites).
  - 1. Thickness: ½".
  - 2. Provide permanent safety glazing labeling in locations where required.
- C. Clear laminated safety glass (where required by Code for interior lites).
  - 1. Thickness: <sup>1</sup>/<sub>4</sub>".
  - 2. Provide permanent safety glazing labeling in locations where required.

### 2.10 INSULATING-GLASS TYPES

- A. Soft-coated, self-cleaning, low-maintenance coating on Surface #1, tinted insulating glass.
  - 1. Outdoor Lite: self-cleaning, low-maintenance, tinted heat-strengthened float glass, Low-E coating surface #2.
  - 2. Indoor Lite: Heat-strengthened tinted float glass. (Color as selected by Architect)
- B. Soft-coated, self-cleaning, low-maintenance coating on Surface #1, clear insulating glass.

GLAZING 08 80 00 - 5

- M.A. No. H1-22029-DA RMG - 22029
- 1. Outdoor Lite: self-cleaning, low-maintenance coating on Surface #1, clear heat-strengthened clear float glass, Low-E coating surface #2.
- 2. Indoor Lite: Heat-strengthened float glass, clear.

#### **PART 3 - EXECUTION**

### 3.1 GLAZING, GENERAL

- A. Comply with combined written instructions of manufacturers of glass, sealants, gaskets, and other glazing materials, unless more stringent requirements are indicated, including those in referenced glazing publications.
- B. Adjust glazing channel dimensions as required by Project conditions during installation to provide necessary bite on glass, minimum edge and face clearances, and adequate sealant thicknesses, with reasonable tolerances.
- C. Protect glass edges from damage during handling and installation. Remove damaged glass from Project site and legally dispose of off Project site. Damaged glass is glass with edge damage or other imperfections that, when installed, could weaken glass and impair performance and appearance.
- D. Apply primers to joint surfaces where required for adhesion of sealants, as determined by preconstruction testing.
- E. Install setting blocks in sill rabbets, sized and located to comply with referenced glazing publications, unless otherwise required by glass manufacturer. Set blocks in thin course of compatible sealant suitable for heel bead.
- F. Do not exceed edge pressures stipulated by glass manufacturers for installing glass lites.
- G. Provide spacers for glass lites where length plus width is larger than 50 inches (1270 mm).
- H. Provide edge blocking where indicated or needed to prevent glass lites from moving sideways in glazing channel, as recommended in writing by glass manufacturer and according to requirements in referenced glazing publications.

### 3.2 TAPE GLAZING

- A. Position tapes on fixed stops so that, when compressed by glass, their exposed edges are flush with or protrude slightly above sightline of stops.
- B. Install tapes continuously, but not necessarily in one continuous length. Do not stretch tapes to make them fit opening.
- C. Cover vertical framing joints by applying tapes to heads and sills first and then to jambs. Cover horizontal framing joints by applying tapes to jambs and then to heads and sills.
- D. Place joints in tapes at corners of opening with adjoining lengths butted together, not lapped. Seal joints in tapes with compatible sealant approved by tape manufacturer.
- E. Apply heel bead of elastomeric sealant.
- F. Center glass lites in openings on setting blocks and press firmly against tape by inserting dense compression gaskets formed and installed to lock in place against faces of removable stops. Start gasket applications at corners and work toward centers of openings.
- G. Apply cap bead of elastomeric sealant over exposed edge of tape.

GLAZING 08 80 00 - 6

## 3.3 SEALANT GLAZING (WET)

- A. Install continuous spacers, or spacers combined with cylindrical sealant backing, between glass lites and glazing stops to maintain glass face clearances and to prevent sealant from extruding into glass channel and blocking weep systems until sealants cure. Secure spacers or spacers and backings in place and in position to control depth of installed sealant relative to edge clearance for optimum sealant performance.
- B. Force sealants into glazing channels to eliminate voids and to ensure complete wetting or bond of sealant to glass and channel surfaces.
- C. Tool exposed surfaces of sealants to provide a substantial wash away from glass.

### 3.4 CLEANING AND PROTECTION

- A. Protect exterior glass from damage immediately after installation by attaching crossed streamers to framing held away from glass. Do not apply markers to glass surface. Remove nonpermanent labels and clean surfaces.
- B. Protect glass from contact with contaminating substances resulting from construction operations. If, despite such protection, contaminating substances do come into contact with glass, remove substances immediately as recommended in writing by glass manufacturer.
- C. Examine glass surfaces adjacent to or below exterior concrete and other masonry surfaces at frequent intervals during construction, but not less than once a month, for buildup of dirt, scum, alkaline deposits, or stains; remove as recommended in writing by glass manufacturer.
- D. Remove and replace glass that is broken, chipped, cracked, or abraded or that is damaged from natural causes, accidents, and vandalism, during construction period.
- E. Carefully examine edges of insulating glass panels for defects and reject any panels exhibiting abrasion, puncture, or evidence of "misting" or condensation within interstitial space.

END OF SECTION 08 80 00

GLAZING 08 80 00 - 7

This page intentionally left blank

#### SECTION 09 22 16 - NON-STRUCTURAL METAL FRAMING

### PART 1 - GENERAL

#### 1.1 SUMMARY

#### A. Section Includes:

- 1. Non-load-bearing steel framing systems for interior partitions.
- 2. Suspension systems for interior ceilings and soffits.

#### 1.2 ACTION SUBMITTALS

A. Product Data: For each type of product.

#### 1.3 INFORMATIONAL SUBMITTALS

- A. Product Certificates: For each type of code-compliance certification for studs and tracks.
- B. Evaluation reports for [embossed, high-strength steel studs and tracks] [firestop tracks] [post-installed anchors] [and] [power-actuated fasteners].

### 1.4 QUALITY ASSURANCE

A. Code-Compliance Certification of Studs and Tracks: Provide documentation that framing members are certified according to the product-certification program of [the Certified Steel Stud Association] [the Steel Framing Industry Association] [or] [the Steel Stud Manufacturers Association].

### PART 2 - PRODUCTS

### 2.1 PERFORMANCE REQUIREMENTS

- A. Fire-Test-Response Characteristics: For fire-resistance-rated assemblies that incorporate non-load-bearing steel framing, provide materials and construction identical to those tested in assembly indicated, according to ASTM E119 by an independent testing agency.
- B. STC-Rated Assemblies: For STC-rated assemblies, provide materials and construction identical to those tested in assembly indicated on Drawings, according to ASTM E90 and classified according to ASTM E413 by an independent testing agency.

### 2.2 FRAMING SYSTEMS

- A. Framing Members, General: Comply with ASTM C754 for conditions indicated.
  - 1. Steel Sheet Components: Comply with ASTM C645 requirements for steel unless otherwise indicated.
  - 2. Protective Coating: ASTM A653/A653M, G40 (Z120), hot-dip galvanized unless otherwise indicated.
- B. Studs and Tracks: ASTM C645.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
    - a. ClarkDietrich.
    - b. MarinoWARE.
    - c. Or prior approved equal
  - 2. Minimum Base-Steel Thickness: As required by performance requirements for horizontal deflection.
    - a. 20 ga. Typical. Coordinate with details.
    - b. 18 ga. where wall extends to structure, see partition types.
    - c. 16 ga. where wall height exceeds 15'-0"
  - 3. Depth: As indicated on Drawings.
- C. Slip-Type Head Joints: Where indicated, provide[ **one of**] the following:
  - Clip System: Clips designed for use in head-of-wall deflection conditions that provide a positive attachment of studs to tracks while allowing [1-1/2-inch (38-mm)] [2-inch (51-mm)] [2-1/2-inch (64-mm)] [3-inch (76-mm)] < Insert dimension > minimum vertical movement.
    - a. < Double click here to find, evaluate, and insert list of manufacturers and products. >
  - 2. Single Long-Leg Track System: ASTM C645 top track with 2-inch- (51-mm-) deep flanges in thickness not less than indicated for studs, installed with studs friction fit into top track and with continuous bridging located within 12 inches (305 mm) of the top of studs to provide lateral bracing.
  - 3. Double-Track System: ASTM C645 top outer tracks, inside track with 2-inch- (51-mm-) deep flanges in thickness not less than indicated for studs and fastened to studs, and outer track sized to friction-fit over inner track.
  - 4. Deflection Track: Steel sheet top track manufactured to prevent cracking of finishes applied to interior partition framing resulting from deflection of structure above; in thickness not less than indicated for studs and in width to accommodate depth of studs.
    - a. <a href="Oouble click here to find, evaluate,">Oouble click here to find, evaluate, and insert list of manufacturers and products.</a>>

- M.A. No. H1-22029-DA RMG - 22029
- 5. < Double click here to find, evaluate, and insert list of manufacturers and products. >
- D. Flat Strap and Backing Plate: Steel sheet for blocking and bracing in length and width indicated.
  - 1. Minimum Base-Steel Thickness: 0.0329 inch (0.836 mm).
- E. Cold-Rolled Channel Bridging: Steel, 0.0538-inch (1.367-mm) minimum base-steel thickness, with minimum 1/2-inch- (13-mm-) wide flanges.
  - 1. Depth: 1-1/2 inches (38 mm).
  - 2. Clip Angle: Not less than 1-1/2 by 1-1/2 inches (38 by 38 mm), 0.068-inch- (1.72-mm-) thick, galvanized steel.
- F. Cold-Rolled Furring Channels: 0.053-inch (1.34-mm) uncoated-steel thickness, with minimum 1/2-inch- (13-mm-) wide flanges.
  - 1. Depth: As indicated on Drawings.
  - 2. Furring Brackets: Adjustable, corrugated-edge-type steel sheet with minimum uncoated-steel thickness of 0.0329 inch (0.8 mm).
  - 3. Tie Wire: ASTM A641/A641M, Class 1 zinc coating, soft temper, 0.062-inch- (1.59-mm-) diameter wire, or double strand of 0.048-inch- (1.21-mm-) diameter wire.

#### 2.3 AUXILIARY MATERIALS

- A. General: Provide auxiliary materials that comply with referenced installation standards.
  - 1. Fasteners for Steel Framing: Of type, material, size, corrosion resistance, holding power, and other properties required to fasten steel members to substrates.

### PART 3 - EXECUTION

### 3.1 INSTALLATION, GENERAL

- A. Installation Standard: ASTM C754.
  - 1. Gypsum Board Assemblies: Also comply with requirements in ASTM C840 that apply to framing installation.
- B. Install framing and accessories plumb, square, and true to line, with connections securely fastened.
- C. Install supplementary framing, and blocking to support fixtures, equipment services, heavy trim, grab bars, toilet accessories, furnishings, or similar construction.
- D. Install bracing at terminations in assemblies.
- E. Do not bridge building control and expansion joints with non-load-bearing steel framing members. Frame both sides of joints independently.

### 3.2 INSTALLING FRAMED ASSEMBLIES

- A. Install framing system components according to spacings indicated, but not greater than spacings required by referenced installation standards for assembly types.
- B. Where studs are installed directly against exterior masonry walls or dissimilar metals at exterior walls, install isolation strip between studs and exterior wall.
- C. Install studs so flanges within framing system point in same direction.
- D. Install tracks at floors and overhead supports. Extend framing full height to structural supports or substrates above suspended ceilings except where partitions are indicated to terminate at suspended ceilings. Continue framing around ducts that penetrate partitions above ceiling.
  - 1. Slip-Type Head Joints: Where framing extends to overhead structural supports, install to produce joints at tops of framing systems that prevent axial loading of finished assemblies.
  - 2. Door Openings: Screw vertical studs at jambs to jamb anchor clips on door frames; install track section (for cripple studs) at head and secure to jamb studs.
    - a. Install two studs at each jamb unless otherwise indicated.
    - b. Install cripple studs at head adjacent to each jamb stud, with a minimum 1/2-inch (13-mm) clearance from jamb stud to allow for installation of control joint in finished assembly.
    - c. Extend jamb studs through suspended ceilings and attach to underside of overhead structure.
  - 3. Other Framed Openings: Frame openings other than door openings the same as required for door openings unless otherwise indicated. Install framing below sills of openings to match framing required above door heads.
  - 4. Sound-Rated Partitions: Install framing to comply with sound-rated assembly indicated.

### E. Direct Furring:

- 1. Screw to wood framing.
- 2. Attach to concrete or masonry with stub nails, screws designed for masonry attachment, or powder-driven fasteners spaced 24 inches (610 mm) o.c.
- F. Installation Tolerance: Install each framing member so fastening surfaces vary not more than 1/8 inch (3 mm) from the plane formed by faces of adjacent framing.

END OF SECTION 09 22 16

This page intentionally left blank

#### SECTION 09 29 00 - GYPSUM BOARD

### PART 1 - GENERAL

#### 1.1 SUMMARY

#### A. Section Includes:

- 1. Interior gypsum board.
- 2. Tile backing panels.

#### 1.2 ACTION SUBMITTALS

- A. Product Data: For the following:
  - 1. Gypsum board, Type X.
  - 2. Mold-resistant gypsum board.
  - 3. Glass-mat, water-resistant backing board.
  - 4. Cementitious backer units.
  - 5. Interior trim.
  - 6. Joint treatment materials.
  - 7. Laminating adhesive.
  - 8. Sound-attenuation blankets.
  - 9. Acoustical sealant.

#### PART 2 - PRODUCTS

### 2.1 PERFORMANCE REQUIREMENTS

A. Fire-Resistance-Rated Assemblies: For fire-resistance-rated assemblies, provide materials and construction identical to those tested in assembly indicated according to ASTM E119 by an independent testing agency.

### 2.2 GYPSUM BOARD, GENERAL

A. Size: Provide maximum lengths and widths available that will minimize joints in each area and that correspond with support system indicated.

#### 2.3 INTERIOR GYPSUM BOARD

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. American Gypsum.
  - 2. CertainTeed Gypsum.
  - 3. Georgia-Pacific Gypsum LLC.

GYPSUM BOARD 09 29 00 - 1

- 4. <u>National Gypsum Company</u>.
- 5. USG Corporation.
- B. Gypsum Board, Type X: ASTM C1396/C1396M.
  - 1. Thickness: 5/8 inch (15.9 mm).
  - 2. Long Edges: Tapered and featured (rounded or beveled) for prefilling.
  - 3. Mold Resistance: ASTM D3273, score of 10 as rated according to ASTM D3274.
- C. Mold-Resistant Gypsum Board (**Restrooms**): ASTM C1396/C1396M. With moisture- and mold-resistant core and paper surfaces.
  - 1. Core: 5/8 inch (15.9 mm), Type X.
  - 2. Long Edges: Tapered.
  - 3. Mold Resistance: ASTM D3273, score of 10 as rated according to ASTM D3274.

#### 2.4 TILE BACKING PANELS

- A. Cementitious Backer Units: ANSI A118.9 and ASTM C1288 or ASTM C1325, with manufacturer's standard edges.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. CertainTeed Corporation.
    - b. James Hardie Building Products, Inc.
    - c. National Gypsum Company.
    - d. USG Corporation.
  - 2. Thickness: 5/8 inch (15.9 mm).
  - 3. Mold Resistance: ASTM D3273, score of 10 as rated according to ASTM D3274.

#### 2.5 TRIM ACCESSORIES

- A. Interior Trim: ASTM C1047.
  - 1. Material: Galvanized or aluminum-coated steel sheet, rolled zinc, plastic, or paper-faced galvanized-steel sheet.
  - 2. Shapes:
    - a. Cornerbead.
    - b. Bullnose bead.
    - c. LC-Bead: J-shaped; exposed long flange receives joint compound.
    - d. L-Bead: L-shaped; exposed long flange receives joint compound.
    - e. U-Bead: J-shaped; exposed short flange does not receive joint compound.
    - f. Expansion (control) joint.

GYPSUM BOARD 09 29 00 - 2

g. Curved-Edge Cornerbead: With notched or flexible flanges.

### 2.6 JOINT TREATMENT MATERIALS

- A. General: Comply with ASTM C475/C475M.
- B. Joint Tape:
  - 1. Interior Gypsum Board: Paper.
  - 2. Exterior Gypsum Soffit Board: Paper.
  - 3. Glass-Mat Gypsum Sheathing Board: 10-by-10 glass mesh.
  - 4. Tile Backing Panels: As recommended by panel manufacturer.
- C. Joint Compound for Interior Gypsum Board: For each coat, use formulation that is compatible with other compounds applied on previous or for successive coats.
  - 1. Prefilling: At open joints, rounded or beveled panel edges, and damaged surface areas, use setting-type taping compound.
  - 2. Embedding and First Coat: For embedding tape and first coat on joints, fasteners, and trim flanges, use drying-type, all-purpose compound.
    - a. Use setting-type compound for installing paper-faced metal trim accessories.
  - 3. Fill Coat: For second coat, use drying-type, all-purpose compound.
  - 4. Finish Coat: For third coat, use drying-type, all-purpose compound.
  - 5. Skim Coat: For final coat of Level 5 finish, use drying-type, all-purpose compound.
- D. Joint Compound for Tile Backing Panels:
  - 1. Glass-Mat, Water-Resistant Backing Panel: As recommended by backing panel manufacturer.
  - 2. Cementitious Backer Units: As recommended by backer unit manufacturer.

## 2.7 AUXILIARY MATERIALS

- A. Provide auxiliary materials that comply with referenced installation standards and manufacturer's written instructions.
- B. Laminating Adhesive: Adhesive or joint compound recommended for directly adhering gypsum panels to continuous substrate.
- C. Steel Drill Screws: ASTM C1002 unless otherwise indicated.
  - 1. Use screws complying with ASTM C954 for fastening panels to steel members from 0.033 to 0.112 inch (0.84 to 2.84 mm) thick.
  - 2. For fastening cementitious backer units, use screws of type and size recommended by panel manufacturer.

GYPSUM BOARD 09 29 00 - 3

- M.A. No. H1-22029-DA RMG - 22029
- D. Sound-Attenuation Blankets: ASTM C665, Type I (blankets without membrane facing) produced by combining thermosetting resins with mineral fibers manufactured from glass, slag wool, or rock wool.
  - 1. Fire-Resistance-Rated Assemblies: Comply with mineral-fiber requirements of assembly.
- E. Acoustical Sealant: As specified in Section 07 92 19 "Acoustical Joint Sealants."
- F. Thermal Insulation: As specified in Section 07 21 00 "Thermal Insulation."
- G. Vapor Retarder: As specified in Section 07 26 00 "Vapor Retarders."

#### PART 3 - EXECUTION

## 3.1 INSTALLATION AND FINISHING OF PANELS

- A. Examine panels before installation. Reject panels that are wet, moisture damaged, and mold damaged.
- B. Comply with ASTM C840.
- C. Isolate perimeter of gypsum board applied to non-load-bearing partitions at structural abutments. Provide 1/4- to 1/2-inch- (6.4- to 12.7-mm-) wide spaces at these locations and trim edges with edge trim where edges of panels are exposed. Seal joints between edges and abutting structural surfaces with acoustical sealant.
- D. For trim with back flanges intended for fasteners, attach to framing with same fasteners used for panels. Otherwise, attach trim according to manufacturer's written instructions.
- E. Prefill open joints and damaged surface areas.
- F. Apply joint tape over gypsum board joints, except for trim products specifically indicated as not intended to receive tape.
- G. Gypsum Board Finish Levels: Finish panels to levels indicated below and according to ASTM C840:
  - 1. Level 1: Ceiling plenum areas, concealed areas, and where indicated.
  - 2. Level 2: Panels that are substrate for tile.
  - 3. Level 3: Where indicated on Drawings.
  - 4. Level 4: At panel surfaces that will be exposed to view unless otherwise indicated.
    - a. Primer and its application to surfaces are specified in Section 09 91 23 "Interior Painting."
  - 5. Level 5: At surface areas to receive epoxy, gloss, semi-gloss or eggshell paint.
    - a. Primer and its application to surfaces are specified in Section 09 91 23 "Interior Painting."

GYPSUM BOARD 09 29 00 - 4

H. Cementitious Backer Units: Finish according to manufacturer's written instructions.

# 3.2 PROTECTION

- A. Protect installed products from damage from weather, condensation, direct sunlight, construction, and other causes during remainder of the construction period.
- B. Remove and replace panels that are wet, moisture damaged, and mold damaged.

END OF SECTION 09 29 00

GYPSUM BOARD 09 29 00 - 5

540

## SECTION 09 51 13 - ACOUSTICAL PANEL CEILINGS

#### PART 1 - GENERAL

#### 1.1 SUMMARY

- A. Section includes acoustical panels and exposed suspension systems for ceilings.
- B. Products furnished, but not installed under this Section, include anchors, clips, and other ceiling attachment devices to be cast in concrete.

## 1.2 PREINSTALLATION MEETINGS

A. Preinstallation Conference: Conduct conference at Project site.

## 1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Samples: For each exposed product and for each color and texture specified, 6 inches (150 mm) in size.

#### 1.4 DELIVERY, STORAGE, AND HANDLING

- A. Deliver acoustical panels, suspension-system components, and accessories to Project site in original, unopened packages and store them in a fully enclosed, conditioned space where they will be protected against damage from moisture, humidity, temperature extremes, direct sunlight, surface contamination, and other causes.
- B. Before installing acoustical panels, permit them to reach room temperature and a stabilized moisture content.
- C. Handle acoustical panels carefully to avoid chipping edges or damaging units in any way.

## 1.9 MAINTENANCE & EXTRA MATERIALS

- A. Maintenance Instructions: Provide manufacturers standard maintenance and cleaning instructions for all finishes provided.
- B. Extra Materials: Deliver to Owner extra materials described below that match products installed, are packaged with protective covering for storage, and are identified with labels describing contents. Only typical system components are included with attic stock.
  - 1. Acoustical Panels: Full-size units equal to five percent (5 %) of amount installed.
  - 2. Ceiling Suspension System Components: Quantity of each grid and exposed component equal to five percent (5 %) percent of amount installed.

## 1.5 FIELD CONDITIONS

A. Environmental Limitations: Do not install acoustical panel ceilings until spaces are enclosed and weatherproof, wet work in spaces is complete and dry, work above ceilings is complete, and ambient temperature and humidity conditions are maintained at the levels indicated for Project when occupied for its intended use.

1. Pressurized Plenums: Operate ventilation system for not less than 48 hours before beginning acoustical panel ceiling installation.

#### PART 2 - PRODUCTS

## 2.1 PERFORMANCE REQUIREMENTS

- A. Surface-Burning Characteristics: Comply with ASTM E 84; testing by a qualified testing agency. Identify products with appropriate markings of applicable testing agency.
  - 1. Flame-Spread Index: Comply with ASTM E 1264 for Class A materials.
  - 2. Smoke-Developed Index: 50 or less.

# 2.2 ACOUSTICAL PANELS, GENERAL

- A. Source Limitations: Obtain each type of acoustical ceiling panel and supporting suspension system from single source from single manufacturer.
- B. Glass-Fiber-Based Panels: Made with binder containing no urea formaldehyde.
- C. Acoustical Panel Standard: Provide manufacturer's standard panels of configuration indicated that comply with ASTM E 1264 classifications as designated by types, patterns, acoustical ratings, and light reflectances unless otherwise indicated.
- D. Acoustical Panel Colors and Patterns: Match appearance characteristics indicated for each product type.
  - 1. Where appearance characteristics of acoustical panels are indicated by referencing pattern designations in ASTM E 1264 and not manufacturers' proprietary product designations, provide products selected by Architect from each manufacturer's full range that comply with requirements indicated for type, pattern, color, light reflectance, acoustical performance, edge detail, and size.

# 2.3 ACOUSTICAL PANELS (Reference "Room Finish Schedule")

- A. Ceiling "ACT": 3/4" thick mineral fiber, smooth, unperforated face, angled tegular; standard or fire-rated 15/16" exposed grid as required; 24" x 24" panels; Class A, 0-25 Flame Spread per ASTM E84.
  - 1. Armstrong School Zone Fine Fissured #1717
  - 2. USG Ceilings Radar ClimaPlus High NRC/CAC Panels #22523
- B. Classification: Provide panels complying with ASTM E 1264 for type, form, and pattern as follows:
  - 1. Type and Form: Type III, mineral base with painted finish; as indicated on schedule
  - 2. Pattern: As indicated by schedule.

## 2.4 METAL SUSPENSION SYSTEMS, GENERAL

- A. Metal Suspension-System Standard: Provide manufacturer's standard direct-hung metal suspension systems of types, structural classifications, and finishes indicated that comply with applicable requirements in ASTM C 635/C 635M.
  - 1. High-Humidity Finish: Comply with ASTM C 635/C 635M requirements for "Coating Classification for Severe Environment Performance" where high-humidity finishes are indicated.

- B. Attachment Devices: Size for five times the design load indicated in ASTM C 635/C 635M, Table 1, "Direct Hung," unless otherwise indicated. Comply with seismic design requirements.
  - 1. Power-Actuated Fasteners in Concrete: Fastener system of type suitable for application indicated, fabricated from corrosion-resistant materials, with clips or other accessory devices for attaching hangers of type indicated and with capability to sustain, without failure, a load equal to 10 times that imposed by ceiling construction, as determined by testing according to ASTM E 1190, conducted by a qualified testing and inspecting agency.
- C. Wire Hangers, Braces, and Ties: Provide wires complying with the following requirements:
  - 1. Zinc-Coated, Carbon-Steel Wire: ASTM A 641/A 641M, Class 1 zinc coating, soft temper.
  - 2. Stainless-Steel Wire: ASTM A 580/A 580M, Type 304, nonmagnetic.
  - 3. Nickel-Copper-Alloy Wire: ASTM B 164, nickel-copper-alloy UNS No. N04400.
  - 4. Size: Select wire diameter so its stress at three times hanger design load (ASTM C 635/C 635M, Table 1, "Direct Hung") will be less than yield stress of wire, but provide not less than 0.135-inch- (3.5-mm-) diameter wire.

## 2.5 METAL SUSPENSION SYSTEM

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to, the following:
- B. Basis-of-Design Product: Armstrong Prelude XL 15/16". Subject to compliance with requirements, provide product indicated on Drawings or comparable product by one of the following:
  - 1. Armstrong World Industries, Inc.
  - 2. CertainTeed Corp.
  - 3. Chicago Metallic Corporation.
  - 4. USG Interiors, Inc.; Subsidiary of USG Corporation.

#### 2.6 METAL EDGE MOLDINGS AND TRIM

- A. Roll-Formed, Sheet-Metal Edge Moldings and Trim: Type and profile indicated or, if not indicated, manufacturer's standard moldings for edges and penetrations that comply with seismic design requirements; formed from sheet metal of same material, finish, and color as that used for exposed flanges of suspension-system runners.
  - 1. Provide manufacturer's standard edge moldings that fit acoustical panel edge details and suspension systems indicated and that match width and configuration of exposed runners unless otherwise indicated.
  - 2. For circular penetrations of ceiling, provide edge moldings fabricated to diameter required to fit penetration exactly.

#### PART 3 - EXECUTION

## 3.1 EXAMINATION

A. Examine substrates, areas, and conditions, including structural framing to which acoustical panel ceilings attach or abut, with Installer present, for compliance with requirements specified in this and other Sections that affect ceiling installation and anchorage and with requirements

- for installation tolerances and other conditions affecting performance of acoustical panel ceilings.
- B. Examine acoustical panels before installation. Reject acoustical panels that are wet, moisture damaged, or mold damaged.
- C. Proceed with installation only after unsatisfactory conditions have been corrected.

#### 3.2 PREPARATION

A. Measure each ceiling area and establish layout of acoustical panels to balance border widths at opposite edges of each ceiling. Avoid using less-than-half-width panels at borders, and comply with layout shown on reflected ceiling plans.

#### 3.3 INSTALLATION

- A. General: Install acoustical panel ceilings to comply with ASTM C 636/C 636M and seismic design requirements indicated, according to manufacturer's written instructions and CISCA's "Ceiling Systems Handbook."
- B. Suspend ceiling hangers from building's structural members and as follows:
  - 1. Install hangers plumb and free from contact with insulation or other objects within ceiling plenum that are not part of supporting structure or of ceiling suspension system.
  - 2. Splay hangers only where required to miss obstructions; offset resulting horizontal forces by bracing, countersplaying, or other equally effective means.
  - 3. Where width of ducts and other construction within ceiling plenum produces hanger spacings that interfere with location of hangers at spacings required to support standard suspension-system members, install supplemental suspension members and hangers in form of trapezes or equivalent devices.
  - 4. Secure wire hangers to ceiling-suspension members and to supports above with a minimum of three tight turns. Connect hangers directly either to structures or to inserts, eye screws, or other devices that are secure and appropriate for substrate and that will not deteriorate or otherwise fail due to age, corrosion, or elevated temperatures.
  - 5. Do not support ceilings directly from permanent metal forms or floor deck. Fasten hangers to cast-in-place hanger inserts, postinstalled mechanical or adhesive anchors, or power-actuated fasteners that extend through forms into concrete.
  - 6. When steel framing does not permit installation of hanger wires at spacing required, install carrying channels or other supplemental support for attachment of hanger wires.
  - 7. Do not attach hangers to steel deck tabs.
  - 8. Do not attach hangers to steel roof deck. Attach hangers to structural members.
  - 9. Space hangers not more than 48 inches (1200 mm) o.c. along each member supported directly from hangers unless otherwise indicated; provide hangers not more than 8 inches (200 mm) from ends of each member.
  - 10. Size supplemental suspension members and hangers to support ceiling loads within performance limits established by referenced standards and publications.
- C. Secure bracing wires to ceiling suspension members and to supports with a minimum of four tight turns. Suspend bracing from building's structural members as required for hangers, without attaching to permanent metal forms, steel deck, or steel deck tabs. Fasten bracing wires into concrete with cast-in-place or postinstalled anchors.
- D. Install edge moldings and trim of type indicated at perimeter of acoustical ceiling area and where necessary to conceal edges of acoustical panels.

- 1. Screw attach moldings to substrate at intervals not more than 16 inches (400 mm) o.c. and not more than 3 inches (75 mm) from ends, leveling with ceiling suspension system to a tolerance of 1/8 inch in 12 feet (3.2 mm in 3.6 m). Miter corners accurately and connect securely.
- 2. Do not use exposed fasteners, including pop rivets, on moldings and trim.
- E. Install suspension-system runners so they are square and securely interlocked with one another. Remove and replace dented, bent, or kinked members.
- F. Install acoustical panels with undamaged edges and fit accurately into suspension-system runners and edge moldings. Scribe and cut panels at borders and penetrations to provide a neat, precise fit.
  - 1. Arrange directionally patterned acoustical panels as follows:
    - a. As indicated on reflected ceiling plans.
  - 2. For square-edged panels, install panels with edges fully hidden from view by flanges of suspension-system runners and moldings.
  - 3. For reveal-edged panels on suspension-system runners, install panels with bottom of reveal in firm contact with top surface of runner flanges.
  - 4. For reveal-edged panels on suspension-system members with box-shaped flanges, install panels with reveal surfaces in firm contact with suspension-system surfaces and panel faces flush with bottom face of runners.
  - 5. Paint cut edges of panel remaining exposed after installation; match color of exposed panel surfaces using coating recommended in writing for this purpose by acoustical panel manufacturer.

#### 3.4 CLEANING

A. Clean exposed surfaces of acoustical panel ceilings, including trim, edge moldings, and suspension-system members. Comply with manufacturer's written instructions for cleaning and touchup of minor finish damage. Remove and replace ceiling components that cannot be successfully cleaned and repaired to permanently eliminate evidence of damage.

END OF SECTION 09 51 13

#### SECTION 09 65 13 - RESILIENT BASE AND ACCESSORIES

## PART 1 - GENERAL

#### 1.1 SUMMARY

- A. Section Includes:
  - 1. Thermoset-rubber base.

#### 1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Samples: For each exposed product and for each color and texture specified.

## PART 2 - PRODUCTS

#### 2.1 THERMOSET-RUBBER BASE

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. Flexco.
  - 2. <u>Johnsonite</u>; a Tarkett company.
  - 3. Roppe Corporation, USA.
- B. Product Standard: ASTM F1861, Type TS (rubber, vulcanized thermoset), Group I (solid, homogeneous).
  - 1. Style and Location:
    - a. Style A, Straight: Provide in areas with carpet.
    - b. Style B, Cove: Provide in areas with resilient floor coverings.
- C. Thickness: 0.125 inch (3.2 mm).
- D. Height: 4 inches (102 mm).
- E. Lengths: Coils in manufacturer's standard length.
- F. Outside Corners: Preformed.
- G. Inside Corners: Job formed.
- H. Colors: as selected from manufacturers full range.

## 2.2 RUBBER MOLDING ACCESSORY

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. Roppe Corporation, USA.
  - 2. <u>VPI Corporation</u>.
- B. Description: Rubber carpet edge for glue-down applications and transition strips.

#### 2.3 INSTALLATION MATERIALS

- A. Trowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based or blended hydraulic-cement-based formulation provided or approved by resilient-product manufacturer for applications indicated.
- B. Adhesives: Water-resistant type recommended by resilient-product manufacturer for resilient products and substrate conditions indicated.
- C. Floor Polish: Provide protective, liquid floor-polish products recommended by resilient stair-tread manufacturer.

## PART 3 - EXECUTION

#### 3.1 PREPARATION

- A. Prepare substrates according to manufacturer's written instructions to ensure adhesion of resilient products.
- B. Concrete Substrates for Resilient Stair Accessories: Prepare horizontal surfaces according to ASTM F710.
  - 1. Verify that substrates are dry and free of curing compounds, sealers, and hardeners.
  - 2. Remove substrate coatings and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by manufacturer. Do not use solvents.
  - 3. Alkalinity and Adhesion Testing: Perform tests recommended by manufacturer. Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by manufacturer in writing, but not less than 5 or more than 9 pH.
  - 4. Moisture Testing: Perform tests so that each test area does not exceed 200 sq. ft. (18.6 sq. m), and perform no fewer than three tests in each installation area and with test areas evenly spaced in installation areas.
    - a. Anhydrous Calcium Chloride Test: ASTM F1869. Proceed with installation only after substrates have maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. (1.36 kg of water/92.9 sq. m) in 24 hours.
    - b. Relative Humidity Test: Using in-situ probes, ASTM F2170. Proceed with installation only after substrates have a maximum 75 percent relative humidity level measurement.
- C. Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps and ridges to produce a uniform and smooth substrate.

- D. Do not install resilient products until materials are the same temperature as space where they are to be installed.
- E. Immediately before installation, sweep and vacuum clean substrates to be covered by resilient products.

#### 3.2 RESILIENT BASE INSTALLATION

- A. Comply with manufacturer's written instructions for installing resilient base.
- B. Apply resilient base to walls, columns, pilasters, casework and cabinets in toe spaces, and other permanent fixtures in rooms and areas where base is required.
- C. Install resilient base in lengths as long as practical without gaps at seams and with tops of adjacent pieces aligned.
- D. Tightly adhere resilient base to substrate throughout length of each piece, with base in continuous contact with horizontal and vertical substrates.
- E. Do not stretch resilient base during installation.
- F. On masonry surfaces or other similar irregular substrates, fill voids along top edge of resilient base with manufacturer's recommended adhesive filler material.
- G. Preformed Corners: Install preformed corners before installing straight pieces.

## 3.3 RESILIENT ACCESSORY INSTALLATION

- A. Comply with manufacturer's written instructions for installing resilient accessories.
- B. Resilient Molding Accessories: Butt to adjacent materials and tightly adhere to substrates throughout length of each piece. Install reducer strips at edges of floor covering that would otherwise be exposed.

#### 3.4 CLEANING AND PROTECTION

- A. Comply with manufacturer's written instructions for cleaning and protecting resilient products.
- B. Floor Polish: Remove soil, adhesive, and blemishes from resilient stair treads before applying liquid floor polish.
  - 1. Apply three coat(s).
- C. Cover resilient products subject to wear and foot traffic until Substantial Completion.

END OF SECTION 09 65 13

#### SECTION 09 65 19 - RESILIENT TILE FLOORING

## PART 1 - GENERAL

#### 1.1 SUMMARY

- A. Section Includes:
  - 1. Enhanced Vinyl composition floor tile.

# 1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Samples: For each exposed product and for each color and pattern specified.

## 1.3 CLOSEOUT SUBMITTALS

A. Maintenance data.

## 1.4 QUALITY ASSURANCE

A. Installer Qualifications: An entity that employs installers and supervisors who are competent in techniques required by manufacturer for floor tile installation.

#### PART 2 - PRODUCTS

# 2.1 PERFORMANCE REQUIREMENTS

- A. Fire-Test-Response Characteristics: For resilient floor tile, as determined by testing identical products according to ASTM E648 or NFPA 253 by a qualified testing agency.
  - 1. Critical Radiant Flux Classification: Class I, not less than 0.45 W/sq. cm.

# 2.2 ENHANCED VINYL COMPOSITION FLOOR TILE – EVT (LVT)

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - 1. Armstrong World Industries, Inc.
  - 2. Mannington Mills, Inc.
  - 3. Pateraft.
  - 4. Mohawk Commercial.

- B. Tile Standard: ASTM F1066, Class 3, surface pattern.
- C. Wearing Surface: Embossed.
- D. Colors and Patterns: as selected by owner see allowances.

#### 2.3 INSTALLATION MATERIALS

- A. Trowelable Leveling and Patching Compounds: Latex-modified, portland-cement-based or blended hydraulic-cement-based formulation provided or approved by floor tile manufacturer for applications indicated.
- B. Adhesives: Water-resistant type recommended by floor tile and adhesive manufacturers to suit floor tile and substrate conditions indicated.

#### **PART 3 - EXECUTION**

#### 3.1 PREPARATION

- A. Prepare substrates according to floor tile manufacturer's written instructions to ensure adhesion of resilient products.
- B. Concrete Substrates: Prepare according to ASTM F710.
  - 1. Verify that substrates are dry and free of curing compounds, sealers, and hardeners.
  - 2. Remove substrate coatings and other substances that are incompatible with adhesives and that contain soap, wax, oil, or silicone, using mechanical methods recommended by floor tile manufacturer. Do not use solvents.
  - 3. Alkalinity and Adhesion Testing: Perform tests recommended by floor tile manufacturer. Proceed with installation only after substrate alkalinity falls within range on pH scale recommended by manufacturer in writing, but not less than 5 or more than 9 pH.
  - 4. Moisture Testing: Perform tests so that each test area does not exceed 200 sq. ft. (18.6 sq. m), and perform no fewer than three tests in each installation area and with test areas evenly spaced in installation areas.
    - a. Anhydrous Calcium Chloride Test: ASTM F1869. Proceed with installation only after substrates have maximum moisture-vapor-emission rate of 3 lb of water/1000 sq. ft. (1.36 kg of water/92.9 sq. m) in 24 hours.
    - b. Relative Humidity Test: Using in-situ probes, ASTM F2170. Proceed with installation only after substrates have a maximum 75 percent relative humidity level measurement.
- C. Fill cracks, holes, and depressions in substrates with trowelable leveling and patching compound; remove bumps and ridges to produce a uniform and smooth substrate.
- D. Do not install floor tiles until materials are the same temperature as space where they are to be installed.

- M.A. No. H1-22029-DA RMG - 22029
- 1. At least 48 hours in advance of installation, move resilient floor tile and installation materials into spaces where they will be installed.
- E. Immediately before installation, sweep and vacuum clean substrates to be covered by resilient floor tile.

#### 3.2 FLOOR TILE INSTALLATION

- A. Comply with manufacturer's written instructions for installing floor tile.
- B. Lay out floor tiles from center marks established with principal walls, discounting minor offsets, so tiles at opposite edges of room are of equal width. Adjust as necessary to avoid using cut widths that equal less than one-half tile at perimeter.
  - 1. Lay tiles square with room axis in pattern indicated.
- C. Match floor tiles for color and pattern by selecting tiles from cartons in the same sequence as manufactured and packaged, if so numbered. Discard broken, cracked, chipped, or deformed tiles.
  - 1. Lay tiles in pattern of colors and sizes indicated.
- D. Scribe, cut, and fit floor tiles to butt neatly and tightly to vertical surfaces and permanent fixtures including built-in furniture, cabinets, pipes, outlets, and door frames.
- E. Extend floor tiles into toe spaces, door reveals, closets, and similar openings. Extend floor tiles to center of door openings.
- F. Maintain reference markers, holes, and openings that are in place or marked for future cutting by repeating on floor tiles as marked on substrates. Use chalk or other nonpermanent marking device.
- G. Install floor tiles on covers for telephone and electrical ducts, building expansion-joint covers, and similar items in installation areas. Maintain overall continuity of color and pattern between pieces of tile installed on covers and adjoining tiles. Tightly adhere tile edges to substrates that abut covers and to cover perimeters.
- H. Adhere floor tiles to substrates using a full spread of adhesive applied to substrate to produce a completed installation without open cracks, voids, raising and puckering at joints, telegraphing of adhesive spreader marks, and other surface imperfections.

END OF SECTION 09 65 19

#### SECTION 09 91 00 - PAINTING

#### PART 1 - GENERAL

#### 1.1 DESCRIPTION OF WORK

- A. The extent of painting work is shown on drawings and schedules, and as herein specified.
- B. The work includes painting and finishing of interior and exterior exposed items and surfaces throughout project, except as otherwise indicated.
  - 1. Surface preparation, priming and coats of paint specified are in addition to shop-priming and surface treatment specified under other sections of work.
- C. The work includes field painting of exposed bare and covered pipes and ducts (including color coding), and of hangers, exposed steel and iron work, and primed metal surfaces of equipment installed under mechanical and electrical work, except as otherwise indicated.
- D. "Paint" as used herein means all coating systems materials, including primers, emulsions, enamels, stains, sealers and fillers, and other applied materials whether used as prime intermediate or finish coats.
- E. Paint exposed surfaces whether or not colors are designated in "schedules", except where natural finish of material is specifically noted as a surface not to be painted. Where items or surfaces are not specifically mentioned, paint same as adjacent similar materials or areas. If color or finish is not designated, Architect will select these from standard colors available for materials systems specified.
- F. Pre-Finished Items: Unless otherwise indicated, do not include painting when factory-finishing or installer finishing is specified for such items as (but not limited to) toilet enclosures, prefinished partition systems, acoustic materials, architectural woodwork and casework, finished mechanical and electrical equipment including light fixtures, switchgear and distribution cabinets, elevator entrance frames, doors and equipment.
- G. Concealed Surfaces: Unless otherwise indicated, such as back priming of all exterior wood trim, painting is not required on surfaces such as walls or ceiling in concealed areas and generally inaccessible areas, foundation spaces, furred areas, utility tunnels, pipe spaces, duct shafts and elevator shafts.
- H. Finished Metal Surfaces: Metal surfaces of anodized aluminum, stainless steel, chromium plate, copper, bronze and similar finished materials will not require finish painting, unless otherwise indicated.
- I. Operating Parts and Labels: Moving parts of operating units, mechanical and electrical parts, such as valve and damper operators, linkages, sinkages, sensing devices, motor and fan shafts will not require finish painting, unless otherwise indicated. Do not paint over any code-required labels, such as Underwriters Laboratories and Factory Mutual, or any equipment identification, performance rating, name or nomenclature panels.
- J. Refer to Section 03 35 43 Polished Concrete for staining and finishing of concrete floor.

#### 1.2 SUBMITTALS

- A. Product Data: Submit manufacturer's technical information including paint label analysis and application instructions for each material proposed for use.
- B. Schedule of painting systems for each condition and substrate material.
- C. Samples: Submit samples for Architect's review of color and texture only. Provide a listing of material and application for each coat of each finish sample. Include indication of all coats specified.
  - 1. On 12"x 12" representative of actual material to be painted, provide two samples of each color and material, with texture to simulate actual conditions. Resubmit samples as requested by Architect until acceptable sheen, color, and texture is achieved.
- D. Coating Maintenance Manual: Upon conclusion of the project, the Contractor or paint manufacturer/supplier shall furnish a coating maintenance manual, such as Sherwin-Williams "Custodian Project Color and Product Information" report or equal. Manual shall include an Area Summary with finish schedule, Area Detail designating where each product/color/finish was used, product data pages, Material Safety Data Sheets, care and cleaning instructions, touchup procedures, and color samples of each color and finish used.

## 1.3 DELIVERY AND STORAGE

- A. Deliver materials to job site in original, new and unopened packages and containers bearing manufacturer's name and label, and following information.
- B. Name or title of material. Color name and number.

## 1.4 JOB CONDITIONS

- A. Apply water-base paints only when temperature of surfaces to be painted and surrounding air temperatures are between 50 degrees F (10 degrees C.) And 90 Degrees F (32 degrees C.) Unless otherwise permitted by paint manufacturer's printed instructions.
- B. Apply solvent-thinned paints only when temperature of surfaces to be painted and surrounding air temperatures are between 45 degrees F (7 degrees C.) And 95 degrees F (35 degrees C.), unless otherwise permitted by paint manufacturer's printed instruction.
- C. DO NOT APPLY PAINT in snow, rain, fog or mist; or when relative humidity exceeds 85% or damp or wet surfaces: Unless otherwise permitted by paint manufacturer's printed instructions. Painting may be continued during inclement weather if areas and surfaces to be painted are closed-in and heated within temperature limits specified by paint manufacturer during application and drying periods.
- D. Paint Coordination: Provide finish coats which are compatible with prime paints used. Review other sections of these specification in which prime paints are to be provided to ensure compatibility of total coatings system for various substrates. Upon request from other trades, furnish information on characteristics of finish materials proposed for use, to ensure compatible prime coats are used. Provide barrier coats over incompatible primers or remove and re-prime as required. Notify Architect in writing of any anticipated problems using specified coating systems with substrates primed by others.

#### PART 2 - PRODUCTS

# 2.1 MATERIAL QUALITY

- A. Provide best quality grade of various types of coating as regularly manufactured by acceptable paint materials manufacturers. Materials not displaying manufacturer's identification as a standard, best-grade product will not be acceptable.
  - 1. Proprietary names used to designate colors or materials are not intended to imply that products of named manufacturers are required to exclusion of equivalent products of other manufacturers.
  - 2. Federal Specifications establish minimum acceptable quality for paint materials. Provide written certification from paint manufacturer that materials provided meet or exceed these minimums.
  - 3. Manufacturer's products which comply with coating qualitative requirements of applicable Federal Specifications, yet differ in quantitative requirements, may be considered for use when acceptable to Architect. Furnish material data and manufacturer's certificate of performance to Architect for any proposed substitutions.
- B. Provide undercoat paint produced by same manufacturer as finish coats. Use only thinners approved by paint manufacturer, and use only within recommended limits.
- C. Paint Coordination: Provide finish coats which are compatible with prime
- D. A "Schedule of Paint Systems" is at the end of this section.

# 2.2 MAINTENANCE MATERIALS

A. Provide minimum of one unopened gallon, for each type and color of paint used in the project.

## PART 3 -EXECUTION

3.1 INSPECTION: Applicator must examine areas and conditions under which painting work is to be applied and notify Contractor in writing of conditions detrimental to proper and timely completion of work. Do not proceed with work until unsatisfactory conditions have been corrected in a manner acceptable to Applicator's acceptance of surfaces and conditions within any particular area. Do not paint over dirt, rust, scale, grease, moisture, scuffed surfaces, or conditions otherwise detrimental to formation of a durable, homogeneous paint film.

#### 3.2 SURFACE PREPARATION

- A. General: Perform preparation and cleaning procedures in accordance with paint manufacturer's instructions and as herein specified for each particular substrate condition. Remove hardware, hardware accessories, machined surfaces, plates, lighting fixtures, and similar items in place and not to be finish-painted, or provide surface-applied protection sheeting prior to surface preparation and painting operations. Remove, if necessary, items within space for unencumbered painting of each space or area; reinstall removed items when operations are complete. Clean surfaces to be painted before applying paint or surface treatments. Remove oil and grease prior to mechanical cleaning. Program cleaning so that will not fall onto wet, newly-painted surfaces.
- B. Cementitious Materials: Prepare Cementitious surfaces of concrete and fiber-cement board to be painted by removing efflorescence, chalk, dust, dirt, grease, oils and by roughening as required to remove glaze. Determine alkalinity and moisture content of surfaces to be painted by performing appropriate tests. If surfaces are found to be sufficiently alkaline to cause blistering and burning of finish paint, correct this condition before application of paint.

- Do not paint over surfaces where moisture content exceeds that permitted in manufacturer's printed directions.
- C. Clean concrete floor surfaces scheduled as "Sealed Concrete" with commercial solution of muriatic acid, or other etching cleaner. Flush floor with clean water to neutralize acid, and allow to thoroughly dry before painting.
- Prepare concrete floor surface scheduled as "Painted Concrete"

  For surface preparation, refer to SSPC-SP13/NACE 6, or ICRI No. 310.2R, CSP 1-3.

  Surfaces should be thoroughly clean and dry. Concrete and mortar must be cured at least 28 days @ 75°F (24°C). Remove all loose mortar and foreign material. Surface must be free of laitance, concrete dust, dirt, form release agents, moisture curing membranes, loose cement and hardeners. Fill bug holes, air pockets and other voids with Steel-Seam FT910. Primer required.

## E. Wood:

- 1. Clean wood surfaces to be painted of dirt, oil, or other foreign substances with scrapers, mineral spirits, and sandpaper, as required.
- 2. Sandpaper smooth those finished surfaces exposed to view, and wipe dust from surface. Scrape and clean small, dry, seasoned knots and apply a thin coat of while shellac or other recommended knot sealer, before application of priming coat. After priming, fill holes and imperfections in finish surfaces with putty or plastic woodfiller. Sandpaper smooth when dried.
- 3. Prime, stain or seal wood required to be job-painted immediately upon delivery to job. Prime edges, ends, faces-undersides, and backsides of such wood, including cabinets, counters, cases, paneling.
- 4. When transparent finish is required, use spar varnish for back-priming.
- 5. Back-prime paneling on interior partitions only where masonry plaster, or other wet wall construction occurs on backside.
- 6. Seal tops, bottoms, and cut-outs of un-primed wood doors with a heavy coat of varnish or equivalent sealer immediately upon delivery to job.
- F. Galvanized Surfaces: Clean surfaces with non-petroleum based solvent to be free of oil, laitence and surface contamination.

#### 3.3 MATERIALS PREPARATION

- A. Mix and prepare painting materials in accordance with manufacturer's directions.
- B. Store materials which are not in actual use in tightly covered containers. Maintain used containers in storage. Mixing of paint must be in a clean condition, free of foreign materials and residue.
- C. Stir materials before application to produce a mixture of uniform density, and stir as required during application. Do not stir surface film into material. Remove film if necessary and strain material before using.

#### 3.4 APPLICATION

- A. General: Apply paint in accordance with manufacturer's directions. Use applicators and techniques best suited for substrate and type of material being applied.
  - 1. Apply additional coats when undercoats, stains or other conditions show through final coat of paint, until paint film is of uniform finish, color and appearance. Give special attention to insure that surfaces, including edges, corners, crevices, welds, and exposed fasteners receive a dry film thickness equivalent to that of flat surfaces.
  - 2. Paint surfaces behind movable equipment and furniture same as similar exposed surfaces. Paint surfaces behind permanently-fixed equipment or furniture with prime coat only before final installation of equipment.
  - 3. Paint interior surfaces of ducts, where visible through registers or grilles, with flat, non-specular black paint.

- M.A. No. H1-22029-DA RMG - 22029
- 4. Paint access panels, and removable or hinged covers to match exposed surfaces.
- 5. Finish exterior doors on tops, bottoms and side edges same as exterior faces, unless otherwise indicated.
- Sand lightly between each succeeding enamel or varnish coat. 6.
- Omit first coat (primer) on metal surfaces which have been shop-primed and touch-up 7. painted, unless otherwise indicated.
- В. Scheduling Painting: Apply first-coat material to surfaces that have been cleaned, pretreated or otherwise prepared for painting as soon as practicable after preparation and before subsequent surface deterioration.
- Allow sufficient time between successive coatings to permit proper drying. Do not re-coat C. until paint has dried to where it feels firm, does not deform or feel sticky under moderate thumb pressure, and application of another coat of paint does not cause lifting or loss of adhesion of the undercoat.
- Minimum Coating Thickness: Apply materials at not less than manufacturer's recommended D. spreading rate, to establish a total dry film thickness as indicated or, if not indicated, as recommended by coating manufacturer.
- Prime Coats: Apply prime coat of material which is required to be painted or finished, and E. which has not been prime coated by others.
- Re-coat primed and sealed surfaces where there is evidence of suction spots or unsealed F. areas in first coat to assure a finish coat with no burn-through or other defects due to insufficient sealing.
- Pigmented (Opaque) Finishes: Completely cover to provide an opaque, smooth surface of G. uniform finish, color, appearance and coverage. Cloudiness, color spotting, holidays, laps, brush marks, runs, sags, ropiness or other surface imperfections will not be acceptable.
- H. Transparent (Clear) Finishes: Use multiple coats to produce glass-smooth surface film of even luster. Provide a finish free of laps, cloudiness, color irregularity, runs brush marks, orange peel, nail holes, or other surface imperfections.
- I. Completed Work: Match approved samples for color, texture, and coverage. Remove, refinish or repaint work not in compliance with specified requirements.
- J. Stained or clear sealed finished wood:
  - Apply sanding sealer to raise grain and sand smooth to produce flat and smooth 1. surface.
  - 2. Apply successive coats of polyurethane varnish, allow to completely dry before light sanding to produce smooth, flat surface, free of sanding marks, scratches, etc.

#### 3.5 FIELD QUALITY CONTROL

The right is reserved by Owner to invoke material testing procedure at any time, and any A. number of times during period of field painting.

#### 3.6 **CLEAN-UP AND PROTECTION**

- Clean-Up: During progress of work remove from site discarded paint materials, rubbish, A. cans and rags at end of each work day.
- Upon completion of painting work, clean window glass and other paint-spattered surfaces. В. Remove spattered paint by proper methods of washing and scraping using care not to scratch or otherwise damage finished surfaces.
- C. Protection: Protect work of other trades, whether to be painted or not, against damage by painting and finishing or work. Correct any damage by cleaning, repairing or replacing, and repainting, as acceptable to Architect.
- D. Provide "Wet Paint" signs as required to protect newly-painted finishes. Remove temporary protective wrappings provided by others for protection of their work, after completion of painting operations.
- At the completion of work of other trades, touch-up and restore all damaged or defaced E.

09 91 00 - 5 **PAINTING** 

painted surfaces.

## 3.7 EXTERIOR PAINT SCHEDULE

- A. General: Provide the following paint systems for the various substrates, as indicated.
- B. Basis-of-Design Product: Subject to compliance with requirements, provide product indicated or comparable product by one of the following:
  - 1. Sherwin-Williams (S-W)
  - 2. Pittsburg Paints (PPG).
  - 3. Master Coating Technologies, Inc. (Scuffmaster)
- C. FERROUS GALVANIZED METAL (Door Frames & Miscellaneous Metal): High Performance Urethane Modified Alkyd Semi-Gloss Finish: 2 Finish coats over primer.
  - 1st Coat: S-W: Pro Industrial Pro-Cryl Universal Acrylic Primer, B66W00310.
    PPG: Pitt-Tech Plus Interior Exterior DTM Industrial Primer, 90-912
  - 2<sup>nd</sup> Coat: S-W: Pro Industrial Waterbased Alkyd Urethane SG Finish, B53-1150 Series. PPG: SpeedHide Int/Ext Urethane Modified Gloss Oil, 6-270 Series.
  - 3<sup>rd</sup> Coat: S-W: Pro Industrial Waterbased Alkyd Urethane SG Finish, B53-1150 Series. PPG: SpeedHide Int/Ext Urethane Modified Gloss Oil, 6-270 Series.

#### D. EXPOSED EXTERIOR BEAMS and/or COLUMNS

High Performance Acrylic Semi-Gloss Finish 2 finish coats over primer.

1st Coat: S-W: Pro-Cryl Universal Primer, B66W310.

PPG: Pitt-Tech Plus Interior Exterior DTM Industrial Primer, 90-912

2<sup>nd</sup> Coat: S-W: SherCryl HPA Acrylic S.G. Finish, B66W351.

PPG: Pitt-Tech Plus S/G, 90-1210 Series.

3rd Coat: S-W: SherCryl HPA Acrylic S.G. Finish, B66W351.

PPG: Pitt-Tech Plus S/G, 90-1210 Series.

# E. CONCRETE MASONRY UNITS/FACE BRICK

1<sup>ST</sup> Coat: S-W Loxon Masonry Primer, LX2W50

PPG: PermaCrete High Build 100% Acrylic Primer, 4-2 Series

2nd Coat: S-W Superpaint Exterior Latex Acrylic, Satin, A89 Series.

PPG: Sun Proof Exterior 100% Acrylic Satin – 76-45Xi

3rd Coat: S-W Superpaint Exterior Latex Acrylic, Satin, A89 Series.

PPG: Sun Proof Exterior 100% Acrylic Satin – 76-45Xi

#### 3.8 INTERIOR PAINT SCHEDULE

General: Provide the following paint systems for the various interior substrates:

## A. DRYWALL (Typical)

1st Coat: S-W: ProMar 200 Zero VOC Interior Latex Primer, B28W02600

PPG: Speedhide 6 Series Acrylic Primer

2nd Coat: S-W: Pro Industiral Waterbased Catalyzed Epoxy Eg-Shel, B73-360 Series

PPG: Pitt-Glaze 16-901/902 Series Acrylic Epoxy

3rd Coat: S-W: Pro Industiral Waterbased Catalyzed Epoxy Eg-Shel, B73-360 Series

# PPG: Pitt-Glaze 16-901/902 Series Acrylic Epoxy

B. DRYWALL (Corridors) Two-component polyurethane-fortified coating and cross-linker.

1st Coat: Primer - Scuffmaster Primemaster Primer/Sealer; sand after dry, and wipe all sanding dust from surface.

S-W: ProMar 200 Zero VOC Interior Primer, B28W2600 Series

2<sup>nd</sup> Coat: Base - Scuffmaster Premium-Coat 200 (color as selected) S-W: Pro Industrial PreCatalyzed Waterbased Epoxy Eg-Shel,

K45W00151 3<sup>rd</sup> Coat: Finish - Scuffmaster ScrubTough (color as selected) applied to match finish

> standard; eggshell finish. S-W: Pro Industiral Waterbased Catalyzed Epoxy Eg-Shel, B73-360 Series

# C. CONCRETE MASONRY UNITS with

Epoxy Finish over block filler/primer

1<sup>st</sup> Coat - Block Filler/primer:

S-W: Prep-Rite Interior/Exterior Block Filler B25W25 PPG:Aquapon 97-685 (20.0 mils WFT)

2<sup>nd</sup> Coat: S-W: Water Based Catalyzed Epoxy S/G B70 Series PPG: Aquapon 97-130

3<sup>rd</sup> Coat: S-W: Water Based Catalyzed Epoxy S/G B70 Series PPG: Aquapon 97-130

#### D. FERROUS GALVANIZED METAL

High Performance Acrylic Semi-Gloss Finish over primer.

1st Coat: S-W: Pro-Cryl Universal Primer, B66W310.

PPG: Pitt-Tech Plus Interior Exterior DTM Industrial Primer, 90-912

2<sup>nd</sup> Coat: S-W: Pro Industrial Waterbased Alkyd Urethane SG Finish, B53-1150 Series.

PPG: SpeedHide Int/Ext Urethane Modified Gloss Oil, 6-270 Series.

3<sup>rd</sup> Coat: S-W: Pro Industrial Waterbased Alkyd Urethane SG Finish, B53-1150 Series. PPG: SpeedHide Int/Ext Urethane Modified Gloss Oil, 6-270 Series.

#### E. INTERIOR WOOD - PAINTED

Water-Based Eg Shell Epoxy Finish: 2 Finish coats over primer.

1st Coat: S-W: Premium Wall & Wood Primer, B28W8111

PPG: Seal Grip Primeline Wood Undercoat, 17-9517.

2<sup>nd</sup> Coat: S-W: Pro Industiral Waterbased Catalyzed Epoxy Eg-Shel, B73-360 Series

PPG: Pitt-Glaze 16-901/902 Series Acrylic Epoxy

3<sup>rd</sup> Coat: S-W: Pro Industiral Waterbased Catalyzed Epoxy Eg-Shel, B73-360 Series PPG: Pitt-Glaze 16-901/902 Series Acrylic Epoxy

## F. CONCRETE FLOOR SEALER

(opaque where noted "Sealed Concrete on Finish Schedule"):

1st Coat: S-W: H & C Shield-Crete Solid Pigmented 2-Part Epoxy Floor Coating.

PPG: AquaPon WB Solid Pigmented 2-Part Epoxy.

2<sup>nd</sup> Coat: S-W: H & C Shield-Crete Solid Pigmented 2-Part Epoxy Floor Coating.

PPG: AquaPon WB Solid Pigmented 2-Part Epoxy.

END OF SECTION 09 91 00

PAINTING 09 91 00 - 7

562

#### SECTION 10 14 26 SIGNAGE AND GRAPHICS

#### PART 1 GENERAL

#### 1.01 SECTION INCLUDES

A. ADA compliant signage with raised copy, Grade II Braille and raised numbers, text and pictograms.

#### 1.02 SUBMITTALS

- A. Manufacturers must submit 3 references showing comparable products for projects completed within the last 5 years.
- B. Manufacturer's Data: Submit manufacturer's descriptive literature and specifications, including color samples of material for selection, as applicable for approval.
- C. Submit shop drawings listing sign styles, lettering and locations, and overall dimensions of each sign.
- D. Submit full size sample sign of type, style and color specified including method of attachment.
- E. Submit manufacturer's standard warranty information.

#### 1.03 REFERENCES

A. Signage package shall meet the requirements of the Americans with Disabilities Act - 2010 Accessibility Guidelines for Buildings and Facilities.

#### PART 2 PRODUCTS

#### 2.01 MANUFACTURER

- A. Acceptable Manufacturers:
  - 1. Best Manufacturing Sign Systems; Phone 1-800-235-2378; Fax 1-970-249-0223; <a href="https://www.bestsigns.com">www.bestsigns.com</a>
  - 2. Mohawk Sign Systems, Inc.; Phone 518-842-5303; Fax 518-842-5306; <a href="https://www.mohawksign.com">www.mohawksign.com</a>
  - 3. ASI; Phone (214) 352-9140; Fax (214) 352-9741; www.asisignage.com

## 2.02 PRODUCTS

- A. Best HC300 ADA System
- B. Mohawk Series 200A
- C. ASI EmBoss

## 2.03 FABRICATION

- A. ADA-Compliant System with the following characteristics:
  - 1. Tactile characters/symbols shall be raised 1/32 inch from sign plate face. Signs shall be of one-piece construction.
  - 2. Text shall be accompanied by Grade 2 braille.

SIGNAGE 10 14 26 - 1

564

M.A. No. H1-22029-DA

RMG - 22029

- 3. Raised inset border with 1/2" radius corners.
- 4. All letters, numbers and/or symbols shall contrast with their background either light characters on a dark background or dark characters on a light background. Characters and background matte finish.

#### 2.04 SIGNAGE

- A. Plaque material: Melamine plastic laminate, approximately 1/8" thick (1/4" thick for slot signs), with background painted a contrasting color and rated non-static, fire-retardant and self-extinguishing. Plastic laminate will be impervious to most acids, alkalies, alcohol, solvents, abrasives and boiling water.
- B. Lettering style: Upper case, sans serif or simple serif typeface as selected by Architect.
- C. Sizes of letters and numbers shall be as follows:
  - 1. Room numbers shall be 5/8" high.
  - 2. Lettering for room usage and directional identification shall be 5/8" high.
  - 3. Lettering for restroom identification shall be 5/8" high, corresponding symbols shall be 3" high.
- D. Letters and numbers shall be left offset on sign.
- E. Grade 2 braille shall be placed directly below last line of letters or numbers, except for room number signs, where they shall be placed directly behind the last number.
- F. Provide one at each door opening, each restroom, and one "Exit" sign at each exterior door and stair door.

#### 2.05 SIGN TYPES

- A. Typical restroom signs with ADA symbol.
- B. Typical room identification signs with numerals and names.
- C. Typical low-level "Exit" & "Stairs" signs.

## PART 3 EXECUTION

## 3.01 EXAMINATION

A. Examine installation areas to insure that proper conditions exist for timely completion of installation.

## 3.02 PREPARATION

- A. Verify that mounting locations and height for each sign will comply with ADA Accessibility Guidelines.
- B. Mounting locations should be smooth and free of all dirt, dust, grease, etc.

## 3.03 INSTALLATION

A. Mount signs level and plumb using manufacturer's recommended standard mounting (no exposed fasteners) utilizing combination of vinyl foam tape <u>and</u> clear structural silicone sealant.

SIGNAGE 10 14 26 - 2

565

B. Remove excess adhesives, etc. from exposed sign surfaces using method recommended by adhesive manufacturer. Clean sign surfaces.

END OF SECTION 10 14 26

SIGNAGE 10 14 26 - 3

566

## SECTION 10 28 00 - TOILET, BATH, AND LAUNDRY ACCESSORIES

## **PART 1 - GENERAL**

#### 1.1 SUMMARY

- A. Section Includes:
  - 1. Public-use washroom accessories.
  - 2. Underlayatory guards.
  - 3. Custodial accessories.

#### 1.2 ACTION SUBMITTALS

A. Product Data: For each type of product.

## 1.3 INFORMATIONAL SUBMITTALS

A. Sample warranties.

#### 1.4 CLOSEOUT SUBMITTALS

A. Maintenance data.

#### 1.5 WARRANTY

- A. Manufacturer's Special Warranty for Mirrors: Manufacturer agrees to repair or replace mirrors that fail in materials or workmanship within specified warranty period.
  - 1. Warranty Period: 15 years from date of Substantial Completion.

## **PART 2 - PRODUCTS**

# 2.1 PERFORMANCE REQUIREMENTS

- A. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
- B. Structural Performance: Design accessories and fasteners to comply with the following requirements:
  - 1. Grab Bars: Installed units are able to resist 250 lbf concentrated load applied in any direction and at any point.

## 2.2 PUBLIC-USE WASHROOM ACCESSORIES

- A. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - a. American Specialties, Inc.
  - b. <u>Bobrick Washroom Equipment, Inc.</u>
  - c. Bradley Corporation.
  - d. Or equal approved prior to bid.
- B. Toilet Tissue (Roll) Dispenser **TPD**:
  - 1. American Specialties Model 0042.
  - 2. Bobrick Model B-2890
  - 3. Description: single 10" roll dispenser.
  - 4. Mounting: Surface mounted.
  - 5. Noncontrol delivery with standard spindle.
  - 6. Material and Finish: Stainless Satin.
- C. Combination Towel (Folded) Dispenser/Waste Receptacle PTW:
  - 1. American Specialties Model 04697-4
  - 2. Bobrick Model B-3942
  - 3. Description: Combination unit for dispensing C-fold or multifold towels.
  - 4. Mounting: Semirecessed.
    - a. Designed for minimum 2-inch wall depth.
  - 5. Minimum Towel-Dispenser Capacity: 600 C-fold or 800 multifold paper towels.
  - 6. Minimum Waste-Receptacle Capacity: 12 gal..
  - 7. Material and Finish: Stainless steel, ASTM A480/A480M No. 4 finish (satin).
  - 8. Lockset: Tumbler type for towel-dispenser compartment and waste receptacle.
  - 9. Lockset: Tumbler type.
  - 10. Refill Indicator: Window type.
- D. Automatic Soap Dispenser **SD**:
  - 1. American Specialties Model 0360
  - 2. Bobrick Model B-2012
  - 3. Description: Automatic dispenser with infrared sensor to detect presence of hands; battery powered; designed for dispensing soap in liquid or lotion lather form.
  - 4. Mounting: Surface mounted.
  - 5. Capacity: min. 30 oz.
  - 6. Materials: stainless steel.
  - 7. Refill Indicator: LED indicator.
  - 8. Low-Battery Indicator: LED indicator.
- E. Grab Bar **AGB**:
  - 1. American Specialties Model: 3800
  - 2. Bobrick Model: B6806
  - 3. Mounting: Flanges with concealed fasteners.
  - 4. Material: Stainless steel, 0.05 inch thick.
    - a. Finish: Smooth, ASTM A480/A480M No. 4 finish (satin).
  - 5. Outside Diameter: 1-1/2 inches.
  - 6. Configuration and Length: 42" at side, 36" at rear.
- F. Mirror Unit **MR1**:
  - 1. Frame: Stainless steel channel.

- a. Corners: Welded and ground smooth.
- 2. Size: 18"W x 36"H, unless noted otherwise.

# 2.3 UNDERLAVATORY GUARDS

#### A. Underlayatory Guard:

- 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
  - a. Buckaroos, Inc.
  - b. Plumberex Specialty Products, Inc.
  - c. Truebro by IPS Corporation.
- 2. Description: Insulating pipe covering for supply and drain piping assemblies that prevents direct contact with and burns from piping; allow service access without removing coverings.
- 3. Material and Finish: Antimicrobial, molded plastic, white.
- 4. Location: Provide at all wall mounted sinks and lavatories or where piping is exposed to direct contact.

#### 2.4 CUSTODIAL ACCESSORIES

- A. Custodial Mop and Broom Holder Rack **MHR**:
  - 1. Description: Unit with shelf, hooks, holders, and rod suspended beneath shelf.
  - 2. Length: 26 inches.
  - 3. Mop/Broom Holders: Three, spring-loaded, rubber hat, cam type.
  - 4. Location: Provide one above each mop sink.
  - 5. Material and Finish: Stainless steel, ASTM A480/A480M No. 4 finish (satin).
    - a. Shelf: Not less than nominal 0.05-inch- thick stainless steel.
    - b. Rod: Approximately 1/4-inch- diameter stainless steel.

#### 2.5 FABRICATION

A. Keys: Provide universal keys for internal access to accessories for servicing and resupplying. Provide minimum of six keys to Owner's representative.

#### **PART 3 - EXECUTION**

#### 3.1 INSTALLATION

- A. Install accessories according to manufacturers' written instructions, using fasteners appropriate to substrate indicated and recommended by unit manufacturer. Install units level, plumb, and firmly anchored in locations and at heights indicated.
  - 1. Remove temporary labels and protective coatings.
- B. Grab Bars: Install to comply with specified structural-performance requirements.

#### **END OF SECTION 10 28 00**

# SECTION 10 44 13 - FIRE PROTECTION CABINETS

## PART 1 - GENERAL

#### 1.1 SUMMARY

- A. Section Includes:
  - 1. Fire-protection cabinets for portable fire extinguishers.

## 1.2 PREINSTALLATION CONFERENCE

A. Preinstallation Conference: Conduct conference at Project site.

## 1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings: For fire-protection cabinets.
- C. Samples: For each type of exposed finish required.

#### 1.4 CLOSEOUT SUBMITTALS

A. Maintenance data.

#### 1.5 COORDINATION

- A. Coordinate size of fire-protection cabinets to ensure that type and capacity of fire extinguishers indicated are accommodated.
- B. Coordinate sizes and locations of fire-protection cabinets with wall depths.

#### PART 2 - PRODUCTS

## 2.1 PERFORMANCE REQUIREMENTS

A. Fire-Rated Fire-Protection Cabinets: Listed and labeled to comply with requirements in ASTM E814 for fire-resistance rating of walls where they are installed.

# 2.2 FIRE-PROTECTION CABINET < Insert drawing designation>

- A. Cabinet Type: Suitable for fire extinguisher.
  - 1. <u>Manufacturers:</u> Subject to compliance with requirements, provide products by one of the following:
    - a. Fire-End & Croker Corporation.
    - b. Guardian Fire Equipment, Inc.
    - c. <u>Larsens Manufacturing Company</u>.
    - d. Modern Metal Products, Division of Technico Inc.
    - e. <u>Nystrom</u>.
    - f. Strike First Corporation of America (The).
- B. Cabinet Construction: Nonrated.
  - 1. Fire-Rated Cabinets: Construct fire-rated cabinets with double walls fabricated from 0.043-inch- (1.09-mm-) thick cold-rolled steel sheet lined with minimum 5/8-inch- (16-mm-) thick fire-barrier material. Provide factory-drilled mounting holes.
- C. Cabinet Material: Stainless steel sheet.
- D. Semirecessed Cabinet: One-piece combination trim and perimeter door frame overlapping surrounding wall surface, with exposed trim face and wall return at outer edge (backbend).
  - 1. Rolled-Edge Trim: 2-1/2-inch (64-mm) backbend depth.
- E. Cabinet Trim Material: Stainless steel sheet.
- F. Door Material: Stainless steel sheet.
- G. Door Style: Center glass panel with frame.
- H. Door Glazing: Clear float glass.
- I. Door Hardware: Manufacturer's standard door-operating hardware of proper type for cabinet type, trim style, and door material and style indicated.
- J. Accessories:
  - 1. Mounting Bracket: Manufacturer's standard steel, designed to secure fire extinguisher to fire-protection cabinet, of sizes required for types and capacities of fire extinguishers indicated, with plated or baked-enamel finish.
  - 2. Lettered Door Handle: One-piece, cast-iron door handle with the word "FIRE" embossed into face.
  - 3. Door Lock: Cam lock that allows door to be opened during emergency by pulling sharply on door handle.
  - 4. Identification: Lettering complying with authorities having jurisdiction for letter style, size, spacing, and location. Locate as indicated.
    - a. Identify fire extinguisher in fire-protection cabinet with the words "FIRE EXTINGUISHER."

- 1) Location: Applied to cabinet door.
- 2) Application Process: Pressure-sensitive vinyl letters.
- 3) Lettering Color: Black.
- 4) Orientation: Vertical.

#### K. Materials:

- 1. Stainless Steel: ASTM A240/A240M or ASTM A666, Type 304.
  - a. Finish: ASTM A480/A480M No. 4 directional satin finish,.
- 2. Clear Float Glass: ASTM C1036, Type I, Class 1, Quality q3, 3 mm thick.

#### 2.3 FABRICATION

A. Fire-Protection Cabinets: Provide manufacturer's standard box (tub) with trim, frame, door, and hardware to suit cabinet type, trim style, and door style indicated.

## PART 3 - EXECUTION

#### 3.1 INSTALLATION

- A. Prepare recesses for semirecessed fire-protection cabinets as required by type and size of cabinet and trim style.
- B. Install fire-protection cabinets in locations and at mounting heights indicated or, if not indicated, at heights acceptable to authorities having jurisdiction.
- C. Fire-Protection Cabinets: Fasten cabinets to structure, square and plumb.
- D. Identification: Apply vinyl lettering at locations indicated.
- E. Adjust fire-protection cabinet doors to operate easily without binding. Verify that integral locking devices operate properly.

#### END OF SECTION 10 44 13